## **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

## FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 16, 2021

# WM TECHNOLOGY, INC.

(Exact name of registrant as specified in its charter)

Delaware

(State or Other Jurisdiction of Incorporation)

001-39021 (Commission File Number)

98-1605615 (I.R.S. Employer Identification No.)

41 Discovery

Irvine, California

(Address of principal executive offices)

92618

(Zip Code)

(844) 933-3627 (Registrant's telephone number, including area code) Silver Spike Acquisition Corp. 660 Madison Ave., Suite 1600 New York, New York 10065

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligations of the registrant under any of the following provisions:

Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240-13e-4(c))

## Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading symbol(s)	Name of each exchange on which registered
Class A Common Stock, \$0.0001 par value per	MAPS	The Nasdaq Global Select Market
share		
Warrants, each whole warrant exercisable for one	MAPSW	The Nasdaq Global Select Market
share of Class A Common Stock at an exercise		
price of \$11.50 per share		

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b–2 of the Securities Exchange Act of 1934 (§240.12b–2 of this chapter).

Emerging growth company  $\boxtimes$ 

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.  $\Box$ 

#### **INTRODUCTORY NOTE**

Due to the large number of events reported under the specified items of Form 8-K, this Current Report on Form 8-K is being filed in two parts. An amendment to this Form 8-K is being submitted for filing on the same date to include additional matters under Items 3.03, 5.03, 5.05 and 8.01 of Form 8-K.

On June 16, 2021 (the "Closing Date"), WM Technology, Inc., a Delaware corporation (the "Company" or "WM Technology") (f/k/a Silver Spike Acquisition Corp. ("Silver Spike")), consummated the previously announced merger (the "Closing") pursuant to that certain Agreement and Plan of Merger, dated December 10, 2020 (the "Merger Agreement"), by and among Silver Spike, Silver Spike Merger Sub LLC, a Delaware limited liability company and a wholly owned direct subsidiary of Silver Spike ("Merger Sub"), WM Holding Company, LLC, a Delaware limited liability company (when referred to in its pre-Business Combination (as defined below) capacity, "Legacy WMH"), and Ghost Media Group, LLC, a Nevada limited liability company, solely in its capacity as the initial holder representative (the "Holder Representative"). The Company's shareholders approved the business combination (the "Business Combination") and the change of Silver Spike's jurisdiction of incorporation from the Cayman Islands to the State of Delaware by deregistering as an exempted company in the Cayman Islands and domesticating and continuing as a corporation formed under the laws of the State of Delaware (the "Domestication") at a special meeting of stockholders held on June 10, 2021 (the "Second Special Meeting"). Prior to the Second Special Meeting, the Company's shareholders approved an amendment to the Amended and Restated Memorandum and Articles of Association of the Company to extend the date by which, if the Company has not consummated a merger, amalgamation, share exchange, asset acquisition, share purchase, reorganization or similar business combination with one or more businesses, the Company must (i) cease all operations except for the purpose of winding up, (ii) redeem all of the Company's Class A ordinary shares included as part of the units sold in the Company's initial public offering that was consummated on August 19, 2019 and (iii) liquidate and dissolve, from February 12, 2021 to July 10, 2021 at a special meeting of stockholders held on January 13, 2021 (the "First Special Meeting"). In connection with the First Special Meeting, the Second Special Meeting and the Business Combination, holders of 11,437 Silver Spike's Class A ordinary shares ("Class A Ordinary Shares"), or 0.05% of the shares with redemption rights, exercised their right to redeem their shares for cash at a redemption price of approximately \$10.17 per share, for an aggregate redemption amount of \$116,314.29.

Pursuant to the terms of the Merger Agreement, the Business Combination was effected through the merger (the "Merger") of Merger Sub with and into Legacy WMH, with Legacy WMH surviving as a subsidiary of Silver Spike (when referred to in its post-Business Combination capacity, "WMH LLC"). On the Closing Date, the Company changed its name from Silver Spike Acquisition Corp. to WM Technology, Inc.

Upon the consummation of the Domestication, each of Silver Spike's currently issued and outstanding Class A Ordinary Shares and Class B ordinary shares ("Class B Ordinary Shares") automatically converted by operation of law, on a one-for-one basis, into shares of Class A common stock of the Company ("Class A Common Stock"). Similarly, all of Silver Spike's outstanding warrants became warrants to acquire shares of Class A Common Stock, and no other changes were made to the terms of any outstanding warrants.

The Business Combination was accomplished through what is commonly referred to as an "Up-C" structure, which is often used by partnerships and limited liability companies undertaking an initial public offering. The Up-C structure allowed the Legacy WMH equity holders to retain their equity ownership in WMH LLC, an entity that is classified as a partnership for U.S. federal income tax purposes, in the form of WMH LLC Class A units ("Class A Units") and WMH LLC Class P units ("Class P Units") and provides potential future tax benefits for both the Company and the WMH LLC equity holders when they ultimately exchange their pass-through interests for shares of Class A Common Stock. The Company is a holding company, and immediately after the consummation of the Business Combination, its principal asset is its ownership interest and managing member interest in WMH LLC.

A description of the Business Combination and the terms of the Merger Agreement are included in the final prospectus and definitive proxy statement, dated May 25, 2021 (the "Proxy Statement/Prospectus") filed by the Company with the Securities and Exchange Commission (the "SEC") in the section entitled "*Proposal No. 1—The Business Combination Proposal*" beginning on page 160 of the Proxy Statement/Prospectus.

The foregoing description of the Merger Agreement is a summary only and is qualified in its entirety by the full text of the Merger Agreement, a copy of which is attached hereto as Exhibit 2.1, which is incorporated herein by reference.

On December 10, 2020, a number of purchasers subscribed to purchase from the Company an aggregate of 32,500,000 shares of the Company's Class A Common Stock (the "PIPE Shares"), for a purchase price of \$10.00 per share and an aggregate purchase price of \$325,000,000, pursuant to separate subscription agreements (each, a "Subscription Agreement"). The sale of PIPE Shares was consummated concurrently with the Closing. A description of the Subscription Agreements is included in the Proxy Statement/Prospectus in the section entitled "*The Business Combination—Related Agreements—PIPE Subscription Agreement*" beginning on page 125 of the Proxy Statement/Prospectus.

The foregoing description of the Subscription Agreements is a summary only and is qualified in its entirety by the full text of the Form of Subscription Agreement, a copy of the form of which is attached hereto as Exhibit 10.4, which is incorporated herein by reference.

#### Item 1.01. Entry into a Material Definitive Agreement.

#### **Exchange Agreement**

The Company, WMH LLC and the WMH LLC equity holders entered into an exchange agreement (the "Exchange Agreement") at the time of the Business Combination. The terms of the Exchange Agreement provide the WMH LLC equity holders (or certain permitted transferees thereof) with the right from time to time at and after 180 days following the Business Combination to exchange their Class A Units (together with an equal number of shares of Class V common stock of the Company (the "Class V Common Stock" and together with the Class A Common Stock, the "Common Stock")) for shares of Class A Common Stock on a one-for-one basis, subject to customary conversion rate adjustments for stock splits, stock dividends and reclassifications, and to exchange their vested Class P Units for shares of Class A Common Stock with a value equal to the value of such Class A Common Stock. A description of the Exchange Agreement is included in the Proxy Statement/Prospectus in the section entitled "*The Business Combination—Related Agreements—Exchange Agreement*" beginning on page 122 of the Proxy Statement/Prospectus.

The foregoing description of the Exchange Agreement is qualified in its entirety by the full text of the form of Exchange Agreement, a copy of which is attached hereto as Exhibit 4.4 and incorporated herein by reference.

#### Amended and Restated Registration Rights Agreement

In connection with the Closing, Silver Spike Sponsor, LLC ("Silver Spike Sponsor") and certain WMH LLC equity holders entered into an amended and restated registration rights agreement (the "Amended and Restated Registration Rights Agreement"). The terms of the Amended & Restated Registration Rights Agreement are described in the Proxy Statement/Prospectus in the section entitled "Business Combination—Related Agreements—Amended and Restated Registration Rights Agreement" beginning on page 123 of the Proxy Statement/Prospectus.

The foregoing description of the Amended and Restated Registration Rights Agreement is qualified in its entirety by the full text of the Amended and Restated Registration Rights Agreement, a copy of which is attached hereto as Exhibit 4.2 and incorporated herein by reference.

#### **Tax Receivable Agreement**

Upon the consummation of the Business Combination, the Company, the Holder Representative and the Class A Unit holders entered into the tax receivable agreement, (the "Tax Receivable Agreement"). The terms of the Tax Receivable Agreement are described in the Proxy Statement/Prospectus in the section entitled "*Business Combination—Related Agreements— Tax Receivable Agreement*" beginning on page 124 of the Proxy Statement/Prospectus.

The foregoing description of the Tax Receivable Agreement is qualified in its entirety by the full text of the Tax Receivable Agreement, a copy of which is attached hereto as Exhibit 4.5 and incorporated herein by reference.

#### **Amended Operating Agreement**

Concurrently with the Closing, the Legacy WMH operating agreement was amended and restated in its entirety to become the Fourth Amended and Restated Operating Agreement (the "Amended Operating Agreement"). The terms of the Amended Operating Agreement are described in the Proxy Statement/Prospectus in the section entitled "*Business Combination—Related Agreements— Amended Operating Agreement*" beginning on page 125 of the Proxy Statement/Prospectus.

The foregoing description of the Amended Operating Agreement is qualified in its entirety by the full text of the Amended Operating Agreement, a copy of which is attached hereto as Exhibit 4.6 and incorporated herein by reference.

#### Item 2.01. Completion of Acquisition or Disposition of Assets.

The disclosure set forth in the "Introductory Note" above is incorporated by reference into this Item 2.01.

#### FORM 10 INFORMATION

Prior to the Closing, the Company was a shell company (as defined in Rule 12b-2 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) with no operations, formed as a vehicle to effect a business combination with one or more operating businesses. After the Closing, the Company became a holding company whose only assets consist of equity interests in WMH LLC. Item 2.01(f) of Form 8-K states that if the registrant was a shell company, as the Company was immediately before the Merger, then the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10. Accordingly, the Company is providing below the information that would be included in a Form 10 if it were to file a Form 10. Please note that the information provided below relates to the combined company after the consummation of the Business Combination, unless otherwise specifically indicated or the context otherwise requires.

#### **Cautionary Note Regarding Forward-Looking Statements**

This Current Report on Form 8-K and in documents incorporated herein by reference may contain "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act") and Section 21E of the Exchange Act. The Company's forward-looking statements include, but are not limited to, statements regarding the Company's or its management team's expectations, hopes, beliefs, intentions or strategies regarding the future. Statements that are not historical facts. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "continue," "could," "estimate," "expect," "intends," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking.

The forward-looking statements contained in this Current Report on Form 8-K and in documents incorporated herein are based on the Company's current expectations and beliefs concerning future developments and their potential effects on us taking into account information currently available to us. There can be no assurance that future developments affecting us will be those that we have anticipated. These forward-looking statements involve a number of risks, uncertainties (many of which are difficult to predict and beyond the Company's control) or other assumptions that may cause actual results or performance to be materially different from those expressed or implied by these forward-looking statements.



As a result of a number of known and unknown risks and uncertainties, the Company's actual results or performance may be materially different from those expressed or implied by these forward-looking statements. Some factors that could cause actual results to differ include:

- changes in domestic and foreign business, market, financial, political and legal conditions;
- the Company's ability to recognize the anticipated benefits of the Business Combination, which may be affected by, among other things, competition and the ability of the Company to grow and manage growth profitably following the Closing;
- costs related to the Business Combination;
- changes in applicable laws or regulations;
- the outcome of any legal proceedings against the Company;
- the financial and business performance of the Company, including financial projections and business metrics and any underlying assumptions thereunder;
- future global, regional or local economic and market conditions affecting the cannabis industry;
- the development, effects and enforcement of laws and regulations, including with respect to the cannabis industry;
- the Company's ability to successfully capitalize on new and existing cannabis markets, including its ability to successfully monetize its solutions in those markets;
- the Company's ability to develop new products and solutions, bring them to market in a timely manner, and make enhancements to its platform and the Company's ability to maintain and grow its two sided digital network, including its ability to acquire and retain paying customers; the effects of competition on the Company's future business;
- changes in the Company's strategy, future operations, financial position, estimated revenues and losses, projected costs, prospects and plans;
- developments and projections relating to the Company's competitors and industry;
- the impact of health epidemics, including the COVID-19 pandemic, on the Company's business and the actions the Company may take in response thereto;
- expectations regarding the time during which we will be an emerging growth company under the Jumpstart Our Business Startups Act of 2012, as amended;
- the Company's future capital requirements and sources and uses of cash;
- the Company's ability to obtain funding for its future operations;
- the Company's business, expansion plans and opportunities;
- the outcome of any known and unknown litigation and regulatory proceedings; and
- other risks and uncertainties set forth in the Proxy Statement/Prospectus in the section entitled "Risk Factors" beginning on page 55 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

These forward-looking statements are based on information available as of the date of this Current Report on Form 8-K, and current expectations, forecasts and assumptions, and involve a number of risks and uncertainties. Accordingly, forward-looking statements in this Current Report on Form 8-K and in any document incorporated herein by reference should not be relied upon as representing the Company's views as of any subsequent date, and we do not undertake any obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

<sup>4</sup> 

#### **Business and Properties**

The business and properties of Silver Spike and Legacy WMH prior to the Business Combination are described in the Proxy Statement/Prospectus in the sections entitled *"Business of Silver Spike"* beginning on page 199 and *"Business of WMH"* beginning on page 201 of the Proxy Statement/Prospectus, which are incorporated herein by reference.

#### **Risk Factors**

The risks associated with the Company's business are described in the Proxy Statement/Prospectus in the section entitled *"Risk Factors"* beginning on page 55 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### **Selected Historical Financial Information**

The selected historical consolidated financial information and other data for the three months ended March 31, 2021 and 2020 and the years ended December 31, 2020 and 2019 for Legacy WMH is included in the Proxy Statement/Prospectus in the section entitled *"Selected Historical Consolidated Financial and Operating Data of WMH"* beginning on page 49 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### **Unaudited Condensed Consolidated Financial Statements**

The unaudited condensed consolidated financial statements as of and for the three months ended March 31, 2021 of Legacy WMH have been prepared in accordance with U.S. generally accepted accounting principles and pursuant to the regulations of the SEC and are included in the Proxy Statement/Prospectus beginning on page FS-18 of the Proxy Statement/Prospectus, which are incorporated herein by reference.

These unaudited condensed consolidated financial statements should be read in conjunction with the historical audited financial statements of Legacy WMH as of and for the year ended December 31, 2020 and the related notes included in the Proxy Statement/Prospectus beginning on page F-3 of the Proxy Statement/Prospectus, which are incorporated herein by reference.

## **Unaudited Pro Forma Condensed Combined Financial Information**

The unaudited pro forma condensed combined financial information of the Company as of and for the three months ended March 31, 2021 and for the year ended December 31, 2020 is set forth in Exhibit 99.1 hereto and is incorporated herein by reference.

## Management's Discussion and Analysis of Financial Condition and Results of Operations

The Management's Discussion and Analysis of Financial Condition and Results of Operations of Legacy WMH prior to the Business Combination are described in the Proxy Statement/Prospectus in the section entitled "*Management's Discussion and Analysis of Financial Condition and Results of Operations of WMH*" beginning on page 218 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

## **Directors and Executive Officers**

Information with respect to the Company's directors and executive officers after the Closing is set forth in the Proxy Statement/Prospectus in the sections entitled "*Management After the Business Combination*" and "*Executive Compensation*" beginning on page 243 and page 250, respectively, of the Proxy Statement/Prospectus, which are incorporated herein by reference.

#### Directors

Upon the Closing, the size of the board of directors of the Company (the "Board") was set at nine members. Upon the Closing, each of William Healy, Orrin Devinsky, Richard M. Goldman and Kenneth H. Landis resigned as directors of the Company, and each of Tony Aquila, Christopher Beals, Douglas Francis, Brenda Freeman, Olga Gonzalez, Scott Gordon, Justin Hartfield and Fiona Tan were elected to serve as directors on the Board. In addition, there was one vacancy on the Board upon the Closing.

Mr. Beals and Ms. Tan were appointed to serve as Class I directors, with terms expiring at the Company's 2021 annual meeting of stockholders; Mr. Aquila, Mses. Freeman and Gonzalez were appointed to serve as Class II directors, with terms expiring at the Company's 2022 annual meeting of stockholders; and Messrs. Francis, Gordon and Hartfield were appointed to serve as Class III directors, with terms expiring at the Company's 2023 annual meeting of stockholders; Biographical information for these individuals is set forth in the Proxy Statement/Prospectus in the section entitled "*Management After the Business Combination—Executive Officers and Directors*" beginning on page 243 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### **Independence of Directors**

The Board has determined that each of the directors of the Company other than Messrs. Beals and Francis qualify as independent directors, as defined under the listing rules of The Nasdaq Stock Market LLC (the "Nasdaq listing rules"), and that the Board consists of a majority of "independent directors," as defined under the rules of the SEC and Nasdaq listing rules relating to director independence requirements.

#### **Committees of the Board of Directors**

Effective upon the consummation of the Business Combination, the standing committees of the Board consist of an audit committee (the "Audit Committee"), a compensation committee (the "Compensation Committee") and a nominating and corporate governance committee (the "Nominating and Corporate Governance Committee"). Each of the committees reports to the Board.

Effective upon the consummation of the Business Combination, the Board appointed Mses. Gonzalez, Freeman and Mr. Gordon to serve on the Audit Committee, with Ms. Gonzalez as chair of the Audit Committee. The Board appointed Messrs. Aquila, Gordon and Francis to serve on the Compensation Committee, with Messr. Aquila as chair of the Compensation Committee. The Board appointed Mses. Tan and Freeman to serve on the Nominating and Corporate Governance Committee, with Ms. Freeman as chair of the Nominating and Corporate Governance Committee.

#### **Executive Officers**

Effective upon the consummation of the Business Combination, the Board appointed the following individuals as the Company's executive officers: Christopher Beals to serve as Chief Executive Officer, Steven Jung to serve as Chief Operations Officer, Arden Lee to serve as Chief Financial Officer, Juanjo Feijoo to serve as Chief Marketing Officer, Justin Dean to serve as Chief Technology Officer and Brian Camire to serve as General Counsel. Effective upon the consummation of the Business Combination, each of Scott Gordon, William Healy and Gregory M. Gentile resigned as the Chief Executive Officer, President and Chief Financial Officer, respectively. The biographical information for the new executive officers set forth in the Proxy Statement/Prospectus in the section entitled "*Management After the Business Combination*" beginning on page 243 of the Proxy Statement/Prospectus, is incorporated herein by reference.

#### **Director Compensation**

Information with respect to the compensation of the Company's directors is set forth in the Proxy Statement/Prospectus in the sections entitled *"Executive Compensation—Director Compensation"* beginning on page 253 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

## **Executive Compensation**

Information with respect to the compensation of Company's executive officers is set forth in the Proxy Statement in the section entitled "Executive Compensation" beginning on page 250 of the Proxy Statement, which is incorporated herein by reference. The foregoing description of the compensation of the Company's executive officers is qualified in its entirety by the full text of the offer letters of Messrs. Beals, Dean and Jung, copies of which are attached hereto as Exhibit 10.9, Exhibit 10.10 and Exhibit 10.11, respectively, and incorporated herein by reference.

#### Strategic Advisor Agreement

In connection with the Closing, Mr. Jung entered into a Strategic Advisor Agreement (the "Strategic Advisor Agreement") with Ghost Management Group, LLC ("Ghost Management"), an indirect subsidiary of the Company. Pursuant to the Strategic Advisor Agreement, Mr. Jung's last day of active employment in his current role as President and Chief Operating Officer of the Company and Ghost Management will be June 30, 2021, and he will transition to a paid employee advisory role from July 1, 2021 through July 6, 2022 (the "Advisory Services Period"). The Strategic Advisor Agreement provides for Mr. Jung to be paid his regular salary during the Advisory Services Period and the continued vesting of the Class P units of WMH LLC held by Mr. Jung during the Advisory Services Period, each subject to a maximum aggregate amount, and a general release of claims in favor of Ghost Management.

The foregoing description of the Strategic Advisor Agreement is qualified in its entirety by the full text of the Strategic Advisor Agreement, a copy of which is attached hereto as Exhibit 10.17 and incorporated herein by reference.

#### **Beneficial Ownership of Securities**

The following table sets forth information known to the Company regarding (i) the actual beneficial ownership of the Company's Common Stock as of the Closing Date, after giving effect to the Closing, by:

- each person known by the Company to be the beneficial owner of more than 5% of the Company's outstanding shares of Common Stock;
- each of the Company's named executive officers and directors;
- all executive officers and directors of the Company, as a group.

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options and warrants that are currently exercisable or exercisable within 60 days.

The beneficial ownership percentages set forth in the table below are based on 129,240,912 shares of the Company's Common Stock issued and outstanding as of the Closing Date and do not take into account the issuance of any shares of Class A Common Stock upon (i) the exercise of 19,500,000 warrants, each exercisable for one share of Class A Common Stock at a price of \$11.50 per share (the "Warrants") to purchase an aggregate of 19,500,000 shares of Class A Common Stock or (ii) the exchange of Class P Units. Unless otherwise noted in the footnotes to the following table, and subject to applicable community property laws, the persons and entities named in the table have sole voting and investment power with respect to their beneficially owned Class A Common Stock.

Name of Beneficial Owner <sup>(1)</sup>	Number of Shares of Class A Common Stock Beneficially Owned	Number of Shares of Class V Common Stock Beneficially Owned	Combined Total Voting Power
Directors and Named Executive Officers:		C 1CC 010	4.00/
Christopher Beals	—	6,166,819	4.8%
Justin Dean		_	_
Steven Jung	-	—	—
Tony Aquila <sup>(2)</sup>	5,000,000		3.9%
Douglas Francis <sup>(3)</sup>	—	27,700,849	21.4%
Justin Hartfield <sup>(4)</sup>		29,328,310	22.7%
Scott Gordon	—	—	—
Fiona Tan	—	—	—
Olga Gonzalez		_	
Brenda Freeman	_	_	_
All Directors and Executive Officers of the Company as a Group (10 Individuals)	5,000,000	54,726,787	46.2%
Five Percent Holders:			
Silver Spike Sponsor, LLC <sup>(5)</sup>	9,750,000	_	7.5%
Ghost Media Group, LLC <sup>(3)(4)</sup>	—	8,469,191	6.6%

- (1) Unless otherwise noted, the business address of each of the following entities or individuals is 41 Discovery, Irvine, California 92618.
- (2) Includes 5,000,000 shares in the aggregate of shares of Class A Common Stock held by AFV Partners SPV-5 LLC ("AFV 5"), AFV Partners SPV-6 LLC ("AFV 6") and a controlled affiliated entity of Tony Aquila upon the completion of the business combination pursuant to the PIPE subscription financing. Mr. Aquila is the Chairman and CEO of AFV Partners LLC, which exercises ultimate voting and investment power with respect to the shares held by AFV 5 and AFV 6. Furthermore, Mr. Aquila will personally hold a portion of the shares of Class A Common Stock and will be the sole member with ultimate voting and investment power with respect to the shares held by the controlled entity to be formed to hold the shares of Class A Common Stock. As such, Mr. Aquila may be deemed to be a beneficial owner of the shares held by AFV 5, AFV 6 and the controlled affiliated entity. The business address of the reporting person is 2126 Hamilton Road Suite 260, Argyle, TX 76226.
- (3) Includes 17,162,485 shares of Class V Common Stock held by Mr. Francis, 8,469,191 shares of Class V Common Stock held by Ghost Media Group, LLC, 600,618 shares of Class V Common Stock held by Genco Incentives, LLC and 1,468,555 shares of Class V Common Stock held by WM Founders Legacy I, LLC. Ghost Media Group, LLC is controlled by Messrs. Francis and Hartfield and WM Founders Legacy I, LLC and Genco Incentives, LLC are controlled by Mr. Francis. Accordingly, Mr. Francis may be deemed to be a beneficial owner of the Class A Units held by Ghost Media Group, LLC, Genco Incentives, LLC and WM Founders Legacy I, LLC.
- (4) Includes 19,288,160 shares of Class V Common Stock held by Mr. Hartfield, 8,469,191 shares of Class V Common Stock held by Ghost Media Group, LLC and 1,570,959 shares of Class V Common Stock held by WM Founders Legacy II, LLC. Ghost Media Group, LLC is controlled by Messrs. Hartfield and Francis and WM Founders Legacy II, LLC is controlled by Mr. Hartfield. Accordingly, Mr. Hartfield may be deemed to be a beneficial owner of the shares held by Ghost Media Group, LLC and WM Founders Legacy II, LLC.
- (5) Includes 6,250,000 shares of Class A common stock held by Silver Spike Sponsor, LLC and 3,500,000 shares of Class A common stock held by Silver Spike Opportunities I, LLC (the "SPV"). Silver Spike Sponsor, LLC and Silver Spike Capital, LLC, the manager of the SPV, are both controlled by Silver Spike Holdings, LP. Accordingly, Silver Spike Holdings, LP may be deemed to be a beneficial owner of the shares held by Silver Spike Sponsor, LLC and the shares held by the SPV.

#### **Certain Relationships and Related Transactions**

The certain relationships and related party transactions of the Company are described in the Proxy Statement/Prospectus in the section entitled "*Certain WMH Relationships and Related Party Transactions*" beginning on page 259 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### **Legal Proceedings**

Information about legal proceedings is set forth in the Proxy Statement/Prospectus in the section "Annex I—Silver Spike's Annual Report on Form 10-K/A, filed with the SEC on May 12, 2021" on page I-52 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

#### Market Information and Holders

Prior to the Business Combination, the Class A Common Stock and the Warrants were historically quoted on The Nasdaq Stock Market under the symbols "SSPKU" and "SSPKW," respectively. On June 16, 2021, the Class A Common Stock and Warrants began trading on The Nasdaq Global Select Market under the new trading symbols "MAPS" and "MAPSW," respectively.

As of the Closing Date and following the consummation of the Business Combination, the Company had 63,738,563 shares of Class A Common Stock issued and outstanding held of record by 70 holders, 65,502,347 shares of Class V Common Stock issued and outstanding held of record by 23 holders and 19,499,993 Warrants outstanding held of record by two holders. No market exists for the Class V Common Stock.

On June 16, 2021, in connection with the Closing, all of the units previously issued by Silver Spike separated into their component parts of one share of Class A Common Stock and one half (1/2) of one Warrant to purchase one share of Class A Common Stock, and the units ceased trading on The Nasdaq Capital Market.

#### Dividends

The Company has not paid any cash dividends on the Class A Ordinary Shares, Class A Common Stock or Class V Common Stock to date. The payment of cash dividends in the future will be dependent upon the Company's revenues and earnings, if any, capital requirements, general financial condition, contractual restrictions and other factors that the Board may deem relevant and will be within the discretion of the Board at such time. In addition, the Company's ability to pay dividends may be limited by covenants of any existing and future outstanding indebtedness the Company or its subsidiaries incur. The Company does not anticipate declaring any cash dividends to holders of Common Stock in the foreseeable future.

#### **Recent Sales of Unregistered Securities**

Reference is made to the disclosure set forth below under Item 3.02 of this Current Report on Form 8-K concerning the issuance and sale by the Company of certain unregistered securities, which is incorporated herein by reference.

#### Description of Registrant's Securities to be Registered

#### **Class A Common Stock**

A description of the Class A Common Stock is included in the Proxy Statement/Prospectus in the section entitled "*Description of Securities—Authorized and Outstanding Stock—Class A Common Stock*" beginning on page 261 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### Warrants

A description of the Company's warrants is included in the Proxy Statement/Prospectus in the section entitled "*Description of Securities—Warrants*" beginning on page 263 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

## **Indemnification of Directors and Officers**

Information about indemnification of the Company's directors and officers is set forth in the Proxy Statement/Prospectus in the section entitled "*Management After the Business Combination—Limitation of Liability and Indemnification*" beginning on page 248 of the Proxy Statement/Prospectus, which is incorporated herein by reference. The disclosure set forth in Item 1.01 of this Current Report on Form 8-K under the section entitled "*Indemnification Agreements*" is incorporated by reference into this Item 2.01.

#### **Financial Statements and Supplementary Data**

The information set forth in Item 9.01 of this Current Report on Form 8-K is incorporated herein by reference.

#### Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

Reference is made to the disclosure set forth under Item 4.01 of this Report relating to the changes in certifying accountant.

#### **Financial Statements and Exhibits**

The information set forth in Item 9.01 of this Current Report on Form 8-K is incorporated herein by reference.

#### Item 3.02. Unregistered Sales of Equity Securities.

The disclosure set forth in the "Introductory Note" above is incorporated by reference into this Item 3.02.

The securities issued in connection with the Business Combination Agreement and Subscription Agreements have not been registered under the Securities Act in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act.

The issuance of Class A Common Stock upon automatic conversion of Class B ordinary shares at the Closing and the issuance of Class V Common Stock to WMH LLC equity holders have not been registered under the Securities Act in reliance on the exemption from registration provided by Rule 506(b) of Regulation D under the Securities Act.

#### Item 4.01 Changes in Registrant's Certifying Accountant.

On June 16, 2021, the Board approved the engagement of Baker Tilly US, LLP ("Baker Tilly") as the Company's independent registered public accounting firm to audit the Company's consolidated financial statements for the year ending December 31, 2021. Baker Tilly served as the independent registered public accounting firm of Legacy WMH prior to the Business Combination. Accordingly, Marcum LLP ("Marcum"), the Company's independent registered public accounting firm prior to the Business Combination, was informed on the Closing Date that it would be dismissed and replaced by Baker Tilly as the Company's independent registered public accounting firm.

Marcum's report on the Company's balance sheets as of December 31, 2020 and 2019, the related statements of operations, stockholders' equity and cash flows for the year ended December 31, 2020 and for the period from June 7, 2019 (inception) to December 31, 2019, and the related notes to the financial statements (collectively, the "financial statements") did not contain any adverse opinion or disclaimer of opinion, nor were they qualified or modified as to uncertainty, audit scope or accounting principles, except for the substantial doubt about the Company's ability to continue as a going concern.

During the period from June 7, 2019 (inception) to December 31, 2020 and the subsequent interim period through June 21, 2021, there were no: (i) disagreements with Marcum on any matter of accounting principles or practices, financial statement disclosures or audited scope or procedures, which disagreements if not resolved to Marcum's satisfaction would have caused Marcum to make reference to the subject matter of the disagreement in connection with its report or (ii) reportable events as defined in Item 304(a)(1)(v) of Regulation S-K under the Exchange Act.

During the period from June 7, 2019 (inception) to December 31, 2020, and the interim period through March 31, 2021, the Company did not consult Baker Tilly with respect to either (i) the application of accounting principles to a specified transaction, either completed or proposed; or the type of audit opinion that might be rendered on the Company's financial statements, and no written report or oral advice was provided to the Company by Baker Tilly that Baker Tilly concluded was an important factor considered by the Company in reaching a decision as to the accounting, auditing or financial reporting issue; or (ii) any matter that was either the subject of a disagreement, as that term is described in Item 304(a)(1)(iv) of Regulation S-K under the Exchange Act and the related instructions to Item 304 of Regulation S-K under the Exchange Act, or a reportable event, as that term is defined in Item 304(a)(1)(v) of Regulation S-K under the Exchange Act.

The Company has provided Marcum with a copy of the disclosures made by the Company in response to this Item 4.01 and has requested that Marcum furnish the Company with a letter addressed to the SEC stating whether it agrees with the statements made by the Company in response to this Item 4.01 and, if not, stating the respects in which it does not agree. A letter from Marcum is attached hereto as Exhibit 16.1.

#### Item 5.01. Changes in Control of the Registrant.

The information set forth in the section entitled "*Introductory Note*" and in the section entitled "*Security Ownership of Certain Beneficial Owners and Management*" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

As a result of the consummation of the Business Combination , a change of control of Silver Spike has occurred, and the stockholders of Silver Spike as of immediately prior to the Closing held 49.3% of the outstanding voting power and shares of Common Stock immediately following the Closing.

## Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information set forth in the sections entitled "*Directors and Executive Officers*" and "*Certain Relationships and Related Transactions*" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

#### WM Technology, Inc. 2021 Equity Incentive Plan

At the Second Special Meeting, the stockholders of the Company considered and approved the WM Technology, Inc. 2021 Equity Incentive Plan (the "<u>2021 Plan</u>"). The 2021 Plan was previously approved, subject to stockholder approval, by the board of directors of Silver Spike on December 9, 2020, and on the Closing Date, the Board ratified the approval of the 2021 Plan. The 2021 Plan became effective immediately upon the Closing.

A description of the 2021 Plan is included in the Proxy Statement/Prospectus in the section entitled "*Proposal No. 11—The Equity Incentive Plan Proposal*" beginning on page 174 of the Proxy Statement/Prospectus, which is incorporated herein by reference. The foregoing description of the 2021 Plan is qualified in its entirety by the full text of the 2021 Plan, which is attached hereto as Exhibit 10.7 and incorporated herein by reference.

#### WM Technology, Inc. 2021 Employee Stock Purchase Plan

At the Second Special Meeting, the stockholders of the Company considered and approved the WM Technology, Inc. 2021 Employee Stock Purchase Plan (the "<u>2021 ESPP</u>"). The 2021 ESPP was previously approved, subject to stockholder approval, by the board of directors of Silver Spike on December 9, 2020, and on the Closing Date, the Board ratified the approval of the 2021 ESPP. The 2021 ESPP became effective immediately upon the Closing.

A description of the 2021 ESPP is included in the Proxy Statement/Prospectus in the section entitled "*Proposal No. 12—The Employee Stock Purchase Plan Proposal*" beginning on page 180 of the Proxy Statement/Prospectus, which is incorporated herein by reference. The foregoing description of the 2021 ESPP is qualified in its entirety by the full text of the 2021 ESPP, which is attached hereto as Exhibit 10.8 and incorporated herein by reference.

#### Item 5.06. Change in Shell Company Status.

As a result of the Merger, which fulfilled the definition of a business combination as required by the Amended and Restated Certificate of Incorporation of the Company, as in effect immediately prior to the Closing, the Company ceased to be a shell company (as defined in Rule 12b-2 of the Exchange Act) as of the Closing. A description of the Business Combination and the terms of the Merger Agreement are included in the Proxy Statement/Prospectus in the section entitled "*Proposal No. 1—The Business Combination Proposal*" beginning on page 160 of the Proxy Statement/Prospectus, which is incorporated herein by reference.

#### Item 9.01. Financial Statements and Exhibits.

#### (a) Financial Statements of Businesses Acquired.

The unaudited condensed consolidated financial statements of Legacy WMH as of March 31, 2021 and for the three months ended March 31, 2021 and 2020 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-18 of the Proxy Statement/Prospectus and are incorporated herein by reference.

The audited consolidated financial statements of WMH as of and for the year ended December 31, 2021 and December 31, 2020 and the related notes are included in the Proxy Statement/Prospectus beginning on page F-3 of the Proxy Statement/Prospectus and are incorporated herein by reference.

The unaudited condensed consolidated financial statements of Silver Spike as of March 31, 2021 and for the three months ended March 31, 2021 and 2010 and the related notes are included in the Proxy Statement/Prospectus beginning on page J-1 of the Proxy Statement/Prospectus and are incorporated herein by reference.

The audited consolidated financial statements of Silver Spike as of and for the year ended December 31, 2020 and December 31, 2021 and the related notes are included in the Proxy Statement/Prospectus beginning on page I-82 of the Proxy Statement/Prospectus and are incorporated herein by reference.

#### (b) Pro Forma Financial Information.

The unaudited pro forma condensed combined financial information of the Company as of and for the three months ended March 31, 2021 and for the year ended December 31, 2021 is set forth in Exhibit 99.1 hereto and is incorporated herein by reference.

(d) Exhibits.

Exhibit No.	Description
2.1+	Agreement and Plan of Merger, dated December 10, 2020, by and among Silver Spike, Merger Sub, WMH, and the Holde
	Representative named therein (incorporated by reference to Exhibit 2.1 to the Current Report on Form 8-K filed on December 10, 2020)
<u>3.1</u>	Certificate of Incorporation of the Company, dated June 15, 2021.
<u>3.2</u>	Amended and Restated Bylaws of the Company, dated June 16, 2021.
<u>4.1</u>	Form of Common Stock Certificate of the Company.
<u>4.2</u>	Form of Warrant Certificate of the Company.
3.2 4.1 4.2 4.3	Warrant Agreement, dated August 7, 2019, between the Company and Continental Stock Transfer & Trust Company, as warrant ager
	(incorporated by reference to Exhibit 4.1 filed on Silver Spike's Current Report on Form 8-K, filed by the Company on August 12
	2019).
<u>10.1</u>	Exchange Agreement, dated as of June 16, 2021, by and among the Company, Silver Spike Sponsor and the other parties thereto.
<u>10.2</u>	Tax Receivable Agreement, dated as of June 16, 2021, by and among the Company and the other parties thereto.
<u>10.3</u>	Fourth Amended and Restated Operating Agreement of WMH LLC.
<u>10.4</u>	Form of Subscription Agreement (incorporated by reference to Exhibit 10.1 to the Current Report on Form 8-K filed on December 10
	2020).
<u>10.5</u>	Amended and Restated Registration Rights Agreement, dated as of June 16, 2021, by and among the Company, Silver Spike Sponsor an
	the other parties thereto.
<u>10.6#</u>	Form of Indemnification Agreement by and between the Company and its directors and officers.
<u>10.7#</u>	WM Technology, Inc. 2021 Equity Incentive Plan.
<u>10.7(a)#</u>	Form of Stock Option Grant Notice.
<u>10.7(b)#</u>	Form of RSU Award Grant Notice.
<u>10.8#</u>	WM Technology, Inc. 2021 Employee Stock Purchase Plan.
<u>10.9#</u>	Offer letter by and between Ghost Management Group, LLC and Christopher Beals, dated July 31, 2015.
<u>10.10#</u>	Offer letter by and between Ghost Management Group, LLC and Justin Dean, dated October 3, 2018.
<u>10.11#</u>	Offer letter by and between Ghost Management Group, LLC and Steven Jung, dated May 17, 2017.
<u>10.12</u>	Lease by and between the Irvine Company LLC and Ghost Media Group, LLC, dated November 11, 2013, as amended.
<u>10.13</u>	First Amendment to Lease and Consent to Assignment by and between Discovery Business Center LLC, as successor-in-interest to the
	Irvine Company LLC, and Ghost Management Group, LLC, as successor-in-interest to Ghost Media Group, LLC, dated January 2
	2016.
<u>10.14</u>	Second Amendment to Lease, by and between Discovery Business Center LLC and Ghost Management Group, LLC, dated April 2
	2017.
<u>10.15</u>	Third Amendment to Lease, by and between Discovery Business Center LLC and Ghost Management Group, LLC, dated December 29
	2017.
<u>10.16</u>	Fourth Amendment to Lease, by and between Discovery Business Center LLC and Ghost Management Group, LLC, dated May 3, 2018
<u>10.17</u>	Strategic Advisor Agreement, by and between the Company and Steven Jung, dated June 21, 2021.
<u>16.1</u>	Letter from Marcum LLP to the SEC, dated June 21, 2021.
<u>21.1</u>	List of Subsidiaries
<u>99.1</u>	Unaudited pro forma condensed combined financial information of the Company as of and for the three months ended March 31, 202 and for the year ended December 31, 2020.

<sup>+</sup> The schedules and exhibits to this agreement have been omitted pursuant to Item 601(a)(5) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished to the SEC upon request.

<sup>#</sup> Indicates management contract or compensatory plan or arrangement.

## SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: June 21, 2021

## WM TECHNOLOGY, INC.

/s/ Arden Lee

By:

Arden Lee Chief Financial Officer

## **CERTIFICATE OF INCORPORATION**

of

## WM TECHNOLOGY, INC.

1. <u>Name</u>. The name of the Corporation is WM Technology, Inc. (the "<u>Corporation</u>").

2. <u>Address; Registered Office and Agent</u>. The address of the Corporation's registered office in the State of Delaware is c/o Corporation Service Company, 251 Little Falls Drive, City of Wilmington, County of New Castle, State of Delaware 19808 and the name of its registered agent at such address is Corporation Service Company.

3. <u>Purposes</u>. The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (as from time to time in effect, the "<u>General Corporation Law</u>").

4. <u>Number of Shares</u>.

4.1 The total number of shares of all classes of stock that the Corporation shall have authority to issue is 2,075,000,000 shares, consisting of: (i) 2,000,000,000 shares of common stock, divided into (a) 1,500,000,000 shares of Class A common stock, with the par value of \$0.0001 per share (the "<u>Class A Common Stock</u>") and (b) 500,000,000 shares of Class V common stock, with the par value of \$0.0001 per share (the "<u>Class V Common Stock</u>" and, together with Class A Common Stock, the "<u>Common Stock</u>"); and (ii) 75,000,000 shares of preferred stock, with the par value of \$0.0001 per share (the "<u>Preferred Stock</u>").

4.2 Subject to the rights of the holders of any one or more series of Preferred Stock then outstanding, the number of authorized shares of any class of the Common Stock or the Preferred Stock may be increased or decreased, in each case by the affirmative vote of the holders of a majority of the total voting power of the outstanding shares of capital stock of the Corporation entitled to vote thereon, voting together as a single class, irrespective of the provisions of Section 242(b)(2) of the General Corporation Law, and no vote of the holders of any class of the Common Stock or the Preferred Stock voting separately as a class will be required therefor. Notwithstanding the immediately preceding sentence, the number of authorized shares of any particular class may not be decreased below the number of shares of such class then outstanding, plus:

(i) in the case of Class A Common Stock, the number of shares of Class A Common Stock issuable in connection with (x) the exchange of all outstanding shares of Class V Common Stock, together with the corresponding Class A LLC Units, pursuant to the Exchange Agreement (y) the exchange of all Class A LLC Units issuable to holders of Class P LLC Units upon conversion pursuant to the Exchange Agreement (assuming for this purpose that the Participation Threshold of each Class P LLC Unit is zero at the time of conversion) and (z) the exercise of outstanding options, warrants, exchange rights, conversion rights or similar rights for Class A Common Stock;

(ii) in the case of Class V Common Stock, the number of shares of Class V Common Stock issuable in connection with the exercise of outstanding options, warrants, exchange rights, conversion rights or similar rights, including upon conversion of LTIP LLC Units pursuant to the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC, for Class V Common Stock.

5. <u>Classes of Shares</u>. The designation, relative rights, preferences and limitations of the shares of each class of stock are as follows:

#### 5.1 <u>Common Stock</u>.

(i) <u>Voting Rights</u>.

(1) Each holder of Class A Common Stock will be entitled to one vote for each share of Class A Common Stock held of record by such holder on all matters on which stockholders generally are entitled to vote, and each holder of Class V Common Stock will be entitled to one vote for each share of Class V Common Stock held of record by such holder on all matters on which stockholders generally are entitled to vote, except that, in each case, to the fullest extent permitted by law and subject to Section 5.1(i)(2), holders of shares of each class of Common Stock, as such, will have no voting power with respect to, and will not be entitled to vote on, any amendment to this Certificate of Incorporation (including any certificate of designations relating to any series of Preferred Stock) that relates solely to the terms of any outstanding Preferred Stock if the holders of such Preferred Stock are entitled to vote as a separate class thereon under this Certificate of Incorporation (including any certificate of designations relating to any series of Preferred Stock) or under the General Corporation Law.

(2) (a) The holders of the outstanding shares of Class A Common Stock shall be entitled to vote separately upon any amendment to this Certificate of Incorporation (including by merger, consolidation, reorganization or similar event) that would alter or change the powers, preferences or special rights of such class of Common Stock in a manner that is disproportionately adverse as compared to the Class V Common Stock and (b) the holders of the outstanding shares of Class V Common Stock shall be entitled to vote separately upon any amendment to this Certificate of Incorporation (including by merger, consolidation, reorganization or similar event) that would alter or change the powers, preferences or special rights of such class of Common Stock in a manner that is disproportionately adverse as compared to the Class A Common Stock, it being understood that any merger, consolidation or other business combination shall not be deemed an amendment hereof if such merger, consolidation or other business combination shall not be deemed an amendment hereof if such merger, consolidation or other business combination Transaction" permitted by Section 3.07 of the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC. (3) Except as otherwise required in this Certificate of Incorporation or by applicable law, the holders of Common Stock will vote together as a single class on all matters (or, if any holders of Preferred Stock are entitled to vote together with the holders of Common Stock, as a single class with the holders of Preferred Stock).

#### (ii) <u>Dividends; Stock Splits or Combinations</u>.

(1) Subject to applicable law and the rights, if any, of the holders of any outstanding series of Preferred Stock or any class or series of stock having a preference senior to or the right to participate with the Class A Common Stock with respect to the payment of dividends, such dividends and other distributions of cash, stock or property may be declared and paid on the Class A Common Stock out of the assets of the Corporation that are by law available therefor, at the times and in the amounts as the board of directors of the Corporation (the "<u>Board</u>") in its discretion may determine.

(2) Except as provided in Section 5.1(ii)(3) with respect to stock dividends, dividends of cash or property may not be declared or paid on shares of Class V Common Stock.

(3) In no event will any stock dividend, stock split, reverse stock split, combination of stock, reclassification or recapitalization be declared or made on any class of Common Stock (each, a "<u>Stock Adjustment</u>") unless (a) a corresponding Stock Adjustment for all other classes of Common Stock not so adjusted at the time outstanding is made in the same proportion and the same manner and (b) the Stock Adjustment has been reflected in the same economically equivalent manner on all Class A LLC Units. Stock dividends with respect to each class of Common Stock may only be paid with shares of stock of the same class of Common Stock.

(iii) Liquidation. In the event of any voluntary or involuntary liquidation, dissolution or winding-up of the affairs of the Corporation, after payment or provision for payment of the debts and other liabilities of the Corporation and of the preferential and other amounts, if any, to which the holders of Preferred Stock are entitled, if any, the holders of all outstanding shares of Class A Common Stock will be entitled to receive, *pari passu*, an amount per share equal to the par value thereof, and thereafter the holders of all outstanding shares of Class A Common Stock will be entitled to receive the remaining assets of the Corporation available for distribution ratably in proportion to the number of shares of Class A Common Stock. Without limiting the rights of the holders of Class V Common Stock to exchange their shares of Class V Common Stock, together with the corresponding Class A LLC Units constituting the remainder of any Paired Interests in which such shares are included, for shares of Class A Common Stock in accordance with the Exchange Agreement (or for the consideration payable in respect of shares of Class V Common Stock in such voluntary or involuntary liquidation, dissolution or winding-up), the holders of shares of Class V Common Stock, as such, will not be entitled to receive, with respect to such shares, any assets of the Corporation in excess of the par value thereof, in the event of any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Corporation.

5.2 <u>Preferred Stock</u>. Shares of Preferred Stock may be issued from time to time in one or more series of any number of shares, <u>provided</u> that the aggregate number of shares issued and not retired of any and all such series shall not exceed the total number of shares of Preferred Stock hereinabove authorized, and with such powers, including voting powers, if any, and the designations, preferences and relative, participating, optional or other special rights, if any, and any qualifications, limitations or restrictions thereof, all as shall hereafter be stated and expressed in the resolution or resolutions providing for the designation and issue of such shares of Preferred Stock from time to time adopted by the Board pursuant to authority so to do which is hereby expressly vested in the Board. The powers, including voting powers, if any, preferences and relative, participating, optional and other special rights of each series of Preferred Stock, and the qualifications, limitations or restrictions thereof, if any, may differ from those of any and all other series at any time outstanding.

## 6. <u>Class V Common Stock</u>.

6.1 <u>Retirement of Class V Shares</u>. No holder of Class V Common Stock may transfer shares of Class V Common Stock to any person unless such holder transfers a corresponding number of Class A LLC Units to the same person in accordance with the provisions of the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC, as such agreement may be amended from time to time in accordance with the terms thereof. If any outstanding share of Class V Common Stock ceases to be held by a holder of the corresponding Class A LLC Unit, such share shall automatically and without further action on the part of the Corporation or any holder of Class V Common Stock be transferred to the Corporation for no consideration and retired.

6.2 <u>Reservation of Shares of Class A Common Stock</u>. The Corporation will at all times reserve and keep available out of its authorized and unissued shares of Class A Common Stock, solely for the purpose of the issuance in connection with (i) the exchange of Paired Interests, the number of shares of Class A Common Stock that are issuable upon conversion of all outstanding Paired Interests, pursuant to the Exchange Agreement and (ii) the exchange of Class A LLC Units issuable to holders of Class P LLC Units upon conversion pursuant to the Exchange Agreement (assuming for this purpose that the Participation Threshold of each Class P LLC Unit is zero at the time of conversion). The Corporation covenants that all the shares of Class A Common Stock that are issued upon the exchange of such Paired Interests or Class P LLC Units will, upon issuance, be validly issued, fully paid and non-assessable.

6.3 Taxes. The issuance of shares of Class A Common Stock upon the exercise by holders of LLC Units of their right under the Exchange Agreement to exchange Paired Interests or Class P LLC Units for shares of Class A Common Stock will be made without charge to such holders for any transfer taxes, stamp taxes or duties or other similar tax in respect of the issuance; provided, however, that if any such shares of Class A Common Stock are to be issued in a name other than that of the then record holder of the Paired Interests or Class P LLC Units being exchanged (or The Depository Trust Company or its nominee for the account of a participant of The Depository Trust Company that will hold the shares for the account of such holder), then such holder and/or the Person in whose name such shares are to be delivered, shall pay to the Corporation the amount of any tax that may be payable in respect of any transfer involved in the issuance or shall establish to the reasonable satisfaction of the Corporation that the tax has been paid or is not payable.

6.4 <u>Preemptive Rights</u>. To the extent Class A LLC Units are issued pursuant to the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC to anyone other than the Corporation or a wholly owned subsidiary of the Corporation, an equivalent number of shares of Class V Common Stock (subject to adjustment as set forth herein) shall be issued to the same Person to which such Class A LLC Units are issued at par.

## 7. <u>Board of Directors</u>.

7.1 <u>Number of Directors</u>.

(i) The business and affairs of the Corporation shall be managed by, or under the direction of, the Board. Unless and except to the extent that the By-laws of the Corporation (as such By-laws may be amended from time to time, the "<u>By-laws</u>") shall so require, the election of the directors of the Corporation (the "<u>Directors</u>") need not be by written ballot. Except as otherwise provided for or fixed pursuant to the provisions of Section 5.2 of this Certificate of Incorporation relating to the rights of the holders of any series of Preferred Stock to elect additional Directors, the total number of Directors constituting the entire Board shall, (a) as of the date of this Certificate of Incorporation, be nine (9) and (b) thereafter, shall be fixed exclusively by one or more resolutions adopted from time to time by the Board.

(ii) During any period when the holders of any series of Preferred Stock have the right to elect additional Directors as provided for or fixed pursuant to the provisions of Section 5.2 ("<u>Preferred Stock Directors</u>"), upon the commencement, and for the duration, of the period during which such right continues: (i) the then total authorized number of Directors shall automatically be increased by such specified number of Preferred Stock Directors, and the holders of the related Preferred Stock shall be entitled to elect the Preferred Stock Directors pursuant to the provisions of the Board's designation for the series of Preferred Stock and (ii) each such Preferred Stock Director shall serve until such Preferred Stock Director's successor shall have been duly elected and qualified, or until such Preferred Stock Director's right to hold such office terminates pursuant to such provisions, whichever occurs earlier, subject to his or her earlier death, disqualification, resignation or removal. Except as otherwise provided by the Board in the resolution or resolutions establishing such series, whenever the holders of any series of Preferred Stock having such right to elect Preferred Stock Directors are divested of such right pursuant to the provisions of such stock, the terms of office of all such Preferred Stock Directors elected by the holders of such Preferred Stock, or elected to fill any vacancies resulting from the death, resignation, disqualification or removal of such Preferred Stock Directors, shall forthwith terminate and the total and authorized number of Directors shall be reduced accordingly.

7.2 <u>Staggered Board</u>. The Board (other than Preferred Stock Directors) shall be divided into three (3) classes, as nearly equal in number as possible, designated Class I, Class II and Class III. Class I Directors shall initially serve until the first annual meeting of stockholders following the adoption of this Certificate of Incorporation; Class II Directors shall initially serve until the second annual meeting of stockholders following the adoption of this Certificate of Incorporation; and Class III Directors shall initially serve until the third annual meeting of stockholders following the adoption of this Certificate of Incorporation; and Class III Directors of each class the term of which shall then expire shall be elected to hold office for a term ending on the date of the third annual meeting of stockholders next following the annual meeting at which such director was elected. In case of any increase or decrease, from time to time, in the number of Directors (other than Preferred Stock Directors), the number of Directors in each class shall be apportioned as nearly equal as possible. In the event of any change in the number of directors, the Board shall apportion any newly created directorships among, or reduce the number of directorships in, such class or classes as shall equalize, as nearly as possible, the number of directors in each class. In no event will a decrease in the number of directors shorten the term of any incumbent director. As of the date of this Certificate of Incorporation, the names and mailing addresses of the persons who are to serve initially as directors of each Class are:

	Name	Mailing Address
Class I	Chris Beal	41 Discovery, Irvine, California 92618
	Fiona Tan	41 Discovery, Irvine, California 92618
	Vacancy	N/A
Class II	Olga Gonzalez	41 Discovery, Irvine, California 92618
	Tony Aquila	41 Discovery, Irvine, California 92618
	Brenda Freeman	41 Discovery, Irvine, California 92618
Class III	Justin Hartfield	41 Discovery, Irvine, California 92618
	Douglas Francis	41 Discovery, Irvine, California 92618
	Scott Gordon	41 Discovery, Irvine, California 92618

7.3 <u>Vacancies and Newly Created Directorships</u>. Subject to any limitations imposed by applicable law and the rights of the holders of any one or more series of Preferred Stock then outstanding, newly created directorships resulting from any increase in the authorized number of Directors or any vacancies on the Board resulting from death, resignation, retirement, disqualification, removal from office or other cause shall, unless the Board determines by resolution that any such vacancies or newly created directorships shall be filled by the stockholders and except as otherwise provided by applicable law, be filled only by the affirmative vote of a majority of the remaining Directors then in office, even if less than a quorum of the Board, and not by the stockholders. Any Director so chosen shall hold office until the next election of the class for which such Director shall have been chosen and until his or her successor shall be duly elected and qualified or until such Director's earlier death, disqualification, resignation or removal. No decrease in the number of Directors shall shorten the term of any Director then in office.

7.4 <u>Removal of Directors</u>. Subject to any limitations imposed by applicable law, except for Preferred Stock Directors, any Director or the entire Board may be removed from office at any time, but only for cause by the affirmative vote of the holders of at least a majority of the total voting power of the outstanding shares of capital stock of the Corporation entitled to vote generally in the election of Directors, voting together as a single class.

## 8. <u>Meetings of Stockholders</u>.

8.1 <u>Action by Written Consent</u>. Any action required or permitted to be taken by the stockholders of the Corporation may be effected only at a duly called annual or special meeting of stockholders of the Corporation and may not be effected by any consent in writing by such stockholders; <u>provided</u>, <u>however</u>, that any action required or permitted to be taken by the holders of Class V Common Stock, voting separately as a class, may be effected by the consent in writing of the holders of a majority of the total voting power of the Class V Common Stock entitled to vote thereon, voting together as a single class in lieu of a duly called annual or special meeting of holders of Class V Common Stock.

8.2 <u>Meetings of Stockholders</u>. (i) An annual meeting of stockholders for the election of directors to succeed those whose terms expire and for the transaction of such other business as may properly come before the meeting shall be held at such place, on such date, and at such time as the Board shall determine.

(ii) Subject to any special rights of the holders of any series of Preferred Stock, and to the requirements of applicable law, special meetings of stockholders of the Corporation may be called only by the chairperson of the Board, the chief executive officer of the Corporation or at the direction of the Board pursuant to a written resolution adopted by a majority of the total number of Directors that the Corporation would have if there were no vacancies. Any business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes stated in the notice of meeting. The ability of holders of Common Stock to call a special meeting of the stockholders is hereby specifically denied.

(iii) Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws of the Corporation.

8.3 <u>No Cumulative Voting; Election of Directors by Written Ballot</u>. There shall be no cumulative voting in the election of directors. Unless and except to the extent that the By-laws shall so require, the election of the Directors need not be by written ballot.

9. <u>Indemnification</u>.

9.1 <u>Limited Liability</u>. The liability of the directors for monetary damages shall be eliminated to the fullest extent permitted by applicable law.

9.2 <u>Right to Indemnification</u>.

9.3 To the fullest extent permitted by applicable law, the Corporation shall provide indemnification of (and advancement of expenses to) Directors, officers, employees and agents of the Corporation (and any other persons to which applicable law permits the Company to provide indemnification) through Bylaw provisions, agreements with such agents or other persons, vote of stockholders or disinterested directors or otherwise in excess of the indemnification and advancement otherwise permitted by such applicable law. If applicable law is amended after approval by the stockholders of this Article 9 to authorize corporate action further eliminating or limiting the personal liability of Directors, then the liability of a director to the Corporation shall be eliminated or limited to the fullest extent permitted by applicable law as so amended. Any repeal or modification of this Article 9 in effect at the time of the alleged occurrence of any act or omission to act giving rise to liability or indemnification.

9.4 <u>Insurance</u>. The Corporation shall have power to purchase and maintain insurance on behalf of any Person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any expense, liability or loss incurred by such Person in any such capacity or arising out of such Person's status as such, whether or not the Corporation would have the power to indemnify such Person against such liability under the General Corporation Law.

9.5 <u>Nonexclusivity of Rights</u>. The rights and authority conferred in this Article 9 shall not be exclusive of any other right that any Person may otherwise have or hereafter acquire.

10. <u>Adoption, Amendment or Repeal of By-Laws</u>. In furtherance and not in limitation of the powers conferred by law, the Board is expressly authorized to make, alter, amend or repeal the By-laws. Any adoption, amendment or repeal of the Bylaws of the Corporation by the Board shall require the approval of a majority of the authorized number of Directors. The stockholders shall also have power to adopt, amend or repeal the Bylaws of the Corporation; provided, however, that, in addition to any vote of the holders of any class or series of stock of the Corporation required by law or by this Certificate of Incorporation, such action by stockholders shall require the affirmative vote of the holders of at least 66 2/3% of the voting power of all of the then-outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class.

11. Adoption, Amendment and Repeal of Certificate. Subject to Article 5, the Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by the General Corporation Law, and all rights, preferences and privileges of whatsoever nature conferred upon stockholders, Directors or any other Persons whomsoever by and pursuant to this Certificate of Incorporation in its present form or as hereafter amended, are granted and held subject to this reservation. Notwithstanding anything to the contrary contained in this Certificate of Incorporation, and notwithstanding that a lesser percentage may be permitted from time to time by applicable law, no provision of Sections 7.2, 7.3 and 7.4 of Article 7, Sections 8.1 and 8.2 of Article 8 or Article 9, 10, 12 or 13 may be altered, amended or repealed in any respect, nor may any provision or oby-law inconsistent therewith be adopted, unless in addition to any other vote required by this Certificate of Incorporation is approved by, in addition to any other vote otherwise required by law, the affirmative vote of the holders of sixty-six and two-thirds percent (66 and 2/3%) of the total voting power of the outstanding shares of capital stock of the Corporation entitled to vote generally in the election of Directors, voting together as a single class, at a meeting of the stockholders called for that purpose.

### 12. Forum for Adjudication of Disputes.

Unless the Corporation consents in writing to the selection of an alternative forum, the Court of 12.1 Chancery of the State of Delaware shall (or, if and only if the Court of Chancery of the State of Delaware lacks subject matter jurisdiction, any state court located within the State of Delaware or, if and only if all such state courts lack subject matter jurisdiction, the federal district court for the District of Delaware), to the fullest extent permitted by applicable law, be the sole and exclusive forum for the following types of actions or proceedings under Delaware statutory or common law: (A) any derivative action or proceeding brought on behalf of the Corporation; (B) any action or proceeding (including any class action) asserting a claim of breach of a fiduciary duty owed by any current or former director, officer or other employee of the Corporation to the Corporation or the Corporation's stockholders; (C) any action or proceeding (including any class action) asserting a claim against the Corporation or any current or former director, officer or other employee of the Company arising out of or pursuant to any provision of the General Corporation Law, this Certificate of Incorporation or the Bylaws (as each may be amended from time to time); (D) any action or proceeding (including any class action) to interpret, apply, enforce or determine the validity of this Certificate of Incorporation or the Bylaws of the Corporation (including any right, obligation or remedy thereunder); (E) any action or proceeding as to which the General Corporation Law confers jurisdiction to the Court of Chancery of the State of Delaware; or (F) any action asserting a claim against the Corporation or any director, officer or other employee of the Corporation governed by the internal affairs doctrine, in all cases to the fullest extent permitted by law and subject to the court's having personal jurisdiction over the indispensable parties named as defendants. This Article 12 shall not apply to suits brought to enforce a duty or liability created by the Securities Exchange Act of 1934, as amended, or any other claim for which the federal courts have exclusive jurisdiction.

12.2 If any action the subject matter of which is within the scope of Section 12.1 is filed in a court other than a court located within the State of Delaware (a "<u>Foreign Action</u>") in the name of any stockholder, such stockholder shall be deemed to have consented to (i) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce Section 12.1 (an "<u>Enforcement Action</u>") and (ii) having service of process made upon such stockholder in any such Enforcement Action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder.

12.3 Unless the Company consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended.

12.4 Any person or entity purchasing, holding, owning or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and to have consented to the provisions of this Article 12.

13. <u>Severability</u>. If any provision or provisions of this Certificate of Incorporation shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Certificate of Incorporation (including, without limitation, each portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (ii) to the fullest extent possible, the provisions of this Certificate of Incorporation (including, without limitation, each portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its Directors, officers, employees and agents from personal liability in respect of their good faith service to or for the benefit of the Corporation to the fullest extent permitted by law.

14. <u>Corporate Opportunity</u>. To the extent allowed by law, the doctrine of corporate opportunity, or any other analogous doctrine, shall not apply with respect to the Corporation or any of its officers or Directors, or any of their respective affiliates, and the Corporation renounces any expectancy that any of the directors or officers of the Corporation will offer any such corporate opportunity of which he or she may become aware to the Corporation, except, the doctrine of corporate opportunity shall apply with respect to any of the directors or officers of the Corporation and (i) such opportunity that was offered to such person solely in his or her capacity as a director or officer of the Corporation and (i) such opportunity is one the Corporation is legally and contractually permitted to undertake and would otherwise be reasonable for the Corporation to pursue and (ii) the director or officer is permitted to refer that opportunity to the Corporation without violating any legal obligation.

15. <u>Definitions</u>. As used in this Certificate of Incorporation, unless the context otherwise requires or as set forth in another Article or Section of this Certificate of Incorporation, the term:

(a) "<u>Affiliate</u>" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person; <u>provided</u>, that (i) neither the Corporation nor any of its subsidiaries will be deemed an Affiliate of any stockholder of the Corporation or any of such stockholders' Affiliates and (ii) no stockholder of the Corporation will be deemed an Affiliate of any other stockholder of the Corporation, in each case, solely by reason of any investment in the Corporation (including any representatives of such stockholder serving on the Board).

- (b) "<u>Board</u>" is defined in Section 5.1(ii)(1).
- (c) "<u>By-laws</u>" is defined in Section 7.1.
- (d) "<u>Certificate of Incorporation</u>" is defined in the recitals.
- (e) "<u>Class A Common Stock</u>" is defined in Section 4.1.

(f) "<u>Class A LLC Unit</u>" means a unit of WM Holding Company, LLC designated as a Class A Unit pursuant to the Fourth Amended and Restated Operating Agreement of WM Holding Company.

(g) "<u>Class P LLC Unit</u>" means a unit of WM Holding Company, LLC designated as a Class P Unit pursuant to the Fourth Amended and Restated Operating Agreement of WM Holding Company.

- (h) "<u>Class V Common Stock</u>" is defined in Section 4.1.
- (i) "<u>Common Stock</u>" is defined in Section 4.1.

(j) "<u>control</u>" (including the terms "<u>controlling</u>" and "<u>controlled</u>"), with respect to the relationship between or among two or more Persons, means the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of such subject Person, whether through the ownership of voting securities, as trustee or executor, by contract or otherwise.

- (k) "<u>Corporation</u>" means WM Technology, Inc.
- (l) "<u>Director</u>" is defined in Section 7.1.
- (m) "<u>Enforcement Action</u>" is defined in Section 12.2.

(n) "<u>Exchange Agreement</u>" means the Exchange Agreement, dated as of June 16, 2021, by and among WM Technology, Inc., a Delaware corporation, WM Holding Company, LLC and the holders from time to time party hereto.

(o) "<u>Foreign Action</u>" is defined in Section 12.2.

(p) "<u>Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC</u>" means the Fourth Amended and Restated Operating Agreement, dated as of June 16, 2021, by and among the Corporation, the Post-Acquisition LLC Members and the other Persons that may become parties thereto from time to time, as the same may be amended, restated, supplemented and/or otherwise modified, from time to time. (q) "<u>General Corporation Law</u>" is defined in the recitals.

(r) "<u>LLC Unit</u>" means a nonvoting interest unit of WM Holding Company, LLC.

(s) "<u>LTIP LLC Unit</u>" means a unit of WM Holding Company, LLC designated as a LTIP Unit pursuant to the Fourth Amended and Restated Operating Agreement of WM Holding Company.

(t) "<u>Paired Interest</u>" means one Class A LLC Unit together with one share of Class V Common Stock, subject to adjustment pursuant to the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC.

(u) "<u>Participation Threshold</u>" shall have the meaning ascribed to it in the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC.

(v) "<u>Person</u>" means any individual, partnership, firm, corporation, limited liability company, association, trust, unincorporated organization or other entity.

(w) "<u>Post-Acquisition LLC Members</u>" means holders of LLC Units that sign the Fourth Amended and Restated Operating Agreement of WM Holding Company, LLC.

(x) "<u>Preferred Stock</u>" is defined in Section 4.1.

(y) "<u>Preferred Stock Directors</u>" is defined in Section 7.1.

(z) "<u>Stock Adjustment</u>" is defined in Section 5.1(ii)(3).

(aa) "<u>WM Holding Company, LLC</u>" means WM Holding Company, LLC, a Delaware limited liability company or any successor thereto.

16. <u>Incorporator</u>. The name and mailing address of the incorporator is:

NameMailing AddressScott Gordon660 Madison Avenue, Suite 1600<br/>New York, New York 10065

[Remainder of page intentionally left blank.]

IN WITNESS WHEREOF, this Certificate of Incorporation of WM Technology, Inc. has been duly executed by the incorporator below this 15<sup>th</sup> day of June, 2021.

By: /s/ Scott Gordon

Name: Scott Gordon Title: Incorporator

[Signature Page to Certificate of Incorporation]

## **BY-LAWS**

of

## WM TECHNOLOGY, INC.

(A Delaware Corporation)

## TABLE OF CONTENTS

PAGE

#### ARTICLE 1 Definitions

#### ARTICLE 2 Stockholders

Section 2.01.	Place of Meetings	2
Section 2.02.	Annual Meetings; Stockholder Proposals	2
Section 2.03.	Special Meetings	6
Section 2.04.	Record Date	6
Section 2.05.	Notice of Meetings of Stockholders	7
Section 2.06.	Waivers of Notice	8
Section 2.07.	List of Stockholders	8
Section 2.08.	Quorum of Stockholders; Adjournment	8
Section 2.09.	Voting; Proxies	9
Section 2.10.	Voting Procedures and Inspectors at Meetings of Stockholders	9
Section 2.11.	Conduct of Meetings; Adjournment	10
Section 2.12.	Order of Business	10
Section 2.13.	Written Consent of Stockholders Without a Meeting	10
	ARTICI F 3	

#### ARTICLE 3 Directors

Section 3.01.	General Powers	11
Section 3.02.	Term of Office	11
Section 3.03.	Nominations of Directors	11
Section 3.04.	Nominee and Director Qualifications	14
Section 3.05.	Resignation	15
Section 3.06.	Compensation	15
Section 3.07.	Regular Meetings	15
Section 3.08.	Special Meetings	15
Section 3.09.	Telephone Meetings	15
Section 3.10.	Adjourned Meetings	15
Section 3.11.	Notice Procedure	16
Section 3.12.	Waiver of Notice	16
Section 3.13.	Organization	16
Section 3.14.	Quorum of Directors	16
Section 3.15.	Action by Majority Vote	16
Section 3.16.	Action Without Meeting	16

	ARTICLE 4	
	Committees of the Board	
	ARTICLE 5 Officers	
	Ufficers	
Section 5.01.	Positions; Election	17
Section 5.02.	Term of Office	18
Section 5.03.	Chairman	18
Section 5.04.	Vice Chairman	18
Section 5.05.	Chief Executive Officer	18
Section 5.06.	President	19
Section 5.07.	Vice Presidents	19
Section 5.08.	Secretary	20
Section 5.09.	Treasurer	20
Section 5.10.	Assistant Secretaries and Assistant Treasurers	19
	ARTICLE 6	
	General Provisions	
Section 6.01.	Certificates Representing Shares	20
Section 6.02.	Transfer and Registry Agents	20
Section 6.03.	Lost, Stolen or Destroyed Certificates	20
Section 6.04.	Form of Records	20
Section 6.05.	Seal	21
Section 6.06.	Fiscal Year	21
Section 6.07.	Amendments	21
Section 6.08.	Conflict with Applicable Law or Certificate of Incorporation	21

ii

#### ARTICLE 1 DEFINITIONS

As used in these By-laws, unless the context otherwise requires, the term:

"Assistant Secretary" means an Assistant Secretary of the Corporation.

"Assistant Treasurer" means an Assistant Treasurer of the Corporation.

"Board" means the Board of Directors of the Corporation.

"By-laws" means the By-laws of the Corporation, as amended and restated.

"Certificate of Incorporation" means the Certificate of Incorporation of the Corporation, as amended and restated.

"Chairman" means the Chairman of the Board and includes any Executive Chairman.

"Chief Executive Officer" means the Chief Executive Officer of the Corporation.

"**control**" (including the terms "controlling" and "controlled"), with respect to the relationship between or among two or more persons, means the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of such subject person, whether through the ownership of voting securities, as trustee or executor, by contract or otherwise.

"Corporation" means WM Technology, Inc..

"Derivative" is defined in Section 2.02(d)(iii).

"Directors" means the directors of the Corporation.

**"Exchange Act**" means the Securities Exchange Act of 1934, as amended, and any successor law or statute, and the rules and regulations promulgated thereunder.

"Executive Chairman" means the Executive Chairman of the Board.

"General Corporation Law" means the General Corporation Law of the State of Delaware, as amended.

"**law**" means any U.S. or non-U.S. federal, state or local law (statutory, common or otherwise), constitution, treaty, convention, ordinance, code, rule, regulation, order, injunction, judgment, decree, ruling or other similar requirement enacted, adopted, promulgated or applied by a governmental authority (including any department, court, agency or official, or non-governmental self-regulatory organization, agency or authority and any political subdivision or instrumentality thereof).

"Lead Independent Director" is defined in Section 3.17.

"**Nominating Stockholder**" is defined in Section 3.03(b).

"**Notice of Business**" is defined in Section 2.02(c).

"Notice of Nomination" is defined in Section 3.03(c).

"Notice Record Date" is defined in Section 2.04(a).

**"Office of the Corporation**" means the executive office of the Corporation, anything in Section 131 of the General Corporation Law to the contrary notwithstanding.

"President" means the President of the Corporation.

"**Proponent**" is defined in Section 2.02(d)(i).

"Public Disclosure" is defined in Section 2.02(i).

"SEC" means the Securities and Exchange Commission.

"Secretary" means the Secretary of the Corporation.

"Stockholder Associated Person" is defined in Section 2.02(j).

"Stockholder Business" is defined in Section 2.02(b).

"Stockholder Information" is defined in Section 2.02(d)(iii).

"Stockholder Nominees" is defined in Section 3.03(b).

"Stockholders" means the stockholders of the Corporation.

"Treasurer" means the Treasurer of the Corporation.

"Vice President" means a Vice President of the Corporation.

"Voting Commitment" is defined in Section 3.04.

"Voting Record Date" is defined in Section 2.04(a).

## ARTICLE 2

## STOCKHOLDERS

Section 2.01. *Place of Meetings*. Meetings of Stockholders may be held within or without the State of Delaware, at such place or solely by means of remote communication or otherwise, as may be designated by the Board from time to time. The Board may, in its sole discretion, determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication as provided under the General Corporation Law.

Section 2.02. Annual Meetings; Stockholder Proposals.

(a) A meeting of Stockholders for the election of Directors and other business shall be held annually at such date and time as may be designated by the Board from time to time. The Board may postpone, reschedule or cancel any annual meeting of stockholders previously scheduled by the Board.

(b) At an annual meeting of the Stockholders, only business (other than business relating to the nomination or election of Directors, which is governed by Section 3.03) that has been properly brought before the Stockholder meeting in accordance with the procedures set forth in this Section 2.02 shall be conducted. To be properly brought before a meeting of Stockholders, such business must be brought before the meeting (i) by or at the direction of the Board or any committee thereof or (ii) by a Stockholder who (A) was a Stockholder of record of the Corporation when the notice required by this Section 2.02 is delivered to the Secretary and at the time of the meeting, (B) is entitled to vote at the meeting and (C) complies with the notice and other provisions of this Section 2.02. Subject to Section 2.02(k), and except with respect to nominations or elections of Directors, which are governed by Section 3.03, Section 2.02(b)(ii) is the exclusive means by which a Stockholder may bring business before a meeting of Stockholders; *provided* that if Rule 14a-8 of the Exchange Act (or any successor rule) is applicable, a Stockholder may not bring business before any meeting if the Stockholder fails to meet the requirements of such rule. Any business brought before a meeting in accordance with Section 2.02(b)(ii) is referred to as "**Stockholder Business**."

(c) Subject to Section 2.02(k), at any annual meeting of Stockholders, all proposals of Stockholder Business must be made by timely written notice given by or on behalf of a Stockholder of record of the Corporation (the "**Notice of Business**") and must otherwise be a proper matter for Stockholder action. To be timely, the Notice of Business must be delivered personally or mailed to, and received at, the Office of the Corporation, addressed to the Secretary, by no earlier than one hundred and twenty (120) days and no later than ninety (90) days before the first anniversary of the date of the prior year's annual meeting of Stockholders; *provided, however*, that if (i) the annual meeting of Stockholders is advanced by more than thirty (30) days, or delayed by more than sixty (60) days, from the first anniversary of the prior year's annual meeting of Stockholders or (ii) no annual meeting was held during the prior year, the notice by the Stockholder to be timely must be received (A) no earlier than one hundred and twenty (120) days before such annual meeting and (B) no later than the later of ninety (90) days before such annual meeting and the tenth day after the day on which the notice of such annual meeting was made by mail or Public Disclosure. In no event shall an adjournment, postponement or deferral, or Public Disclosure of an adjournment, postponement or deferral, of a Stockholder meeting of the Notice of Business.

(d) The Notice of Business must set forth:

(i) the name and record address of each Stockholder proposing Stockholder Business (the "**Proponent**"), as they appear on the Corporation's books;

(ii) the name and address of any Stockholder Associated Person;

as to each Proponent and any Stockholder Associated Person, (A) the class or series and number of shares (iii) of stock directly or indirectly held of record and beneficially by the Proponent or Stockholder Associated Person, (B) the date such shares of stock were acquired, (C) a description of any agreement, arrangement or understanding, direct or indirect, with respect to such Stockholder Business between or among the Proponent, any Stockholder Associated Person or any others (including their names) acting in concert with any of the foregoing, (D) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, hedging transactions, warrant, convertible security, stock appreciation right or similar right with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class of securities and/or borrowed or loaned shares) that has been entered into, directly or indirectly, as of the date of the Proponent's notice by, or on behalf of, the Proponent or any Stockholder Associated Person, the effect or intent of which is to mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of the Proponent or any Stockholder Associated Person with respect to shares of stock of the Corporation or with a value derived in whole or in part from the value or decrease in value of any class or series of stock of the Corporation, whether or not such instrument or right shall be subject to settlement in the underlying class or series of stock of the Corporation or otherwise (a "**Derivative**"), (E) a description in reasonable detail of any proxy (including revocable proxies), contract, arrangement, understanding or other relationship pursuant to which the Proponent or Stockholder Associated Person has a right to vote any shares of stock of the Corporation, (F) any rights to dividends on the stock of the Corporation owned beneficially by the Proponent or Stockholder Associated Person that are separated or separable from the underlying stock of the Corporation, (G) any proportionate interest in stock of the Corporation or Derivatives held, directly or indirectly, by a general or limited partnership in which the Proponent or Stockholder Associated Person is a general partner or, directly or indirectly, beneficially owns an interest in a general partner and (H) any performance-related fees (other than an asset-based fee) that the Proponent or Stockholder Associated Person is entitled to based on any increase or decrease in the value of stock of the Corporation or Derivatives thereof, if any, as of the date of such notice. The information specified in Section 2.02(d)(i) to (iii) is referred to herein as "Stockholder Information";

(iv) Stockholder Information with respect to any stock or other interests of the Corporation held by members of the Proponent's or Stockholder Associated Person's immediate family sharing the same household;

(v) a representation to the Corporation that each Proponent is a holder of record of stock of the Corporation entitled to vote at the meeting and intends to appear in person or by proxy at the meeting to propose such Stockholder Business;

(vi) a brief description of the Stockholder Business desired to be brought before the annual meeting, the text of the proposal (including the text of any resolutions proposed for consideration and, if such business includes a proposal to amend the By-laws, the language of the proposed amendment) and the reasons for conducting such Stockholder Business at the meeting;

(vii) any material interest of each Proponent and any Stockholder Associated Person in such Stockholder Business;

(viii) a representation to the Corporation as to whether the Proponent intends (A) to deliver a proxy statement and form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt such Stockholder Business or (B) otherwise to solicit proxies from the Stockholders in support of such Stockholder Business;

(ix) all other information that would be required to be filed with the SEC if the Proponents or Stockholder Associated Persons were participants in a solicitation subject to Section 14 of the Exchange Act; and

(x) a representation and covenant for the benefit of the Corporation that the Proponents shall provide any other information reasonably requested by the Corporation.

(e) The Proponents shall also provide any other information reasonably requested by the Corporation within ten (10) business days after such request.

(f) In addition, the Proponent shall further update and supplement the information provided to the Corporation in the Notice of Business or upon the Corporation's request pursuant to Section 2.02(e) as needed, so that such information shall be true and correct as of the record date for the meeting and as of the date that is the later of five (5) business days before the meeting or any adjournment or postponement thereof. Such update and supplement must be delivered personally or mailed to, and received at, the Office of the Corporation, addressed to the Secretary, by no later than five (5) business days after the record date for the meeting (in the case of the update and supplement required to be made as of the record date), and not later than two (2) business days before the meeting or any adjournment or postponement thereof).

(g) The person presiding over the meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting in accordance with the procedures set forth in this Section 2.02, and if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

(h) If the Proponent (or a qualified representative of the Proponent) does not appear at the meeting of Stockholders to present the Stockholder Business, such business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For purposes of this Section 2.02, to be considered a qualified representative of the Stockholder, a person must be a duly authorized officer, manager or partner of such Stockholder or must be authorized by a writing executed by such Stockholder or an electronic transmission delivered by such Stockholder to act for such Stockholder as proxy at the meeting of Stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of Stockholders.

(i) **"Public Disclosure**" of any date or other information means disclosure thereof by a press release reported by the Dow Jones News Services, Associated Press or comparable U.S. national news service or in a document publicly filed by the Corporation with the SEC pursuant to Sections 13, 14 or 15(d) of the Exchange Act.

(j) "**Stockholder Associated Person**" means, with respect to any Stockholder, (i) any other beneficial owner of stock of the Corporation that is owned by such Stockholder and (ii) any person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, the Stockholder or such beneficial owner.

(k) The notice requirements of this Section 2.02 shall be deemed satisfied with respect to Stockholder proposals that have been properly brought under Rule 14a-8 of the Exchange Act and that are included in a proxy statement that has been prepared by the Corporation to solicit proxies for such annual meeting. Further, nothing in this Section 2.02 shall be deemed to affect any rights of the holders of any series of preferred stock of the Corporation pursuant to any applicable provision of the Certificate of Incorporation.

Section 2.03. *Special Meetings.* Special meetings of the Stockholders may be called only in the manner set forth in the Certificate of Incorporation. Notice of every special meeting of the Stockholders shall state the purpose or purposes of such meeting. Except as otherwise required by law, the business conducted at a special meeting of Stockholders shall be limited exclusively to the business set forth in the Corporation's notice of meeting, and the Board shall have exclusive authority to determine the business included in such notice. The Chairman, the Chief Executive Officer or the Board may postpone, reschedule or cancel any special meeting of stockholders previously called by any of them.

## Section 2.04. Record Date.

For the purpose of determining the Stockholders entitled to notice of any meeting of Stockholders or any (a) adjournment thereof, unless otherwise required by the Certificate of Incorporation or applicable law, the Board may fix a record date (the "Notice Record Date"), which record date shall not precede the date on which the resolution fixing the record date was adopted by the Board and shall not be more than sixty (60) or less than ten (10) days before the date of such meeting. The Notice Record Date shall also be the record date for determining the Stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such Notice Record Date, that a later date on or before the date of the meeting shall be the date for making such determination (the "Voting Record Date"). For the purposes of determining the Stockholders entitled to express consent to corporate action in writing without a meeting, unless otherwise required by the Certificate of Incorporation or applicable law, the Board may fix a record date, which record date shall not precede the date on which the resolution fixing the record date was adopted by the Board and shall not be more than ten (10) days after the date on which the record date was fixed by the Board. For the purposes of determining the Stockholders entitled to (i) receive payment of any dividend or other distribution or allotment of any rights, (ii) exercise any rights in respect of any change, conversion or exchange of stock or (iii) take any other lawful action, unless otherwise required by the Certificate of Incorporation or applicable law, the Board may fix a record date, which record date shall not precede the date on which the resolution fixing the record date was adopted by the Board and shall not be more than sixty (60) days prior to such action.

## (b) If no such record date is fixed:

(i) the record date for determining Stockholders entitled to notice of, and to vote at, a meeting of Stockholders shall be at the close of business on the day next preceding the day on which notice is given or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held;

(ii) the record date for determining Stockholders entitled to express consent to corporate action in writing without a meeting (unless otherwise provided in the Certificate of Incorporation), when no prior action by the Board is required by applicable law, shall be the first day on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation in accordance with applicable law; and when prior action by the Board is required by applicable law, the record date for determining Stockholders entitled to express consent to corporate action in writing without a meeting shall be at the close of business on the date on which the Board takes such prior action; and

(iii) when a determination of Stockholders of record entitled to notice of, or to vote at, any meeting of Stockholders has been made as provided in this Section 2.04, such determination shall apply to any adjournment thereof, unless the Board fixes a new Voting Record Date for the adjourned meeting, in which case the Board shall also fix such Voting Record Date or a date earlier than such date as the new Notice Record Date for the adjourned meeting.

Notice of Meetings of Stockholders. Whenever, under the provisions of applicable law, the Certificate of Section 2.05. Incorporation or these By-laws, Stockholders are required or permitted to take any action at a meeting, notice shall be given stating the place, if any, date and hour of the meeting; the means of remote communication, if any, by which Stockholders and proxy holders may be deemed to be present in person and vote at such meeting; the Voting Record Date, if such date is different from the Notice Record Date; and, in the case of a special meeting, the purposes for which the meeting is called. Unless otherwise provided by these By-laws or applicable law, notice of any meeting shall be given, not less than ten (10) nor more than sixty (60) days before the date of the meeting, to each Stockholder entitled to vote at such meeting as of the Notice Record Date. If mailed, such notice shall be deemed to be given when deposited in the U.S. mail, with postage prepaid, and directed to the Stockholder at his or her address as it appears on the records of the Corporation. An affidavit of the Secretary, an Assistant Secretary or the transfer agent of the Corporation that the notice required by this Section 2.05 has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. If a meeting is adjourned to another time or place, notice need not be given of the adjourned meeting if the time and place thereof are announced at the meeting at which the adjournment is taken. Any business that might have been transacted at the meeting as originally called may be transacted at the adjourned meeting. If, however, the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each Stockholder of record entitled to vote at the meeting. If, after the adjournment, a new Voting Record Date is fixed for the adjourned meeting, the Board shall fix a new Notice Record Date in accordance with Section 2.04(b)(iii) hereof and shall give notice of such adjourned meeting to each Stockholder entitled to vote at such meeting as of the Notice Record Date.

Section 2.06. *Waivers of Notice*. Whenever the giving of any notice to Stockholders is required by applicable law, the Certificate of Incorporation or these By-laws, a waiver thereof, given by the person entitled to said notice, whether before or after the event as to which such notice is required, shall be deemed equivalent to notice. Attendance by a Stockholder at a meeting shall constitute a waiver of notice of such meeting except when the Stockholder attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business on the ground that the meeting has not been lawfully called or convened. Any Stockholder so waiving notice of such meeting shall be bound by the proceedings of any such meeting in all respects as if due notice thereof had been given. Neither the business to be transacted at, nor the purposes of, any regular or special meeting of the Stockholders need be specified in any waiver of notice.

Section 2.07. *List of Stockholders*. The Secretary shall prepare and make available, at least ten (10) days before every meeting of Stockholders, a complete, alphabetical list of the Stockholders entitled to vote at the meeting, and showing the address of each Stockholder and the number of shares registered in the name of each Stockholder. Such list may be examined by any Stockholder, the Stockholder's agent or attorney, at the Stockholder's expense, for any purpose germane to the meeting, for a period of at least ten (10) days prior to the meeting, during ordinary business hours at the principal place of business of the Corporation or on a reasonably accessible electronic network as provided by applicable law. If the meeting is to be held at a place, the list shall also be produced and kept at the time and place of the meeting during the whole time thereof and may be inspected by any Stockholder who is present. If the meeting is held solely by means of remote communication, the list shall also be open for inspection as provided by applicable law. Except as provided by applicable law, the stock ledger shall be the only evidence as to who are the Stockholders entitled to examine the list of Stockholders or to vote in person or by proxy at any meeting of Stockholders.

Section 2.08. *Quorum of Stockholders; Adjournment.* Except as otherwise provided by these By-laws, at each meeting of Stockholders, the presence in person or by proxy of the holders of a majority of the voting power of all outstanding shares of stock entitled to vote at the meeting of Stockholders shall constitute a quorum for the transaction of any business at such meeting, except that, where a separate vote by a class or series of classes of shares is required, a quorum shall consist of no less than a majority of the voting power of all outstanding shares of stock of such class or series of classes, as applicable. In the absence of a quorum, the holders of a majority in voting power of the shares of stock present in person or represented by proxy at any meeting of Stockholders, including an adjourned meeting, may adjourn such meeting to another time and place. Shares of its own stock belonging to the Corporation or to another corporation, if a majority of the shares entitled to vote in the election of Directors of such other corporation is held, directly or indirectly, by the Corporation, shall neither be entitled to vote nor be counted for quorum purposes; *provided, however*, that the foregoing shall not limit the right of the Corporation to vote stock, including but not limited to its own stock, held by it in a fiduciary capacity.

Section 2.09. *Voting; Proxies.* Unless otherwise provided by the General Corporation Law or in the Certificate of Incorporation, every Stockholder entitled to vote at any meeting of Stockholders shall be entitled to one vote for each share of stock held by such Stockholder which has voting power upon the matter in question. At any meeting of Stockholders, all matters other than the election of Directors, except as otherwise provided by the Certificate of Incorporation, these By-laws or any applicable law, shall be decided by the affirmative vote of a majority in voting power of shares of stock present in person or represented by proxy and entitled to vote thereon. At all meetings of Stockholders for the election of Directors, a plurality of the votes cast shall be sufficient to elect Directors. Each Stockholder entitled to vote at a meeting of Stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for such Stockholder by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy expressly provides for a longer period. A proxy shall be irrevocable power. A Stockholder may revoke any proxy that is not irrevocable by attending the meeting and voting in person or by delivering to the Secretary a revocation of the proxy or by delivering a new proxy bearing a later date.

Section 2.10. Voting Procedures and Inspectors at Meetings of Stockholders. The Board, in advance of any meeting of Stockholders, shall appoint one or more inspectors, who may be employees of the Corporation, to act at the meeting and make a written report thereof. The Board may designate one or more persons as alternate inspectors to replace any inspector who fails to act. If no inspector or alternate is able to act at a meeting, the person presiding at the meeting shall appoint one or more inspectors to act at the meeting. Each inspector, before entering upon the discharge of his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspectors shall (a) ascertain the number of shares outstanding and the voting power of each, (b) determine the shares represented at the meeting and the validity of proxies and ballots, (c) count all votes and ballots, (d) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors and (e) certify their determination of the number of shares represented at the meeting and their count of all votes and ballots. The inspectors may appoint or retain other persons or entities to assist the inspectors in the performance of their duties. Unless otherwise provided by the Board, the date and time of the opening and the closing of the polls for each matter upon which the Stockholders will vote at a meeting shall be determined by the person presiding at the meeting and shall be announced at the meeting. No ballot, proxies, votes or any revocation thereof or change thereto, shall be accepted by the inspectors after the closing of the polls unless the Court of Chancery of the State of Delaware upon application by a Stockholder shall determine otherwise. In determining the validity and counting of proxies and ballots cast at any meeting of Stockholders, the inspectors may consider such information as is permitted by applicable law. No person who is a candidate for office at an election may serve as an inspector at such election.

Conduct of Meetings; Adjournment. The Board may adopt such rules and procedures for the conduct of Section 2.11. Stockholder meetings as it deems appropriate. At each meeting of Stockholders, the Chairman or, in the absence of the Chairman, the Chief Executive Officer or, in the absence of the Chairman, and the Chief Executive Officer, the President or, if there is no Chairman, Chief Executive Officer or President, or if they are absent, a Vice President and, in the case that more than one Vice President shall be present, that Vice President designated by the Board (or in the absence of any such designation, the most senior Vice President present), shall preside over the meeting. Except to the extent inconsistent with the rules and procedures as adopted by the Board, the person presiding over the meeting of Stockholders shall have the right and authority to convene, adjourn and reconvene the meeting from time to time, to prescribe such additional rules and procedures and to do all such acts as, in the judgment of such person, are appropriate for the proper conduct of the meeting. Such rules and procedures, whether adopted by the Board or prescribed by the person presiding over the meeting, may include (a) the establishment of an agenda or order of business for the meeting, (b) rules and procedures for maintaining order at the meeting and the safety of those present, (c) limitations on attendance at or participation in the meeting to Stockholders of record of the Corporation, their duly authorized and constituted proxies or such other persons as the person presiding over the meeting shall determine, (d) restrictions on entry to the meeting after the time fixed for the commencement thereof and (e) limitations on the time allotted to questions or comments by participants. The person presiding over any meeting of Stockholders, in addition to making any other determinations that may be appropriate to the conduct of the meeting, may determine and declare to the meeting that a matter or business was not properly brought before the meeting and if such presiding person should so determine, he or she shall so declare to the meeting and any such matter or business not properly brought before the meeting shall not be transacted or considered. Unless and to the extent determined by the Board or the person presiding over the meeting, meetings of Stockholders shall not be required to be held in accordance with the rules of parliamentary procedure. The Secretary or, in his or her absence, one of the Assistant Secretaries, shall act as secretary of the meeting. If none of the officers above designated to act as the person presiding over the meeting or as secretary of the meeting shall be present, a person presiding over the meeting or a secretary of the meeting, as the case may be, shall be designated by the Board and, if the Board has not so acted, in the case of the designation of a person to act as secretary of the meeting, designated by the person presiding over the meeting. To the extent permitted by applicable law, meetings of stockholders may be conducted by remote communications, including by webcast.

Section 2.12. *Order of Business*. The order of business at all meetings of Stockholders shall be as determined by the person presiding over the meeting.

Written Consent of Stockholders Without a Meeting. If, and only if, the Certificate of Incorporation Section 2.13. expressly permits action to be taken at any annual or special meeting of Stockholders without a meeting, without prior notice and without a vote, then a consent or consents in writing, setting forth the action to be so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered (by hand or by certified or registered mail, return receipt requested) to the Corporation by delivery to its registered office in the State of Delaware, the Office of the Corporation or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of Stockholders are recorded. Every written consent shall bear the date of signature of each Stockholder who signs the consent, and no written consent shall be effective to take the corporate action referred to therein unless, within sixty (60) days of the earliest dated consent delivered in the manner required by this Section 2.13, written consents signed by a sufficient number of holders to take action are delivered to the Corporation as aforesaid. Prompt notice of the taking of the corporate action without a meeting by less than unanimous written consent shall, to the extent required by applicable law, be given to those Stockholders who have not consented in writing and who, if the action had been taken at a meeting, would have been entitled to notice of the meeting if the record date for such meeting had been the date that written consents signed by a sufficient number of holders to take the action were delivered to the Corporation.

#### ARTICLE 3 Directors

Section 3.01. *General Powers*. The business and affairs of the Corporation shall be managed by, or under the direction of, the Board. The Board may adopt such rules and procedures, not inconsistent with the Certificate of Incorporation, these By-laws or applicable law, as it may deem proper for the conduct of its meetings and the management of the Corporation.

Section 3.02. *Classes of Directors; Term of Office*. Subject to the rights of the holders of any series of Preferred Stock to elect additional Directors under specified circumstances, following the adoption of these Bylaws, the Directors shall be divided into three classes designated as Class I, Class II and Class III, respectively. The Board is authorized to assign members of the Board already in office to such classes at the time the classification becomes effective. At the first annual meeting of Stockholders following the adoption of these Bylaws, the term of office of the Class I Directors shall expire and Class I Directors shall be elected for a full term of three years. At the second annual meeting of Stockholders following the adoption of these Bylaws, the term of office of the Class II Directors shall be elected for a full term of three years. At the shall expire and Class II Directors shall be elected for a full term of three years. At the third annual meeting of Stockholders following the adoption of these Bylaws, the term of office of the Class III Directors shall expire and Class III Directors shall be elected for a full term of three years. At the third annual meeting of Stockholders following the adoption of these Bylaws, the term of office of the Class III Directors shall be elected for a full term of three years. At each succeeding annual meeting of Stockholders, Directors shall be elected for a full term of three years to succeed the Directors of the class whose terms expire at such annual meeting.

Section 3.03. *Nominations of Directors*. Notwithstanding the foregoing provisions of this Section 3.02, each Director shall serve until his or her successor is duly elected and qualified or until his or her earlier death, resignation or removal. No decrease in the number of Directors constituting the Board shall shorten the term of any incumbent Director.

(a) Subject to Section 3.03(k), only persons who are nominated in accordance with the procedures set forth in this Section 3.03 are eligible for election as Directors.

(b) Nominations of persons for election to the Board may only be made at a meeting properly called for the election of Directors and only (i) by or at the direction of the Board or any committee thereof or (ii) by a Stockholder who (A) was a Stockholder of record of the Corporation when the notice required by this Section 3.03 is delivered to the Secretary and at the time of the meeting, (B) is entitled to vote for the election of Directors at the meeting and (C) complies with the notice and other provisions of this Section 3.03. Subject to Section 3.03(k), Section 3.03(b)(ii) is the exclusive means by which a Stockholder may nominate a person for election to the Board. Persons nominated in accordance with Section 3.03(b)(ii) are referred to as "**Stockholder Nominees**." A Stockholder nominating persons for election to the Board is referred to as the "**Nominating Stockholder**."

(c) Subject to Section 3.03(k), all nominations of Stockholder Nominees must be made by timely written notice given by or on behalf of a Stockholder of record of the Corporation (the "**Notice of Nomination**"). To be timely, the Notice of Nomination must be delivered personally or mailed to and received at the Office of the Corporation, addressed to the attention of the Secretary, by the following dates:

(i) in the case of the nomination of a Stockholder Nominee for election to the Board at an annual meeting of Stockholders, no earlier than one hundred and twenty (120) days and no later than ninety (90) days before the first anniversary of the date of the prior year's annual meeting of Stockholders; *provided, however*, that if (A) the annual meeting of Stockholders is advanced by more than thirty (30) days, or delayed by more than sixty (60) days, from the first anniversary of the prior year's annual meeting of Stockholders or (B) no annual meeting was held during the prior year, the notice by the Stockholder to be timely must be received (1) no earlier than one hundred and twenty (120) days before such annual meeting and (2) no later than the later of ninety (90) days before such annual meeting and the tenth day after the day on which the notice of such annual meeting was made by mail or Public Disclosure; and

(ii) in the case of the nomination of a Stockholder Nominee for election to the Board at a special meeting of Stockholders, no earlier than one hundred and twenty (120) days before and no later than the later of ninety (90) days before such special meeting and the tenth day after the day on which the notice of such special meeting was made by mail or Public Disclosure.

(d) Notwithstanding anything to the contrary, if the number of Directors to be elected to the Board at a meeting of Stockholders is increased and there is no Public Disclosure by the Corporation naming the nominees for the additional directorships at least one hundred (100) days before the first anniversary of the preceding year's annual meeting, a Notice of Nomination shall also be considered timely, but only with respect to nominees for the additional directorships, if it shall be delivered personally and received at the Office of the Corporation, addressed to the attention of the Secretary, no later than the close of business on the tenth day following the day on which such Public Disclosure is first made by the Corporation.

(e) In no event shall an adjournment, postponement or deferral, or Public Disclosure of an adjournment, postponement or deferral, of an annual or special meeting commence a new time period (or extend any time period) for the giving of the Notice of Nomination.

(f) The Notice of Nomination shall set forth:

(i) the Stockholder Information with respect to each Nominating Stockholder and Stockholder Associated Person;

(ii) a representation to the Corporation that each Nominating Stockholder is a holder of record of stock of the Corporation entitled to vote at the meeting and intends to appear in person or by proxy at the meeting to propose such nomination;

(iii) all information regarding each Stockholder Nominee and Stockholder Associated Person that would be required to be disclosed in a solicitation of proxies subject to Section 14 of the Exchange Act, the written consent of each Stockholder Nominee to being named in a proxy statement as a nominee and to serve if elected and a completed signed questionnaire, representation and agreement required by Section 3.04;

(iv) a description of all direct and indirect compensation and other material monetary agreements, arrangements and understandings during the past three years, and any other material relationships, between or among a Nominating Stockholder, Stockholder Associated Person or their respective associates, or others acting in concert therewith, including all information that would be required to be disclosed pursuant to Rule 404 promulgated under Regulation S-K if the Nominating Stockholder, Stockholder Associated Person or any person acting in concert therewith were the "registrant" for purposes of such rule and the Stockholder Nominee were a director or executive of such registrant;

(v) Stockholder Information with respect to any stock or other interests of the Corporation held by members of the Nominating Stockholder's or its Stockholder Associated Person's immediate family sharing the same household;

(vi) a representation to the Corporation as to whether each Nominating Stockholder intends (A) to deliver a proxy statement and form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve the nomination or (B) otherwise to solicit proxies from Stockholders in support of such nomination;

(vii) all other information that would be required to be filed with the SEC if the Nominating Stockholders and Stockholder Associated Persons were participants in a solicitation subject to Section 14 of the Exchange Act; and

(viii) a representation and covenant for the benefit of the Corporation that the Nominating Stockholders shall provide any other information reasonably requested by the Corporation.

(g) The Nominating Stockholders shall also provide any other information reasonably requested by the Corporation within ten (10) business days after such request.

(h) In addition, the Nominating Stockholders shall further update and supplement the information provided to the Corporation in the Notice of Nomination or upon the Corporation's request pursuant to Section 3.03(g) as needed, so that such information shall be true and correct as of the record date for the meeting and as of the date that is five (5) business days before the meeting or any adjournment or postponement thereof. Such update and supplement must be delivered personally or mailed to, and received at, the Office of the Corporation, addressed to the Secretary, by no later than five (5) business days after the record date for the meeting (in the case of the update and supplement required to be made as of the record date), and not later than two (2) business days before the date for the meeting (in the case of the update and supplement required to be made as of five (5) business days before the meeting or any adjournment or postponement or postponement thereof).

(i) The person presiding over the meeting shall, if the facts warrant, determine and declare to the meeting, that the nomination was not made in accordance with the procedures set forth in this Section 3.03, and, if he or she should so determine, he or she shall so declare to the meeting and the defective nomination shall be disregarded.

(j) If the Stockholder (or a qualified representative of the Stockholder) does not appear at the applicable Stockholder meeting to nominate the Stockholder Nominees, such nomination shall be disregarded and such business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. For purposes of this Section 3.03, to be considered a qualified representative of the Stockholder, a person must be a duly authorized officer, manager or partner of such Stockholder or must be authorized by a writing executed by such Stockholder or an electronic transmission delivered by such Stockholder to act for such Stockholder as proxy at the meeting of Stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of Stockholders.

(k) Nothing in this Section 3.03 shall be deemed to affect any rights of the holders of any series of preferred stock of the Corporation pursuant to any applicable provision of the Certificate of Incorporation.

*Nominee and Director Qualifications.* Unless the Board determines otherwise, to be eligible to be a Section 3.04. nominee for election or reelection as a Director, a person must deliver (in accordance with the time periods prescribed for delivery of notice by the Board) to the Secretary at the Office of the Corporation a written questionnaire with respect to the background and qualification of such person and the background of any other person or entity on whose behalf the nomination is being made (which questionnaire shall be provided by the Secretary upon written request) and a written representation and agreement (in the form provided by the Secretary upon written request) that such person (a) is not and will not become a party to (i) any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such person will act or vote as a Director on any issue or question (a "Voting Commitment") that has not been disclosed to the Corporation or (ii) any Voting Commitment that could limit or interfere with such person's ability to comply with such person's fiduciary duties as a Director under applicable law, (b) is not and will not become a party to any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a Director that has not been disclosed therein, and (c) in such person's individual capacity and on behalf of any person or entity on whose behalf the nomination is being made, would be in compliance, and will comply with all applicable publicly disclosed corporate governance, conflict of interest, confidentiality and stock ownership and trading and other policies and guidelines of the Corporation that are applicable to Directors.

Section 3.05. *Resignation*. Any Director may resign at any time by notice given in writing or by electronic transmission to the Corporation. Such resignation shall take effect at the date of receipt of such notice or at such later time as is therein specified, and, unless otherwise specified in such resignation, the acceptance of such resignation shall not be necessary to make it effective. If no such specification is made, the resignation shall be deemed effective at the time of delivery of the resignation to the Corporation. When one or more Directors shall resign from the Board, effective at a future date, a majority of the Directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each Director so chosen shall hold office for the unexpired portion of the term of the Director whose place shall be vacated and until his or her successor shall have been duly elected and qualified.

Section 3.06. *Compensation*. Each Director, in consideration of his or her service as such, shall be entitled to receive from the Corporation such amount per annum or such fees (payable in cash or equity) for attendance at Directors' meetings, or both, as the Board may from time to time determine, together with reimbursement for the reasonable out-of-pocket expenses, if any, incurred by such Director in connection with the performance of his or her duties. Each Director who shall serve as a member of any committee of Directors in consideration of serving as such shall be entitled to such additional amount per annum or such fees for attendance at committee meetings, or both, as the Board may from time to time determine, together with reimbursement for the reasonable out-of-pocket expenses, if any, incurred by such Director in the performance of his or her duties. Nothing contained in this Section 3.06 shall preclude any Director from serving the Corporation or its subsidiaries in any other capacity and receiving proper compensation therefor.

Section 3.07. *Regular Meetings*. Regular meetings of the Board may be held without notice at such times and at such places within or without the State of Delaware as may be determined from time to time by the Board or its Chairman.

Section 3.08. *Special Meetings*. Special meetings of the Board may be held at such times and at such places within or without the State of Delaware as may be determined by the Chairman, or the Chief Executive Officer on at least twenty-four (24) hours' notice to each Director given by one of the means specified in Section 3.11 hereof other than by mail, or on at least three (3) days' notice if given by mail.

Section 3.09. *Telephone Meetings*. Board or Board committee meetings may be held by means of telephone conference or other communications equipment by means of which all persons participating in the meeting can hear each other. Participation by a Director in a meeting pursuant to this Section 3.09 shall constitute presence in person at such meeting.

Section 3.10. *Adjourned Meetings*. A majority of the Directors present at any meeting of the Board, including an adjourned meeting, whether or not a quorum is present, may adjourn and reconvene such meeting to another time and place. At least twenty-four (24) hours' notice of any adjourned meeting of the Board shall be given to each Director whether or not present at the time of the adjournment, if such notice shall be given by one of the means specified in Section 3.11 hereof other than by mail, or at least three (3) days' notice if by mail. Any business may be transacted at an adjourned meeting that might have been transacted at the meeting as originally called.

Section 3.11. *Notice Procedure*. Subject to Section 3.08 and 3.12 hereof, whenever notice is required to be given to any Director by applicable law, the Certificate of Incorporation or these By-laws, such notice shall be deemed given effectively if given in person or by telephone, mail or electronic mail addressed to such Director at such Director's address or email address, as applicable, as it appears on the records of the Corporation, facsimile or by other means of electronic transmission.

Section 3.12. *Waiver of Notice*. Whenever the giving of any notice to Directors is required by applicable law, the Certificate of Incorporation or these By-laws, a waiver thereof, in writing signed by the Director entitled to the notice, whether before or after such notice is required, shall be deemed equivalent to notice. Attendance by a Director at a meeting shall constitute a waiver of notice of such meeting except when the Director attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business on the ground that the meeting was not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special Board or committee meeting need be specified in any waiver of notice.

Section 3.13. *Organization*. At each meeting of the Board, the Chairman or, in the absence of the Chairman, the Lead Independent Director has not been appointed or is absent, the Chief Executive Officer or, in the absence of the Chairman, the Chief Executive Officer, another Director selected by the Board shall preside. The Secretary shall act as secretary at each meeting of the Board. If the Secretary is absent from any meeting of the Board, an Assistant Secretary shall perform the duties of secretary at such meeting; and in the absence from any such meeting of the Secretary and all Assistant Secretaries, the person presiding at the meeting may appoint any person to act as secretary of the meeting.

Section 3.14. *Quorum of Directors*. The presence in person of a majority of the total members of the Board shall be necessary and sufficient to constitute a quorum for the transaction of business at any meeting of the Board.

Section 3.15. *Action by Majority Vote*. Except as otherwise expressly required by these By-laws, or the Certificate of Incorporation, the vote of a majority of the Directors present at a meeting at which a quorum is present shall be the act of the Board; provided that to the extent one or more Directors recuses himself or herself from an act, the act of a majority of the remaining Directors present shall be the act of the Board.

Section 3.16. *Action Without Meeting.* Unless otherwise restricted by these By-laws, any action required or permitted to be taken at any meeting of the Board or of any committee thereof may be taken without a meeting if all Directors or members of such committee, as the case may be, consent thereto in writing or by electronic transmission, and the writings or electronic transmissions are filed with the minutes of proceedings of the Board or committee.

Section 3.17. *Chairman and Lead Independent Director*. The Chairman, if appointed and when present, shall preside at all meetings of the Stockholders and the Board. The Chairman shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers, as the Board shall designate from time to time. The Chairman, or if the Chairman is not an independent director, one of the independent directors, may be designated by the Board as lead independent director to serve until replaced by the Board ("**Lead Independent Director**"). The Lead Independent Director will perform such other duties as may be established or delegated by the Board.

# ARTICLE 4

## Committees of the Board

The Board may, by resolution, designate one or more committees, each committee to consist of one or more of the Directors of the Corporation. The Board may, by resolution, adopt charters for one or more of such committees. The Board may designate one or more Directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of such committee. If a member of a committee shall be absent from any meeting, or disqualified from voting thereat, the remaining member or members present at the meeting and not disgualified from voting, whether or not such member or members constitute a quorum, may, by a unanimous vote, appoint another member of the Board to act at the meeting in the place of any such absent or disgualified member. Any such committee, to the extent permitted by applicable law, and to the extent provided in the resolution of the Board designating such committee or the charter for such committee, shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the Corporation and may authorize the seal of the Corporation to be affixed to all papers that may require it to the extent so authorized by the Board. The Board may remove any Director from any committee at any time, with or without cause. Unless the Board provides otherwise, at all meetings of such committee, a majority of the then authorized members of the committee shall constitute a guorum for the transaction of business, and the vote of a majority of the members of the committee present at any meeting at which there is a quorum shall be the act of the committee. Each committee shall keep regular minutes of its meetings. Unless the Board provides otherwise, each committee designated by the Board may make, alter and repeal rules and procedures for the conduct of its business. In the absence of such rules and procedures, each committee shall conduct its business in the same manner as the Board conducts its business pursuant to Article 3.

## ARTICLE 5

#### Officers

Section 5.01. *Positions; Election.* The Board may from time to time elect officers of the Corporation, which may include a Chief Executive Officer, President, Chief Financial Officer, Vice Presidents, Secretary, Treasurer and any other officers as it may deem proper or may delegate to any elected officer of the Corporation the power to appoint and remove any such officers and to prescribe their respective terms of office, authorities and duties. Any number of offices may be held by the same person. Should the Corporation or any of its Subsidiaries enter into any management services or similar agreement with another entity (each as may be amended, supplemented, restated or replaced from time to time), the officers of the Corporation may be the officers or employees of such entity to the extent permitted by applicable law.

Section 5.02. *Term of Office*. Each officer of the Corporation shall hold office for such terms as may be determined by the Board or, except with respect to his or her own office, the Chief Executive Officer, or until such officer's successor is elected and qualifies or until such officer's earlier death, resignation or removal. Any officer may resign at any time upon written notice to the Corporation. Such resignation shall take effect at the date of receipt of such notice or at such later time as is therein specified, and, unless otherwise specified, the acceptance of such resignation shall not be necessary to make it effective. The resignation of an officer shall be without prejudice to the contract rights of the Corporation, if any. Any officer may be removed at any time with or without cause by the Board or, in the case of appointed officers, by any elected officer upon whom such power of removal shall have been conferred by the Board. Any vacancy occurring in any office of the Corporation may be filled by the Board or, in the case of appointed officer upon whom such power of appointent shall have been conferred by the Board. The election or appointment of an officer shall not of itself create contract rights.

Section 5.03. *Chief Executive Officer*. The Chief Executive Officer shall have general supervision over, and direction of, the business and affairs of the Corporation, subject, however, to the control of the Board and of any duly authorized committee of the Board. The Chief Executive Officer shall preside at all meetings of the Stockholders and at all meetings of the Board at which the Chairman and, with respect to meetings of the Board, the Lead Independent Director (if there be one), are not present. The Chief Executive Officer may sign and execute in the name of the Corporation deeds, mortgages, bonds, contracts and other instruments, except in cases in which the signing and execution thereof shall be expressly delegated by resolution of the Board or by these By-laws to some other officer or agent of the Corporation, or shall be required by applicable law otherwise to be signed or executed and, in general, the Chief Executive Officer shall perform all duties incident to the office of Chief Executive Officer of a corporation and such other duties as may be determined from time to time by the Board.

Section 5.04. *President*. The President shall have duties incident to the office of President, and any other duties as may from time to time be assigned to the President by the Chief Executive Officer (if the President and Chief Executive Officer are not the same person) or the Board and subject to the control of the Chief Executive Officer (if the President and Chief Executive Officer are not the same person) and the Board in each case. The President shall preside at all meetings of the Stockholders at which the Chairman, and the Chief Executive Officer are not present. The President may sign and execute in the name of the Corporation deeds, mortgages, bonds, contracts and other instruments, except in cases in which the signing and execution thereof shall be expressly delegated by the Board or by these By-laws to some other officer or agent of the Corporation, or shall be required by applicable law otherwise to be signed or executed.

Section 5.05. *Vice Presidents*. Vice Presidents shall have the duties incident to the office of Vice President and any other duties that may from time to time be assigned to the Vice President by the Chief Executive Officer, the President or the Board. A Vice President shall preside at all meetings of the Stockholders at which the Chairman, the Chief Executive Officer and the President are not present. Any Vice President may sign and execute in the name of the Corporation deeds, mortgages, bonds, contracts or other instruments, except in cases in which the signing and execution thereof shall be expressly delegated by the Board or by these By-laws to some other officer or agent of the Corporation, or shall be required by applicable law otherwise to be signed or executed.

Section 5.06. *Chief Financial Officer*. The Chief Financial Officer shall keep or cause to be kept the books of account of the Corporation in a thorough and proper manner and shall render statements of the financial affairs of the Corporation in such form and as often as required by the Board or the Chief Executive Officer, or if no Chief Executive officer is then serving, the President. The Chief Financial Officer, subject to the order of the Board, shall have the custody of all funds and securities of the Corporation. The Chief Financial Officer shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board or the Chief Executive Officer, or if no Chief Executive Officer is then serving, the President shall designate from time to time. To the extent that a Chief Financial Officer has been appointed and no Treasurer has been appointed, all references in these Bylaws to the Treasurer shall be deemed references to the Chief Financial Officer, and each Treasurer and Perform the duties of the Chief Financial Officer in the absence or disability of the Chief Financial Officer, and each Treasurer and shall also perform such other duties and have such other duties and have such other duties commonly incident to the officer, or if no Chief Executive Officer, and each Treasurer and Assistant Treasurer and each controller and assistant controller shall perform other duties commonly incident to the office and shall also perform such other duties and have such other powers as the Board or the Chief Executive Officer, or if no Chief Executive Officer, or if no Chief Executive Officer is then serving, the President shall designate from time to time.

Section 5.07. *Secretary*. The Secretary shall attend all meetings of the Board and of the Stockholders, record all the proceedings of the meetings of the Board and of the Stockholders in a book to be kept for that purpose and perform like duties for committees of the Board, when required. The Secretary shall give, or cause to be given, notice of all special meetings of the Board and of the Stockholders and perform such other duties as may be prescribed by the Board, the Chief Executive Officer or the President. The Secretary shall have custody of the corporate seal of the Corporation and the Secretary or an Assistant Secretary shall have authority to affix the same on any instrument that may require it, and when so affixed, the seal may be attested by the signature of the Secretary or by the signature of such Assistant Secretary. The Board may give general authority to any other officer to affix the seal of the Corporation and to attest the same by such officer's signature. The Secretary or an Assistant Secretary may also attest all instruments signed by the Executive Chairman, Chief Executive Officer, President or any Vice President. The Secretary shall have charge of all the books, records and papers of the Corporation relating to its organization and management, see that the reports, statements and other documents required by applicable law are properly kept and filed and, in general, perform all duties incident to the office of secretary of a corporation and such other duties as may from time to time be assigned to the Secretary by the Board, the Chief Executive Officer or the President.

Section 5.08. *Treasurer*. The Treasurer shall have charge and custody of, and be responsible for, all funds, securities and notes of the Corporation, receive and give receipts for moneys due and payable to the Corporation from any sources whatsoever; deposit all such moneys and valuable effects in the name and to the credit of the Corporation in such depositaries as may be designated by the Board, against proper vouchers, cause such funds to be disbursed by checks or drafts on the authorized depositaries of the Corporation signed in such manner as shall be determined by the Board and be responsible for the accuracy of the amounts of all moneys so disbursed, regularly enter or cause to be entered in books or other records maintained for the purpose full and adequate account of all moneys received or paid for the account of the Corporation, have the right to require from time to time reports or statements giving such information as the Treasurer may desire with respect to any and all financial transactions of the Corporation from the officers or agents transacting the same, render to the Chief Executive Officer, the President or the Board, whenever the Chief Executive Officer, the President or the Board shall require the Treasurer so to do, an account of the financial condition of the Corporation and of all financial transactions of the Corporation, disburse the funds of the Corporation as ordered by the Board and, in general, perform all duties incident to the office of Treasurer of a corporation and such other duties as may from time to time be assigned to the Treasurer by the Board, the Chief Executive Officer or the President.

Section 5.09. *Assistant Secretaries and Assistant Treasurers*. Assistant Secretaries and Assistant Treasurers shall perform such duties as shall be assigned to them by the Secretary or by the Treasurer, respectively, or by the Board, the Chief Executive Officer or the President.

### ARTICLE 6

#### GENERAL PROVISIONS

Section 6.01. *Certificates Representing Shares*. The shares of stock of the Corporation may be represented by certificates or all of such shares shall be uncertificated shares that may be evidenced by a book-entry system maintained by the registrar of such stock, or a combination of both. If shares are represented by certificates (if any), such certificates shall be in the form approved by the Board. The certificates representing shares of stock of each class shall be signed by, or in the name of, the Corporation by the Chief Executive Officer, the President or any Vice President, the Chief Financial Officer and by the Secretary, any Assistant Secretary, the Treasurer or any Assistant Treasurer. Any or all such signatures may be facsimiles. Although any officer, transfer agent or registrar whose manual or facsimile signature is affixed to such a certificate ceases to be such officer, transfer agent or registrar before such certificate has been issued, it may nevertheless be issued by the Corporation with the same effect as if such officer, transfer agent or registrar were still such at the date of its issue.

Section 6.02. *Transfer and Registry Agents*. The Corporation may from time to time maintain one or more transfer offices or agents and registry offices or agents at such place or places as may be determined from time to time by the Board.

Section 6.03. *Lost, Stolen or Destroyed Certificates.* The Corporation may issue a new certificate of stock in the place of any certificate theretofore issued by it, alleged to have been lost, stolen or destroyed, and the Corporation may require the owner of the lost, stolen or destroyed certificate or his legal representative to give the Corporation a bond sufficient to indemnify it against any claim that may be made against it on account of the alleged loss, theft or destruction of any such certificate or the issuance of such new certificate.

Section 6.04. *Form of Records*. Any records maintained by the Corporation in the regular course of its business, including its stock ledger, books of account and minute books, may be maintained on any information storage device or method; *provided* that the records so kept can be converted into clearly legible paper form within a reasonable time. The Corporation shall so convert any records so kept upon the request of any person entitled to inspect such records pursuant to applicable law.

Section 6.05. *Seal*. The corporate seal shall have the name of the Corporation inscribed thereon and shall be in such form as may be approved from time to time by the Board. The seal may be used by causing it or a facsimile thereof to be impressed or affixed or otherwise reproduced.

Section 6.06. Fiscal Year. The fiscal year of the Corporation shall be determined by the Board.

Section 6.07. *Amendments*. These By-laws may be altered, amended or repealed in accordance with the Certificate of Incorporation and the General Corporation Law.

Section 6.08. *Conflict with Applicable Law or Certificate of Incorporation*. These By-laws are adopted subject to any applicable law and the Certificate of Incorporation. Whenever these By-laws may conflict with any applicable law or the Certificate of Incorporation, such conflict shall be resolved in favor of such law or the Certificate of Incorporation.

## ARTICLE 7

### INDEMNIFICATION

Section 7.01. *Directors and Executive Officers*. The Corporation shall indemnify its Directors and executive officers (for the purposes of this Article 7, "executive officers" shall have the meaning defined in Rule 3b-7 promulgated under the Exchange Act) to the extent not prohibited by the General Corporation Law or any other applicable law; provided, however, that the Corporation may modify the extent of such indemnification by individual contracts with its Directors and executive officers; and, provided, further, that the Corporation shall not be required to indemnify any Director or executive officer in connection with any proceeding (or part thereof) initiated by such person unless (i) such indemnification is expressly required to be made by law, (ii) the proceeding was authorized by the Board, (iii) such indemnification is provided by the Corporation, in its sole discretion, pursuant to the powers vested in the Corporation under the General Corporation Law or any other applicable law or (iv) such indemnification is required to be made under Section 7.04.

Section 7.02. *Other Officers, Employees and Other Agents.* The Corporation shall have power to indemnify its other officers, employees and other agents as set forth in the General Corporation Law or any other applicable law. The Board shall have the power to delegate the determination of whether indemnification shall be given to any such person except executive officers to such officers or other persons as the Board shall determine.

Section 7.03. *Expenses*. The Corporation shall advance to any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was a Director or executive officer, of the Corporation, or is or was serving at the request of the Corporation as a Director or executive officer of another Corporation, partnership, joint venture, trust or other enterprise, prior to the final disposition of the proceeding, promptly following request therefor, all expenses incurred by any Director or executive officer in connection with such proceeding provided, however, that if the General Corporation Law requires, an advancement of expenses incurred by a Director or executive officer in his or her capacity as a Director or executive officer (and not in any other capacity in which service was or is rendered by such indemnitee, including, without limitation, service to an employee benefit plan) shall be made only upon delivery to the Corporation of an undertaking (hereinafter an "undertaking"), by or on behalf of such indemnitee, to repay all amounts so advanced if it shall ultimately be determined by final judicial decision from which there is no further right to appeal (hereinafter a "final adjudication") that such indemnitee is not entitled to be indemnified for such expenses under this section or otherwise.

Notwithstanding the foregoing, unless otherwise determined pursuant to Section 7.05, no advance shall be made by the Corporation to an executive officer of the Corporation (except by reason of the fact that such executive officer is or was a director of the Corporation in which event this paragraph shall not apply) in any action, suit or proceeding, whether civil, criminal, administrative or investigative, if a determination is reasonably and promptly made (i) by a majority vote of Directors who were not parties to the proceeding, even if not a quorum, or (ii) by a committee of such Directors designated by a majority vote of such Directors, even though less than a quorum, or (iii) if there are no such Directors, or such Directors so direct, by independent legal counsel in a written opinion, that the facts known to the decision-making party at the time such determination is made demonstrate clearly and convincingly that such person acted in bad faith or in a manner that such person did not believe to be in or not opposed to the best interests of the Corporation.

Section 7.04. *Enforcement.* Without the necessity of entering into an express contract, all rights to indemnification and advances to directors and executive officers under this Bylaw shall be deemed to be contractual rights and be effective to the same extent and as if provided for in a contract between the Corporation and the director or executive officer. Any right to indemnification or advances granted by this section to a Director or executive officer shall be enforceable by or on behalf of the person holding such right in any court of competent jurisdiction if (i) the claim for indemnification or advances is denied, in whole or in part, or (ii) no disposition of such claim is made within ninety (90) days of request therefor. To the extent permitted by law, the claimant in such enforcement action, if successful in whole or in part, shall be entitled to be paid also the expense of prosecuting the claim. In connection with any claim for indemnification, the Corporation shall be entitled to raise as a defense to any such action that the claimant has not met the standards of conduct that make it permissible under the General Corporation Law or any other applicable law for the Corporation to indemnify the claimant for the amount claimed. In connection with any claim by an executive officer of the Corporation (except in any action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such executive officer is or was a director of the Corporation) for advances, the Corporation shall be entitled to raise a defense as to any such action clear and convincing evidence that such person acted in bad faith or in a manner that such person did not believe to be in or not opposed to the best interests of the Corporation, or with respect to any criminal action or proceeding that such person acted without reasonable cause to believe that his or her conduct was lawful. Neither the failure of the Corporation (including its Board, independent legal counsel or its stockholders) to have made a determination prior to the commencement of such action that indemnification of the claimant is proper in the circumstances because he has met the applicable standard of conduct set forth in the General Corporation Law or any other applicable law, nor an actual determination by the Corporation (including its Board, independent legal counsel or its stockholders) that the claimant has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that claimant has not met the applicable standard of conduct. In any suit brought by a Director or executive officer to enforce a right to indemnification or to an advancement of expenses hereunder, the burden of proving that the Director or executive officer is not entitled to be indemnified, or to such advancement of expenses, under this section or otherwise shall be on the Corporation.

Section 7.05. *Non-Exclusivity of Rights*. The rights conferred on any person by this Bylaw shall not be exclusive of any other right which such person may have or hereafter acquire under any applicable statute, provision of the Certificate of Incorporation, Bylaws, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in his or her official capacity and as to action in another capacity while holding office. The Corporation is specifically authorized to enter into individual contracts with any or all of its Directors, officers, employees or agents respecting indemnification and advances, to the fullest extent not prohibited by the General Corporation Law, or by any other applicable law.

Section 7.06. *Survival of Rights*. The rights conferred on any person by this Bylaw shall continue as to a person who has ceased to be a Director, officer, employee or other agent and shall inure to the benefit of the heirs, executors and administrators of such a person.

Section 7.07. *Insurance*. To the fullest extent permitted by the General Corporation Law or any other applicable law, the Corporation, upon approval by the Board, may purchase insurance on behalf of any person required or permitted to be indemnified pursuant to this section.

Section 7.08. *Amendments*. Any repeal or modification of this section shall only be prospective and shall not affect the rights under this Bylaw in effect at the time of the alleged occurrence of any action or omission to act that is the cause of any proceeding against any agent of the Corporation.

Section 7.09. *Saving Clause*. If this Bylaw or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Corporation shall nevertheless indemnify each Director and executive officer to the full extent not prohibited by any applicable portion of this section that shall not have been invalidated, or by any other applicable law. If this section shall be invalid due to the application of the indemnification provisions of another jurisdiction, then the Corporation shall indemnify each Director and executive officer to the full extent under any other applicable law.

Section 7.10. *Certain Definitions*. For the purposes of this Article 7, the following definitions shall apply:

(a) The term "proceeding" shall be broadly construed and shall include, without limitation, the investigation, preparation, prosecution, defense, settlement, arbitration and appeal of, and the giving of testimony in, any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative.

(b) The term "expenses" shall be broadly construed and shall include, without limitation, court costs, attorneys' fees, witness fees, fines, amounts paid in settlement or judgment and any other costs and expenses of any nature or kind incurred in connection with any proceeding.

(c) The term the "Corporation" shall include, in addition to the resulting Corporation, any constituent Corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent Corporation, or is or was serving at the request of such constituent Corporation as a director, officer, employee or agent of another Corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under the provisions of this section with respect to the resulting or surviving Corporation as he would have with respect to such constituent Corporation if its separate existence had continued.

(d) References to a "Director," "executive officer," "officer," "employee," or "agent" of the Corporation shall include, without limitation, situations where such person is serving at the request of the Corporation as, respectively, a director, executive officer, officer, employee, trustee or agent of another Corporation, partnership, joint venture, trust or other enterprise.

(e) References to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; and references to "serving at the request of the Corporation" shall include any service as a Director, officer, employee or agent of the Corporation which imposes duties on, or involves services by, such Director, officer, employee, or agent with respect to an employee benefit plan, its participants, or beneficiaries; and a person who acted in good faith and in a manner he reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Corporation" as referred to in this section.

NUMBER

NUMBER C-SHARES SEE REVERSE FOR CERTAIN DEFINITIONS CUSIP 92971A 109

## WM Technology, Inc.

## INCORPORATED UNDER THE LAWS OF THE STATE OF DELAWARE COMMON STOCK

This Certifies that

is the owner of

FULLY PAID AND NON-ASSESSABLE SHARES OF THE PAR VALUE OF \$0.0001 EACH OF THE COMMON STOCK OF

## WM Technology, Inc. (THE "COMPANY")

transferable on the books of the Company in person or by duly authorized attorney upon surrender of this certificate properly endorsed.

This certificate is not valid unless countersigned by the Transfer Agent and registered by the Registrar.

Witness the seal of the Company and the facsimile signatures of its duly authorized officers.

Chief Executive Officer

[Corporate Seal] Delaware

Chief Financial Officer

#### WM Technology, Inc.

The Company will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof of the Company and the qualifications, limitations, or restrictions of such preferences and/or rights. This certificate and the shares represented thereby are issued and shall be held subject to all the provisions of the Company's amended and restated certificate of incorporation and all amendments thereto and resolutions of the Board of Directors providing for the issue of securities (copies of which may be obtained from the secretary of the Company), to all of which the holder of this certificate by acceptance hereof assents. The following abbreviations, when used in the inscription on the face of this certificate, shall be construed as though they were written out in full according to applicable laws or regulations:

TEN COM	 as tenants in common	UNIF GIFT MIN ACT—	Custodian
TEN ENT	 as tenants by the entireties		(Cust)
JT TEN	 as joint tenants with right of survivors	oint tenants with right of survivorship and not as tenants in common	
			(Minor)
			(State)

Additional abbreviations may also be used though not in the above list.

For value received, hereby sells, assigns and transfers unto

### (PLEASE INSERT SOCIAL SECURITY OR OTHER IDENTIFYING NUMBER(S) OF ASSIGNEE(S))

(PLEASE PRINT OR TYPEWRITE NAME(S) AND ADDRESS(ES), INCLUDING ZIP CODE, OF ASSIGNEE(S))

shares of the capital stock represented by the within Certificate, and hereby irrevocably constitutes and appoints

Attorney to transfer the said stock on the books of the within named Company with full power of substitution in the premises.

Dated:

NOTICE: THE SIGNATURE(S) TO THIS ASSIGNMENT MUST CORRESPOND WITH THE NAME AS WRITTEN UPON THE FACE OF THE CERTIFICATE IN EVERY PARTICULAR, WITHOUT ALTERATION OR ENLARGEMENT OR ANY CHANGE WHATEVER.

Signature(s) Guaranteed:

By

THE SIGNATURE(S) MUST BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15 (OR ANY SUCCESSOR RULE).

Exhibit 4.2

[Form of Warrant Certificate]

[FACE]

Number

## WARRANTS THIS WARRANT SHALL BE VOID IF NOT EXERCISED PRIOR TO THE EXPIRATION OF THE EXERCISE PERIOD PROVIDED FOR IN THE WARRANT AGREEMENT DESCRIBED BELOW

**WM TECHNOLOGY, INC.** Incorporated Under the Laws of the State of Delaware

CUSIP 92971A 117

## Warrant Certificate

**This Warrant Certificate certifies that**, or registered assigns, is the registered holder of warrant(s) evidenced hereby (the "Warrants" and each, a "Warrant") to purchase Class A Common Stock, \$0.0001 par value ("Class A Common Stock"), of WM Technology, Inc., a Delaware Corporation (the "Company"). Each Warrant entitles the holder, upon exercise during the period set forth in the Warrant Agreement referred to below, to receive from the Company that number of fully paid and non-assessable shares of Class A Common Stock as set forth below, at the exercise price (the "Exercise Price") as determined pursuant to the Warrant Agreement, payable in lawful money (or through "cashless exercise" as provided for in the Warrant Agreement) of the United States of America upon surrender of this Warrant Certificate and payment of the Exercise Price at the office or agency of the Warrant Agent referred to below, subject to the conditions set forth herein and in the Warrant Agreement. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Each whole Warrant is initially exercisable for one fully paid and non-assessable share of Class A Common Stock. No fractional shares will be issued upon exercise of any Warrant. If, upon the exercise of Warrants, a holder would be entitled to receive a fractional interest in a share of Class A Common Stock, the Company will, upon exercise, round down to the nearest whole number the number of shares of Class A Common Stock to be issued to the Warrant holder. The number of shares of Class A Common Stock to adjustment upon the occurrence of certain events as set forth in the Warrant Agreement.

The initial Exercise Price per one share of Class A Common Stock for any Warrant is equal to \$11.50 per share. The Exercise Price is subject to adjustment upon the occurrence of certain events as set forth in the Warrant Agreement.

Subject to the conditions set forth in the Warrant Agreement, the Warrants may be exercised only during the Exercise Period and to the extent not exercised by the end of such Exercise Period, such Warrants shall become void.

Reference is hereby made to the further provisions of this Warrant Certificate set forth on the reverse hereof and such further provisions shall for all purposes have the same effect as though fully set forth at this place.

This Warrant Certificate shall not be valid unless countersigned by the Warrant Agent, as such term is used in the Warrant Agreement.

This Warrant Certificate shall be governed by and construed in accordance with the internal laws of the State of New York.

WM TECHNOLOGY, INC.

By:				
Name:				
Title:				
•				

CONTINENTAL STOCK TRANSFER & TRUST COMPANY, AS WARRANT AGENT

By:		
Name:		
Title:		

### [Form of Warrant Certificate]

## [Reverse]

The Warrants evidenced by this Warrant Certificate are part of a duly authorized issue of Warrants entitling the holder on exercise to receive Class A ordinary shares and are issued or to be issued pursuant to a Warrant Agreement dated as of August 7, 2019 (the "*Warrant Agreement*"), duly executed and delivered by the Company to Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "*Warrant Agent*"), which Warrant Agreement is hereby incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, limitation of rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the holders (the words "*holders*" or "*holder*" meaning the Registered Holders or Registered Holder, respectively) of the Warrants. A copy of the Warrant Agreement may be obtained by the holder hereof upon written request to the Company. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Warrants may be exercised at any time during the Exercise Period set forth in the Warrant Agreement. The holder of Warrants evidenced by this Warrant Certificate may exercise them by surrendering this Warrant Certificate, with the form of Election to Purchase set forth hereon properly completed and executed, together with payment of the Exercise Price as specified in the Warrant Agreement (or through "*cashless exercise*" as provided for in the Warrant Agreement) at the principal corporate trust office of the Warrant Agent. In the event that upon any exercise of Warrants evidenced hereby the number of Warrants exercised shall be less than the total number of Warrants evidenced hereby, there shall be issued to the holder hereof or his, her or its assignee, a new Warrant Certificate evidencing the number of Warrants not exercised.

Notwithstanding anything else in this Warrant Certificate or the Warrant Agreement, no Warrant may be exercised unless at the time of exercise (i) a registration statement covering the issuance of the shares of Class A Common Stock to be issued upon exercise is effective under the Securities Act and (ii) a prospectus thereunder relating to the shares of Class A Common Stock is current, except through "*cashless exercise*" as provided for in the Warrant Agreement.

The Warrant Agreement provides that upon the occurrence of certain events the number of shares of Class A Common Stock issuable upon exercise of the Warrants set forth on the face hereof may, subject to certain conditions, be adjusted. If, upon exercise of a Warrant, the holder thereof would be entitled to receive a fractional interest in a share of Class A Common Stock, the Company shall, upon exercise, round down to the nearest whole number of shares of Class A Common Stock to be issued to the holder of the Warrant.

Warrant Certificates, when surrendered at the principal corporate trust office of the Warrant Agent by the Registered Holder thereof in person or by legal representative or attorney duly authorized in writing, may be exchanged, in the manner and subject to the limitations provided in the Warrant Agreement, but without payment of any service charge, for another Warrant Certificate or Warrant Certificates of like tenor evidencing in the aggregate a like number of Warrants.

Upon due presentation for registration of transfer of this Warrant Certificate at the office of the Warrant Agent a new Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee(s) in exchange for this Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any tax or other governmental charge imposed in connection therewith.

The Company and the Warrant Agent may deem and treat the Registered Holder(s) hereof as the absolute owner(s) of this Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone), for the purpose of any exercise hereof, of any distribution to the holder(s) hereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary. Neither the Warrants nor this Warrant Certificate entitles any holder hereof to any rights of a shareholder of the Company.

#### **Election to Purchase**

### (To Be Executed Upon Exercise of Warrant)

The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, to receive shares of Class A Common Stock and herewith tenders payment for such shares of Class A Common Stock to the order of WM Technology, Inc. (the "*Company*") in the amount of \$\_\_\_\_\_\_ in accordance with the terms hereof. The undersigned requests that a certificate for such shares of Class A Common Stock be registered in the name of \_\_\_\_\_\_, whose address is \_\_\_\_\_\_. If said number of shares of Class A Common Stock is less than all of the shares of Class A Common Stock be registered in the name of shares of Class A Common Stock is less than all of the shares of Class A Common Stock be registered to shares of Class A Common Stock is less than all of the shares of Class A Common Stock be registered in the remaining balance of such shares of Class A Common Stock is less than all of the shares of Class A Common Stock be registered in the remaining balance of such shares of Class A Common Stock be registered in the remaining balance of such shares of Class A Common Stock be registered in the remaining balance of such shares of Class A Common Stock be registered in the name of \_\_\_\_\_\_, whose address is \_\_\_\_\_\_\_ and that such Warrant Certificate be delivered to \_\_\_\_\_\_\_, whose address is \_\_\_\_\_\_\_.

In the event that the Warrant has been called for redemption by the Company pursuant to <u>Section 6</u> of the Warrant Agreement and the Company has required cashless exercise pursuant to <u>Section 6.3</u> of the Warrant Agreement, the number of shares of Class A Common Stock that this Warrant is exercisable for shall be determined in accordance with <u>subsection 3.3.1(b)</u> and <u>Section 6.3</u> of the Warrant Agreement.

In the event that the Warrant is a Private Placement Warrant that is to be exercised on a "cashless" basis pursuant to <u>subsection 3.3.1(c)</u> of the Warrant Agreement, the number of shares of Class A Common Stock that this Warrant is exercisable for shall be determined in accordance with <u>subsection 3.3.1(c)</u> of the Warrant Agreement.

In the event that the Warrant is to be exercised on a "cashless" basis pursuant to <u>Section 7.4</u> of the Warrant Agreement, the number of shares of Class A Common Stock that this Warrant is exercisable for shall be determined in accordance with <u>Section 7.4</u> of the Warrant Agreement.

In the event that the Warrant may be exercised, to the extent allowed by the Warrant Agreement, through cashless exercise (i) the number of shares of Class A Common Stock that this Warrant is exercisable for would be determined in accordance with the relevant section of the Warrant Agreement which allows for such cashless exercise and (ii) the holder hereof shall complete the following: The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, through the cashless exercise provisions of the Warrant Agreement, to receive shares of Class A Common Stock. If said number of shares is less than all of the shares of Class A Common Stock purchasable hereunder (after giving effect to the cashless exercise), the undersigned requests that a new Warrant Certificate representing the remaining balance of such shares of Class A Common Stock be registered in the name of \_\_\_\_\_\_, whose address is \_\_\_\_\_\_ and that such Warrant Certificate be delivered to \_\_\_\_\_\_\_, whose address is \_\_\_\_\_\_.

[Signature Page Follows]

Date: , 20

(Signature)

(Address)

(Tax Identification Number)

Signature Guaranteed:

THE SIGNATURE(S) MUST BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15 (OR ANY SUCCESSOR RULE) UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED).

Exhibit 10.1

### **EXECUTION COPY**

## **EXCHANGE AGREEMENT**

EXCHANGE AGREEMENT (this "<u>Agreement</u>"), dated as of June 16, 2021, among WM Technology, Inc., a Delaware corporation, WM Holding Company, LLC, a Delaware limited liability company, and the holders from time to time party hereto, other than the Corporation (as defined herein), of LLC Units (as defined herein) from time to time party hereto.

WHEREAS, the parties hereto desire to provide for the exchange of Paired Interests (as defined herein) or Class P Units (as defined herein), after their conversion to Class A Units, for shares of Class A Common Stock (as defined herein), on the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants and undertakings contained herein and for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

## **ARTICLE I**

#### SECTION 1.1 Definitions

The following definitions shall be for all purposes, unless otherwise clearly indicated to the contrary, applied to the terms used in this Agreement.

"<u>Appraiser FMV</u>" means the fair market value of a share of Class A Common Stock as determined by an independent appraiser mutually agreed upon by the Corporation and the relevant Exchanging Member, whose determination shall be final and binding for those purposes for which Appraiser FMV is used in this Agreement. Appraiser FMV shall be the fair market value determined without regard to any discounts for minority interest, illiquidity or other discounts. The cost of any independent appraisal in connection with the determination of Appraiser FMV in accordance with this Agreement shall be borne by OpCo.

"Board" means has the meaning given to such term in the OpCo LLC Agreement.

"<u>Business Day</u>" means a day other than a Saturday, Sunday or other day on which commercial banks in New York, New York, San Francisco, California or Wilmington, Delaware are authorized or required by Law to close.

"<u>Cash Exchange Class A 5-Day VWAP</u>" means the arithmetic average of the VWAP for each of the five (5) consecutive Trading Days ending on the Trading Day immediately prior to the Exchange Notice Date (in the case of an Unrestricted Exchange) or the Exchange Date (in the case of any other Exchange).

"Cash Exchange Notice" has the meaning set forth in Section 2.1(c) of this Agreement.

"<u>Cash Exchange Payment</u>" means with respect to a particular Exchange for which the Corporation has elected to make a Cash Exchange Payment in accordance with <u>Section 2.1(c)</u>:

(a) if the shares of Class A Common Stock trade on a National Securities Exchange or automated or electronic quotation system, an amount of cash equal to the product of: (x) the number of shares of Class A Common Stock that would have been received by the Exchanging Member in the Exchange for that portion of the Exchanged Units subject to the Exchange set forth in the Cash Exchange Notice if OpCo or the Corporation, as applicable, had paid the Stock Exchange Payment with respect to such number of Exchanged Units, and (y) the Cash Exchange Class A 5-Day VWAP; or

(b) if shares of Class A Common Stock are not then traded on a National Securities Exchange or automated or electronic quotation system, as applicable, an amount of cash equal to the product of (x) the number of shares of Class A Common Stock that would have been received by the Exchanging Member in the Exchange for that portion of the Exchanged Units subject to the Exchange set forth in the Cash Exchange Notice if OpCo or the Corporation, as applicable, had paid the Stock Exchange Payment with respect to such number of Exchanged Units, and (y) the Appraiser FMV of one (1) share of Class A Common Stock that would be obtained in an arms-length transaction between an informed and willing buyer and an informed and willing seller, neither of whom is under any compulsion to buy or sell, respectively, and without regard to the particular circumstances of the buyer or seller.

"<u>Change of Control</u>" has the meaning given to such term in the Tax Receivable Agreement, <u>provided</u> that, for the avoidance of doubt, any event that constitutes both a Pubco Offer and a Change of Control of the Corporation shall be considered a Pubco Offer for purposes of this Agreement.

"<u>Class A Common Stock</u>" means the Class A common stock, par value \$0.0001 per share, of the Corporation.

"<u>Class V Common Stock</u>" means the Class V common stock, par value \$0.0001 per share, of the Corporation.

"<u>Class A Unit</u>" means each unit of limited liability interest in OpCo designated as a "Class A Unit" pursuant to the OpCo LLC Agreement.

"<u>Class P Unit</u>" means each unit of limited liability interest in OpCo designated as a "Class P Unit" pursuant to the OpCo LLC Agreement.

"<u>Class P Unit Exchange Rate</u>" means, with respect to any Class P Unit subject to an Exchange Notice, the quotient of (a) the difference between the Per Unit Equity Value on the Exchange Date and the Participation Threshold applicable to such Class P Unit, divided by (b) the Per Unit Equity Value on the Exchange Date.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Corporation</u>" means WM Technology, Inc., a Delaware corporation, and any successor thereto.

"Direct Exchange" has the meaning set forth in Section 2.6 of this Agreement.

"<u>Direct Exchange Election Notice</u>" has the meaning set forth in <u>Section 2.6</u> of this Agreement.

"Exchange" has the meaning set forth in Section 2.1(a) of this Agreement.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exchange Blackout Period" means (i) any "black out" or similar period under the Corporation's policies covering trading in the Corporation's securities to which the applicable Exchanging Member is subject (or will be subject at such time as it owns Class A Common Stock), which period restricts the ability of such Exchanging Member to immediately resell shares of Class A Common Stock to be delivered to such Exchanging Member in connection with a Stock Exchange Payment and (ii) the period of time commencing on (x) the date of the declaration of a dividend by the Corporation and ending on the first day following (y) the record date determined by the board of directors of the Corporation with respect to such dividend declared pursuant to clause (x), which period of time shall be no longer than 10 Business Days; <u>provided</u> that in no event shall an Exchange Blackout Period which respect to clause (ii) of the definition hereof occur more than four (4) times per calendar year.

"Exchange Date" means, in the case of any Unrestricted Exchange, the date that is five (5) Business Days after the date the Exchange Notice is given pursuant to <u>Section 2.1(b)</u>, unless the Exchanging Member submits a written request to extend such date and the Corporation in its sole discretion agrees in writing to such extension, and in any other case, the Quarterly Exchange Date; <u>provided</u>, that if the Exchange Date for any Exchange with respect to which the Corporation elects to make a Stock Exchange Payment would otherwise fall within any Exchange Blackout Period, then the Exchange Date shall occur on the next Business Day following the end of such Exchange Blackout Period.

"<u>Exchange Notice Date</u>" means, with respect to an Exchange, the date the applicable Exchange Notice is delivered in accordance with <u>Section 2.1(b)</u>.

"<u>Exchange Rate</u>" means, at any time, the number of shares of Class A Common Stock for which an Exchanged Unit is entitled to be exchanged at such time. On the date of this Agreement, the Exchange Rate shall be 1 for 1, subject to adjustment pursuant to <u>Section 2.4</u> hereof.

"<u>Exchanged Units</u>" means any Class A Units (including, for the avoidance of doubt, Class A Units received pursuant to <u>Section 2.2(b)(ii)</u> in exchange for Class P Units subject to an Exchange Notice) to be Exchanged for the Cash Exchange Payment or Stock Exchange Payment, as applicable, on the applicable Exchange Date.

"<u>Exchanging Member</u>" means, with respect to any Exchange, the LLC Unitholder exchanging Units pursuant to <u>Section</u> <u>2.1(a)</u> of this Agreement.

"Exchange Notice" has the meaning set forth in <u>Section 2.1(b)</u> of this Agreement.

"LLC Unit" means each Unit as defined in the OpCo LLC Agreement.

"LLC Unitholder" means each holder of one or more LLC Units that may from time to time be a party to this Agreement.

"Managing Member" has the meaning given to such term in the OpCo LLC Agreement.

"<u>National Securities Exchange</u>" means a securities exchange that has registered with the SEC under Section 6 of the Exchange Act.

"<u>OpCo</u>" means WM holding Company, LLC, a Delaware limited liability company, and any successor thereto.

"<u>OpCo LLC Agreement</u>" means the Fourth Amended and Restated Limited Liability Company Agreement of OpCo, dated on or about the date hereof, as such agreement may be amended from time to time.

"Paired Interest" means one Class A Unit and one share of Class V Common Stock.

"Participation Threshold" has the meaning given to such term in the OpCo LLC Agreement.

"<u>Per Unit Equity Value</u>" means, as of any particular time, the amount to which each holder of a Class A Unit would be entitled in respect of such Class A Unit if the aggregate equity value of the Company as of such time (as reasonably determined by the Managing Member based on the public trading price of Class A Common Stock) were distributed to the Members in accordance with Section 4.02 of the OpCo LLC Agreement (assuming for these purposes that any "Tax Distributions" were made pro rata in accordance with "Class A Percentage Interests"/"LTIP Percentage Interests" (as such terms are defined in the OpCo LLC Agreement)).

"<u>Permitted Exchange Event</u>" means any of the following events, which has or is occurring, or is otherwise satisfied, as of the Exchange Date:

(i) The Exchange is part of one or more Exchanges by an LLC Unitholder and any related persons (within the meaning of Section 267(b) or 707(b)(1) of the Code) that is part of a "block transfer" within the meaning of Treasury Regulations Section 1.7704-1(e)(2) (for this purpose, treating the Managing Member as a "general partner" within the meaning of Treasury Regulations Section 1.7704-1(e)(1),

(ii) The Exchange is in connection with a Pubco Offer or Change of Control; <u>provided</u> that any such Exchange pursuant to this clause (ii) shall be effective immediately prior to the consummation of the closing of the Pubco Offer or Change of Control date (and, for the avoidance of doubt, shall not be effective if such Pubco Offer is not consummated or Change of Control does not occur), or

(iii) The Exchange is permitted by the Managing Member, in its sole discretion, in connection with circumstances not otherwise set forth herein, if the Managing Member determines, after consultation with its outside legal counsel and tax advisor, that the Company would not be treated as a "publicly traded partnership" under Section 7704 of the Code (or any successor or similar provision) as a result of or in connection with such Exchange.

"<u>Permitted Transferee</u>" has the meaning given to such term in <u>Section 3.1</u> of this Agreement.

"<u>Person</u>" means any individual, estate, corporation, partnership, limited partnership, limited liability company, limited company, joint venture, trust, unincorporated or governmental organization or any agency or political subdivision thereof.

"<u>Private Placement Safe Harbor</u>" means the "private placement" safe harbor set forth in Treasury Regulations Section 1.7704-1(h)(1).

"Pubco Offer" has the meaning set forth in Section 2.7 of this Agreement.

"Quarterly Exchange Date" means, either (x) for each fiscal quarter, the first (1st) Business Day occurring after the sixtieth (60th) day after the expiration of the applicable Quarterly Exchange Notice Period or (y) such other date as the Corporation shall determine in its sole discretion; <u>provided</u> that such date is at least sixty (60) days after the expiration of the Quarterly Exchange Notice Period; <u>provided further</u> that the Corporation shall use commercially reasonable efforts to ensure that at least one Quarterly Exchange Date occurs each fiscal quarter.

"Quarterly Exchange Notice Period" means, for each fiscal quarter, the period commencing on the third (3rd) Business Day after the day on which the Corporation releases its earnings for the prior fiscal period, beginning with the first such date that falls on or after the waiver or expiration of any contractual lock-up period relating to the shares of the Corporation that may be applicable to an LLC Unitholder (or such other date within such quarter as the Corporation shall determine in its sole discretion) and ending five (5) Business Days thereafter. Notwithstanding the foregoing, the Corporation may change the definition of Quarterly Exchange Notice Period with respect to any Quarterly Exchange Notice Period scheduled to occur in a calendar quarter subsequent to the then-current calendar quarter if (x) the revised definition provides for a Quarterly Exchange Notice Period occurring at least once in each calendar quarter, (y) the first Quarterly Exchange Notice Period pursuant to the revised definition will occur no less than 10 Business Days from the date written notice of such change is sent to each LLC Unitholder (other than the Corporation) and (z) the revised definition, together with the revised Quarterly Exchange Date resulting therefrom, do not materially adversely affect the ability of the LLC Unitholders to exercise their Exchange rights pursuant to this Agreement.

"Redemption" has the meaning set forth in Section 2.1(a) of this Agreement.

"Restricted Retraction Notice" has the meaning set forth in Section 2.1(d) of this Agreement.

"<u>Stock Exchange Payment</u>" means, with respect to the portion of any Exchange for which a Cash Exchange Notice is not delivered by the Corporation, on behalf of OpCo, a number of shares of Class A Common Stock equal to the product of the number of Exchanged Units multiplied by the Exchange Rate.

"<u>Tax Receivable Agreement</u>" means that certain Tax Receivable Agreement, dated as of June 15, 2021, by and among the Corporation and the other parties thereto.

"<u>Trading Day</u>" means a day on which the Nasdaq Stock Market or such other principal United States securities exchange on which shares of Class A Common Stock are listed, quoted or admitted to trading and is open for the transaction of business (unless such trading shall have been suspended for the entire day).

"<u>Unrestricted Exchanges</u>" means any Exchange that is in connection with a Permitted Exchange Event or that occurs during a period in which OpCo meets the requirements of the Private Placement Safe Harbor.

"<u>Unvested Units</u>" has the meaning given to such term in the OpCo LLC Agreement.

"<u>VWAP</u>" means the daily per share volume-weighted average price of shares of Class A Common Stock on the Nasdaq Stock Market or such other principal United States securities exchange on which shares of Class A Common Stock are listed, quoted or admitted to trading, as displayed under the heading "Bloomberg VWAP" on the Bloomberg page designated for shares of Class A Common Stock (or its equivalent successor if such page is not available) in respect of the period from the open of trading on such Trading Day until the close of trading on such Trading Day (or if such volume-weighted average price is unavailable, (a) the per share volume- weighted average price of a share of Class A Common Stock on such Trading Day (determined without regard to afterhours trading or any other trading outside the regular trading session or trading hours), or (b) if such determination is not feasible, the market price per share of Class A Common Stock, in either case as determined by a nationally recognized independent investment banking firm retained in good faith for this purpose by the Managing Member).

#### **ARTICLE II**

## SECTION 2.1 Exchange Procedure

(a) From and after 180 days following the date of the consummation of the transactions described in the Corporation's Registration Statement on Form S-4 (File No. 333-252186), each LLC Unitholder (other than the Corporation) shall be entitled, upon the terms and subject to the conditions hereof, to surrender Paired Interests or Class P Units (other than Unvested Units) to OpCo in exchange for the delivery of the Stock Exchange Payment or, at the election of the Corporation, the Cash Exchange Payment (such exchange, a "<u>Redemption</u>" and, together with a Direct Exchange (as defined below), an "<u>Exchange</u>"); provided, that (absent a waiver by the Managing Member) any such Exchange is for a minimum of the lesser of (i) 10,000 LLC Units (which minimum shall be equitably adjusted in accordance with any adjustments to the Exchange Rate) and (ii) all of the LLC Units (other than Unvested Units) held by such LLC Unitholder.

An LLC Unitholder shall exercise its right to make an Exchange as set forth in Section 2.1(a) above by (b) delivering to OpCo, with a copy to the Corporation, a written election of exchange in respect of the Paired Interests or Class P Units (other than Unvested Units) to be exchanged substantially in the form of Exhibit A hereto (an "Exchange Notice") in accordance with this Section 2.1(b). An LLC Unitholder may deliver an Exchange Notice with respect to an Unrestricted Exchange at any time, and, in any other case, during the Quarterly Exchange Notice Period preceding the desired Exchange Date. An Exchange Notice with respect to an Unrestricted Exchange may specify that the Exchange is to be contingent (including as to timing) upon the consummation of a purchase by another Person (whether in a tender or exchange offer, an underwritten offering or otherwise) of the Class A Common Stock into which the Exchanged Units are exchangeable, or contingent (including as to timing) upon the closing of an announced merger, consolidation or other transaction or event in which such Class A Common Stock would be exchanged or converted or become exchangeable for or convertible into cash or other securities or property. Notwithstanding anything to the contrary contained in this Agreement, if, in connection with an Exchange in accordance with this Section 2.1, a filing is required under the Hart-Scott-Rodino Antitrust Improvements Act of 1976 ("HSR Act"), then the Exchange Date with respect to all Exchanged Units which would be exchanged into shares of Class A Common Stock resulting from such Exchange shall be delayed until the earlier of (i) such time as the required filing under the HSR Act has been made and the waiting period applicable to such Exchange under the HSR Act shall have expired or been terminated or (ii) such filing is no longer required, at which time such Exchange shall automatically occur without any further action by the holders of any such Exchanged Units. Each of the LLC Unitholders and the Corporation agree to promptly take all actions required to make such filing under the HSR Act and the filing fee for such filing shall be paid by OpCo.

(c) Within three (3) Business Days of the giving of an Exchange Notice, the Corporation, on behalf of OpCo, may elect to settle all or a portion of the Exchange in cash in an amount equal to the Cash Exchange Payment (in lieu of Class A Common Stock) by giving written notice of such election to the Exchanging Member within such three (3) Business Day period (such notice, the "<u>Cash Exchange Notice</u>"). The Cash Exchange Notice shall set forth the portion of the Exchanged Units which will be exchanged for cash in lieu of Class A Common Stock. Any portion of the Exchange not settled for a Cash Exchange Payment shall be settled for a Stock Exchange Payment.

The Exchanging Member may elect to retract its Exchange Notice with respect to an Unrestricted (d) Exchange by giving written notice of such election to OpCo, with a copy to the Corporation, no later than (1) Business Day prior to the Exchange Date. Subject to the last two sentences of this <u>Section 2.1(d)</u>, if, in the case of an Exchange that is not an Unrestricted Exchange, the Cash Exchange Class A 5-Day VWAP (determined treating the final date of such period as the Exchange Date) decreases by more than 10% from the Cash Exchange Class A 5-Day VWAP (determined treating the final date of such period as the date of delivery of an Exchange Notice), the Exchanging Member may elect to retract its Exchange Notice by giving written notice of such election (a "Restricted Retraction Notice") to OpCo, with a copy to the Corporation, no later than three (3) Business Days prior to the Exchange Date. The giving of any notice pursuant to this <u>Section 2.1(d)</u> shall terminate all of the Exchanging Member's, the Corporation's and OpCo's rights and obligations under this Article II arising from such retracted Exchange Notice (but not, for the avoidance of doubt, from any Exchange Notice not retracted or that may be delivered in the future). An Exchanging Member may deliver a Restricted Retraction Notice only once in every twelve (12)-month period (and any additional Restricted Retraction Notice delivered by such Exchanging Member within such twelve (12)-month period shall be deemed null and void *ab initio* and ineffective with respect to the revocation of the Exchange specified therein). An Exchanging Member who revokes an Exchange pursuant to a Restricted Retraction Notice may not participate in the Exchange to occur on the next Quarterly Exchange Date immediately following the Quarterly Exchange Date with respect to which the Restricted Retraction Notice pertains.

(e) Notwithstanding anything to the contrary in this Agreement, if the Corporation closes an underwritten distribution of the shares of Class A Common Stock and the LLC Unitholders (other than, or in addition to, the Corporation) were entitled to resell shares of Class A Common Stock in connection therewith (by the exercise by such LLC Unitholders of Exchange rights or otherwise) (a "<u>Secondary Offering</u>"), then, the immediately succeeding Quarterly Exchange Date shall be automatically cancelled and of no force or effect (and no LLC Unitholder shall be entitled to exercise its Exchange right or deliver a Quarterly Exchange Date Notice with respect to an Exchange that is not an Unrestricted Exchange in respect of such Quarterly Exchange Date). Notwithstanding anything to the contrary in this Agreement (a) for so long as the Company does not meet the requirements of the Private Placement Safe Harbor, any Secondary Offering (other than that pursuant to which all Exchanges are Unrestricted Exchanges) shall only be undertaken if, during the applicable taxable year, the total number of Quarterly Exchange Dates and prior Secondary Offerings (other than any pursuant to which all Exchanges are Unrestricted Exchanges occur is three (3) or fewer and (b) the Company and the Corporation shall not be deemed to have failed to comply with their respective obligations under the Corporation's Amended and Restated Registration Rights Agreement, dated as of June 15, 2021, as amended from time to time, if a Secondary Offering cannot be undertaken due to the restriction set forth in the preceding clause (a).

## SECTION 2.2 Exchange Payment

(a) The Exchange shall be consummated on the Exchange Date.

(b) With respect to any Class P Units specified in an Exchange Notice, immediately prior to the transactions contemplated in <u>Section 2.2(c)</u> or (<u>d</u>), as applicable, (i) the Exchanging Member shall transfer and surrender such Class P Units to OpCo free and clear of all liens and encumbrances, (ii) OpCo shall issue such Exchanging Member a number of Class A Units that is equal to the product of (A) the number of such Class P Units and (B) the Class P Unit Exchange Rate (<u>provided</u> that if the number of Class A Units determined by this calculation is a negative number, it shall be deemed to be zero (0)), and (iii) such newly issued Class A Units shall constitute Exchanged Units, which will be immediately surrendered to OpCo or the Corporation in exchange for the Stock Exchange Payment or Cash Exchange Payment pursuant to <u>Section 2.2(c)</u> or (<u>d</u>), as applicable.

(c) On the Exchange Date (to be effective immediately prior to the close of business on the Exchange Date), in the case of a Redemption, (i) the Corporation shall contribute to OpCo, for delivery to the Exchanging Member (x) the Stock Exchange Payment with respect to any Exchanged Units not subject to a Cash Exchange Notice and (y) the Cash Exchange Payment with respect to any Exchanged Units subject to a Cash Exchange Notice, (ii) the Exchanging Member shall transfer and surrender the Exchanged Units to OpCo (provided that, in the case of Paired Interests, the Exchanging Member shall surrender the corresponding number of shares of Class V Common Stock to the Corporation and the Corporation shall cancel such shares), free and clear of all liens and encumbrances, (iii) OpCo shall issue to the Corporation a number of Class A Units equal to the number of Exchanged Units surrendered pursuant to <u>clause (ii)</u>, (iv) solely to the extent necessary in connection with a Redemption, with respect to the Class A Common Stock to maintain a one-to-one ratio between the number of Class A Units owned by the Corporation, directly or indirectly, and the number of outstanding shares of Class A Common Stock, taking into account the issuance in <u>clause (iii)</u>, any Stock Exchange Payment, and any other action taken in connection with this <u>Section 2.2</u>, and (v) OpCo shall (x) cancel the redeemed Exchanged Units and (y) transfer to the Exchanging Member the Cash Exchange Payment and/or the Stock Exchange Payment, as applicable.

(d) On the Exchange Date (to be effective immediately prior to the close of business on the Exchange Date), in the case of a Direct Exchange, (i) the Corporation shall deliver to the Exchanging Member, (x) the Stock Exchange Payment with respect to any Exchanged Units not subject to a Cash Exchange Notice and (y) the Cash Exchange Payment with respect to any Exchanged Units subject to a Cash Exchange Notice, (ii) the Exchanging Member shall transfer to the Corporation the Exchanged Units and the corresponding shares of Class V Common Stock (it being understood that the Corporation shall cancel the surrendered shares of Class V Common Stock), free and clear of all liens and encumbrances, and (iii) solely to the extent necessary in connection with a Direct Exchange, the Corporation shall undertake all actions, including an issuance, reclassification, distribution, division or recapitalization, with respect to the shares of Class A Common Stock to maintain a one-to-one ratio between the number of Class A Units owned by the Corporation, directly or indirectly, and the number of outstanding shares of Class A Common Stock, any Stock Exchange Payment, and any other action taken in connection with this <u>Section 2.2</u>.

(e) Upon the Exchange of all of an LLC Unitholder's LLC Units, such LLC Unitholder shall cease to be a Member (as such term is defined in the OpCo LLC Agreement) of OpCo.

## SECTION 2.3 Expenses and Restrictions.

(a) Except as expressly set forth in this Agreement, OpCo and each exchanging LLC Unitholder shall bear its own expenses in connection with the consummation of any Exchange, whether or not any such Exchange is ultimately consummated, except that OpCo shall bear any transfer taxes, stamp taxes or duties, or other similar taxes in connection with, or arising by reason of, any Exchange; provided, however, that if any shares of Class A Common Stock are to be delivered in a name other than that of the LLC Unitholder that requested the Exchange, then such LLC Unitholder and/or the person in whose name such shares are to be delivered shall pay to OpCo the amount of any transfer taxes, stamp taxes or duties, or other similar taxes in connection with, or arising by reason of, such Exchange or shall establish to the reasonable satisfaction of OpCo that such tax has been paid or is not payable.

(b) Notwithstanding anything to the contrary herein, the Corporation or OpCo shall use commercially reasonable efforts to restrict issuances of LLC Units in an amount sufficient for the Company to be eligible for the Private Placement Safe Harbor, and, to the extent that the Corporation or OpCo determine that OpCo does not meet the requirements of the Private Placement Safe Harbor at any point in any taxable year, the Corporation or OpCo may impose such restrictions on Exchanges during such taxable year as the Corporation or OpCo may determine to be necessary or advisable so that OpCo is not treated as a "publicly traded partnership" under Section 7704 of the Code; <u>provided</u>, that restrictions imposed pursuant to this <u>Section 2.3(b)</u> shall not apply to any Unrestricted Exchange. Notwithstanding anything to the contrary herein, no Exchange shall be permitted (and, if attempted, shall be void *ab initio*) if, in the good faith determination of the Corporation or of OpCo, such an Exchange would pose a material risk that OpCo would be a "publicly traded partnership" under Section 7704 of the Code.

(c) For the avoidance of doubt, and notwithstanding anything to the contrary herein, an LLC Unitholder shall not be entitled to effect an Exchange to the extent the Corporation determines that such Exchange (i) would be prohibited by law or regulation (including, without limitation, the unavailability of any requisite registration statement filed under the U.S. Securities Act of 1933, as amended (the "<u>Securities Act</u>"), or any exemption from the registration requirements thereunder) or (ii) would not be permitted under any other agreements with the Corporation or its subsidiaries to which such LLC Unitholder may be party (including, without limitation, the OpCo LLC Agreement) or any written policies of the Corporation related to unlawful or inappropriate trading applicable to its directors, officers or other personnel.

(d) The Corporation may adopt reasonable procedures for the implementation of the exchange provisions set forth in this <u>Article II</u>, including, without limitation, procedures for the giving of notice of an election of exchange.

SECTION 2.4 <u>Adjustment</u>. The Exchange Rate shall be adjusted accordingly if there is: (a) any subdivision (by any unit split, unit distribution, reclassification, reorganization, recapitalization or otherwise) or combination (by reverse unit split, reclassification, reorganization, recapitalization or otherwise) of the Class A Units that is not accompanied by an identical subdivision or combination of the Class A Common Stock or (b) any subdivision (by any stock split, stock dividend or distribution, reclassification, recapitalization or otherwise) or combination (by reverse stock split, reclassification, reorganization, recapitalization or otherwise) of the Class A Common Stock that is not accompanied by an identical subdivision or combination of the Class A Units. If there is any reclassification, reorganization, recapitalization or other similar transaction in which the Class A Common Stock are converted or changed into another security, securities or other property, then upon any subsequent Exchange, an exchanging LLC Unitholder shall be entitled to receive the amount of such security, securities or other property that such exchanging LLC Unitholder would have received if such Exchange had occurred immediately prior to the effective time of such reclassification, reorganization, recapitalization or other similar transaction, taking into account any adjustment as a result of any subdivision (by any split, distribution or dividend, reclassification, recapitalization or otherwise) or combination (by reverse split, reclassification, recapitalization or otherwise) of such security, securities or other property that occurs after the effective time of such reclassification, reorganization, recapitalization or other similar transaction. Except as may be required in the immediately preceding sentence, no adjustments in respect of distributions shall be made upon the exchange of any LLC Unit.

#### SECTION 2.5 Class A Common Stock to be Issued.

(a) The Corporation shall at all times reserve and keep available out of its authorized but unissued Class A Common Stock, solely for the purpose of issuance upon an Exchange, such number of shares of Class A Common Stock as may be deliverable upon any such Exchange; <u>provided</u>, that nothing contained herein shall be construed to preclude OpCo from satisfying its obligations in respect of the Exchange of the Exchanged Units by delivery of shares of Class A Common Stock which are held in the treasury of the Corporation or are held by OpCo or any of their subsidiaries or by delivery of purchased shares of Class A Common Stock (which may or may not be held in the treasury of the Corporation or held by any subsidiary thereof), or by delivery of the Cash Exchange Payment. The Corporation and OpCo covenant that all Class A Common Stock issued upon an Exchange will, upon issuance, be validly issued, fully paid and non-assessable.

(b) The Corporation and OpCo shall at all times ensure that the execution and delivery of this Agreement by each of the Corporation and OpCo and the consummation by each of the Corporation and OpCo of the transactions contemplated hereby (including without limitation, the issuance of the Class A Common Stock) have been duly authorized by all necessary corporate or limited liability company action, as the case may be, on the part of the Corporation and OpCo, including, but not limited to, all actions necessary to ensure that the acquisition of shares of Class A Common Stock pursuant to the transactions contemplated hereby, to the fullest extent of the Corporation's board of directors' power and authority and to the extent permitted by law, shall not be subject to any "moratorium," "control share acquisition," "business combination," "fair price" or other form of anti-takeover laws and regulations of any jurisdiction that may purport to be applicable to this Agreement or the transactions contemplated hereby.

(c) The Corporation and OpCo covenant and agree that, to the extent that a registration statement under the Securities Act is effective and available for shares of Class A Common Stock to be delivered with respect to any Exchange, shares that have been registered under the Securities Act shall be delivered in respect of such Exchange. In the event that any Exchange in accordance with this Agreement is to be effected at a time when any required registration has not become effective or otherwise is unavailable, upon the request and with the reasonable cooperation of the LLC Unitholder requesting such Exchange, the Corporation and OpCo shall use commercially reasonable efforts to promptly facilitate such Exchange pursuant to any reasonable efforts to list the Class A Common Stock required to be delivered upon exchange prior to such delivery upon each national securities exchange or inter-dealer quotation system upon which the outstanding Class A Common Stock may be listed or traded at the time of such delivery.

**SECTION 2.6** Direct Exchange. Notwithstanding anything to the contrary in this Article II, the Corporation may, in its sole and absolute discretion, elect to effect on the Exchange Date the Exchange of Exchanged Units for the Cash Exchange Payment and/or the Stock Exchange Payment, as the case may be (and subject to the terms of Section 2.2(c) and (d)), through a direct exchange of such Exchanged Units and with such consideration between the Exchanging Member and the Corporation (a "Direct Exchange"). Upon such Direct Exchange pursuant to this Section 2.6, the Corporation shall acquire the Exchanged Units and shall be treated for all purposes of this Agreement as the owner of such Units; provided, that, any such election by the Corporation shall not relieve OpCo of its obligation arising with respect to such applicable Exchange Notice. The Corporation may, at any time prior to an Exchange Date, deliver written notice (an "Direct Exchange Election Notice") to OpCo and the Exchanging Member setting forth its election to exercise its right to consummate a Direct Exchange; provided that such election does not prejudice the ability of the parties to consummate an Exchange or Direct Exchange on the Exchange Date. A Direct Exchange Election Notice may be revoked by the Corporation at any time; provided that any such revocation does not prejudice the ability of the parties to consummate an Exchange or Direct Exchange on the Exchange Date. The right to consummate a Direct Exchange in all events shall be exercisable for all the Exchanged Units that would otherwise have been subject to an Exchange. Except as otherwise provided in this Section 2.6, a Direct Exchange shall be consummated pursuant to the same timeframe and in the same manner as the relevant Exchange would have been consummated had the Corporation not delivered a Direct Exchange Election Notice.

## SECTION 2.7. Pubco Offer or Change of Control.

(a) In the event that a tender offer, share exchange offer, issuer bid, take-over bid, recapitalization or similar transaction with respect to Class A Common Stock (a "<u>Pubco Offer</u>") is proposed by the Corporation or is proposed to the Corporation or its stockholders and approved by the Board or is otherwise effected or to be effected with the consent or approval of the Board or the Corporation will undergo a Change of Control, the LLC Unitholders shall be permitted to deliver an Exchange Notice (which Exchange Notice shall be effective immediately prior to the consummation of such Pubco Offer or Change of Control (and, for the avoidance of doubt, shall be contingent upon such Pubco Offer or Change of Control and not be effective if such Pubco Offer or Change of Control is not consummated)). In the case of a Pubco Offer proposed by the Corporation, the Corporation will use its reasonable best efforts expeditiously and in good faith to take all such actions and do all such things as are necessary or desirable to enable and permit the LLC Unitholders to participate in such Pubco Offer to the same extent or on an economically equivalent basis as the holders of shares of Class A Common Stock without discrimination (but taking into account, for the avoidance of doubt, the Class P Unit Exchange Rate in the case of any Class P Units).

(b) The Corporation shall send written notice to the Company and the LLC Unitholders at least thirty (30) days prior to the closing of the transactions contemplated by the Pubco Offer or the Change of Control date notifying them of their rights pursuant to this <u>Section 2.7</u>, and setting forth, in the case of a Pubco Offer, (i) a copy of the written proposal or agreement pursuant to which the Pubco Offer will be effected, (ii) the consideration payable in connection therewith, (iii) the terms and conditions of transfer and payment and (iv) the date and location of and procedures for selling LLC Units, or in the case of a Change of Control, (x) a description of the event constituting the Change of Control, (y) the date of the Change of Control, and (z) a copy of any written proposals or agreement relating thereto. In the event that the information set forth in such notice changes from that set forth in the initial notice, a subsequent notice shall be delivered by the Corporation no less than seven (7) days prior to the closing of the Pubco Offer or date of the Change of Control.

## **ARTICLE III**

SECTION 3.1 <u>Additional LLC Unitholders</u>. To the extent an LLC Unitholder validly transfers any or all of such holder's LLC Units to another person in a transaction in accordance with, and not in contravention of, the OpCo LLC Agreement or any other agreement or agreements with the Corporation or any of its subsidiaries to which a transferring LLC Unitholder may be party, then such transferee (each, a "<u>Permitted Transferee</u>") shall have the right to execute and deliver a joinder to this Agreement, substantially in the form of <u>Exhibit B</u> hereto, whereupon such Permitted Transferee shall become an LLC Unitholder hereunder. To the extent OpCo issues LLC Units in the future, OpCo shall be entitled, in its sole discretion, to make any holder of such LLC Units an LLC Unitholder hereunder through such holder's execution and delivery of a joinder to this Agreement, substantially in the form of <u>Exhibit B</u> hereto.

SECTION 3.2 <u>Addresses and Notices</u>. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by courier service, by fax, by electronic mail (delivery receipt requested) or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be as specified in a notice given in accordance with this <u>Section 3.2</u>):

(a) If to the Corporation, to:

41 Discovery, Irvine, CA 92618 Attn: General Counsel Email: legal@weedmaps.com

(b) If to OpCo, to:

41 Discovery, Irvine, CA 92618 Attn: General Counsel Email: legal@ weedmaps.com

(c) If to any LLC Unitholder, to the address or other contact information set forth in the records of OpCo from time to time.

SECTION 3.3 <u>Further Action</u>. The parties shall execute and deliver all documents, provide all information and take or refrain from taking action as may be necessary or appropriate to achieve the purposes of this Agreement.

SECTION 3.4 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of all of the parties and, to the extent permitted by this Agreement, their successors, executors, administrators, heirs, legal representatives and assigns. No LLC Unitholder may assign its rights under this Agreement without the consent of the Corporation and OpCo.

SECTION 3.5 <u>Severability</u>. If any term or other provision of this Agreement is held to be invalid, illegal or incapable of being enforced by any rule of law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions is not affected in any manner materially adverse to any party. Upon a determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

SECTION 3.6 <u>Amendment</u>. The provisions of this Agreement may be amended only by the affirmative vote or written consent of each of (i) the Corporation, (ii) OpCo and (iii) LLC Unitholders holding at least a majority of the then outstanding LLC Units (excluding LLC Units held by the Corporation); <u>provided</u> that no amendment may materially, disproportionately and adversely affect the rights of an LLC Unitholder (other than the Corporation and its subsidiaries) without the consent of such LLC Unitholder (or, if there is more than one such LLC Unitholder that is so affected, without the consent of a majority in interest of such affected LLC Unitholders (other than the Corporation and its subsidiaries) in accordance with their holdings of LLC Units).

SECTION 3.7 <u>Waiver</u>. No failure by any party to insist upon the strict performance of any covenant, duty, agreement or condition of this Agreement or to exercise any right or remedy consequent upon a breach thereof shall constitute waiver of any such breach of any other covenant, duty, agreement or condition.

# SECTION 3.8 Submission to Jurisdiction; Waiver of Jury Trial.

Any and all disputes which cannot be settled amicably with respect to this Agreement, including any (a) action (at law or in equity), claim, litigation, suit, arbitration, hearing, audit, review, inquiry, proceeding, investigation or ancillary claims of any party, arising out of, relating to or in connection with the validity, negotiation, execution, interpretation, performance or non-performance of this Agreement or any matter arising out of or in connection with this Agreement and the rights and obligations arising hereunder or thereunder, or for recognition and enforcement of any judgment in respect of this Agreement and the rights and obligations arising hereunder or thereunder brought by a party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Chancery Court, or if such court shall not have jurisdiction, any federal court located in the State of Delaware, or, if neither of such courts shall have jurisdiction, any other Delaware state court. Each of the parties hereby irrevocably submits with regard to any such dispute for itself and in respect of its property, generally and unconditionally, to the sole and exclusive personal jurisdiction of the aforesaid courts and agrees that it will not bring any dispute relating to this Agreement or any of the transactions contemplated by this Agreement in any court other than the aforesaid courts. Each party irrevocably consents to service of process in any dispute in any of the aforesaid courts by the mailing of copies thereof by registered or certified mail, postage prepaid, or by recognized overnight delivery service, to such party at such party's address referred to in Section 3.2. Each party hereby irrevocably and unconditionally waives, and agrees not to assert as a defense, counterclaim or otherwise, in any action brought by any party with respect to this Agreement (i) any claim that it is not personally subject to the jurisdiction of the aforesaid courts for any reason other than the failure to serve process in accordance with this Section 3.8; (ii) any claim that it or its property is exempt or immune from the jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise); or (iii) any objection which such party may now or hereafter have (A) to the laying of venue of any of the aforesaid actions arising out of or in connection with this Agreement brought in the courts referred to above; (B) that such action brought in any such court has been brought in an inconvenient forum and (C) that this Agreement, or the subject matter hereof or thereof, may not be enforced in or by such courts.

(b) To the extent that any party has or hereafter may acquire any immunity from jurisdiction of any court or from any legal process (whether through service or notice, attachment prior to judgment, attachment in aid of execution, execution or otherwise) with respect to itself, or to such party's property, each such party hereby irrevocably waives such immunity in respect of such party's obligations with respect to this Agreement.

(c) EACH PARTY ACKNOWLEDGES THAT IT IS KNOWINGLY AND VOLUNTARILY AGREEING TO THE CHOICE OF DELAWARE LAW TO GOVERN THIS AGREEMENT AND TO THE JURISDICTION OF DELAWARE COURTS IN CONNECTION WITH PROCEEDINGS BROUGHT HEREUNDER. THE PARTIES INTEND THIS TO BE AN EFFECTIVE CHOICE OF DELAWARE LAW AND AN EFFECTIVE CONSENT TO JURISDICTION AND SERVICE OF PROCESS UNDER 6 DEL. C. § 2708.

(d) EACH PARTY, FOR ITSELF AND ITS AFFILIATES, HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ALL RIGHT TO TRIAL BY JURY IN ANY ACTION OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THE ACTIONS OF THE PARTIES OR THEIR RESPECTIVE AFFILIATES PURSUANT TO THIS AGREEMENT OR THE OTHER TRANSACTION DOCUMENTS IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF OR THEREOF.

(e) The parties hereby waive, to the fullest extent permitted by applicable law, any objection which they now or hereafter may have to personal jurisdiction or to the laying of venue of any such ancillary suit, action or proceeding brought in any court referred to in <u>Section 3.8(a)</u> and such parties agree not to plead or claim the same.

SECTION 3.9 <u>Counterparts</u>. This Agreement may be executed and delivered (including by facsimile transmission or by e-mail delivery of a ".pdf" format data file) in one or more counterparts, and by the different parties hereto in separate counterparts, each of which when executed and delivered shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Copies of executed counterparts transmitted by telecopy, by e-mail delivery of a ".pdf" format data file or other electronic transmission service shall be considered original executed counterparts for purposes of this <u>Section 3.9</u>.

SECTION 3.10 Tax Treatment. This Agreement shall be treated as part of the partnership agreement of OpCo as described in Section 761(c) of the Code and Sections 1.704-1(b)(2)(ii)(h) and 1.761-1(c) of the Treasury Regulations promulgated thereunder. As required by the Code and the Treasury Regulations, the parties shall report any Exchange consummated hereunder as a taxable sale of the Exchanged Units (together with an equal number of shares of Class V Common Stock, if applicable) by an LLC Unitholder to the Corporation in exchange for (i) the payment by the Corporation of the Stock Exchange Payment, the Cash Exchange Payment, or other applicable consideration to the Exchanging Member, and, if applicable, (ii) corresponding payments under the Tax Receivable Agreement, and no party shall take a contrary position on any income tax return, amendment thereof or communication with a taxing authority unless an alternate position is permitted under the Code and Treasury Regulations and the Corporation consents in writing, such consent not to be unreasonably withheld, conditioned, or delayed. Further, in connection with any Exchange consummated hereunder, OpCo and/or the Corporation shall provide the exchanging LLC Unitholder with all reasonably necessary information to enable the exchanging LLC Unitholder to file its income Tax returns for the taxable year that includes the Exchange, including information with respect to Code Section 751 assets (including relevant information regarding "unrealized receivables" or "inventory items") and Section 743(b) basis adjustments as soon as practicable and in all events within 60 days following the close of such taxable year (and use commercially reasonable efforts to provide estimates of such information within 90 days of the applicable Exchanges). Within thirty (30) days following the Exchange Date, the Corporation shall deliver a Section 743 notification to OpCo in accordance with Treasury Regulations Section 1.743-1(k)(2).

Withholding. The Corporation and OpCo shall be entitled to deduct and withhold from any payments SECTION 3.11 made to an LLC Unitholder pursuant to any Exchange consummated under this Agreement all Taxes that each of the Corporation and OpCo is required to deduct and withhold with respect to such payments under the Code (and any other provision of applicable law, including, without limitation, under Section 1445 and Section 1446(f) of the Code). In connection with any Exchange, the Exchanging Member shall, to the extent it is legally entitled to deliver such form, deliver to the Corporation or OpCo, as applicable, a certificate, dated as of the Exchange Date, in a form reasonably acceptable to the Corporation certifying as to such Exchanging Member's taxpayer identification number and that such Exchanging Member is a not a foreign person for purposes of Section 1445 and Section 1446(f) of the Code (which certificate may be an Internal Revenue Service Form W-9 if then sufficient for such purposes under applicable law) (such certificate, a "Non-Foreign Person Certificate"). If an Exchanging Member is unable to provide a Non-Foreign Person Certificate in connection with an Exchange, then (i) such Exchanging Member shall provide a certificate substantially in the form described in Proposed Regulations Section 1.1446(f)-2(c)(2)(ii)(B) or (ii) OpCo shall deliver a certificate reasonably acceptable to the Corporation and substantially in the form described in Proposed Regulations Section 1.1446(f)-2(c)(2)(ii)(C), in each case setting forth the liabilities of OpCo allocated to the Exchanged Units subject to the Exchange under Section 752 of the Code, and the Corporation or OpCo, as applicable, shall be permitted to withhold on the amount realized by such Exchanging Member in respect of such Exchange as provided in Section 1446(f) of the Code and Proposed Regulations thereunder. The Corporation or OpCo, as applicable, may at their sole discretion reduce the Class A Common Stock issued to a LLC Unitholder in an Exchange in an amount that corresponds to the amount of the required withholding described in the immediately preceding sentence and all such amounts shall be treated as having been paid to such LLC Unitholder.

SECTION 3.12 <u>Specific Performance</u>. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that such parties shall be entitled to specific performance of the terms and provisions hereof, in addition to any other remedy to which they are entitled at law or in equity.

SECTION 3.13 Independent Nature of LLC Unitholders' Rights and Obligations. The obligations of each LLC Unitholder hereunder are several and not joint with the obligations of any other LLC Unitholder, and no LLC Unitholder shall be responsible in any way for the performance of the obligations of any other LLC Unitholder hereunder. The decision of each LLC Unitholder to enter into this Agreement has been made by such LLC Unitholder independently of any other LLC Unitholder. Nothing contained herein, and no action taken by any LLC Unitholder pursuant hereto, shall be deemed to constitute the LLC Unitholders as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the LLC Unitholders are in any way acting in concert or as a group with respect to such obligations or the transactions contemplated hereby. The Corporation acknowledges that the LLC Unitholders are not acting in concert or as a group, and the Corporation will not assert any such claim, with respect to such obligations or the transactions contemplated hereby.

SECTION 3.14 <u>Applicable Law</u>. This Agreement shall be governed by, and construed in accordance with, the law of the State of Delaware, without regards to its principles of conflicts of laws.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered, all as of the date first set forth above.

WM Technology, Inc.

By: /s/ Christopher Beals

Name:Christopher BealsTitle:Chief Executive Officer

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered, all as of the date first set forth above.

WM Holding Company, LLC

By: /s/ Christopher Beals

Name:Christopher BealsTitle:Chief Executive Officer

#### EXHIBIT A

# EXCHANGE NOTICE

WM Technology, Inc. Attn: General Counsel 41 Discovery, Irvine, CA 92618

WM Holding Company, LLC Attn: General Counsel 41 Discovery, Irvine, CA 92618

Reference is hereby made to the Exchange Agreement, dated as of June 15, 2021 (as amended from time to time, the "<u>Exchange Agreement</u>"), among WM Holding Company, LLC, a Delaware limited liability company (together with any successor thereto, "**OpCo**"), WM Technology, Inc., a Delaware corporation ("**Pubco**") and managing member of OpCo, and the LLC Unitholders from time to time party thereto (each, a "**Holder**"). Capitalized terms used but not defined herein shall have the meanings given to them in the Exchange Agreement.

The undersigned Holder hereby transfers [the number of Class A Units plus shares of Class V Common Stock set forth below (together, the "**Paired Interests**")][the number of Class P Units] in Exchange for shares of Class A Common Stock to be issued in its name as set forth below, or the Cash Exchange Payment, as applicable, as set forth in the Exchange Agreement.

Legal Name of Holder:			
Address:			
Number of [Paired Interests] [Class P Units] to be Exchanged:			
Brokerage Account Details:			

The undersigned hereby represents and warrants that (i) the undersigned has full legal capacity to execute and deliver this Exchange Notice and to perform the undersigned's obligations hereunder; (ii) this Exchange Notice has been duly executed and delivered by the undersigned and is the legal, valid and binding obligation of the undersigned enforceable against it in accordance with the terms thereof or hereof, as the case may be, subject to applicable bankruptcy, insolvency and similar laws affecting creditors' rights generally and the availability of equitable remedies; (iii) the [Paired Interests][ Class P Units] subject to this Exchange Notice are being transferred to Pubco or OpCo, as applicable, free and clear of any pledge, lien, security interest, encumbrance, equities or claim; and (iv) no consent, approval, authorization, order, registration or qualification of any third party or with any court or governmental agency or body having jurisdiction over the undersigned or the [Paired Interests][ Class P Units] subject to this Exchange Notice is required to be obtained by the undersigned for the transfer of such [Paired Interests][ Class P Units] to PubCo or OpCo, as applicable.

The undersigned hereby irrevocably constitutes and appoints any officer of PubCo or of OpCo as the attorney of the undersigned, with full power of substitution and resubstitution in the premises, to do any and all things and to take any and all actions that may be necessary to transfer to PubCo or OpCo, as applicable, the [Paired Interests][ Class P Units] subject to this Exchange Notice and to deliver to the undersigned the Stock Exchange Payment or Cash Exchange Payment, as applicable, to be delivered in exchange therefor.

IN WITNESS WHEREOF the undersigned, by authority duly given, has caused this Exchange Notice to be executed and delivered by the undersigned or by its duly authorized attorney.

Name:

Dated:

#### EXHIBIT B

# JOINDER

This Joinder Agreement ("Joinder Agreement") is a joinder to the Exchange Agreement, dated as of June 15, 2021 (as amended from time to time, the "Exchange Agreement"), among WM Technology, Inc., a Delaware corporation (together with any successor thereto, the "<u>Corporation</u>"), WM Holding Company, LLC, a Delaware limited liability company, and each of the LLC Unitholders from time to time party thereto. Capitalized terms used but not defined in this Joinder Agreement shall have their meanings given to them in the Exchange Agreement. This Joinder Agreement shall be governed by, and construed in accordance with, the law of the State of Delaware. In the event of any conflict between this Joinder Agreement and the Exchange Agreement, the terms of this Joinder Agreement shall control.

The undersigned hereby joins and enters into the Exchange Agreement having acquired LLC Units in WM Holding Company, LLC. By signing and returning this Joinder Agreement to the Corporation, the undersigned accepts and agrees to be bound by and subject to all of the terms and conditions of and agreements of an LLC Unitholder contained in the Exchange Agreement, with all attendant rights, duties and obligations of an LLC Unitholder thereunder. The parties to the Exchange Agreement shall treat the execution and delivery hereof by the undersigned as the execution and delivery of the Exchange Agreement by the undersigned and, upon receipt of this Joinder Agreement by the Corporation and by WM Holding Company, LLC, the signature of the undersigned set forth below shall constitute a counterpart signature to the signature page of the Exchange Agreement.

# Exhibit 10.2

**EXECUTION COPY** 

TAX RECEIVABLE AGREEMENT among WM TECHNOLOGY, INC. and THE PERSONS NAMED HEREIN Dated as of June 16, 2021

# TABLE OF CONTENTS

		Page
ARTICLE I DEF	INITIONS	1
Section 1.1	Definitions	1
Section 1.1	Definitions	I
ARTICLE II DE	FERMINATION OF CERTAIN REALIZED TAX BENEFIT	8
		0
Section 2.1	Basis Adjustment	8
Section 2.2	Tax Benefit Schedule	8
Section 2.3	Procedures, Amendments	9
		-
ARTICLE III TA	X BENEFIT PAYMENTS	10
Section 3.1	Payments	10
Section 3.2	No Duplicative Payments	11
Section 3.3	Pro Rata Payments; Coordination of Benefits With Other Tax Receivable Agreements	11
ARTICLE IV TE	RMINATION	11
Section 4.1	Early Termination and Breach of Agreement	11
Section 4.2	Early Termination Notice	13
Section 4.3	Payment upon Early Termination	13
ARTICLE V SU	BORDINATION AND LATE PAYMENTS	13
Section 5.1	Subordination	13
Section 5.2	Late Payments by the Corporate Taxpayer	13
		14
ARTICLE VI NO	) DISPUTES; CONSISTENCY; COOPERATION	14
Section 6.1	Desticination in the Comparente Temperarie and OpCo's Terr Metters	14
Section 6.2	Participation in the Corporate Taxpayer's and OpCo's Tax Matters	14
Section 6.3	Consistency Cooperation	14
Section 0.5	Cooperation	14
ARTICI E VII M	ISCELLANEOUS	14
ARTICLE VII W	ISCELLANEOUS	14
Section 7.1	Notices	14
Section 7.2	Counterparts	15
Section 7.3	Entire Agreement; No Third Party Beneficiaries	15
Section 7.4	Governing Law	15
Section 7.5	Severability	15
Section 7.6	Successors; Assignment; Amendments; Waivers	15
Section 7.7	Titles and Subtitles	16
Section 7.8	Resolution of Disputes	16
Section 7.9	Reconciliation	16
Section 7.10	Withholding	17
Section 7.11	Admission of the Corporate Taxpayer into a Consolidated Group; Transfers of Corporate Assets	17
Section 7.12	Confidentiality	18
Section 7.13	Change in Law	18
Section 7.14	Independent Nature of TRA Parties' Rights and Obligations	18
Section 7.15	TRA Party Representative	19

#### TAX RECEIVABLE AGREEMENT

This TAX RECEIVABLE AGREEMENT (this "<u>Agreement</u>"), dated as of June 16, 2021, is hereby entered into by and among WM Technology, Inc., a Delaware corporation (the "<u>Corporate Taxpayer</u>"), and each of the other persons from time to time party hereto (the "<u>TRA Parties</u>").

#### RECITALS

WHEREAS, the TRA Parties directly or indirectly hold limited liability company units (the "<u>Units</u>") in WM Holding Company, LLC, a Delaware limited liability company ("<u>OpCo</u>"), which is classified as a partnership for United States federal income tax purposes;

WHEREAS, the Corporate Taxpayer, OpCo, Silver Spike Merger Sub LLC, a Delaware limited liability company ("<u>Opco Merger Sub"</u>), and other parties thereto entered into that certain Agreement and Plan of Merger, dated December 10, 2020 (as further amended or modified in whole or in part from time to time in accordance with such Agreement, the "<u>Merger</u> <u>Agreement</u>"), pursuant to which, among other things, Opco Merger Sub merged with and into OpCo with OpCo surviving (the "<u>Merger</u>") and the Corporate Taxpayer acquired certain Units in (i) a contribution governed by Section 721 of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>") and (ii) in a taxable transaction under Section 741 of the Code (the "<u>Purchase</u>");

WHEREAS, following the Merger, the Corporate Taxpayer is the managing member of OpCo, and holds and will hold, directly and/or indirectly, Units;

WHEREAS, following the Merger, the Units held by the TRA Parties, together with Class V common stock of the Corporate Taxpayer, may be exchanged for Class A common stock of the Corporate Taxpayer (the "<u>Class A Shares</u>") constituting the Stock Exchange Payment or, alternatively, at the election of the Corporate Taxpayer, the Cash Exchange Payment (an "<u>Exchange</u>"), pursuant to the provisions of the LLC Agreement (as defined below) and the Exchange Agreement, dated as of June 15, 2021, among the Corporate Taxpayer, OpCo, and the holders of Units from time to time party thereto, as amended from time to time (the "<u>Exchange Agreement</u>");

WHEREAS, OpCo and each of its direct and indirect subsidiaries treated as a partnership for United States federal income tax purposes currently have and will have in effect an election under Section 754 of the Code for the Taxable Year (as defined below) that includes the Closing Date and each subsequent Taxable Year in which a taxable acquisition (including a deemed taxable acquisition under Section 707(a) of the Code) of Units by the Corporate Taxpayer from the TRA Parties for Class A Shares or other consideration occurs;

WHEREAS, as a result of the Purchase and Exchanges, the income, gain, loss, expense and other Tax (as defined below) items of the Corporate Taxpayer may be affected by the Basis Adjustments (as defined below) and deductions attributable to any payment (including amounts attributable to Imputed Interest (as defined below)) made under this Agreement (collectively, the "Tax Attributes");

WHEREAS, the parties to this Agreement desire to make certain arrangements with respect to the effect of the Tax Attributes on the liability for Taxes of the Corporate Taxpayer;

NOW, THEREFORE, in consideration of the foregoing and the respective covenants and agreements set forth herein, and intending to be legally bound hereby, the parties hereto agree as follows:

## ARTICLE I

#### **DEFINITIONS**

Section 1.1 <u>Definitions</u>. As used in this Agreement, the terms set forth in this Article I shall have the following meanings (such meanings to be equally applicable to both the singular and plural forms of the terms defined).

"<u>Actual Tax Liability</u>" means, with respect to any Taxable Year, the actual liability for Taxes, which shall not be less than zero, of (i) the Corporate Taxpayer and (ii) without duplication, OpCo and its Subsidiaries, but only with respect to Taxes imposed on OpCo and its Subsidiaries and allocable to the Corporate Taxpayer or to the other members of the consolidated group of which the Corporate Taxpayer is the parent, *provided*, that, if applicable, such amounts shall be determined in accordance with a Determination (including interest imposed in respect thereof under applicable law).

"<u>Affiliate</u>" means, with respect to any Person, any other Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such first Person.

"Agreed Rate" means LIBOR plus 100 basis points.

"<u>Attributable</u>" means the portion of any Tax Attribute of the Corporate Taxpayer or Subsidiaries or OpCo and its Subsidiaries that is attributable to a TRA Party and shall be determined by reference to the Tax Attributes, under the following principles:

(i) any Purchase Basis Adjustments shall be determined separately with respect to each TRA Party and are Attributable to each TRA Party in an amount equal to the total Purchase Basis Adjustments relating to the interests in OpCo exchanged by such TRA Party in the Purchase;

(ii) any Exchange Basis Adjustments shall be determined separately with respect to each TRA Party and are Attributable to a TRA Party in an amount equal to the total Exchange Basis Adjustments relating to the Units Exchanged by such TRA Party; and

(iii) any deduction to the Corporate Taxpayer or its Subsidiaries, as applicable, with respect to a Taxable Year in respect of any payment (including amounts attributable to Imputed Interest) made under this Agreement is Attributable to the Person that is required to include the Imputed Interest or other payment in income (without regard to whether such Person is actually subject to Tax thereon).

"<u>Basis Adjustment</u>" means a Purchase Basis Adjustment or an Exchange Basis Adjustment.

A "<u>Beneficial Owner</u>" of a security is a Person who directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares: (i) voting power, which includes the power to vote, or to direct the voting of, such security and/or (ii) investment power, which includes the power to dispose of, or to direct the disposition of, such security. The terms "<u>Beneficially Own</u>" and "<u>Beneficial Ownership</u>" shall have correlative meanings.

"Board" means the Board of Directors of the Corporate Taxpayer.

"<u>Business Day</u>" means a day other than a Saturday, Sunday or other day on which commercial banks in New York, New York, San Francisco, California or Wilmington, Delaware are authorized or required by Law to close.

"<u>Cash Exchange Payment</u>" has the meaning set forth in the Exchange Agreement.

"<u>Change of Control</u>" means the occurrence of any of the following events:

(i) any Person or any group of Persons acting together which would constitute a "group" for purposes of Section 13(d) of the Exchange Act or any successor provisions thereto (excluding (a) a corporation or other entity owned, directly or indirectly, by the stockholders of the Corporate Taxpayer in substantially the same proportions as their ownership of stock of the Corporate Taxpayer or (b) a group of Persons in which one or more of the Permitted Investors or Affiliates of Permitted Investors directly or indirectly hold Beneficial Ownership of securities representing more than 50% of the total voting power held by such group) is or becomes the Beneficial Owner, directly or indirectly, of securities of the Corporate Taxpayer representing more than 50% of the combined voting power of the Corporate Taxpayer's then outstanding voting securities; or

(ii) the following individuals cease for any reason to constitute a majority of the number of directors of the Corporate Taxpayer then serving: individuals who, on the Closing Date, constitute the Board and any new director whose appointment or election by the Board or nomination for election by the Corporate Taxpayer's shareholders was approved or recommended by a vote of at least two-thirds of the directors then still in office who either were directors on the Closing Date or whose appointment, election or nomination for election was previously so approved or recommended by the directors referred to in this clause (ii); or

(iii) there is consummated a merger or consolidation of the Corporate Taxpayer with any other corporation or other entity, and, immediately after the consummation of such merger or consolidation, either (x) the board of directors immediately prior to the merger or consolidation does not constitute at least a majority of the board of directors of the company surviving the merger or, if the surviving company is a Subsidiary, the ultimate parent thereof, or (y) the voting securities of the Corporate Taxpayer immediately prior to such merger or consolidation do not continue to represent or are not converted into more than 50% of the combined voting power of the then outstanding voting securities of the Person resulting from such merger or consolidation or, if the surviving company is a Subsidiary, the ultimate parent thereof; or

(iv) the shareholders of the Corporate Taxpayer approve a plan of complete liquidation or dissolution of the Corporate Taxpayer or there is consummated an agreement or series of related agreements for the sale, lease or other disposition, directly or indirectly, by the Corporate Taxpayer of all or substantially all of the assets of the Corporate Taxpayer and its Subsidiaries, taken as a whole, other than such sale or other disposition by the Corporate Taxpayer of all or substantially all of the assets of the combined voting power of the voting securities of which are owned by shareholders of the Corporate Taxpayer in substantially the same proportions as their ownership of the Corporate Taxpayer immediately prior to such sale.

Notwithstanding the foregoing, except with respect to clause (ii) and clause (iii)(x) above, a "Change of Control" shall not be deemed to have occurred by virtue of the consummation of any transaction or series of integrated transactions immediately following which the record holders of the shares of the Corporate Taxpayer immediately prior to such transaction or series of transactions continue to have substantially the same proportionate ownership in, and own substantially all of the shares of, an entity which owns all or substantially all of the assets of the Corporate Taxpayer immediately following such transaction or series of transactions.

"<u>Closing Date</u>" means the date of the consummation of the transactions contemplated by the Merger Agreement.

"<u>Control</u>" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise.

"<u>Corporate Taxpayer Return</u>" means the United States federal and/or state and/or local and/or foreign Tax Return, as applicable, of the Corporate Taxpayer filed with respect to Taxes of any Taxable Year.

"<u>Cumulative Net Realized Tax Benefit</u>" for a Taxable Year means the cumulative amount of Realized Tax Benefits for all Taxable Years of the Corporate Taxpayer, up to and including such Taxable Year, net of the cumulative amount of Realized Tax Detriments for the same period. The Realized Tax Benefit and Realized Tax Detriment for each Taxable Year shall be determined based on the most recent Tax Benefit Schedules or Amended Schedules, if any, in existence at the time of such determination.

"Default Rate" means the LIBOR plus 500 basis points.

"<u>Determination</u>" shall have the meaning ascribed to such term in Section 1313(a) of the Code or similar provision of state, foreign or local tax law, as applicable, or any other event (including the execution of IRS Form 870-AD) that finally and conclusively establishes the amount of any liability for Tax.

"<u>Early Termination Date</u>" means the date of an Early Termination Notice for purposes of determining the Early Termination Payment.

"Early Termination Rate" means the LIBOR plus 150 basis points.

"Exchange" is defined in the Recitals of this Agreement.

"Exchange Basis Adjustment" means the adjustment to the Tax basis of a Reference Asset under Sections 732, 734(b) and/or 1012 of the Code (in situations where, as a result of one or more Exchanges, OpCo becomes an entity that is disregarded as separate from its owner for United States federal income Tax purposes) or under Sections 734(b), 743(b), 754 and/or 755 of the Code (in situations where, following an Exchange, OpCo remains in existence as an entity treated as a partnership for United States federal income Tax purposes) and, in each case, comparable sections of state, local, and foreign Tax laws, as a result of an Exchange and the payments made pursuant to this Agreement. The amount of any Exchange Basis Adjustment shall be determined using the Market Value with respect to such Exchange, except, for the avoidance of doubt, as otherwise required by a Determination. For the avoidance of doubt, payments under this Agreement shall not be treated as resulting in an Exchange Basis Adjustment to the extent such payments are treated as Imputed Interest, and the amount of any Basis Adjustment resulting from an Exchange of one or more Units shall be determined without regard to any Pre-Exchange Transfer of such Units and as if any such Pre-Exchange Transfer had not occurred.

"Exchange Date" means the date of any Exchange.

"Exchange Notice" shall have the meaning set forth in the LLC Agreement.

"<u>Hypothetical Tax Liability</u>" means, with respect to any Taxable Year, an amount, not less than zero, equal to the liability for Taxes of (i) the Corporate Taxpayer and (ii) without duplication, OpCo and its Subsidiaries, but only with respect to Taxes imposed on OpCo and its Subsidiaries and allocable to the Corporate Taxpayer or to the other members of the consolidated group of which the Corporate Taxpayer is the parent, in each case determined using the same methods, elections, conventions and similar practices used in computing the Actual Tax Liability, but, in each case, (a) calculating depreciation, amortization or similar deductions and income, gain or loss using the Non-Stepped Up Tax Basis as reflected on the Basis Schedule including amendments thereto for the Taxable Year and (b) excluding any deduction attributable to any payment (including amounts attributable to Imputed Interest) made under this Agreement for the Taxable Year. For the avoidance of doubt, Hypothetical Tax Liability shall be determined without taking into account the carryover or carryback of any Tax item (or portions thereof) that is attributable to a Tax Attribute, as applicable.

"<u>Imputed Interest</u>" in respect of a TRA Party shall mean any interest imputed under Section 1272, 1274 or 483 or other provision of the Code and any similar provision of state, local and foreign Tax law with respect to the Corporate Taxpayer's payment obligations in respect of such TRA Party under this Agreement.

"IRS" means the United States Internal Revenue Service.

"LIBOR" means during any period, an interest rate per annum equal to the one-year LIBOR reported, on the date two calendar days prior to the first day of such period, on the Telerate Page 3750 (or if such screen shall cease to be publicly available, as reported on Reuters Screen page "LIBOR01" or by any other publicly available source of such market rate) for London interbank offered rates for United States dollar deposits for such period. Notwithstanding the foregoing sentence: (i) if the Corporate Taxpaver reasonably determines, in good faith consultation with the TRA Party Representative, on or prior to the relevant date of determination that the relevant London interbank offered rate for U.S. dollar deposits has been discontinued or such rate has ceased to be published permanently or indefinitely, then "LIBOR" for the relevant interest period shall be deemed to refer to a substitute or successor rate that the Corporate Taxpayer reasonably determines, in good faith consultation with the TRA Party Representative, after consulting an investment bank of national standing in the United States and other reasonable sources, to be (a) the industry-accepted successor rate to the relevant London interbank offered rate for U.S. dollar deposits or (b) if no such industry-accepted successor rate exists, the most comparable substitute or successor rate to the relevant London interbank offered rate for U.S. dollar deposits; and (ii) if the Corporate Taxpayer has determined a substitute or successor rate in accordance with the foregoing, the Corporate Taxpayer may reasonably determine, in good faith consultation with the TRA Party Representative, after consulting an investment bank of national standing in the United States and other reasonable sources, any relevant methodology for calculating such substitute or successor rate, including any adjustment factor it reasonably determines, in good faith consultation with the TRA Party, is needed to make such substitute or successor rate comparable to the relevant London interbank offered rate for U.S. dollar deposits, in a manner that is consistent with industry-accepted practices for such substitute or successor rate. In the event that the TRA Party Representative disagrees with any determination by the Corporate Taxpayer set forth in this paragraph, and such disagreement is not resolved within thirty (30) days of submission by the TRA Party Representative of notice of such disagreement to the Corporate Taxpayer, such disagreement shall be deemed a "Reconciliation Dispute," and shall be subject to the Reconciliation Procedures set forth in Section 7.9 hereof.

"<u>LLC Agreement</u>" means, with respect to OpCo, the Fourth Amended and Restated Limited Liability Company Agreement of OpCo, dated on or about the date hereof, as amended from time to time.

"<u>Market Value</u>" shall mean on any date, (a) if the Class A Shares trade on a national securities exchange or automated or electronic quotation system, the arithmetic average of the high trading and the low trading price on such date (or if such date is not a trading day, the immediately preceding trading day) or (b) if the Class A Shares are not then traded on a national securities exchange or automated or electronic quotation system, as applicable, the "Appraiser FMV" (as defined in the Exchange Agreement) on such date of one (1) Class A Share that would be obtained in an arms-length transaction between an informed and willing buyer and an informed and willing seller, neither of whom is under any compulsion to buy or sell, respectively, and without regard to the particular circumstances of the buyer or seller.

"<u>Non-Stepped Up Tax Basis</u>" means, with respect to any Reference Asset at any time, the Tax basis that such asset would have had at such time if no Basis Adjustments had been made.

"<u>Payment Date</u>" means any date on which a payment is required to be made pursuant to this Agreement.

"<u>Person</u>" means any individual, corporation, firm, partnership, joint venture, limited liability company, estate, trust, business association, organization, governmental entity or other entity.

"<u>Pre-Exchange Transfer</u>" means any transfer (including upon the death of a Member) or distribution in respect of one or more Units (a) that occurs prior to an Exchange of such Units, and (b) to which Section 743(b) or 734(b) of the Code applies.

"<u>Purchase Basis Adjustment</u>" means the adjustment to the tax basis of a Reference Asset under Sections 734(b), 743(b), 754 and/or 755 of the Code and, in each case, comparable sections of state and local tax laws, as a result of the Purchase and the payments made pursuant to this Agreement in respect of the Purchase. For the avoidance of doubt, payments made under this Agreement shall not be treated as resulting in a Purchase Basis Adjustment to the extent such payments are treated as Imputed Interest.

"<u>Realized Tax Benefit</u>" means, for a Taxable Year, the excess, if any, of the Hypothetical Tax Liability over the Actual Tax Liability. If all or a portion of the actual liability for such Taxes for the Taxable Year arises as a result of an audit by a Taxing Authority of any Taxable Year, such liability shall not be included in determining the Realized Tax Benefit unless and until there has been a Determination.

"<u>Realized Tax Detriment</u>" means, for a Taxable Year, the excess, if any, of the Actual Tax Liability over the Hypothetical Tax Liability. If all or a portion of the actual liability for such Taxes for the Taxable Year arises as a result of an audit by a Taxing Authority of any Taxable Year, such liability shall not be included in determining the Realized Tax Detriment unless and until there has been a Determination.

"<u>Reference Asset</u>" means an asset that is held by OpCo, or by any of its direct or indirect Subsidiaries treated as a partnership or disregarded entity (but only if such indirect Subsidiaries are held only through Subsidiaries treated as partnerships or disregarded entities) for purposes of the applicable Tax, at the time of the Purchase or an Exchange, as relevant. A Reference Asset also includes any asset the Tax basis of which is determined, in whole or in part, for purposes of the applicable Tax, by reference to the Tax basis of an asset that is described in the preceding sentence, including for U.S. federal income Tax purposes, any asset that is "substituted basis property" under Section 7701(a)(42) of the Code with respect to a Reference Asset.

"<u>Schedule</u>" means any of the following: (a) a Basis Schedule, (b) a Tax Benefit Schedule, or (c) the Early Termination Schedule.

"<u>Stock Exchange Payment</u>" has the meaning set forth in the Exchange Agreement.

"<u>Subsidiaries</u>" means, with respect to any Person, as of any date of determination, any other Person as to which such Person, owns, directly or indirectly, or otherwise controls more than 50% of the voting power or other similar interests or the sole general partner interest or managing member or similar interest of such Person.

"<u>Subsidiary Stock</u>" means any stock or other equity interest in any subsidiary entity of OpCo that is treated as a C corporation for United States federal income tax purposes.

"<u>Tax Attributes</u>" has the meaning set forth in the Recitals.

"<u>Tax Benefit Payment</u>" has the meaning set forth in Section 3.1(b).

"<u>Tax Benefit Schedule</u>" has the meaning set forth in Section 2.2(a).

"<u>Tax Return</u>" means any return, declaration, report, or similar statement filed or required to be filed with respect to Taxes (including any attached schedules), including any information return, claim for refund, amended return and declaration of estimated Tax.

"<u>Taxable Year</u>" means a taxable year of the Corporate Taxpayer as defined in Section 441(b) of the Code or comparable section of state, local or foreign Tax law, as applicable (and, therefore, for the avoidance of doubt, may include a period of less than 12 months for which a Tax Return is made), ending on or after the Closing Date.

"<u>Taxes</u>" means any and all United States federal, state, local and foreign taxes, assessments or similar charges that are based on or measured with respect to net income or profits, whether as an exclusive or an alternative basis, and including franchise taxes that are based on or measured with respect to net income or profits, and any interest, penalties, or additions related to such amounts or imposed in respect thereof under applicable law.

"<u>Taxing Authority</u>" shall mean any domestic, federal, national, state, county or municipal or other local government, any subdivision, agency, commission or authority thereof, or any quasi-governmental body exercising any taxing authority or any other authority exercising Tax regulatory authority.

"<u>TRA Disinterested Majority</u>" means a majority of the directors of the Board who are disinterested as determined by the Board in accordance with the DGCL with respect to the matter being considered by the Board; provided that to the extent a matter being considered by the Board is required to be considered by disinterested directors under the rules of the National Securities Exchange on which the Class A Common Stock is then listed, the Securities Act or the Exchange Act, such rules with respect to the definition of disinterested director shall apply solely with respect to such matter.

"<u>TRA Party Representative</u>" means, initially, Ghost Media Group, LLC, or, if Ghost Media Group, LLC becomes unable to perform the TRA Party Representative's responsibilities hereunder or resigns from such position, either (x) a replacement TRA Party Representative selected by Ghost Media Group, LLC, or (y) if Ghost Media Group, LLC has not selected a replacement TRA Party Representative at or prior to the time of such inability or resignation, that TRA Party or committee of TRA Parties determined by a plurality vote of the TRA Parties ratably in accordance with their right to receive Early Termination Payments hereunder if all TRA Parties had fully Exchanged their Units for Class A Shares or other consideration and the Corporate Taxpayer had exercised its right of early termination on the date of the most recent Exchange.

"<u>Treasury Regulations</u>" means the final, temporary and proposed regulations under the Code promulgated from time to time (including corresponding provisions and succeeding provisions) as in effect for the relevant taxable period.

"Valuation Assumptions" shall mean, as of an Early Termination Date, the assumptions that in each Taxable Year ending on or after such Early Termination Date, (a) the Corporate Taxpayer will have taxable income sufficient to fully utilize deductions arising from the Tax Attributes (other than any items addressed in clause (b)) during such Taxable Year or future Taxable Years (including, for the avoidance of doubt, deductions and other Tax items arising from Tax Attributes that would result from future Tax Benefit Payments that would be paid in accordance with the Valuation Assumptions, further assuming that such applicable future payments would be paid on the due date (including extensions) for filing the Corporate Taxpaver Return for the applicable Taxable Year) in which such deductions would become available (b) any loss carryovers generated by deductions arising from Tax Attributes that are available as of the date of such Early Termination Date, will be used by the Corporate Taxpayer on a pro rata basis from the Early Termination Date through the scheduled expiration date thereof or, if there is no such scheduled expiration date, the tenth anniversary of the generation of such loss carryover, (c) the United States federal, state, local and foreign income tax rates that will be in effect for each such Taxable Year will be those specified for each such Taxable Year by the Code and other law as in effect on the Early Termination Date, except to the extent any change to such Tax rates for such Taxable Year has already been enacted into law as of the Early Termination Date, (d) any non-amortizable, nondepreciable Reference Assets (other than any Subsidiary Stock) will be disposed of on the fifteenth anniversary of the Purchase or Exchange which gave rise to the applicable Basis Adjustment and any short-term investments will be disposed of 12 months following the Early Termination Date; provided that, in the event of a Change of Control, such non-amortizable, non-depreciable assets shall be deemed disposed of at the time of sale of the relevant asset (if earlier than such fifteenth anniversary), (e) any Subsidiary Stock will never be disposed of and (f) if, at the Early Termination Date, there are Units that have not been Exchanged, then each such Unit is Exchanged in a fully taxable transaction for the Market Value of the Class A Shares that would be transferred if the Exchange occurred on the Early Termination Date.

Term	Section
Agreement	Recitals
Amended Schedule	Section 2.3(b)
Class A Shares	Recitals
Code	Recitals
Corporate Taxpayer	Recitals
Early Termination Effective Date	Section 4.2
Early Termination Notice	Section 4.2
Early Termination Schedule	Section 4.2
Early Termination Payment	Section 4.3(b)
Basis Schedule	Section 2.1
Expert	Section 7.9
Joinder Requirement	Section 7.6(a)
Liquidity Exceptions	Section 4.1(b)
Mandatory Assignment	Section 7.6(c)

Material Objection Notice	Section 4.2
Merger Agreement	Recitals
Objection Notice	Section 2.3(a)
Reconciliation Dispute	Section 7.9
Reconciliation Procedures	Section 2.3(a)
Senior Obligations	Section 5.1
Tax Benefit Payment	Section 3.1(b)
Tax Benefit Schedule	Section 2.2
TRA Party	Recitals
Units	Recitals

## **ARTICLE II**

#### **DETERMINATION OF CERTAIN REALIZED TAX BENEFIT**

Section 2.1 <u>Basis Adjustment</u>. Within one hundred twenty (120) calendar days after the filing of the United States federal income tax return of the Corporate Taxpayer for the Taxable Year that includes the Closing Date and each Taxable Year in which an Exchange has been effected, the Corporate Taxpayer shall deliver to each TRA Party who received (or is deemed to receive) cash or Class A Shares in such Taxable Year pursuant to the Purchase or an Exchange, as applicable, a schedule (the "<u>Basis Schedule</u>") that shows, in reasonable detail necessary to perform the calculations required by this Agreement, (a) the actual Tax basis and the Non-Stepped Up Tax Basis of the Reference Assets as of the Closing Date and each applicable Exchange Date, (b) the Exchange Basis Adjustment with respect to the Reference Assets Attributable to such TRA Party as a result of the Exchanges effected in such Taxable Year and prior Taxable Years by such TRA Party, calculated in the aggregate, (c) the Purchase Basis Adjustment Attributable to such TRA Party for the Taxable Year of the Closing, (d) the period (or periods) over which the Reference Assets are amortizable and/or depreciable and (e) the period (or periods) over which each Basis Adjustment in respect of such TRA Party is amortizable and/or depreciable. Each Basis Schedule will become final as provided in Section 2.3(b) (subject to the procedures set forth in Section 2.3(b)).

#### Section 2.2 <u>Tax Benefit Schedule</u>.

(a) <u>Tax Benefit Schedule</u>. Within one hundred and twenty (120) calendar days after the filing of the United States federal income tax return of the Corporate Taxpayer for each Taxable Year, the Corporate Taxpayer shall provide to each TRA Party who has received (or is deemed to receive) cash or Class A Shares pursuant to the Purchase or an Exchange, as applicable, a schedule showing, in reasonable detail, the calculation of the Tax Benefit Payment, if any, any Realized Tax Benefit and any Realized Tax Detriment, as applicable, Attributable to such TRA Party for such Taxable Year (a "<u>Tax Benefit Schedule</u>"). Each Tax Benefit Schedule will become final as provided in Section 2.3(a) and may be amended as provided in Section 2.3(b) (subject to the procedures set forth in Section 2.3(b)).

(b) Applicable Principles. Subject to Section 3.3(a), the Realized Tax Benefit or Realized Tax Detriment for each Taxable Year is intended to measure the decrease or increase in the Actual Tax Liability for such Taxable Year attributable to the Tax Attributes, determined using a "with and without" methodology. For the avoidance of doubt, the Actual Tax Liability will take into account the deduction of the portion of the Tax Benefit Payment that must be accounted for as interest under the Code based upon the characterization of Tax Benefit Payments as additional consideration payable by the Corporate Taxpaver for the Units acquired in the Purchase or Exchange. Carryovers or carrybacks of any Tax item attributable to the Tax Attributes shall be considered to be subject to the rules of the Code and the Treasury Regulations or the appropriate provisions of U.S. state and local and foreign income and franchise Tax law, as applicable, governing the use, limitation and expiration of carryovers or carrybacks of the relevant type. If a carryover or carryback of any Tax item includes a portion that is attributable to any Tax Attribute and another portion that is not, such portions shall be considered to be used in accordance with the "with and without" methodology. The parties agree that (i) (A) all Tax Benefit Payments in respect of the Purchase (to the extent permitted by applicable law and other than amounts accounted for as Imputed Interest) are intended to be treated and shall be reported for all purposes, including Tax purposes, as additional contingent consideration to the applicable TRA Parties for the sale of interests in OpCo at the Closing that has the effect of creating Purchase Basis Adjustments, and (B) all Tax Benefit Payments in respect of an Exchange are intended to be treated and shall be reported for all purposes, including Tax purposes, as additional contingent consideration to the applicable TRA Party for such Exchange that has the effect of creating Exchange Basis Adjustments, in each case, to Reference Assets for the Corporate Taxpayer in the year of payment, (ii) as a result, such additional Basis Adjustments will be incorporated into the calculation for the Taxable Year of the applicable payment and into the calculations for subsequent Taxable Years, as appropriate, and (iii) the Actual Tax Liability shall take into account the deduction of the portion of the Tax Benefit Payment that must be accounted for as Imputed Interest under applicable law and (iv) the liability for U.S. federal income Taxes of the Corporate Taxpayer and the amount of taxable income of the Corporate Taxpayer for U.S. federal income Tax purposes as determined for purposes of calculating the Actual Tax Liability and the Hypothetical Tax Liability shall include, without duplication, such liability for Taxes and such taxable income that is economically borne by or allocated to the Corporate Taxpayer as a result of the provisions of Sections 5.07 and 5.08 of the LLC Agreement; provided, however, that such liability for Taxes and such taxable income shall be included in the Hypothetical Tax Liability and the Actual Tax Liability subject to the adjustments and assumptions set forth in the definitions thereof and, to the extent any such amount is taken into account on an Amended Schedule, such amount shall adjust a Tax Benefit Payment, as applicable, in accordance with Section 2.3(b).

## Section 2.3 Procedures, Amendments.

(a) Procedure. Every time the Corporate Taxpayer delivers to a TRA Party an applicable Schedule under this Agreement, including any Amended Schedule delivered pursuant to Section 2.3(b), and any Early Termination Schedule or amended Early Termination Schedule, the Corporate Taxpaver shall also (x) deliver to such TRA Party supporting schedules, valuation reports, if any, and work papers, as determined by the Corporate Taxpayer or requested by such TRA Party, providing reasonable detail regarding data and calculations that were relevant for the preparation of the Schedule, (y) indicate which accounting firm, if any, assisted with the preparation of the Schedule, and (z) allow the TRA Party Representative and its advisors reasonable access to the appropriate representatives at the Corporate Taxpayer and (at the cost and expense of OpCo) at the relevant accounting firm that prepared the applicable Schedule, if applicable, in connection with the review of such Schedule. Without limiting the generality of the preceding sentence, the Corporate Taxpayer shall ensure that each Tax Benefit Schedule or Early Termination Schedule delivered to a TRA Party, together with any supporting schedules and work papers, provides a reasonably detailed presentation of the calculation of the Actual Tax Liability (the "with" calculation), the Hypothetical Tax Liability (the "without" calculation), and identifies any material assumptions or operating procedures or principles that were used for purposes of such calculations. An applicable Schedule or amendment thereto shall become final and binding on all parties thirty (30) calendar days from the date on which all relevant TRA Parties are treated as having received the applicable Schedule or amendment thereto under Section 7.1 unless the TRA Party Representative (i) within thirty (30) calendar days from such date provides the Corporate Taxpayer with notice of an objection to such Schedule ("Objection Notice") or (ii) provides a written waiver of such right of any Objection Notice within the period described in clause (i) above, in which case such Schedule or amendment thereto shall become binding on the date such waiver is received by the Corporate Taxpayer. If the Corporate Taxpayer and the TRA Party Representative, for any reason, are unable to successfully resolve the issues raised in the Objection Notice within thirty (30) calendar days after receipt by the Corporate Taxpayer of an Objection Notice, the Corporate Taxpayer and the TRA Party Representative shall employ the reconciliation procedures as described in Section 7.9 of this Agreement (the "Reconciliation Procedures").

(b) <u>Amended Schedule</u>. The applicable Schedule for any Taxable Year may be amended from time to time by the Corporate Taxpayer (i) in connection with a Determination affecting such Schedule, (ii) to correct inaccuracies in the Schedule, including those identified as a result of the receipt of additional factual information relating to a Taxable Year after the date the Schedule was provided to a TRA Party, (iii) to comply with the Expert's determination under the Reconciliation Procedures, (iv) to reflect a change in the Realized Tax Benefit or Realized Tax Detriment for such Taxable Year attributable to a carryback or carryforward of a loss or other Tax item to such Taxable Year, (v) to reflect a change in the Realized Tax Benefit or Realized Tax Detriment for such Taxable Year attributable to an amended Tax Return filed for such Taxable Year, or (vi) to adjust an applicable Basis Schedule to take into account payments made pursuant to this Agreement (any such Schedule, an "Amended Schedule"). The Corporate Taxpayer shall provide an Amended Schedule to each TRA Party within thirty (30) calendar days of the occurrence of an event referenced in clauses (i) through (vi) of the preceding sentence. In the event a Schedule is amended after such Schedule becomes final pursuant to Section 2.3(a) or, if applicable, Section 7.9, (A) the Amended Schedule shall not be taken into account in calculating any Tax Benefit Payment in the Taxable Year to which the amendment relates but instead shall be taken into account in calculating the Cumulative Net Realized Tax Benefit for the Taxable Year in which the amendment actually occurs, and (B) as a result of the foregoing, any increase of the Net Tax Benefit attributable to an Amended Schedule shall not accrue the Interest Amount (or any other interest hereunder) until after the due date (without extensions) for filing the United States federal income tax return of the Corporate Taxpayer for the Taxable Year in which the amendment actually occurs.

#### **ARTICLE III**

#### TAX BENEFIT PAYMENTS

#### Section 3.1 Payments.

(a) <u>Payments</u>. Within five (5) Business Days after a Tax Benefit Schedule delivered to a TRA Party becomes final in accordance with Section 2.3(a), or, if applicable, Section 7.9, the Corporate Taxpayer shall pay such TRA Party for such Taxable Year the Tax Benefit Payment determined pursuant to Section 3.1(b) that is Attributable to such TRA Party. Each such Tax Benefit Payment shall be made by wire transfer of immediately available funds to the bank account previously designated by such TRA Party to the Corporate Taxpayer or as otherwise agreed by the Corporate Taxpayer and such TRA Party. The payments provided for pursuant to the above sentence shall be computed separately for each TRA Party. For the avoidance of doubt, no Tax Benefit Payment shall be made in respect of estimated tax payments, including federal estimated income tax payments. Notwithstanding anything herein to the contrary, at the election of a TRA Party (specified in the Exchange Notice with respect to an applicable Exchange, or by providing written notice to the Corporate Taxpayer at the Closing with respect to the Purchase), the aggregate Tax Benefit Payments in respect of such Purchase or Exchange (other than amounts accounted for as interest under the Code) shall not exceed, as specified by a TRA Party, 50% of the fair market value of the Class A Shares, or cash received in the relevant Purchase or Exchange. Without limiting the Corporate Taxpayer's ability to make offsets against Tax Benefit Payments to the extent permitted by Section 3.4, no TRA Party shall be required under any circumstances to make a payment or return a payment to the Corporate Taxpayer in respect of any portion of any Tax Benefit Payment previously paid by the Corporate Taxpayer to such TRA Party (including any portion of any Early Termination Payment).

(b) A "Tax Benefit Payment" in respect of a TRA Party for a Taxable Year means an amount, not less than zero, equal to the sum of the portion of the Net Tax Benefit that is Attributable to such TRA Party and the Interest Amount with respect thereto. For the avoidance of doubt, for Tax purposes, the Interest Amount shall not be treated as interest (to the extent permitted by applicable law and other than amounts accounted for as Imputed Interest) but instead shall be treated as additional consideration for the acquisition of Units in the applicable Purchase or Exchange, unless otherwise required by law. Subject to Section 3.3(a), the "Net Tax Benefit" for a Taxable Year shall be an amount equal to the excess, if any, of eighty-five percent (85%) of the Cumulative Net Realized Tax Benefit as of the end of such Taxable Year, over the total amount of payments previously made under this Section 3.1 (excluding payments attributable to Interest Amounts); provided that if there is no such excess (or a deficit exists) no TRA Party shall be required to make payment (or return a payment) to the Corporate Taxpayer in respect of any portion of any previously made Tax Benefit Payment. The "Interest Amount" shall equal the interest on the Net Tax Benefit calculated at the Agreed Rate from the due date (without extensions) for filing the Corporate Taxpaver Return with respect to Taxes for such Taxable Year until the payment date under Section 3.1(a). The Net Tax Benefit and the Interest Amount shall be determined separately with respect to each Purchase or Exchange, on a Unit by Unit basis by reference to the resulting Basis Adjustment to the Corporate Taxpayer. Notwithstanding the foregoing, for each Taxable Year ending on or after the date of a Change of Control that occurs after the Closing Date, all Tax Benefit Payments, whether paid with respect to the Units that were Exchanged (i) prior to the date of such Change of Control or (ii) on or after the date of such Change of Control, shall be calculated by utilizing Valuation Assumptions (a), (c) and (d), substituting in each case the terms "the closing date of a Change of Control" for an "Early Termination Date."

Section 3.2 <u>No Duplicative Payments</u>. It is intended that the provisions of this Agreement will result in the payments specified in Section 3.1 being made to the TRA Parties and will not result in duplicative payment of any amount (including interest) required under this Agreement. The provisions of this Agreement shall be construed in the appropriate manner to ensure such intentions are realized.

# Section 3.3 Pro Rata Payments; Coordination of Benefits With Other Tax Receivable Agreements.

(a) Notwithstanding anything in Section 3.1 to the contrary, to the extent that the aggregate Tax benefit of the Corporate Taxpayer with respect to the Basis Adjustments or Imputed Interest, as such terms are defined in this Agreement, is limited in a particular Taxable Year because the Corporate Taxpayer does not have sufficient taxable income, the Net Tax Benefit for the Corporate Taxpayer shall be allocated among all TRA Parties eligible for payments under this Agreement in proportion to the respective amounts of Net Tax Benefit that would have been allocated to each such TRA Party if the Corporate Taxpayer had sufficient taxable income so that there were no such limitation.

(b) If for any reason (including as contemplated by Section 3.3(a)) the Corporate Taxpayer does not fully satisfy its payment obligations to make all Tax Benefit Payments due under this Agreement in respect of a particular Taxable Year, then the Corporate Taxpayer and the TRA Parties agree that no Tax Benefit Payment shall be made in respect of any subsequent Taxable Year until all Tax Benefit Payments in respect of prior Taxable Years have been made in full.

(c) Any Tax Benefit Payment or Early Termination Payment required to be made by the Corporate Taxpayer to the TRA Parties under this Agreement shall rank senior in right of payment to any principal, interest or other amounts due and payable in respect of any similar agreement ("<u>Other Tax Receivable Obligations</u>"). The effect of any other similar agreement shall not be taken into account in respect of any calculations made hereunder.

Section 3.4 <u>Overpayments</u>. To the extent the Corporate Taxpayer makes a payment to a TRA Party in respect of a particular Taxable Year under Section 3.1(a) in an amount in excess of the amount of such payment that should have been made to such TRA Party in respect of such Taxable Year (taking into account Section 3.3) under the terms of this Agreement, then such TRA Party shall not receive further payments under Section 3.1(a) until such TRA Party has foregone an amount of payments equal to such excess.

## ARTICLE IV

## **TERMINATION**

## Section 4.1 Early Termination and Breach of Agreement.

(a) The Corporate Taxpayer may, with the prior written consent of the TRA Disinterested Majority, terminate this Agreement with respect to all amounts payable to the TRA Parties and with respect to all of the Units held by the TRA Parties at any time by paying to each TRA Party the Early Termination Payment in respect of such TRA Party; <u>provided</u>, <u>however</u>, that this Agreement shall only terminate upon the receipt of the Early Termination Payment by all TRA Parties; <u>provided further</u> that the Corporate Taxpayer may withdraw any notice to execute its termination rights under this Section 4.1(a) prior to the time at which any Early Termination Payment of the Early Termination Payment by the Corporate Taxpayer, none of the TRA Parties or the Corporate Taxpayer shall have any further payment obligations under this Agreement, other than for any (i) Tax Benefit Payment due and payable that remains unpaid as of the Early Termination Notice (except to the extent that the amount described in clause (ii) is included in the Early Termination Payment). If an Exchange occurs after the Corporate Taxpayer makes all of the required Early Termination Payments, the Corporate Taxpayer shall have no obligations under this Agreement with respect to such Exchange.

(b) In the event that the Corporate Taxpayer (1) breaches any of its material obligations under this Agreement. whether as a result of failure to make any payment when due, failure to honor any other material obligation required hereunder or (2)(A) the Corporate Taxpayer commences any case, proceeding or other action (i) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts or (ii) seeking an appointment of a receiver, trustee, custodian, conservator or other similar official for it or for all or any substantial part of its assets, or it shall make a general assignment for the benefit of creditors or (B) there shall be commenced against the Corporate Taxpaver any case, proceeding or other action of the nature referred to in clause (A) above that remains undismissed or undischarged for a period of sixty (60) days, all obligations hereunder shall be accelerated and such obligations shall be calculated as if an Early Termination Notice had been delivered on the date of such breach and shall include, but not be limited to, (i) the Early Termination Payments calculated as if an Early Termination Notice had been delivered on the date of a breach, (ii) any Tax Benefit Payment in respect of a TRA Party agreed to by the Corporate Taxpayer and such TRA Party as due and payable but unpaid as of the date of a breach, and (iii) any Tax Benefit Payment in respect of any TRA Party due for the Taxable Year ending with or including the date of a breach; provided, that procedures similar to the procedures of Section 4.2 shall apply with respect to the determination of the amount payable by the Corporate Taxpayer pursuant to this sentence. Notwithstanding the foregoing, in the event that the Corporate Taxpayer breaches this Agreement, each TRA Party shall be entitled to elect to receive the amounts set forth in clauses (i), (ii) and (iii) above or to seek specific performance of the terms hereof. The parties agree that the failure to make any payment due pursuant to this Agreement within three (3) months of the date such payment is due shall be deemed to be a breach of a material obligation under this Agreement for all purposes of this Agreement, and that it will not be considered to be a breach of a material obligation under this Agreement to make a payment due pursuant to this Agreement within three (3) months of the date such payment is due. Notwithstanding anything in this Agreement to the contrary, it shall not be a breach of this Agreement if the Corporate Taxpaver fails to make any Tax Benefit Payment when due to the extent that the Corporate Taxpaver (x) has insufficient funds, or cannot make such payment as a result of obligations imposed in connection with any Senior Obligations, and cannot take commercially reasonable actions to obtain sufficient funds, to make such payment or (y) would become insolvent as a result of making such payment (in each case, as determined by the Board in good faith) (clauses (x) and (y) together, the "Liquidity Exceptions"); provided that the interest provisions of Section 5.2 shall apply to such late payment, and provided further that if the Liquidity Exceptions apply and the Corporate Taxpayer declares or pays any dividend of cash to its shareholders while any Tax Benefit Payment is due and payable and remains unpaid, then the Liquidity Exceptions shall no longer apply.

(c) In the event of a Change of Control, the Corporate Taxpayer shall provide written notice of such Change of Control to the TRA Parties in accordance with the procedures set forth in Section 2.7 of the Exchange Agreement and the TRA Party Representative shall have the option, upon written notice to the Corporate Taxpayer, to cause acceleration of all unpaid payment obligations with respect to Units that have been Exchanged prior to or in connection with such Change of Control, which shall be calculated as if an Early Termination Notice had been delivered on the date of such Change of Control and shall include, without duplication, (1) the Early Termination Payments calculated with respect to such TRA Parties as if the Early Termination Date is the date of such Change of Control, (2) any Tax Benefit Payment due and payable and that remains unpaid as of the date of such Change of Control, and (3) any Tax Benefit Payment in respect of any TRA Party due for the Taxable Year ending with or including the date of such Change of Control. In the event of a Change of Control, any Early Termination Payment described in the preceding sentence shall be calculated utilizing the Valuation Assumptions (a), (b), (c), substituting in each case and in the lead-in to such definition, the terms "date of a Change of Control" for an "Early Termination Date." Any Exchanges with respect to which a payment has been made under this Section 4.1(c) shall be excluded in calculating any future Tax Benefit Payments, or Early Termination Payments, and this Agreement shall have no further application to such Exchanges.

Section 4.2 <u>Early Termination Notice</u>. If the Corporate Taxpayer chooses to exercise its right of early termination in accordance with Section 4.1(a) above, the Corporate Taxpayer shall deliver to each TRA Party a notice ("<u>Early Termination Notice</u>") and a schedule (the "<u>Early Termination Schedule</u>") specifying the Corporate Taxpayer's intention to exercise such right and showing in reasonable detail the calculation of the Early Termination Payment(s) due for each TRA Party. Each Early Termination Schedule shall become final and binding on all parties thirty (30) calendar days from the first date on which all TRA Parties are treated as having received such Schedule or amendment thereto under Section 7.1 unless, prior to such thirtieth calendar day, the TRA Party Representative (a) provides the Corporate Taxpayer with notice of a material Objection to such Schedule made in good faith ("<u>Material Objection Notice</u>") or (b) provides a written waiver of such right of a Material Objection Notice, in which case such Schedule will become binding on the date the waiver is received by the Corporate Taxpayer (the "<u>Early Termination Effective Date</u>"). If the Corporate Taxpayer and the TRA Party Representative, for any reason, are unable to successfully resolve the issues raised in such notice within thirty (30) calendar days after receipt by the Corporate Taxpayer of the Material Objection Notice, the Corporate Taxpayer and the TRA Party Representative shall employ the Reconciliation Procedures in which case such Schedule shall become binding ten (10) calendar days after the conclusion of the Reconciliation Procedures.

## Section 4.3 Payment upon Early Termination.

(a) Within three (3) Business Days after the Early Termination Effective Date, the Corporate Taxpayer shall pay to each TRA Party an amount equal to the Early Termination Payment in respect of such TRA Party. Such payment shall be made by wire transfer of immediately available funds to a bank account or accounts designated by each TRA Party or as otherwise agreed by the Corporate Taxpayer and such TRA Party.

(b) "<u>Early Termination Payment</u>" in respect of a TRA Party shall equal the present value, discounted at the Early Termination Rate as of the applicable Early Termination Effective Date, of all Tax Benefit Payments in respect of such TRA Party that would be required to be paid by the Corporate Taxpayer beginning from the Early Termination Date and assuming that (i) the Valuation Assumptions in respect of such TRA Party are applied and (ii) for each Taxable Year, the Tax Benefit Payment is paid on the due date, assuming an extension, of the U.S. federal income tax return of the Corporate Taxpayer and (iii) for purposes of calculating the Early Termination Rate, LIBOR shall be LIBOR as of the date of the Early Termination Notice. For the avoidance of doubt, an Early Termination Payment shall be made to each applicable TRA Party regardless of whether such TRA Party has exchanged all of its Units as of the Early Termination Effective Date.

#### ARTICLE V

## SUBORDINATION AND LATE PAYMENTS

Section 5.1 <u>Subordination</u>. Notwithstanding any other provision of this Agreement to the contrary, any Tax Benefit Payment or Early Termination Payment required to be made by the Corporate Taxpayer to the TRA Parties under this Agreement shall rank subordinate and junior in right of payment to any principal, interest or other amounts due and payable in respect of any obligations in respect of indebtedness for borrowed money of the Corporate Taxpayer and its Subsidiaries ("<u>Senior Obligations</u>"), shall rank senior in right of payment to any principal, interest or other amounts due and payable in respect of any Other Tax Receivable Obligation, and shall rank *pari passu* with all current or future unsecured obligations of the Corporate Taxpayer that are not Senior Obligations or Other Tax Receivable Obligations. To the extent that any payment under this Agreement is not permitted to be made at the time payment is due as a result of this Section 5.1 and the terms of agreements governing Senior Obligations, such payment obligation nevertheless shall accrue for the benefit of TRA Parties and the Corporate Taxpayer shall make such payments at the first opportunity that such payment. To the extent the Corporate Taxpayer or its Subsidiaries (including OpCo and its Subsidiaries) incur, create or assume any Senior Obligations after the date hereof, the Corporate Taxpayer shall, and shall cause its Subsidiaries to, use commercially reasonable efforts to ensure that such indebtedness permitts the amounts payable hereunder to be paid.

Section 5.2 <u>Late Payments by the Corporate Taxpayer</u>. The amount of all or any portion of any Tax Benefit Payment or Early Termination Payment not made to the TRA Parties when due under the terms of this Agreement, whether as a result of Section 5.1 or otherwise, shall be payable together with any interest thereon, computed at the Default Rate and commencing from the date on which such Tax Benefit Payment or Early Termination Payment was due and payable.

#### **ARTICLE VI**

#### **NO DISPUTES; CONSISTENCY; COOPERATION**

Section 6.1 <u>Participation in the Corporate Taxpayer's and OpCo's Tax Matters</u>. Except as otherwise provided in this Agreement, the Merger Agreement or the LLC Agreement, the Corporate Taxpayer shall have full responsibility for, and sole discretion over, all Tax matters concerning the Corporate Taxpayer and OpCo, including the preparation, filing or amending of any Tax Return and defending, contesting or settling any issue pertaining to Taxes. Notwithstanding the foregoing, the Corporate Taxpayer shall notify the TRA Party Representative in writing of the commencement of, and keep the TRA Party Representative reasonably informed with respect to, the portion of any audit of the Corporate Taxpayer and OpCo or any of OpCo's Subsidiaries by a Taxing Authority the outcome of which is reasonably expected to affect the rights and obligations of a TRA Party under this Agreement, and shall provide to the TRA Party Representative reasonable opportunity to participate in or provide information and other input to the Corporate Taxpayer, OpCo and their respective advisors concerning the conduct of any such portion of such audit; provided, however, that the Corporate Taxpayer and OpCo shall not be required to take any action that is inconsistent with any provision of the LLC Agreement.

Section 6.2 <u>Consistency</u>. The Corporate Taxpayer and the TRA Parties agree to report and cause to be reported for all purposes, including federal, state, local and foreign Tax purposes and financial reporting purposes, all Tax-related items (including the Basis Adjustments and each Tax Benefit Payment) in a manner consistent with that set forth in this Agreement or specified by the Corporate Taxpayer in any Schedule (or Amended Schedule, as applicable) required to be provided by or on behalf of the Corporate Taxpayer under this Agreement that is final and binding on the parties unless otherwise required by law. The Corporate Taxpayer shall (and shall cause OpCo and its other Subsidiaries to) use commercially reasonable efforts (for the avoidance of doubt, taking into account the interests and entitlements of all TRA Parties under this Agreement) to defend the Tax treatment contemplated by this Agreement and any Schedule (or Amended Schedule, as applicable) in any audit, contest or similar proceeding with any Taxing Authority.

Section 6.3 <u>Cooperation</u>. Each of the TRA Parties shall (a) furnish to the Corporate Taxpayer in a timely manner such information, documents and other materials as the Corporate Taxpayer may reasonably request for purposes of making any determination or computation necessary or appropriate under this Agreement, preparing any Tax Return or contesting or defending any audit, examination or controversy with any Taxing Authority, (b) make itself available to the Corporate Taxpayer and its representatives to provide explanations of documents and materials and such other information as the Corporate Taxpayer or its representatives may reasonably request in connection with any of the matters described in clause (a) above, and (c) reasonably cooperate in connection with any such matter, and the Corporate Taxpayer shall reimburse each such TRA Party for any reasonable third-party costs and expenses incurred pursuant to this Section.

## **ARTICLE VII**

#### **MISCELLANEOUS**

Section 7.1 <u>Notices</u>. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be deemed duly given and received (a) on the date of delivery if delivered personally, or by facsimile or email with confirmation of transmission by the transmitting equipment or (b) on the first Business Day following the date of dispatch if delivered by a recognized next-day courier service. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice:

If to the Corporate Taxpayer, to:

41 Discovery, Irvine, California 92618 Attn: General Counsel Email: legal@weedmaps.com

If to the TRA Parties, to the address and other contact information set forth in the records of OpCo from time to time.

Any party may change its address, fax number or email by giving the other party written notice of its new address, fax number or email in the manner set forth above.

Section 7.2 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties, it being understood that all parties need not sign the same counterpart. Delivery of an executed signature page to this Agreement by facsimile transmission shall be as effective as delivery of a manually signed counterpart of this Agreement.

Section 7.3 <u>Entire Agreement; Third Party Beneficiaries</u>. This Agreement (together with all Exhibits and Schedules to this Agreement), the Merger Agreement (together with the Ancillary Agreements) the LLC Agreement, and the Confidentiality Agreement constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof. This Agreement shall be binding upon and inure solely to the benefit of each party hereto and their respective successors and permitted assigns, and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

Section 7.4 <u>Governing Law</u>. This Agreement shall be governed by, and construed in accordance with, the law of the State of Delaware, without regard to the conflicts of laws principles thereof that would mandate the application of the laws of another jurisdiction.

Section 7.5 <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any law or public policy, all other terms and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

## Section 7.6 Successors; Assignment; Amendments; Waivers.

(a) Each TRA Party may assign any of its rights under this Agreement to any Person as long as such transferee has executed and delivered, or, in connection with such transfer, executes and delivers, a joinder to this Agreement, in form and substance reasonably satisfactory to the Corporate Taxpayer (the "Joinder Requirement"), agreeing to become a TRA Party for all purposes of this Agreement; provided, however, that to the extent any TRA Party sells, exchanges, distributes, or otherwise transfers Units to any Person (other than the Corporate Taxpayer or the OpCo) in accordance with the terms of the Exchange Agreement and/or LLC Agreement, such TRA Party shall have the option to assign to the transferee of such Units its rights under this Agreement with respect to such transferred Units; provided, further, that such transferee has satisfied the Joinder Requirement. For the avoidance of doubt, if a TRA Party transfers Units in accordance with the terms of the Exchange Agreement and/or LLC Agreement but does not assign to the transferee of such Units its rights under this Agreement with respect to such transferred Units, such TRA Party shall continue to be entitled to receive the Tax Benefit Payments arising in respect of a subsequent Exchange of such Units and such transferee may not enforce the provisions of this Agreement. Notwithstanding any other provision of this Agreement, an assignee of only rights to receive a Tax Benefit Payment in connection with an Exchange has no rights under this Agreement other than to enforce its right to receive a Tax Benefit Payment pursuant to this Agreement. The Corporate Taxpayer may not assign any of its rights or obligations under this Agreement to any Person (other than in connection with a Mandatory Assignment) without the prior written consent of the TRA Party Representative (not to be unreasonably withheld, conditioned or delayed). Any purported assignment in violation of the terms of this Section 7.6 shall be null and void.

(b) No provision of this Agreement may be amended unless such amendment is approved in writing by the Corporate Taxpayer (as determined by the TRA Disinterested Majority) and by the TRA Party Representative and no provision of this Agreement may be waived unless such waiver is in writing and signed by the party against whom the waiver is to be effective (or, in the case of a waiver by all TRA Parties, signed by the TRA Party Representative); <u>provided</u> that no such amendment or waiver shall be effective if such amendment or waiver will have a disproportionate and adverse effect on the payments certain TRA Parties will or may receive under this Agreement unless such amendment or waiver is consented in writing by the TRA Parties disproportionately and adversely affected who would be entitled to receive at least majority of the total amount of the Early Termination Payments payable to all TRA Parties disproportionately and adversely affected hereunder if the Corporate Taxpayer had exercised its right of early termination on the date of the most recent Exchange prior to such amendment or waiver (excluding, for purposes of this sentence, all payments made to any TRA Party pursuant to this Agreement since the date of such most recent Exchange).

(c) All of the terms and provisions of this Agreement shall be binding upon, shall inure to the benefit of and shall be enforceable by the parties hereto and their respective successors, assigns, heirs, executors, administrators and legal representatives. The Corporate Taxpayer shall require and cause any direct or indirect successor (whether by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Corporate Taxpayer, by written agreement, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Corporate Taxpayer would be required to perform if no such succession had taken place (any such assignment, a "<u>Mandatory Assignment</u>").

Section 7.7 <u>Titles and Subtitles</u>. The titles of the sections and subsections of this Agreement are for convenience of reference only and are not to be considered in construing this Agreement.

# Section 7.8 Waiver of Jury Trial, Jurisdiction.

(a) EACH PARTY TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ALL RIGHTS TO TRIAL BY JURY IN ANY PROCEEDING BROUGHT TO RESOLVE ANY DISPUTE BETWEEN OR AMONG ANY OF THE PARTIES (WHETHER ARISING IN CONTRACT, TORT OR OTHERWISE) ARISING OUT OF, CONNECTED WITH, RELATED OR INCIDENTAL TO THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT AND/OR THE RELATIONSHIPS ESTABLISHED AMONG THE PARTIES HEREUNDER. THE PARTIES HERETO FURTHER WARRANT AND REPRESENT THAT EACH HAS REVIEWED THIS WAIVER WITH ITS LEGAL COUNSEL, AND THAT EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL.

(b) Subject to Section 7.9, each of the parties submits to the exclusive jurisdiction of first, the Chancery Court of the State of Delaware or if such court declines jurisdiction, then to the Federal District Court for the District of Delaware, in any action, suit or proceeding arising out of or relating to this Agreement, agrees that all claims in respect of such action, suit or proceeding shall be heard and determined in any such court and agrees not to bring any action, suit or proceeding arising out of or relating in this Section 7.8, however, shall affect the right of any party to serve legal process in any other manner permitted by law or at equity. Each party agrees that a final judgment in any action, suit or proceeding so brought shall be conclusive and may be enforced by suit on the judgment or in any other manner provided by law or at equity. The parties hereby waive, to the fullest extent permitted by applicable law, any objection which they now or hereafter may have to personal jurisdiction or to the laying of venue of any such ancillary suit, action or proceeding brought in any court referred to in this Section 7.8 and such parties agree not to plead or claim the same.

Section 7.9 Reconciliation. In the event that the Corporate Taxpaver and the TRA Party Representative are unable to resolve a disagreement with respect to the matters (x) governed by Sections 2.3 and 4.2 or (y) described in the definition of "LIBOR" within the relevant period designated in this Agreement ("Reconciliation Dispute"), the Reconciliation Dispute shall be submitted for determination to a nationally recognized expert (the "Expert") in the particular area of disagreement mutually acceptable to both parties. The Expert shall be a partner or principal in a nationally recognized accounting or law firm, and unless the Corporate Taxpayer and the TRA Party Representative agree in writing otherwise, the Expert shall not, and the firm that employs the Expert shall not, have any material relationship with the Corporate Taxpayer or the TRA Party Representative or other actual or potential conflict of interest. If the Corporate Taxpayer and the TRA Party Representative are unable to agree on an Expert within fifteen (15) calendar days of receipt by the respondent(s) of written notice of a Reconciliation Dispute, the Expert shall be appointed by the International Chamber of Commerce Centre for Expertise. The Expert shall resolve any matter relating to the Basis Schedule or an amendment thereto or the Early Termination Schedule or an amendment thereto within thirty (30) calendar days and shall resolve any matter relating to a Tax Benefit Schedule or an amendment thereto within fifteen (15) calendar days or as soon thereafter as is reasonably practicable, in each case after the matter has been submitted to the Expert for resolution. Notwithstanding the preceding sentence, if the matter is not resolved before any payment that is the subject of a disagreement would be due (in the absence of such disagreement) or any Tax Return reflecting the subject of a disagreement is due, the undisputed amount shall be paid on the date prescribed by this Agreement and such Tax Return may be filed as prepared by the Corporate Taxpayer, subject to adjustment or amendment upon resolution. The costs and expenses relating to the engagement of such Expert or amending any Tax Return shall be borne by the Corporate Taxpayer except as provided in the next sentence. The Corporate Taxpaver and the TRA Party Representative shall bear their own costs and expenses of such proceeding, unless (i) the Expert adopts the TRA Party Representative's position, in which case the Corporate Taxpayer shall reimburse the TRA Party Representative for any reasonable out-of-pocket costs and expenses in such proceeding, or (ii) the Expert adopts the Corporate Taxpaver's position, in which case the TRA Party Representative shall reimburse the Corporate Taxpaver for any reasonable out-of-pocket costs and expenses in such proceeding. Any dispute as to whether a dispute is a Reconciliation Dispute within the meaning of this Section 7.9 shall be decided by the Expert. The Expert shall finally determine any Reconciliation Dispute and the determinations of the Expert pursuant to this Section 7.9 shall be binding on the Corporate Taxpayer and each of the TRA Parties and may be entered and enforced in any court having jurisdiction.

Section 7.10 Withholding. The Corporate Taxpayer shall be entitled to deduct and withhold from any payment payable pursuant to this Agreement such amounts as the Corporate Taxpayer is required to deduct and withhold with respect to the making of such payment under the Code or any provision of state, local or foreign tax law provided, however, that the Corporate Taxpayer shall use commercially reasonable efforts to notify and shall reasonably cooperate with the applicable TRA Party prior to the making of such deductions and withholding payments to determine whether any such deductions or withholding payments (other than any deduction or withholding required by reason of such TRA Party's failure to comply with the last sentence of this Section 7.10) are required under applicable law and in obtaining any available exemption or reduction of, or otherwise minimizing to the extent permitted by applicable law, such deduction and withholding. To the extent that amounts are so withheld and paid over to the appropriate Taxing Authority by the Corporate Taxpayer, such withholding was made. Each TRA Party shall promptly provide the Corporate Taxpayer with any applicable Tax forms and certifications (including IRS Form W-9) or the applicable version of IRS Form W-8) reasonably requested by the Corporate Taxpayer in connection with determining whether any such deductions and withholdings are required under the Code or any provision of state, local or foreign tax law.

#### Section 7.11 Admission of the Corporate Taxpayer into a Consolidated Group; Transfers of Corporate Assets.

(a) If the Corporate Taxpayer is or becomes a member of an affiliated, consolidated, combined or unitary group of corporations that files a consolidated, combined or unitary income Tax Return pursuant to Sections 1501 et seq. of the Code or any corresponding provisions of state, local or foreign Tax law, then: (i) the provisions of this Agreement shall be applied with respect to the group as a whole; and (ii) Tax Benefit Payments, Early Termination Payments and other applicable items hereunder shall be computed with reference to the consolidated, combined or unitary taxable income of the group as a whole.

(b) If any Person the income of which is included in the income of the Corporate Taxpaver or the Corporate Taxpaver's affiliated or consolidated group transfers one or more assets to a corporation (or a Person classified as a corporation for U.S. federal income tax purposes) with which such entity does not file a consolidated Tax Return pursuant to Section 1501 of the Code or any corresponding provisions of state, local or foreign Tax law, such Person, for purposes of calculating the amount of any Tax Benefit Payment or Early Termination Payment due hereunder, shall be treated as having disposed of such asset in a fully taxable transaction on the date of such transfer. The consideration deemed to be received in a transaction contemplated in the prior sentence shall be equal to the fair market value of the deemed transferred asset, plus (i) the amount of debt to which such asset is subject, in the case of a transfer of an encumbered asset or (ii) the amount of debt allocated to such asset, in the case of a transfer of a partnership interest. The transactions described in this Section 7.11(b) shall be taken into account in determining the Realized Tax Benefit or Realized Tax Detriment, as applicable, for such Taxable Year based on the income, gain or loss deemed allocated to the Corporate Taxpaver using the Non-Stepped Up Tax Basis of the Reference Assets in calculating its Hypothetical Tax Liability for such Taxable Year and using the actual Tax basis of the Reference Assets in calculating its Actual Tax Liability, determined using the "with and without" methodology. Thus, for example, in determining the Hypothetical Tax Liability of the Corporate Taxpayer the taxable income of the Corporate Taxpayer shall be determined by treating OpCo as having sold the applicable Reference Asset for its fair market value, recovering any basis applicable to such Reference Asset (using the Non-Stepped Up Tax Basis), while the Actual Tax Liability of the Corporate Taxpayer would be determined by recovering the actual Tax basis of the Reference Asset that reflects any Basis Adjustments. For purposes of this Section 7.11, a transfer of a partnership interest shall be treated as a transfer of the transferring partner's share of each of the assets and liabilities of that partnership.

#### Section 7.12 Confidentiality.

(a) Each TRA Party and each of their assignees acknowledges and agrees that the information of the Corporate Taxpayer is confidential and, except in the course of performing any duties as necessary for the Corporate Taxpayer and its Affiliates, as required by law or legal process or to enforce the terms of this Agreement, such person shall keep and retain in confidence in accordance with this Agreement, and not disclose to any Person, any confidential matters acquired pursuant to this Agreement of the Corporate Taxpayer and its Affiliates and successors, concerning OpCo and its Affiliates and successors or the Members, learned by the TRA Party heretofore or hereafter. This Section 7.12 shall not apply to (i) any information that has been made publicly available by the Corporate Taxpayer or any of its Affiliates, becomes public knowledge (except as a result of an act of the TRA Party in violation of this Agreement) or is generally known, (ii) the disclosure of information to the extent necessary for the TRA Party to assert its rights hereunder or defend itself in connection with any action or proceeding arising out of, or relating to, this Agreement, (iii) any information that was in the possession of, or becomes available to, the TRA Party from a source other than the Corporate Taxpayer, its Affiliates or its or their respective representatives (provided that such source is not known by the TRA Party to be bound by a legal, contractual or fiduciary confidentiality obligation not to disclose such information) and (iv) the disclosure of information to the extent necessary for the TRA Party to prepare and file its Tax Returns, to respond to any inquiries regarding the same from any governmental or taxing authority or to prosecute or defend any action, proceeding or audit by any governmental or taxing authority with respect to such returns. Notwithstanding anything to the contrary herein, each TRA Party and each of their assignees (and each employee, representative or other agent of the TRA Party or its assignees, as applicable) may disclose to any and all Persons the tax treatment and tax structure of the Corporate Taxpayer, OpCo and their Affiliates, and any of their transactions, and all materials of any kind (including opinions or other tax analyses) that are provided to the TRA Party relating to such tax treatment and tax structure.

(b) If a TRA Party or an assignee commits a breach, or threatens to commit a breach, of any of the provisions of this Section 7.12, the Corporate Taxpayer shall have the right and remedy to seek to have the provisions of this Section 7.12 specifically enforced by injunctive relief or otherwise by any court of competent jurisdiction without the need to post any bond or other security. Such rights and remedies shall be in addition to, and not in lieu of, any other rights and remedies available at law or in equity.

Section 7.13 <u>Change in Law</u>. Notwithstanding anything herein to the contrary, if, in connection with an actual or proposed change in law, a TRA Party reasonably believes that the existence of this Agreement could cause income (other than income arising from receipt of a payment under this Agreement) recognized by the TRA Party upon any Exchange by such TRA Party to be treated as ordinary income rather than capital gain (or otherwise taxed at ordinary income rates) for United States federal income tax purposes or would have other material adverse tax consequences to such TRA Party, then at the election of such TRA Party and to the extent specified by such TRA Party, this Agreement (i) shall cease to have further effect with respect to such TRA Party, (ii) shall not apply to an Exchange by such TRA Party occurring after a date specified by such TRA Party, or (iii) shall otherwise be amended in a manner determined by such TRA Party; <u>provided</u> that such amendment shall not result in an increase in payments under this Agreement at any time as compared to the amounts and times of payments that would have been due in the absence of such amendment.

Section 7.14 Independent Nature of TRA Parties' Rights and Obligations. The obligations of each TRA Party hereunder are several and not joint with the obligations of any other TRA Party, and no TRA Party shall be responsible in any way for the performance of the obligations of any other TRA Party hereunder. The decision of each TRA Party to enter into this Agreement has been made by such TRA Party independently of any other TRA Party. Nothing contained herein, and no action taken by any TRA Party pursuant hereto, shall be deemed to constitute the TRA Parties as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the TRA Parties are in any way acting in concert or as a group with respect to such obligations or the transactions contemplated hereby and the Corporate Taxpayer acknowledges that the TRA Parties are not acting in concert or as a group, and the Corporate Taxpayer will not assert any such claim, with respect to such obligations or the transactions contemplated hereby.

## Section 7.15 TRA Party Representative.

(a) Without further action of any of the Corporate Taxpayer, the TRA Party Representative or any TRA Party, and as partial consideration in respect of the benefits conferred by this Agreement, the TRA Party Representative is hereby irrevocably constituted and appointed as the TRA Party Representative, with full power of substitution, to take any and all actions and make any decisions required or permitted to be taken by the TRA Party Representative under this Agreement.

(b) If at any time the TRA Party Representative shall incur out of pocket expenses in connection with the exercise of its duties hereunder, upon written notice to the Corporate Taxpayer from the TRA Party Representative of documented costs and expenses (including fees and disbursements of counsel and accountants) incurred by the TRA Party Representative in connection with the performance of its rights or obligations under this Agreement and the taking of any and all actions in connection therewith, the Corporate Taxpayer shall reduce the future payments (if any) due to the TRA Parties hereunder pro rata by the amount of such expenses which it shall instead remit directly to the TRA Party Representative. In connection with the performance of its rights and obligations under this Agreement and the taking of any and all actions therewith, the TRA Party Representative shall not be required to expend any of its own funds (though, for the avoidance of doubt but without limiting the provisions of this Section 7.15(b), it may do so at any time and from time to time in its sole discretion).

(c) The TRA Party Representative shall not be liable to any TRA Party for any act of the TRA Party Representative arising out of or in connection with the acceptance or administration of its duties under this Agreement, except to the extent any liability, loss, damage, penalty, fine, cost or expense is actually incurred by such TRA Party as a proximate result of the bad faith or willful misconduct of the TRA Party Representative (it being understood that any act done or omitted pursuant to the advice of legal counsel shall be conclusive evidence of such good faith judgment). The TRA Party Representative shall not be liable for, and shall be indemnified by the TRA Parties (on a several but not joint basis) for, any liability, loss, damage, penalty or fine incurred by the TRA Party Representative (and any cost or expense incurred by the TRA Party Representative in connection therewith and herewith and not previously reimbursed pursuant to subsection (b) above) arising out of or in connection with the acceptance or administration of its duties under this Agreement, and such liability, loss, damage, penalty, fine, cost or expense shall be treated as an expense subject to reimbursement pursuant to the provisions of subsection (b) above, except to the extent that any such liability, loss, damage, penalty, fine, cost or expense is the proximate result of the bad faith or willful misconduct of the TRA Party Representative (it being understood that any act done or omitted pursuant to the advice of legal counsel shall be conclusive evidence of such good faith judgment); provided, however, in no event shall any TRA Party be obligated to indemnify the TRA Party Representative hereunder for any liability, loss, damage, penalty, fine, cost or expense to the extent (and only to the extent) that the aggregate amount of all liabilities, losses, damages, penalties, fines, costs and expenses indemnified by such TRA Party hereunder is or would be in excess of the aggregate payments under this Agreement actually remitted to such TRA Party.

(d) Subject to Section 7.6(b), a decision, act, consent or instruction of the TRA Party Representative shall constitute a decision of all TRA Parties and shall be final, binding and conclusive upon each TRA Party, and the Corporate Taxpayer may rely upon any decision, act, consent or instruction of the TRA Party Representative as being the decision, act, consent or instruction of each TRA Party. The Corporate Taxpayer is hereby relieved from any liability to any person for any acts done by the Corporate Taxpayer in accordance with any such decision, act, consent or instruction of the TRA Party Representative.

[The remainder of this page is intentionally blank]

# **CORPORATE TAXPAYER:**

#### WM TECHNOLOGY, INC.

By:/s/ Christopher BealsName:Christopher BealsTitle:Chief Executive Officer

## TRA PARTY REPRESENTATIVE:

#### GHOST MEDIA GROUP, LLC

By: /s/ Justin Hartfield Name: Justin Hartfield Title: Manager

By: /s/ Doug Francis

Name: Doug Francis Title: Manager

# TRA PARTIES:

## **CHRISTOPHER BEALS**

/s/ Christopher Beals (Signature)

# TRA PARTIES:

#### GHOST MEDIA GROUP, LLC

By: /s/ Justin Hartfield Name: Justin Hartfield Title: Manager

By: /s/ Doug Francis

Name: Doug Francis Title: Manager

# TRA PARTIES:

## 420 CASA WAY, LLC

By:/s/ Richard HamName:Richard HamTitle:Managing Partner

### TRA PARTIES:

#### CALILOVE LLC

By: /s/ Sabin Burrell Name: Sabin Burrell Title: Manager

### TRA PARTIES:

#### BLACK DOG MANAGEMENT, LLC

By: /s/ Sabin Burrell Name: Sabin Burrell Title: Manager

### TRA PARTIES:

# CERES HOLDINGS GROUP, LLC

By: /s/ William Rubin Name: William Rubin Title: General Partner

#### TRA PARTIES:

### CLAY W. HAMLIN, IV

/s/ Clay W. Hamlin, IV (Signature)

### TRA PARTIES:

### **DOUG FRANCIS**

/s/ Doug Francis (Signature)

### TRA PARTIES:

#### FCP VENTURES IV, LLC

By: FCP Ventures IV Manager LLC

By: /s/ Nirmal Roy

Name: Nirmal Roy Title: Manager

### TRA PARTIES:

#### GENCO INCENTIVES, LLC

By: /s/ Doug Francis Name: Doug Francis Title: Managing Member

#### TRA PARTIES:

#### INCEPTACON-IV, LLC

By: The Inception Companies, LLC Its: Manager

By: /s/ Brent Cox

Name: Brent Cox Title: Manager

### TRA PARTIES:

#### JUSTIN HARTFIELD

/s/ Justin Hartfield (Signature)

### TRA PARTIES:

#### LBCW HOLDINGS, LP

By:/s/ Stephen ChangName:Stephen ChangTitle:Chief Investment Officer

### TRA PARTIES:

# **OBM HOLDINGS LLC**

By: /s/ Brent Cox Name: Brent Cox Title: Manager

### TRA PARTIES:

#### **REDWOOD INVESTMENT HOLDINGS, LLC**

By: /s/ JJ

Name: Title:

#### TRA PARTIES:

SCP OPPORTUNITIES FUND LLC – SERIES A

By:Sunderland Capital Partners LP<br/>ManagerBy:Sunderland GP LLC<br/>General Partner

By:/s/ Thomas BushleyName:Thomas BushleyTitle:Managing Member

### TRA PARTIES:

#### SEQUOIA INVESTMENT HOLDINGS LLC

By: /s/ Steven Landau Name: Steven Landau Title: Manager

### TRA PARTIES:

#### SEVEN EIGHT INVESTMENTS, LLC

By: /s/ Desiree Hedge

Name: Desiree Hedge Title: President

### TRA PARTIES:

### WM FOUNDERS LEGACY I, LLC

By: /s/ Justin Hartfield Name: Justin Hartfield Title: Board Member

### TRA PARTIES:

### WM FOUNDERS LEGACY I, LLC

By: /s/ Doug Francis Name: Doug Francis Title: Member

### TRA PARTIES:

#### WM FOUNDERS LEGACY II, LLC

By: /s/ Justin Hartfield Name: Justin Hartfield Title: Board Member

### TRA PARTIES:

### WM FOUNDERS LEGACY II, LLC

By: /s/ Doug Francis

Name: Doug Francis Title: Member

### TRA PARTIES:

#### WM INVESTMENT PARTNERS, LLC

By:/s/ Eric LindbergName:Eric LindbergTitle:Chief Executive Officer

### TRA PARTIES:

# WM INVESTMENT PARTNERS II, LLC

By: /s/ Eric Lindberg Name: Eric Lindberg

Title: Chief Executive Officer

### TRA PARTIES:

#### YOUNG FAMILY LIMITED PARTNERSHIP

By:/s/ Morris YoungName:Morris YoungTitle:Trustee

#### FOURTH AMENDED AND RESTATED OPERATING AGREEMENT OF WM HOLDING COMPANY, LLC

#### a Delaware limited liability company

#### Dated as of June 16, 2021

THE LIMITED LIABILITY COMPANY UNITS OF WM HOLDING COMPANY, LLC HAVE NOT BEEN REGISTERED UNDER THE U.S. SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION OR ANY OTHER APPLICABLE SECURITIES LAWS AND ARE BEING SOLD IN RELIANCE UPON EXEMPTIONS FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT AND SUCH LAWS. SUCH UNITS MUST BE ACQUIRED FOR INVESTMENT ONLY AND MAY NOT BE OFFERED FOR SALE, PLEDGED, HYPOTHECATED, SOLD, ASSIGNED OR TRANSFERRED AT ANY TIME EXCEPT IN COMPLIANCE WITH (I) THE SECURITIES ACT, ANY APPLICABLE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION, AND ANY OTHER APPLICABLE SECURITIES LAWS; (II) THE TERMS AND CONDITIONS OF THIS FOURTH AMENDED AND RESTATED OPERATING AGREEMENT; AND (III) ANY OTHER TERMS AND CONDITIONS AGREED TO IN WRITING BETWEEN THE MANAGING MEMBER AND THE APPLICABLE MEMBER. THE UNITS MAY NOT BE TRANSFERRED OF RECORD EXCEPT IN COMPLIANCE WITH SUCH LAWS, THIS FOURTH AMENDED AND RESTATED OPERATING AGREEMENT, AND ANY OTHER TERMS AND CONDITIONS AGREED TO IN WRITING BETWEEN THE MANAGING MEMBER AND THE APPLICABLE MEMBER. THE UNITS MAY NOT BE TRANSFERRED OF RECORD EXCEPT IN COMPLIANCE WITH SUCH LAWS, THIS FOURTH AMENDED AND RESTATED OPERATING AGREEMENT, AND ANY OTHER TERMS AND CONDITIONS AGREED TO IN WRITING BY THE MANAGING MEMBER AND THE APPLICABLE MEMBER. THEREFORE, PURCHASERS AND OTHER TRANSFERES OF SUCH UNITS WILL BE REQUIRED TO BEAR THE RISK OF THEIR INVESTMENT OR ACQUISITION FOR AN INDEFINITE PERIOD OF TIME.

# TABLE OF CONTENTS

		Page		
ARTICLE I DEFINITIONS				
1.01	Definitions	2		
ARTICLE II FORMATION, TERM, PURPOSE AND POWERS				
2.01	Formation	12		
2.02	Name	12		
2.03	Term	12		
2.04	Offices	12		
2.05	Agent for Service of Process; Existence and Good Standing; Foreign Qualification.	12		
2.06	Business Purpose	13		
2.07	Powers of the Company	13		
2.08	Members; Reclassification; Admission of New Members	13		
2.09	Resignation	13		
2.10	Representations of Members	13		
	MANAGEMENT	15		
3.01	Managing Member	15		
3.02	Compensation	15		
3.03	Expenses	15		
3.04	Officers	16		
3.05	Authority of Members	16		
3.06	Action by Written Consent or Ratification	16		
3.07	Restrictions on Termination Transactions	17		
ARTICLE IV DISTRIBUTIONS		18		
4.01	Distributions	18		
4.02	Liquidation Distribution	19		
4.03	Limitations on Distribution	19		
4.04	Earnout Company Units	19		
4.05	Use of Distribution Funds	20		
ARTICLE V C	ARTICLE V CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS; TAX ALLOCATIONS; TAX MATTERS			
5.01	Initial Capital Contributions	20		
5.02	No Additional Capital Contributions	20		
5.03	Capital Accounts	20		
5.04	Allocations of Profits and Losses	21		
5.05	Special Allocations	21		
5.05				

# TABLE OF CONTENTS (continued)

			Page
!	5.06	Tax Allocations	23
!	5.07	Tax Advances	24
!	5.08	Partnership Representative.	24
!	5.09	Other Allocation Provisions	26
!	5.10	Survival	26
ARTICL	E VI BO	OOKS AND RECORDS; REPORTS	26
	6.01	Books and Records	26
ARTICL	E VII C	OMPANY UNITS	28
	7.01	Units.	28
	7.02	Register	29
	7.03	Registered Members	29
	7.04	Issuances, Repurchases and Redemptions, Recapitalizations.	29
	7.05	Class P Units.	31
	7.06	Triggering Events for Management Members.	33
ARTICL	EVIII	<b>FRANSFER RESTRICTIONS</b>	35
8	8.01	Member Transfers	35
1	8.02	Mandatory Exchanges	36
1	8.03	Encumbrances	36
1	8.04	Further Restrictions.	36
1	8.05	Rights of Assignees	38
1	8.06	Admissions, Resignations and Removals	38
1	8.07	Admission of Assignees as Substitute Members	38
1	8.08	Resignation and Removal of Members	39
1	8.09	Withholding	39
1	8.10	Allocations in Respect of Transferred Units	39
ARTICL	E IX DI	SSOLUTION, LIQUIDATION AND TERMINATION	39
9	9.01	No Dissolution	39
	9.02	Events Causing Dissolution	39
2	9.03	Distribution upon Dissolution	40
	9.04	Time for Liquidation	40
:	9.05	Termination	41
:	9.06	Claims of the Members	41
1	9.07	Survival of Certain Provisions	41

# TABLE OF CONTENTS (continued)

			Page		
ARTICI	ARTICLE X LIABILITY AND INDEMNIFICATION 41				
	10.01	Liability of Members	41		
	10.02	Indemnification.	42		
ARTICI	LE XI M	IISCELLANEOUS	44		
	11.01	Severability	44		
	11.02	Notices	44		
	11.03	Cumulative Remedies	45		
	11.04	Binding Effect	45		
	11.05	Interpretation	45		
	11.06	Counterparts	45		
	11.00	Further Assurances	45		
	11.07		45		
		Entire Agreement			
	11.09	Governing Law	45		
	11.10	Submission to Jurisdiction; Waiver of Jury Trial.	46		
	11.11	Expenses	46		
	11.12	Amendments and Waivers	47		
	11.13	No Third Party Beneficiaries	48		
	11.14	Headings	48		
	11.15	Power of Attorney	48		
	11.16	Separate Agreements; Schedules	48		
	11.17	Partnership Status	48		
	11.18	Delivery by Facsimile or Email	49		

-iii-

#### FOURTH AMENDED AND RESTATED OPERATING AGREEMENT OF WM HOLDING COMPANY, LLC

THIS FOURTH AMENDED AND RESTATED OPERATING AGREEMENT (this "Agreement") of WM Holding Company, LLC (the "Company"), is made as of June 16, 2021 (the "Effective Date") by and among WM Technology, Inc., a Delaware corporation, as the Managing Member, and the Members set forth on Schedule I hereto and each other person who is or at any time becomes a Member in accordance with the terms of this Agreement and the Act.

#### RECITALS

**W**HEREAS, the Company was formed as a limited liability company pursuant to the Act upon the filing of the Certificate in the office of the Secretary of State of the State of Delaware on September 4, 2014;

WHEREAS, the Company entered into that certain Restructuring Agreement, dated November 1, 2014 (the "*Original Agreement*"), by and between the Company and certain of the Members, which was amended by that certain First Amendment to the Operating Agreement of WM Holding Company, LLC, entered into by the Company and certain of the Members;

WHEREAS, the Original Agreement was amended and restated in its entirety by that certain Amended and Restated Limited Liability Company Operating Agreement of WM Holding Company, LLC, effective as of October 30, 2015, as amended by that certain Amendment No. 1, dated as of March 22, 2016, and effective as of October 30, 2015, and that certain Amendment No. 2, effective as of March 23, 2017 (collectively, the "*First A&R Operating Agreement*");

WHEREAS, the First A&R Operating Agreement was amended and restated in its entirety by that certain Second Amended and Restated Operating Agreement of WM Holding Company, LLC, effective as of May 7, 2018 (the "Second A&R Operating Agreement");

WHEREAS, the Second A&R Operating Agreement was amended and restated in its entirety by that certain Third Amended and Restated Operating Agreement of WM Holding Company, LLC, effective as of August 15, 2018 (the "*Existing Agreement*");

WHEREAS, concurrently with the effectiveness of this Agreement, in accordance with the Agreement and Plan of Merger, dated as of December 10, 2020 (the "*Merger Agreement*"), by and among Silver Spike Acquisition Corp. (the "*Parent*"), Silver Spike Merger Sub LLC (the "*Merger Sub*"), the Company and Ghost Media Group, LLC, solely in its capacity as the Holder Representative, Merger Sub has merged with and into the Company (the "*Merger*"), with the Company being the entity surviving the Merger;

WHEREAS, pursuant to the Merger, (i) each of the Class A-1 Units, Class A-2 Units and Class A-3 Units (as each is defined in the Existing Agreement) outstanding prior to the effectiveness of this Agreement were cancelled and certain of the holders thereof received the number of Class A Units set forth opposite such Member's name on Schedule I hereto and (ii) each of the Class B Units (as defined in the Existing Agreement) outstanding prior to the effectiveness of this Agreement were cancelled and the holders thereof received the number of Class P Units set forth opposite such Member's name on Schedule I hereto. If hereto, in each case, in accordance with <u>Section 4.01</u> of the Merger Agreement; and

WHEREAS, pursuant to the Merger Agreement, (i) the Members have agreed to amend and restate the Existing Agreement in its entirety as set forth herein and (ii) WM Technology, Inc., by its execution and delivery of this Agreement, is hereby admitted to the Company as a Member and is hereby substituted as Managing Member, and in such capacity shall have the rights and obligations as provided in this Agreement.

**Now, THEREFORE,** in consideration of the premises and agreements of the parties set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Members and the Managing Member hereby agree to amend and restate the Existing Agreement to read in its entirety as follows:

### ARTICLE I

#### **DEFINITIONS**

**1.01 Definitions**. Capitalized terms used herein without definition have the following meanings (such meanings being equally applicable to both the singular and plural form of the terms defined):

"*Act*" means, the Delaware Limited Liability Company Act, 6 Del. C. Section 18-101, et seq., as it may be amended from time to time.

"Adjusted Capital Account Balance" means, with respect to each Member, the balance in such Member's Capital Account adjusted (i) by taking into account the adjustments, allocations and distributions described in Treasury Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) and (6); and (ii) by adding to such balance such Member's share of Company Minimum Gain and Member Nonrecourse Debt Minimum Gain, determined pursuant to Treasury Regulations Sections 1.704-2(g) and 1.704-2(i) (5), and any amounts such Member is obligated to restore pursuant to any provision of this Agreement or by applicable Law. The foregoing definition of Adjusted Capital Account Balance is intended to comply with the provisions of Treasury Regulations Section 1.704-1(b)(2)(ii)(d) and shall be interpreted consistently therewith.

"*Affiliate*" means, with respect to a specified Person, any other Person that directly, or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such specified Person.

"*Agreement*" has the meaning set forth in the preamble of this Agreement.

"Assignee" has the meaning set forth in Section 8.05.

"Assumed Tax Rate" means the highest effective marginal combined U.S. federal, state and local income tax rate (including, without limitation, the tax imposed under Section 1411 of the Code on net investment income) for a Fiscal Year prescribed for an individual or corporate resident in California or New York, New York (whichever results in the application of the highest state and local tax rate for a given type of income), and taking into account (a) the limitations imposed on the deductibility of expenses and other items, (b) the character (e.g., long-term or short-term capital gain or ordinary or exempt income) of the applicable income, and (c) the deductibility of state and local income taxes, to the extent applicable, but not taking into account any deduction under Section 199A of the Code or any similar state or local law), as determined in good faith by the Managing Member. For the avoidance of doubt, the Assumed Tax Rate shall be the same for all Members.

"Available Cash" means, as of a particular date, the amount of cash on hand which the Managing Member, in its reasonable discretion, deems available for distribution to the Members, taking into account all debts, liabilities and obligations of the Company then due and amounts that the Managing Member, in its reasonable discretion, deems necessary to expend or retain for working capital or to place into reserves for customary and usual claims with respect to the Company's operations.

"Board" means the Board of Directors of the Managing Member.

*"Capital Account"* means the separate capital account maintained for each Member in accordance with *Section* **5.03** hereof.

*"Capital Contribution"* means, with respect to any Member, the aggregate amount of money contributed to the Company and the initial Carrying Value of any property (other than money), net of any liabilities assumed by the Company upon contribution or to which such property is subject, contributed to the Company pursuant to *Article V*.

"*Carrying Value*" means, with respect to any Company asset, the asset's adjusted basis for U.S. federal income tax purposes, except that the initial carrying value of assets contributed to the Company shall be their respective gross fair market values on the date of contribution as determined by the Managing Member in its reasonable discretion, and the Carrying Values of all Company assets shall be adjusted to equal their respective fair market values, in accordance with the rules set forth in Treasury Regulation Section 1.704-1(b)(2)(iv)(f), except as otherwise provided herein, as of: (a) the date of the acquisition of any additional limited liability company interest in the Company by any new or existing Member in exchange for more than a de minimis Capital Contribution; (b) the date of the distribution of more than a de minimis amount of Company assets to a Member as consideration for an interest in the Company; (c) the liquidation of the Company within the meaning of Treasury Regulations Section 1.704-1(b)(2)(ii)(g); (d) in connection with the grant of an interest in the Company (other than a de minimis interest) as consideration for the provision of services to or for the benefit of the Company by an existing member acting in a partner capacity, or by a new Member acting in a partner capacity in anticipation of being a Member, (e) the acquisition of an interest in the Company upon the exercise of a noncompensatory option in accordance with Treasury Regulations Section 1.704-1(b)(2)(iv) (s); (f) on the Effective Date in connection with the closing of the transactions contemplated by the Merger Agreement, or (g) any other date specified in the Treasury Regulations; provided, however, that adjustments pursuant to clauses (a), (b), (d) (other than the issuance of LTIP Units) and (f) above shall be made only if such adjustments are deemed necessary or appropriate by the Managing Member in its reasonable discretion to reflect the relative economic interests of the Members; and *provided further*, if any noncompensatory option is outstanding, Carrying Values shall be adjusted in accordance with Treasury Regulations Sections 1.704-1(b)(2)(iv)(f)(1) and 1.704-1(b)(2)(iv)(h)(2). The Carrying Value of any Company asset distributed to any Member shall be adjusted immediately before such distribution to equal its fair market value. In the case of any asset that has a Carrying Value that differs from its adjusted tax basis, Carrying Value shall be adjusted by the amount of depreciation calculated for purposes of the definition of "Profits" and "Losses" rather than the amount of depreciation determined for U.S. federal income tax purposes, and depreciation shall be calculated by reference to Carrying Value rather than tax basis once Carrying Value differs from tax basis. The Carrying Value of Company assets shall be increased (or decreased) to reflect any adjustments to the adjusted basis of such assets pursuant to Code Section 734(b), but only to the extent that such adjustments are taken into account in determining Capital Accounts pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(m); provided, however, that Carrying Values shall not be adjusted pursuant to this sentence to the extent that the Managing Member reasonably determines that an adjustment pursuant to the first sentence of this definition is necessary or appropriate in connection with the transaction that would otherwise result in an adjustment pursuant to this sentence.

"Certificate" means the Certificate of Formation of the Company as filed in the office of the Secretary of State of the State of Delaware on September 4, 2014, as amended.

"Class" means the classes of Units into which the limited liability company interests in the Company may be classified or divided from time to time by the Managing Member pursuant to the provisions of this Agreement. As of the date of this Agreement the only Classes are the Class A Units and Class P Units. Subclasses within a Class shall not be separate Classes for purposes of this Agreement. For all purposes hereunder and under the Act, only such Classes expressly established under this Agreement, including by the Managing Member in accordance with this Agreement, shall be deemed to be a class of limited liability company interests in the Company. For the avoidance of doubt, to the extent that the Managing Member holds limited liability company interests of any Class, the Managing Member shall not be deemed to hold a separate Class of such interests from any other Member because it is the Managing Member.

share.

"Class A Common Stock" means the Class A common stock of the Managing Member, par value \$0.0001 per

"Class A Percentage Interest" means, with respect to any Member, the quotient obtained by dividing the aggregate number of Class A Units then owned by such Member by the aggregate number of Class A Units then owned by all Members; provided that Unvested Units shall not be taken into account in determining such quotient.

"Class A/LTIP Percentage Interest" means, with respect to any Member, the quotient obtained by dividing the aggregate number of Class A Units and LTIP Units then owned by such Member by the aggregate number of Class A Units and LTIP Units then owned by all Members; provided that Unvested Units shall not be taken into account in determining such quotient.

"Class A Units" means the Units of limited liability company interest in the Company designated as the "Class A **Units**" herein and having the rights pertaining thereto as are set forth in this Agreement.

"Class A Unit Capital Account Amount" means, from time to time, the Capital Account a Member would have if such Member held a single Class A Unit.

"Class P Units" means the Units of limited liability company interest in the Company designated as the "Class P **Units**" herein and having the rights pertaining thereto as are set forth in this Agreement.

"Class P Unit Agreement" means a Class P Unit Agreement between the Company and a Management Member as in effect from time to time.

"*Class V Common Stock*" has the meaning set forth in the Managing Member Charter.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"*Company*" has the meaning set forth in the preamble of this Agreement.

"Company Minimum Gain" has the meaning ascribed to the term "partnership minimum gain" set forth in Treasury Regulations Sections 1.704-2(b)(2) and 1.704-2(d).

"Competing Activity" means (a) any type of business activity or operation which relates, directly or indirectly, or is otherwise competitive with, any of the businesses of the Company or any Affiliate of the Company, as such businesses may be conducted (or contemplated to be conducted) from time to time (including, for the avoidance of doubt) any promotional, service provider or any other type of remuneration-providing (whether monetary, in-kind or otherwise) relationship with any entity, business, company, enterprise, joint venture or other business which relates, directly or indirectly, or is otherwise competitive with, any of the businesses of the Company or any Affiliate of the Company, as such businesses may be conducted (or contemplated to be conducted) from time to time); (b) the direct or indirect ownership, receipt or holding of any Equity Interest, debt securities or other financial interest in any entity, business, company, enterprise, joint venture or other business competitive with, any of the businesses may be conducted (or contemplated to be conducted), or is otherwise competitive with, any of the businesses of the Company or any Affiliate of the Company or any Affiliate of the Company or any Affiliate of the Company, as such businesses may be conducted (or contemplated to be conducted) from time to time; (b) the direct or indirect ownership, receipt or holding of any Equity Interest, debt securities or other financial interest in any entity, business, company, enterprise, joint venture or other business which relates, directly or indirectly, or is otherwise competitive with, any of the businesses of the Company or any Affiliate of the Company, as such businesses may be conducted (or contemplated to be conducted) from time to time; or (c) engagement in any activity which has the general effect of being materially detrimental to the business, profitability or prospects of the Company or any Affiliate of the Company. Notwithstanding the foregoing, "Competing Activity," as such term is applied to an

"*Contingencies*" has the meaning set forth in Section 9.03(a).

"*Control*" (including the terms "Controlled by" and "under common Control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, as trustee or executor, by contract or otherwise, including, without limitation, the ownership, directly or indirectly, of securities having the power to elect a majority of the board of directors or similar body governing the affairs of such Person.

*"Covered Transaction"* means any liquidation, dissolution or winding up of the Company (whether occurring through one transaction or a series of related transactions, and whether voluntary or involuntary) and any other sale, redemption or Transfer of Units.

"*Designated Individual*" has the meaning set forth in Section 5.08.

"*Disability*" means the inability of a Person to exercise his or her rights or fulfill his or her obligations to the Company or its Affiliates on account of medically determinable physical or mental illness or incapacity for a period of one-hundred eighty (180) consecutive days or any one-hundred eight (180) days in any twelve (12)-month period, as determined by the Managing Member.

"*Disabling Event*" means the Managing Member ceasing to be the Managing Member of the Company.

*"Earnout Company Units"* has the meaning set forth in the Sponsor Letter Agreement. The Earnout Company Units issued and outstanding as of the Closing Date (as defined in the Merger Agreement) are held by the Managing Member, designated as such as set forth in Schedule I attached hereto.

"Effective Date" has the meaning set forth in the preamble of this Agreement.

*"Eligible Class P Unit"* means a Class P Unit, the Participation Threshold of which is zero (taking into account any adjustments described in clauses (i) and (ii) of Section 7.05(e)).

"*Encumbrance*" means any mortgage, hypothecation, claim, lien, encumbrance, conditional sales or other title retention agreement, right of first refusal, preemptive right, pledge, option, charge, security interest or other similar interest, easement, judgment or imperfection of title of any nature whatsoever.

"Equitized LTIP Series" means an LTIP Series composed of Equitized LTIP Series Units.

"*Equitized LTIP Series Units*" has the meaning set forth in Section 5.03(b).

"*Equity Interests*" means (a) capital stock, membership interests, partnership interests, other equity interests, rights to profits or revenue and any other similar interest in any corporation, partnership, limited liability company or other business entity, (b) any security or other interest convertible into or exchangeable or exercisable for any of the foregoing, whether at the time of issuance or upon the passage of time or the occurrence of some future event and (c) any warrant, option or other right (contingent or otherwise) to acquire any of the foregoing.

"*ERISA*" means The Employee Retirement Income Security Act of 1974, as amended.

*"Exchange Act"* means the U.S. Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

*"Exchange Agreement"* means the exchange agreement dated as of or about the date hereof among the Company, Managing Member, the other Members of the Company from time to time party thereto, and the other parties thereto, as amended from time to time.

*"Exchange Transaction"* means an exchange of (i) Class A Units and Class V Common Stock, or (ii) Class P Units that are Vested Units which are exchanged for Class A Units, in each case for shares of Class A Common Stock of the Managing Member pursuant to, and in accordance with, the Exchange Agreement (including pursuant to a Direct Exchange (as defined in the Exchange Agreement)).

"Existing Agreement" has the meaning set forth in the recitals of this Agreement.

"*Family Group*" means, with respect to a Person who is an individual, (a) such Person's spouse and direct descendants (whether natural or adopted) (collectively, for purposes of this definition, "*relatives*"), and (b) any trust, the trustee of which is such Person and which at all times is and remains solely for the benefit of such Person and/or such Person's relatives.

"First A&R Operating Agreement" has the meaning set forth in the Recitals of this Agreement.

*"Fiscal Year"* means, unless otherwise determined by the Managing Member in its sole discretion in accordance with *Section 11.12*, any twelve-month period commencing on January 1 and ending on December 31.

"GAAP" means accounting principles generally accepted in the United States of America as in effect from time to

time.

"*Incapacity*" means, with respect to any Person, the bankruptcy, dissolution, termination of such Person.

"Income Amount" has the meaning set forth in Section 4.01(c)(ii).

"Indemnitee" means (a) the Managing Member, (b) any additional or substitute Managing Member, (c) any Person who is or was a Partnership Representative, officer or director of the Managing Member or any additional or substitute Managing Member, (d) any Person that is required to be indemnified by the Managing Member as an "indemnitee" in accordance with the certificate of incorporation and/or bylaws of the Managing Member as in effect from time to time, (e) any officer or director of the Managing Member or any additional or substitute Managing Member who is or was serving at the request of the Managing Member or any additional or substitute Managing Member as an officer, director, employee, member, Member, Partnership Representative, agent, fiduciary or trustee of another Person; provided, that a Person shall not be an Indemnitee by reason of providing, on a fee-for-services basis, trustee, fiduciary or custodial services, (f) any Officer, (g) any other Person the Managing Member in its sole discretion designates as an "Indemnitee" for purposes of this Agreement, (h) any former officer or Manager of the Company pursuant to Section 9.02 of the Merger Agreement and (i) any heir, executor or administrator with respect to Persons named in clauses (a) through (h).

"*Law*" means any statute, law, ordinance, regulation, rule, code, executive order, injunction, judgment, decree or other order issued or promulgated by any national, supranational, state, federal, provincial, local or municipal government or any administrative or regulatory body with authority therefrom with jurisdiction over the Company or any Member, as the case may be.

"*Liquidation Agent*" has the meaning set forth in Section 9.03.

"LTIP Series Sub-Account" has the meaning set forth in Section 5.03(b).

"*LTIP Unit*" means a Unit which is designated as an LTIP Unit in the relevant Vesting Agreement or other documentation pursuant to which such LTIP Unit is granted or issued, having the rights, powers, privileges, restrictions, qualifications and limitations set forth in Schedule II hereto or in this Agreement in respect of the holder, as well as the relevant Vesting Agreement or other documentation pursuant to which such LTIP Unit is granted or issued. LTIP Units that are issued on the same date shall be designated as one or more separate series of LTIP Units (each such series, an "*LTIP Series*" and any LTIP Unit in respect of a given series, an "*LTIP Series Unit*").

*"LTIP Unit Member*" means any Person that holds LTIP Units or Class A Units resulting from a conversion of LTIP Units.

"*Malfeasance*" has the meaning set forth in the applicable Service Provider's individual agreement (including any employment agreement, offer letter, severance agreement or equity award agreement) and, if no individual agreement exists, it shall mean (a) any act of fraud, embezzlement, theft, dishonesty, or any misappropriation of any amount of money or other assets or property of the Company or any Affiliate of the Company, or of a customer or Service Provider of the same; (b) any willful failure to perform or negligence in the performance of one's duties or responsibilities to the Company or any Affiliate of the Company; (c) any act that brings the Company or any Affiliate of the Company into public disrepute in a manner that could material damage its business, or any other act which could have a material and adverse effect upon the business, interests or reputation of the Company or any Affiliate of the Company; (d) any conviction of, or plea of nolo contendere to, a felony or other crime involving moral turpitude; (e) any breach of any agreement with the Company or any Affiliate of the Company or any violation of any policy of the Company or any Affiliate of the Company in effect from time to time, or acting against the interest of such entities, including soliciting any present, former or future Service Provider of such entities; or (f) misuse of any confidential, secret, privileged or non-public information relating to the Company or any Affiliate of the Company may consider Malfeasance for purposes hereof or limit in any way the exercise of remedies for malfeasance under any applicable employment, consulting or service agreement.

"Management Member" has the meaning set forth in Section 7.05(a).

"*Managing Member*" means WM Technology, Inc., a corporation incorporated under the laws of the State of Delaware, or any successor Managing Member admitted to the Company in accordance with the terms of this Agreement, in its capacity as the managing member of the Company.

*"Managing Member Charter"* means the certificate of incorporation (or equivalent organizational document) as filed with the secretary of state (or equivalent governmental body or department) of the state in which the Managing Member is incorporated or formed, as applicable, as in effect and amended from time to time.

*"Mark-to-Market Gain"* means gain recognized for Capital Account purposes upon an adjustment to the Carrying Value of any asset, pursuant to the definition of Carrying Value.

*"Member*" means, at any time, each person listed as a Member (including the Managing Member) on the books and records of the Company, in each case for so long as he, she or it remains a Member of the Company as provided hereunder.

"*Member Nonrecourse Debt Minimum Gain*" means an amount with respect to each partner nonrecourse debt (as defined in Treasury Regulations Section 1.704-2(b)(4)) equal to the Company Minimum Gain that would result if such partner nonrecourse debt were treated as a nonrecourse liability (as defined in Treasury Regulations Section 1.704-2(b)(3)) determined in accordance with Treasury Regulations Section 1.704-2(i)(3).

*"Member Nonrecourse Deductions"* has the meaning ascribed to the term "partner nonrecourse deductions" set forth in Treasury Regulations Section 1.704-2(i)(2).

"Member's Required Tax Distribution" has the meaning set forth in Section 4.01(c)(i).

"*Merger*" has the meaning set forth in the recitals of this Agreement.

"*Merger Agreement*" has the meaning set forth in the recitals of this Agreement.

"*Nonrecourse Deductions*" has the meaning set forth in Treasury Regulations Section 1.704-2(b)(1). The amount of Nonrecourse Deductions of the Company for a Fiscal Year equals the net increase, if any, in the amount of Company Minimum Gain of the Company during that fiscal year, determined according to the provisions of Treasury Regulations Section 1.704-2(c).

"*Officer*" means each Person designated as an officer of the Company by the Managing Member pursuant to and in accordance with the provisions of Section 3.04, subject to any resolutions of the Managing Member appointing such Person as an officer of the Company or relating to such appointment.

"Original Agreement" has the meaning set forth in the recitals of this Agreement.

"Original Members" means the Members of the Company as of immediately prior to the Effective Time (as defined in the Merger Agreement) of the Merger.

"Original Member Representative" means Ghost Media Group, LLC or such other Person as may be appointed from time to time by holders of a majority of Units held by Original Members who hold Units at the time of determination.

Unit.

"*Participating Employee Unit*" means any Class P Unit that is both (i) an Eligible Class P Unit and (ii) a Vested

"*Participating Unit*" means, with respect to any Distribution (or other allocation of proceeds) pursuant to Section 4.01(a) or Section 4.02 hereof, any Unit, other than (a) LTIP Units and (b) any Class P Unit that is not a Participating Employee Unit.

*"Partnership Representative"* has means any Person acting as "tax matters partner" or the "partnership representative" pursuant to Section 5.08.

"*Person*" means any individual, estate, corporation, partnership, limited partnership, limited liability company, limited company, joint venture, trust, unincorporated or governmental organization or any agency or political subdivision thereof.

"Primary Indemnification" has the meaning set forth in Section 10.02(a).

"*Proceeding*" has the meaning set forth in Section 10.02(a).

"Profits" and "Losses" means, for each Fiscal Year or other period, the taxable income or loss of the Company, or particular items thereof, determined in accordance with the accounting method used by the Company for U.S. federal income tax purposes with the following adjustments: (a) all items of income, gain, loss or deduction allocated pursuant to Section 5.05 shall not be taken into account in computing such taxable income or loss (but the amounts of items to be specially allocated pursuant to Section 5.05 shall be determined by applying rules analogous to those set forth in this definition of "Profits" and "Losses"); (b) any income of the Company that is exempt from U.S. federal income taxation and not otherwise taken into account in computing Profits and Losses shall be added to such taxable income or loss; (c) if the Carrying Value of any asset differs from its adjusted tax basis for U.S. federal income tax purposes, any gain or loss resulting from a disposition of such asset shall be calculated with reference to such Carrying Value; (d) upon an adjustment to the Carrying Value (other than an adjustment in respect of depreciation) of any asset, pursuant to the definition of Carrying Value, the amount of the adjustment shall be included as gain or loss in computing such taxable income or loss; (e) if the Carrying Value of any asset differs from its adjusted tax basis for U.S. federal income tax purposes, the amount of depreciation, amortization or cost recovery deductions with respect to such asset for purposes of determining Profits and Losses, if any, shall be an amount which bears the same ratio to such Carrying Value as the U.S. federal income tax depreciation, amortization or other cost recovery deductions bears to such adjusted tax basis (provided that if the U.S. federal income tax depreciation, amortization or other cost recovery deduction is zero, the Managing Member may use any reasonable method for purposes of determining depreciation, amortization or other cost recovery deductions in calculating Profits and Losses); (f) to the extent that an adjustment to the adjusted tax basis of any Company asset pursuant to Code Section 734(b) is required pursuant to Treasury Regulation Section 1.704-1(b)(2)(iv)(m)(4) to be taken into account in determining Capital Accounts as a result of a distribution other than in liquidation of a Member's interest in the Company, the amount of such adjustment shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases the basis of the asset) from the disposition of the asset and shall be taken into account for purposes of computing such taxable income or loss and (g) except for items in (a) above, any expenditures of the Company not deductible in computing taxable income or loss, not properly capitalizable and not otherwise taken into account in computing Profits and Losses pursuant to this definition shall be treated as deductible items.

"*Revised Partnership Audit Provisions*" means Code Sections 6221 through 6241, as in effect for taxable years of the Company beginning after December 31, 2017, together with any subsequent amendments thereto, Treasury Regulations promulgated thereunder, and published administrative interpretations thereof, and any comparable provisions of state or local tax law.

"Second A&R Operating Agreement" has the meaning set forth in the Recitals of this Agreement.

*"Securities Act"* means the U.S. Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

*"Service Provider"* means any Member (in his, her or its individual capacity) or other Person, who at the time in question, is employed by or providing services to the Managing Member, the Company or any of their respective Subsidiaries.

"*Similar Law*" means any law or regulation that could cause the underlying assets of the Company to be treated as assets of the Member by virtue of its limited liability company interest in the Company and thereby subject the Company and the Managing Member (or other persons responsible for the investment and operation of the Company's assets) to laws or regulations that are similar to the fiduciary responsibility or prohibited transaction provisions contained in Title I of ERISA or Section 4975 of the Code.

*"Sponsor Letter Agreement"* means that certain Sponsor Letter Agreement, dated December 10, 2020, by and between the Managing Member and the Company.

"*Subsidiary*" means, with respect to any Person, any corporation, limited liability company, partnership, association or business entity of which (i) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (ii) if a limited liability company, partnership, association or other business entity (other than a corporation), a majority of partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a limited liability company, partnership, association or other business entity (other than a corporation) if such Person or Persons shall be allocated a majority of limited liability company, partnership, association or other business entity gains or losses or shall be or control any managing director or general partner of such limited liability company, partnership, association or other business entity. For purposes hereof, references to a "Subsidiary" of the Company shall be given effect only at such times that the Company has one or more Subsidiaries, and, unless otherwise indicated, the term "Subsidiary" refers to a Subsidiary of the Company.

"Tax Advances" has the meaning set forth in Section 5.07.

"Tax Amount" has the meaning set forth in Section 4.01(c)(ii).

"*Tax Distributions*" has the meaning set forth in Section 4.01(c)(ii).

"*Tax Estimation Period*" shall mean each period from January 1 through March 31, from April 1 through May 31, from June 1 through August 31, and from September 1 through December 31 of each taxable year.

*"Tax Receivable Agreement"* means the Tax Receivable Agreement dated as of or about the date hereof among the Company, Managing Member and the other parties from time to time party thereto, as amended from time to time.

"**Termination Transaction**" means any direct or indirect Transfer of all or any portion of the Managing Member's interest in the Company in connection with, or the other occurrence of, (a) a merger, consolidation or other combination involving the Managing Member, on the one hand, and any other Person, on the other, (b) a sale, lease, exchange or other transfer of all or substantially all of the assets of the Managing Member not in the ordinary course of its business, whether in a single transaction or a series of related transactions, (c) a reclassification, recapitalization or change of the outstanding shares of Class A Common Stock (other than a change in par value, or from par value to no par value, or as a result of a stock split, stock dividend or similar subdivision), (d) the adoption of any plan of liquidation or dissolution of the Managing Member, or (e) a direct or indirect Transfer of all or any portion of the Managing Member's interest in the Company, other than a Transfer effected in accordance with Section 3.07(a) or Section 3.07(b).

"*Third A&R Operating Agreement*" has the meaning set forth in the Recitals of this Agreement.

"*Transfer*" means, in respect of any Unit, property or other asset, any sale, assignment, transfer, distribution, exchange, mortgage, pledge, hypothecation or other disposition thereof, whether voluntarily or by operation of Law, directly or indirectly, in whole or in part, including, without limitation, the exchange of any Unit for any other security or the entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Unit, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise. The term "Transferred" shall have a meaning correlative to the foregoing.

"*Transferee*" means any Person that is a permitted transferee of a Member's interest in the Company, or part thereof.

"*Treasury Regulations*" means the income tax regulations, including temporary and proposed regulations, promulgated under the Code, as such regulations may be amended from time to time (including corresponding provisions of succeeding regulations).

"Units" means the Class A Units, Class P Units, and any other Class of Units that is established in accordance with this Agreement, which shall constitute limited liability company interests in the Company as provided in this Agreement and under the Act, entitling the holders thereof to the relative rights, title and interests in the profits, losses, deductions and credits of the Company at any particular time as set forth in this Agreement, and any and all other benefits to which a holder thereof may be entitled as a Member as provided in this Agreement, together with the obligations of such Member to comply with all terms and provisions of this Agreement.

"Unvested LTIP Units" has the meaning set forth in Section 2 of Schedule II hereto.

"*Unvested Units*" means those Units listed as unvested Units in the books and records of the Company, as the same may be amended from time to time in accordance with this Agreement.

"Vested LTIP Units" has the meaning set forth in Section 2 of Schedule II hereto.

"*Vested Units*" means those Units listed as vested Units in the books and records of the Company, as the same may be amended from time to time in accordance with this Agreement.

"Vesting Agreement" has the meaning set forth in Schedule II.

### **ARTICLE II**

### FORMATION, TERM, PURPOSE AND POWERS

**2.01** Formation. The Company was formed as a limited liability company under the provisions of the Act by the filing of the Certificate on September 4, 2014. If requested by the Managing Member, the Members shall promptly execute all certificates and other documents consistent with the terms of this Agreement necessary for the Managing Member to accomplish all filing, recording, publishing and other acts as may be appropriate to comply with all requirements for (a) the formation and operation of a limited liability company under the laws of the State of Delaware, (b) if the Managing Member in its sole discretion deems it advisable, the operation of the Company as a limited liability company, or entity in which the Members have limited liability, in all jurisdictions where the Company proposes to operate and (c) all other filings required to be made by the Company. The rights, powers, duties, obligations and liabilities of the Members shall be determined pursuant to the Act and this Agreement. To the extent that the rights, powers, duties, obligations and liabilities of any Member are different by reason of any provision of this Agreement than they would be in the absence of such provision, this Agreement shall, to the extent permitted by the Act, control. The execution, delivery and filing of the Certificate and each amendment thereto is hereby ratified, approved and confirmed by the Members.

**2.02 Name**. The name of the Company shall be, and the business of the Company shall be conducted under the name of "WM Holding Company, LLC" and all Company business shall be conducted in that name or in such other names that comply with applicable law as the Managing Member in its sole discretion may select from time to time. Subject to the Act, the Managing Member in its sole discretion may change the name of the Company (and amend this Agreement to reflect such change) at any time and from time to time without the consent of any other Person. Prompt notification of any such change shall be given to all Members.

**2.03 Term**. The term of the Company commenced on the date of the filing of the Certificate, and the term shall continue until the dissolution of the Company in accordance with Article IX. The existence of the Company shall continue until cancellation of the Certificate in the manner required by the Act.

**2.04 Offices.** The Company may have offices at such places either within or outside the State of Delaware as the Managing Member from time to time may select in its sole discretion. As of the date hereof, the principal place of business and office of the Company is located at 41 Discovery, Irvine, CA 92618.

### 2.05 Agent for Service of Process; Existence and Good Standing; Foreign Qualification.

(a) The registered office of the Company in the State of Delaware shall be located at c/o Registered Agent Solutions, Inc., 1679 S. Dupont Highway, Suite 100, Dover, Delaware 19901. The name of the registered agent of the Company for service of process on the Company in the State of Delaware at such address shall be Registered Agent Solutions, Inc.

(b) The Managing Member in its sole discretion may take all action which may be necessary or appropriate (i) for the continuation of the Company's valid existence as a limited liability company under the laws of the State of Delaware (and of each other jurisdiction in which such existence is necessary to enable the Company to conduct the business in which it is engaged) and (ii) for the maintenance, preservation and operation of the business of the Company in accordance with the provisions of this Agreement and applicable laws and regulations. The Managing Member in its sole discretion may file or cause to be filed for recordation in the proper office or offices in each other jurisdiction in which the Company is formed or qualified, such certificates (including certificates of formation and fictitious name certificates) and other documents as are required by the applicable statutes, rules or regulations of any such jurisdiction or as are required to reflect the identity of the Members. The Managing Member in its sole discretion may cause the Company to comply, to the extent procedures are available and those matters are reasonably within the control of the Officers, with all requirements necessary to qualify the Company to do business in any jurisdiction other than the State of Delaware.

**2.06 Business Purpose**. The Company was formed for the object and purpose of, and the nature and character of the business to be conducted by the Company is, engaging in any lawful act or activity for which limited liability companies may be formed under the Act.

**2.07 Powers of the Company**. Subject to the limitations set forth in this Agreement, the Company will possess and may exercise all of the powers and privileges granted to it by the Act including, without limitation, the ownership and operation of the assets and other property contributed to the Company by the Members, by any other Law or this Agreement, together with all powers incidental thereto, so far as such powers are necessary or convenient to the conduct, promotion or attainment of the purpose of the Company set forth in Section 2.06.

**2.08 Members; Reclassification; Admission of New Members**. Each of the Persons listed on Schedule I hereto, as the same may be amended from time to time in accordance with this Agreement, by virtue of its execution this Agreement, are admitted as Members of the Company. The rights, duties and liabilities of the Members shall be as provided in the Act, except as is otherwise expressly provided herein, and the Members consent to the variation of such rights, duties and liabilities as provided herein. Subject to Section 8.07 with respect to substitute Members, a Person may be admitted from time to time as a new Member with the written consent of the Managing Member in its sole discretion. Each new Member shall execute and deliver to the Managing Member an appropriate supplement to this Agreement pursuant to which the new Member agrees to be bound by the terms and conditions of this Agreement, as it may be amended from time to time. A new Managing Member or substitute Managing Member may be admitted to the Company solely in accordance with Section 8.06 or Section 9.02(e) hereof.

**2.09 Resignation**. No Member shall have the right to resign as a member of the Company other than following the Transfer of all Units owned by such Member in accordance with Article VIII.

**2.10 Representations of Members**. Each Member severally (and not jointly) represents and warrants to the Company and each other Member as of the date of such Member's admittance to the Company and as of each subsequent date that such Member acquires any additional Units that:

## (a) Organization; Authority

(i) To the extent it is not a natural person, (x) it is duly formed, validly existing and in good standing (if applicable) under the Laws of the jurisdiction of its formation, and if required by Law is duly qualified to conduct business and is in good standing in the jurisdiction of its principal place of business (if not formed in such jurisdiction), and (y) has full corporate, limited liability company, partnership, trust or other applicable power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement and all necessary actions by the board of directors, shareholders, managers, members, partners, trustees, beneficiaries or other Persons necessary for the due authorization, execution, delivery and performance of this Agreement by that Member have been duly taken.

(ii) It has duly executed and delivered this Agreement, and this Agreement is enforceable against such Member in accordance with its terms, subject to bankruptcy, moratorium, insolvency and other Laws generally affecting creditors' rights and general principles of equity (whether applied in a proceeding in a court of law or equity).

(b) Non-Contravention. Its authorization, execution, delivery, and performance of this Agreement does not breach or conflict with or constitute a default under (x) such Member's charter or other governing documents to the extent it is not a natural person, (y) any material obligation under any other material agreement to which that Member is a party or by which it is bound or (z) applicable Law.

(c) **Due Inquiry**. It has had, prior to the execution and delivery of this Agreement, the opportunity to ask questions of and receive answers from representatives of the Company concerning an investment in the Company, as well as the finances, operations, business and prospects of the Company, and the opportunity to obtain additional information to verify the accuracy of all information so obtained, and received all such information about the Company and the Units as it has requested.

(d) **Purpose of Investment**. It is acquiring and holding its Units solely for investment purposes, for its own account and not for the account or benefit of any other Person and not with a view towards the distribution or dissemination thereof, did not decide to enter into this Agreement as a result of any general solicitation or general advertising within the meaning of Rule 502 of Regulation D under the Securities Act, and acknowledges and understands that no United States federal or state agency has passed upon or made any recommendation or endorsement of the offering of any Units.

(e) Transfer Restrictions. It understands the Units are being Transferred in a transaction not involving a public offering within the meaning of the Securities Act and the Units will comprise "*restricted securities*" within the meaning of Rule 144(a)(3) under the Securities Act which shall not be sold, pledged, hypothecated or otherwise Transferred except in accordance with the terms of this Agreement and applicable Law. It agrees that, if in the future it decides to offer, resell, pledge or otherwise Transfer any portion of its Units, such Units may be offered, resold, pledged or otherwise Transferred only pursuant to an effective registration statement under the Securities Act or an applicable exemption from registration and/or qualification under the Securities Act and applicable state securities Laws, and as a condition precedent to any such Transfer, it may be required to deliver to the Company an opinion of counsel satisfactory to the Company, and agrees, absent registration or an exemption with respect to its Units, not to resell any such Units.

(f) Investor Status. It (i) has adequate means of providing for its current needs and possible contingencies, is able to bear the economic risks of its investment for an indefinite period of time and has a sufficient net worth to sustain a loss of its entire investment in the Company in the event such loss should occur, (ii) is sophisticated in financial matters and has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of an investment in the Company, (iii) is, or is controlled by, an "*accredited investor*," as that term is defined in Rule 501(a) of Regulation D, promulgated under the Securities Act, and acknowledges the issuance of Units under this Agreement is being made in reliance on a private placement exemption to "*accredited investors*" within the meaning of Section 501(a) of Regulation D under the Securities Act or similar exemptions under federal and state Law, and (iv) is treated as a single partner within the meaning of Treasury Regulations Section 1.7704-1(h) (determined taking into account the rules of Treasury Regulations Section 1.7704-1(h) (3)).

#### ARTICLE III

### MANAGEMENT

## 3.01 Managing Member

(a) The business, property and affairs of the Company shall be managed under the sole, absolute and exclusive direction of the Managing Member, which may from time to time delegate authority to Officers or to others to act on behalf of the Company.

(b) Without limiting the foregoing provisions of this Section 3.01, the Managing Member shall have the general power to manage or cause the management of the Company (which may be delegated to Officers of the Company), including, without limitation, the following powers:

(i) to develop and prepare a business plan each year which will set forth the operating goals and plans for the Company;

(ii) to execute and deliver or to authorize the execution and delivery of contracts, deeds, leases, licenses, instruments of transfer and other documents on behalf of the Company;

(iii) to make any expenditures, to lend or borrow money, to assume or guarantee, or otherwise contract for, indebtedness and other liabilities, to issue evidences of indebtedness and to incur any other obligations;

(iv) to establish and enforce limits of authority and internal controls with respect to all personnel and functions;

(v) to engage attorneys, consultants and accountants for the Company;

(vi) books of account: and

- (vii) to do all such other acts as shall be authorized in this Agreement or by the Members in writing

to develop or cause to be developed accounting procedures for the maintenance of the Company's

from time to time.

**3.02 Compensation**. The Managing Member shall not be entitled to any compensation for services rendered to the Company in its capacity as Managing Member.

3.03 **Expenses**. The Company shall pay, or cause to be paid, all costs, fees, operating expenses and other expenses of the Company (including the costs, fees and expenses of attorneys, accountants or other professionals) incurred in pursuing and conducting, or otherwise related to, the activities of the Company. The Company shall also, in the sole discretion of the Managing Member, bear and/or reimburse the Managing Member for (i) any costs, fees or expenses incurred by the Managing Member in connection with serving as the Managing Member and (ii) all other expenses allocable to the Company or otherwise incurred by the Managing Member in connection with operating the Company's business (including expenses allocated to the Managing Member by its Affiliates). To the extent that the Managing Member determines in its sole discretion that such expenses are related to the business and affairs of the Managing Member that are conducted through the Company and/or its subsidiaries (including expenses that relate to the business and affairs of the Company and/or its subsidiaries and that also relate to other activities of the Managing Member), the Managing Member may cause the Company to pay or bear all expenses of the Managing Member, including, without limitation, compensation and meeting costs of any board of directors or similar body of the Managing Member, any salary, bonus, incentive compensation and other amounts paid to any Person including Affiliates of the Managing Member to perform services for the Company, litigation costs and damages arising from litigation, accounting and legal costs and franchise taxes, provided that the Company shall not pay or bear any income tax obligations of the Managing Member or any obligations of the Managing Member under the Tax Receivable Agreement. Reimbursements pursuant to this Section 3.03 shall be in addition to any reimbursement to the Managing Member as a result of indemnification pursuant to Section 10.02.

**3.04 Officers**. Subject to the direction and oversight of the Managing Member, the day-to-day administration of the business of the Company may be carried out by persons who may be designated as officers by the Managing Member, with titles including but not limited to "assistant secretary," "assistant treasurer," "chairman," "chief executive officer," "chief financial officer," "chief operating officer," "director," "general counsel," "general manager," "managing director," "president," "principal accounting officer," "secretary," "senior chairman," "senior managing director," "treasurer," "vice chairman," "executive vice president" or "vice president," and as and to the extent authorized by the Managing Member in its sole discretion. The officers of the Company shall have such titles and powers and perform such duties as shall be determined from time to time by the Managing Member and otherwise as shall customarily pertain to such offices. Any number of offices may be held by the same person. In its sole discretion, the Managing Member may choose not to fill any office for any period as it may deem advisable. All officers and other persons providing services to or for the benefit of the Company shall be subject to the supervision and direction of the Managing Member and may be removed, with or without cause, from such office by the Managing Member and the authority, duties or responsibilities of any employee, agent or officer of the Company may be suspended by the Managing Member from time to time, in each case in the sole discretion of the Managing Member. The Managing Member shall not cease to be a Managing Member of the Company as a result of the delegation of any duties hereunder. No officer of the Company, in its capacity as such, shall be considered a Managing Member of the Company by agreement, as a result of the performance of its duties hereunder or otherwise.

3.05 Authority of Members. No Member (other than the Managing Member), in its capacity as such, shall participate in or have any control over the business of the Company. Except as expressly provided herein, the Units do not confer any rights upon the Members to participate in the affairs of the Company described in this Agreement. Except as expressly provided herein, no Member (other than the Managing Member) shall have any right to vote on any matter involving the Company, including with respect to any merger, consolidation, combination or conversion of the Company, or any other matter that a Member might otherwise have the ability to vote on or consent with respect to under the Act, at law, in equity or otherwise. The conduct, control and management of the Company shall be vested exclusively in the Managing Member. In all matters relating to or arising out of the conduct of the operation of the Company, the decision of the Managing Member shall be the decision of the Company. Except as required or permitted by Law, or expressly provided in the ultimate sentence of this Section 3.05 or by separate agreement with the Company, no Member who is not also the Managing Member (and acting in such capacity) shall take any part in the management or control of the operation or business of the Company in its capacity as a Member, nor shall any Member who is not also the Managing Member (and acting in such capacity) have any right, authority or power to act for or on behalf of or bind the Company in his or its capacity as a Member in any respect or assume any obligation or responsibility of the Company or of any other Member. Notwithstanding the foregoing, the Company may from time to time appoint one or more Members as officers or employ one or more Members as employees, and such Members, in their capacity as officers or employees of the Company (and not, for clarity, in their capacity as Members of the Company), may take part in the control and management of the business of the Company to the extent such authority and power to act for or on behalf of the Company has been delegated to them by the Managing Member.

**3.06** Action by Written Consent or Ratification. Any action required or permitted to be taken by the Members pursuant to this Agreement shall be taken if all Members whose consent or ratification is required consent thereto or provide a consent or ratification in writing. Any action required, required to be approved or permitted to be taken by the Managing Member pursuant to this Agreement may be taken or approved, as applicable, by the Managing Member acting pursuant to a writing which evidences its approval of or consent to such action.

**3.07 Restrictions on Termination Transactions**. The Managing Member shall not engage in, or cause or permit, a Termination Transaction, unless either (x) the Termination Transaction has been approved by Members holding a majority of the Class A Units held by all Members (excluding the Managing Member and any Members controlled by the Managing Member) or (y) the following conditions are satisfied:

in connection with any such Termination Transaction. (i) each holder of Class A Units and Class P Units (a) (other than the Managing Member and its wholly owned Subsidiaries) will receive, or will have the right to elect to receive, for each Class A Unit or Class P Unit an amount of cash, securities or other property equal to the product of (x) the number of shares of Class A Common Stock into which a Class A Unit or Class P Unit is then exchangeable pursuant to the Exchange Agreement and (y) the greatest amount of cash, securities or other property paid to a holder of one share of Class A Common Stock in consideration of one share of Class A Common Stock pursuant to the terms of such Termination Transaction; provided, that the condition set forth in this Section 3.07(a)(i) shall be deemed to have been satisfied if, in connection with such Termination Transaction, a purchase, tender or exchange offer shall have been made to and accepted by the holders of a majority of the outstanding shares of Class A Common Stock, and each holder of Class A Units or Class P Units (other than the Managing Member and its wholly owned subsidiaries) will receive, or will have the right to elect to receive, the greatest amount of cash, securities or other property which such holder of Class A Units or Class P Units would have received had such Class A Units or Class P Units been exchanged for shares of Class A Common Stock in an Exchange Transaction immediately prior to the expiration of such purchase, tender or exchange offer, such holder of Class A Units or Class P Units had thereupon accepted such purchase, tender or exchange offer and then such Termination Transaction shall have been consummated (the fair market value, at the time of the Termination Transaction, of the amount specified herein with respect to each Class A Unit or Class P Unit is referred to as the "Transaction Consideration"); and (ii) the Company receives an opinion from nationally recognized tax counsel to the effect that such Termination Transaction will be tax-free to each holder of Class A Units and Class P Units (including the Managing Member and its wholly owned Subsidiaries unless waived by the Managing Member) for U.S. federal income tax purposes (except to the extent of cash, marketable securities or other property received); or

(b) all of the following conditions are met: (i) substantially all of the assets directly or indirectly owned by the Company prior to the announcement of the Termination Transaction are, immediately after the Termination Transaction, owned directly or indirectly by (x) the Company or (y) another limited liability company or limited partnership organized or existing under the laws of the United States, any state thereof, the District of Columbia, or any territory thereof, which is the survivor of a merger, consolidation or combination of assets with the Company (in each case, the "Surviving Company"); (ii) the Surviving Company is classified as a partnership for U.S. Federal income tax purposes; (iii) the Members (other than entities controlled by the Managing Member) that held Class A Units and Class P Units immediately prior to the consummation of such Termination Transaction own a percentage interest of the Surviving Company based on the proportion of the relative fair market value of the net assets of the Company to the other net assets of the Surviving Company immediately prior to the consummation of such transaction; (iv) the rights of such Members with respect to the Surviving Company are at least as favorable as those of Members holding Class A Units and Class P Units (including any rights under the Tax Receivable Agreement, unless such Termination Transaction constitutes a "Change of Control" for purposes of the Tax Receivable Agreement or otherwise results in payments of cash to the TRA Parties (as defined in the Tax Receivable Agreement) equivalent to (and in lieu of) the payments that would be required to be made to such TRA Parties if such Termination Transaction did constitute a "Change of Control" for such purposes) immediately prior to the consummation of such transaction (except to the extent that any such rights are consistent with clause (v) below) and as those applicable to any other Members (not including the Managing Member); and (v) such rights include the right, to the same extent provided to holders of Class A Units and Class P Units pursuant to the Exchange Agreement, to exchange their interests in the Surviving Company for: (1) a number of such publicly traded common equity securities with a fair market value, as of the date of consummation of such Termination Transaction, equal to the Transaction Consideration, subject to antidilution adjustments comparable to those set forth in Section 2.2 of the Exchange Agreement (the "Successor Shares Amount"); and/or (2) cash in an amount equal to the fair market value of the Successor Shares Amount at the time of such exchange, determined in a manner consistent with the definition of "Cash Exchange Payment" as set forth in the Exchange Agreement.

(c) In connection with any Termination Transaction permitted by Section 3.07(b) hereof, the relative fair market values shall be reasonably determined by the Managing Member as of the time of such transaction and, to the extent applicable, shall be no less favorable to the Members than the relative values reflected in the terms of such transaction.

### **ARTICLE IV**

### DISTRIBUTIONS

#### 4.01 Distributions

(a) **Operating Distributions.** The Managing Member, in its sole discretion, may authorize distributions (to the extent of Available Cash) by the Company to the Members at any time and from time to time. Subject to Section 4.01(b) with respect to Class P Units and to Section 4.01(c) with respect to Tax Distributions, all Distributions by the Company other than those made in connection with dissolution of the Company pursuant to Section 4.02, shall be made or allocated to holders of Participating Units and LTIP Units pro rata based on the number of Participating Units and/or LTIP Units held by each such holder; *provided* that any distributions in respect of Unvested LTIP Units shall be held back and shall be payable at the same time as the underlying LTIP Units become Vested LTIP Units, and if such LTIP Units are forfeited, the former holder of such LTIP Units shall have no right to receive such distributions. For the avoidance of doubt, if a Distribution in respect of an Unvested LTIP Unit is held back pursuant to this Section 4.01(a), the LTIP Series Sub-Account in respect of such Unvested LTIP Unit shall be treated as reduced pursuant to and for purposes of applying Section 5.03(b).

(b) Class P Units. For the avoidance of doubt, if the amount to be distributed pursuant to Section 4.01(a) and Section 4.02 with respect to any particular Distribution would cause the amount of any outstanding Class P Unit's Participation Threshold to be reduced to zero, then such Class P Unit shall constitute an Eligible Class P Unit for purposes of Section 4.01(a) and Section 4.02 only after the portion of the amount to be distributed in such Distribution that would cause such Class P Unit's Participating Units (taking into account outstanding Class P Units that have lesser Participation Thresholds (determined immediately prior to such Distribution)). For the avoidance of doubt, if any Class P Unit is an Unvested Unit as of the date of any Distribution, such Unvested Unit shall not participate in such Distribution (but such Distribution may reduce the Participation Threshold of such Unvested Unit).

(c) Tax Distributions (i) With respect to each Member the Company shall calculate the excess of (x)(A) the Income Amount allocated or allocable to such Member for the Tax Estimation Period in question and for all preceding Tax Estimation Periods, if any, within the taxable year containing such Tax Estimation Period multiplied by (B) the Assumed Tax Rate over (y) the aggregate amount of all prior Tax Distributions in respect of such taxable year and any Distributions made to such Member pursuant to Section 4.01(a) and Section 4.02, with respect to the Tax Estimation Period in question and any previous Tax Estimation Period falling in the taxable year containing the applicable Tax Estimation Period referred to in (x)(A) (the amount so calculated pursuant to this sentence is herein referred to as a "*Member's Required Tax Distribution*"); *provided*, however, that the Managing Member may make adjustments in its reasonable discretion to reflect transactions occurring during the taxable year. For purposes of this Agreement, the "*Income Amount*" for a Tax Estimation Period shall equal, with respect to any Member, the net taxable income of the Company allocated or allocable to such Member for such Tax Estimation Period (excluding any compensation paid to a Member outside of this Agreement). For purposes of computing the Tax Amount, the net taxable income shall be determined without regard to any special adjustments of tax items required as a result of any election under Section 754 of the Code, including adjustments required by Sections 734 and 743 of the Code.

(ii) At least five (5) days before the quarterly due date for payment by corporations or individuals (whichever is earlier) on a calendar year under the Code, the Company shall distribute (to the extent of Available Cash) to the Members *pro rata* based upon the number of Units held by each such Member, an aggregate amount of cash sufficient to provide each such Member with a distribution at least equal to such other Member's Required Tax Distribution (*provided* that notwithstanding the foregoing, the Members shall only receive distributions in respect of their Class P Units or LTIP Units to the extent of their Member's Required Tax Distribution for such period in respect of such Class P Units or LTIP Units (i.e., which may not result in a *pro rata* distribution in respect of their Class P Units or the LTIP Units, as applicable), and shall not receive any amount in excess of such amount in respect of their Class P Units or LTIP Units, distributed pursuant to this Section 4.01(c), "*Tax Distributions*"). Any Tax Distributions shall be treated in all respects as advances against future distributions pursuant to Section 4.01(a) and Section 4.02; *provided* that, any Tax Distributions made with respect to Class P Units or LTIP Units which subsequently convert into Class A Units pursuant to Section 5.03(b) shall be treated in all respects as advances against any such future distributions made with respect to such Class A Units.

(iii) Notwithstanding anything to the contrary herein, no Tax Distributions will be required to be made with respect to items arising with respect to any Covered Transaction, although any unpaid Tax Distributions with respect to any Tax Estimation Period, or portion thereof, ending before a Covered Transaction shall continue to be required to be paid prior to any Distributions being made under Section 4.01(a) and Section 4.02.

**4.02 Liquidation Distribution**. Subject to Section 4.01(b) with respect to Class P Units and Section 4.01(c) with respect to Tax Distributions, all Distributions by the Company, and all proceeds (whether received by the Company or directly by the Members) in connection with dissolution of the Company shall be made or allocated among the holders of Participating Units and Vested LTIP Units *pro rata* based on the number of Participating Units and Vested LTIP Units held by each such holder.

**4.03 Limitations on Distribution**. Notwithstanding any provision to the contrary contained in this Agreement, the Managing Member shall not make a distribution to any Member if such distribution would (i) violate Section 18-607 of the Act or other applicable Law, or (ii) in the case of LTIP Units, to the extent such distribution or payment would cause the balance of a Member's LTIP Series Sub-Account in respect of such LTIP Series Units to be less than zero.

**4.04 Earnout Company Units**. Notwithstanding anything in this Article IV to the contrary, for all purposes of this Article IV, any distributions that would be made to the Managing Member pursuant to this Article IV in respect of the Class A Units held by the Managing Member that are Earnout Company Units and that have not satisfied the earnout criteria applicable to the shares of Class A Common Stock set forth in the Sponsor Letter Agreement at the time such distribution is made shall be held back and recorded by the Company and such amounts shall either be (i) released to the Managing Member at such time (if any) as such Earnout Company Units satisfy the earnout criteria set forth in the Sponsor Letter Agreement or (ii) released to the Company at such time as such Earnout Company Units are forfeited to the Company in accordance with the Sponsor Letter Agreement.

**4.05 Use of Distribution Funds**. The Managing Member shall use distributions received from the Company for payment of taxes, obligations under the Tax Receivable Agreement, liabilities or expenses, to loan funds to the Company in accordance with this Agreement, for the payment of dividends to its shareholders or for other general corporate purposes as determined in the sole discretion of the Managing Member; provided that the Managing Member may not use such distributions to acquire any Units, except as otherwise provided in <u>Section 7.04</u>.

### ARTICLE V

### CAPITAL CONTRIBUTIONS; CAPITAL ACCOUNTS; TAX ALLOCATIONS; TAX MATTERS

**5.01 Initial Capital Contributions**. The Members have made, on or prior to the date hereof, Capital Contributions and, in exchange, the Company has issued to the Members the number of Class A Units as specified in the books and records of the Company.

**5.02** No Additional Capital Contributions. No Member shall be required to make additional Capital Contributions to the Company without the consent of such Member or permitted to make additional capital contributions to the Company without the consent of the Managing Member, which may be granted or withheld in its sole discretion.

#### 5.03 Capital Accounts.

(a) A separate capital account (a "*Capital Account*") shall be established and maintained for each Member in accordance with the provisions of Treasury Regulations Section 1.704-1(b)(2)(iv) and this Section 5.03. The Company may adjust the Capital Accounts of its Members to reflect revaluations of the property of any Subsidiary of the Company that is treated as a partnership (or entity disregarded from a partnership) for U.S. federal income tax purposes. The Capital Account of each Member shall be credited with such Member's Capital Contributions, if any, all Profits allocated to such Member pursuant to Section 5.04 and any items of income or gain which are specially allocated pursuant to Section 5.05; and shall be debited with all Losses allocated to such Member pursuant to Section 5.04, any items of loss or deduction of the Company specially allocated to such Member pursuant to Section 5.05, and all cash and the Carrying Value of any property (net of liabilities assumed by such Member and the liabilities to which such property is subject) distributed by the Company to such Member. Any references in any section of this Agreement to the Capital Account of a Member shall be deemed to refer to such Capital Account as the same may be credited or debited from time to time as set forth above. In the event of any transfer of any interest in the Company in accordance with the terms of this Agreement, the Transferee shall succeed to the Capital Account of the transferor to the extent it relates to the transferred interest.

(b) A separate sub-account (an "*LTIP Series Sub-Account*") shall be established and maintained for each Member in respect of each LTIP Series Unit held by such Member. The balance of each LTIP Series Sub-Account shall initially be zero and shall be adjusted as provided in the previous paragraph as if the LTIP Series Sub-Account was a Capital Account and the Member only held the LTIP Series Units of such LTIP Series held by such Member. If at any time the aggregate LTIP Series Sub-Accounts of an LTIP Series equal the product of the number of LTIP Series Units in such LTIP Series and the Class A Unit Capital Account Amount (as determined at such time), the LTIP Series Units of such LTIP Series shall be converted automatically into (i) a separate sub-class of LTIP Series Units (*"Equitized LTIP Series Units*"), if such LTIP Series Units are Unvested LTIP Units, or (ii) Class A Units, if such LTIP Series Units are Vested LTIP Units. LTIP Series Sub-Accounts shall continue to be maintained for Equitized LTIP Series Units. If an Equitized LTIP Series Unit shall be converted automatically into a Class A Unit once the aggregate LTIP Series Sub-Accounts for the Equitized LTIP Series to which such Equitized LTIP Series Unit belongs equal the product of the number of the LTIP Units in such Equitized LTIP Series and the Class A Unit Capital Account Amount (as determined at such time). Upon the automatic conversion of a Vested LTIP Unit into a Class A Unit pursuant to this Section 5.03, the Managing Member shall issue one share of Class V Common Stock per each such converted Class A Unit to the holder thereof.

**5.04 Allocations of Profits and Losses.** Except as otherwise provided in this Agreement, Profits and Losses (and, to the extent necessary, individual items of income, gain or loss or deduction of the Company) shall be allocated in a manner such that the Capital Account of each Member after giving effect to the special allocations set forth in Section 5.05 is, as nearly as possible, equal (proportionately) to (i) the distributions that would be made pursuant to Article IX if the Company were dissolved, its affairs wound up and its assets sold for cash equal to their Carrying Value, all Company liabilities were satisfied in cash in accordance with their terms (limited with respect to each nonrecourse liability to the Carrying Value of the assets securing such liability) and all remaining or resulting cash was distributed to the Members in accordance with Section 9.03, minus (ii) such Member's share of Company Minimum Gain and Member Nonrecourse Debt Minimum Gain, computed immediately prior to the hypothetical sale of assets. For purposes of this Article V, (i) no Profits or Losses or items thereof will be allocated in respect of any LTIP Unit pursuant to this Section 5.04 in excess of the amount allocated to a Class A Unit under this Section 5.04, and (ii) each Unvested Unit shall be treated as a Vested Unit. Notwithstanding the foregoing, the Managing Member shall make such adjustments to Capital Accounts as it determines in its reasonable discretion to be appropriate to ensure allocations are made in accordance with a Member's interest in the Company.

**5.05 Special Allocations**. Notwithstanding any other provision in this Article V:

(a) Minimum Gain Chargeback. If there is a net decrease in Company Minimum Gain or Member Nonrecourse Debt Minimum Gain (determined in accordance with the principles of Treasury Regulations Sections 1.704-2(d) and 1.704-2(i)) during any Company taxable year, the Members shall be specially allocated items of Company income and gain for such year (and, if necessary, subsequent years) in an amount equal to their respective shares of such net decrease during such year, determined pursuant to Treasury Regulations Section 1.704-2(i)(4). The items to be so allocated shall be determined in accordance with Treasury Regulations Sections 1.704-2(i)(4) and 1.704-2(j)(2). This Section 5.05(a) is intended to comply with the minimum gain chargeback requirements in such Treasury Regulations Sections and shall be interpreted consistently therewith; including that no chargeback shall be required to the extent of the exceptions provided in Treasury Regulations Sections 1.704-2(i)(4).

(b) Qualified Income Offset. If any Member unexpectedly receives any adjustments, allocations, or distributions described in Treasury Regulations Section 1.704-1(b)(2)(ii)(d)(4), (5) or (6), items of Company income and gain shall be specially allocated to such Member in an amount and manner sufficient to eliminate the deficit balance in such Member's Adjusted Capital Account Balance created by such adjustments, allocations or distributions as promptly as possible; provided that an allocation pursuant to this Section 5.05(b) shall be made only to the extent that a Member would have a deficit Adjusted Capital Account Balance in excess of such sum after all other allocations provided for in this Article V have been tentatively made as if this Section 5.05(b) were not in this Agreement. This Section 5.05(b) is intended to comply with the "qualified income offset" requirement of the Code and shall be interpreted consistently therewith.

(c) Gross Income Allocation. If any Member has a deficit Capital Account at the end of any Fiscal Year which is in excess of the sum of (i) the amount such Member is obligated to restore, if any, pursuant to any provision of this Agreement, and (ii) the amount such Member is deemed to be obligated to restore pursuant to the penultimate sentences of Treasury Regulations Section 1.704-2(g)(1) and 1.704-2(i)(5), each such Member shall be specially allocated items of Company income and gain in the amount of such excess as quickly as possible; provided that an allocation pursuant to this Section 5.05(c) shall be made only if and to the extent that a Member would have a deficit Capital Account in excess of such sum after all other allocations provided for in this Article V have been tentatively made as if Section 5.05(b) and this Section 5.05(c) were not in this Agreement.

(d) LTIP Unit Gain Allocation. Prior to making any allocations pursuant to Section 5.04 for an applicable period, gain recognized on the sale of all or substantially all of the Company's assets and any Mark-to-Market Gain shall be allocated to the Capital Accounts and LTIP Series Sub-Accounts of the Members in a manner such that, to the extent possible, each LTIP Series converts to Equitized LTIP Series Units or Class A Units pursuant to Section 5.03(b), subject to the following principles as interpreted and applied by the Managing Member in good faith:

(i) To the extent such gain is insufficient to cause all LTIP Units to convert to Equitized LTIP Series Units or Class A Units, gain shall be allocated with respect to each LTIP Series (other than any Equitized LTIP Series) based on the order in which each such LTIP Series was issued beginning with the LTIP Series that has been outstanding the longest.

(ii) The provisions of this Agreement, including this Section, are intended to ensure that holders of LTIP Units receive "profits interests" within the meaning of Revenue Procedure 93-27, 1993-2 C.B. 343 and 2001-43, 2001-2 C.B. 191. In this regard, it is the intention of the parties to this Agreement that any allocation of gain to an LTIP Series Unit (other than an Equitized LTIP Series Unit) be limited to gain that is economically accrued after the date such LTIP Series Unit is issued ("Available Gains"). If the Managing Member subsequently determines that an allocation of gain other than Available Gains was made to an LTIP Unit (other than an Equitized LTIP Series Unit) or that its determination of the aggregate value of the Capital Accounts was otherwise incorrect, it may adjust the values of the aggregate Capital Accounts or other values (and make correlative changes to the allocations previously made and to the Capital Accounts of the Members) or distributions made pursuant to this Agreement to ensure that the intended treatment applies; *provided* that such adjustments shall be made, to the maximum extent possible, in a manner that does not adversely affect any Member holding Class A Units, with respect to such Units.

(e) Equitized LTIP Series Unit Loss Allocation. If the Equitized LTIP Series Sub-Account with respect to an Equitized LTIP Series Unit exceeds the Class A Unit Capital Account Amount, or would exceed the Class A Unit Capital Account Amount after giving effect to the allocations specified under Section 5.05(d) (for example, as a result of a distribution being made in respect of Class A Units under Section 4.01), a priority allocation of Losses (or items thereof) or other adjusting allocations shall be made to such Equitized LTIP Series Sub-Account in an amount necessary to eliminate such excess or, if there are insufficient Losses (or items thereof) to do so, to reduce such excess to the maximum extent possible.

(f) Nonrecourse Deductions. Nonrecourse Deductions shall be allocated to the Members holding Class A Units, Class P Units and Equitized LTIP Series Units in accordance with their respective Class A Percentage Interest. For the purpose of determining the Class A Percentage Interest in the foregoing sentence, all Equitized LTIP Series Units and Class P Units shall be treated as Class A Units.

(g) Member Nonrecourse Deductions. Member Nonrecourse Deductions for any taxable period shall be allocated to the Member who bears the economic risk of loss with respect to the liability to which such Member Nonrecourse Deductions are attributable in accordance with Treasury Regulations Section 1.704-2(i)(1).

(h) Section 754 Adjustments. To the extent an adjustment to the adjusted tax basis of any Company asset, pursuant to Code Section 734(b) or Section 743(b) is required, pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(m)(2) or Section 1.704-1(b)(2)(iv)(m)(4), to be taken into account in determining Capital Accounts as the result of a distribution to a Member in complete liquidation of such Member's interest in the Company, the amount of such adjustment to Capital Accounts shall be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases such basis) and such gain or loss shall be specially allocated to the Members in accordance with their interests in the Company in the event Treasury Regulations Section 1.704-1(b)(2)(iv)(m)(2) applies, or to the Member to whom such distribution was made in the event Treasury Regulations Section 1.704-1(b)(2)(iv)(m)(4) applies.

(i) Ameliorative Allocations. Any special allocations of income or gain pursuant to Sections 5.05(a), 5.05(b) or 5.05(c) hereof shall be taken into account in computing subsequent allocations pursuant to Section 5.04 and this Section 5.05(i), so that the net amount of any items so allocated and all other items allocated to each Member shall, to the extent possible, be equal to the net amount that would have been allocated to each Member if such allocations pursuant to Sections 5.05(a), 5.05(b) or 5.05(c) had not occurred.

(j) Allocations Relating to Taxable Issuance of Company Units. Any income, gain, loss, or deduction realized as a direct or indirect result of the issuance of Units by the Company to a Member (the "*Issuance Items*") shall be allocated among the Members so that, to the extent possible, the net amount of such Issuance Items, together with all other allocations under this Agreement to each Member shall be equal to the net amount that would have been allocated to each such Member if the Issuance Items had not been realized. The forfeiture allocations described in Proposed Regulations Section 1.704-1(b)(4)(xii)(C) (2005), and the allocations to which they relate, shall be treated as Issuance Items.

(k) **Forfeiture Allocation**. In the event that the Units of any Member are forfeited, then for the fiscal year of such forfeiture or other period (as determined by the Managing Member):

(i) items of income, gain, loss, and deduction shall be excluded from the calculation of Profits and Losses and shall be specially allocated to the Member whose Units have been forfeited so as to cause such Member's Capital Account to equal such Member's distribution entitlements under Section 4.1 after giving effect to the adjustment in the Member's Class A/LTIP Percentage Interest resulting from the applicable forfeiture;

(ii) the Managing Member may elect to apply another allocation or Capital Account adjustment method to a Unit forfeiture as it reasonably deems appropriate in lieu of the method set forth in this Section 5.05(k).

**5.06 Tax Allocations**. For income tax purposes, each item of income, gain, loss and deduction of the Company shall be allocated among the Members in the same manner as the corresponding items of Profits and Losses and specially allocated items are allocated for Capital Account purposes; *provided* that in the case of any asset the Carrying Value of which differs from its adjusted tax basis for U.S. federal income tax purposes, income, gain, loss and deduction with respect to such asset shall be allocated solely for income tax purposes in accordance with the principles of Sections 704(b) and (c) of the Code (in any manner determined by the Managing Member and permitted by the Code and Treasury Regulations, provided that the prior written consent of Original Member Representative (and its successors or assigns) shall be required for use of any method other than the traditional method (without curative allocations) described in Treasury Regulation Section 1.704-3(b)) so as to take account of the difference between Carrying Value and adjusted basis of such asset. Notwithstanding the foregoing, the Managing Member shall make such allocations for tax purposes as it determines in its reasonable discretion, subject to, for so long as the Original Members collectively own at least 10% of the Units, the prior written consent, not to be unreasonably withheld, conditioned or delayed, of the Original Member Representative (and its successors or assigns), to be appropriate to ensure allocations are made in accordance with a Member's interest in the Company.

5.07 Tax Advances. If the Company or any other Person in which the Company holds an interest is required by Law to withhold or to make tax payments on behalf of or with respect to any Member, or the Company is subjected to tax itself (including any amounts withheld from amounts directly or indirectly payable to the Company or to any other Person in which the Company holds an interest) by reason of the status of any Member as such or that is specifically attributable to a Member (including federal, state, local or foreign withholding, personal property, unincorporated business or other taxes, the amount of any taxes arising under the Revised Partnership Audit Provisions, the amount of any taxes imposed under Code Section 1446(f). and any interest, penalties, additions to tax, and expenses related to any such amounts) ("Tax Advances"), the Managing Member may cause the Company to withhold such amounts and cause the Company to make such tax payments as so required. All Tax Advances made on behalf of a Member shall be repaid by reducing the amount of the current or next succeeding distribution or distributions which would otherwise have been made to such Member or, if such distributions are not sufficient for that purpose, by so reducing the proceeds of liquidation otherwise payable to such Member. For all purposes of this Agreement such Member shall be treated as having received the amount of the distribution that is equal to the Tax Advance. Each Member hereby agrees to indemnify and hold harmless the Company and the other Members from and against any liability (including, without limitation, any liability for taxes, penalties, additions to tax or interest other than any penalties, additions to tax or interest imposed as a result of the Company's failure to withhold or make a tax payment on behalf of such Member which withholding or payment is required pursuant to applicable Law) with respect to income attributable to or distributions or other payments to such Member. For the avoidance of doubt, any income taxes, penalties, additions to tax and interest payable by the Company or any fiscally transparent entity in which the Company owns an interest shall be treated as specifically attributable to the Members and shall be allocated among the Members such that the burden of (or any diminution in distributable proceeds resulting from) any such amounts is borne by those Members to whom such amounts are specifically attributable (whether as a result of their status, actions, inactions or otherwise, including pursuant to an allocation made under Section 5.08), in each case as reasonably determined by the Managing Member. For the avoidance of doubt, any taxes, penalties, and interest payable under the Revised Partnership Audit Provisions by the Company or any fiscally transparent entity in which the Company owns an interest shall be treated as specifically attributable to the Members of the Company, and the Managing Member shall use commercially reasonable efforts to allocate the burden of (or any diminution in distributable proceeds resulting from) any such taxes, penalties or interest to those Members to whom such amounts are specifically attributable (whether as a result of their status, actions, inactions or otherwise), as reasonably determined by the Managing Member.

### 5.08 Partnership Representative.

(a) The Original Member Representative is hereby designated as the Company's "tax matters partner" for U.S. federal income tax purposes under Section 6231(a)(7) of the Code, as in effect for taxable years of the Company beginning on or before December 31, 2017, and as the Company's "partnership representative" as that term is defined in the Revised Partnership Audit Provisions for taxable years of the Company beginning after December 31, 2017 and ending prior to January 1, 2021. The Managing Member is hereby designated as the "partnership representative" as that term is defined in Revised Partnership Audit Provisions for taxable years of the Company beginning on or after January 1, 2021. In addition, the Managing Member is hereby authorized to designate or remove any other Person selected by the Managing Member as the Partnership Representative. For each Fiscal Year in which the Partnership Representative is an entity, the Company shall appoint an individual identified by the Partnership Representative for such Fiscal Year to act on its behalf (the "Designated Individual") in accordance with the applicable Regulations or analogous provisions of state or local Law. Each Member hereby expressly consents to such designations and agrees to take, and that the Managing Member is authorized to take (or cause the Company to take), such other actions as may be necessary or advisable pursuant to Treasury Regulations or other Internal Revenue Service or Treasury guidance or state or local Law to cause such designations or evidence such Member's consent to such designations.

(b) Subject to this Section 5.08, the Partnership Representative shall have the sole authority to act on behalf of the Company in connection with, make all relevant decisions regarding application of, and to exercise the rights and powers provided for in the Revised Partnership Audit Provisions, including making any elections under the Revised Partnership Audit Provisions or any decisions to settle, compromise, challenge, litigate or otherwise alter the defense of any action, audit or examination before the IRS or any other tax authority (each, an "*Audit*"), and to expend Company funds for professional services and other expenses reasonably incurred in connection therewith.

Without limiting the foregoing, the Partnership Representative shall give prompt written notice to the (c) Original Member Representative of the commencement of any Audit of the Company or any of its Subsidiaries the resolution of which would reasonably be expected to have a disproportionate (compared to the Managing Member) and material adverse effect on the Original Members (a "Specified Audit"). The Partnership Representative shall (i) keep the Original Member Representative reasonably informed of the material developments and status of any such Specified Audit, (ii) permit the Original Member Representative (or its designee) to participate (including using separate counsel), in each case at the Original Members' sole cost and expense, in any such Specified Audit, and (iii) promptly notify the Original Member Representative of receipt of a notice of a final partnership adjustment (or equivalent under applicable Laws) or a final decision of a court or IRS Independent Office of Appeals panel (or equivalent body under applicable Laws) with respect to such Specified Audit. The Partnership Representative or the Company shall promptly provide the Original Member Representative with copies of all material correspondence between the Partnership Representative or the Company (as applicable) and any governmental entity in connection with such Specified Audit and shall give the Original Member Representative a reasonable opportunity to review and comment on any material correspondence, submission (including settlement or compromise offers) or filing in connection with any such Specified Audit. Additionally, the Partnership Representative shall not (and the Company shall not (and shall not authorize the Partnership Representative to)) settle, compromise or abandon any Specified Audit in a manner that would reasonably be expected to have a disproportionate (compared to the Managing Member) and material adverse effect on the Original Members without the Original Member Representative's prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned). The Partnership Representative shall obtain the prior written consent of the Original Member Representative (which consent shall not be unreasonably withheld, delayed or conditioned) before (i) making an election under Section 6226(a) of the Code (or any analogous provision of state or local Law) or (ii) taking any material action under the Revised Partnership Audit Provisions that would reasonably be expected to have a disproportionate (compared to the Managing Member) and material adverse effect on the Original Members, in the case of clauses (i) and (ii).

(d) All expenses incurred by the Partnership Representative or Designated Individual in connection with its duties as partnership representative or designated individual, as applicable, shall be expenses of the Company (including, for the avoidance of doubt, any costs and expenses incurred in connection with any claims asserted against the Partnership Representative or Designated Individual, as applicable, except to the extent the Partnership Representative or Designated Individual is determined to have performed its duties in the manner described in the final sentence of this Section 5.08(d)), and the Company shall reimburse and indemnify the Partnership Representative or Designated Individual, as applicable, for all such expenses and costs. Nothing herein shall be construed to restrict the Partnership Representative or Designated Individual from engaging lawyers, accountants, tax advisers, or other professional advisers or experts to assist the Partnership Representative or Designated Individual shall be liable to the Company, any Member or any Affiliate thereof for any costs or losses to any Persons, any diminution in value or any liability whatsoever arising as a result of the performance of its duties pursuant to this Section 5.08 absent (i) willful breach of any provision of this Section 5.08 or (ii) bad faith, fraud, gross negligence or willful misconduct on the part of the Partnership Representative or Designated Individual, as applicable.

(e) The Company, the Partnership Representative, and the Members expressly agree to be bound by the terms of Section 9.04 of the Merger Agreement. Notwithstanding anything to the contrary contained in this Agreement, in the event of any conflict between Section 9.04 of the Merger Agreement and this Agreement, Section 9.04 of the Merger Agreement shall control.

**5.09 Other Allocation Provisions**. Certain of the foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with Treasury Regulations Section 1.704-1(b) and shall be interpreted and applied in a manner consistent with such regulations. In addition to amendments effected in accordance with Section 11.12 or otherwise in accordance with this Agreement, Sections 5.03, 5.04 and 5.05 may also, so long as any such amendment does not materially change the relative economic interests of the Members, be amended at any time by the Managing Member if necessary, in the opinion of tax counsel to the Company, to comply with such regulations or any applicable Law.

**5.10** Survival. Sections 5.07 and 5.08 shall be interpreted to apply to Members and former Members and shall survive the Transfer of a Member's Units and the termination, dissolution, liquidation and winding up of the Company and, for this purpose to the extent not prohibited by applicable Law, the Company shall be treated as continuing in existence.

### **ARTICLE VI**

### **BOOKS AND RECORDS; REPORTS**

### 6.01 Books and Records

(a) At all times during the continuance of the Company, the Company shall prepare and maintain separate books of account for the Company in accordance with GAAP.

(b) Except as limited by Section 6.01(c), each Member shall have the right to receive, for a purpose reasonably related to such Member's interest as a Member in the Company, upon reasonable written demand stating the purpose of such demand and at such Member's own expense:

(i) a copy of the Certificate and this Agreement and all amendments thereto, together with a copy of the executed copies of all powers of attorney pursuant to which the Certificate and this Agreement and all amendments thereto have been executed; and

(ii) promptly after their becoming available, copies of the Company's U.S. federal income tax returns for the three most recent years.

(c) The Managing Member may keep confidential from the Members, for such period of time as the Managing Member determines in its sole discretion, (i) any information that the Managing Member reasonably believes to be in the nature of trade secrets or (ii) other information the disclosure of which the Managing Member believes is not in the best interests of the Company, could damage the Company or its business or that the Company is required by law or by agreement with any third party to keep confidential, including without limitation, information as to the Units held by any other Member. With respect to any schedules, annexes or exhibits to this Agreement, each Member (other than the Managing Member) shall only be entitled to receive and review any such schedules, annexes and exhibits relating to such Member and shall not be entitled to receive or review any schedules, annexes or exhibits relating to any other Member (other than the Managing Member).

(d) The Managing Member shall cause to be prepared and filed all necessary federal and state income tax returns for the Company, including making any tax elections. At the Company's expense, the Managing Member, within 75 days of the close of the Fiscal Year, shall use commercially reasonable efforts to furnish to each Member that was a Member during such Fiscal Year a Schedule K-1 and such other tax information reasonably required for federal, state and local income tax reporting purposes. The Company shall use commercially reasonable efforts to provide to each Person that was a Member during the Fiscal Year (a) by May 15th, August 15th and November 15th of such Fiscal Year, with an estimate of the taxable income, gains, deductions, losses and other items for, respectively, the first, second and third fiscal quarters that such Person will be required to include in its taxable income and (b) by February 15th of such Fiscal Year, with an estimate of the taxable income, gains, deductions, losses and other items of such Person to be reflected on the Schedule K-1 of such Person for the prior Fiscal Year. The Company also shall provide the Members with such other information as may be reasonably requested for purposes of allowing the Members to prepare and file their own tax returns, *provided* that any costs or expenses with respect to the foregoing shall be borne by the requesting Member.

(e) The Managing Member shall make the following elections on the appropriate tax returns and shall not rescind them without the prior written consent of the Original Member Representative (*provided* that the election described in clause (ii) below cannot be rescinded without the prior written consent of the all the Members):

(i) to adopt an appropriate federal income tax method of accounting and to keep the Company's books and records on such income-tax method;

(ii) to have in effect (and to cause each direct or indirect subsidiary that is treated as a partnership for U.S. federal income tax purposes to have in effect) an election, pursuant to Section 754 of the Code (and any similar election for state or local tax purposes), to adjust the tax basis of Company properties, for the taxable year of the Company that includes the Effective Date and each subsequent taxable year in which an Exchange Transaction occurs; and

(iii) any other available election that the Managing Member deems appropriate; provided that, for so long as the Original Members collectively own at least 10% of the Units, the Managing Member shall consult in good faith with the Original Member Representative with respect to any material tax election with respect to the Company that could reasonably be expected to have a disproportionate (as compared to the Managing Member) and adverse effect on the Original Members, and not make such election without the Original Member Representative's prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned).

No Member may make an election for the Company to be excluded from the application of the provisions of subchapter K of chapter 1 of subtitle A of the Code or any similar provisions of applicable state law, and no provision of this Agreement shall be construed to sanction or approve such an election.

#### **ARTICLE VII**

### **COMPANY UNITS**

### 7.01 Units.

(a) Limited liability company interests in the Company shall be represented by Units. At the execution of this Agreement, the Units are comprised of three Classes:

(i) *"Class A Units*". Immediately after giving effect to the transactions contemplated by the Merger Agreement, each Member holds the number of Class A Units set forth opposite such Member's name on **Exhibit A** attached hereto.

(ii) "*Class P Units*". Class P Units shall consist of those Class P Units currently outstanding and those Class P Units to be issued from time to time under Section 7.05 and the applicable Class P Unit Agreements relating to such Class P Units. Class P Units shall have all the rights, privileges, preferences, and obligations as are specifically provided for in such Class P Unit Agreements and in this Agreement for Class P Units, and as may otherwise be generally applicable to all classes of Units, unless such application is specifically limited to one or more other classes of Units. Notwithstanding anything to the contrary contained herein or in such Class P Unit Agreements, the Class P Units shall not be entitled to vote on any matter subject to a vote of the Members, except as otherwise required by law.

(iii) "*LTIP Units*." LTIP Units shall consist of those Units to be issued under Schedule II hereto and the applicable Vesting Agreements relating to such LTIP Units. LTIP Units shall have all the rights, privileges, preferences, and obligations as are specifically provided for in such Vesting Agreements and in this Agreement (including Schedule II hereto) for LTIP Units, and as may otherwise be generally applicable to all classes of Units, unless such application is specifically limited to one or more other classes of Units. Notwithstanding anything to the contrary contained herein (including Schedule II hereto) or in such Vesting Agreements, the LTIP Units shall not be entitled to vote on any matter subject to a vote of the Members, except as otherwise required by law.

(b) Subject to Section 7.04, the Managing Member in its sole discretion may establish and issue, from time to time in accordance with such procedures as the Managing Member shall determine from time to time, additional Units, in one or more classes or series of Units, or other Company securities, at such price, and with such designations, preferences and relative, participating, optional or other special rights, powers and duties (which may be senior to existing Units, classes and series of Units or other Company securities), as shall be determined by the Managing Member without the approval of any Member or any other Person who may acquire an interest in any of the Units, including (i) the right of such Units to share in Profits and Losses or items thereof; (ii) the right of such Units to share in Company distributions; (iii) the rights of such Units upon dissolution and winding up of the Company; (iv) whether, and the terms and conditions upon which, the Company may or shall be required to redeem such Units (including sinking fund provisions); (v) whether such Units are issued with the privilege of conversion or exchange and, if so, the terms and conditions of such conversion or exchange; (vi) the terms and conditions upon which such Units will be issued, evidenced by certificates and assigned or transferred; (vii) the method for determining the Class A Percentage Interest and Class A/LTIP Percentage Interest, as applicable, as to such Units; (viii) the terms and conditions of the issuance of such Units (including, without limitation, the amount and form of consideration, if any, to be received by the Company in respect thereof, the Managing Member being expressly authorized, in its sole discretion, to cause the Company to issue such Units for less than fair market value); and (ix) the right, if any, of the holder of such Units to vote on Company matters, including matters relating to the relative designations, preferences, rights, powers and duties of such Units. Notwithstanding any other provision of this Agreement, the Managing Member in its sole discretion, without the approval of any Member or any other Person, is authorized (i) to issue Units or other Company securities of any newly established class or any existing class to Members or other Persons who may acquire an interest in the Company; (ii) to amend this Agreement to reflect the creation of any such new class, the issuance of Units or other Company securities of such class, and the admission of any Person as a Member which has received Units or other Company securities; and (iii) to effect the combination, subdivision and/or reclassification of outstanding Units as may be necessary or appropriate to give economic effect to equity investments in the Company by the Managing Member that are not accompanied by the issuance by the Company to the Managing Member of additional Units and to update the books and records of the Company accordingly. Except as expressly provided in this Agreement to the contrary, any reference to "Units" shall include the Class A Units, Class P Units, LTIP Units, and Units of any other class or series that may be established in accordance with this Agreement. All Units of a particular class shall have identical rights in all respects as all other Units of such class, except in each case as otherwise specified in this Agreement.

(c) Notwithstanding anything to the contrary in this Agreement, the Managing Member shall not cause or permit the Company to issue, or authorize the issuance of, any Units unless the Managing Member has a sufficient number of Class A Common Stock authorized, available and reserved for issuance upon an exchange of such newly issued Units for Class A Common Stock pursuant to an Exchange Transaction.

**7.02 Register**. The books and records of the Company shall be the definitive record of ownership of each Unit and all relevant information with respect to each Member. Unless the Managing Member in its sole discretion shall determine otherwise, Units shall be uncertificated and recorded in the books and records of the Company.

**7.03 Registered Members.** The Company shall be entitled to recognize the exclusive right of a Person registered on its records as the owner of Units for all purposes and shall not be bound to recognize any equitable or other claim to or interest in Units on the part of any other Person, whether or not it shall have express or other notice thereof, except as otherwise provided by the Act or other applicable Law.

# 7.04 Issuances, Repurchases and Redemptions, Recapitalizations.

# (a) <u>Issuances by the Managing Member</u>

(i) Subject to Section 7.04(a)(ii) and the Exchange Agreement, if, at any time after the Closing Date, the Managing Member sells or issues shares of Class A Common Stock or any other Equity Interests of the Managing Member (other than Class V Common Stock), (x) the Company shall concurrently issue to the Managing Member an equal number of Class A Units (if the Managing Member issues Class A Common Stock), or an equal number of such other Equity Interests of the Equity Interests issued by the Managing Member (if the Managing Member issues Equity Interests other than Class A Common Stock), and with substantially the same rights to dividends and distributions (including distributions upon liquidation) and other economic rights as those of such Equity Interests of the Managing Member so issued and (y) the Managing Member shall concurrently contribute to the Company, the net proceeds or other property received by the Managing Member, if any, for such Class A Common Stock or other Equity Interest.

(ii) Notwithstanding anything to the contrary contained in <u>Section 7.04(a)(i)</u> or <u>Section 7.04(a)(ii)</u>, this <u>Section 7.04(a)</u> shall not apply to (x) the issuance and distribution to holders of Class A Common Stock or other Equity Interests of the Managing Member of rights to purchase Equity Interests of the Managing Member under a "poison pill" or similar shareholder rights plan (and upon exchange of Class A Units for Class A Common Stock, such Class A Common Stock will be issued together with a corresponding right under such plan) or (y) the issuance under the Managing Member's employee benefit plans of any warrants, options, stock appreciation right, restricted stock, restricted stock units, performance based award or other rights to acquire Equity Interests of the Managing Member in connection with the exercise or settlement of such warrants, options, stock appreciation right, restricted stock (including as set forth in clause (iii) below, as applicable).

(iii) In the event any outstanding Equity Interest of the Managing Member is exercised or otherwise converted and, as a result, any shares of Class A Common Stock or other Equity Interests of the Managing Member are issued (including as a result of the exercise of warrants of the Managing Member), (x) the corresponding Equity Interest outstanding at the Company, if any, shall be similarly exercised or otherwise converted, if applicable, (y) an equivalent number of Class A Units or equivalent Equity Interests of the Company shall be issued to the Managing Member as required by the first sentence of <u>Section 7.04(a)(i)</u>, and (z) the Managing Member shall concurrently contribute to the Company the net proceeds received by the Managing Member from any such exercise or conversion.

(b) <u>New Company Equity Interests</u>. Except pursuant to the Exchange Agreement, (x) the Company may not issue any additional Class A Units or Equity Interests of the Company to the Managing Member or any of its Subsidiaries (other than the Company and its Subsidiaries) unless substantially simultaneously therewith the Managing Member or such Subsidiary issues or transfers an equal number of newly-issued shares of Class A Common Stock of the Managing Member (or relevant Equity Interest of such Subsidiary) to another Person or Persons and contributes the net proceeds therefrom to the Company, and (y) the Company may not issue any other Equity Interests of the Company to the Managing Member or any of its Subsidiaries (other than the Company and its Subsidiaries) unless substantially simultaneously therewith the Managing Member or such Subsidiaries (other than the Company and its Subsidiaries) unless substantially simultaneously therewith the Managing Member or such Subsidiaries (other than the Company and its Subsidiaries) unless substantially simultaneously therewith the Managing Member or such Subsidiaries or transfers, to another Person, an equal number of newly-issued shares of Equity Interests of the Managing Member or such Subsidiary with substantially the same rights to dividends and distributions (including distributions upon liquidation) and other economic rights as those of such Equity Interests of the Company and contributes the net proceeds therefrom to the Company.

### (c) <u>Repurchases and Redemptions</u>.

(i) Subject to <u>Section 7.05</u>, neither the Managing Member nor any of its Subsidiaries (other than the Company and its Subsidiaries) may redeem, repurchase or otherwise acquire (A) Class A Common Stock pursuant to a Board approved repurchase plan or program (or otherwise in connection with a transaction approved by the Board) unless substantially simultaneously therewith the Company redeems, repurchases or otherwise acquires from the Managing Member or such Subsidiary an equal number of Class A Units for the same price per security, if any, or (B) any other Equity Interests of the Managing Member or any of its Subsidiaries (other than the Company and its Subsidiaries) pursuant to a Board approved repurchase plan or program (or otherwise in connection with a transaction approved by the Board) unless substantially simultaneously therewith the Company redeems, repurchases or otherwise acquires from the Managing Member or such subsidiaries (other than the Company and its Subsidiaries) pursuant to a Board approved repurchase plan or program (or otherwise in connection with a transaction approved by the Board) unless substantially simultaneously therewith the Company redeems, repurchases or otherwise acquires from the Managing Member or such Subsidiary an equal number of the corresponding class or series of Equity Interests of the Company with the same rights to dividends and distributions (including distributions upon liquidation) and other economic rights as those of such Equity Interests of the Managing Member or such Subsidiary for the same price per security, if any.

(ii) Subject to Section 7.06, the Company may not redeem, repurchase or otherwise acquire (x) any Class A Units from the Managing Member or any of its Subsidiaries (other than the Company and its Subsidiaries) unless substantially simultaneously the Managing Member or such Subsidiary redeems, repurchases or otherwise acquires pursuant to a Board approved repurchase plan or program (or otherwise in connection with a transaction approved by the Board) an equal number of shares of Class A Common Stock for the same price per security from holders thereof or (y) any other Equity Interests of the Company from the Managing Member or any of its Subsidiaries (other than the Company and its Subsidiaries) unless substantially simultaneously the Managing Member or such Subsidiary redeems, repurchases or otherwise acquires pursuant to a Board approved repurchase plan or program (or otherwise in connection with a transaction approved by the Board) for the same price per security an equal number of Equity Interests of the Managing Member or such Subsidiary redeems, repurchases or otherwise acquires pursuant to a Board approved repurchase plan or program (or otherwise in connection with a transaction approved by the Board) for the same price per security an equal number of Equity Interests of the Managing Member (or such Subsidiary) of a corresponding class or series with substantially the same rights to dividends and distributions (including distributions upon liquidation) and other economic rights as those of such Equity Interests of the Managing Member or such Subsidiary.

# (d) <u>Equity Subdivisions and Combinations</u>.

(i) The Company shall not in any manner effect any subdivision (by any equity split, equity distribution, reclassification, recapitalization or otherwise) or combination (by reverse equity split, reclassification, recapitalization or otherwise) of the outstanding Equity Interests of the Company unless accompanied by an identical subdivision or combination, as applicable, of the outstanding related class or series of Equity Interest of the Managing Member, with corresponding changes made with respect to any other exchangeable or convertible Equity Interests of the Company and the Managing Member.

(ii) the Managing Member shall not in any manner effect any subdivision (by any equity split, equity distribution, reclassification, recapitalization or otherwise) or combination (by reverse equity split, reclassification, recapitalization or otherwise) of any class or series of Equity Interest of the Managing Member, unless accompanied by an identical subdivision or combination, as applicable, of the outstanding related class or series of Equity Interest of the Company, with corresponding changes made with respect to any applicable exchangeable or convertible Equity Interests of the Company and the Managing Member.

(e) <u>General Authority</u>. For the avoidance of doubt, but subject to <u>Section 7.01</u>, <u>Section 7.02</u>, <u>Section 7.04</u> and <u>Section 7.05</u>, the Company and the Managing Member shall be permitted to undertake all actions, including an issuance, redemption, reclassification, distribution, division or recapitalization, with respect to the Class A Units as the Managing Member determines is necessary to maintain at all times a one-to-one ratio between (i) the number of Class A Units owned by the Members, directly or indirectly, and the number of outstanding shares of Class A Common Stock, and (ii) the number of outstanding shares of Class V Common Stock held by any Member other than the Managing Member and the number of Class A Units held by such Member disregarding, for purposes of maintaining the one-to-one ratios in clause (i), (A) options, rights or securities of the Managing Member issued under any plan involving the issuance of any Equity Interests that are convertible into or exercisable or exchangeable for shares of Class A Common Stock, (B) treasury stock, or (C) preferred stock or other debt or equity securities (including warrants, options or rights) issued by the Managing Member that are convertible or into or exercisable or exchangeable for Class A Common Stock (but in each case prior to such conversion, exercise or exchange).

# 7.05 Class P Units.

(a) From time to time, the Managing Member shall have the power and discretion to approve the issuance of Class P Units to any director, employee, officer, consultant or other service provider of the Managing Member, the Company or any Subsidiary of the Company (each such person, a "**Management Member**"). The Managing Member shall have power and discretion to approve which directors, employees, officers, consultants or other service providers shall be offered and issued such Class P Units, the number of Class P Units to be offered and issued to each Management Member and the purchase price and other terms and conditions with respect thereto.

(b) The provisions of this <u>Section 7.05</u> are designed to provide incentives to directors, employees, officers, consultants or other service providers of the Company or its Subsidiaries. This <u>Section 7.05</u>, together with the other terms of this Agreement and the Class P Unit Agreements relating to Class P Units, are intended to be a compensatory benefit plan within the meaning of Rule 701 of the Securities Act, and, unless and until the Company's Equity Interests are publicly traded, the issuance of Class P Units are, to the extent permitted by applicable federal securities laws, intended to qualify for the exemption from registration under Rule 701 of the Securities Act.

(c) On the date hereof, the Managing Member will establish and document in the books and records of the Company the current Participation Threshold (as defined below) amount, and, if applicable, vesting schedule, with respect to Class P units that are outstanding on the Effective Date.

(d) On the date of each future grant of Class P Units to a Management Member, the Managing Member will establish (and document in the applicable Class P Unit Agreement) an initial "*Participation Threshold*" amount with respect to each such Class P Unit granted on such date. The Participation Threshold with respect to each Class P Unit will be at least equal to the amount a Class A Unit would receive on the date of issuance of such Class P Unit in a hypothetical liquidation of the Company on the date of issuance of such Class P Unit in which the Company sold its assets for their Fair Market Value, satisfied its liabilities (excluding any nonrecourse liabilities to the extent the balance of such liabilities exceeds the fair market value of the assets that secure them) and distributed the net proceeds to the holders of Units in liquidation of the Company. The determination by the Managing Member of each Participation Threshold (as reasonably determined by the Managing Member based on the public trading price of Class A Common Stock) shall be final, conclusive and binding on all Members. Each Class P Unit is intended to be a "profits interest" within the meaning of IRS Revenue Procedures 93-27 and 2001-43 and is issued with the intention that under current interpretations of the Code the recipient will not realize income upon the issuance of the Class P Unit, and that neither the Company nor any Member is entitled to any deduction either immediately or through depreciation or amortization as a result of the issuance of such Class P Unit.

follows:

(e) Each Class P Unit's Participation Threshold shall be adjusted after the grant of such Class P Unit as

(i) In the event of any Distribution pursuant to <u>Section 4.01(a) or Section 4.02</u>, the Participation Threshold of each Class P Unit outstanding at the time of such Distribution shall be reduced (but not below zero) by the amount distributable to the holder of a single Class A Unit in connection with such Distribution (determined pursuant to <u>Section 4.01(a)</u> or <u>Section 4.02</u> of this Agreement and taking into account all Class P Units that are entitled to participate in such Distribution as contemplated by <u>Section 4.01</u>; and

(ii) If the Company at any time subdivides (by any Unit split, Unit dividend or otherwise) its outstanding Units into a greater number of Units, the Participation Threshold of each Class P Unit in effect immediately prior to such subdivision shall be proportionately reduced, and if the Company at any time combines (by reverse Unit split or otherwise) its outstanding Units into a smaller number of Units, the Participation Threshold of each Class P Unit in effect immediately prior to such combination shall be proportionately increased.

(f) In connection with any approved issuance of Class P Units to a Management Member hereunder, such Management Member shall, if it has not already done so, execute a counterpart to this Agreement (or a joinder to this Agreement in a form acceptable to the Company), accepting and agreeing to be bound by all terms and conditions hereof, and shall enter into such other documents and instruments to effect such purchase (including, without limitation, a Class P Unit Agreement) as are required by the Managing Member and in connection therewith, be admitted as a member of the Company, if not already a Member.

(g) If the Managing Member so determines, the Class P Units issued to any Management Member shall become vested in accordance with the vesting schedule determined by the Managing Member in connection with the issuance of such Class P Units (and reflected in the relevant Class P Unit Agreement), which may be time-based or performance-based. Notwithstanding any other provision in this <u>Section 7.05</u>, each recipient of a Class P Unit hereby agrees that such recipient shall make a valid and timely election in respect of such Unit, upon receipt thereof, pursuant to Section 83(b) of the Code and promptly provide evidence of such election to the Company.

(h) By executing this Agreement, each Member authorizes and directs the Company to elect to have the "Safe Harbor" described in the proposed Revenue Procedure set forth in Internal Revenue Service Notice 2005-43 (the "IRS Notice") apply to any interest in the Company transferred to a service provider by the Company on or after the effective date of such Revenue Procedure in connection with services provided to the Company, including the Class P Units and LTIP Units. For purposes of making such Safe Harbor election, the Partnership Representative is hereby designated as the "member who has responsibility for federal income tax reporting" by the Company and, accordingly, execution of such Safe Harbor election by the Partnership Representative constitutes execution of a "Safe Harbor Election" in accordance with Section 3.03(1) of the IRS Notice, including, without limitation, the requirement that each Member shall prepare and file all federal income tax returns reporting the income tax effects of each "Safe Harbor Partnership Interest" issued by the Company in a manner consistent with the requirements of the IRS Notice. A Member's obligations to comply with the requirements of this <u>Section 7.05(h</u>), shall survive such Member's ceasing to be a member of the Company and/or the dissolution, liquidation, winding up and termination of the Company, and, for purposes of this <u>Section 7.05(h</u>), the Company shall be treated as continuing in existence.

# 7.06 Triggering Events for Management Members.

(a) Upon the occurrence of any of the following events (each, a "*Triggering Event*") with respect to a Management Member, the Company shall have the option to purchase all or a portion of such Management Member's (the "*Affected Member*") Class P Units (the "*Affected Units*") on the date that the Triggering Event occurred at the price and on the terms set forth in this <u>Section 7.06</u>:

(i) the attachment of, execution against, levy upon or seizure of the Management Member's Units (other than an attachment solely for jurisdictional purposes) unless (and for only so long as) counsel of the Company determines that such Management Member is in good faith contesting such attachment, execution, levy or other seizure;

(ii) (A) an assignment by the Management Member for the benefit of creditors, which assignment includes such Management Member's Units, or (B) a petition for bankruptcy by or on behalf of the Management Member;

(iii) the death or Disability of the Management Member;

(iv) if the Management Member is a legal entity, the winding up an dissolution of such Management Member or the merger or reorganization of such Management Member as a result of which such Management Member does not survive as an entity or the Persons indirectly or directly controlling such Management Member immediately prior to such merger or reorganization no longer control such Management Member;

(v) the possession by any Person of the Management Member's Units or of a claim to, lien on, interest in or encumbrance upon such Management Member's Units, other than a Person that acquired such Units, claim, lien, interest or encumbrance upon the Managing Member's prior written consent in accordance with the conditions hereof;

(vi) the termination of the Management Member's employment or services to the Managing Member, the Company or any Subsidiary of the Company for Malfeasance;

(vii) the Transfer or attempted Transfer (whether voluntary or involuntary) of Units in violation of this Agreement or any other applicable agreement; and

(viii) the Management Member engages in Competing Activity.

Within ten (10) calendar days after the occurrence of any of the above Triggering Events, the Affected Member (or the Affected Member's representative) shall provide a written communication setting forth the details of such event (the "*Triggering Event Communication*") to the Managing Member. Failure of the Affected Member (or the Affected Member's representative) to provide such Triggering Event Communication shall in no way prevent the Company from exercising its rights or relieve such Affected Member (the Affected Member's representative) from satisfying its obligations under this Agreement. Notwithstanding anything to contrary in this <u>Section 7.06</u>, (x) the Class P Unit Agreement with respect to certain Class P Units can provide an alternative definition of Triggering Event for purposes of this Agreement, in which case the definition of Triggering Event in the Class P Unit Agreement shall control, and (y) in the event of a Triggering Event set forth in <u>Section 7.06(a)(vi) or Section 7.06(a)</u> (<u>viii</u>), the Affected Units shall, at the Company's option, be automatically forfeited by the Affected Member rather than repurchased for the Triggering Event Purchase Price.

(b) If an Affected Member has Transferred all or any portion of his or her Class P Units to a Transferee, this Section 7.06 shall also apply, mutatis mutandis, to such Transferee as though such Transferee was the Management Member referenced in such Section. This Section 7.06 shall be binding upon each Management Member and his or her heirs, executor, administrator, guardian or other legal representative (collectively, his or her "*representative*"), and his or her Transferees.

(c) Upon the occurrence of a Triggering Event, the Company shall have the option, exercisable at the sole discretion of the Managing Member, to purchase all or any of the Affected Units held by the Affected Member at the price determined in accordance with this <u>Section 7.06</u> (the "*Triggering Event Purchase Price*"). The Company may exercise its right to purchase (the "*Call Option*") the Affected Units by providing written notice (the "*Call Option Exercise Notice*") to the Affected Member at any time following occurrence of the Triggering Event. Upon the occurrence of a Triggering Event, the Affected Member shall forfeit any Class P Units which have not vested pursuant to the terms of the applicable Class P Unit Agreement.

(d) In the event the Company provides the Call Option Exercise Notice within thirty (30) calendar days after receipt of the Triggering Event Communication, the Triggering Event Purchase Price shall be calculated as follows:

(i) If the Triggering Event is any event set forth in <u>Section 7.06(a)(i)</u> through (v) or <u>Section 7.06(a)</u> (vii), the Triggering Event Purchase Price shall equal the fair market value of the Affected Units that a willing buyer would pay to a willing seller in an arm's length transaction, as determined by the Managing Member (the "*Fair Market Value*"), on the date that the Triggering Event occurred with respect to each Affected Unit that is a Vested Unit.

(ii) If the Triggering Event is an event set forth in <u>Section 7.06(a)(vi)</u> or <u>Section 7.06(a)(viii)</u> and the Company does not exercise its option under <u>Section 7.06(a)</u> to cause the Affected Units to be automatically forfeited, the Triggering Event Purchase Price shall equal ten percent (10%) of the Fair Market Value on the date the Triggering Event occurred with respect to each Affected Unit that is a Vested Unit.

(e) In the event the Company provides the Call Option Exercise Notice following the thirtieth (30<sup>th</sup>) calendar day after receipt of the Triggering Event Communication, the Triggering Event Purchase Price shall be calculated as follows:

(i) If the Triggering Event is any event set forth in <u>Section 7.06(a)(i)</u> through (v) or <u>Section 7.06(a)</u> (vii), the Triggering Event Purchase Price shall equal the Fair Market Value on the date that the Call Option Exercise Notice was issued with respect to each Affected Unit that is a Vested Unit.

(ii) If the Triggering Event is an event set forth in <u>Section 7.06(a)(vi)</u> or <u>Section 7.06(a)(viii</u>), the Triggering Event Purchase Price shall equal ten percent (10%) of the Fair Market Value on the date the Call Option Exercise Notice was issued with respect to each Affected Unit that is a Vested Unit.

#### **ARTICLE VIII**

### TRANSFER RESTRICTIONS

#### 8.01 Member Transfers

Except as otherwise agreed to in writing between the Managing Member and the applicable Member and (a) reflected in the books and records of the Company or as otherwise provided in this Article VIII, no Member or Assignee thereof may Transfer all or any portion of its Units or other interest in the Company (or beneficial interest therein) without the prior consent of the Managing Member, which consent may be given or withheld, or made subject to such conditions (including, without limitation, the receipt of such legal opinions and other documents that the Managing Member may require) as are determined by the Managing Member, in each case in the Managing Member's sole discretion, and which consent may be in the form of a plan or program entered into or approved by the Managing Member, in its sole discretion. Any such determination in the Managing Member's sole discretion in respect of Units shall be final and binding. Such determinations need not be uniform and may be made selectively among Members, whether or not such Members are similarly situated, and shall not constitute the breach of any duty hereunder or otherwise existing at law, in equity or otherwise. Any purported Transfer of Units that is not in accordance with, or subsequently violates, this Agreement shall be, to the fullest extent permitted by law, null and void. If a Member transfers all or a portion of its Class A Units to a transferee in compliance with this Agreement, the Member shall surrender a number of shares of Class V Common Stock to the Managing Member equal to the number of transferred Class A Units, such shares of Class V Common Stock will be immediately cancelled, and the Managing Member shall issue the same number of shares of Class V Common Stock to such transferee upon its admittance to the Company as a Class A Member.

(b) Notwithstanding anything otherwise to the contrary in this Section 8.01, without the consent of the Managing Member or any other Person, each Member that is a Member holding at least 5% of the Class A Percentage Interest may Transfer all or any portion of its Class A Units in a Transfer that complies with Section 8.04, unless the Managing Member timely and reasonably objects in accordance with Section 8.04, so long as such transfer does not increase the number of Members of the Company.

(c) Notwithstanding anything otherwise to the contrary in this Section 8.01, each Member may Transfer Units in Exchange Transactions pursuant to, and in accordance with, the Exchange Agreement; *provided* that in the case of any Member other than a Member holding at least 5% of the Class A Percentage Interest, that such Exchange Transactions shall be effected in compliance with reasonable policies that the Managing Member may adopt or promulgate from time to time and advise the Members of in writing (including policies requiring the use of designated administrators or brokers) in its reasonable discretion; *provided*, further, that if such policies conflict with the terms of the Exchange Agreement, the provisions of the Exchange Agreement shall apply in lieu thereof to any Exchange Transaction to the extent of such conflict.

(d) Notwithstanding anything otherwise to the contrary in this Section 8.01, (i) an individual Member may Transfer all or any portion of his or her Units without consideration to any member of his or her Family Group or (ii) any Affiliate of such Member (including any partner, shareholder or member controlling or under common control with such Member and Affiliated investment fund or vehicle of such Member), but excluding any Affiliate under this <u>clause (ii)</u> who operates or engages in a business which competes with the business of Managing Member or the Company, in each case, in a Transfer that complies with Section 8.04 and (iii) the Managing Member may implement other policies and procedures to permit the Transfer of Units by the other Members for personal planning purposes and any such Transfer effected in compliance with such policies and procedures shall not require the prior consent of the Managing Member.

**8.02 Mandatory Exchanges**. The Managing Member may in its sole discretion at any time and from time to time, without the consent of any Member or other Person, cause to be Transferred to the Managing Member in an Exchange Transaction any and all Units, except for Units held by any Member holding at least 5% of the Class A Percentage Interest. Any such determinations by the Managing Member need not be uniform and may be made selectively among Members, whether or not such Members are similarly situated.

**8.03** Encumbrances. No Member or Assignee may create an Encumbrance with respect to all or any portion of its Units (or any beneficial interest therein) other than Encumbrances that run in favor of the Member unless the Managing Member consents in writing thereto, which consent may be given or withheld, or made subject to such conditions as are determined by the Managing Member, in the Managing Member's sole discretion. Consent of the Managing Member shall be withheld until the holder of the Encumbrance acknowledges the terms and conditions of this Agreement. Any purported Encumbrance that is not in accordance with this Agreement shall be, to the fullest extent permitted by law, null and void.

# 8.04 Further Restrictions.

(a) Units issued from time to time after the date of this Agreement, including Units issued under equity incentive plans of the Company or the Managing Member (or upon settlement of awards granted under such plans), may be subject to such additional or other terms and conditions, including with regard to vesting, forfeiture, minimum retained ownership and Transfer, as may be agreed between the Managing Member and the applicable Member and reflected in the books and records of the Company. Such requirements, provisions and restrictions need not be uniform and may be waived or released by the Managing Member in its sole discretion with respect to all or a portion of the Units owned by any one or more Members at any time and from time to time, and shall not constitute the breach of any duty hereunder or otherwise existing at law, in equity or otherwise.

(b) Notwithstanding any contrary provision in this Agreement, in no event may any Transfer of a Unit (other than in accordance with the Exchange Agreement) be made by any Member or Assignee if the Managing Member determines that:

Unit;

(i)

such Transfer is made to any Person who lacks the legal right, power or capacity to own such

(ii) except pursuant to an Exchange Transaction, such Transfer would require the registration of such transferred Unit or of any Class of Unit pursuant to any applicable U.S. federal or state securities laws (including, without limitation, the Securities Act or the Exchange Act) or other non-U.S. securities laws (including Canadian provincial or territorial securities laws) or would constitute a non-exempt distribution pursuant to applicable provincial or state securities laws;

(iii) such Transfer would cause (i) all or any portion of the assets of the Company to (A) constitute "plan assets" (under ERISA, the Code or any applicable Similar Law) of any existing or contemplated Member, or (B) be subject to the provisions of ERISA, Section 4975 of the Code or any applicable Similar Law, or (ii) the Managing Member to become a fiduciary with respect to any existing or contemplated Member, pursuant to ERISA, any applicable Similar Law, or otherwise;

(iv) to the extent requested by the Managing Member, the Company does not receive such legal and/or tax opinions and written instruments (including, without limitation, copies of any instruments of Transfer and such Assignee's consent to be bound by this Agreement as an Assignee) that are in a form satisfactory to the Managing Member, as determined in the Managing Member's sole discretion; *provided* that no such legal and/or tax opinions shall be required for a Transfer by a Member holding at least 5% of the Class A Percentage Interest; or

(v) the Managing Member shall reasonably determine that such Transfer would pose a material risk that the Company would be treated as a "*publicly traded partnership*" within the meaning of Section 7704 of the Code and the regulations promulgated thereunder.

All determinations with respect to this Section 8.04 shall be made by the Managing Member in its sole discretion; provided, however, that all such determinations with respect to a Member holding at least 5% of the Class A Percentage Interest shall be made by the Managing Member exercising its reasonable discretion.

(c) In addition, notwithstanding any contrary provision in this Agreement, to the extent the Managing Member shall reasonably determine that interests in the Company do not meet the requirements of Treasury Regulation Section 1.7704-1(h) (determined taking into account the rules of Treasury Regulations Section 1.7704-1(h)(3), provided that, for such purpose, the Company and the Managing Member shall assume that each Original Member is treated as a single partner within the meaning of Regulations Section 1.7704-1(h) (determined taking into account the rules of Regulations Section 1.7704-1(h)(3)) unless otherwise required by applicable Law), the Managing Member may impose such restrictions on the Transfer of Units or other interests in the Company as the Managing Member may reasonably determine to be necessary or advisable so that the Company is not treated as a "*publicly traded partnership*" within the meaning of Section 7704 of the Code and the regulations promulgated thereunder.

(d) Transfers of Units (other than pursuant to an Exchange Transaction) that are otherwise permitted by this Article VIII may only be made on the first day of a fiscal quarter of the Company, unless the Managing Member otherwise agrees.

(e) Units that are Unvested Units, LTIP Units, or Class P Units may not be Transferred (other than pursuant to an Exchange Transaction) without the Managing Member's prior written consent, which may be given or withheld in its sole discretion.

(f) To the fullest extent permitted by law, any Transfer in violation of this Article VIII shall be deemed null and void *ab initio* and of no effect.

**8.05 Rights of Assignees**. Subject to Section 8.04(b), the Transferee of any permitted Transfer pursuant to this Article VIII will be an assignee only ("*Assignee*"), and only will receive, to the extent transferred, the distributions and allocations of income, gain, loss, deduction, credit or similar item to which the Member which transferred its Units would be entitled, and such Assignee will not be entitled or enabled to exercise any other rights or powers of a Member, such other rights, and all obligations relating to, or in connection with, such interest remaining with the transferring Member. The transferring Member will remain a Member even if it has transferred all of its Units to one or more Assignees until such time as the Assignee(s) is admitted to the Company as a Member pursuant to Section 8.07.

# 8.06 Admissions, Resignations and Removals

(a) No Person may be admitted to the Company as an additional Managing Member or substitute Managing Member without the prior written consent of each incumbent Managing Member, which consent may be given or withheld, or made subject to such conditions as are determined by each incumbent Managing Member, in each case in the sole discretion of each incumbent Managing Member. A Managing Member will not be entitled to resign as a Managing Member of the Company unless another Managing Member shall have been admitted hereunder (and not have previously been removed or resigned).

(b) No Member will be removed or entitled to resign from being a Member of the Company except in accordance with Section 8.08 hereof. Any additional Managing Member or substitute Managing Member admitted as a Managing Member of the Company pursuant to this Section 8.06 is hereby authorized to, and shall, continue the Company without dissolution.

(c) Except as otherwise provided in Article IX or the Act, no admission, substitution, resignation or removal of a Member will cause the dissolution of the Company. To the fullest extent permitted by law, any purported admission, resignation or removal that is not in accordance with this Agreement shall be null and void.

**8.07** Admission of Assignees as Substitute Members. An Assignee will become a substitute Member only if and when each of the following conditions is satisfied:

(a) the Managing Member consents in writing to such admission, which consent may be given or withheld, or made subject to such conditions as are determined by the Managing Member, in each case in the Managing Member's sole discretion;

(b) if required by the Managing Member, the Managing Member receives written instruments (including, without limitation, copies of any instruments of Transfer and such Assignee's consent to be bound by this Agreement as a substitute Member) that are in a form satisfactory to the Managing Member (as determined in its sole discretion);

(c) if required by the Managing Member, the Managing Member receives an opinion of counsel satisfactory to the Managing Member to the effect that such Transfer is in compliance with this Agreement and all applicable Law; and

(d) if required by the Managing Member, the parties to the Transfer, or any one of them, pays all of the Company's reasonable expenses connected with such Transfer (including, but not limited to, the reasonable legal and accounting fees of the Company).

**8.08 Resignation and Removal of Members**. Subject to Section 8.05, if a Member (other than the Managing Member) ceases to hold any Units then such Member shall cease to be a Member and to have the power to exercise any rights or powers of a member of the Company, and shall be deemed to have resigned from the Company.

**8.09** Withholding. In the event any transfer is permitted pursuant to this Article VIII, the transferring parties shall demonstrate to the satisfaction of the Managing Member either that no withholding is required in connection with such transfer under applicable U.S. federal, state, local or non-U.S. law (including under Section 1445 or 1446 of the Code) or that any amounts required to be withheld in connection with such transfer under applicable U.S. federal, state, local or non-U.S. law (including under Section 1446 or non-U.S. law (including under Section 1446 or non-U.S. law (including under Section 1446 or non-U.S. law)

**8.10** Allocations in Respect of Transferred Units. With regard to the Managing Member's acquisition of the Acquired Surviving Company Units (as defined in the Merger Agreement), Profits or Losses shall be allocated to the Members of the Company so as to take into account the varying interests of the Members in the Company using an "interim closing of the books" method in a manner that complies with the provisions of Section 706 of the Code and the Treasury Regulations thereunder. If during any taxable year there is any other change in any Member's Units in the Company and allocate the Profits or Losses to the Members of the Company so as to take into account the varying interests of the tax advisors to the Company and allocate the Profits or Losses to the Members of the Company so as to take into account the varying interests of the Members in the Company using an "interim closing of the books" method in a manner that complies with the provisions of Section 706 of the Code and the Treasury Regulations thereunder; provided, however, that such allocations may instead be made in another manner that complies with the provisions of Section 706 of the Code and the Treasury Regulations thereunder; provided, however, that such allocations may instead be made in another manner that complies with the provisions of Section 706 of the Code and the Treasury Regulations thereunder; provided, however, that such allocations thereunder and that is selected by the Managing Member (with the prior written consent of the Original Member Representative, not to be unreasonably withheld, conditioned or delayed); provided that, the Original Member Representative shall not have the consent right described in this Section 8.10 in the event that the Original Members collectively own less than 10% of the Units.

### ARTICLE IX

# DISSOLUTION, LIQUIDATION AND TERMINATION

**9.01 No Dissolution**. Except as required by the Act, the Company shall not be dissolved by the admission of additional Members or resignation of Members in accordance with the terms of this Agreement. The Company may be dissolved, liquidated, wound up and terminated only pursuant to the provisions of this Article IX, and the Members hereby irrevocably waive any and all other rights they may have to cause a dissolution of the Company or a sale or partition of any or all of the Company assets.

**9.02 Events Causing Dissolution**. The Company shall be dissolved and its affairs shall be wound up upon the occurrence of any of the following events:

(a) the entry of a decree of judicial dissolution of the Company under Section 18-802 of the Act upon the finding by a court of competent jurisdiction that it is not reasonably practicable to carry on the business of the Company in conformity with this Agreement;

- (b) any event which makes it unlawful for the business of the Company to be carried on by the Members;
- (c) the written consent of all Members;
- (d) at any time there are no Members, unless the Company is continued in accordance with the Act;

(e) the Incapacity, withdrawal or resignation of the Managing Member or the occurrence of a Disabling Event with respect to the Managing Member; *provided* that the Company will not be dissolved or required to be wound up in connection with any of the events specified in this Section 9.02(e) if: (i) at the time of the occurrence of such event there is at least one other Managing Member of the Company who is hereby authorized to, and elects to, carry on the business of the Company; or (ii) all remaining Members consent to or ratify the continuation of the business of the Company and the appointment of another Managing Member of the Company, effective as of the event that caused the Managing Member to cease to be a Managing Member of the Company, within 120 days following the occurrence of any such event, which consent shall be deemed (and if requested each Member shall provide a written consent or ratification) to have been given for all Members if the holders of more than 50% of the Vested Units then outstanding agree in writing to so continue the business of the Company; or

(f) the determination of the Managing Member in its reasonable discretion; *provided* that in the event of a dissolution pursuant to this clause (f), the relative economic rights of each Class of Units immediately prior to such dissolution shall be preserved to the greatest extent practicable with respect to distributions made to Members pursuant to Section 9.03 below in connection with the winding up of the Company, taking into consideration tax and other legal constraints that may adversely affect one or more parties hereto and subject to compliance with applicable laws and regulations, unless, and to the extent that, with respect to any Class of Units, holders of not less than 90% of the Units of such Class consent in writing to a treatment other than as described above.

**9.03 Distribution upon Dissolution**. Upon dissolution, the Company shall not be terminated and shall continue until the winding up of the affairs of the Company is completed. Upon the winding up of the Company, the Managing Member, or any other Person designated by the Managing Member (the "*Liquidation Agent*"), shall take full account of the assets and liabilities of the Company and shall, unless the Managing Member determines otherwise, liquidate the assets of the Company as promptly as is consistent with obtaining the fair value thereof. The proceeds of any liquidation shall be applied and distributed in the following order:

(a) First, to the satisfaction of debts and liabilities of the Company (including satisfaction of all indebtedness to Members and/or their Affiliates to the extent otherwise permitted by law) including the expenses of liquidation, and including the establishment of any reserve which the Liquidation Agent shall deem reasonably necessary for any contingent, conditional or unmatured contractual liabilities or obligations of the Company ("*Contingencies*"). Any such reserve may be paid over by the Liquidation Agent to any attorney-at-law, or acceptable party, as escrow agent, to be held for disbursement in payment of any Contingencies and, at the expiration of such period as shall be deemed advisable by the Liquidation Agent for distribution of the balance in the manner hereinafter provided in this Section 9.03; and

(b) The balance, if any, to the Members in accordance with Section 4.02.

**9.04 Time for Liquidation**. A reasonable amount of time shall be allowed for the orderly liquidation of the assets of the Company and the discharge of liabilities to creditors so as to enable the Liquidation Agent to minimize the losses attendant upon such liquidation.

**9.05 Termination**. The Company shall terminate when all of the assets of the Company, after payment of or due provision for all debts, liabilities and obligations of the Company, shall have been distributed to the holders of Units in the manner provided for in this Article IX, and the Certificate shall have been cancelled in the manner required by the Act.

**9.06 Claims of the Members.** The Members shall look solely to the Company's assets for the return of their Capital Contributions, and if the assets of the Company remaining after payment of or due provision for all debts, liabilities and obligations of the Company are insufficient to return such Capital Contributions, the Members shall have no recourse against the Company or any other Member or any other Person. No Member with a negative balance in such Member's Capital Account shall have any obligation to the Company or to the other Members or to any creditor or other Person to restore such negative balance during the existence of the Company, upon dissolution or termination of the Company or otherwise, except to the extent required by the Act.

**9.07 Survival of Certain Provisions**. Notwithstanding anything to the contrary in this Agreement, the provisions of Sections 5.07, 10.01, 10.02, 11.09 and 11.10 shall survive the termination of the Company.

### ARTICLE X

#### LIABILITY AND INDEMNIFICATION

#### 10.01 Liability of Members

(a) No Member and no Affiliate, manager, member, employee or agent of a Member shall be liable for any debt, obligation or liability of the Company or of any other Member or have any obligation to restore any deficit balance in its Capital Account solely by reason of being a Member of the Company, except to the extent required by the Act.

(b) This Agreement is not intended to, and does not, create or impose any duty (including any fiduciary duty) on any of the Members (including without limitation, the Managing Member) hereto or on their respective Affiliates. Further, notwithstanding any other provision of this Agreement or any duty otherwise existing at law or in equity, the parties hereto agree that no Member or Managing Member shall, to the fullest extent permitted by law, have duties (including fiduciary duties) to any other Member or to the Company, and in doing so, recognize, acknowledge and agree that their duties and obligations to one another and to the Company are only as expressly set forth in this Agreement; *provided, however*, that each Member shall have the duty to act in accordance with the implied contractual covenant of good faith and fair dealing.

(c) To the extent that, at law or in equity, any Member (including without limitation, the Managing Member) has duties (including fiduciary duties) and liabilities relating thereto to the Company, to another Member or to another Person who is a party to or is otherwise bound by this Agreement, the Members (including without limitation, the Managing Member) acting under this Agreement will not be liable to the Company, to any such other Member or to any such other Person who is a party to or is otherwise bound by this Agreement, for their good faith reliance on the provisions of this Agreement. The provisions of this Agreement, to the extent that they restrict or eliminate the duties and liabilities relating thereto of any Member (including without limitation, the Managing Member) otherwise existing at law or in equity, are agreed by the Members to replace to that extent such other duties and liabilities of the Members relating thereto (including without limitation, the Managing Member).

(d) The Managing Member may consult with legal counsel, accountants and financial or other advisors selected by it, and any act or omission taken by the Managing Member on behalf of the Company or in furtherance of the interests of the Company in good faith in reliance upon and in accordance with the advice of such Person as to matters the Managing Member reasonably believes to be within such Person's professional or expert competence shall be conclusively presumed to have been done or omitted in good faith and in accordance with such opinion or advice, and the Managing Member will be fully protected in so acting or omitting to act so long as such counsel or accountants or financial or other advisors were selected with reasonable care.

(e) Notwithstanding any other provision of this Agreement or otherwise applicable provision of law or equity, whenever in this Agreement the Managing Member is permitted or required to make a decision (i) in its "sole discretion" or under a grant of similar authority or latitude, such Managing Member shall be entitled to consider only such interests and factors as it desires, including its own interests, and shall, to the fullest extent permitted by applicable Law, have no duty or obligation to give any consideration to any interest of or factors affecting the Company or the Members, or (ii) in its "good faith" or under another expressed standard, such Managing Member shall act under such express standard and shall not be subject to any other or different standards.

## **10.02** Indemnification.

Exculpation and Indemnification. Notwithstanding any other provision of this Agreement, whether (a) express or implied, to the fullest extent permitted by law, no Indemnitee shall be liable to the Company or any Member for any act or omission in relation to the Company or this Agreement or any transaction contemplated hereby taken or omitted by an Indemnitee unless such Indemnitee's conduct constituted fraud, bad faith or willful misconduct. To the fullest extent permitted by law, as the same exists or hereafter be amended (but in the case of any such amendment, only to the extent that such amendment permits the Company to provide broader indemnification rights than such law permitted the Company to provide prior to such amendment), the Company shall indemnify any Indemnitee who was or is made or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding (brought in the right of the Company or otherwise), whether civil, criminal, administrative, arbitrative or investigative, and whether formal or informal (hereinafter a "Proceeding"), including appeals, by reason of his or her or its status as an Indemnitee or by reason of any action alleged to have been taken or omitted to be taken by Indemnitee in such capacity, for and against all loss and liability suffered and expenses (including attorneys' fees), judgments, fines and amounts paid in settlement reasonably incurred by such Indemnitee in connection with such action, suit or proceeding, including appeals; provided that such Indemnitee shall not be entitled to indemnification hereunder if, but only to the extent that, such Indemnitee's conduct constituted fraud, bad faith or willful misconduct. Notwithstanding the preceding sentence, except as otherwise provided in Section 10.02(c), the Company shall be required to indemnify an Indemnitee in connection with any action, suit or proceeding (or part thereof) (i) commenced by such Indemnitee only if the commencement of such action, suit or proceeding (or part thereof) by such Indemnitee was authorized by the Managing Member, and (ii) by or in the right of the Company only if the Managing Member has provided its prior written consent. The indemnification of an Indemnitee of the type identified in clause (e) of the definition of Indemnitee shall be secondary to any and all indemnification to which such Indemnitee is entitled from the relevant other Person (including any payment made to such Indemnitee under any insurance policy issued to or for the benefit of such Person or Indemnitee) (the "Primary Indemnification"), and will only be paid to the extent the Primary Indemnification is not paid and/or does not provide coverage (e.g., a self-insured retention amount under an insurance policy). No such Person shall be entitled to contribution or indemnification from or subrogation against the Company. The indemnification of any other Indemnitee shall, to the extent not in conflict with such policy, be secondary to any and all payment to which such Indemnitee is entitled from any relevant insurance policy issued to or for the benefit of the Company or any Indemnitee.

(b) Advancement of Expenses. To the fullest extent permitted by law, the Company shall promptly pay reasonable expenses (including attorneys' fees) incurred by any Indemnitee in appearing at, participating in or defending any Proceeding in advance of the final disposition of such Proceeding, including appeals, upon presentation of an undertaking on behalf of such Indemnitee to repay such amount if it shall ultimately be determined that such Indemnitee is not entitled to be indemnified under this Section 10.02 or otherwise. Notwithstanding the preceding sentence, except as otherwise provided in Section 10.02(c), the Company shall be required to pay expenses of an Indemnitee in connection with any Proceeding (or part thereof) (i) commenced by such Indemnitee only if the commencement of such action, suit or proceeding (or part thereof) by such Indemnitee was authorized by the Managing Member and (ii) by or in the right of the Company only if the Managing Member has provided its prior written consent.

(c) Unpaid Claims. If a claim for indemnification (following the final disposition of such Proceeding) or advancement of expenses under this Section 10.02 is not paid in full within 30 days after a written claim therefor by any Indemnitee has been received by the Company, such Indemnitee may file proceedings to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense of prosecuting such claim. In any such action the Company shall have the burden of proving that such Indemnitee is not entitled to the requested indemnification or advancement of expenses under applicable Law.

(d) **Insurance**. (i) To the fullest extent permitted by law, the Company may purchase and maintain insurance on behalf of any person described in Section 10.02(a) against any liability asserted against such person, whether or not the Company would have the power to indemnify such person against such liability under the provisions of this Section 10.02 or otherwise.

(ii) In the event of any payment by the Company under this Section 10.02, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of the Indemnitee from any relevant other Person or under any insurance policy issued to or for the benefit of the Company, such relevant other Person, or any Indemnitee. Each Indemnitee agrees to execute all papers required and take all action necessary to secure such rights, including the execution of such documents as are necessary to enable the Company to bring suit to enforce any such rights in accordance with the terms of such insurance policy or other relevant document. The Company shall pay or reimburse all expenses actually and reasonably incurred by the Indemnitee in connection with such subrogation.

(iii) The Company shall not be liable under this Section 10.02 to make any payment of amounts otherwise indemnifiable hereunder (including, but not limited to, judgments, fines and amounts paid in settlement, and excise taxes with respect to an employee benefit plan or penalties) if and to the extent that the applicable Indemnitee has otherwise actually received such payment under this Section 10.02 or any insurance policy, contract, agreement or otherwise.

(e) Non-Exclusivity of Rights. The provisions of this Section 10.02 shall be applicable to all actions, claims, suits or proceedings made or commenced after the date of this Agreement, whether arising from acts or omissions to act occurring before or after its adoption. The provisions of this Section 10.02 shall be deemed to be a contract between the Company and each person entitled to indemnification under this Section 10.02 (or legal representative thereof) who serves in such capacity at any time while this Section 10.02 and the relevant provisions of applicable Law, if any, are in effect, and any amendment, modification or repeal hereof shall not affect any rights or obligations then existing with respect to any state of facts or any action, suit or proceeding then or theretofore existing, or any action, suit or proceeding thereafter brought or threatened based in whole or in part on any such state of facts. If any provision of this Section 10.02 shall be found to be invalid or limited in application provided in this Section 10.02 shall neither be exclusive of, nor be deemed in limitation of, any rights to which any person may otherwise be or become entitled or permitted by contract, this Agreement or as a matter of law, both as to actions in such person's official capacity and actions in any other capacity, it being the policy of the Company that indemnification of any person whom the Company is obligated to indemnify pursuant to Section 10.02(a) shall be made to the fullest extent permitted by law.

For purposes of this Section 10.02, references to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; and references to "serving at the request of the Company" shall include any service as a director, officer, employee or agent of the Company which imposes duties on, or involves services by, such director, officer, employee, or agent with respect to an employee benefit plan, its participants, or beneficiaries.

This Section 10.02 shall not limit the right of the Company, to the extent and in the manner permitted by law, to indemnify and to advance expenses to, and purchase and maintain insurance on behalf of, persons other than persons described in Section 10.02(a).

#### **ARTICLE XI**

### MISCELLANEOUS

**11.01 Severability**. If any term or other provision of this Agreement is held to be invalid, illegal or incapable of being enforced by any rule of Law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions is not affected in any manner materially adverse to any party. Upon a determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

**11.02 Notices**. All notices, requests, claims, demands and other communications hereunder shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by courier service (delivery receipt requested), by electronic mail or by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified in a notice given in accordance with this Section 11.02):

(a) If to the Company, to:
 WM Holding Company, LLC
 41 Discovery,
 Irvine, California 92618
 Attention: General Counsel

(b) If to any Member other than the Managing Member, to such Member at the address of such Member as set forth on Exhibit A

(c) If to the Managing Member, to:

WM Technology, Inc. 41 Discovery, Irvine, California 92618 Attention: General Counsel

**11.03 Cumulative Remedies**. The rights and remedies provided by this Agreement are cumulative and the use of any one right or remedy by any party shall not preclude or waive its right to use any or all other remedies. Said rights and remedies are given in addition to any other rights the parties may have by Law.

**11.04 Binding Effect**. This Agreement shall be binding upon and inure to the benefit of all of the parties and, to the extent permitted by this Agreement, their successors, executors, administrators, heirs, legal representatives and assigns.

**11.05 Interpretation**. Throughout this Agreement, nouns, pronouns and verbs shall be construed as masculine, feminine, neuter, singular or plural, whichever shall be applicable. Unless otherwise specified, all references herein to "Articles," "Sections" and paragraphs shall refer to corresponding provisions of this Agreement.

Each party hereto acknowledges and agrees that the parties hereto have participated collectively in the negotiation and drafting of this Agreement and that he or she or it has had the opportunity to draft, review and edit the language of this Agreement; accordingly, it is the intention of the parties that no presumption for or against any party arising out of drafting all or any part of this Agreement will be applied in any dispute relating to, in connection with or involving this Agreement. Accordingly, the parties hereby waive to the fullest extent permitted by law the benefit of any rule of law or any legal decision that would require that in cases of uncertainty, the language of a contract should be interpreted most strongly against the party who drafted such language.

**11.06 Counterparts**. This Agreement may be executed and delivered (including by email or facsimile transmission) in one or more counterparts, and by the different parties hereto in separate counterparts, each of which when executed and delivered shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Copies of executed counterparts transmitted by telecopy or other electronic transmission service shall be considered original executed counterparts for purposes of this Section 11.06.

**11.07 Further Assurances**. Each Member shall perform all other acts and execute and deliver all other documents as may be necessary or appropriate to carry out the purposes and intent of this Agreement.

**11.08** Entire Agreement. This Agreement constitutes the entire agreement among the parties hereto pertaining to the subject matter hereof and supersedes all prior agreements and understandings, whether oral or written, pertaining thereto (including, without limitation, the Existing Agreement).

**11.09 Governing Law**. This Agreement shall be governed by, and construed in accordance with, the law of the State of Delaware.

#### 11.10 Submission to Jurisdiction; Waiver of Jury Trial.

(a) Any and all disputes which cannot be settled amicably with respect to this Agreement, including any action (at law or in equity), claim, litigation, suit, arbitration, hearing, audit, review, inquiry, proceeding or investigation or ancillary claims of any party, arising out of, relating to or in connection with the validity, negotiation, execution, interpretation, performance or non-performance of this Agreement or any matter arising out of or in connection with this Agreement and the rights and obligations arising hereunder or thereunder, or for recognition and enforcement of any judgment in respect of this Agreement and the rights and obligations arising hereunder or thereunder brought by a party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Chancery Court, if such court shall not have jurisdiction, any federal court located in the State of Delaware, or, if neither of such courts shall have jurisdiction, any other Delaware state court. Each of the parties hereby irrevocably submits with regard to any such dispute for itself and in respect of its property, generally and unconditionally, to the sole and exclusive personal jurisdiction of the aforesaid courts and agrees that it will not bring any dispute relating to this Agreement or any of the transactions contemplated by this Agreement in any court other than the aforesaid courts. Each party irrevocably consents to service of process in any dispute in any of the aforesaid courts by the mailing of copies thereof by registered or certified mail, postage prepaid, or by recognized overnight delivery service, to such party at such party's address referred to in Section 11.02. Each party hereby irrevocably and unconditionally waives, and agrees not to assert as a defense, counterclaim or otherwise, in any action brought by any party with respect to this Agreement (i) any claim that it is not personally subject to the jurisdiction of the aforesaid courts for any reason other than the failure to serve process in accordance with this Section 11.10; (ii) any claim that it or its property is exempt or immune from the jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise); or (iii) any objection which such party may now or hereafter have (A) to the laying of venue of any of the aforesaid actions arising out of or in connection with this Agreement brought in the courts referred to above; (B) that such action brought in any such court has been brought in an inconvenient forum and (C) that this Agreement, or the subject matter hereof or thereof, may not be enforced in or by such courts.

(b) To the extent that any party has or hereafter may acquire any immunity from jurisdiction of any court or from any legal process (whether through service or notice, attachment prior to judgment, attachment in aid of execution, execution or otherwise) with respect to itself, or to such party's property, each such party hereby irrevocably waives such immunity in respect of such party's obligations with respect to this Agreement.

(c) EACH PARTY ACKNOWLEDGES THAT IT IS KNOWINGLY AND VOLUNTARILY AGREEING TO THE CHOICE OF DELAWARE LAW TO GOVERN THIS AGREEMENT AND TO THE JURISDICTION OF DELAWARE COURTS IN CONNECTION WITH PROCEEDINGS BROUGHT HEREUNDER. THE PARTIES INTEND THIS TO BE AN EFFECTIVE CHOICE OF DELAWARE LAW AND AN EFFECTIVE CONSENT TO JURISDICTION AND SERVICE OF PROCESS UNDER 6 DEL. C. § 2708.

(d) EACH PARTY, FOR ITSELF AND ITS AFFILIATES, HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ALL RIGHT TO TRIAL BY JURY IN ANY ACTION OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THE ACTIONS OF THE PARTIES OR THEIR RESPECTIVE AFFILIATES PURSUANT TO THIS AGREEMENT OR THE OTHER TRANSACTION DOCUMENTS IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF OR THEREOF.

**11.11 Expenses**. Except as otherwise specified in this Agreement, the Company shall be responsible for all costs and expenses, including, without limitation, fees and disbursements of counsel, financial advisors and accountants, incurred by the Members and the Company in connection with the preparation, negotiation, and operation of this Agreement.

## 11.12 Amendments and Waivers

(a) This Agreement (including the Annexes hereto) may be amended, supplemented, waived or modified by the Managing Member in its sole discretion without the approval of any other Member or other Person so long as such amendment is executed and delivered to the Company by the Original Member Representative: *provided* that no amendment, including any amendment effected by way of merger, consolidation or transfer of all or substantially all the assets of the Company, may materially and adversely affect the rights of a holder of Units, as such, other than on a pro rata basis with other holders of Units of the same Class without the consent of such holder (or, if there is more than one such holder that is so affected, without the consent of a majority in interest of such affected holders in accordance with their holdings of such Class of Units); provided further, that notwithstanding the foregoing, the Managing Member may, without the written consent of any Member or any other Person, amend, supplement, waive or modify any provision of this Agreement, including Schedule I, and execute, swear to, acknowledge, deliver, file and record whatever documents may be required in connection therewith, to reflect: (1) any amendment, supplement, waiver or modification that the Managing Member determines in its reasonable discretion to be necessary or appropriate in connection with the creation, authorization or issuance of Units or any Class or series of equity interest in the Company pursuant to Section 7.01 hereof; (2) the admission, substitution, or withdrawal of Members in accordance with this Agreement, pursuant to Section 8.07 hereof; (3) a change in the name of the Company, the location of the principal place of business of the Company, the registered agent of the Company or the registered office of the Company; (4) any amendment, supplement, waiver or modification that the Managing Member determines in its reasonable discretion to be necessary or appropriate to address changes in U.S. federal income tax regulations, legislation or interpretation; and/or (5) a change in the Fiscal Year or taxable year of the Company and any other changes that the Managing Member determines to be necessary or appropriate as a result of a change in the Fiscal Year or taxable year of the Company including a change in the dates on which distributions are to be made by the Company. If an amendment has been approved in accordance with this agreement, such amendment shall be adopted and effective with respect to all Members. Upon obtaining such approvals as may be required by this Agreement, and without further action or execution on the part of any other Member or other Person, any amendment to this Agreement may be implemented and reflected in a writing executed solely by the Managing Member and the other Members shall be deemed a party to and bound by such amendment.

(b) No failure or delay by any party in exercising any right, power or privilege hereunder (other than a failure or delay beyond a period of time specified herein) shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by Law.

(c) The Managing Member may, in its sole discretion, unilaterally amend this Agreement on or before the effective date of the final regulations to provide for (i) the election of a safe harbor under Proposed Treasury Regulation Section 1.83-3(l) (or any similar provision) under which the fair market value of a Company interest (or interest in an entity treated as a partnership for U.S. federal income tax purposes) that is transferred is treated as being equal to the liquidation value of that interest, (ii) an agreement by the Company and each of its Members to comply with all of the requirements set forth in such regulations and Notice 2005-43 (and any other guidance provided by the Internal Revenue Service with respect to such election) with respect to all Company interests (or interest in an entity treated as a partnership for U.S. federal income tax purposes) transferred in connection with the performance of services while the election remains effective, (iii) the allocation of items of income, gains, deductions and losses required by the final regulations similar to Proposed Treasury Regulation Section 1.704-1(b) (4)(xii)(b) and (c), 1.704-1(b)(2)(iv)(b)(1) and any other related amendments.

(d) Except as may be otherwise required by law in connection with the winding-up, liquidation, or dissolution of the Company, each Member hereby irrevocably waives any and all rights that it may have to maintain an action for judicial accounting or for partition of any of the Company's property.

47

**11.13 No Third Party Beneficiaries**. This Agreement shall be binding upon and inure solely to the benefit of the parties hereto and their permitted assigns and successors and nothing herein, express or implied, is intended to or shall confer upon any other Person or entity, any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement (other than pursuant to Section 10.02 hereof); *provided, however*, that each employee, officer, director, agent or indemnitee of any Person who is bound by this Agreement or its Affiliates is an intended third party beneficiary of Section 11.10 and shall be entitled to enforce its rights thereunder.

**11.14 Headings**. The headings and subheadings in this Agreement are included for convenience and identification only and are in no way intended to describe, interpret, define or limit the scope, extent or intent of this Agreement or any provision hereof.

**11.15 Power of Attorney**. Each Member, by its execution hereof, hereby makes, constitutes and appoints the Managing Member as its true and lawful agent and attorney in fact, with full power of substitution and full power and authority in its name, place and stead, to make, execute, sign, acknowledge, swear to, record and file (a) this Agreement and any amendment to this Agreement that has been consented to and adopted as herein provided; (b) all amendments to the Certificate required or permitted by law or the provisions of this Agreement; (c) all certificates and other instruments (including consents and ratifications which the Members have agreed to provide upon a matter receiving the agreed support of Members) deemed advisable by the Managing Member to carry out the provisions of this Agreement and Law or to permit the Company to become or to continue as a limited liability company or entity wherein the Members have limited liability in each jurisdiction where the Company may be doing business; (d) all instruments that the Managing Member deems appropriate to reflect a change or modification of this Agreement or the Company in accordance with this Agreement; (e) all conveyances and other instruments or papers deemed advisable by the Managing Member to effect the liquidation and termination of the Company; and (f) all fictitious or assumed name certificates required or permitted (in light of the Company's activities) to be filed on behalf of the Company.

**11.16 Separate Agreements; Schedules**. Notwithstanding any other provision of this Agreement, including Section 11.12, the Managing Member in its sole discretion may, or may cause the Company to, without the approval of any Member or other Person, enter into separate subscription, letter or other agreements with individual Members that have become or will become Members after the date hereof with respect to any matter, which have the effect of establishing rights under, or altering, supplementing or amending the terms of, this Agreement. The parties hereto agree that any terms contained in any such separate agreement shall govern with respect to such future Member(s) party thereto notwithstanding the provisions of this Agreement. The Managing Member in its sole discretion may from time to time execute and deliver to the Members schedules which set forth information contained in the books and records of the Company and any other matters deemed appropriate by the Managing Member. Such schedules shall be for information purposes only and shall not be deemed to be part of this Agreement for any purpose whatsoever. Notwithstanding anything to the contrary, solely for U.S. federal income tax purposes, this Agreement, the Tax Receivable Agreement, the Exchange Agreement and any other separate agreement described in this Section 11.16 shall constitute a "partnership agreement" within the meaning of Section 761 of the Code.

**11.17 Partnership Status**. The Members intend to treat the Company as a partnership for U.S. federal income tax purposes and notwithstanding anything to the contrary herein, no election to the contrary shall be made.

48

**11.18 Delivery by Facsimile or Email**. This Agreement, the agreements referred to herein, and each other agreement or instrument entered into in connection herewith or therewith or contemplated hereby or thereby, and any amendments hereto or thereto, to the extent signed and delivered by means of a facsimile machine or email with scan or facsimile attachment, shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. No party hereto or to any such agreement or instrument shall raise the use of a facsimile machine or email to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine or email as a defense to the formation or enforceability of a contract, and each such party forever waives any such defense.

## [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

COMPANY: WM HOLDING COMPANY, LLC By: /s/ Christopher Beals Name: Christopher Beals Title: Chief Executive Officer

MANAGING MEMBER:WM TECHNOLOGY, INC.By:/s/ Christopher BealsName:Christopher BealsTitle:Chief Executive Officer

OTHER MEMBERS: CHRISTOPHER BEALS /s/ Christopher Beals (Signature)

OTHER MEMBERS: ARDEN LEE /s/ Arden Lee (Signature)

OTHER MEMBERS: BRIAN CAMIRE /s/ Brian Camire (Signature)

OTHER MEMBERS: JUANJO FEIJOO /s/ Juanjo Feijoo (Signature)

OTHER MEMBERS: JUSTIN DEAN /s/ Justin Dean (Signature)

OTHER MEMBERS: 420 CASA WAY, LLC By: /s/ Richard Ham Name: Richard Ham Title: Managing Partner

OTHER MEMBERS: CALILOVE LLC By: /s/ Sabin Burrell Name: Sabin Burrell Title: Manager

OTHER MEMBERS:BLACK DOG MANAGEMENT, LLCBy:/s/ Sabin BurrellName:Sabin BurrellTitle:Manager

OTHER MEMBERS:CERES HOLDINGS GROUP, LLCBy:/s/ William RubinName:William RubinTitle:General Partner

OTHER MEMBERS: CLAY W. HAMLIN, IV /s/ Clay W. Hamlin, IV (Signature)

OTHER MEMBERS: DOUG FRANCIS /s/ Doug Francis (Signature)

OTHER MEMBERS: FCP VENTURES IV, LLC By: FCP Ventures IV Manager LLC

By:/s/ Nirmal RoyName:Nirmal RoyTitle:Manager

OTHER MEMBERS:CERES HOLDINGS GROUP, LLCBy:/s/ William RubinName:William RubinTitle:General Partner

OTHER MEMBERS:GENCO INCENTIVES, LLCBy:/s/ Doug FrancisName:Doug FrancisTitle:Managing Member

OTHER MEMBERS: GHOST MEDIA GROUP, LLC By: /s/ Justin Hartfield Name: Justin Hartfield Title: Manager

By:/s/ Doug FrancisName:Doug FrancisTitle:Manager

OTHER MEMBERS:INCEPTACON-IV, LLCBy:The Inception Companies, LLCIts:Manager

By: /s/ Brent Cox

Name: Brent Cox Title: Manager

OTHER MEMBERS: JUSTIN HARTFIELD /s/ Justin Hartfield (Signature)

OTHER MEMBERS:LBCW HOLDINGS, LPBy:/s/ Stephen ChangName:Stephen ChangTitle:Chief Investment Officer

OTHER MEMBERS:OBM HOLDINGS LLCBy:/s/ Brent CoxName:Brent CoxTitle:Manager

OTHER MEMBERS: REDWOOD INVESTMENT HOLDINGS LLC By: /s/ JJ Name: Title:

OTHER MEMBERS:SCP OPPORTUNITIES FUND LLC – SERIES ABy:Sunderland Capital Partners LPIts:ManagerBy:Sunderland GP LLCIts:General PartnerBy:/s/ Thomas BushleyName:Thomas BushleyTitle:Managing Member

OTHER MEMBERS: SEQUOIA INVESTMENT HOLDINGS LLC By: /s/ Steven Landau Name: Steven Landau Title: Manager

OTHER MEMBERS:SEVEN EIGHT INVESTMENTS, LLCBy:/s/ Desiree HedgeName:Desiree HedgeTitle:President

OTHER MEMBERS: STEVEN JUNG /s/ Steven Jung (Signature)

OTHER MEMBERS:WM FOUNDERS LEGACY I, LLCBy:/s/ Justin HartfieldName:Justin HartfieldTitle:Board member

OTHER MEMBERS:WM FOUNDERS LEGACY I, LLCBy:/s/ Doug FrancisName:Doug FrancisTitle:Member

OTHER MEMBERS:WM FOUNDERS LEGACY II, LLCBy:/s/ Justin HartfieldName:Justin HartfieldTitle:Board member

OTHER MEMBERS:WM FOUNDERS LEGACY II, LLCBy:/s/ Doug FrancisName:Doug FrancisTitle:Member

OTHER MEMBERS:WM INVESTMENT PARTNERS, LLCBy:/s/ Eric LindbergName:Eric LindbergTitle:Chief Executive Officer

 OTHER MEMBERS:

 WM INVESTMENT PARTNERS II, LLC

 By:
 /s/ Eric Lindberg

 Name:
 Eric Lindberg

 Title:
 Chief Executive Officer

OTHER MEMBERS:YOUNG FAMILY LIMITED PARTNERSHIPBy:/s/ Morris YoungName:Morris YoungTitle:Trustee

SCHEDULE I

### **SCHEDULE II**

- 1. **Designation.** A class of Units in the Company designated as "LTIP Units" is hereby established. LTIP Units are intended to qualify as "profits interests" in the Company. The number of LTIP Units that may be issued by the Company shall not be limited.
- 2. Vesting. LTIP Units may, in the sole discretion of the Managing Member, be issued subject to vesting, forfeiture and additional restrictions on transfer pursuant to the terms of an award, vesting or other similar agreement (a "Vesting Agreement"). The terms of any Vesting Agreement may be modified by the Managing Member from time to time in its sole discretion, subject to any restrictions on amendment imposed by the relevant Vesting Agreement or by the terms of any stock incentive plan pursuant to which the LTIP Units are issued, if applicable. LTIP Units that have vested and are no longer subject to forfeiture under the terms of a Vesting Agreement are referred to as "Vested LTIP Units;" all other LTIP Units are referred to as "Unvested LTIP Units."
- 3. Forfeiture or Transfer of Unvested LTIP Units. Unless otherwise specified in the relevant Vesting Agreement, upon the occurrence of any event specified in a Vesting Agreement resulting in either the forfeiture of any LTIP Units or the repurchase thereof by the Company at a specified purchase price, then, upon the occurrence of the circumstances resulting in such forfeiture or repurchase by the Company, the relevant LTIP Units shall immediately, and without any further action, be treated as cancelled and no longer outstanding for any purpose. Unless otherwise specified in the relevant Vesting Agreement, no consideration or other payment shall be due with respect to any LTIP Units that have been forfeited; *provided* that with respect to any distribution declared with a record date prior to the effective date of such forfeiture, such forfeited LTIP Units shall be included in calculating the applicable holder's Class A/LTIP Percentage Interest in accordance with Article IV of this Agreement.
- 4. **Legend.** Any certificate evidencing an LTIP Unit shall bear an appropriate legend indicating that additional terms, conditions and restrictions on transfer, including without limitation provisions set forth in the Vesting Agreement, apply to the LTIP Unit.
- 5. Adjustments. If an LTIP Unit Adjustment Event (as defined below) occurs, then the Managing Member shall make a corresponding adjustment to the LTIP Units to maintain the same correspondence between Class A Units and LTIP Units as existed prior to such LTIP Unit Adjustment Event. The following shall be "LTIP Unit Adjustment Events:" (A) the Company makes a distribution on all outstanding Class A Units in Units, (B) the Company subdivides the outstanding Class A Units into a greater number of Units or combines the outstanding Class A Units into a smaller number of Units, or (C) the Company issues any Units in exchange for its outstanding Class A Units by way of a reclassification or recapitalization. If more than one LTIP Unit Adjustment Event occurs, the adjustment to the LTIP Units need be made only once using a single formula that takes into account each and every LTIP Unit Adjustment Event as if all LTIP Unit Adjustment Events occurred simultaneously. If the Company takes an action affecting the Class A Units other than actions specifically described above as LTIP Unit Adjustment Events and in the opinion of the Managing Member such action would require an adjustment to the LTIP Units to maintain the correspondence between Class A Units and LTIP Units as it existed prior to such action, the Managing Member shall make such adjustment to the LTIP Units, to the extent permitted by law and by the terms of any Vesting Agreement or stock incentive plan pursuant to which the LTIP Units have been issued, in such manner and at such time as the Managing Member, in its sole discretion, may determine to be appropriate under the circumstances to maintain such correspondence. If an adjustment is made to the LTIP Units as herein provided, the Company shall promptly file in the books and records of the Company an officer's certificate setting forth such adjustment and a brief statement of the facts requiring such adjustment, which certificate shall be conclusive evidence of the correctness of such adjustment absent manifest error. Promptly after filing such certificate, the Company shall mail or otherwise provide notice to each holder of LTIP Units setting forth the adjustment to his or her LTIP Units and the effective date of such adjustment.

6. Members' Rights to Transfer. Subject to the terms of the relevant Vesting Agreement or other document pursuant to which LTIP Units are granted, a LTIP Unit Member may not transfer all or any portion of his or her LTIP Units.

#### 7. Allocations and Distributions.

- 7.1. All distributions shall be made to holders of LTIP Units in accordance with the provisions of Article IV of this Agreement.
- 7.2. All allocations, including allocations of Profit and Loss of the Company, special allocations and allocations upon final liquidation, shall be made to holders of LTIP Units in accordance with Article IV of this Agreement.

**EXECUTION COPY** 

### AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

THIS AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "*Agreement*"), dated as of June 16, 2021, is made and entered into by and among WM Technology, Inc., a Delaware corporation (the "*Company*"), Silver Spike Sponsor, LLC, a Delaware limited liability company (the "*Sponsor*"), and the other undersigned parties listed under Holders on the signature pages hereto (each, a "*Holder*" and, collectively, the "*Holders*").

# RECITALS

WHEREAS, concurrently with the execution of this Agreement, the Company is entering into that certain Agreement and Plan of Merger (the "*Merger Agreement*") by and among the Company, Silver Spike Merger Sub LLC, a Delaware limited liability company and wholly owned subsidiary of the Company, WM Holding Company, LLC, a Delaware limited liability company ("*WM Holding*") and Ghost Media Group, LLC, a Nevada limited liability company, solely in its capacity as the initial holder representative under the Merger Agreement, to effect the business combination with WM Holding (the "*Business Combination*");

**WHEREAS**, the Company and the Sponsor are parties to that certain Registration Rights Agreement dated as of August 7, 2019 (the "*Original Agreement*"), pursuant to which the Company granted the Sponsor certain registration rights with respect to certain securities of the Company; and

**WHEREAS**, as a condition of, and as a material inducement for WM Holding to enter into and consummate the transactions contemplated by the Merger Agreement, the Company and the Sponsor have agreed to amend and restate the Original Agreement in order to provide certain registration rights relating to the registration of shares of Common Stock (as defined below) held by the equityholders of WM Holding as of and contingent upon the closing of the Business Combination.

**NOW, THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree that the Original Agreement is hereby amended and restated in its entirety, as of and contingent upon the closing of the Business Combination, as follows:

#### ARTICLE I DEFINITIONS

1.1 <u>Definitions</u>. The terms defined in this Article I shall, for all purposes of this Agreement, have the respective meanings set forth below:

"Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or any principal financial officer of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed and (iii) the Company has a bona fide business purpose for not making such information public.

"Agreement" shall have the meaning given in the Preamble.

"Board" shall mean the Board of Directors of the Company.

"*Business Day*" means any day, other than a Saturday or a Sunday, that is neither a legal holiday nor a day on which banking institutions are generally authorized or required by law or regulation to close in the City of New York, New York.

"Commission" shall mean the Securities and Exchange Commission.

"*Common Stock*" shall mean the Class A shares of common stock, par value \$0.0001 per share, of the Company outstanding immediately following the transactions contemplated by the Merger Agreement.

"Common Stock Equivalents" shall mean any rights, warrants, options, convertible securities or indebtedness, exchangeable securities or indebtedness, or other rights, exercisable for or convertible or exchangeable into, directly or indirectly, Class A Common Stock and securities convertible or exchangeable into Class A Common Stock, whether at the time of issuance or upon the passage of time or the occurrence of such future event, including Units and the Class V Common Stock of the Company.

"*Company*" shall have the meaning given in the Preamble.

"*Demanding Holders*" shall have the meaning given in <u>subsection 2.1.1</u>.

"*Demand Registration*" shall have the meaning given in <u>subsection 2.1.2</u>.

"Demand Registration Period" shall have the meaning given in subsection 2.1.2.

"Demand Registration Request" shall have the meaning given in subsection 2.1.2.

"Effectiveness Date" shall have the meaning given in <u>subsection 2.1.1</u>.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.

"Filing Date" shall have the meaning given in <u>subsection 2.1.1(a)</u>.

*"Form S-1"* shall mean Form S-1 for the registration of securities under the Securities Act promulgated by the Commission.

"Form S-1 Shelf" shall have the meaning given in subsection 2.1.6.

"Form S-3" shall have the meaning given in <u>subsection 2.4</u>.

"Form S-3 Shelf" shall have the meaning given in <u>subsection 2.1.6</u>.

"Founder Shares Lock-up Period" shall mean, with respect to the Common Stock held by the Sponsor from and after the closing of the Business Combination, the period ending on the earlier of (A) one year after the date hereof or (B) subsequent to the date hereof, (x) if the last reported sale price of the Common Stock equals or exceeds \$12.00 per share (as adjusted for share splits, share dividends, rights issuances, subdivisions, reorganizations, recapitalizations and the like) for any 20 trading days within any 30-trading day period commencing at least 150 days after the date of the closing of the Business Combination or (y) in any case, if, after the date hereof, the Company completes a liquidation, merger, amalgamation, share exchange, reorganization or other similar transaction that results in all of the Company's public shareholders having the right to exchange their shares of common stock for cash, securities or other property.

"Holders" shall have the meaning given in the Preamble.

"*Insider Letter*" shall mean that certain letter agreement, dated as of August 7, 2019, by and among the Company, the Sponsor and each of the Company's officers, directors and director nominees.

"Maximum Number of Securities" shall have the meaning given in subsection 2.1.4.

"Merger Agreement" shall have the meaning set forth in the Recitals hereto.

# "Minimum Demand Threshold" shall mean \$10.0 million.

"*Misstatement*" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus or necessary to make the statements in a Registration Statement or Prospectus (in the case of a Prospectus, in the light of the circumstances under which they were made) not misleading.

"Original Agreement" shall have the meaning set forth in the Recitals hereto.

"*Permitted Transferees*" shall mean any person or entity to whom the Sponsor is permitted to transfer such Registrable Securities prior to the expiration of the Founder Shares Lock-up Period or Private Placement Lock-up Period, as the case may be, under the Insider Letter and any other applicable agreement between the Sponsor and the Company and to any transferee thereafter.

"Piggyback Registration" shall have the meaning given in subsection 2.2.1.

"Private Placement Lock-up Period" shall mean, with respect to Private Placement Warrants that are held by the initial purchasers of such Private Placement Warrants or their Permitted Transferees, and any of the shares of Common Stock issued or issuable upon the exercise or conversion of the Private Placement Warrants and that are held by the initial purchasers of the Private Placement Warrants or their Permitted Transferees, the period ending 30 days after the completion of the Business Combination.

"*Private Placement Warrants*" shall mean the 7,000,000 warrants purchased on a private placement occurring simultaneously with the closing of the Company's initial public offering.

"*Prospectus*" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Registrable Security" shall mean (a) the shares of Common Stock, (b) the Private Placement Warrants (including any shares of Common Stock issued or issuable upon the exercise of any such Private Placement Warrants), (c) any shares of Common Stock issuable upon the exercise, conversion or exchange of Common Stock Equivalents, and (d) any other equity security of the Company issued or issuable with respect to any such shares of Common Stock or Common Stock Equivalents by way of a share dividend or share split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates or book entries credits for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no volume, manner of sale or other restrictions or limitations); or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"*Registration*" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

*"Registration Expenses"* shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:

(A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Common Stock is then listed;

(B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);

(C) printing, messenger, telephone and delivery expenses;

(D) reasonable fees and disbursements of counsel for the Company;

(E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and

(F) reasonable fees and expenses of one legal counsel selected by the holders of a majority-in-interest of the Registrable Securities to be registered for offer and sale in the applicable Registration.

"*Registration Statement*" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.

"*Requesting Holder*" shall have the meaning given in <u>subsection 2.1.1</u>.

"Restricted Securities" shall have the meaning given in subsection 3.6.1.

"Securities Act" shall mean the Securities Act of 1933, as amended from time to time.

"Sponsor" shall have the meaning given in the Recitals hereto.

"Transactions" shall have the meaning set forth in the Recitals.

*"Underwriter"* shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.

*"Underwritten Registration"* or *"Underwritten Offering"* shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

"Units" shall mean Surviving Company Membership Units as defined in the Merger Agreement.

### ARTICLE II REGISTRATIONS

# 2.1 Demand Registration.

## 2.1.1 Request for Registration.

(a) Subject to the provisions of subsection 2.1.4, subsection 2.1.6 and Section 2.4, at any time and from time to time on or after the date the Company consummates the initial Business Combination, either (i) one or more Holders (other than the Sponsor or its affiliates or transferees) or (ii) the Sponsor or its affiliates or transferees, in either case of clause (i) or (ii) representing Registrable Securities with a total offering price reasonably expected to exceed, in the aggregate, the Minimum Demand Threshold, may make a written demand for Registration of all or part of their Registrable Securities, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand, a "Demand Registration" and such persons making such written demand, the "Demanding Holders"). The Company shall, within ten (10) days of the Company's receipt of the Demand Registration, notify, in writing, all other Holders of Registrable Securities of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "*Requesting* Holder") shall so notify the Company, in writing, within five (5) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s) to the Company, such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall (i) file a Registration Statement in respect of all Registrable Securities requested by the Demanding Holders and Requesting Holder(s) pursuant such Demand Registration, not more than forty five (45) days immediately after the Company's receipt of the Demand Registration, and (ii) shall effect the registration thereof as soon as practicable thereafter. Under no circumstances shall the Company be obligated to effect more than an (x) aggregate of three (3) Registrations pursuant to a Demand Registration initiated by one or more Holders (other than the Sponsor or its affiliates or transferees) and (y) an aggregate of three (3) Registrations pursuant to a Demand Registration initiated by the Sponsor or its affiliates or transferees, in each case under this subsection 2.1.1 with respect to any or all Registrable Securities; provided, however, that a Registration shall not be counted for such purposes unless a Form S-1 or any similar long-form registration statement that may be available at such time ("Form S-1") has become effective and all of the Registrable Securities requested by the Requesting Holders to be registered on behalf of the Requesting Holders in such Form S-1 Registration have been sold, in accordance with Section 3.1 of this Agreement. For the avoidance of doubt, each of (i) the holders of a majority-in-interest of the Registrable Securities held by the Holders and (ii) the Sponsor shall be permitted to exercise a Demand Registration Statement pursuant to this subsection 2.1.1 with respect to their Registrable Securities.

2.1.2 Effective Registration. Notwithstanding the provisions of subsection 2.1.1 above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (i) the Registration Statement filed with the Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission and (ii) the Company has complied with all of its obligations under this Agreement with respect thereto; provided, further, that if, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elects to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election; provided, further, that the Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration becomes effective or is subsequently terminated.

2.1.3 <u>Underwritten Offering</u>. Subject to the provisions of <u>subsection 2.1.4</u>, <u>subsection 2.1.6</u> and <u>Section 2.4</u> hereof, if a majority-in-interest of the Demanding Holders so elect and such Demanding Holders advise the Company as part of its Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of each Demanding Holder and Requesting Holder to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein; provided that such Demanding Holder(s) (a) reasonably expect aggregate gross proceeds in excess of the Minimum Demand Threshold from such Underwritten Offerings (it being understood that the Company shall not be required to conduct more than two Underwritten Offerings where the expected aggregate proceeds are below \$25,000,000 but in excess of the Minimum Demand Threshold in any 12-month period) or (b) reasonably expects to sell all of the Registrable Securities held by such Holder in such Underwritten Offering but in no event less than \$5,000,000 in aggregate gross proceeds. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.1.3</u> shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by a majority-in-interest of the holders initiating the Demand Registration.

2.1.4 <u>Reduction of Underwritten Offering</u>. If the managing Underwriter or Underwriters in an Underwritten Registration pursuant to a Demand Registration, in good faith, advises the Company and the Requesting Holders in writing that the dollar amount or number of Registrable Securities that such Holders desire to sell, taken together with all other shares of Common Stock or other equity securities that the Company desires to sell and the shares of Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other shareholders who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders and Requesting Holders (pro rata based on the respective number of Registrable Securities that each Demanding Holder and Requesting Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Requesting Holders have requested be included in such Underwritten Registration) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the shares of Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.

2.1.5 Demand Registration Withdrawal. Any Demanding Holder or Requesting Holder shall have the right to withdraw from a Registration pursuant to such Demand Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Registration prior to (x) in the case of a Demand Registration not involving an Underwritten Offering, the effectiveness of the Registration Statement filed with the Commission with respect to the Registration of their Registrable Securities pursuant to such Demand Registration or (y) in the case of a Demand Registration involving an Underwritten Offering, the pricing of such Underwritten Offering; provided, however, that upon withdrawal by a majority-in-interest of the Demanding Holders initiating a Demand Registration, the Company shall cease all efforts to secure effectiveness of the applicable Registration Statement or complete the Underwritten Offering, as applicable. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration prior to its withdrawal under this <u>subsection 2.1.5</u>.

2.1.6 <u>Shelf Registration</u>. The Company shall file within 45 days of the Closing, and use commercially reasonable efforts to cause to be declared effective as soon as practicable thereafter, a Registration Statement for a Shelf Registration on Form S-1 (the "<u>Form S-1 Shelf</u>") or, if the Company is eligible to use a Registration Statement on Form S-3, a Shelf Registration on Form S-3 (the "<u>Form S-3 Shelf</u>" and together with the Form S-1 Shelf, each a "<u>Shelf</u>"), in each case, covering the resale of all the Registrable Securities (determined as of two business days prior to such filing) on a delayed or continuous basis. Such Shelf shall provide for the resale of the Registrable Securities included therein pursuant to any method or combination of methods legally available to, and requested by, any Holder named therein. Notwithstanding anything to the contrary herein, to the extent there is an active Shelf under this <u>subsection 2.1.6</u>, covering a Holder's or Holders' Registrable Securities, such Holder shall not have rights to make a Demand Registration with respect to <u>subsection 2.1.1</u>. Notwithstanding anything to the contrary herein, to the extent there is an active Shelf under this <u>subsection 2.1.6</u>, covering a Holder's or Holders' Registrable Securities, and such Holder or Holders qualify as Demanding Holders pursuant to subsection 2.1.1 and wish to request an Underwritten Offering from such Shelf, such Underwritten Offering shall follow the procedures of <u>subsection 2.1.4</u>, (including <u>subsection 2.1.3</u> and <u>subsection 2.1.4</u>) but such Underwritten Offering shall be made from the Shelf and shall count against the number of Demand Registrations that may be made pursuant to <u>subsection 2.1.1</u>.

2.1.7 <u>Holder Information Required for Participation in Underwritten Offering</u>. At least five (5) business days prior to the first anticipated filing date of a Registration Statement pursuant to this <u>Section 2</u>, the Company shall use reasonable best efforts to notify each Holder in writing (which may be by email) of the information reasonably necessary about the Holder to include such Holder's Registrable Securities in such Registration Statement. Notwithstanding anything else in this Agreement, the Company shall not be obligated to include such Holder's Registrable Securities to the extent the Company has not received such information, and received any other reasonably requested agreements or certificates, on or prior to the second (2<sup>nd</sup>) business day prior to the first anticipated filing date of a Registration Statement pursuant to this <u>Section 2</u>.

### 2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If, at any time on or after the date hereof, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of shareholders of the Company (or by the Company and by the shareholders of the Company including, without limitation, pursuant to Section 2.1 hereof), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing shareholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities as soon as practicable but not less than ten (10) days before the anticipated filing date of such Registration Statement, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration, a "*Piggyback Registration*"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its best efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Registration that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of the shares of Common Stock that the Company desires to sell, taken together with (i) the shares of Common Stock, if any, as to which Registrable Securities hereunder (ii) the Registrable Securities as to which registration has been requested pursuant to <u>Section 2.2</u> hereof, and (iii) the shares of Common Stock, if any, as to which Registration piggy-back registration rights of other shareholders of the Company, exceeds the Maximum Number of Securities, then:

(a) If the Registration is undertaken for the Company's account, the Company shall include in any such Registration (A) first, the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to <u>subsection 2.2.1</u> hereof, pro rata, based on the respective number of Registrable Securities that each Holder has so requested, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other shareholders of the Company, which can be sold without exceeding the Maximum Number of Securities; and

(b) If the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (A) first, the shares of Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to <u>subsection 2.2.1</u>, pro rata, based on the respective number of Registrable Securities that each Holder has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Holders have requested to be included in such Underwritten Registration, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the shares of Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.

2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection 2.2.3</u>.

2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.1</u> hereof.

2.3 <u>Registrations on Form S-3</u>. The Holders of Registrable Securities may at any time, and from time to time, request in writing that the Company, pursuant to Rule 415 under the Securities Act (or any successor rule promulgated thereafter by the Commission), register the resale of any or all of their Registrable Securities on Form S-3 or any similar short-form registration statement that may be available at such time ("Form S-3"); provided, however, that the Company shall not be obligated to effect such request through an Underwritten Offering. Within five (5) days of the Company's receipt of a written request from a Holder or Holders of Registrable Securities for a Registration on Form S-3, the Company shall promptly give written notice of the proposed Registration on Form S-3 to all other Holders of Registrable Securities, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in such Registration on Form S-3 shall so notify the Company, in writing, within ten (10) days after the receipt by the Holder of the notice from the Company. As soon as practicable thereafter, but not more than twelve (12) days after the Company's initial receipt of such written request for a Registration on Form S-3, the Company shall register all or such portion of such Holder's Registrable Securities as are specified in such written request, together with all or such portion of Registrable Securities of any other Holder or Holders joining in such request as are specified in the written notification given by such Holder or Holders; provided, however, that the Company shall not be obligated to effect any such Registration pursuant to Section 2.3 hereof if (i) a Form S-3 is not available for such offering; or (ii) the Holders of Registrable Securities, together with the Holders of any other equity securities of the Company entitled to inclusion in such Registration, propose to sell the Registrable Securities and such other equity securities (if any) at any aggregate price to the public of less than \$25,000,000.

2.4 <u>Restrictions on Registration Rights</u>. If (A) during the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company-initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to <u>subsection 2.1.1</u> and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (B) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (C) in the good faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating that in the good faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement. In such event, the Company shall have the right to defer such filing for a period of not more than thirty (30) days; <u>provided</u>, <u>however</u>, that the Company shall not defer its obligation in this manner more than once in any 12-month period.

2.5 <u>Lock-Up</u>. Notwithstanding anything to the contrary in this Agreement, the Company shall not be obligated to effect any Demand Registration or Piggyback Registration of (i) any shares of Common Stock subject to the Founder Shares Lock-Up Period prior to the Founder Shares Lock-Up Period applicable to such shares of Common Stock or (ii) any Private Placement Warrants during the Private Placement Lock-Up Period. Nothing in this Section 2.4 shall limit the Company's obligation to register all of the Registrable Securities, including such shares of Common Stock subject to the Founder Shares Lock-Up Period and Private Placement Warrants, on the Shelf Registration Statement pursuant to Section 2.4.

# ARTICLE III COMPANY PROCEDURES

3.1 <u>General Procedures</u>. If at any time on or after the date hereof the Company is required to effect the Registration of Registrable Securities, the Company shall use its best efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall, as expeditiously as possible:

3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;

3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by the majority-in-interest of the Holders with Registrable Securities registered on such Registration Statement or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;

3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;

3.1.4 prior to any public offering of Registrable Securities, use its best efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;

3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed;

3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;

3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;

3.1.8 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus furnish a copy thereof to each seller of such Registrable Securities or its counsel;

3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in <u>Section</u> <u>3.4</u> hereof;

3.1.10 permit a representative of the Holders, the Underwriters, if any, and any attorney or accountant retained by such Holders or Underwriter to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; <u>provided</u>, <u>however</u>, that such representatives or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;

3.1.11 obtain a "cold comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Registration, in customary form and covering such matters of the type customarily covered by "cold comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;

3.1.12 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders;

3.1.13 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;

3.1.14 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);

3.1.15 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$25,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and

3.1.16 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.

3.2 <u>Registration Expenses</u>. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "*Registration Expenses*," reasonable fees and expenses of legal counsel representing the Holders.

3.3 <u>Requirements for Participation in Underwritten Offerings</u>. No person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person (i) agrees to sell such person's securities on the basis provided in any underwriting arrangements approved by the Company and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.

3.4 <u>Suspension of Sales; Adverse Disclosure</u>. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until he, she or it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until he, she or it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time, but in no event more than thirty (30) days, determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this <u>Section 3.4</u>.

3.5 <u>Reporting Obligations</u>. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell the shares of Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

#### 3.6 Lock-Up Restrictions.

3.6.1 During the applicable Lock-Up Periods, none of the Holders shall offer, sell, contract to sell, pledge, grant any option to purchase, make any short sale or otherwise dispose of or distribute any shares of Common Stock that are subject to an applicable Lock-Up Period or any securities convertible into, exercisable for, exchangeable for or that represent the right to receive shares of Common Stock that are subject to an applicable Lock-Up Period, whether now owned or hereinafter acquired, that is owned directly by such Holder (including securities held as a custodian) or with respect to which such Holder has beneficial ownership within the rules and regulations of the Commission (such securities that are subject to an applicable Lock-Up Period, the "*Restricted Securities*"), other than any transfer to an affiliate of a Holder or to a Permitted Transferee, as applicable. The foregoing restriction is expressly agreed to preclude each Holder, as applicable, from engaging in any hedging or other transaction with respect to Restricted Securities even if such Restricted Securities would be disposed of by someone other than such Holder. Such prohibited hedging or other transactions include any short sale or any purchase, sale or grant of any right (including any put or call option) with respect to any of the Restricted Securities of the applicable Holder, or with respect to any security that includes, relates to, or derives any significant part of its value from such Restricted Securities.

3.6.2 Each Holder hereby represents and warrants that it now has and, except as contemplated by this <u>subsection 3.6.2</u> for the duration of the applicable Lock-Up Period, will have good and marketable title to its Restricted Securities, free and clear of all liens, encumbrances, and claims that could impact the ability of such Existing Holder to comply with the foregoing restrictions Each Existing Holder agrees and consents to the entry of stop transfer instructions with the Company's transfer agent and registrar against the transfer of any Restricted Securities during the applicable Lock-Up Period.

### ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

#### 4.1 Indemnification.

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and agents and each person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including attorneys' fees) caused by any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.

4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any untrue statement of material fact contained in the Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.

4.1.3 Any person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and only other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling person of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.

4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>subsection 4.1.5</u> were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this <u>subsection 4.1.5</u>. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this subsection 4.1.5 from any person who was not guilty of such fraudulent misrepresentation.

#### ARTICLE V MISCELLANEOUS

5.1 <u>Notices</u>. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, electronic mail, telecopy, telegram or facsimile. Each notice or communication that is mailed, delivered, or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third business day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, electronic mail, telecopy, telegram or facsimile, at such time as it is delivered to the addressee (with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: 1114 6th Ave, 41st Floor, New York, New York, 10036, Attention: Chief Financial Officer with a copy to Cooley LLP, Attention: Dave Peinsipp, Kristin Vanderpas, Garth Osterman and Peter Byrne, 101 California Street, 5<sup>th</sup> Floor, San Francisco, CA 94111, , and, if to any Holder, at such Holder's address or facsimile number as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this <u>Section 5.1</u>.

#### 5.2 Assignment; No Third Party Beneficiaries.

5.2.1 This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part.

5.2.2 Prior to the expiration of the Founder Shares Lock-up Period or the Private Placement Lock-up Period, as the case may be, the Sponsor may not assign or delegate the Sponsor's rights, duties or obligations under this Agreement, in whole or in part, except in connection with a transfer of Registrable Securities by the Sponsor to a Permitted Transferee, but only if such Permitted Transferee agrees to become bound by the transfer restrictions set forth in this Agreement and other applicable agreements. Any other Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, if (i) the transferee receives Registrable Securities that constitute at least 1% of the Company's Common Stock and/or Common Stock Equivalents, (ii) such transfer is not pursuant to Rule 144 under the Securities Act or a registration statement filed pursuant to this Agreement and (iii) the transferee agrees to become bound by the transferee agrees to become bound by the transferee agrees to become bound by the transfer restrictions set forth in this Agreement and (iii) the transfer is not pursuant to Rule 144 under the Securities Act or a registration statement filed pursuant to this Agreement and (iii) the transferee agrees to become bound by the transfer restrictions set forth in this Agreement and other applicable agreements.

5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the Holders, which shall include Permitted Transferees.

5.2.4 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and <u>Section 5.2</u> hereof.

5.2.5 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 5.1</u> shall be null and void.

5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced.

5.4 <u>Governing Law; Venue</u>. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THE AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN NEW YORK COUNTY IN THE STATE OF NEW YORK. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.

5.5 <u>Amendments and Modifications</u>. Upon the written consent of the Company and the Holders of at least a majority-ininterest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in his, her or its capacity as a holder of the shares of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.

5.6 <u>Other Registration Rights</u>. The Company represents and warrants that no person, other than a Holder of Registrable Securities or those certain investors that agreed on or about the date hereof to purchase shares of Common Stock in a transaction exempt from registration under the Securities Act pursuant to those certain Subscription Agreements dated on or about the date hereof, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

5.7 Term. This Agreement shall terminate upon the earlier of (i) the tenth anniversary of the date hereof or (ii) the date as of which (A) all of the Registrable Securities have been sold pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (B) the Holders of all Registrable Securities are permitted to sell the Registrable Securities without registration pursuant to Rule 144 (or any similar provision) under the Securities Act without limitation on the amount of securities sold or the manner of sale. The provisions of <u>Section 3.5</u> and Article IV shall survive any termination.

# [SIGNATURE PAGES FOLLOW]

WM TECHNOLOGY, INC., a Delaware corporation By: /s/ Christopher Beals Name: Christopher Beals Title: Chief Executive Officer

HOLDERS: SILVER SPIKE SPONSOR, LLC, a Delaware limited liability company By: /s/ Greg Gentile

Name: Greg Gentile Title: Manager

HOLDERS: CHRISTOPHER BEALS /s/ Christopher Beals (Signature)

HOLDERS: ARDEN LEE /s/ Arden Lee (Signature)

HOLDERS: BRIAN CAMIRE /s/ Brian Camire (Signature)

HOLDERS: JUANJO FEIJOO /s/ Juanjo Feijoo (Signature)

HOLDERS: JUSTIN DEAN /s/ Justin Dean (Signature)

HOLDERS: DOUG FRANCIS /s/ Doug Francis (Signature)

HOLDERS: GENCO INCENTIVES, LLC By: /s/ Doug Francis Name: Doug Francis Title: Managing Member

#### HOLDERS: GHOST MEDIA GROUP, LLC

By: /s/ Justin Hartfield

Name: Justin Hartfield Title: Manager

By: /s/ Doug Francis

Name: Doug Francis Title: Manager

HOLDERS: JUSTIN HARTFIELD /s/ Justin Hartfield (Signature)

HOLDERS: STEVEN JUNG /s/ Steven Jung (Signature)

HOLDERS: WM FOUNDERS LEGACY I, LLC By: /s/ Justin Hartfield Name: Justin Hartfield Title: Board Member

HOLDERS: WM FOUNDERS LEGACY I, LLC By: /s/ Doug Francis Name: Doug Francis Title: Member

HOLDERS: WM FOUNDERS LEGACY II, LLC By: /s/ Justin Hartfield Name: Justin Hartfield Title: Board Member

HOLDERS: WM FOUNDERS LEGACY II, LLC By: /s/ Doug Francis Name: Doug Francis

Title: Member

#### WM TECHNOLOGY, INC.

### INDEMNIFICATION AGREEMENT

This **INDEMNIFICATION AGREEMENT** (this "*Agreement*") is dated as of \_\_\_\_\_\_, 2021 and is between WM Technology, Inc., a Delaware corporation (the "*Company*"), and \_\_\_\_\_\_("*Indemnitee*").

### RECITALS

A. Indemnitee's service to the Company substantially benefits the Company.

**B.** Individuals are reluctant to serve as directors or officers of corporations or in certain other capacities unless they are provided with adequate protection through insurance or indemnification against the risks of claims and actions against them arising out of such service.

**C.** Indemnitee does not regard the protection currently provided by applicable law, the Company's governing documents and any insurance as adequate under the present circumstances, and Indemnitee may not be willing to serve as a director or officer without additional protection.

**D.** In order to induce Indemnitee to continue to provide services to the Company, it is reasonable, prudent and necessary for the Company to contractually obligate itself to indemnify, and to advance expenses on behalf of, Indemnitee as permitted by applicable law.

**E.** This Agreement is a supplement to and in furtherance of the indemnification provided in the Company's certificate of incorporation and bylaws, and any resolutions adopted pursuant thereto, and this Agreement shall not be deemed a substitute therefor, nor shall this Agreement be deemed to limit, diminish or abrogate any rights of Indemnitee thereunder.

### AGREEMENT

The parties agree as follows:

# **1. Definitions**.

(a) "Beneficial Owner" shall have the meaning given to such term in Rule 13d-3 under the Securities Exchange Act of 1934, as amended; provided, however, that "Beneficial Owner" shall exclude any Person otherwise becoming a Beneficial Owner solely by reason of (i) the stockholders of the Company approving a merger of the Company with another Person, or entering into tender or support agreements relating thereto, provided such merger was approved by the Company's board of directors, or (ii) the Company's board of directors approving a sale of securities by the Company to such Person.

(b) A "*Change in Control*" shall be deemed to occur upon the earliest to occur after the date of this Agreement of any of the following events:

(i) *Acquisition of Stock by Third Party.* Any Person (as defined below) becomes the Beneficial Owner, directly or indirectly, of securities of the Company representing 50% or more of the combined voting power of the Company's then outstanding securities;

1

(ii) Change in Board Composition. During any period of two consecutive years (not including any period prior to the execution of this Agreement), individuals who at the beginning of such period constituted the Company's board of directors and any Approved Directors cease for any reason to constitute at least a majority of the members of the Company's board of directors. "Approved Directors" means new directors (other than a director designated by a person who has entered into an agreement with the Company to effect a transaction described in Sections 1(b)(i), 1(b)(iii) or 1(b)(iv)) whose election or nomination by the board of directors (or, if applicable, by the Company's stockholders) was approved by a vote of at least two thirds of the directors then still in office who either were directors at the beginning of such two-year period or whose election or nomination for election was previously so approved;

(iii) *Corporate Transactions.* The effective date of a merger or consolidation of the Company with any other entity, other than a merger or consolidation that would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity) more than 50% of the combined voting power of the voting securities of the surviving entity outstanding immediately after such merger or consolidation and with the power to elect a majority of the board of directors or other governing body of such surviving entity; or

(iv) *Liquidation*. The approval by the Company's board of directors of a complete liquidation or the dissolution of the Company or an agreement for the sale, lease or disposition by the Company of all or substantially all of the Company's assets; or

(v) *Other Events.* Any other event of a nature that would be required to be reported in response to Item 6(e) of Schedule 14A of Regulation 14A (or in response to any similar item on any similar schedule or form) promulgated under the Securities Exchange Act of 1934, as amended, whether or not the Company is then subject to such reporting requirement.

(c) *"Corporate Status"* describes the status of a person who is or was a director, trustee, general partner, managing member, officer, employee, agent or fiduciary of the Company or any other Enterprise.

(d) *"DGCL"* means the General Corporation Law of the State of Delaware.

(e) *"Disinterested Director"* means a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.

(f) *"Enterprise"* means the Company and any other corporation, partnership, limited liability company, joint venture, trust, employee benefit plan or other enterprise of which Indemnitee is or was serving at the request of the Company as a director, trustee, general partner, managing member, officer, employee, agent or fiduciary.

(g) "*Expenses*" include all reasonable and actually incurred attorneys' fees, retainers, court costs, transcript costs, fees and costs of experts, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, and all other disbursements or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be a witness in, or otherwise participating in, a Proceeding. Expenses also include (i) Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the premium, security for, and other costs relating to any cost bond, supersede as bond or other appeal bond or their equivalent, and (ii) for purposes of Section 10(d), Expenses incurred by Indemnitee in connection with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement or under any directors' and officers' liability insurance policies maintained by the Company. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.

2

(h) "Independent Counsel" means a law firm, or a partner or member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent (i) the Company, any Enterprise or Indemnitee in any matter material to any such party (other than as Independent Counsel with respect to matters concerning Indemnitee under this Agreement, or other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term Independent Counsel shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement.

(i) "*Person*" shall have the meaning set forth in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, as amended; *provided*, *however*, that Person shall exclude (i) the Company, (ii) any trustee or other fiduciary holding securities under an employee benefit plan of the Company, and (iii) any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.

(j) "Proceeding" means any threatened, pending or completed action, suit, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or proceeding, whether brought in the right of the Company or otherwise and whether of a civil, criminal, administrative or investigative nature, whether formal or informal, including any appeal therefrom and including without limitation any such Proceeding pending as of the date of this Agreement, in which Indemnitee was, is or will be involved as a party, a potential party, a non-party witness or otherwise by reason of (i) the fact that Indemnitee is or was a director or officer of the Company, (ii) any action taken by Indemnitee or any action or inaction on Indemnitee's part while acting as a director or officer of the Company, or (iii) the fact that he or she is or was serving at the request of the Company as a director, trustee, general partner, managing member, officer, employee, agent or fiduciary of the Company or any other Enterprise, in each case whether or not serving in such capacity at the time any liability or Expense is incurred for which indemnification or advancement of expenses can be provided under this Agreement.

(k) "to the fullest extent permitted by applicable law" means to the fullest extent permitted by all applicable laws, including without limitation: (i) the fullest extent permitted by DGCL as of the date of this Agreement and (ii) the fullest extent authorized or permitted by any amendments to or replacements of the DGCL adopted after the date of this Agreement that increase the extent to which a corporation may indemnify its officers and directors.

(I) In connection with any Proceeding relating to an employee benefit plan: references to "*fines*" shall include any excise taxes assessed on a person with respect to any employee benefit plan; references to "*serving at the request of the Company*" shall include any service as a director, officer, employee or agent of the Company which imposes duties on, or involves services by, such director, officer, employee or agent with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner he or she reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "*not opposed to the best interests of the Company*" as referred to in this Agreement.

2. Indemnity in Third-Party Proceedings. The Company shall indemnify Indemnitee in accordance with the provisions of this Section 2 if Indemnitee is, or is threatened to be made, a party to or witness or other participant in any Proceeding, other than a Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 2, Indemnitee shall be indemnified to the fullest extent permitted by applicable law against all Expenses, judgments, fines and amounts paid in settlement actually and reasonably incurred by Indemnitee or on his or her behalf in connection with such Proceeding or any claim, issue or matter therein.

**3. Indemnity in Proceedings by or in the Right of the Company**. The Company shall indemnify Indemnitee in accordance with the provisions of this Section 3 if Indemnitee is, or is threatened to be made, a party to or a witness or other participant in any Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 3, Indemnitee shall be indemnified to the fullest extent permitted by applicable law against all Expenses incurred by Indemnitee or on his or her behalf in connection with such Proceeding or any claim, issue or matter therein.

4. Indemnification for Expenses of a Party Who is Wholly or Partly Successful. Notwithstanding any other provision of this Agreement, in circumstances where indemnification is not available under Section 2 or 3, as the case may be, to the fullest extent permitted by law and to the extent that Indemnitee is a party to, and is successful (on the merits or otherwise) in defense of, any Proceeding or any claim, issue or matter therein, the Company shall indemnify Indemnitee against all Expenses incurred by Indemnitee or on Indemnitee's behalf in connection therewith. For purposes of this Section 4, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

**5. Exclusions**. Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnity in connection with any Proceeding (or any part of any Proceeding):

(a) for which payment has actually been made to or on behalf of Indemnitee under any statute, insurance policy, indemnity provision, vote or otherwise, except with respect to any excess beyond the amount paid;

(b) for an accounting or disgorgement of profits pursuant to Section 16(b) of the Securities Exchange Act of 1934, as amended, or similar provisions of federal, state or local statutory law or common law, if Indemnitee is held liable therefor (including pursuant to any settlement arrangements);

(c) for any reimbursement of the Company by Indemnitee of any bonus or other incentive-based or equitybased compensation or of any profits realized by Indemnitee from the sale of securities of the Company, as required in each case under the Securities Exchange Act of 1934, as amended (including any such reimbursements that arise from an accounting restatement of the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 (the "*Sarbanes-Oxley Act*"), or the payment to the Company of profits arising from the purchase and sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act), if Indemnitee is held liable therefor (including pursuant to any settlement arrangements);

(d) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees, agents or other indemnitees, unless (i) the Company's board of directors authorized the Proceeding (or the relevant part of the Proceeding) prior to its initiation, (ii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law, (iii) otherwise authorized in Section 10(d) or (iv) otherwise required by applicable law; provided, for the avoidance of doubt, Indemnitee shall not be deemed for purposes of this paragraph, to have initiated any Proceeding (or any part of a Proceeding) by reason of (i) having asserted any affirmative defenses in connection with a claim not initiated by Indemnitee or (ii) having made any counterclaim (whether permissive or mandatory) in connection with any claim not initiated by Indemnitee; or

(e) if prohibited by the DGCL or other applicable law.

6. Advances of Expenses. The Company shall advance the Expenses incurred by Indemnitee in connection with any Proceeding prior to its final disposition, and such advancement shall be made as soon as reasonably practicable, but in any event no later than 10 days, after the receipt by the Company of a written statement or statements requesting such advances from time to time (which shall include invoices received by Indemnitee in connection with such Expenses but, in the case of invoices in connection with legal services, any references to legal work performed or to expenditure made that would cause Indemnitee to waive any privilege accorded by applicable law shall not be included with the invoice). Advances shall be unsecured and interest free and made without regard to Indemnitee's ability to repay such advances. Indemnitee hereby undertakes to repay any advance to the extent that it is ultimately determined that Indemnitee is not entitled to be indemnified by the Company, *except*, with respect to advances of expenses made pursuant to Section 10(c), in which case Indemnitee makes the undertaking provided in Section 10(c). This Section 6 shall not apply to the extent advancement is prohibited by law and shall not apply to any Proceeding (or any part of any Proceeding) referenced in Section 5(b) or 5(c) prior to a determination that Indemnitee is not entitled to be indemnified by the Company.

# 7. Procedures for Notification and Defense of Claim.

(a) Indemnitee shall notify the Company in writing of any matter with respect to which Indemnitee intends to seek indemnification or advancement of Expenses as soon as reasonably practicable following the receipt by Indemnitee of notice thereof. The written notification to the Company shall include, in reasonable detail, a description of the nature of the Proceeding and the facts underlying the Proceeding. The failure by Indemnitee to notify the Company will not relieve the Company from any liability that it may have to Indemnitee hereunder or otherwise than under this Agreement, and any delay in so notifying the Company shall not constitute a waiver by Indemnitee of any rights, except to the extent that such failure or delay materially prejudices the Company.

(b) If, at the time of the receipt of a notice of a Proceeding pursuant to the terms hereof, the Company has directors' and officers' liability insurance in effect that may be applicable to the Proceeding, the Company shall give prompt notice of the commencement of the Proceeding to the insurers in accordance with the procedures set forth in the applicable policies. The Company shall thereafter take all commercially reasonable action to cause such insurers to pay, on behalf of Indemnitee, all amounts payable as a result of such Proceeding in accordance with the terms of such policies.

(c) In the event the Company may be obligated to make any indemnity in connection with a Proceeding, the Company shall be entitled to assume the defense of such Proceeding with counsel approved by Indemnitee, which approval shall not be unreasonably withheld, conditioned or delayed, upon the delivery to Indemnitee of written notice of its election to do so. After delivery of such notice, approval of such counsel by Indemnitee and the retention of such counsel by the Company, the Company will not be liable to Indemnitee for any fees or expenses of counsel subsequently incurred by Indemnitee with respect to the same Proceeding. Notwithstanding the Company's assumption of the defense of any such Proceeding, the Company shall be obligated to pay the fees and expenses of Indemnitee's separate counsel to the extent (i) the employment of separate counsel by Indemnitee is authorized by the Company, (ii) counsel for the Company shall have reasonably concluded that there is a conflict of interest between the Company and Indemnitee in the conduct of any such defense such that Indemnitee needs to be separately represented, (iii) the Company is not financially or legally able to perform its indemnification obligations, or (iv) the Company shall not have retained, or shall not continue to retain, counsel to defend such Proceeding. Regardless of any provision in this Agreement, Indemnitee shall have the right to employ counsel in any Proceeding at Indemnitee's personal expense. The Company shall not be entitled, without the consent of Indemnitee, to assume the defense of any claim brought by or in the right of the Company.

(d) Indemnitee shall give the Company such information and cooperation in connection with the Proceeding as may be reasonably appropriate.

(e) The Company shall not be liable to indemnify Indemnitee for any settlement of any Proceeding (or any part thereof) effected without the Company's prior written consent, which shall not be unreasonably withheld, conditioned or delayed. The Company acknowledges that a settlement or other disposition short of final judgment may be successful if it permits a party to avoid expense, delay, distraction, disruption and uncertainty. In the event that any action, claim or proceeding to which Indemnitee is a party is resolved in a settlement to which the Company has given its prior written consent, such settlement shall be treated as a success on the merits in the settled action, suit or proceeding.

(f) The Company shall not settle any Proceeding (or any part thereof) in a manner that imposes any penalty or liability on Indemnitee not paid by the Company without Indemnitee's prior written consent, which shall not be unreasonably withheld, conditioned or delayed.

### 8. Procedures upon Application for Indemnification.

(a) To obtain indemnification, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and as is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification following the final disposition of the Proceeding. Any delay in providing the request will not relieve the Company from its obligations under this Agreement, except to the extent such failure is prejudicial.

Upon written request by Indemnitee for indemnification pursuant to Section 8(a), a determination with **(b)** respect to Indemnitee's entitlement thereto shall be made no later than 30 days after the Company's receipt of Indemnitee's written request for indemnification as follows, provided that a Change in Control shall not have occurred: (i) by a majority vote of the Disinterested Directors, even though less than a quorum of the Company's board of directors; (ii) by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum of the Company's board of directors; (iii) if there are no such Disinterested Directors or, if a majority of Disinterested Directors so direct, by Independent Counsel in a written opinion to the Company's board of directors, a copy of which shall be delivered to Indemnitee; or (iv) if so directed by the Company's board of directors, by the stockholders of the Company. If a Change in Control shall have occurred, a determination with respect to Indemnitee's entitlement to indemnification shall be made by Independent Counsel in a written opinion to the Company's board of directors, a copy of which shall be delivered to Indemnitee. If the Company does not deliver a determination that Indemnitee is not entitled to indemnification within 30 days after the Company's receipt of Indemnitee's written request for indemnification, then the requisite determination of entitlement to indemnification shall be deemed to have been made and Indemnitee shall be entitled to such indemnification absent a prohibition of such indemnification under applicable law. If it is determined that Indemnitee is entitled to indemnification, payment to Indemnitee shall be made within ten days after such determination. Indemnitee shall cooperate with the person, persons or entity making the determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information that is not privileged or otherwise protected from disclosure and that is reasonably available to Indemnitee and reasonably necessary to such determination. Any costs or expenses (including attorneys' fees and disbursements) actually and reasonably incurred by Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company, to the extent permitted by applicable law.

In the event the determination of entitlement to indemnification is to be made by Independent Counsel **(c)** pursuant to Section 8(b), the Independent Counsel shall be selected as provided in this Section 8(c). If a Change in Control shall not have occurred, the Independent Counsel shall be selected by the Company's board of directors, and the Company shall give written notice to Indemnitee advising him or her of the identity of the Independent Counsel so selected. If a Change in Control shall have occurred, the Independent Counsel shall be selected by Indemnitee (unless Indemnitee shall request that such selection be made by the Company's board of directors, in which event the preceding sentence shall apply), and Indemnitee shall give written notice to the Company advising it of the identity of the Independent Counsel so selected. In either event, Indemnitee or the Company, as the case may be, may, within ten days after such written notice of selection shall have been given, deliver to the Company or to Indemnitee, as the case may be, a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 1, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected shall act as Independent Counsel. If such written objection is so made and substantiated, the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or a court has determined that such objection is without merit. If, within 20 days after the later of (i) submission by Indemnitee of a written request for indemnification pursuant to Section 8(a) and (ii) the final disposition of the Proceeding, the parties have not agreed upon an Independent Counsel, either the Company or Indemnitee may petition a court of competent jurisdiction for resolution of any objection that shall have been made by the Company or Indemnitee to the other's selection of Independent Counsel and for the appointment as Independent Counsel of a person selected by the court or by such other person as the court shall designate, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel under Section 8(b). Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 10(a), the Independent Counsel shall be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).

(d) The Company shall pay the reasonable fees and expenses of any Independent Counsel and to fully indemnify such counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.

#### 9. **Presumptions and Effect of Certain Proceedings**.

(a) In making a determination with respect to entitlement to indemnification hereunder, the person, persons or entity making such determination shall, to the fullest extent not prohibited by law, presume that Indemnitee is entitled to indemnification under this Agreement, and the Company shall, to the fullest extent not prohibited by law, have the burden of proof to overcome that presumption by clear and convincing evidence.

(b) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of *nolo contendere* or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner that he or she reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that his or her conduct was unlawful.

(c) For purposes of any determination of good faith, Indemnitee shall be deemed to have acted in good faith to the extent Indemnitee relied in good faith on (i) the records or books of account of the Enterprise, including financial statements, (ii) information supplied to Indemnitee by the officers of the Enterprise in the course of their duties, (iii) the advice of legal counsel for the Enterprise or its board of directors or counsel selected by any committee of the board of directors or (iv) information or records given or reports made to the Enterprise by an independent certified public accountant, an appraiser, investment banker or other expert selected with reasonable care by the Enterprise or its board of directors or any committee of the board of directors. The provisions of this Section 9(c) shall not be deemed to be exclusive or to limit in any way the other circumstances in which Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement. Whether or not the foregoing provisions of this Section 9(c) are satisfied, it shall in any event be presumed that Indemnitee has at all times acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Company. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

(d) Neither the knowledge, actions nor failure to act of any other director, officer, agent or employee of the Enterprise shall be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement.

(e) The Company acknowledges that a settlement or other disposition short of final judgment may be successful if it permits a party to avoid expense, delay, distraction, disruption and uncertainty. In the event that any Proceeding to which Indemnitee is a party is resolved in any manner other than by adverse judgment against Indemnitee (including, without limitation, settlement of such action, claim or proceeding with or without payment of money or other consideration) it shall be presumed that Indemnitee has been successful on the merits or otherwise in such Proceeding. Anyone seeking to overcome this presumption shall have the burden of proof and the burden of persuasion by clear and convincing evidence.

## **10.** Remedies of Indemnitee.

Subject to Section 10(e), in the event that (i) a determination is made pursuant to Section 8(b) that (a) Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses is not timely made pursuant to Section 6 or 10(d), (iii) no determination of entitlement to indemnification shall have been made pursuant to Section 8 within 30 days after the later of the receipt by the Company of the request for indemnification or the final disposition of the Proceeding, (iv) payment of indemnification pursuant to this Agreement is not made (A) within 10 days after a determination has been made that Indemnitee is entitled to indemnification or (B) with respect to indemnification pursuant to Sections 4, 5 and 10(d), within 30 days after receipt by the Company of a written request therefor, or (v) the Company or any other person or entity takes or threatens to take any action to declare this Agreement void or unenforceable, or institutes any litigation or other action or proceeding designed to deny, or to recover from, Indemnitee the benefits provided or intended to be provided to Indemnitee hereunder, Indemnitee shall be entitled to an adjudication by a court of competent jurisdiction of his or her entitlement to such indemnification or advancement of Expenses. Alternatively, Indemnitee, at his or her option, may seek an award in arbitration with respect to his or her entitlement to such indemnification or advancement of Expenses, to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules of the American Arbitration Association. Indemnitee shall commence such proceeding seeking an adjudication or an award in arbitration within 12 months following the date on which Indemnitee first has the right to commence such proceeding pursuant to this Section 10(a); provided, however, that the foregoing clause shall not apply in respect of a proceeding brought by Indemnitee to enforce his or her rights under Section 4. The Company shall not oppose Indemnitee's right to seek any such adjudication or award in arbitration in accordance with this Agreement.

(b) Neither (i) the failure of the Company, its board of directors, any committee or subgroup of the board of directors, Independent Counsel or stockholders to have made a determination that indemnification of Indemnitee is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor (ii) an actual determination by the Company, its board of directors, any committee or subgroup of the board of directors, Independent Counsel or stockholders that Indemnitee has not met the applicable standard of conduct, shall create a presumption that Indemnitee has or has not met the applicable standard of conduct, shall create a presumption that Indemnitee has or has not met the applicable standard of conduct, shall have been made pursuant to Section 8 that Indemnitee is not entitled to indemnification, any judicial proceeding or arbitration commenced pursuant to this Section 10 shall be conducted in all respects as a *de novo* trial, or arbitration commenced pursuant to this Section 10, the Company shall, to the fullest extent not prohibited by law, have the burden of proving Indemnitee is not entitled to indemnification or advancement of Expenses, as the case may be, and the burden of proof shall be by clear and convincing evidence.

(c) To the fullest extent not prohibited by law, the Company shall be precluded from asserting in any judicial proceeding or arbitration commenced pursuant to this Section 10 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court or before any such arbitrator that the Company is bound by all the provisions of this Agreement. If a determination shall have been made pursuant to Section 10 that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding or arbitration commenced pursuant to this Section 10, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statements not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law.

(d) To the extent not prohibited by law, the Company shall indemnify Indemnitee against all Expenses incurred by Indemnitee in connection with any action for indemnification or advancement of Expenses from the Company under this Agreement, any other agreement, the Company's certificate of incorporation or bylaws or under any directors' and officers' liability insurance policies maintained by the Company to the extent Indemnitee is successful in such action, and, if requested by Indemnitee, shall (as soon as reasonably practicable, but in any event no later than 30 days, after receipt by the Company of a written request therefor) advance such Expenses to Indemnitee, subject to the provisions of Section 6. Indemnitee hereby undertakes to repay such advances to the extent the Indemnitee is ultimately unsuccessful in such action or arbitration.

**11. Contribution**. To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee, the Company, in lieu of indemnifying Indemnitee, shall contribute to the amounts incurred by Indemnitee, whether for Expenses, judgments, fines or amounts paid or to be paid in settlement, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the events and transactions giving rise to such Proceeding; and (ii) the relative fault of Indemnitee and the Company (and its other directors, officers, employees and agents) in connection with such events and transactions.

12. Non-exclusivity. The rights of indemnification and to receive advancement of Expenses as provided by this Agreement shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Company's certificate of incorporation or bylaws, any agreement, a vote of stockholders or a resolution of directors, or otherwise. To the extent that a change in Delaware law, whether by statute or judicial decision, permits greater indemnification or advancement of Expenses than would be afforded currently under the Company's certificate of incorporation and bylaws and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change, subject to the restrictions expressly set forth herein or therein. Except as expressly set forth herein, no right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. Except as expressly set forth herein, the assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.

**Primary Responsibility**. The Company acknowledges that to the extent Indemnitee is serving as a director on the 13. Company's board of directors at the request or direction of a private equity or venture capital fund or other entity and/or certain of its affiliates (collectively, the "Secondary Indemnitors"), Indemnitee may have certain rights to indemnification and advancement of expenses provided by such Secondary Indemnitors. The Company agrees that, as between the Company and the Secondary Indemnitors, the Company is primarily responsible for amounts required to be indemnified or advanced under the Company's certificate of incorporation or bylaws or this Agreement and any obligation of the Secondary Indemnitors to provide indemnification or advancement for the same amounts is secondary to those Company obligations. To the extent not in contravention of any insurance policy or policies providing liability or other insurance for the Company or any director, trustee, general partner, managing member, officer, employee, agent or fiduciary of the Company or any other Enterprise, the Company waives any right of contribution or subrogation against the Secondary Indemnitors with respect to the liabilities for which the Company is primarily responsible under this Section 13. In the event of any payment by the Secondary Indemnitors of amounts otherwise required to be indemnified or advanced by the Company under the Company's certificate of incorporation or bylaws or this Agreement, the Secondary Indemnitors shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee for indemnification or advancement of expenses under the Company's certificate of incorporation or bylaws or this Agreement or, to the extent such subrogation is unavailable and contribution is found to be the applicable remedy, shall have a right of contribution with respect to the amounts paid. The Secondary Indemnitors are express third-party beneficiaries of the terms of this Section 13.

**14. No Duplication of Payments**. Subject to Section 13, the Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable hereunder (or for which advancement is provided hereunder) if and to the extent that Indemnitee has otherwise actually received payment for such amounts under any insurance policy, contract, agreement or otherwise.

**15. Insurance**. To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, trustees, general partners, managing members, officers, employees, agents or fiduciaries of the Company or any other Enterprise, Indemnitee shall be covered by such policy or policies to the same extent as the most favorably-insured persons under such policy or policies in a comparable position.

**16. Subrogation**. Subject to Section 13, in the event of any payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee, who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.

17. Services to the Company. Indemnitee agrees to serve as a director or officer of the Company or, at the request of the Company, as a director, trustee, general partner, managing member, officer, employee, agent or fiduciary of another Enterprise, for so long as Indemnitee is duly elected or appointed or until Indemnitee tenders his or her resignation or is removed from such position. Indemnitee may at any time and for any reason resign from such position (subject to any other contractual obligation or any obligation imposed by operation of law), in which event the Company shall have no obligation under this Agreement to continue Indemnitee in such position. This Agreement shall not be deemed an employment contract between the Company (or any of its subsidiaries or any Enterprise) and Indemnitee. Indemnitee may be discharged at any time for any reason, with or without cause, with or without notice, except as may be otherwise expressly provided in any executed, written employment contract between Indemnitee and the Company (or any of its subsidiaries or any Enterprise), any existing formal severance policies adopted by the Company's board of directors or, with respect to service as a director or officer of the Company, the Company's certificate of incorporation or bylaws or the DGCL. No such document shall be subject to any oral modification thereof.

**18. Duration**. This Agreement shall continue until and terminate upon the later of (a) five years after the date that Indemnitee shall have ceased to serve as a director or officer of the Company or as a director, trustee, general partner, managing member, officer, employee, agent or fiduciary of any other Enterprise, as applicable; or (b) one year after the final termination of any Proceeding, including any appeal, then pending in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any Proceeding commenced by Indemnitee pursuant to Section 10 relating thereto.

**19. Successors**. This Agreement shall be binding upon the Company and its successors and assigns, including any direct or indirect successor, by purchase, merger, consolidation or otherwise, to all or substantially all of the business or assets of the Company, and shall inure to the benefit of Indemnitee and Indemnitee's heirs, executors and administrators. Further, the Company shall require and cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company, by written agreement, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

**20.** Severability. Nothing in this Agreement is intended to require or shall be construed as requiring the Company to do or fail to do any act in violation of applicable law. The Company's inability, pursuant to court order or other applicable law, to perform its obligations under this Agreement shall not constitute a breach of this Agreement. If any provision or provisions of this Agreement shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (i) the validity, legality and enforceability of the remaining provisions of this Agreement (including without limitation, each portion of any section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and shall remain enforceable to the fullest extent permitted by law; (ii) such provision or provisions shall be deemed reformed to the extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (iii) to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion of any section of this Agreement containing any such provision for any section of this Agreement (including, without limitation, each portion of any section of this Agreement (including, without limitation, each portion of any section of this Agreement containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to give effect to the intent of the parties hereto; and (iii) to the fullest extent possible, the provision held to be invalid, illegal or unenforceable, shall be construed so as to give effect to the intent manifested thereby.

**21. Enforcement**. The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director or officer of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as a director or officer of the Company.

**22. Entire Agreement**. This Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof; *provided, however*, that this Agreement is a supplement to and in furtherance of the Company's certificate of incorporation and bylaws and applicable law.

**23. Modification and Waiver**. No supplement, modification or amendment to this Agreement shall be binding unless executed in writing by the parties hereto. No amendment, alteration or repeal of this Agreement shall adversely affect any right of Indemnitee under this Agreement in respect of any action taken or omitted by such Indemnitee in his or her Corporate Status prior to such amendment, alteration or repeal. No waiver of any of the provisions of this Agreement shall constitute or be deemed a waiver of any other provision of this Agreement nor shall any waiver constitute a continuing waiver.

**24. Notices.** All notices and other communications required or permitted hereunder shall be in writing and shall be mailed by registered or certified mail, postage prepaid, sent by facsimile or electronic mail or otherwise delivered by hand, messenger or courier service addressed:

(a) if to Indemnitee, to Indemnitee's address, facsimile number or electronic mail address as shown on the signature page of this Agreement or in the Company's records, as may be updated in accordance with the provisions hereof; or

(b) if to the Company, to 41 Discovery, Irvine, California 92618, Attention: Chief Executive Officer or at such other current address as the Company shall have furnished to Indemnitee.

Each such notice or other communication shall for all purposes of this Agreement be treated as effective or having been given (i) if delivered by hand, messenger or courier service, when delivered (or if sent *via* a nationally-recognized overnight courier service, freight prepaid, specifying next-business-day delivery, one business day after deposit with the courier), or (ii) if sent via mail, at the earlier of its receipt or five days after the same has been deposited in a regularly-maintained receptacle for the deposit of the United States mail, addressed and mailed as aforesaid, or (iii) if sent via facsimile, upon confirmation of facsimile transfer or, if sent via electronic mail, upon confirmation of delivery when directed to the relevant electronic mail address, if sent during normal business hours of the recipient, or if not sent during normal business hours of the recipient, then on the recipient's next business day.

25. Applicable Law and Consent to Jurisdiction. This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. Except with respect to any arbitration commenced by Indemnitee pursuant to Section 10(a), the Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Delaware Court of Chancery, and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court of Chancery for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) appoint, to the extent such party is not otherwise subject to service of process in the State of Delaware, The Corporation Service Company, Wilmington, Delaware 19808 as its agent in the State of Delaware as such party's agent for acceptance of legal process in connection with any such action or proceeding against such party with the same legal force and validity as if served upon such party personally within the State of Delaware, (iv) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court of Chancery, and (v) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Court of Chancery has been brought in an improper or inconvenient forum.

**26. Counterparts**. This Agreement may be executed in two or more counterparts, each of which shall for all purposes be deemed to be an original but all of which together shall constitute one and the same Agreement. This Agreement may also be executed and delivered by facsimile signature and in counterparts, each of which shall for all purposes be deemed to be an original but all of which together shall constitute one and the same Agreement. Only one such counterpart signed by the party against whom enforceability is sought needs to be produced to evidence the existence of this Agreement.

**27. Captions**. The headings of the paragraphs of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.

### (signature page follows)

The parties are signing this Indemnification Agreement as of the date stated in the introductory sentence.

WM	TECHNOLOGY,	INC.
----	-------------	------

WM	TECHNOLOGY, INC.
By:	
Name	e:
Title:	
14	

The parties are signing this Indemnification Agreement as of the date stated in the introductory sentence.

ī	[INDEMNITEE NAME]
	Address:
15	5

WM TECHNOLOGY, INC. 2021 Equity Incentive Plan

Adopted by the Board of Directors: June 16, 2021 Approved by the Stockholders: June 10, 2021

# TABLE OF CONTENTS

1.	General.	1
2.	Shares Subject to the Plan.	1
3.	ELIGIBILITY AND LIMITATIONS.	2
4.	Options and Stock Appreciation Rights.	3
5.	Awards Other Than Options and Stock Appreciation Rights.	7
6.	OpCo Units.	9
7.	Adjustments upon Changes in Common Stock; Other Corporate Events.	10
8.	Administration.	12
9.	TAX WITHHOLDING	15
10.	Miscellaneous.	16
11.	COVENANTS OF THE COMPANY.	19
12.	Additional Rules for Awards Subject to Section 409A.	19
13.	Severability.	23
14.	Termination of the Plan.	23
15.	Definitions.	24

#### Page

### 1. General.

(a) **Plan Purpose.** The Company, by means of the Plan, seeks to secure and retain the services of Employees, Directors and Consultants, to provide incentives for such persons to exert maximum efforts for the success of the Company and any Affiliate and to provide a means by which such persons may be given an opportunity to benefit from increases in value of the Common Stock through the granting of Awards.

(b) Available Awards. The Plan provides for the grant of the following Awards: (i) Incentive Stock Options; (ii) Nonstatutory Stock Options; (iii) SARs; (iv) Restricted Stock Awards; (v) RSU Awards; (vi) Performance Awards; and (vii) Other Awards.

(c) Adoption Date; Effective Date. The Plan will come into existence on the Adoption Date, but no Award may be granted prior to the Effective Date.

# 2. SHARES SUBJECT TO THE PLAN.

(a) Share Reserve. Subject to adjustment in accordance with Section 2(c) and any adjustments as necessary to implement any Capitalization Adjustments, the aggregate number of shares of Common Stock that may be issued pursuant to Awards will not exceed 19,209,796 shares of Common Stock (equal to eleven percent (11%) of the sum of (i) the number of shares of Common Stock outstanding as of the consummation of the transactions contemplated by the Merger Agreement and (ii) the number of shares of Common Stock underlying securities convertible into Common Stock). In addition, subject to any adjustments as necessary to implement any Capitalization Adjustments, such aggregate number of shares of Common Stock will automatically increase on January 1 of each year for a period of ten years commencing on January 1, 2022 and ending on (and including) January 1, 2031, in an amount equal to five percent (5%) of the total number of shares of the Company's capital stock outstanding on December 31 of the preceding year; provided, however that the Board may act prior to January 1<sup>st</sup> of a given year to provide that the increase for such year will be a lesser number of shares of Common Stock.

**(b) Aggregate Incentive Stock Option Limit.** Notwithstanding anything to the contrary in Section 2(a) and subject to any adjustments as necessary to implement any Capitalization Adjustments, the aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options is 57,629,388 shares (equal to three hundred percent (300%) of the total number of shares of Common Stock initially reserved for issuance under Section 2(a)).

# (c) Share Reserve Operation.

(i) Limit Applies to Common Stock Issued Pursuant to Awards. For clarity, the Share Reserve is a limit on the number of shares of Common Stock that may be issued pursuant to Awards and does not limit the granting of Awards, except that the Company will keep available at all times the number of shares of Common Stock reasonably required to satisfy its obligations to issue shares pursuant to such Awards. Shares may be issued in connection with a merger or acquisition as permitted by, as applicable, Nasdaq Listing Rule 5635(c), NYSE Listed Company Manual Section 303A.08, NYSE American Company Guide Section 711 or other applicable rule, and such issuance will not reduce the number of shares available for issuance under the Plan.

(ii) Actions that Do Not Constitute Issuance of Common Stock and Do Not Reduce Share Reserve. The following actions do not result in an issuance of shares under the Plan and accordingly do not reduce the number of shares subject to the Share Reserve and available for issuance under the Plan: (1) the expiration or termination of any portion of an Award without the shares covered by such portion of the Award having been issued; (2) the settlement of any portion of an Award in cash (*i.e.*, the Participant receives cash rather than Common Stock); (3) the withholding of shares that would otherwise be issued by the Company to satisfy the exercise, strike or purchase price of an Award; or (4) the withholding of shares that would otherwise be issued by the Company to satisfy a tax withholding obligation in connection with an Award.

(iii) **Reversion of Previously Issued Shares of Common Stock to Share Reserve.** The following shares of Common Stock previously issued pursuant to an Award and accordingly initially deducted from the Share Reserve will be added back to the Share Reserve and again become available for issuance under the Plan: (1) any shares that are forfeited back to or repurchased by the Company because of a failure to meet a contingency or condition required for the vesting of such shares; (2) any shares that are reacquired by the Company to satisfy the exercise, strike or purchase price of an Award; and (3) any shares that are reacquired by the Company to satisfy a tax withholding obligation in connection with an Award.

### **3.** Eligibility and Limitations.

(a) Eligible Award Recipients. Subject to the terms of the Plan, Employees, Directors and Consultants are eligible to receive Awards.

# (b) Specific Award Limitations.

(i) Limitations on Incentive Stock Option Recipients. Incentive Stock Options may be granted only to Employees of the Company or a "parent corporation" or "subsidiary corporation" thereof (as such terms are defined in Sections 424(e) and (f) of the Code).

(ii) Incentive Stock Option \$100,000 Limitation. To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and any Affiliates) exceeds \$100,000 (or such other limit established in the Code) or otherwise does not comply with the rules governing Incentive Stock Options, the Options or portions thereof that exceed such limit (according to the order in which they were granted) or otherwise do not comply with such rules will be treated as Nonstatutory Stock Options, notwithstanding any contrary provision of the applicable Option Agreement(s).

(iii) Limitations on Incentive Stock Options Granted to Ten Percent Stockholders. A Ten Percent Stockholder may not be granted an Incentive Stock Option unless (i) the exercise price of such Option is at least 110% of the Fair Market Value on the date of grant of such Option and (ii) the Option is not exercisable after the expiration of five years from the date of grant of such Option.

(iv) Limitations on Nonstatutory Stock Options and SARs. Nonstatutory Stock Options and SARs may not be granted to Employees, Directors and Consultants who are providing Continuous Service only to any "parent" of the Company (as such term is defined in Rule 405) unless the stock underlying such Awards is treated as "service recipient stock" under Section 409A because the Awards are granted pursuant to a corporate transaction (such as a spin off transaction) or unless such Awards otherwise comply with the distribution requirements of Section 409A.

(c) Aggregate Incentive Stock Option Limit. The aggregate maximum number of shares of Common Stock that may be issued pursuant to the exercise of Incentive Stock Options is the number of shares specified in Section 2(b).

(d) Non-Employee Director Compensation Limit. The aggregate value of all compensation granted or paid, as applicable, to any individual for service as a Non-Employee Director with respect to any calendar year, including Awards granted and cash fees paid by the Company to such Non-Employee Director, will not exceed (i) \$750,000 in total value or (ii) in the event such Non-Employee Director is first appointed or elected to the Board during such calendar year, \$1,000,000 in total value, in each case calculating the value of any equity awards based on the grant date fair value of such equity awards for financial reporting purposes. The limitations in this Section 3(d) shall apply commencing with the first calendar year that begins following the Effective Date.

#### 4. **OPTIONS AND STOCK APPRECIATION RIGHTS.**

Each Option and SAR will have such terms and conditions as determined by the Board. Each Option will be designated in writing as an Incentive Stock Option or Nonstatutory Stock Option at the time of grant; provided, however, that if an Option is not so designated, then such Option will be a Nonstatutory Stock Option, and the shares purchased upon exercise of each type of Option will be separately accounted for. Each SAR will be denominated in shares of Common Stock equivalents. The terms and conditions of separate Options and SARs need not be identical; provided, however, that each Option Agreement and SAR Agreement will conform (through incorporation of provisions hereof by reference in the Award Agreement or otherwise) to the substance of each of the following provisions:

(a) **Term.** Subject to Section 3(b) regarding Ten Percent Stockholders, no Option or SAR will be exercisable after the expiration of ten years from the date of grant of such Award or such shorter period specified in the Award Agreement.

(b) Exercise or Strike Price. Subject to Section 3(b) regarding Ten Percent Stockholders, the exercise or strike price of each Option or SAR will not be less than 100% of the Fair Market Value on the date of grant of such Award. Notwithstanding the foregoing, an Option or SAR may be granted with an exercise or strike price lower than 100% of the Fair Market Value on the date of grant of such Award if such Award is granted pursuant to an assumption of or substitution for another option or stock appreciation right pursuant to a Corporate Transaction and in a manner consistent with the provisions of Sections 409A and, if applicable, 424(a) of the Code.

(c) Exercise Procedure and Payment of Exercise Price for Options. In order to exercise an Option, the Participant must provide notice of exercise to the Plan Administrator in accordance with the procedures specified in the Option Agreement or otherwise provided by the Company. The Board has the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to utilize a particular method of payment. The exercise price of an Option may be paid, to the extent permitted by Applicable Law and as determined by the Board, by one or more of the following methods of payment to the extent set forth in the Option Agreement:

(i) by cash or check, bank draft or money order payable to the Company;

(ii) pursuant to a "cashless exercise" program developed under Regulation T as promulgated by the Federal Reserve Board that, prior to the issuance of the Common Stock subject to the Option, results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the exercise price to the Company from the sales proceeds;

(iii) by delivery to the Company (either by actual delivery or attestation) of shares of Common Stock that are already owned by the Participant free and clear of any liens, claims, encumbrances or security interests, with a Fair Market Value on the date of exercise that does not exceed the exercise price, provided that (1) at the time of exercise the Common Stock is publicly traded, (2) any remaining balance of the exercise price not satisfied by such delivery is paid by the Participant in cash or other permitted form of payment, (3) such delivery would not violate any Applicable Law or agreement restricting the redemption of the Common Stock, (4) any certificated shares are endorsed or accompanied by an executed assignment separate from certificate, and (5) such shares have been held by the Participant for any minimum period necessary to avoid adverse accounting treatment as a result of such delivery;

(iv) if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement pursuant to which the Company will reduce the number of shares of Common Stock issuable upon exercise by the largest whole number of shares with a Fair Market Value on the date of exercise that does not exceed the exercise price, provided that (1) such shares used to pay the exercise price will not be exercisable thereafter and (2) any remaining balance of the exercise price not satisfied by such net exercise is paid by the Participant in cash or other permitted form of payment; or

Law.

(v) in any other form of consideration that may be acceptable to the Board and permissible under Applicable

(d) Exercise Procedure and Payment of Appreciation Distribution for SARs. In order to exercise any SAR, the Participant must provide notice of exercise to the Plan Administrator in accordance with the SAR Agreement. The appreciation distribution payable to a Participant upon the exercise of a SAR will not be greater than an amount equal to the excess of (i) the aggregate Fair Market Value on the date of exercise of a number of shares of Common Stock equal to the number of Common Stock equivalents that are vested and being exercised under such SAR, over (ii) the strike price of such SAR. Such appreciation distribution may be paid to the Participant in the form of Common Stock or cash (or any combination of Common Stock and cash) or in any other form of payment, as determined by the Board and specified in the SAR Agreement.

(e) **Transferability.** Options and SARs may not be transferred to third party financial institutions for value. The Board may impose such additional limitations on the transferability of an Option or SAR as it determines. In the absence of any such determination by the Board, the following restrictions on the transferability of Options and SARs will apply, provided that except as explicitly provided herein, neither an Option nor a SAR may be transferred for consideration and *provided*, *further*, that if an Option is an Incentive Stock Option, such Option may be deemed to be a Nonstatutory Stock Option as a result of such transfer:

(i) **Restrictions on Transfer.** An Option or SAR will not be transferable, except by will or by the laws of descent and distribution, and will be exercisable during the lifetime of the Participant only by the Participant; provided, however, that the Board may permit transfer of an Option or SAR in a manner that is not prohibited by applicable tax and securities laws upon the Participant's request, including to a trust if the Participant is considered to be the sole beneficial owner of such trust (as determined under Section 671 of the Code and applicable state law) while such Option or SAR is held in such trust, provided that the Participant and the trustee enter into a transfer and other agreements required by the Company.

(ii) **Domestic Relations Orders.** Notwithstanding the foregoing, subject to the execution of transfer documentation in a format acceptable to the Company and subject to the approval of the Board or a duly authorized Officer, an Option or SAR may be transferred pursuant to a domestic relations order.

(f) Vesting. The Board may impose such restrictions on or conditions to the vesting and/or exercisability of an Option or SAR as determined by the Board. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, vesting of Options and SARs will cease upon termination of the Participant's Continuous Service.

(g) **Termination of Continuous Service for Cause.** Except as explicitly otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service is terminated for Cause, the Participant's Options and SARs will terminate and be forfeited immediately upon such termination of Continuous Service, and the Participant will be prohibited from exercising any portion (including any vested portion) of such Awards on and after the date of such termination of Continuous Service and the Participant will have no further right, title or interest in such forfeited Award, the shares of Common Stock subject to the forfeited Award, or any consideration in respect of the forfeited Award.

(h) **Post-Termination Exercise Period Following Termination of Continuous Service for Reasons Other than Cause.** Subject to Section 4(i), if a Participant's Continuous Service terminates for any reason other than for Cause, the Participant may exercise his or her Option or SAR to the extent vested, but only within the following period of time or, if applicable, such other period of time provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate; provided, however, that in no event may such Award be exercised after the expiration of its maximum term (as set forth in Section 4(a)):

(i) three months following the date of such termination if such termination is a termination without Cause (other than any termination due to the Participant's Disability or death);

(ii) 12 months following the date of such termination if such termination is due to the Participant's Disability;

(iii) 18 months following the date of such termination if such termination is due to the Participant's death; or

(iv) 18 months following the date of the Participant's death if such death occurs following the date of such termination but during the period such Award is otherwise exercisable (as provided in (i) or (ii) above).

Following the date of such termination, to the extent the Participant does not exercise such Award within the applicable Post-Termination Exercise Period (or, if earlier, prior to the expiration of the maximum term of such Award), such unexercised portion of the Award will terminate, and the Participant will have no further right, title or interest in the terminated Award, the shares of Common Stock subject to the terminated Award, or any consideration in respect of the terminated Award.

(i) Restrictions on Exercise; Extension of Exercisability. A Participant may not exercise an Option or SAR at any time that the issuance of shares of Common Stock upon such exercise would violate Applicable Law. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service terminates for any reason other than for Cause and, at any time during the last thirty days of the applicable Post-Termination Exercise Period: (i) the exercise of the Participant's Option or SAR would be prohibited solely because the issuance of shares of Common Stock upon such exercise would violate Applicable Law, or (ii) the immediate sale of any shares of Common Stock upon such exercise would violate the Company's Trading Policy, then the applicable Post-Termination Exercise Period will be extended to the last day of the calendar month that commences following the date the Award would otherwise expire, with an additional extension of the exercise period to the last day of the next calendar month to apply if any of the foregoing restrictions apply at any time during such extended exercise period, generally without limitation as to the maximum permitted number of extensions); provided, however, that in no event may such Award be exercised after the expiration of its maximum term (as set forth in Section 4(a)).

(j) Non-Exempt Employees. No Option or SAR, whether or not vested, granted to an Employee who is a nonexempt employee for purposes of the Fair Labor Standards Act of 1938, as amended, will be first exercisable for any shares of Common Stock until at least six months following the date of grant of such Award. Notwithstanding the foregoing, in accordance with the provisions of the Worker Economic Opportunity Act, any vested portion of such Award may be exercised earlier than six months following the date of grant of such Award in the event of (i) such Participant's death or Disability, (ii) a Corporate Transaction in which such Award is not assumed, continued or substituted, (iii) a Change in Control, or (iv) such Participant's retirement (as such term may be defined in the Award Agreement or another applicable agreement or, in the absence of any such definition, in accordance with the Company's then current employment policies and guidelines). This Section 4(j) is intended to operate so that any income derived by a non-exempt employee in connection with the exercise or vesting of an Option or SAR will be exempt from his or her regular rate of pay.

(k) Whole Shares. Options and SARs may be exercised only with respect to whole shares of Common Stock or their equivalents.

# 5. Awards Other Than Options and Stock Appreciation Rights.

(a) **Restricted Stock Awards and RSU Awards.** Each Restricted Stock Award and RSU Award will have such terms and conditions as determined by the Board; provided, however, that each Restricted Stock Award Agreement and RSU Award Agreement will conform (through incorporation of the provisions hereof by reference in the Award Agreement or otherwise) to the substance of each of the following provisions:

# (i) Form of Award.

(1) RSAs: To the extent consistent with the Company's Bylaws, at the Board's election, shares of Common Stock subject to a Restricted Stock Award may be (i) held in book entry form subject to the Company's instructions until such shares become vested or any other restrictions lapse, or (ii) evidenced by a certificate, which certificate will be held in such form and manner as determined by the Board. Unless otherwise determined by the Board, a Participant will have voting and other rights as a stockholder of the Company with respect to any shares subject to a Restricted Stock Award.

(2) RSUs: An RSU Award represents a Participant's right to be issued on a future date the number of shares of Common Stock that is equal to the number of restricted stock units subject to the RSU Award. As a holder of an RSU Award, a Participant is an unsecured creditor of the Company with respect to the Company's unfunded obligation, if any, to issue shares of Common Stock in settlement of such Award and nothing contained in the Plan or any RSU Agreement, and no action taken pursuant to its provisions, will create or be construed to create a trust of any kind or a fiduciary relationship between a Participant and the Company or an Affiliate or any other person. A Participant will not have voting or any other rights as a stockholder of the Company with respect to any RSU Award (unless and until shares are actually issued in settlement of a vested RSU Award).

# (ii) Consideration.

(1) RSA: A Restricted Stock Award may be granted in consideration for (A) cash or check, bank draft or money order payable to the Company, (B) past services to the Company or an Affiliate, or (C) any other form of consideration (including future services) as the Board may determine and permissible under Applicable Law.

(2) RSU: Unless otherwise determined by the Board at the time of grant, an RSU Award will be granted in consideration for the Participant's services to the Company or an Affiliate, such that the Participant will not be required to make any payment to the Company (other than such services) with respect to the grant or vesting of the RSU Award, or the issuance of any shares of Common Stock pursuant to the RSU Award. If, at the time of grant, the Board determines that any consideration must be paid by the Participant (in a form other than the Participant's services to the Company or an Affiliate) upon the issuance of any shares of Common Stock in settlement of the RSU Award, such consideration may be paid in any form of consideration as the Board may determine and permissible under Applicable Law.

(iii) Vesting. The Board may impose such restrictions on or conditions to the vesting of a Restricted Stock Award or RSU Award as determined by the Board. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, vesting of Restricted Stock Awards and RSU Awards will cease upon termination of the Participant's Continuous Service.

(iv) Termination of Continuous Service. Except as otherwise provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service terminates for any reason, (i) the Company may receive through a forfeiture condition or a repurchase right any or all of the shares of Common Stock held by the Participant under his or her Restricted Stock Award that have not vested as of the date of such termination as set forth in the Restricted Stock Award Agreement and (ii) any portion of his or her RSU Award that has not vested will be forfeited upon such termination and the Participant will have no further right, title or interest in the RSU Award, the shares of Common Stock issuable pursuant to the RSU Award, or any consideration in respect of the RSU Award.

(v) **Dividends and Dividend Equivalents.** Dividends or dividend equivalents may be paid or credited, as applicable, with respect to any shares of Common Stock subject to a Restricted Stock Award or RSU Award, as determined by the Board and specified in the Award Agreement.

(vi) Settlement of RSU Awards. An RSU Award may be settled by the issuance of shares of Common Stock or cash (or any combination thereof) or in any other form of payment, as determined by the Board and specified in the RSU Award Agreement. At the time of grant, the Board may determine to impose such restrictions or conditions that delay such delivery to a date following the vesting of the RSU Award.

(b) **Performance Awards**. With respect to any Performance Award, the length of any Performance Period, the Performance Goals to be achieved during the Performance Period, the other terms and conditions of such Award, and the measure of whether and to what degree such Performance Goals have been attained will be determined by the Board.

(c) Other Awards. Other forms of Awards valued in whole or in part by reference to, or otherwise based on, Common Stock, including the appreciation in value thereof (e.g., options or stock rights with an exercise price or strike price less than 100% of the Fair Market Value at the time of grant) may be granted either alone or in addition to Awards provided for under Section 4 and the preceding provisions of this Section 5. Subject to the provisions of the Plan, the Board will have sole and complete discretion to determine the persons to whom and the time or times at which such Other Awards will be granted, the number of shares of Common Stock (or the cash equivalent thereof) to be granted pursuant to such Other Awards and all other terms and conditions of such Other Awards.

### 6. OPCO UNITS.

(a) General. Awards may be granted under the Plan in the form of any class of limited liability company interests in WM OpCo, the entity through which the Company conducts its business and an entity that has elected to be treated as a partnership for federal income tax purposes ("*OpCo Units*") established pursuant to the OpCo LLC Agreement. Awards of OpCo Units shall be valued by reference to, or otherwise determined by reference to or based on, shares of Common Stock. OpCo Units awarded under the Plan may be (1) convertible, exchangeable or redeemable for other limited liability company interests in WM OpCo (including OpCo Units of a different class or series) or shares of Common Stock, or (2) valued by reference to the book value, fair value or performance of WM OpCo. Awards of OpCo Units may be intended to qualify as "profits interests" within the meaning of IRS Revenue Procedure 93-27, as clarified by IRS Revenue Procedure 2001-43, with respect to a Participant in the Plan who is rendering services to or for the benefit of WM OpCo, including its subsidiaries.

Share Calculations. For purposes of calculating the number of shares of Common Stock underlying an award of (b) OpCo Units relative to the total number of shares of Common Stock available for issuance under the Plan, the Committee shall establish in good faith the maximum number of shares of Common Stock to which a Participant receiving such award of OpCo Units may be entitled upon fulfillment of all applicable conditions set forth in the relevant award documentation, including vesting conditions, partnership capital account allocations, value accretion factors, conversion ratios, exchange ratios and other similar criteria. If and when any such conditions are no longer capable of being met, in whole or in part, the number of shares of Common Stock underlying such awards of OpCo Units shall be reduced accordingly by the Committee, and the number of shares of Common Stock shall be increased by one share of Common Stock for each share so reduced. Awards of OpCo Units may be granted either alone or in addition to other awards granted under the Plan. The Committee shall determine the eligible Participants to whom, and the time or times at which, awards of OpCo Units shall be made; the number of OpCo Units to be awarded; the price, if any, to be paid by the Participant for the acquisition of such OpCo Units (which may be less than the fair value of the OpCo Unit); and the restrictions and conditions applicable to such award of OpCo Units. Conditions may be based on continuing employment (or other service relationship), computation of financial metrics and/or achievement of pre-established performance goals and objectives, with related length of the service period for vesting, minimum or maximum performance thresholds, measurement procedures and length of the performance period to be established by the Committee at the time of grant, in its sole discretion. The Committee may allow awards of OpCo Units to be held through a limited partnership, or similar "look-through" entity, and the Committee may require such limited partnership or similar entity to impose restrictions on its partners or other beneficial owners that are not inconsistent with the provisions of this Section 6. The provisions of the grant of OpCo Units need not be the same with respect to each Participant.

(c) Dividends and Distributions. The Award Agreement or other award documentation in respect of an award of OpCo Units may provide that the recipient of OpCo Units shall be entitled to receive, currently or on a deferred or contingent basis, dividends or dividend equivalents with respect to the number of shares of Common Stock underlying the award or other distributions from WM OpCo prior to vesting (whether based on a period of time or based on attainment of specified performance conditions), as determined at the time of grant by the Committee, in its sole discretion, and the Committee may provide that such amounts (if any) shall be deemed to have been reinvested in additional shares of Common Stock or OpCo Units.

## 7. Adjustments upon Changes in Common Stock; Other Corporate Events.

(a) **Capitalization Adjustments**. In the event of a Capitalization Adjustment, the Board shall appropriately and proportionately adjust: (i) the class(es) and maximum number of shares of Common Stock subject to the Plan and the maximum number of shares by which the Share Reserve may annually increase pursuant to Section 2(a); (ii) the class(es) and maximum number of shares that may be issued pursuant to the exercise of Incentive Stock Options pursuant to Section 2(a); and (iii) the class(es) and number of securities and exercise price, strike price or purchase price of Common Stock subject to outstanding Awards. The Board shall make such adjustments in a matter that it deems equitable in its sole discretion, and its determination shall be final, binding and conclusive. Notwithstanding the foregoing, no fractional shares or rights for fractional shares of Common Stock shall be created in order to implement any Capitalization Adjustment. The Board shall determine an appropriate equivalent benefit, if any, for any fractional shares or rights to fractional shares that might be created by the adjustments referred to in the preceding provisions of this Section.

(b) Dissolution or Liquidation. Except as otherwise provided in the Award Agreement, in the event of a dissolution or liquidation of the Company, all outstanding Awards (other than Awards consisting of vested and outstanding shares of Common Stock not subject to a forfeiture condition or the Company's right of repurchase) will terminate immediately prior to the completion of such dissolution or liquidation, and the shares of Common Stock subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Award is providing Continuous Service, provided, however, that the Board may determine to cause some or all Awards to become fully vested, exercisable and/or no longer subject to repurchase or forfeiture (to the extent such Awards have not previously expired or terminated) before the dissolution or liquidation is completed but contingent on its completion.

(c) **Corporate Transaction.** The following provisions will apply to Awards in the event of a Corporate Transaction unless otherwise provided in the instrument evidencing the Award or any other written agreement between the Company or any Affiliate and the Participant or unless otherwise expressly provided by the Board at the time of grant of an Award.

(i) Awards May Be Assumed. In the event of a Corporate Transaction, any surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) may assume or continue any or all Awards outstanding under the Plan or may substitute similar awards for Awards outstanding under the Plan (including but not limited to, awards to acquire the same consideration paid to the stockholders of the Company pursuant to the Corporate Transaction), and any reacquisition or repurchase rights held by the Company in respect of Common Stock issued pursuant to Awards may be assigned by the Company to the successor of the Company (or the successor's parent company, if any), in connection with such Corporate Transaction. A surviving corporation or acquiring corporation (or its parent) may choose to assume or continue only a portion of an Award or substitute a similar award for only a portion of an Award, or may choose to assume or continue the Awards held by some, but not all Participants. The terms of any assumption, continuation or substitution will be set by the Board.

(ii) Awards Held by Current Participants. In the event of a Corporate Transaction in which the surviving corporation or acquiring corporation (or its parent company) does not assume or continue such outstanding Awards or substitute similar awards for such outstanding Awards, then with respect to Awards that have not been assumed, continued or substituted and that are held by Participants whose Continuous Service has not terminated prior to the effective time of the Corporate Transaction (referred to as the "Current Participants"), the vesting of such Awards (and, with respect to Options and Stock Appreciation Rights, the time when such Awards may be exercised) will be accelerated in full to a date prior to the effective time of such Corporate Transaction (contingent upon the effectiveness of the Corporate Transaction) as the Board determines (or, if the Board does not determine such a date, to the date that is five (5) days prior to the effective time of the Corporate Transaction), and such Awards will terminate for no consideration if not exercised (if applicable) at or prior to the effective time of the Corporate Transaction, and any reacquisition or repurchase rights held by the Company with respect to such Awards will lapse (contingent upon the effectiveness of the Corporate Transaction). With respect to the vesting of Performance Awards that will accelerate upon the occurrence of a Corporate Transaction pursuant to this subsection (ii) and that have multiple vesting levels depending on the level of performance, unless otherwise provided in the Award Agreement or unless otherwise provided by the Board, the vesting of such Performance Awards will accelerate at 100% of the target level upon the occurrence of the Corporate Transaction. With respect to the vesting of Awards that will accelerate upon the occurrence of a Corporate Transaction pursuant to this subsection (ii) and are settled in the form of a cash payment, such cash payment will be made no later than 30 days following the occurrence of the Corporate Transaction.

(iii) Awards Held by Persons other than Current Participants. In the event of a Corporate Transaction in which the surviving corporation or acquiring corporation (or its parent company) does not assume or continue such outstanding Awards or substitute similar awards for such outstanding Awards, then with respect to Awards that have not been assumed, continued or substituted and that are held by persons other than Current Participants, such Awards will terminate for no consideration if not exercised (if applicable) prior to the occurrence of the Corporate Transaction; provided, however, that any reacquisition or repurchase rights held by the Company with respect to such Awards will not terminate and may continue to be exercised notwithstanding the Corporate Transaction.

(iv) Payment for Awards in Lieu of Exercise. Notwithstanding the foregoing, in the event an Award will terminate if not exercised prior to the effective time of a Corporate Transaction, the Board may provide, in its sole discretion, that the holder of such Award may not exercise such Award but will receive a payment, in such form as may be determined by the Board, equal in value, at the effective time, to the excess, if any, of (1) the value of the property the Participant would have received upon the exercise of the Award (including, at the discretion of the Board, any unvested portion of such Award), over (2) any exercise price payable by such holder in connection with such exercise.

(d) **Appointment of Stockholder Representative.** As a condition to the receipt of an Award under this Plan, a Participant will be deemed to have agreed that the Award will be subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on the Participant's behalf with respect to any escrow, indemnities and any contingent consideration.

(e) No Restriction on Right to Undertake Transactions. The grant of any Award under the Plan and the issuance of shares pursuant to any Award does not affect or restrict in any way the right or power of the Company or the stockholders of the Company to make or authorize any adjustment, recapitalization, reorganization or other change in the Company's capital structure or its business, any merger or consolidation of the Company, any issue of stock or of options, rights or options to purchase stock or of bonds, debentures, preferred or prior preference stocks whose rights are superior to or affect the Common Stock or the rights thereof or which are convertible into or exchangeable for Common Stock, or the dissolution or liquidation of the Company, or any sale or transfer of all or any part of its assets or business, or any other corporate act or proceeding, whether of a similar character or otherwise.

### 8. Administration.

(a) Administration by Board. The Board will administer the Plan unless and until the Board delegates administration of the Plan to a Committee or Committees, as provided in subsection (c) below.

**(b) Powers of Board.** The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:

(i) To determine from time to time (1) which of the persons eligible under the Plan will be granted Awards; (2) when and how each Award will be granted; (3) what type or combination of types of Award will be granted; (4) the provisions of each Award granted (which need not be identical), including the time or times when a person will be permitted to receive an issuance of Common Stock or other payment pursuant to an Award; (5) the number of shares of Common Stock or cash equivalent with respect to which an Award will be granted to each such person; (6) the Fair Market Value applicable to an Award; and (7) the terms of any Performance Award that is not valued in whole or in part by reference to, or otherwise based on, the Common Stock, including the amount of cash payment or other property that may be earned and the timing of payment.

(ii) To construe and interpret the Plan and Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Award Agreement, in a manner and to the extent it deems necessary or expedient to make the Plan or Award fully effective.

(iii) To settle all controversies regarding the Plan and Awards granted under it.

(iv) To accelerate the time at which an Award may first be exercised or the time during which an Award or any part thereof will vest, notwithstanding the provisions in the Award Agreement stating the time at which it may first be exercised or the time during which it will vest.

(v) To prohibit the exercise of any Option, SAR or other exercisable Award during a period of up to 30 days prior to the consummation of any pending stock dividend, stock split, combination or exchange of shares, merger, consolidation or other distribution (other than normal cash dividends) of Company assets to stockholders, or any other change affecting the shares of Common Stock or the share price of the Common Stock including any Corporate Transaction, for reasons of administrative convenience.

(vi) To suspend or terminate the Plan at any time. Suspension or termination of the Plan will not Materially Impair rights and obligations under any Award granted while the Plan is in effect except with the written consent of the affected Participant.

(vii) To amend the Plan in any respect the Board deems necessary or advisable; provided, however, that stockholder approval will be required for any amendment to the extent required by Applicable Law. Except as provided above, rights under any Award granted before amendment of the Plan will not be Materially Impaired by any amendment of the Plan unless (1) the Company requests the consent of the affected Participant, and (2) such Participant consents in writing.

(viii) To submit any amendment to the Plan for stockholder approval.

(ix) To approve forms of Award Agreements for use under the Plan and to amend the terms of any one or more Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Award Agreement, subject to any specified limits in the Plan that are not subject to Board discretion; *provided however*, that, a Participant's rights under any Award will not be Materially Impaired by any such amendment unless (1) the Company requests the consent of the affected Participant, and (2) such Participant consents in writing.

(x) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan or Awards.

(xi) To adopt such procedures and sub-plans as are necessary or appropriate to permit and facilitate participation in the Plan by, or take advantage of specific tax treatment for Awards granted to, Employees, Directors or Consultants who are foreign nationals or employed outside the United States (provided that Board approval will not be necessary for immaterial modifications to the Plan or any Award Agreement to ensure or facilitate compliance with the laws of the relevant foreign jurisdiction).

(xii) To effect, at any time and from time to time, subject to the consent of any Participant whose Award is Materially Impaired by such action, (1) the reduction of the exercise price (or strike price) of any outstanding Option or SAR; (2) the cancellation of any outstanding Option or SAR and the grant in substitution therefor of (A) a new Option, SAR, Restricted Stock Award, RSU Award or Other Award, under the Plan or another equity plan of the Company, covering the same or a different number of shares of Common Stock, (B) cash and/or (C) other valuable consideration (as determined by the Board); or (3) any other action that is treated as a repricing under generally accepted accounting principles.

### (c) Delegation to Committee.

(i) General. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to another Committee or a subcommittee of the Committee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board will thereafter be to the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan as may be adopted from time to time by the Board. Each Committee may retain the authority to concurrently administer the Plan with Committee or subcommittee to which it has delegated its authority hereunder and may, at any time, revest in such Committee some or all of the powers previously delegated. The Board may retain the authority to concurrently administer the Plan with any Committee and may, at any time, revest in the Board some or all of the powers previously delegated.

(ii) **Rule 16b-3 Compliance.** To the extent an Award is intended to qualify for the exemption from Section 16(b) of the Exchange Act that is available under Rule 16b-3 of the Exchange Act, the Award will be granted by the Board or a Committee (or a subcommittee thereof) that consists solely of two or more Non-Employee Directors, as determined under Rule 16b-3(b)(3) of the Exchange Act and thereafter any action establishing or modifying the terms of the Award will be approved by the Board or a Committee (or a subcommittee) meeting such requirements to the extent necessary for such exemption to remain available.

(d) Effect of Board's Decision. All determinations, interpretations and constructions made by the Board or any Committee with respect to the Plan or any Awards granted under it (including any Award Agreements) will not be subject to review by any person and will be final, binding and conclusive on all persons.

(e) **Delegation to an Officer.** The Board or any Committee may delegate to one or more Officers the authority to do one or both of the following (i) designate Employees who are not Officers to be recipients of Options and SARs (and, to the extent permitted by Applicable Law, other types of Awards) and, to the extent permitted by Applicable Law, the terms thereof, and (ii) determine the number of shares of Common Stock to be subject to such Awards granted to such Employees; provided, however, that the resolutions or charter adopted by the Board or any Committee evidencing such delegation will specify the total number of shares of Common Stock that may be subject to the Awards granted by such Officer and that such Officer may not grant an Award to himself or herself. Any such Awards will be granted on the applicable form of Award Agreement most recently approved for use by the Board or the Committee, unless otherwise provided in the resolutions approving the delegation authority. Notwithstanding anything to the contrary herein, neither the Board nor any Committee may delegate to an Officer who is acting solely in the capacity of an Officer (and not also as a Director) the authority to determine the Fair Market Value.

#### 9. TAX WITHHOLDING

(a) Withholding Authorization. As a condition to acceptance of any Award under the Plan, a Participant authorizes withholding from payroll and any other amounts payable to such Participant, and otherwise agree to make adequate provision for (including), any sums required to satisfy any U.S. federal, state, local and/or foreign tax or social insurance contribution withholding obligations of the Company or an Affiliate, if any, which arise in connection with the exercise, vesting or settlement of such Award, as applicable. Accordingly, a Participant may not be able to exercise an Award even though the Award is vested, and the Company shall have no obligation to issue shares of Common Stock subject to an Award, unless and until such obligations are satisfied.

(b) Satisfaction of Withholding Obligation. To the extent permitted by the terms of an Award Agreement, the Company may, in its sole discretion, satisfy any U.S. federal, state, local and/or foreign tax or social insurance withholding obligation relating to an Award by any of the following means or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with the Award; (iii) withholding cash from an Award settled in cash; (iv) withholding payment from any amounts otherwise payable to the Participant; (v) by allowing a Participant to effectuate a "cashless exercise" pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board; or (vi) by such other method as may be set forth in the Award Agreement.

No Obligation to Notify or Minimize Taxes; No Liability to Claims. Except as required by Applicable Law the  $(\mathbf{c})$ Company has no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Award. Furthermore, the Company has no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of an Award or a possible period in which the Award may not be exercised. The Company has no duty or obligation to minimize the tax consequences of an Award to the holder of such Award and will not be liable to any holder of an Award for any adverse tax consequences to such holder in connection with an Award. As a condition to accepting an Award under the Plan, each Participant (i) agrees to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from such Award or other Company compensation and (ii) acknowledges that such Participant was advised to consult with his or her own personal tax, financial and other legal advisors regarding the tax consequences of the Award and has either done so or knowingly and voluntarily declined to do so. Additionally, each Participant acknowledges any Option or SAR granted under the Plan is exempt from Section 409A only if the exercise or strike price is at least equal to the "fair market value" of the Common Stock on the date of grant as determined by the Internal Revenue Service and there is no other impermissible deferral of compensation associated with the Award. Additionally, as a condition to accepting an Option or SAR granted under the Plan, each Participant agrees not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the Internal Revenue Service asserts that such exercise price or strike price is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the Internal Revenue Service.

(d) Withholding Indemnification. As a condition to accepting an Award under the Plan, in the event that the amount of the Company's and/or its Affiliate's withholding obligation in connection with such Award was greater than the amount actually withheld by the Company and/or its Affiliates, each Participant agrees to indemnify and hold the Company and/or its Affiliates harmless from any failure by the Company and/or its Affiliates to withhold the proper amount.

## **10. M**ISCELLANEOUS.

(a) **Source of Shares.** The stock issuable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market or otherwise.

(b) Use of Proceeds from Sales of Common Stock. Proceeds from the sale of shares of Common Stock pursuant to Awards will constitute general funds of the Company.

(c) Corporate Action Constituting Grant of Awards. Corporate action constituting a grant by the Company of an Award to any Participant will be deemed completed as of the date of such corporate action, unless otherwise determined by the Board, regardless of when the instrument, certificate, or letter evidencing the Award is communicated to, or actually received or accepted by, the Participant. In the event that the corporate records (e.g., Board consents, resolutions or minutes) documenting the corporate action approving the grant contain terms (e.g., exercise price, vesting schedule or number of shares) that are inconsistent with those in the Award Agreement or related grant documents as a result of a clerical error in the Award Agreement or related grant documents will have no legally binding right to the incorrect term in the Award Agreement or related grant documents.

(d) **Stockholder Rights.** No Participant will be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to such Award unless and until (i) such Participant has satisfied all requirements for exercise of the Award pursuant to its terms, if applicable, and (ii) the issuance of the Common Stock subject to such Award is reflected in the records of the Company.

(e) No Employment or Other Service Rights. Nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award granted pursuant thereto will confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Award was granted or affect the right of the Company or an Affiliate to terminate at will and without regard to any future vesting opportunity that a Participant may have with respect to any Award (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate, or (iii) the service of a Director pursuant to the Bylaws of the Company or an Affiliate is incorporated, as the case may be. Further, nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award will constitute any promise or commitment by the Company or an Affiliate regarding the fact or nature of future positions, future work assignments, future compensation or any other term or condition of employment or service or confer any right or benefit under the Award or the Plan unless such right or benefit has specifically accrued under the terms of the Award Agreement and/or Plan.

(f) Change in Time Commitment. In the event a Participant's regular level of time commitment in the performance of his or her services for the Company and any Affiliates is reduced (for example, and without limitation, if the Participant is an Employee of the Company and the Employee has a change in status from a full-time Employee to a part-time Employee or takes an extended leave of absence) after the date of grant of any Award to the Participant, the Board may determine, to the extent permitted by Applicable Law, to (i) make a corresponding reduction in the number of shares or cash amount subject to any portion of such Award that is scheduled to vest or become payable after the date of such change in time commitment, and (ii) in lieu of or in combination with such a reduction, extend the vesting or payment schedule applicable to such Award. In the event of any such reduction, the Participant will have no right with respect to any portion of the Award that is so reduced or extended.

(g) **Execution of Additional Documents.** As a condition to accepting an Award under the Plan, the Participant agrees to execute any additional documents or instruments necessary or desirable, as determined in the Plan Administrator's sole discretion, to carry out the purposes or intent of the Award, or facilitate compliance with securities and/or other regulatory requirements, in each case at the Plan Administrator's request.

(h) Electronic Delivery and Participation. Any reference herein or in an Award Agreement to a "written" agreement or document will include any agreement or document delivered electronically, filed publicly at www.sec.gov (or any successor website thereto) or posted on the Company's intranet (or other shared electronic medium controlled by the Company to which the Participant has access). By accepting any Award the Participant consents to receive documents by electronic delivery and to participate in the Plan through any on-line electronic system established and maintained by the Plan Administrator or another third party selected by the Plan Administrator. The form of delivery of any Common Stock (e.g., a stock certificate or electronic entry evidencing such shares) shall be determined by the Company.

(i) **Clawback/Recovery.** All Awards granted under the Plan will be subject to recoupment in accordance with any clawback policy that the Company is required to adopt pursuant to the listing standards of any national securities exchange or association on which the Company's securities are listed or as is otherwise required by the Dodd-Frank Wall Street Reform and Consumer Protection Act or other Applicable Law and any clawback policy that the Company otherwise adopts, to the extent applicable and permissible under Applicable Law. In addition, the Board may impose such other clawback, recovery or recoupment provisions in an Award Agreement as the Board determines necessary or appropriate, including but not limited to a reacquisition right in respect of previously acquired shares of Common Stock or other cash or property upon the occurrence of Cause. No recovery of compensation under such a clawback policy will be an event giving rise to a Participant's right to voluntary terminate employment upon a "resignation for good reason," or for a "constructive termination" or any similar term under any plan of or agreement with the Company.

(j) Securities Law Compliance. A Participant will not be issued any shares in respect of an Award unless either (i) the shares are registered under the Securities Act; or (ii) the Company has determined that such issuance would be exempt from the registration requirements of the Securities Act. Each Award also must comply with other Applicable Law governing the Award, and a Participant will not receive such shares if the Company determines that such receipt would not be in material compliance with Applicable Law.

(k) Transfer or Assignment of Awards; Issued Shares. Except as expressly provided in the Plan or the form of Award Agreement, Awards granted under the Plan may not be transferred or assigned by the Participant. After the vested shares subject to an Award have been issued, or in the case of Restricted Stock and similar awards, after the issued shares have vested, the holder of such shares is free to assign, hypothecate, donate, encumber or otherwise dispose of any interest in such shares provided that any such actions are in compliance with the provisions herein, the terms of the Trading Policy and Applicable Law.

(I) Effect on Other Employee Benefit Plans. The value of any Award granted under the Plan, as determined upon grant, vesting or settlement, shall not be included as compensation, earnings, salaries, or other similar terms used when calculating any Participant's benefits under any employee benefit plan sponsored by the Company or any Affiliate, except as such plan otherwise expressly provides. The Company expressly reserves its rights to amend, modify, or terminate any of the Company's or any Affiliate's employee benefit plans.

(m) **Deferrals.** To the extent permitted by Applicable Law, the Board, in its sole discretion, may determine that the delivery of Common Stock or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Award may be deferred and may also establish programs and procedures for deferral elections to be made by Participants. Deferrals by will be made in accordance with the requirements of Section 409A.

(n) Section 409A. Unless otherwise expressly provided for in an Award Agreement, the Plan and Award Agreements will be interpreted to the greatest extent possible in a manner that makes the Plan and the Awards granted hereunder exempt from Section 409A, and, to the extent not so exempt, in compliance with the requirements of Section 409A. If the Board determines that any Award granted hereunder is not exempt from and is therefore subject to Section 409A, the Award Agreement evidencing such Award will incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code, and to the extent an Award Agreement is silent on terms necessary for compliance, such terms are hereby incorporated by reference into the Award Agreement. Notwithstanding anything to the contrary in this Plan (and unless the Award Agreement specifically provides otherwise), if the shares of Common Stock are publicly traded, and if a Participant holding an Award that constitutes "deferred compensation" under Section 409A is a "specified employee" for purposes of Section 409A, no distribution or payment of any amount that is due because of a "separation from service" (as defined in Section 409A without regard to alternative definitions thereunder) will be issued or paid before the date that is six months and one day following the date of such Participant's "separation from service" or, if earlier, the date of the Participant's death, unless such distribution or payment can be made in a manner that complies with Section 409A, and any amounts so deferred will be paid in a lump sum on the day after such six month period elapses, with the balance paid thereafter on the original schedule.

(0) **CHOICE OF LAW.** This Plan and any controversy arising out of or relating to this Plan shall be governed by, and construed in accordance with, the internal laws of the State of Delaware, without regard to conflict of law principles that would result in any application of any law other than the law of the State of Delaware.

### **11.** COVENANTS OF THE COMPANY.

(a) Compliance with Law. The Company will seek to obtain from each regulatory commission or agency, as may be deemed to be necessary, having jurisdiction over the Plan such authority as may be required to grant Awards and to issue and sell shares of Common Stock upon exercise or vesting of the Awards; provided, however, that this undertaking will not require the Company to register under the Securities Act the Plan, any Award or any Common Stock issued or issuable pursuant to any such Award. If, after reasonable efforts and at a reasonable cost, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary or advisable for the lawful issuance and sale of Common Stock under the Plan, the Company will be relieved from any liability for failure to issue and sell Common Stock upon exercise or vesting of such Awards unless and until such authority is obtained. A Participant is not eligible for the grant of an Award or the subsequent issuance of Common Stock pursuant to the Award if such grant or issuance would be in violation of any Applicable Law.

### 12. Additional Rules for Awards Subject to Section 409A.

(a) **Application.** Unless the provisions of this Section of the Plan are expressly superseded by the provisions in the form of Award Agreement, the provisions of this Section shall apply and shall supersede anything to the contrary set forth in the Award Agreement for a Non-Exempt Award.

(b) Non-Exempt Awards Subject to Non-Exempt Severance Arrangements. To the extent a Non-Exempt Award is subject to Section 409A due to application of a Non-Exempt Severance Arrangement, the following provisions of this subsection (b) apply.

(i) If the Non-Exempt Award vests in the ordinary course during the Participant's Continuous Service in accordance with the vesting schedule set forth in the Award Agreement, and does not accelerate vesting under the terms of a Non-Exempt Severance Arrangement, in no event will the shares be issued in respect of such Non-Exempt Award any later than the later of: (i) December 31<sup>st</sup> of the calendar year that includes the applicable vesting date, or (ii) the 60<sup>th</sup> day that follows the applicable vesting date.

(ii) If vesting of the Non-Exempt Award accelerates under the terms of a Non-Exempt Severance Arrangement in connection with the Participant's Separation from Service, and such vesting acceleration provisions were in effect as of the date of grant of the Non-Exempt Award and, therefore, are part of the terms of such Non-Exempt Award as of the date of grant, then the shares will be earlier issued in settlement of such Non-Exempt Award upon the Participant's Separation from Service in accordance with the terms of the Non-Exempt Severance Arrangement, but in no event later than the 60<sup>th</sup> day that follows the date of the Participant's Separation from Service. However, if at the time the shares would otherwise be issued the Participant is subject to the distribution limitations contained in Section 409A applicable to "specified employees," as defined in Section 409A(a)(2)(B)(i) of the Code, such shares shall not be issued before the date that is six months following the date of such Participant's Separation from Service, or, if earlier, the date of the Participant's death that occurs within such six month period.

(iii) If vesting of a Non-Exempt Award accelerates under the terms of a Non-Exempt Severance Arrangement in connection with a Participant's Separation from Service, and such vesting acceleration provisions were not in effect as of the date of grant of the Non-Exempt Award and, therefore, are not a part of the terms of such Non-Exempt Award on the date of grant, then such acceleration of vesting of the Non-Exempt Award shall not accelerate the issuance date of the shares, but the shares shall instead be issued on the same schedule as set forth in the Grant Notice as if they had vested in the ordinary course during the Participant's Continuous Service, notwithstanding the vesting acceleration of the Non-Exempt Award. Such issuance schedule is intended to satisfy the requirements of payment on a specified date or pursuant to a fixed schedule, as provided under Treasury Regulations Section 1.409A-3(a)(4).

(c) **Treatment of Non-Exempt Awards Upon a Corporate Transaction for Employees and Consultants.** The provisions of this subsection (c) shall apply and shall supersede anything to the contrary set forth in the Plan with respect to the permitted treatment of any Non-Exempt Award in connection with a Corporate Transaction if the Participant was either an Employee or Consultant upon the applicable date of grant of the Non-Exempt Award.

(i) Vested Non-Exempt Awards. The following provisions shall apply to any Vested Non-Exempt Award in connection with a Corporate Transaction:

(1) If the Corporate Transaction is also a Section 409A Change in Control then the Acquiring Entity may not assume, continue or substitute the Vested Non-Exempt Award. Upon the Section 409A Change in Control the settlement of the Vested Non-Exempt Award will automatically be accelerated and the shares will be immediately issued in respect of the Vested Non-Exempt Award. Alternatively, the Company may instead provide that the Participant will receive a cash settlement equal to the Fair Market Value of the shares that would otherwise be issued to the Participant upon the Section 409A Change in Control.

(2) If the Corporate Transaction is not also a Section 409A Change in Control, then the Acquiring Entity must either assume, continue or substitute each Vested Non-Exempt Award. The shares to be issued in respect of the Vested Non-Exempt Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of the Fair Market Value of the shares made on the date of the Corporate Transaction.

(ii) Unvested Non-Exempt Awards. The following provisions shall apply to any Unvested Non-Exempt Award unless otherwise determined by the Board pursuant to subsection (e) of this Section.

(1) In the event of a Corporate Transaction, the Acquiring Entity shall assume, continue or substitute any Unvested Non-Exempt Award. Unless otherwise determined by the Board, any Unvested Non-Exempt Award will remain subject to the same vesting and forfeiture restrictions that were applicable to the Award prior to the Corporate Transaction. The shares to be issued in respect of any Unvested Non-Exempt Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of Fair Market Value of the shares made on the date of the Corporate Transaction.

(2) If the Acquiring Entity will not assume, substitute or continue any Unvested Non-Exempt Award in connection with a Corporate Transaction, then such Award shall automatically terminate and be forfeited upon the Corporate Transaction with no consideration payable to any Participant in respect of such forfeited Unvested Non-Exempt Award. Notwithstanding the foregoing, to the extent permitted and in compliance with the requirements of Section 409A, the Board may in its discretion determine to elect to accelerate the vesting and settlement of the Unvested Non-Exempt Award upon the Corporate Transaction, or instead substitute a cash payment equal to the Fair Market Value of such shares that would otherwise be issued to the Participant, as further provided in subsection (e)(ii) below. In the absence of such discretionary election by the Board, any Unvested Non-Exempt Award shall be forfeited without payment of any consideration to the affected Participants if the Acquiring Entity will not assume, substitute or continue the Unvested Non-Exempt Awards in connection with the Corporate Transaction.

(3) The foregoing treatment shall apply with respect to all Unvested Non-Exempt Awards upon any Corporate Transaction, and regardless of whether or not such Corporate Transaction is also a Section 409A Change in Control.

(d) Treatment of Non-Exempt Awards Upon a Corporate Transaction for Non-Employee Directors. The following provisions of this subsection (d) shall apply and shall supersede anything to the contrary that may be set forth in the Plan with respect to the permitted treatment of a Non-Exempt Director Award in connection with a Corporate Transaction.

(i) If the Corporate Transaction is also a Section 409A Change in Control then the Acquiring Entity may not assume, continue or substitute the Non-Exempt Director Award. Upon the Section 409A Change in Control the vesting and settlement of any Non-Exempt Director Award will automatically be accelerated and the shares will be immediately issued to the Participant in respect of the Non-Exempt Director Award. Alternatively, the Company may provide that the Participant will instead receive a cash settlement equal to the Fair Market Value of the shares that would otherwise be issued to the Participant upon the Section 409A Change in Control pursuant to the preceding provision.

(ii) If the Corporate Transaction is not also a Section 409A Change in Control, then the Acquiring Entity must either assume, continue or substitute the Non-Exempt Director Award. Unless otherwise determined by the Board, the Non-Exempt Director Award will remain subject to the same vesting and forfeiture restrictions that were applicable to the Award prior to the Corporate Transaction. The shares to be issued in respect of the Non-Exempt Director Award shall be issued to the Participant by the Acquiring Entity on the same schedule that the shares would have been issued to the Participant if the Corporate Transaction had not occurred. In the Acquiring Entity's discretion, in lieu of an issuance of shares, the Acquiring Entity may instead substitute a cash payment on each applicable issuance date, equal to the Fair Market Value of the shares that would otherwise be issued to the Participant on such issuance dates, with the determination of Fair Market Value made on the date of the Corporate Transaction.

(e) If the RSU Award is a Non-Exempt Award, then the provisions in this Section 11(e) shall apply and supersede anything to the contrary that may be set forth in the Plan or the Award Agreement with respect to the permitted treatment of such Non-Exempt Award:

(i) Any exercise by the Board of discretion to accelerate the vesting of a Non-Exempt Award shall not result in any acceleration of the scheduled issuance dates for the shares in respect of the Non-Exempt Award unless the earlier issuance of the shares upon the applicable vesting dates would be in compliance with the requirements of Section 409A.

(ii) The Company explicitly reserves the right to earlier settle any Non-Exempt Award to the extent permitted and in compliance with the requirements of Section 409A, including pursuant to any of the exemptions available in Treasury Regulations Section 1.409A-3(j)(4)(ix).

(iii) To the extent the terms of any Non-Exempt Award provide that it will be settled upon a Change in Control or Corporate Transaction, to the extent it is required for compliance with the requirements of Section 409A, the Change in Control or Corporate Transaction event triggering settlement must also constitute a Section 409A Change in Control. To the extent the terms of a Non-Exempt Award provides that it will be settled upon a termination of employment or termination of Continuous Service, to the extent it is required for compliance with the requirements of Section 409A, the termination event triggering settlement must also constitute a Separation From Service. However, if at the time the shares would otherwise be issued to a Participant in connection with a "separation from service" such Participant is subject to the distribution limitations contained in Section 409A applicable to "specified employees," as defined in Section 409A(a)(2)(B)(i) of the Code, such shares shall not be issued before the date that is six months following the date of the Participant's Separation From Service, or, if earlier, the date of the Participant's death that occurs within such six month period.

(iv) The provisions in this subsection (e) for delivery of the shares in respect of the settlement of an RSU Award that is a Non-Exempt Award are intended to comply with the requirements of Section 409A so that the delivery of the shares to the Participant in respect of such Non-Exempt Award will not trigger the additional tax imposed under Section 409A, and any ambiguities herein will be so interpreted.

### 13. SEVERABILITY.

If all or any part of the Plan or any Award Agreement is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity shall not invalidate any portion of the Plan or such Award Agreement not declared to be unlawful or invalid. Any Section of the Plan or any Award Agreement (or part of such a Section) so declared to be unlawful or invalid shall, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid.

### **14. TERMINATION OF THE PLAN.**

The Board may suspend or terminate the Plan at any time. No Incentive Stock Options may be granted after the tenth anniversary of the earlier of: (i) the Adoption Date, or (ii) the date the Plan is approved by the Company's stockholders. No Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

### **15. D**EFINITIONS.

As used in the Plan, the following definitions apply to the capitalized terms indicated below:

(a) *"Acquiring Entity"* means the surviving or acquiring corporation (or its parent company) in connection with a Corporate Transaction.

(b) *"Adoption Date"* means the date the Plan is first approved by the Board or Compensation Committee.

(c) "*Affiliate*" means, at the time of determination, any "parent" or "subsidiary" of the Company as such terms are defined in Rule 405 promulgated under the Securities Act. The Board may determine the time or times at which "parent" or "subsidiary" status is determined within the foregoing definition.

(d) "*Applicable Law*" means shall mean any applicable securities, federal, state, foreign, material local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, listing rule, regulation, judicial decision, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Body (including under the authority of any applicable self-regulating organization such as the Nasdaq Stock Market, New York Stock Exchange, or the Financial Industry Regulatory Authority).

(e) "*Award*" means any right to receive Common Stock, cash or other property granted under the Plan (including an Incentive Stock Option, a Nonstatutory Stock Option, a Restricted Stock Award, an RSU Award, a SAR, OpCo Units, a Performance Award or any Other Award).

(f) "*Award Agreement*" means a written agreement between the Company and a Participant evidencing the terms and conditions of an Award. The Award Agreement generally consists of the Grant Notice and the agreement containing the written summary of the general terms and conditions applicable to the Award and which is provided to a Participant along with the Grant Notice.

(g) "*Board*" means the Board of Directors of the Company (or its designee). Any decision or determination made by the Board shall be a decision or determination that is made in the sole discretion of the Board (or its designee), and such decision or determination shall be final and binding on all Participants.

(h) "*Capitalization Adjustment*" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Award after the Effective Date without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, reverse stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or any similar equity restructuring transaction, as that term is used in Statement of Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.

(i) "*Cause*" has the meaning ascribed to such term in any written agreement between the Participant and the Company defining such term and, in the absence of such agreement, such term means, with respect to a Participant, the occurrence of any of the following events: (i) such Participant's commission of (or attempted commission of), or participation in, a fraud or act of dishonesty against the Company; (ii) such Participant's material violation of any contract or agreement between the Participant and the Company or of any statutory duty owed to the Company; (iii) such Participant's unauthorized use or disclosure of the Company's confidential information or trade secrets; (iv) such Participant's gross or willful misconduct; or (v) such Participant's conviction of, or plea of *nolo contendere* to, a felony or crime of moral turpitude. The determination that a termination of the Participant's Continuous Service is either for Cause or without Cause will be made by the Board with respect to Participants who are not executive officers of the Company. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Awards held by such Participant will have no effect upon any determination of the rights or obligations of the Company or such Participant for any other purpose.

(j) "*Change in Control*" or "*Change of Control*" means the occurrence, in a single transaction or in a series of related transactions, of any one or more of the following events; provided, however, to the extent necessary to avoid adverse personal income tax consequences to the Participant in connection with an Award, also constitutes a Section 409A Change in Control:

(i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control shall not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities, or (C) solely because the level of Ownership held by any Exchange Act Person (the *"Subject Person"*) exceeds the designated percentage threshold of the outstanding voting securities as a result of a repurchase or other acquisition of voting securities by the Company reducing the number of shares outstanding, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control shall be deemed to occur;

(ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the stockholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than 50% of the combined outstanding voting power of the surviving Entity in such merger, consolidation or similar transaction or (B) more than 50% of the combined outstanding voting power of the parent of the surviving Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction;

(iii) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries to an Entity, more than 50% of the combined voting power of the voting securities of which are Owned by stockholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition; or

(iv) individuals who, on the date the Plan is adopted by the Board, are members of the Board (the "*Incumbent Board*") cease for any reason to constitute at least a majority of the members of the Board; provided, however, that if the appointment or election (or nomination for election) of any new Board member was approved or recommended by a majority vote of the members of the Incumbent Board then still in office, such new member shall, for purposes of this Plan, be considered as a member of the Incumbent Board.

Notwithstanding the foregoing or any other provision of this Plan, (A) the term Change in Control shall not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company, and (B) the definition of Change in Control (or any analogous term) in an individual written agreement between the Company or any Affiliate and the Participant that is specifically intended to apply to an Award granted under the Plan shall supersede the foregoing definition with respect to Awards subject to such agreement; provided, however, that if no definition of Change in Control or any analogous term is set forth in such an individual written agreement, the foregoing definition shall apply.

(k) "*Code*" means the Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.

(I) *"Committee*" means the Compensation Committee and any other committee of Directors to whom authority has been delegated by the Board or Compensation Committee in accordance with the Plan.

- (m) *"Common Stock"* means the Class A common stock of the Company.
- (n) *"Company"* means WM Technology, Inc., a Delaware corporation.
- (o) *"Compensation Committee*" means the Compensation Committee of the Board.

(p) "*Consultant*" means any person, including an advisor, who is (i) engaged by the Company or an Affiliate to render consulting or advisory services and is compensated for such services, or (ii) serving as a member of the board of directors of an Affiliate and is compensated for such services. However, service solely as a Director, or payment of a fee for such service, will not cause a Director to be considered a "Consultant" for purposes of the Plan. Notwithstanding the foregoing, a person is treated as a Consultant under this Plan only if a Form S-8 Registration Statement under the Securities Act is available to register either the offer or the sale of the Company's securities to such person.

"Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an (q) Employee, Director or Consultant, is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Director or Consultant or a change in the Entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, will not terminate a Participant's Continuous Service (unless otherwise determined by the Board in its sole discretion): provided, however, that if the Entity for which a Participant is rendering services ceases to qualify as an Affiliate, as determined by the Board, such Participant's Continuous Service will be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or to a Director will not constitute an interruption of Continuous Service (unless otherwise determined by the Board in its sole discretion). To the extent permitted by law, the Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service will be considered interrupted in the case of (i) any leave of absence approved by the Board or chief executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. Notwithstanding the foregoing, a leave of absence will be treated as Continuous Service for purposes of vesting in an Award only to such extent as may be provided in the Company's leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law. In addition, to the extent required for exemption from or compliance with Section 409A, the determination of whether there has been a termination of Continuous Service will be made, and such term will be construed, in a manner that is consistent with the definition of "separation from service" as defined under Treasury Regulation Section 1.409A-1(h) (without regard to any alternative definition thereunder).

(r) *"Corporate Transaction"* means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:

(i) a sale or other disposition of all or substantially all, as determined by the Board, of the consolidated assets of the Company and its Subsidiaries;

(ii) a sale or other disposition of at least 50% of the outstanding securities of the Company;

(iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or

(iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.

(s) "*Director*" means a member of the Board.

(t) *"determine" or "determined"* means as determined by the Board or the Committee (or its designee) in its sole discretion.

(u) "*Disability*" means, with respect to a Participant, such Participant is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than 12 months, as provided in Section 22(e)(3) of the Code, and will be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.

(v) *"Effective Date"* means June 16, 2021.

(w) *"Employee"* means any person employed by the Company or an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.

- (x) *"Employer"* means the Company or the Affiliate of the Company that employs the Participant.
- (y) *"Entity"* means a corporation, partnership, limited liability company or other entity.

(z) *"Exchange Act"* means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

(aa) "Exchange Act Person" means any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act), except that "Exchange Act Person" will not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company or any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to a registered public offering of such securities, (iv) an Entity Owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their Ownership of stock of the Company; or (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities.

**(bb)** *"Fair Market Value"* means, as of any date, unless otherwise determined by the Board, the value of the Common Stock (as determined on a per share or aggregate basis, as applicable) determined as follows:

(i) If the Common Stock is listed on any established stock exchange or traded on any established market, the Fair Market Value will be the closing sales price for such stock as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in the Common Stock) on the date of determination, as reported in a source the Board deems reliable.

(ii) If there is no closing sales price for the Common Stock on the date of determination, then the Fair Market Value will be the closing selling price on the last preceding date for which such quotation exists.

(iii) In the absence of such markets for the Common Stock, or if otherwise determined by the Board, the Fair Market Value will be determined by the Board in good faith and in a manner that complies with Sections 409A and 422 of the Code.

(cc) "*Governmental Body*" means any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or regulatory body, or quasi-governmental body of any nature (including any governmental division, department, administrative agency or bureau, commission, authority, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or Entity and any court or other tribunal, and for the avoidance of doubt, any Tax authority) or other body exercising similar powers or authority; or (d) self-regulatory organization (including the Nasdaq Stock Market, New York Stock Exchange, and the Financial Industry Regulatory Authority).

(dd) "*Grant Notice*" means the notice provided to a Participant that he or she has been granted an Award under the Plan and which includes the name of the Participant, the type of Award, the date of grant of the Award, number of shares of Common Stock subject to the Award or potential cash payment right, (if any), the vesting schedule for the Award (if any) and other key terms applicable to the Award.

(ee) *"Incentive Stock Option*" means an option granted pursuant to Section 4 of the Plan that is intended to be, and qualifies as, an "incentive stock option" within the meaning of Section 422 of the Code.

(ff) "*Materially Impair*" means any amendment to the terms of the Award that materially adversely affects the Participant's rights under the Award. A Participant's rights under an Award will not be deemed to have been Materially Impaired by any such amendment if the Board, in its sole discretion, determines that the amendment, taken as a whole, does not materially impair the Participant's rights. For example, the following types of amendments to the terms of an Award do not Materially Impair the Participant's rights under the Award: (i) imposition of reasonable restrictions on the minimum number of shares subject to an Option that may be exercised; (ii) to maintain the qualified status of the Award as an Incentive Stock Option under Section 422 of the Code; (iii) to change the terms of an Incentive Stock Option in a manner that disqualifies, impairs or otherwise affects the qualified status of the Award as an Incentive Stock Option under Section 422 of the Code; (iv) to clarify the manner of exemption from, or to bring the Award into compliance with or qualify it for an exemption from, Section 409A; or (v) to comply with other Applicable Laws.

(a) *"Merger Agreement"* means that certain Agreement and Plan of Merger, dated as of December 10, 2020, by and among Silver Spike Acquisition Corp. (*"Silver Spike"*), a Cayman Islands exempted company, Silver Spike Merger Sub LLC, a Delaware limited liability company and a wholly owned direct subsidiary of Silver Spike, WM Holding Company, LLC, a Delaware limited liability company, and Ghost Media Group, LLC, a Nevada limited liability company.

(gg) "*Non-Employee Director*" means a Director who either (i) is not a current employee or officer of the Company or an Affiliate, does not receive compensation, either directly or indirectly, from the Company or an Affiliate for services rendered as a consultant or in any capacity other than as a Director (except for an amount as to which disclosure would not be required under Item 404(a) of Regulation S-K promulgated pursuant to the Securities Act ("*Regulation S-K*")), does not possess an interest in any other transaction for which disclosure would be required under Item 404(a) of Regulation S-K, and is not engaged in a business relationship for which disclosure would be required pursuant to Item 404(b) of Regulation S-K; or (ii) is otherwise considered a "non-employee director" for purposes of Rule 16b-3.

(hh) "*Non-Exempt Award*" means any Award that is subject to, and not exempt from, Section 409A, including as the result of (i) a deferral of the issuance of the shares subject to the Award which is elected by the Participant or imposed by the Company, (ii) the terms of any Non-Exempt Severance Agreement.

(ii) *"Non-Exempt Director Award*" means a Non-Exempt Award granted to a Participant who was a Director but not an Employee on the applicable grant date.

(jj) "Non-Exempt Severance Arrangement" means a severance arrangement or other agreement between the Participant and the Company that provides for acceleration of vesting of an Award and issuance of the shares in respect of such Award upon the Participant's termination of employment or separation from service (as such term is defined in Section 409A(a) (2)(A)(i) of the Code (and without regard to any alternative definition thereunder) ("Separation from Service") and such severance benefit does not satisfy the requirements for an exemption from application of Section 409A provided under Treasury Regulations Section 1.409A-1(b)(4), 1.409A-1(b)(9) or otherwise.

(kk) *"Nonstatutory Stock Option"* means any option granted pursuant to Section 4 of the Plan that does not qualify as an Incentive Stock Option.

(ll) *"Officer*" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act.

(mm) "*OpCo LLC Agreement*" means the Fifth Amended and Restated Operating Agreement of WM Holding Company, LLC, as amended or amended and restated from time to time.

(nn) *"OpCo Unit"* means an Award granted under Section 6 of the Plan.

**(00)** *"Option*" means an Incentive Stock Option or a Nonstatutory Stock Option to purchase shares of Common Stock granted pursuant to the Plan.

**(pp)** "*Option Agreement*" means a written agreement between the Company and the Optionholder evidencing the terms and conditions of the Option grant. The Option Agreement includes the Grant Notice for the Option and the agreement containing the written summary of the general terms and conditions applicable to the Option and which is provided to a Participant along with the Grant Notice. Each Option Agreement will be subject to the terms and conditions of the Plan.

(qq) *"Optionholder*" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.

(rr) *"Other Award*" means an award based in whole or in part by reference to the Common Stock which is granted pursuant to the terms and conditions of Section 5(c).

(ss) *"Other Award Agreement"* means a written agreement between the Company and a holder of an Other Award evidencing the terms and conditions of an Other Award grant. Each Other Award Agreement will be subject to the terms and conditions of the Plan.

(tt) "Own," "Owned," "Owner," "Ownership" means that a person or Entity will be deemed to "Own," to have "Owned," to be the "Owner" of, or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.

(uu) "*Participant*" means an Employee, Director or Consultant to whom an Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Award.

(vv) "*Performance Award*" means an Award that may vest or may be exercised or a cash award that may vest or become earned and paid contingent upon the attainment during a Performance Period of certain Performance Goals and which is granted under the terms and conditions of Section 5(b) pursuant to such terms as are approved by the Board. In addition, to the extent permitted by Applicable Law and set forth in the applicable Award Agreement, the Board may determine that cash or other property may be used in payment of Performance Awards. Performance Awards that are settled in cash or other property are not required to be valued in whole or in part by reference to, or otherwise based on, the Common Stock.

(ww) "*Performance Criteria*" means the one or more criteria that the Board will select for purposes of establishing the Performance Goals for a Performance Period. The Performance Criteria that will be used to establish such Performance Goals may be based on any measure of performance selected by the Board.

"Performance Goals" means, for a Performance Period, the one or more goals established by the Board for the (xx) Performance Period based upon the Performance Criteria. Performance Goals may be based on a Company-wide basis, with respect to one or more business units, divisions, Affiliates, or business segments, and in either absolute terms or relative to the performance of one or more comparable companies or the performance of one or more relevant indices. Unless specified otherwise by the Board (i) in the Award Agreement at the time the Award is granted or (ii) in such other document setting forth the Performance Goals at the time the Performance Goals are established, the Board will appropriately make adjustments in the method of calculating the attainment of Performance Goals for a Performance Period as follows: (1) to exclude restructuring and/or other nonrecurring charges; (2) to exclude exchange rate effects; (3) to exclude the effects of changes to generally accepted accounting principles; (4) to exclude the effects of any statutory adjustments to corporate tax rates; (5) to exclude the effects of items that are "unusual" in nature or occur "infrequently" as determined under generally accepted accounting principles; (6) to exclude the dilutive effects of acquisitions or joint ventures; (7) to assume that any business divested by the Company achieved performance objectives at targeted levels during the balance of a Performance Period following such divestiture; (8) to exclude the effect of any change in the outstanding shares of common stock of the Company by reason of any stock dividend or split, stock repurchase, reorganization, recapitalization, merger, consolidation, spin-off, combination or exchange of shares or other similar corporate change, or any distributions to common stockholders other than regular cash dividends; (9) to exclude the effects of stock based compensation and the award of bonuses under the Company's bonus plans; (10) to exclude costs incurred in connection with potential acquisitions or divestitures that are required to expense under generally accepted accounting principles; and (11) to exclude the goodwill and intangible asset impairment charges that are required to be recorded under generally accepted accounting principles. In addition, the Board retains the discretion to reduce or eliminate the compensation or economic benefit due upon attainment of Performance Goals and to define the manner of calculating the Performance Criteria it selects to use for such Performance Period. Partial achievement of the specified criteria may result in the payment or vesting corresponding to the degree of achievement as specified in the Award Agreement or the written terms of a Performance Cash Award.

**(yy)** "*Performance Period*" means the period of time selected by the Board over which the attainment of one or more Performance Goals will be measured for the purpose of determining a Participant's right to vesting or exercise of an Award. Performance Periods may be of varying and overlapping duration, at the sole discretion of the Board.

(zz) "*Plan*" means this WM Technology, Inc. 2021 Equity Incentive Plan, as amended from time to time.

(aaa) "*Plan Administrator*" means the person, persons, and/or third-party administrator designated by the Company to administer the day to day operations of the Plan and the Company's other equity incentive programs.

**(bbb)** *"Post-Termination Exercise Period"* means the period following termination of a Participant's Continuous Service within which an Option or SAR is exercisable, as specified in Section 4(h).

(ccc) *"Prospectus"* means the document containing the Plan information specified in Section 10(a) of the Securities Act.

(ddd) *"Restricted Stock Award"* or *"RSA"* means an Award of shares of Common Stock which is granted pursuant to the terms and conditions of Section 5(a).

**(eee)** *"Restricted Stock Award Agreement"* means a written agreement between the Company and a holder of a Restricted Stock Award evidencing the terms and conditions of a Restricted Stock Award grant. The Restricted Stock Award Agreement includes the Grant Notice for the Restricted Stock Award and the agreement containing the written summary of the general terms and conditions applicable to the Restricted Stock Award and which is provided to a Participant along with the Grant Notice. Each Restricted Stock Award Agreement will be subject to the terms and conditions of the Plan.

(fff) "*RSU Award*" or "*RSU*" means an Award of restricted stock units representing the right to receive an issuance of shares of Common Stock which is granted pursuant to the terms and conditions of Section 5(a).

**(ggg)** "*RSU Award Agreement*" means a written agreement between the Company and a holder of an RSU Award evidencing the terms and conditions of an RSU Award grant. The RSU Award Agreement includes the Grant Notice for the RSU Award and the agreement containing the written summary of the general terms and conditions applicable to the RSU Award and which is provided to a Participant along with the Grant Notice. Each RSU Award Agreement will be subject to the terms and conditions of the Plan.

(hhh) "*Rule 16b-3*" means Rule 16b-3 promulgated under the Exchange Act or any successor to Rule 16b-3, as in effect from time to time.

(iii) *"Rule 405"* means Rule 405 promulgated under the Securities Act.

(jjj) *"Section 409A"* means Section 409A of the Code and the regulations and other guidance thereunder.

(kkk) "*Section 409A Change in Control*" means a change in the ownership or effective control of the Company, or in the ownership of a substantial portion of the Company's assets, as provided in Section 409A(a)(2)(A)(v) of the Code and Treasury Regulations Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder).

(III) *"Securities Act"* means the Securities Act of 1933, as amended.

(mmm) "Share Reserve" means the number of shares available for issuance under the Plan as set forth in Section 2(a).

(nnn) *"Stock Appreciation Right"* or *"SAR"* means a right to receive the appreciation on Common Stock that is granted pursuant to the terms and conditions of Section 4.

(000) "*SAR Agreement*" means a written agreement between the Company and a holder of a SAR evidencing the terms and conditions of a SAR grant. The SAR Agreement includes the Grant Notice for the SAR and the agreement containing the written summary of the general terms and conditions applicable to the SAR and which is provided to a Participant along with the Grant Notice. Each SAR Agreement will be subject to the terms and conditions of the Plan.

**(ppp)** "*Subsidiary*" means, with respect to the Company, (i) any corporation of which more than 50% of the outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, stock of any other class or classes of such corporation will have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than 50%.

(qqq) "*Ten Percent Stockholder*" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or any Affiliate.

(**rrr**) "*Trading Policy*" means the Company's policy permitting certain individuals to sell Company shares only during certain "window" periods and/or otherwise restricts the ability of certain individuals to transfer or encumber Company shares, as in effect from time to time.

(sss) *"Unvested Non-Exempt Award*" means the portion of any Non-Exempt Award that had not vested in accordance with its terms upon or prior to the date of any Corporate Transaction.

(ttt) *"Vested Non-Exempt Award*" means the portion of any Non-Exempt Award that had vested in accordance with its terms upon or prior to the date of a Corporate Transaction.

(uuu) *"WM OpCo"* means WM Holding Company, LLC, a Delaware limited liability company.

1

# WM TECHNOLOGY, INC. Stock Option Grant Notice (2021 Equity Incentive Plan)

WM Technology, Inc. (the "*Company*"), pursuant to the Company's 2021 Equity Incentive Plan (the "*Plan*"), has granted to you ("*Optionholder*") an option to purchase the number of shares of the Common Stock set forth below (the "*Option*"). Your Option is subject to all of the terms and conditions as set forth herein and in the Plan, and the Stock Option Agreement and the Notice of Exercise, all of which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Stock Option Agreement shall have the meanings set forth in the Plan or the Stock Option Agreement, as applicable.

Type of Grant:

[Incentive Stock Option] <u>OR</u> [Nonstatutory Stock Option]

Exercise andSubject to the Optionholder's Continuous Service through each applicable vesting date, the Option will vest as follows:Vesting Schedule:

**Optionholder Acknowledgements:** By your signature below or by electronic acceptance or authentication in a form authorized by the Company, you understand and agree that:

- The Option is governed by this Stock Option Grant Notice (this "*Grant Notice*"), and the provisions of the Plan and the Stock Option Agreement and the Notice of Exercise, all of which are made a part of this document. Unless otherwise provided in the Plan, this Grant Notice and the Stock Option Agreement (together, the "*Option Agreement*") may not be modified, amended or revised except in a writing signed by you and a duly authorized officer of the Company.
- [If the Option is an Incentive Stock Option, it (plus other outstanding Incentive Stock Options granted to you) cannot be first *exercisable* for more than \$100,000 in value (measured by exercise price) in any calendar year. Any excess over \$100,000 is a Nonstatutory Stock Option.]

- You consent to receive this Grant Notice, the Stock Option Agreement, the Plan, the Prospectus and any other Plan-related documents by
  electronic delivery and to participate in the Plan through an on-line or electronic system established and maintained by the Company or another
  third party designated by the Company.
- You have read and are familiar with the provisions of the Plan, the Stock Option Agreement, the Notice of Exercise and the Prospectus. In the event of any conflict between the provisions in this Grant Notice, the Option Agreement, the Notice of Exercise, or the Prospectus and the terms of the Plan, the terms of the Plan shall control.
- The Option Agreement sets forth the entire understanding between you and the Company regarding the acquisition of Common Stock and supersedes all prior oral and written agreements, promises and/or representations on that subject with the exception of other equity awards previously granted to you and any written employment agreement, offer letter, severance agreement, written severance plan or policy, or other written agreement between the Company and you in each case that specifies the terms that should govern this Option.
- Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, Uniform Electronic Transactions Act or other applicable law) or other transmission method and any counterpart so delivered will be deemed to have been duly and validly delivered and be valid and effective for all purposes.

WM TECHNOLOGY, INC.

**O**PTIONHOLDER:

By:

Title:

Date:

Date:

Signature

ATTACHMENTS: Stock Option Agreement, 2021 Equity Incentive Plan, Notice of Exercise

Signature

### ATTACHMENT I

# WM TECHNOLOGY, INC. STOCK OPTION AGREEMENT (2021 EQUITY INCENTIVE PLAN)

As reflected by your Stock Option Grant Notice ("*Grant Notice*"), WM Technology, Inc. (the "*Company*") has granted you an option under the Company's 2021 Equity Incentive Plan (the "*Plan*") to purchase a number of shares of Common Stock at the exercise price indicated in your Grant Notice (the "*Option*"). Capitalized terms not explicitly defined in this Agreement but defined in the Grant Notice or the Plan shall have the meanings set forth in the Grant Notice or Plan, as applicable. The terms of your Option as specified in the Grant Notice and this Stock Option Agreement constitute your Option Agreement.

The general terms and conditions applicable to your Option are as follows:

**1. GOVERNING PLAN DOCUMENT.** Your Option is subject to all the provisions of the Plan, including but not limited to the provisions in:

(a) Section 7 regarding the impact of a Capitalization Adjustment, dissolution, liquidation, or Corporate Transaction on your Option;

(b) Section 10(e) regarding the Company's retained rights to terminate your Continuous Service notwithstanding the grant of the Option; and

(c) Section 9(c) regarding the tax consequences of your Option.

Your Option is further subject to all interpretations, amendments, rules and regulations, which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the Option Agreement and the provisions of the Plan, the provisions of the Plan shall control.

2. Exercise.

(a) You may generally exercise the vested portion of your Option for whole shares of Common Stock at any time during its term by delivery of payment of the exercise price and applicable withholding taxes and other required documentation to the Plan Administrator in accordance with the exercise procedures established by the Plan Administrator, which may include an electronic submission. Please review Sections 4(i), 4(j) and 8(b)(v) of the Plan, which may restrict or prohibit your ability to exercise your Option during certain periods.

(b) To the extent permitted by Applicable Law, you may pay your Option exercise price as follows:

(i) cash, check, bank draft or money order;

(ii) subject to Company and/or Committee consent at the time of exercise, pursuant to a "cashless exercise" program as further described in Section 4(c)(ii) of the Plan if at the time of exercise the Common Stock is publicly traded;

(iii) subject to Company and/or Committee consent at the time of exercise, by delivery of previously owned shares of Common Stock as further described in Section 4(c)(iii) of the Plan; or

(iv) subject to Company and/or Committee consent at the time of exercise, if the Option is a Nonstatutory Stock Option, by a "net exercise" arrangement as further described in Section 4(c)(iv) of the Plan.

(c) By accepting your Option, you agree that you will not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale with respect to any shares of Common Stock or other securities of the Company held by you, for a period of one hundred eighty (180) days following the effective date of a registration statement of the Company filed under the Securities Act or such longer period as the underwriters or the Company will request to facilitate compliance with FINRA Rule 2241 or any successor or similar rules or regulation-(the "*Lock-Up Period*"); *provided, however*, that nothing contained in this section will prevent the exercise of a repurchase option, if any, in favor of the Company during the Lock-Up Period. You further agree to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to your shares of Common Stock until the end of such period. You also agree that any transferee of any shares of Common Stock (or other securities) of the Company held by you will be bound by this Section 2(c). The underwriters of the Company's stock are intended third party beneficiaries of this Section 2(c) and will have the right, power and authority to enforce the provisions hereof as though they were a party hereto.

**3. TERM.** You may not exercise your Option before the commencement of its term or after its term expires. The term of your Option commences on the Date of Grant and expires upon the earliest of the following:

(a) immediately upon the termination of your Continuous Service for Cause;

(b) three months after the termination of your Continuous Service for any reason other than Cause, Disability or death;

(c) 12 months after the termination of your Continuous Service due to your Disability;

(d) 18 months after your death if you die during your Continuous Service;

(e) immediately upon a Corporate Transaction if the Board has determined that the Option will terminate in connection with a Corporate Transaction,

(f) the Expiration Date indicated in your Grant Notice; or

(g) the day before the 10th anniversary of the Date of Grant.

Notwithstanding the foregoing, if you die during the period provided in Section 3(b) or 3(c) above, the term of your Option shall not expire until the earlier of (i) 18 months after your death, (ii) upon any termination of the Option in connection with a Corporate Transaction, (iii) the Expiration Date indicated in your Grant Notice, or (iv) the day before the tenth anniversary of the Date of Grant. Additionally, the Post-Termination Exercise Period of your Option may be extended as provided in Section 4(i) of the Plan.

To obtain the federal income tax advantages associated with an Incentive Stock Option, the Code requires that at all times beginning on the date of grant of your Option and ending on the day three months before the date of your Option's exercise, you must be an employee of the Company or an Affiliate, except in the event of your death or Disability. If the Company provides for the extended exercisability of your Option under certain circumstances for your benefit, your Option will not necessarily be treated as an Incentive Stock Option if you exercise your Option more than three months after the date your employment terminates.

4. WITHHOLDING OBLIGATIONS. As further provided in Section 9 of the Plan: (a) you may not exercise your Option unless the applicable tax withholding obligations are satisfied, and (b) at the time you exercise your Option, in whole or in part, or at any time thereafter as requested by the Company, you hereby authorize withholding from payroll and any other amounts payable to you, and otherwise agree to make adequate provision for (including by means of a "cashless exercise" pursuant to a program developed under Regulation T as promulgated by the Federal Reserve Board to the extent permitted by the Company), any sums required to satisfy the federal, state, local and foreign tax withholding obligations, if any, which arise in connection with the exercise of your Option in accordance with the withholding procedures established by the Company. Accordingly, you may not be able to exercise your Option, unless and until such obligations are satisfied. In the event that the amount of the Company's withholding obligation in connection with your Option was greater than the amount actually withheld by the Company, you agree to indemnify and hold the Company harmless from any failure by the Company to withhold the proper amount.

**5. INCENTIVE STOCK OPTION DISPOSITION REQUIREMENT.** If your Option is an Incentive Stock Option, you must notify the Company in writing within 15 days after the date of any disposition of any of the shares of the Common Stock issued upon exercise of your Option that occurs within two years after the date of your Option grant or within one year after such shares of Common Stock are transferred upon exercise of your Option.

**6. TRANSFERABILITY.** Except as otherwise provided in Section 4(e) of the Plan, your Option is not transferable, except by will or by the applicable laws of descent and distribution, and is exercisable during your life only by you.

**7. CORPORATE TRANSACTION.** Your Option is subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on your behalf with respect to any escrow, indemnities and any contingent consideration.

8. No LIABILITY FOR TAXES. As a condition to accepting the Option, you hereby (a) agree to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from the Option or other Company compensation and (b) acknowledge that you were advised to consult with your own personal tax, financial and other legal advisors regarding the tax consequences of the Option and have either done so or knowingly and voluntarily declined to do so. Additionally, you acknowledge that the Option is exempt from Section 409A only if the exercise price is at least equal to the "fair market value" of the Common Stock on the date of grant as determined by the Internal Revenue Service and there is no other impermissible deferral of compensation associated with the Option. Additionally, as a condition to accepting the Option, you agree not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates in the event that the Internal Revenue Service asserts that such exercise is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the Internal Revenue Service asserts that such exercise is less than the "fair market value" of the Common Stock on the date of grant as subsequently determined by the Internal Revenue Service.

**9. SEVERABILITY.** If any part of this Option Agreement or the Plan is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity will not invalidate any portion of this Option Agreement or the Plan not declared to be unlawful or invalid. Any Section of this Option Agreement (or part of such a Section) so declared to be unlawful or invalid will, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid

**10. OTHER DOCUMENTS.** You hereby acknowledge receipt of or the right to receive a document providing the information required by Rule 428(b)(1) promulgated under the Securities Act, which includes the Prospectus. In addition, you acknowledge receipt of the Company's Trading Policy.

**11. Q**UESTIONS. If you have questions regarding these or any other terms and conditions applicable to your Option, including a summary of the applicable federal income tax consequences please see the Prospectus.

\* \* \* \*

ATTACHMENT II

**2021 EQUITY INCENTIVE PLAN** 

# ATTACHMENT III

# WM TECHNOLOGY, INC. NOTICE OF EXERCISE (2021 Equity Incentive Plan)

WM Technology, Inc. 41 Discovery Irvine, California 92618

This constitutes notice to WM Technology, Inc. (the "*Company*") that I elect to purchase the below number of shares of Common Stock of the Company (the "*Shares*") by exercising my Option for the price set forth below. Capitalized terms not explicitly defined in this Notice of Exercise but defined in the Stock Option Grant Notice, Stock Option Agreement or 2021 Equity Incentive Plan (the "*Plan*") shall have the meanings set forth in the Stock Option Grant Notice, Stock Option Agreement or Plan, as applicable. Use of certain payment methods is subject to Company and/or Committee consent and certain additional requirements set forth in the Stock Option Agreement and the Plan.

Type of option (check one):	Incentive	Nonstatutory
Date of Grant:		
Number of Shares as to which Option is exercised:		
Certificates to be issued in name of:		
Total exercise price:	\$	
Cash, check, bank draft or money order delivered herewith:	\$	
Value of Shares delivered herewith:	\$	
Regulation T Program (cashless exercise)	\$	
Value of Shares pursuant to net exercise:	\$	

Date of Exercise: \_\_\_\_\_

By this exercise, I agree (i) to provide such additional documents as you may require pursuant to the terms of the Plan, (ii) to satisfy the tax withholding obligations, if any, relating to the exercise of this Option as set forth in the Stock Option Agreement, and (iii) if this exercise relates to an incentive stock option, to notify you in writing within 15 days after the date of any disposition of any of the Shares issued upon exercise of this Option that occurs within two years after the Date of Grant or within one year after such Shares are issued upon exercise of this Option.

I further agree that, if required by the Company (or a representative of the underwriters) in connection with the first underwritten registration of the offering of any securities of the Company under the Securities Act, I will not sell, dispose of, transfer, make any short sale of, grant any option for the purchase of, or enter into any hedging or similar transaction with the same economic effect as a sale with respect to any shares of Common Stock or other securities of the Company for a period of one hundred eighty (180) days following the effective date of a registration statement of the Company filed under the Securities Act (or such longer period as the underwriters or the Company shall request to facilitate compliance with FINRA Rule 2241 or any successor or similar rule or regulation) (the "*Lock-Up Period*"). I further agree to execute and deliver such other agreements as may be reasonably requested by the Company or the underwriters that are consistent with the foregoing or that are necessary to give further effect thereto. In order to enforce the foregoing covenant, the Company may impose stop-transfer instructions with respect to securities subject to the foregoing restrictions until the end of such period.

Very truly yours,

# WM TECHNOLOGY, INC. RSU Award Grant Notice (WM Technology, Inc. 2021 Equity Incentive Plan)

WM Technology, Inc. (the "*Company*") has awarded to you (the "*Participant*") the number of restricted stock units specified and on the terms set forth below in consideration of your services (the "*RSU Award*"). Your RSU Award is subject to all of the terms and conditions as set forth herein and in the WM Technology, Inc. 2021 Equity Incentive Plan (the "*Plan*") and the Award Agreement (the "*Agreement*"), which are attached hereto and incorporated herein in their entirety. Capitalized terms not explicitly defined herein but defined in the Plan or the Agreement shall have the meanings set forth in the Plan or the Agreement.

Participant:			
Date of Grant:			
Vesting Commencement	Date:		
Number of Restricted Ste	ock Units:		
Vesting Schedule:	[		].
	Mate sith standing the four gains and	ting shall to main stars and the Deutisia and	where the second s

Notwithstanding the foregoing, vesting shall terminate upon the Participant's termination of Continuous Service.

**Issuance Schedule:** One share of Common Stock will be issued at the time set forth in Section 5 of the Agreement for each restricted stock unit which vests.

**Participant Acknowledgements:** By your signature below or by electronic acceptance or authentication in a form authorized by the Company, you understand and agree that:

- The RSU Award is governed by this RSU Award Grant Notice (the "*Grant Notice*"), and the provisions of the Plan and the Agreement, all of which are made a part of this document. Unless otherwise provided in the Plan, this Grant Notice and the Agreement (together, the "*RSU Award Agreement*") may not be modified, amended or revised except in a writing signed by you and a duly authorized officer of the Company.
- You have read and are familiar with the provisions of the Plan, the RSU Award Agreement and the Prospectus. In the event of any conflict between the provisions in the RSU Award Agreement, or the Prospectus and the terms of the Plan, the terms of the Plan shall control.
- The RSU Award Agreement sets forth the entire understanding between you and the Company regarding the acquisition of Common Stock and supersedes all prior oral and written agreements, promises and/or representations on that subject with the exception of: (i) other equity awards previously granted to you, and (ii) any written employment agreement, offer letter, severance agreement, written severance plan or policy, or other written agreement between the Company and you in each case that specifies the terms that should govern this RSU Award.

WM TECHNOLOGY, INC.:

By:				
	Signature		Signature	
Title:		Date:		
Date:				
ATTACHMENTS:	RSU Award Agreement, 2021 Equity Incentive Plan			

**PARTICIPANT:** 

### WM TECHNOLOGY, INC. 2021 Equity Incentive Plan

# AWARD AGREEMENT (RSU AWARD)

As reflected by your RSU Award Grant Notice ("*Grant Notice*"), WM Technology, Inc. (the "*Company*") has granted you a RSU Award under the WM Technology, Inc. 2021 Equity Incentive Plan (the "*Plan*") for the number of restricted stock units as indicated in your Grant Notice (the "*RSU Award*"). The terms of your RSU Award as specified in this Award Agreement for your RSU Award (the "*Agreement*") and the Grant Notice constitute your "*RSU Award Agreement*". Defined terms not explicitly defined in this Agreement but defined in the Grant Notice or the Plan shall have the same definitions as in the Grant Notice or Plan, as applicable.

The general terms applicable to your RSU Award are as follows:

**1. GOVERNING PLAN DOCUMENT.** Your RSU Award is subject to all the provisions of the Plan, including but not limited to the provisions in:

(a) Section 7 of the Plan regarding the impact of a Capitalization Adjustment, dissolution, liquidation, or Corporate Transaction on your RSU Award;

(b) Section 10(e) of the Plan regarding the Company's retained rights to terminate your Continuous Service notwithstanding the grant of the RSU Award; and

(c) Section 9(c) of the Plan regarding the tax consequences of your RSU Award.

Your RSU Award is further subject to all interpretations, amendments, rules and regulations, which may from time to time be promulgated and adopted pursuant to the Plan. In the event of any conflict between the RSU Award Agreement and the provisions of the Plan, the provisions of the Plan shall control.

2. GRANT OF THE RSU AWARD. This RSU Award represents your right to be issued on a future date the number of shares of the Company's Common Stock that is equal to the number of restricted stock units indicated in the Grant Notice as modified to reflect any Capitalization Adjustment and subject to your satisfaction of the vesting conditions set forth therein (the "*Restricted Stock Units*"). Any additional Restricted Stock Units that become subject to the RSU Award pursuant to Capitalization Adjustments as set forth in the Plan and the provisions of Section 3 below, if any, shall be subject, in a manner determined by the Board, to the same forfeiture restrictions, restrictions on transferability, and time and manner of delivery as applicable to the other Restricted Stock Units covered by your RSU Award.

**3. DIVIDENDS.** You shall receive no benefit or adjustment to this RSU Award with respect to any cash dividend, stock dividend or other distribution that does not result from a Capitalization Adjustment; provided, however, that this sentence will not apply with respect to any shares of Common Stock that are delivered to you in connection with your RSU Award after such shares have been delivered to you.

4. WITHHOLDING OBLIGATIONS. As further provided in Section 9 of the Plan, you hereby authorize withholding from payroll and any other amounts payable to you, and otherwise agree to make adequate provision for, any sums required to satisfy the federal, state, local and foreign tax withholding obligations, if any, which arise in connection with your RSU Award (the "*Withholding Obligation*") in accordance with the withholding procedures established by the Company. Unless the Withholding Obligation is satisfied, the Company shall have no obligation to deliver to you any Common Stock in respect of the RSU Award. In the event the Withholding Obligation of the Company arises prior to the delivery to you of Common Stock or it is determined after the delivery of Common Stock to you that the amount of the Withholding Obligation was greater than the amount withheld by the Company, you agree to indemnify and hold the Company harmless from any failure by the Company to withhold the proper amount.

# 5. DATE OF ISSUANCE.

(a) The issuance of shares in respect of the Restricted Stock Units is intended to comply with Treasury Regulations Section 1.409A-1(b)(4) and will be construed and administered in such a manner. Subject to the satisfaction of the Withholding Obligation, if any, in the event one or more Restricted Stock Units vests, the Company shall issue to you one (1) share of Common Stock for each Restricted Stock Unit that vests on the applicable vesting date(s) (subject to any adjustment under Section 3 above, and subject to any different provisions in the Grant Notice). Each issuance date determined by this paragraph is referred to as an "Original Issuance Date."

(b) If the Original Issuance Date falls on a date that is not a business day, delivery shall instead occur on the next following business day. In addition, if:

(i) the Original Issuance Date does not occur (1) during an "open window period" applicable to you, as determined by the Company in accordance with the Company's then-effective policy on trading in Company securities, or (2) on a date when you are otherwise permitted to sell shares of Common Stock on an established stock exchange or stock market (including but not limited to under a previously established written trading plan that meets the requirements of Rule 10b5-1 under the Exchange Act and was entered into in compliance with the Company's policies (a "10b5-1 Arrangement)), and

(ii) either (1) a Withholding Obligation does not apply, or (2) the Company decides, prior to the Original Issuance Date, (A) not to satisfy the Withholding Obligation by withholding shares of Common Stock from the shares otherwise due, on the Original Issuance Date, to you under this Award, and (B) not to permit you to enter into a "same day sale" commitment with a broker-dealer (including but not limited to a commitment under a 10b5-1 Arrangement) and (C) not to permit you to pay your Withholding Obligation in cash,

(iii) then the shares that would otherwise be issued to you on the Original Issuance Date will not be delivered on such Original Issuance Date and will instead be delivered on the first business day when you are not prohibited from selling shares of the Company's Common Stock in the open public market, but in no event later than December 31 of the calendar year in which the Original Issuance Date occurs (that is, the last day of your taxable year in which the Original Issuance Date occurs), or, <u>if and only if</u> permitted in a manner that complies with Treasury Regulations Section 1.409A-1(b)(4), no later than the date that is the 15th day of the third calendar month of the applicable year following the year in which the shares of Common Stock under this Award are no longer subject to a "substantial risk of forfeiture" within the meaning of Treasury Regulations Section 1.409A-1(d).

apply.

(c)

To the extent the RSU Award is a Non-Exempt RSU Award, the provisions of Section 12 of the Plan shall

**6. TRANSFERABILITY.** Except as otherwise provided in the Plan, your RSU Award is not transferable, except by will or by the applicable laws of descent and distribution.

**7. CORPORATE TRANSACTION.** Your RSU Award is subject to the terms of any agreement governing a Corporate Transaction involving the Company, including, without limitation, a provision for the appointment of a stockholder representative that is authorized to act on your behalf with respect to any escrow, indemnities and any contingent consideration.

8. No LIABILITY FOR TAXES. As a condition to accepting the RSU Award, you hereby (a) agree to not make any claim against the Company, or any of its Officers, Directors, Employees or Affiliates related to tax liabilities arising from the RSU Award or other Company compensation and (b) acknowledge that you were advised to consult with your own personal tax, financial and other legal advisors regarding the tax consequences of the RSU Award and have either done so or knowingly and voluntarily declined to do so.

**9. SEVERABILITY.** If any part of this Agreement or the Plan is declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity will not invalidate any portion of this Agreement or the Plan not declared to be unlawful or invalid. Any Section of this Agreement (or part of such a Section) so declared to be unlawful or invalid will, if possible, be construed in a manner which will give effect to the terms of such Section or part of a Section to the fullest extent possible while remaining lawful and valid.

**10. OTHER DOCUMENTS.** You hereby acknowledge receipt of or the right to receive a document providing the information required by Rule 428(b)(1) promulgated under the Securities Act, which includes the Prospectus. In addition, you acknowledge receipt of the Company's Trading Policy.

**11. QUESTIONS.** If you have questions regarding these or any other terms and conditions applicable to your RSU Award, including a summary of the applicable federal income tax consequences please see the Prospectus.

### WM TECHNOLOGY, INC.

# 2021 Employee Stock Purchase Plan

# Adopted by the Board of Directors: June 16, 2021

#### Approved by the Stockholders: June 10, 2021

#### 1. GENERAL; PURPOSE.

(a) The Plan provides a means by which Eligible Employees of the Company and certain Designated Companies may be given an opportunity to purchase shares of Common Stock. The Plan permits the Company to grant a series of Purchase Rights to Eligible Employees under an Employee Stock Purchase Plan.

(b) The Plan includes two components: a 423 Component and a Non-423 Component. The Company intends (but makes no undertaking or representation to maintain) the 423 Component to qualify as an Employee Stock Purchase Plan. The provisions of the 423 Component, accordingly, will be construed in a manner that is consistent with the requirements of Section 423 of the Code. In addition, this Plan authorizes grants of Purchase Rights under the Non-423 Component that do not meet the requirements of an Employee Stock Purchase Plan. Except as otherwise provided in the Plan or determined by the Board, the Non-423 Component will operate and be administered in the same manner as the 423 Component. In addition, the Company may make separate Offerings which vary in terms (provided that such terms are not inconsistent with the provisions of the Plan or the requirements of an Employee Stock Purchase Plan to the extent the Offering is made under the 423 Component), and the Company will designate which Designated Company is participating in each separate Offering.

(c) The Company, by means of the Plan, seeks to retain the services of Eligible Employees, to secure and retain the services of new Employees and to provide incentives for such persons to exert maximum efforts for the success of the Company and its Related Corporations.

#### 2. Administration.

(a) The Board will administer the Plan unless and until the Board delegates administration of the Plan to a Committee or Committees, as provided in Section 2(c).

(b) The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:

(i) To determine how and when Purchase Rights will be granted and the provisions of each Offering (which need not be identical).

(ii) To designate from time to time (A) which Related Corporations of the Company will be eligible to participate in the Plan as Designated 423 Companies, (B) which Related Corporations or Affiliates will be eligible to participate in the Plan as Designated Non-423 Companies, (C) which Affiliates or Related Corporations may be excluded from participation in the Plan, and (D) which Designated Companies will participate in each separate Offering (to the extent that the Company makes separate Offerings).

(iii) To construe and interpret the Plan and Purchase Rights, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan, in a manner and to the extent it deems necessary or expedient to make the Plan fully effective.

(iv) To settle all controversies regarding the Plan and Purchase Rights granted under the Plan.

(v) To suspend or terminate the Plan at any time as provided in Section 12.

(vi) To amend the Plan at any time as provided in Section 12.

(vii) Generally, to exercise such powers and to perform such acts as it deems necessary or expedient to promote the best interests of the Company and its Related Corporations and to carry out the intent that the Plan be treated as an Employee Stock Purchase Plan with respect to the 423 Component.

(viii) To adopt such procedures and sub-plans as are necessary or appropriate to permit participation in the Plan by Employees who are foreign nationals or employed outside the United States.

(c) The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to a subcommittee any of the administrative powers the Committee is authorized to exercise (and references to the Board in this Plan and in any applicable Offering Document will thereafter be to the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. The Board may retain the authority to concurrently administer the Plan with the Committee and may, at any time, revest in the Board some or all of the powers previously delegated. Whether or not the Board has delegated administration of the Plan to a Committee, the Board will have the final power to determine all questions of policy and expediency that may arise in the administration of the Plan.

(d) All determinations, interpretations and constructions made by the Board will not be subject to review by any person and will be final, binding and conclusive on all persons.

3. SHARES OF COMMON STOCK SUBJECT TO THE PLAN.

(a) Subject to the provisions of Section 11(a) relating to Capitalization Adjustments, the maximum number of shares of Common Stock that may be issued under the Plan will not exceed 2,619,543 shares of Common Stock (equal to one and one-half percent (1.5%) of the sum of (i) the number of shares of Common Stock outstanding as of the consummation of the transactions contemplated by the Merger Agreement and (ii) the number of shares of Common Stock underlying securities convertible into Common Stock) (the "*Initial Share Reserve*"), plus the number of shares of Common Stock that are automatically added on January 1st of each year for a period of up to ten years, commencing on January 1, 2022 and ending on (and including) January 1, 2031, in an amount equal to the lesser of (x) one percent (1.0%) of the total number of shares of Capital Stock outstanding on December 31st of the preceding calendar year, and (y) 5,239,086 shares of Common Stock (equal to two hundred percent (200%) of the Initial Share Reserve). Notwithstanding the foregoing, the Board may act prior to the first day of any calendar year to provide that there will be no January 1<sup>st</sup> increase in the share reserve for such calendar year will be a lesser number of shares of Common Stock than would otherwise occur pursuant to the preceding sentence.

(b) If any Purchase Right granted under the Plan terminates without having been exercised in full, the shares of Common Stock not purchased under such Purchase Right will again become available for issuance under the Plan.

(c) The stock purchasable under the Plan will be shares of authorized but unissued or reacquired Common Stock, including shares repurchased by the Company on the open market.

### 4. GRANT OF PURCHASE RIGHTS; OFFERING.

(a) The Board may from time to time grant or provide for the grant of Purchase Rights to Eligible Employees under an Offering (consisting of one or more Purchase Periods) on an Offering Date or Offering Dates selected by the Board. Each Offering will be in such form and will contain such terms and conditions as the Board will deem appropriate, and with respect to the 423 Component, will comply with the requirement of Section 423(b)(5) of the Code that all Employees granted Purchase Rights will have the same rights and privileges. The terms and conditions of an Offering shall be incorporated by reference into the Plan and treated as part of the Plan. The provisions of separate Offerings need not be identical, but each Offering will include (through incorporation of the provisions of this Plan by reference in the document comprising the Offering or otherwise) the period during which the Offering will be effective, which period will not exceed 27 months beginning with the Offering Date, and the substance of the provisions contained in Sections 5 through 8, inclusive.

(b) If a Participant has more than one Purchase Right outstanding under the Plan, unless he or she otherwise indicates in forms delivered to the Company: (i) each form will apply to all of his or her Purchase Rights under the Plan, and (ii) a Purchase Right with a lower exercise price (or an earlier-granted Purchase Right, if different Purchase Rights have identical exercise prices) will be exercised to the fullest possible extent before a Purchase Right with a higher exercise price (or a later-granted Purchase Right if different Purchase Rights have identical exercise prices) will be exercised to the fullest possible extent before a Purchase Right with a higher exercise price (or a later-granted Purchase Right if different Purchase Rights have identical exercise prices) will be exercised.

(c) The Board will have the discretion to structure an Offering so that if the Fair Market Value of a share of Common Stock on the first Trading Day of a new Purchase Period within that Offering is less than or equal to the Fair Market Value of a share of Common Stock on the Offering Date for that Offering, then (i) that Offering will terminate immediately as of that first Trading Day, and (ii) the Participants in such terminated Offering will be automatically enrolled in a new Offering beginning on the first Trading Day of such new Purchase Period.

# 5. ELIGIBILITY.

(a) Purchase Rights may be granted only to Employees of the Company or, as the Board may designate in accordance with Section 2(b), to Employees of a Related Corporation or an Affiliate. Except as provided in Section 5(b), an Employee will not be eligible to be granted Purchase Rights unless, on the Offering Date, the Employee has been in the employ of the Company, the Related Corporation or the Affiliate, as the case may be, for such continuous period preceding such Offering Date as the Board may require, but in no event will the required period of continuous employment be equal to or greater than two years. In addition, the Board may provide that no Employee will be eligible to be granted Purchase Rights unless, on the Offering Date, such Employee's customary employment with the Company, the Related Corporation or the Affiliate, as applicable, is more than 20 hours per week and more than five months per calendar year or such other criteria as the Board may determine consistent with Section 423 of the Code with respect to the 423 Component.

(b) The Board may provide that each person who, during the course of an Offering, first becomes an Eligible Employee will, on a date or dates specified in the Offering which coincides with the day on which such person becomes an Eligible Employee or which occurs thereafter, receive a Purchase Right under that Offering, which Purchase Right will thereafter be deemed to be a part of that Offering. Such Purchase Right will have the same characteristics as any Purchase Rights originally granted under that Offering, as described herein, except that:

(i) the date on which such Purchase Right is granted will be the "Offering Date" of such Purchase Right for all purposes, including determination of the exercise price of such Purchase Right;

(ii) the period of the Offering with respect to such Purchase Right will begin on its Offering Date and end coincident with the end of such Offering; and

(iii) the Board may provide that if such person first becomes an Eligible Employee within a specified period of time before the end of the Offering, he or she will not receive any Purchase Right under that Offering.

(c) No Employee will be eligible for the grant of any Purchase Rights under the 423 Component if, immediately after any such Purchase Rights are granted, such Employee owns stock possessing five percent or more of the total combined voting power or value of all classes of stock of the Company or of any Related Corporation. For purposes of this Section 5(c), the rules of Section 424(d) of the Code will apply in determining the stock ownership of any Employee, and stock which such Employee may purchase under all outstanding Purchase Rights and options will be treated as stock owned by such Employee.

(d) As specified by Section 423(b)(8) of the Code, an Eligible Employee may be granted Purchase Rights under the 423 Component only if such Purchase Rights, together with any other rights granted under all Employee Stock Purchase Plans of the Company and any Related Corporations, do not permit such Eligible Employee's rights to purchase stock of the Company or any Related Corporation to accrue at a rate which, when aggregated, exceeds \$25,000 of Fair Market Value of such stock (determined at the time such rights are granted, and which, with respect to the Plan, will be determined as of their respective Offering Dates) for each calendar year in which such rights are outstanding at any time.

(e) Officers of the Company and any Designated Company, if they are otherwise Eligible Employees, will be eligible to participate in Offerings under the Plan. Notwithstanding the foregoing, the Board may provide in an Offering that Employees who are highly compensated Employees within the meaning of Section 423(b)(4)(D) of the Code will not be eligible to participate.

(f) Notwithstanding anything in this Section 5 to the contrary, in the case of an Offering under the Non-423 Component, an Eligible Employee (or group of Eligible Employees) may be excluded from participation in the Plan or an Offering if the Board has determined, in its sole discretion, that participation of such Eligible Employee(s) is not advisable or practical for any reason.

6. PURCHASE RIGHTS; PURCHASE PRICE.

(a) On each Offering Date, each Eligible Employee, pursuant to an Offering made under the Plan, will be granted a Purchase Right to purchase up to that number of shares of Common Stock purchasable either with a percentage or with a maximum dollar amount, as designated by the Board, but in either case not exceeding 15% of such Employee's earnings (as defined by the Board in each Offering) during the period that begins on the Offering Date (or such later date as the Board determines for a particular Offering) and ends on the date stated in the Offering, which date will be no later than the end of the Offering.

(b) The Board will establish one or more Purchase Dates during an Offering on which Purchase Rights granted for that Offering will be exercised and shares of Common Stock will be purchased in accordance with such Offering.

(c) In connection with each Offering made under the Plan, the Board may specify (i) a maximum number of shares of Common Stock that may be purchased by any Participant on any Purchase Date during such Offering, (ii) a maximum aggregate number of shares of Common Stock that may be purchased by all Participants pursuant to such Offering and/or (iii) a maximum aggregate number of shares of Common Stock that may be purchased by all Participants on any Purchase Date under the Offering. If the aggregate purchase of shares of Common Stock issuable upon exercise of Purchase Rights granted under the Offering would exceed any such maximum aggregate number, then, in the absence of any Board action otherwise, a pro rata (based on each Participant's accumulated Contributions) allocation of the shares of Common Stock available will be made in as nearly a uniform manner as will be practicable and equitable.

(d) The purchase price of shares of Common Stock acquired pursuant to Purchase Rights will be not less than the lesser of:

(i) an amount equal to 85% of the Fair Market Value of the shares of Common Stock on the Offering Date;

or

Purchase Date.

an amount equal to 85% of the Fair Market Value of the shares of Common Stock on the applicable

# 7. PARTICIPATION; WITHDRAWAL; TERMINATION.

(ii)

(a) An Eligible Employee may elect to participate in an Offering and authorize payroll deductions as the means of making Contributions by completing and delivering to the Company, within the time specified in the Offering, an enrollment form provided by the Company. The enrollment form will specify the amount of Contributions not to exceed the maximum amount specified by the Board. Each Participant's Contributions will be credited to a bookkeeping account for such Participant under the Plan and will be deposited with the general funds of the Company except where applicable law requires that Contributions be deposited with a third party. If permitted in the Offering, a Participant may begin such Contributions with the first practicable payroll occurring on or after the Offering Date (or, in the case of a payroll date that occurs after the end of the prior Offering but before the Offering, a Participant may thereafter reduce (including to zero) or increase his or her Contributions. If specifically provided in the Offering, in addition to or instead of making Contributions by payroll deductions, a Participant may make Contributions through the payment by cash or check prior to a Purchase Date.

(b) During an Offering, a Participant may cease making Contributions and withdraw from the Offering by delivering to the Company a withdrawal form provided by the Company. The Company may impose a deadline before a Purchase Date for withdrawing. Upon such withdrawal, such Participant's Purchase Right in that Offering will immediately terminate and the Company will distribute as soon as practicable to such Participant all of his or her accumulated but unused Contributions and such Participant's Purchase Right in that Offering shall thereupon terminate. A Participant's withdrawal from that Offering will have no effect upon his or her eligibility to participate in any other Offerings under the Plan, but such Participant will be required to deliver a new enrollment form to participate in subsequent Offerings.

(c) Unless otherwise required by applicable law, Purchase Rights granted pursuant to any Offering under the Plan will terminate immediately if the Participant either (i) is no longer an Employee for any reason or for no reason (subject to any post-employment participation period required by law) or (ii) is otherwise no longer eligible to participate. The Company will distribute to such individual as soon as practicable all of his or her accumulated but unused Contributions.

(d) Unless otherwise determined by the Board, a Participant whose employment transfers or whose employment terminates with an immediate rehire (with no break in service) by or between the Company and a Designated Company or between Designated Companies will not be treated as having terminated employment for purposes of participating in the Plan or an Offering; however, if a Participant transfers from an Offering under the 423 Component to an Offering under the Non-423 Component, the exercise of the Participant's Purchase Right will be qualified under the 423 Component only to the extent such exercise complies with Section 423 of the Code. If a Participant transfers from an Offering under the Non-423 Component to an Offering under the 423 Component, the exercise of the Purchase Right will remain non-qualified under the Non-423 Component. The Board may establish different and additional rules governing transfers between separate Offerings within the 423 Component and Offerings under the 423 Component.

(e) During a Participant's lifetime, Purchase Rights will be exercisable only by such Participant. Purchase Rights are not transferable by a Participant, except by will, by the laws of descent and distribution, or, if permitted by the Company, by a beneficiary designation as described in Section 10.

(f) Unless otherwise specified in the Offering or required by applicable law, the Company will have no obligation to pay interest on Contributions.

# 8. Exercise of Purchase Rights.

(a) On each Purchase Date, each Participant's accumulated Contributions will be applied to the purchase of shares of Common Stock, up to the maximum number of shares of Common Stock permitted by the Plan and the applicable Offering, at the purchase price specified in the Offering. No fractional shares will be issued unless specifically provided for in the Offering.

(b) Unless otherwise provided in the Offering, if any amount of accumulated Contributions remains in a Participant's account after the purchase of shares of Common Stock and such remaining amount is less than the amount required to purchase one share of Common Stock on the final Purchase Date of an Offering, then such remaining amount will be held in such Participant's account for the purchase of shares of Common Stock under the next Offering under the Plan, unless such Participant withdraws from or is not eligible to participate in such next Offering, in which case such amount will be distributed to such Participant after the final Purchase Date without interest (unless the payment of interest is otherwise required by applicable law). If the amount of Contributions remaining in a Participant's account after the purchase of shares of Common Stock on the final Purchase Date of an Offering, then such remaining amount required to purchase one (1) whole share of Common Stock on the final Purchase Date of an Offering, then such remaining amount will be distributed in full to such Participant after the final Purchase Date of an Offering, then such remaining amount will be distributed in full to such Participant after the final Purchase Date of such Offering without interest.

(c) No Purchase Rights may be exercised to any extent unless the shares of Common Stock to be issued upon such exercise under the Plan are covered by an effective registration statement pursuant to the Securities Act and the Plan is in material compliance with all applicable federal, state, foreign and other securities and other laws applicable to the Plan. If on a Purchase Date the shares of Common Stock are not so registered or the Plan is not in such compliance, no Purchase Rights will be exercised on such Purchase Date, and the Purchase Date will be delayed until the shares of Common Stock are subject to such an effective registration statement and the Plan is in material compliance, except that the Purchase Date will in no event be more than 6 months from the Offering Date. If, on the Purchase Date, as delayed to the maximum extent permissible, the shares of Common Stock are not registered and the Plan is not in material compliance with all applicable laws, no Purchase Rights will be exercised and all accumulated but unused Contributions will be distributed to the Participants without interest.

#### 9. COVENANTS OF THE COMPANY.

The Company will seek to obtain from each U.S. federal or state, foreign or other regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Purchase Rights and issue and sell shares of Common Stock thereunder unless the Company determines, in its sole discretion, that doing so would cause the Company to incur costs that are unreasonable. If, after commercially reasonable efforts, the Company is unable to obtain the authority that counsel for the Company deems necessary for the grant of Purchase Rights or the lawful issuance and sale of Common Stock under the Plan, and at a commercially reasonable cost, the Company will be relieved from any liability for failure to grant Purchase Rights and/or to issue and sell Common Stock upon exercise of such Purchase Rights.

#### **10. Designation of Beneficiary.**

(a) The Company may, but is not obligated to, permit a Participant to submit a form designating a beneficiary who will receive any shares of Common Stock and/or Contributions from the Participant's account under the Plan if the Participant dies before such shares and/or Contributions are delivered to the Participant. The Company may, but is not obligated to, permit the Participant to change such designation of beneficiary. Any such designation and/or change must be on a form approved by the Company.

(b) If a Participant dies, and in the absence of a valid beneficiary designation, the Company will deliver any shares of Common Stock and/or Contributions to the executor or administrator of the estate of the Participant. If no executor or administrator has been appointed (to the knowledge of the Company), the Company, in its sole discretion, may deliver such shares of Common Stock and/or Contributions without interest (unless the payment of interest is otherwise required by applicable law) to the Participant's spouse, dependents or relatives, or if no spouse, dependent or relative is known to the Company, then to such other person as the Company may designate.

# 11. Adjustments upon Changes in Common Stock; Corporate Transactions.

(a) In the event of a Capitalization Adjustment, the Board will appropriately and proportionately adjust: (i) the class(es) and maximum number of securities subject to the Plan pursuant to Section 3(a), (ii) the class(es) and maximum number of securities by which the share reserve is to increase automatically each year pursuant to Section 3(a), (iii) the class(es) and number of securities subject to, and the purchase price applicable to outstanding Offerings and Purchase Rights, and (iv) the class(es) and number of securities that are the subject of the purchase limits under each ongoing Offering. The Board will make these adjustments, and its determination will be final, binding and conclusive.

(b) In the event of a Corporate Transaction, then: (i) any surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) may assume or continue outstanding Purchase Rights or may substitute similar rights (including a right to acquire the same consideration paid to the stockholders in the Corporate Transaction) for outstanding Purchase Rights, or (ii) if any surviving or acquiring corporation (or its parent company) does not assume or continue such Purchase Rights or does not substitute similar rights for such Purchase Rights, then the Participants' accumulated Contributions will be used to purchase shares of Common Stock within ten business days prior to the Corporate Transaction under the outstanding Purchase Rights, and the Purchase Rights will terminate immediately after such purchase.

#### 12. Amendment, Termination or Suspension of the Plan.

(a) The Board may amend the Plan at any time in any respect the Board deems necessary or advisable. However, except as provided in Section 11(a) relating to Capitalization Adjustments, stockholder approval will be required for any amendment of the Plan for which stockholder approval is required by applicable law or listing requirements.

(b) The Board may suspend or terminate the Plan at any time. No Purchase Rights may be granted under the Plan while the Plan is suspended or after it is terminated.

(c) Any benefits, privileges, entitlements and obligations under any outstanding Purchase Rights granted before an amendment, suspension or termination of the Plan will not be materially impaired by any such amendment, suspension or termination except (i) with the consent of the person to whom such Purchase Rights were granted, (ii) as necessary to comply with any laws, listing requirements, or governmental regulations (including, without limitation, the provisions of Section 423 of the Code and the regulations and other interpretive guidance issued thereunder relating to Employee Stock Purchase Plans) including without limitation any such regulations or other guidance that may be issued or amended after the date the Plan is adopted by the Board, or (iii) as necessary to obtain or maintain favorable tax, listing, or regulatory treatment. To be clear, the Board may amend outstanding Purchase Rights without a Participant's consent if such amendment is necessary to ensure that the Purchase Right and/or the Plan complies with the requirements of Section 423 of the Code with respect to the 423 Component.

Notwithstanding anything in the Plan or any Offering Document to the contrary, the Board will be entitled to: (i) establish the exchange ratio applicable to amounts withheld in a currency other than U.S. dollars; (ii) permit Contributions in excess of the amount designated by a Participant in order to adjust for mistakes in the Company's processing of properly completed Contribution elections; (iii) establish reasonable waiting and adjustment periods and/or accounting and crediting procedures to ensure that amounts applied toward the purchase of Common Stock for each Participant properly correspond with amounts withheld from the Participant's Contributions; (iv) amend any outstanding Purchase Rights or clarify any ambiguities regarding the terms of any Offering to enable the Purchase Rights to qualify under and/or comply with Section 423 of the Code; and (v) establish other limitations or procedures as the Board determines in its sole discretion advisable that are consistent with the Plan. The actions of the Board pursuant to this paragraph will not be considered to alter or impair any Purchase Rights granted under an Offering as they are part of the initial terms of each Offering and the Purchase Rights granted under each Offering.

#### 13. TAX QUALIFICATION; TAX WITHHOLDING.

(a) Rights granted under the 423 Component are intended to be exempt from the application of Section 409A of the Code under U.S. Treasury Regulation Section 1.409A-1(b)(5)(ii). Purchase Rights granted under the Non-423 Component to U.S. taxpayers are intended to be exempt from the application of Section 409A of the Code under the short-term deferral exception or compliant with Section 409A of the Code and any ambiguities will be construed and interpreted in accordance with such intent.

(b) Although the Company may endeavor to (i) qualify a Purchase Right for special tax treatment under the laws of the United States or jurisdictions outside of the United States or (ii) avoid adverse tax treatment, the Company makes no representation to that effect and expressly disavows any covenant to maintain special or to avoid unfavorable tax treatment, notwithstanding anything to the contrary in this Plan. The Company will be unconstrained in its corporate activities without regard to the potential negative tax impact on Participants.

(c) The Participant will make adequate provision to satisfy the Tax-Related Items withholding obligations, if any, of the Company and/or the applicable Designated Company which arise with respect to Participant's participation in the Plan or upon the disposition of the shares of the Common Stock. The Company and/or the Designated Company may, but will not be obligated to, withhold from the Participant's compensation or any other payments due the Participant the amount necessary to meet such withholding obligations, withholding a sufficient whole number of shares of Common Stock issued following exercise having an aggregate value sufficient to pay the Tax-Related Items or withhold from the proceeds of the sale of shares of Common Stock, either through a voluntary sale or a mandatory sale arranged by the Company or any other method of withholding that the Company and/or the Designated Company deems appropriate. The Company and/or the Designated Company will have the right to take such other action as may be necessary in the opinion of the Company or a Designated Company to satisfy withholding and/or reporting obligations for such Tax-Related Items. The Company shall not be required to issue any shares of Common Stock under the Plan until such obligations are satisfied.

# 14. EFFECTIVE DATE OF PLAN.

The Plan will become effective on June 16, 2021. No Purchase Rights will be exercised unless and until the Plan has been approved by the stockholders of the Company, which approval must be within 12 months before or after the date the Plan is adopted (or if required under Section 12(a) above, materially amended) by the Board.

15. MISCELLANEOUS PROVISIONS.

(a) Proceeds from the sale of shares of Common Stock pursuant to Purchase Rights will constitute general funds of the Company.

(b) A Participant will not be deemed to be the holder of, or to have any of the rights of a holder with respect to, shares of Common Stock subject to Purchase Rights unless and until the Participant's shares of Common Stock acquired upon exercise of Purchase Rights are recorded in the books of the Company (or its transfer agent).

(c) The Plan and Offering do not constitute an employment contract. Nothing in the Plan or in the Offering will in any way alter the at will nature of a Participant's employment or be deemed to create in any way whatsoever any obligation on the part of any Participant to continue in the employ of the Company, a Related Corporation or an Affiliate, or on the part of the Company, a Related Corporation or an Affiliate to continue the employment of a Participant.

(d) The provisions of the Plan will be governed by the laws of the State of Delaware without resort to that state's conflict of laws rules.

# **16. D**EFINITIONS.

As used in the Plan, the following definitions will apply to the capitalized terms indicated below:

(a) "423 *Component*" means the part of the Plan, which excludes the Non-423 Component, pursuant to which Purchase Rights that satisfy the requirements for an Employee Stock Purchase Plan may be granted to Eligible Employees.

(b) "*Affiliate*" means any entity, other than a Related Corporation, in which the Company has an equity or other ownership interest or that is directly or indirectly controlled by, controls, or is under common control with the Company, in all cases, as determined by the Board, whether now or hereafter existing.

(c) "*Board*" means the Board of Directors of the Company.

(d) *"Capital Stock"* means each and every class of common stock of the Company, regardless of the number of votes per share.

(e) "*Capitalization Adjustment*" means any change that is made in, or other events that occur with respect to, the Common Stock subject to the Plan or subject to any Purchase Right after the date the Plan is adopted by the Board without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, large nonrecurring cash dividend, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other similar equity restructuring transaction, as that term is used in Financial Accounting Standards Board Accounting Standards Codification Topic 718 (or any successor thereto). Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.

(f) *"Code*" means the Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.

(g) "*Committee*" means a committee of one or more members of the Board to whom authority has been delegated by the Board in accordance with Section 2(c).

(h) *"Common Stock"* means the Class A common stock of the Company.

(i) *"Company"* means WM Technology, Inc., a Delaware corporation.

(j) *"Contributions*" means the payroll deductions and other additional payments specifically provided for in the Offering that a Participant contributes to fund the exercise of a Purchase Right. A Participant may make additional payments into his or her account if specifically provided for in the Offering, and then only if the Participant has not already had the maximum permitted amount withheld during the Offering through payroll deductions.

(k) *"Corporate Transaction"* means the consummation, in a single transaction or in a series of related transactions, of any one or more of the following events:

(i) a sale or other disposition of all or substantially all, as determined by the Board in its sole discretion, of the consolidated assets of the Company and its subsidiaries;

(ii) a sale or other disposition of more than 50% of the outstanding securities of the Company;

(iii) a merger, consolidation or similar transaction following which the Company is not the surviving corporation; or

(iv) a merger, consolidation or similar transaction following which the Company is the surviving corporation but the shares of Common Stock outstanding immediately preceding the merger, consolidation or similar transaction are converted or exchanged by virtue of the merger, consolidation or similar transaction into other property, whether in the form of securities, cash or otherwise.

(I) "Designated 423 Company" means any Related Corporation selected by the Board as participating in the 423 Component.

(m) "*Designated Company*" means any Designated Non-423 Corporation or Designated 423 Company, provided, however, that at any given time, a Related Corporation participating in the 423 Component shall not be a Related Corporation participating in the Non-423 Component.

(n) *"Designated Non-423 Company"* means any Related Corporation or Affiliate selected by the Board as participating in the Non-423 Component.

(o) *"Director*" means a member of the Board.

(p) "*Eligible Employee*" means an Employee who meets the requirements set forth in the document(s) governing the Offering for eligibility to participate in the Offering, provided that such Employee also meets the requirements for eligibility to participate set forth in the Plan.

(q) "*Employee*" means any person, including an Officer or Director, who is "employed" for purposes of Section 423(b)(4) of the Code by the Company or a Related Corporation or solely with respect to the Non-423 Component, an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.

(r) *"Employee Stock Purchase Plan*" means a plan that grants Purchase Rights intended to be options issued under an "employee stock purchase plan," as that term is defined in Section 423(b) of the Code.

(s) *"Exchange Act"* means the Securities Exchange Act of 1934, as amended and the rules and regulations promulgated thereunder.

(t) *"Fair Market Value"* means, as of any date, the value of the Common Stock determined as follows:

(i) If the Common Stock is listed on any established stock exchange or traded on any established market, the Fair Market Value of a share of Common Stock will be, unless otherwise determined by the Board, the <u>closing sales price</u> for such stock as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in the Common Stock) <u>on the date of determination</u>, as reported in such source as the Board deems reliable. Unless otherwise provided by the Board, if there is no closing sales price for the Common Stock on the date of determination, then the Fair Market Value will be the closing sales price on the last preceding date for which such quotation exists.

(ii) In the absence of such markets for the Common Stock, the Fair Market Value will be determined by the Board in good faith in compliance with applicable laws and in a manner that complies with Sections 409A of the Code.

(u) "*Merger Agreement*" means that certain Agreement and Plan of Merger, dated as of December 10, 2020, by and among Silver Spike Acquisition Corp. ("*Silver Spike*"), a Cayman Islands exempted company, Silver Spike Merger Sub LLC, a Delaware limited liability company and a wholly owned direct subsidiary of Silver Spike, WM Holding Company, LLC, a Delaware limited liability company, and Ghost Media Group, LLC, a Nevada limited liability company.

(v) "*Non-423 Component*" means the part of the Plan, which excludes the 423 Component, pursuant to which Purchase Rights that are not intended to satisfy the requirements for an Employee Stock Purchase Plan may be granted to Eligible Employees.

(w) "*Offering*" means the grant to Eligible Employees of Purchase Rights, with the exercise of those Purchase Rights automatically occurring at the end of one or more Purchase Periods. The terms and conditions of an Offering will generally be set forth in the "*Offering Document*" approved by the Board for that Offering.

(x) *"Offering Date"* means a date selected by the Board for an Offering to commence.

(y) *"Officer*" means a person who is an officer of the Company or a Related Corporation within the meaning of Section 16 of the Exchange Act.

(z) *"Participant"* means an Eligible Employee who holds an outstanding Purchase Right.

(aa) *"Plan"* means this WM Technology, Inc. 2021 Employee Stock Purchase Plan, as amended from time to time, including both the 423 Component and the Non-423 Component.

**(bb)** *"Purchase Date"* means one or more dates during an Offering selected by the Board on which Purchase Rights will be exercised and on which purchases of shares of Common Stock will be carried out in accordance with such Offering.

(cc) *"Purchase Period"* means a period of time specified within an Offering, generally beginning on the Offering Date or on the first Trading Day following a Purchase Date, and ending on a Purchase Date. An Offering may consist of one or more Purchase Periods.

(dd) *"Purchase Right"* means an option to purchase shares of Common Stock granted pursuant to the Plan.

(ee) *"Related Corporation"* means any "parent corporation" or "subsidiary corporation" of the Company whether now or subsequently established, as those terms are defined in Sections 424(e) and (f), respectively, of the Code.

(ff) *"Securities Act"* means the Securities Act of 1933, as amended.

(gg) "*Tax-Related Items*" means any income tax, social insurance, payroll tax, fringe benefit tax, payment on account or other tax-related items arising out of or in relation to a Participant's participation in the Plan, including, but not limited to, the exercise of a Purchase Right and the receipt of shares of Common Stock or the sale or other disposition of shares of Common Stock acquired under the Plan.

(hh) "*Trading Day*" means any day on which the exchange(s) or market(s) on which shares of Common Stock are listed, including but not limited to the NYSE, Nasdaq Global Select Market, the Nasdaq Global Market, the Nasdaq Capital Market or any successors thereto, is open for trading.

ghozigroup

July 31, 2015

<u>Via Email</u> **Chris Beals** <u>chris.beals@gmail.com</u>

Dear Chris:

This is your formal offer of employment with Ghost Management Group, LLC (the "Company"). We are enthusiastic that you will accept our offer by signing where indicated below and returning a copy of this letter to us. If you have any questions, please do not hesitate to contact Jackie Becerra or me.

Position	General Counsel / CLO
Start Date	September 1, 2015
Reporting to	Doug Francis, President & COO
Schedule/Status	Full-Time, exempt employee converting to self-employed partner status as required for Holdings membership unit grant. Hours will very on business necessities. You will not be eligible for overtime pay.
Location:	First 2-3 months will be at our HQ at 41 Discovery in Irvine, CA. Chris will develop our NY office in time, but can work from home (NY) in the interim. There will be heavy travel involved with the position, including a monthly trip to Irvine.
Base Salary	Six Hundred Thousand Dollars (\$600,000) per year. In the event Chris is subject to self-employment tax on any salary amounts, WM Holdings will reimburse any such self-employment tax amounts.

**Bonus and Targets** 

Annual bonus payable at the end of each calendar year which shall be pro rated for any partial calendar years of employment. Such bonus amount shall be payable pro rata in the event Chris is terminated by the Company other than for cause. The initial target bonus amount shall be Five Hundred Thousand Dollars (\$500,000), with such amount being allocated as follows for performance. Chris and Company to discuss option to convert portion of bonus into Holdings equity.

25%	Oversight and management of internal legal and policy teams and
	management of outside counsel; management of counsel expenses
25%	Oversight and management of strategic transactions (e.g. acquisitions,
	divestitures, restructuring); advice on strategic decisions; assistance in
	staffing and growth
25%	Advice and management on "DarkCo" projects; management of legal issues
	of affiliated companies
25%	2015 EBITDA in excess of 20mm
	2016 EBITDA in excess of 30mm
	Subsequent years as mutually agreed

**Job Description** 

• CLO for WM Holding Company, LLC ("**WM Holding**"), all its subsidiaries and all affiliated companies.

- Hands on contract work for all deal flow.
- Quickly meet leading industry attorneys for best practice. Establish structure for our companies.
- Work directly with policy team to aid efforts and drive expansion to the east coast. Work directly to achieve outcomes in municipalities across the country.
- Work with targets to prepare for acquisition. Help set up best practices in acquired companies.
- Develop and manage the NY office.
- Manage our internal legal team and outside counsel. Manage legal costs.
- Assist in providing guidance on strategic opportunities and areas for growth.
- Be the deal flow desk. Contribute to revenue.
- Work with tax team to ensure best practice.
- Act as a mouthpiece for our company and on the issue. Be a thought leader.
- Leverage network to recruit other key positions. Aid in operations where needed.

Benefits Stock	Full Paid Benefits with Cigna PPO Program. 401K (We do not match). Holidays & Sick days TBD. Chris will be granted 3% of all outstanding class A units on a fully diluted basis (the " <b>Incentive Units</b> ") in WM Holdings. Vesting of Incentive Units is over a 3-year term, with the first 0.5% vesting on the anniversary of a 6-month initial cliff and quarterly vesting of 0.25% thereafter. Subject to confirmation with tax and counsel, Incentive Units shall be structured as a transfer with a lapsing call right (in accordance with the vesting schedule) (the " <b>Call Right</b> ") on such Incentive Units in favor of WM Holdings. The Call Right shall only be exercisable in the event of a termination of Chris.
	Subject to confirmation with tax and counsel, WM Holdings shall make a loan to Chris in the amount of the tax due on all Incentive Units that shall be recourse only to the Incentive Units and bear reasonable interest (the " <b>Tax Loan</b> "). The Tax Loan shall have a 7 year term and no amortization or interest payments due prior to the expiry of the term. In the event the Call Right is excercised, the call price for the applicable Incentive Units shall be a pro rata portion of the face value of the Tax Loan.
	Chris will also have 5% of the equity belonging to Doug Francis and Justin Hartfield in current and future marijuana related deal flow.
Expenses	Chris will be issued a company credit card and allowed to expense anything he deems as business.
At Will Employment	As noted above, your employment will be "at-will," with no specified length of employment. Accordingly, either you or the Company may terminate your employment for any reason, with or without cause, and with or without notice, at any time. Further, the Company has the right at any time to change the terms of and conditions of your employment, such as pay, benefits, polices and all other working conditions, as it deems appropriate.
Severance	Pay In the event Chris is let go by the Company for any other reason other than cause (i.e. Violence, theft, fraud, harassment, etc.), the company will pay three months of salary and cover COBRA expenses for health insurance for 6 months time period. Payment is due in regular pay periods over the three months.

The parties hereby acknowledge and agree to the foregoing as of the date set forth above.

/s/ Christopher Beals	/s/ Doug Francis	
Christopher Beals	Doug Francis	
	/s/ Justin Hartfield	
	Justin Hartfiel	

# GHOST MANAGEMENT GROUP, LLC 41 Discovery Irvine, California 92618

October 3, 2018

#### Dear Justin Dean:

This letter is to memorialize the offer of employment (the "<u>Offer</u>") made to **Justin Dean** by Ghost Management Group, LLC (the "<u>Company</u>") and to set forth the specific terms and conditions of your employment.

- Position. The Company offers you the full-time, exempt, regular position of Chief Information Officer at our Irvine, California office located at 41 Discovery, Irvine, California 92618. In this position, your current supervisor is Doug Francis.
- 2) <u>At-Will Employment</u>. Subject to the terms and conditions of this Offer, the Company agrees to employ you and you agree to be employed by the Company commencing no later than **November 5, 2018**. If you have not commenced employment with the Company by this date, this Offer shall be null and void and of no further effect. You shall be employed on an at-will basis, meaning that either the Company or you may, at any time, with or without cause and with or without notice, terminate the employment relationship. You and the Company agree that it is the express intent of each of us that your employment shall be at will. Nothing in this Offer or the relationship between you and the Company now or in the future may be construed or interpreted to create an employment relationship for a specific length of time or any right to continued employment, or any limit on the discretion of the Company to modify terms and conditions of employment. No employee or representative of the Company has the authority to modify this at-will policy except for the CEO of the Company, and any such modification to this at-will employment policy must be in a written agreement signed by both the employee and the CEO of the Company. This constitutes an integrated agreement with respect to the at-will nature of the employment relationship, and there may be no implied or oral agreements that in any way modify this at-will employment policy.
- 3) <u>Duties</u>. Your title will be **Chief Information Officer**. In such capacity, you shall be responsible for all duties commensurate with those generally expected of your title as well as any set forth in your job description. You shall also have such other responsibilities as may be assigned to you from time to time by your manager other senior officer of the Company.
- 4) <u>Compensation and Benefits</u>.
  - a. <u>Salary</u>. You will receive a base salary at the rate of **\$475,000**, annualized, payable in accordance with the Company's normal payroll practices. In addition, you will be eligible for a discretionary bonus, with the attainment of such bonus being based upon performance. Any bonus payable hereunder shall be paid within the "short-term deferral" period provided under Treasury Regulation Section 1.409A-(b)(4).

- b. Equity. You will be granted **3,000** unvested Class B Units (the "Employee Class B Units") in WM Holding Company, LLC, the parent company of the Company (the "Parent"), pursuant to the terms and conditions set forth in an Equity Award Agreement (the "Award Agreement") to be entered into between you and the Parent following your start date with the Company and in accordance with its internal policies regarding the grant of equity incentive units. Your Employee Class B Units will vest on a quarterly basis over four (4) years from your start date with the Company, with the vesting during the first year not accruing until the twelve (12) month anniversary of your start date with the Company (i.e. 1/4 of your Employee Class B Units shall vest on such anniversary) (such date, the "Vesting Cliff"), assuming you are still employed by the Company at such time. Following the Vesting Cliff, your Employee Class B Units shall continue to vest on a prorated quarterly basis provided you are employed by the Company on each such quarterly vesting date. Your Employee Class B Units will be subject to the terms and conditions set forth in the Parent's Amended and Restated Equity Incentive Plan, as amended from time-to-time (the "Incentive Play"). You acknowledge that you have reviewed the Incentive Plan.
- c. <u>Withholdings and Deductions</u>. All payments made under this Offer by the Company shall be subject to all required federal, state, and local withholdings and such deductions as you may instruct the Company to take.
- d. <u>Benefits</u>. You will be entitled to employee benefits on the same basis as those benefits are made available to other similarly situated Company employees. Your rights under any benefit policies or plans adopted by the Company shall be governed solely by the terms of such policies or plans. The Company reserves to itself or its designated administrator the exclusive authority and discretion to determine all issues of eligibility, interpretation and administration of each such benefit plan or policy. The Company or its designated administrator reserves the right to modify or terminate each benefit plan or program with or without prior notice to employees. Details about current benefit plans and programs are available in the office of the Company's benefits administrator.
- e. <u>Vacation</u>. You will receive paid vacation according to the Vacation policy set forth in the Employee Handbook. You will be eligible to accrue paid vacation at the rate set forth in the Employee Handbook. Payment of accrued but unused vacation will be made upon termination of employment.
- f. <u>Paid Sick Leave</u>. You will be eligible for paid sick leave according to the Company Sick Leave Policy set forth in the Employee Handbook.
- g. <u>Exclusive Compensation</u>. You agree that your compensation under this <u>Section 4</u> constitutes the full and exclusive consideration and compensation for all services rendered by you under this Offer.
- 5) <u>Conditions of Employment</u>.
  - a. <u>Policies and Procedures</u>. You agree to adhere to Company policies and procedures, including all policies contained in the Company's Employee Handbook, which you will receive when you begin employment. From time to time, Company policies and procedures may be amended by the Company and will be called to your attention.

- b. <u>Background Check and Eligibility</u>. This Offer is contingent upon a satisfactory background and reference check, which may be conducted in whole or in part by a consumer reporting agency; including, but not limited to, education and employment verification, and proof of your eligibility to work in the United States. You agree to timely complete and return to the Company all documentation provided to you under the Fair Credit Reporting Act for the purpose of completing such background and/or credit check.
- c. <u>Right to Work</u>. For purposes of federal immigration law, you will be required to provide to the Company documentary evidence of your identity and eligibility for employment in the United States. Such documentation must be provided to us within three (3) business days of your date of hire, or our employment relationship with you may be terminated.
- d. <u>Confidential Information, Non-Solicitation and Inventions Assignment Agreement</u>. Enclosed is the Company's Confidential Information, Non-Solicitation and Inventions Assignment Agreement, which you are required to sign as a condition of your employment. Upon your acceptance of this Offer, please return to me a signed copy of that agreement.
- e. <u>Arbitration Agreement</u>. Enclosed is the Company's Mutual Agreement to Arbitrate All Employment-Related Disputes, which you are required to sign as a condition of your employment. Upon your acceptance of this Offer, please return to me a signed copy of that agreement.
- f. <u>Modification</u>. The Company reserves the right to modify your position, duties, compensation, benefits, and/or other terms and conditions of employment at any time in its sole discretion, as allowed by law, except for the at-will employment policy.
- g. <u>No Reliance</u>. You acknowledge that you are not relocating your residence or resigning employment in reliance on any promise or representation by the Company regarding the kind, character, or existence of such work, or the length of time such work will last, or the compensation therefore.
- h. <u>Prior Agreements</u>. This letter supersedes any prior agreements regarding your employment with the Company.

[Remainder of Page Intentionally Left Blank]

If you accept the terms of the foregoing offer of employment, please so indicate by signing and dating below and returning it to my attention *no later than* **October 6, 2018**.

Sincerely, GHOST MANAGEMENT GROUP, LLC

By: /s/ Doug Francis

Doug Francis, CEO

# ACCEPTED AND AGREED:

Justin Dean
Oct 3, 2018
(

# GHOST MANAGEMENT GROUP, LLC 41 Discovery Irvine, California 92618

May 17, 2017

# Dear Steven Jung:

This letter is to memorialize the offer of employment ("**Offer**") made to **Steven Jung** by Ghost Management Group, LLC (the "**Company**") and to set forth the specific terms and conditions of your employment.

- 1) <u>Position</u>. Company offers you the **full-time**, **exempt**, **regular** position of **VP**, **Business Operations** at our Irvine, California location located at **41 Discovery**, **Irvine**, **California 92618**. In this position, your current supervisor is **Doug Francis**.
- 2) <u>At-Will Employment</u>. Subject to the terms and conditions of this Offer, the Company agrees to employ you and you agree to be employed by the Company commencing no later than June 19, 2017. If you have not commenced employment with the Company by this date, this Offer shall be null and void and of no further effect. You shall be employed on an at-will basis, meaning that either the Company or you may, at any time, with or without cause and with or without notice, terminate the employment relationship. You and the Company agree that it is the express intent of each of us that your employment shall be at will. Nothing in this Offer or the relationship between you and the Company now or in the future may be construed or interpreted to create an employment relationship for a specific length of time or any right to continued employment, or any limit on the discretion of the Company to modify terms and conditions of employment. No employee or representative of Company has the authority to modify this at-will policy except for the CEO of Company, and any such modification to this at-will employment policy must be in a written agreement signed by both the employee and the CEO of Company. This constitutes an integrated agreement with respect to the at-will nature of the employment relationship, and there may be no implied or oral agreements that in any way modify this at-will employment policy.
- 3) <u>Duties</u>. Your title will be **VP**, **Business Operations**. In such capacity, you shall be responsible for all duties commensurate with those generally expected of your title as well as any set forth in your job description. You shall also have such other responsibilities as may be assigned to you from time to time.
- 4) <u>Compensation and Benefits</u>.
  - a. <u>Salary</u>. You will receive a base salary at the rate of **\$225,000**, annualized, payable in accordance with the Company's normal payroll practices. In addition, you will be eligible for a bonus equal to an annualized amount of **twenty percent (20%)** of your annual salary, with the attainment of such bonus being based upon the achievement of quantifiable quarterly and annual goals which will be set by you and senior management. Such bonus will be payable on a quarterly basis (up to 5% per quarter of such 20% annualized amount) based on the achievement of the applicable quarterly or annual goals.

- b. <u>Sign-on Bonus</u>: You shall receive a signing bonus of **\$20,000** (the "<u>Signing Bonus</u>"). If you terminate your employment with the Company or are terminated for cause on or prior to the one year anniversary of your start date with the Company, then you must repay the Company the full amount of the Signing Bonus.
- c. Equity. You will be granted 4,354 unvested Class B Units (the "Employee Class B Units") in WM Holding Company, LLC, the parent company of the Company ("Parent"), pursuant to the terms and conditions set forth in an Equity Award Agreement (the "Award Agreement") to be entered into between you and Parent following your start date with the Company and in accordance with its internal policies regarding the grant of equity incentive units. Your Employee Class B Units will vest on a quarterly basis over four (4) years from your start date with the Company (i.e. 1/4 of your Employee Class B Units shall vest on such anniversary) (such date, the "Vesting Cliff"), assuming you are still employed by the Company at such time. Following the Vesting Cliff, your Employee Class B Units shall continue to vest on a prorated quarterly basis provided you are employee Class B Units will be 100% vested, on a Company Sale or Company IPO (as such terms are defined in the Amended and Restated Limited Liability Company Operating Agreement of Parent (the "Parent Operating Agreement")), in each case, in accordance with, and subject to, the terms and conditions set forth in the Parent Operating Agreement.
- d. <u>Withholdings and Deductions</u>. All payments made under this Offer by the Company shall be subject to all required federal, state and local withholdings and such deductions as you may instruct the Company to take.
- e. <u>Benefits</u>. You will be entitled to employee benefits on the same basis as those benefits are made available to other similarly situated Company employees. Your rights under any benefit policies or plans adopted by the Company shall be governed solely by the terms of such policies or plans. The Company reserves to itself or its designated administrator the exclusive authority and discretion to determine all issues of eligibility, interpretation and administration of each such benefit plan or policy. The Company or its designated administrator reserves the right to modify or terminate each benefit plan or program with or without prior notice to employees. Details about current benefit plans and programs are available in the office of the Company's benefits administrator.
- f. <u>Vacation</u>. You will receive paid vacation according to the Vacation policy set forth in the Employee Handbook. You will be eligible to accrue paid vacation at the rate set forth in the Employee Handbook. Payment of accrued but unused vacation will be made upon termination of employment.
- g. <u>Paid Sick Leave</u>. You will receive paid sick leave according to Company Sick Leave Policy set forth in the Employee Handbook.

h. <u>Exclusive Compensation</u>. You agree that your compensation under this <u>Section 4</u> constitutes the full and exclusive consideration and compensation for all services rendered by you under this Offer.

#### 5) <u>Conditions of Employment</u>.

- a. <u>Policies and Procedures</u>. You agree to adhere to Company policies and procedures, including all polices contained in Company's Employee Handbook, which you will receive when you begin employment. From time to time, Company policies and procedures may be amended by the Company and will be called to your attention.
- b. <u>Background Check and Eligibility</u>. This Offer is contingent upon a satisfactory background and reference check, which may be conducted in whole or in part by a consumer reporting agency; including, but not limited to, education and employment verification, and proof of your eligibility to work in the United States. You agree to timely complete and return to the Company all documentation provided to you under the Fair Credit Reporting Act for the purpose of completing such background and/or credit check.
- c. <u>Right to Work</u>. For purposes of federal immigration law, you will be required to provide to the Company documentary evidence of your identity and eligibility for employment in the United States. Such documentation must be provided to us within three (3) business days of your date of hire, or our employment relationship with you may be terminated.
- d. <u>Confidential Information, Non-Solicitation and Inventions Assignment Agreement</u>. Enclosed is the Company's Confidential Information, Non-Solicitation and Inventions Assignment Agreement, which you are required to sign as a condition of your employment. Upon your acceptance of this offer, please return to me a signed copy of that agreement.
- e. <u>Arbitration Agreement</u>. Enclosed is the Company's Mutual Agreement to Arbitrate All Employment-Related Disputes, which you are required to sign as a condition of your employment. Upon your acceptance of this offer, please return to me a signed copy of that agreement.
- f. <u>Modification</u>. Company reserves the right to modify your position, duties, compensation, benefits, and/or other terms and conditions of employment at any time in its sole discretion, as allowed by law, except for the at-will employment policy.
- g. <u>No Reliance</u>. You acknowledge that you are not relocating your residence or resigning employment in reliance on any promise or representation by the Company regarding the kind, character, or existence of such work, or the length of time such work will last, or the compensation therefore.
- h. <u>Prior Verbal Agreements</u>. This letter supersedes any prior verbal agreements regarding your employment with the Company.

[Remainder of Page Intentionally Left Blank]

If you accept the terms of the foregoing offer of employment, please so indicate by signing and dating below and returning it to my attention *no later than* **May 26, 2017**.

4

Sincerely, GHOST MANAGEMENT GROUP, LLC

By: /s/ Doug Francis

Doug Francis, CEO

# ACCEPTED AND AGREED:

/s/ Steven Jung		
Print	Steven Jung	
Name:		
Date:	5/18/17	

LEASE

BETWEEN

# THE IRVINE COMPANY LLC

AND

# GHOST MEDIA GROUP, LLC

#### LEASE

THIS LEASE is made as of November 11, 2013, by and between **THE IRVINE COMPANY LLC**, a Delaware limited liability company, hereafter called "*Landlord*," and **GHOST MEDIA GROUP, LLC**, a Nevada limited liability company, hereafter called "*Tenant*."

#### **ARTICLE 1. BASIC LEASE PROVISIONS**

Each reference in this Lease to the "*Basic Lease Provisions*" shall mean and refer to the following collective terms, the application of which shall be governed by the provisions in the remaining Articles of this Lease.

1. Tenant's Trade Name: N/A

2. **Premises:** Suite No. 100 (The Premises are more particularly described in Section 2.1).

Address of Building: 41 Discovery, Irvine, CA 92618

**Project Description** Discovery Business Center (as shown on Exhibit Y to this Lease)

- 3. **Use of Premises**: General office and for no other use.
- 4. Estimated Commencement Date: March 1, 2014

5. Lease Term: 60 months, plus such additional days as may be required to cause this Lease to expire on the final day of the calendar month.

6. Basic Rent:

Months of Term or Period	Monthly Rate Per Rentable Square Foot	Monthly Basic Rent (rounded to the nearest dollar)
1 to 6	N/A	\$26,668.00
7 to 12	\$1.19	\$53,336.00
13 to 24	\$1.24	\$55,577.00
25 to 36	\$1.30	\$58,266.00
37 to 48	\$1.36	\$60,955.00
49 to 60	\$1.42	\$63,644.00

Notwithstanding the above schedule of Basic Rent to the contrary, as long as Tenant is not in Default (as defined in Section 14.1) under this Lease, Tenant shall be entitled to an abatement of 3 full calendar months of Basic Rent in the aggregate amount of S174,798.00 (i.e. \$58,266.00 per month) (the "*Abated Basic Rent*") for the initial 31<sup>st</sup>, 32<sup>nd</sup>, and 33<sup>rd</sup> full calendar months of the Term (the "*Abatement Period*"). In the event Tenant Defaults at any time during the Term, all Abated Basic Rent shall immediately become due and payable. The payment by Tenant of the Abated Basic Rent in the event of a Default shall not limit or affect any of Landlord's other rights, pursuant to this Lease or at law or in equity. Only Basic Rent shall be abated during the Abatement Period and all other additional rent and other costs and charges specified in this Lease shall remain as due and payable pursuant to the provisions of this Lease.

7. **Expense Recovery Period**: Every twelve month period during the Term (or portion thereof during the first and last Lease years) ending June 30.



8. Floor Area of Premises: approximately 44,820 rentable square feet

#### Floor Area of Building: approximately 44,820 rentable square feet

- 9. Security Deposit: \$175,000.00
- 10. Broker(s): Irvine Realty Company ("Landlord's Broker") and Travers Realty Corporation of Orange County. Inc. ("Tenant's Broker")
- 11. **Parking**: 202 parking spaces, together with the "Additional Parking", in accordance with the provisions set forth in Exhibit F to this Lease.

#### 12. Address for Payments and Notices:

#### LANDLORD

Payment Address: THE IRVINE COMPANY LLC Department #1211 Los Angeles, CA 90084-1211

Notice Address:

THE IRVINE COMPANY LLC 550 Newport Center Drive Newport Beach, CA 92660 Attn: Senior Vice President, Property Operations Irvine Office Properties

with a copy of notices to:

The Irvine Company LLC 550 Newport Center Drive Newport Beach, CA, 92660 Attn: Senior Vice President, Property Operations Irvine Office Properties

**LIST OF LEASE EXHIBITS** (All exhibits, riders and addenda attached to this Lease are hereby incorporated into and made a part of this Lease):

3

**Description of Premises** Exhibit A Exhibit B **Operating Expenses** Utilities and Services Exhibit C Exhibit D Tenant's Insurance Exhibit E **Rules and Regulations** Exhibit F Parking Exhibit G Additional Provisions Exhibit H Hazardous Materials Disclosure Statement [Intentionally Deleted] Exhibit I Exhibit J Survey Form Exhibit X Work Letter Exhibit Y **Project Description** 

#### TENANT

GHOST MEDIA GROUP, LLC 41 Discovery, 100 Irvine, CA 92618

#### **ARTICLE 2. PREMISES**

**2.1.** LEASED PREMISES. Landlord leases to Tenant and Tenant leases from Landlord the Premises shown in Exhibit A (the "**Premises**"), containing approximately the floor area set forth in Item 8 of the Basic Lease Provisions (the "**Floor Area**"). The Premises consist of all of the Floor Area of the building identified in Item 2 of the Basic Lease Provisions (the "**Building**"), which is a portion of the project described in Item 2 (the "**Project**"). Landlord and Tenant stipulate and agree that the Floor Area of Premises set forth in Item 8 of the Basic Lease Provisions is correct.

2.2. ACCEPTANCE OF PREMISES. Tenant acknowledges that neither Landlord nor any representative of Landlord has made any representation or warranty with respect to the Premises, the Building or the Project or the suitability or fitness of either for any purpose, except as expressly set forth in Section 3 of Exhibit G attached hereto and elsewhere in this Lease. Tenant acknowledges that the flooring materials which may be installed within portions of the Premises located on the ground floor of the Building may be limited by the moisture content of the Building slab and underlying soils. The taking of possession or use of the Premises by Tenant for any purpose other than construction shall conclusively establish that the Premises and the Building were in satisfactory condition and in conformity with the provisions of this Lease in all respects, except for those matters which Tenant shall have brought to Landlord's attention on a written punch list. The punch list shall be limited to any items required to be accomplished by Landlord under the Work Letter (if any) attached as Exhibit X, and shall be delivered to Landlord within 30 days after the Commencement Date (as defined herein). If there is no Work Letter, or if no items are required of Landlord under the Work Letter, by taking possession of the Premises Tenant accepts the improvements in their existing condition, and waives any right or claim against Landlord arising out of the condition of the Premises. Nothing contained in this Section 2.2 shall affect the commencement of the Term or the obligation of Tenant to pay rent. Landlord shall diligently complete all punch list items of which it is notified as provided above, but in no event longer than thirty (30) days after receipt from Tenant.

#### **ARTICLE 3. TERM**

**3.1. GENERAL**. The term of this Lease ("*Term*") shall be for the period shown in Item 5 of the Basic Lease Provisions. The Term shall commence ("*Commencement Date*") on the earlier of (a) the date the Premises are deemed "ready for occupancy" (as hereinafter defined) and possession thereof is delivered to Tenant, but not sooner than February 1, 2014, or (b) the date Tenant commences its regular business activities within the Premises. Promptly following request by Landlord, the parties shall memorialize on a form provided by Landlord (the "*Commencement Memorandum*") the actual Commencement Date and the expiration date ("*Expiration Date*") of this Lease; should Tenant fail to execute and return the Commencement Memorandum to Landlord within 10 business days (or provide specific written objections thereto within that period), then Landlord's determination of the Commencement and Expiration Dates as set forth in the Commencement Memorandum shall be conclusive. The Premises shall be deemed "ready for occupancy" when Landlord, to the extent applicable, has substantially completed all the work required to be completed by Landlord pursuant to the Work Letter (if any) attached to this Lease but for minor punch list matters, and has obtained the requisite governmental approvals for Tenant's occupancy in connection with such work.

**3.2. D**ELAY IN POSSESSION. If Landlord, for any reason whatsoever, cannot deliver possession of the Premises to Tenant on or before the Estimated Commencement Date set forth in Item 4 of the Basic Lease Provisions, this Lease shall not be void or voidable nor shall Landlord be liable to Tenant for any resulting loss or damage However, Tenant shall not be liable for any rent until the Commencement Date occurs as provided in Section 3.1 above, except that if Landlord's failure to substantially complete all work required of Landlord pursuant to Section 3.1(i) above is attributable to any action or inaction by Tenant (including without limitation any Tenant Delay described in the Work Letter, if any, attached to this Lease), then the Premises shall be deemed ready for occupancy, and Landlord shall be entitled to full performance by Tenant (including the payment of rent), as of the date Landlord would have been able to substantially complete such work and deliver the Premises to Tenant but for Tenant's delay(s).

#### **ARTICLE 4. RENT AND OPERATING EXPENSES**

**4.1. B**ASIC **R**ENT. From and after the Commencement Date, Tenant shall pay to Landlord without deduction or offset a Basic Rent for the Premises in the total amount shown (including subsequent adjustments, if any) in Item 6 of the Basic Lease Provisions (the "*Basic Rent*"). if the Commencement Date is other than the first day of a calendar month, any rental adjustment shown In Item 6 shall be deemed to occur on the first day of the next calendar month following the specified monthly anniversary of the Commencement Date. The Basic Rent shall be due and payable in advance commencing on the Commencement Date and continuing thereafter on the first day of each successive calendar month of the Term, as prorated for any partial month. No demand, notice or invoice shall be required. An installment in the amount of 1 full month's Basic Rent at the initial rate specified in Item 6 of the Basic Lease Provisions shall be delivered to Landlord concurrently with Tenant's execution of this Lease and shall be applied against the Basic Rent first due hereunder; the next installment of Basic Rent shall be due on the first day of the second calendar month of the Term, which installment shall, if applicable, be appropriately prorated to reflect the amount prepaid for that calendar month.

**4.2. OPERATING EXPENSES.** Tenant shall pay Tenant's Share of Operating Expenses in accordance with Exhibit B of this Lease.

**4.3. SECURITY DEPOSIT.** Concurrently with Tenant's delivery of this Lease, Tenant shall deposit with Landlord the sum, if any, stated in Item 9 of the Basic Lease Provisions (the "*Security Deposit*"), to be held by Landlord as security for the full and faithful performance of Tenant's obligations under this Lease, to pay any rental sums, including without limitation such additional rent as may be owing under any provision hereof, and to maintain the Premises as required by Sections 7.1 and 15.2 or any other provision of this Lease. Upon any breach of the foregoing obligations by Tenant, Landlord may apply all or part of the Security Deposit as full or partial compensation. If any portion of the Security Deposit is so applied, Tenant shall within 5 days after written demand by Landlord deposit cash with Landlord in an amount sufficient to restore the Security Deposit to its original amount. Landlord shall not be required to keep the Security Deposit separate from its general funds, and Tenant shall not be entitled to interest on the Security Deposit. In no event may Tenant utilize all or any portion of the Security Deposit as a payment toward any rental sum due under this Lease. Any unapplied balance of the Security Deposit shall be returned to Tenant or, at Landlord's option, to the last assignee of Tenant's interest in this Lease within 30 days following the termination of this Lease and Tenant's vacation of the Premises. Tenant hereby waives the provisions of Section 1950.7 of the California Civil Code, or any similar or successor laws now or hereafter in effect, in connection with Landlord's application of the Security Deposit to prosit to prospective rent that would have been payable by Tenant but for the early termination due to Tenant's Default (as defined herein).

Provided that no Default has occurred under any provision of this Lease at any time during the Term hereof, then upon the written request of Tenant given at any time following the expiration of the 30th full calendar month of the Term, Landlord shall return a portion of the Security Deposit in the amount of \$104,991.00 to Tenant in the form of credit(s) against Rent next coming due under this Lease.

#### **ARTICLE 5. USES**

**5.1.** Use. Tenant shall use the Premises only for the purposes stated in Item 3 of the Basic Lease Provisions and for no other use whatsoever The uses prohibited under this Lease shall include, without limitation, use of the Premises or a portion thereof for (i) offices of any agency or bureau of the United States or any state or political subdivision thereof; (ii) offices or agencies of any foreign governmental or political subdivision thereof; or (iii) schools, temporary employment agencies or other training facilities which are not ancillary to corporate, executive or professional office use. Tenant shall not do or permit anything to be done in or about the Premises which will in any way interfere with the rights or quiet enjoyment of other occupants of the Building or the Project, or use or allow the Premises to be used for any unlawful purpose, nor shall Tenant permit any nuisance or commit any waste in the Premises or the Project. Tenant shall not perform any work or conduct any business whatsoever in the Project other than inside the Premises. Tenant shall comply at its expense with all present and future laws, ordinances and requirements of all governmental authorities that pertain to Tenant or its use of the Premises. As of the date of this Lease, there has been no inspection of the Building and Project by a Certified Access Specialist as referenced in Section 1938 of the California Civil Code.

5.2. SIGNS. Provided Tenant continues to occupy the entire Premises, Tenant shall have the exclusive right to 2 exterior "*building top*" signs on the Building for Tenant's name and graphics in locations designated by Landlord, subject to Landlord's right of prior approval that such exterior signage is in compliance with the Signage Criteria (defined below). Except as provided in the foregoing, and except for Landlord's standard lobby directory and suite signage identifying Tenant's name and/or logo. Tenant shall have no right to maintain signs in any location in. on or about the Premises, the Building or the Project and shall not place or erect any signs that are visible from the exterior of the Building. The size, design, graphics, material, style, color and other physical aspects of any permitted sign shall be subject to Landlords written determination, as determined reasonably by Landlord, prior to installation, that signage is in compliance with any covenants, conditions or restrictions encumbering the Premises and Landlord's signage program for the Project, as in effect from time to time and approved by the City in which the Premises are located ("Signage Criteria"). Prior to placing or erecting any such signs, Tenant shall obtain and deliver to Landlord a copy of any applicable municipal or other governmental permits and approvals, except to Landlord's standard suite signage. Tenant shall be responsible for all costs of any permitted sign, including, without limitation, the fabrication, installation, maintenance and removal thereof and the cost of any permits therefor, except that Landlord shall pay for the initial installation costs only of the standard lobby directory and suite signage. If Tenant fails to maintain its sign in good condition, or if Tenant fails to remove same upon termination of this Lease and repair and restore any damage caused by the sign or its removal, Landlord may do so at Tenant's expense. Landlord shall have the right to temporarily remove any signs in connection with any repairs or maintenance in or upon the Building. The term "sign" as used in this Section shall include all signs, designs, monuments, displays, advertising materials, logos, banners, projected images, pennants, decals, pictures, notices, lettering, numerals or graphics. Tenant's exterior signage rights under this Section 5.2 belong solely to Ghost Media Group, LLC, a Nevada limited liability company, and any attempted assignment or transfer of such rights shall be void and of no force and effect.

**5.3. HAZARDOUS MATERIALS.** Tenant shall not generate, handle, store or dispose of hazardous or toxic materials (as such materials may be identified in any federal, state or local law or regulation) in the Premises or Project without the prior written consent of Landlord; provided that the foregoing shall not be deemed to proscribe the use by Tenant of customary office supplies in normal quantities so long as such use comports with all applicable laws. Tenant acknowledges that it has read, understands and, if applicable, shall comply with the provisions of Exhibit H to this Lease, if that Exhibit is attached.

## **ARTICLE 6. LANDLORD SERVICES**

**6.1.** UTILITIES AND SERVICES. Landlord and Tenant shall be responsible to furnish those utilities and services to the Premises to the extent provided in Exhibit C, subject to the conditions and payment obligations and standards set forth in this Lease. Landlord shall not be liable for any failure to furnish any services or utilities when the failure is the result of any accident or other cause beyond Landlord's reasonable control, nor shall Landlord be liable for damages resulting from power surges or any breakdown in telecommunications facilities or services. Landlord's temporary inability to furnish any services or utilities shall not entitle Tenant to any damages, relieve Tenant of the obligation to pay rent or constitute a constructive or other eviction of Tenant, except that Landlord shall diligently attempt to restore the service or utility promptly However, if the Premises, or a material portion of the Premises, are made untenantable for a period in excess of 5 consecutive business days as a result of a service interruption that is reasonably within the control of Landlord to correct and through no fault of Tenant and for reasons other than as contemplated in Article 11, then Tenant, as its sole remedy, shall be entitled to receive an abatement of rent payable hereunder during the period beginning on the 6th consecutive business day of the service interruption and ending on the day the service has been restored. Tenant shall comply with all rules and regulations which Landlord may reasonably establish for the provision of services and utilities, and shall cooperate with all reasonable conservation practices established by Landlord.

**6.2. OPERATION AND MAINTENANCE OF COMMON AREAS.** During the Term, Landlord shall operate all Common Areas within the Building and the Project. The term "*Common Areas*" shall mean all areas within the Building and other buildings in the Project which are not held for exclusive use by persons entitled to occupy space, including without limitation parking areas and structures, driveways, sidewalks, landscaped and planted areas, hallways and interior stairwells not located within the premises of any tenant, common electrical rooms, entrances and lobbies, elevators, and restrooms not located within the premises of any tenant.

**6.3.** Use of Common Areas. The occupancy by Tenant of the Premises shall include the use of the Common Areas in common with Landlord and with all others for whose convenience and use the Common Areas may be provided by Landlord, subject, however, to compliance with Rules and Regulations described in Article 17 below. Landlord shall at all times during the Term have exclusive control of the Common Areas, and may restrain or permit any use or occupancy, except as otherwise provided in this Lease or in Landlord's rules and regulations. Tenant shall keep the Common Areas clear of any obstruction or unauthorized use related to Tenant's operations. Landlord may temporarily close any portion of the Common Areas for repairs, remodeling and/or alterations, to prevent a public dedication or the accrual of prescriptive rights, or for any other reasonable purpose. Landlord's temporary closure of any portion of the Common Areas for such purposes shall not deprive Tenant of reasonable access to the Premises.

**6.4. CHANGES AND ADDITIONS BY LANDLORD.** Landlord reserves the right to make alterations or additions to the Building or the Project or to the attendant fixtures, equipment and Common Areas, and such change shall not entitle Tenant to any abatement of rent or other claim against Landlord. No such change shall deprive Tenant of reasonable access to or use of the Premises.

#### **ARTICLE 7. REPAIRS AND MAINTENANCE**

**7.1. TENANT'S MAINTENANCE AND REPAIR.** Subject to Articles 11 and 12, Tenant at its sole expense shall make all repairs necessary to keep the Premises and all improvements and fixtures therein in good condition and repair, excepting ordinary wear and tear. Notwithstanding Section 7.2 below, Tenant's maintenance obligation shall include without limitation all appliances, interior glass, doors, door closures, hardware, fixtures, electrical, plumbing, fire extinguisher equipment and other equipment installed in the Premises and all Alterations constructed by Tenant pursuant to Section 7.3 below, together with any supplemental HVAC equipment servicing only the Premises. All repairs and other work performed by Tenant or its contractors shall be subject to the terms of Sections 7.3 and 7.4 below. Alternatively, should Landlord or its management agent agree to make a repair on behalf of Tenant and at Tenant's request, Tenant shall promptly reimburse Landlord as additional rent for all reasonable costs incurred (including the standard supervision fee) upon submission of an invoice.

LANDLORD'S MAINTENANCE AND REPAIR. Subject to Articles 11 and 12, Landlord shall provide service, 7.2. maintenance and repair with respect to the heating, ventilating and air conditioning ("HVAC") equipment of the Building (exclusive of any supplemental HVAC equipment servicing only the Premises) and shall maintain in good repair the Common Areas, roof, foundations, footings, the exterior surfaces of the exterior walls of the Building (including exterior glass), and the structural, electrical, mechanical and plumbing systems of the Building (including elevators, if any, serving the Building), except to the extent provided in Section 7.1 above. Landlord need not make any other improvements or repairs except as specifically required under this Lease, and nothing contained in this Section 7.2 shall limit Landlord's right to reimbursement from Tenant for maintenance, repair costs and replacement costs as provided elsewhere in this Lease. Notwithstanding any provision of the California Civil Code or any similar or successor laws to the contrary, Tenant understands that it shall not make repairs at Landlord's expense or by rental offset. Except as provided in Section 11.1 and Article 12 below, there shall be no abatement of rent and no liability of Landlord by reason of any injury to or interference with Tenant's business arising from the making of any repairs, alterations or improvements to any portion of the Building, including repairs to the Premises, nor shall any related activity by Landlord constitute an actual or constructive eviction; provided, however, that in making repairs, alterations or improvements, Landlord shall interfere as little as reasonably practicable with the conduct of Tenant's business in the Premises. Tenant hereby waives any and all rights under and benefits of subsection 1 of Section 1932, and Sections 1941 and 1942 of the California Civil Code, or any similar or successor laws now or hereafter in effect.

ALTERATIONS. Except for cosmetic alterations and projects that do not exceed \$150,000.00 during any calendar 7.3. year of the Term, that do not require a permit from the City of Irvine and that satisfy the criteria in the next following sentence (which cosmetic work shall require notice to Landlord but not Landlord's consent), Tenant shall make no alterations, additions, decorations, or improvements (collectively referred to as "Alterations") to the Premises without the prior written consent of Landlord. Landlord's consent shall not be unreasonably withheld as long as the proposed Alterations do not affect the structural. electrical or mechanical components or systems of the Building, are not visible from the exterior of the Premises, do not change the basic floor plan of the Premises, and utilize only Landlord's building standard materials ("Standard Improvements"). Landlord may impose, as a condition to its consent, any requirements that Landlord in its discretion may deem reasonable or desirable. Without limiting the generality of the foregoing, Tenant shall use Landlord's designated mechanical and electrical contractors for all Alterations work affecting the mechanical or electrical systems of the Building. Should Tenant perform any Alterations work that would necessitate any ancillary Building modification or other expenditure by Landlord, then Tenant shall promptly fund the cost thereof to Landlord. Tenant shall obtain all required permits for the Alterations and shall perform the work in compliance with all applicable laws, regulations and ordinances with contractors reasonably acceptable to Landlord, and except for cosmetic Alterations not requiring a permit, Landlord shall be entitled to a supervision fee in the amount of 5% of the cost of the Alterations Any request for Landlord's consent shall be made in writing and shall contain architectural plans describing the work in detail reasonably satisfactory to Landlord. Landlord may elect to cause its architect to review Tenant's architectural plans, and the reasonable cost of the review shall be reimbursed by Tenant. Should the Alterations proposed by Tenant and consented to by Landlord change the floor plan of the Premises, then Tenant shall, at its expense, furnish Landlord with as-built drawings and CAD disks compatible with Landlord's systems. Alterations shall be constructed in a good and workmanlike manner using materials of a quality reasonably approved by Landlord Unless Landlord otherwise agrees in writing, all Alterations affixed to the Premises, including without limitation all Tenant Improvements constructed pursuant to the Work Letter (except as otherwise provided in the Work Letter), but excluding moveable trade fixtures and furniture, shall become the property of Landlord and shall be surrendered with the Premises at the end of the Term, except that Landlord may, by notice to Tenant given at least 30 days prior to the Expiration Date, require Tenant to remove by the Expiration Date, or sooner termination date of this Lease, all or any Alterations (including without limitation all telephone and data cabling) installed either by Tenant or by Landlord at Tenant's request (collectively, the "Required Removables"), and to replace any non-Standard Improvements with the applicable Standard Improvements. Tenant, at the time it requests approval for a proposed Alteration, may request in writing that Landlord advise Tenant whether the Alteration or any portion thereof, is a Required Removable. Within 10 days after receipt of Tenant's request, Landlord shall advise Tenant in writing as to which portions of the subject Alterations are Required Removables. In connection with its removal of Required Removables, Tenant shall repair any damage to the Premises arising from that removal and shall restore the affected area to its pre-existing condition, reasonable wear and tear excepted.

**7.4. MECHANIC'S LIENS.** Tenant shall keep the Premises free from any liens arising out of any work performed, materials furnished, or obligations incurred by or for Tenant. Upon request by Landlord, Tenant shall promptly cause any such lien to be released by posting a bond in accordance with California Civil Code Section 8424 or any successor statute. In the event that Tenant shall not, within 15 days following the imposition of any lien, cause the lien to be released of record by payment or posting of a proper bond, Landlord shall have, in addition to all other available remedies, the right to cause the lien to be released by any means it deems proper, including payment of or defense against the claim giving rise to the lien. All expenses so incurred by Landlord, including Landlord's attorneys' fees, shall be reimbursed by Tenant promptly following Landlord's demand, together with interest from the date of payment by Landlord at the maximum rate permitted by law until paid. Tenant shall give Landlord no less than 20 days' prior notice in writing before commencing construction of any kind on the Premises.

**7.5. ENTRY AND INSPECTION.** Landlord shall at all reasonable times have the right to enter the Premises to inspect them, to supply services in accordance with this Lease, to make repairs and renovations as reasonably deemed necessary by Landlord, and to submit the Premises to prospective or actual purchasers or encumbrance holders (or, during the final twelve months of the Term or when an uncured Default exists, to prospective tenants), all without being deemed to have caused an eviction of Tenant and without abatement of rent except as provided elsewhere in this Lease, If reasonably necessary, Landlord may temporarily close all or a portion of the Premises to perform repairs, alterations and additions. Except in emergencies or to provide Building services, Landlord shall provide Tenant with at least 24-hour prior verbal notice of entry and shall use reasonable efforts to minimize any interference with Tenant's use of the Premises.

#### **ARTICLE 8.** [Intentionally Deleted]

#### **ARTICLE 9. ASSIGNMENT AND SUBLETTING**

#### **9.1. RIGHTS OF PARTIES.**

(a) Except as otherwise specifically provided in this Article 9, Tenant may not, either voluntarily or by operation of law, assign, sublet, encumber, or otherwise transfer all or any part of Tenant's interest in this Lease, or permit the Premises to be occupied by anyone other than Tenant (each, a "*Transfer*"), without Landlord's prior written consent, which consent shall not unreasonably be withheld in accordance with the provisions of Section 9.1(b). For purposes of this Lease, references to any subletting, sublease or variation thereof shall be deemed to apply not only to a sublease effected directly by Tenant, but also to a sub-subletting or an assignment of subtenancy by a subtenant at any level. Except as otherwise specifically provided in this Article 9, no Transfer (whether voluntary, involuntary or by operation of law) shall be valid or effective without Landlord's prior written consent and, at Landlord's election, such a Transfer shall constitute a material default of this Lease.

Except as otherwise specifically provided in this Article 9, if Tenant or any subtenant hereunder desires to (b) transfer an interest in this Lease, Tenant shall first notify Landlord in writing and shall request Landlord's consent thereto. Tenant shall also submit to Landlord in writing: (i) the name and address of the proposed transferee; (ii) the nature of any proposed subtenant's or assignee's business to be carried on in the Premises; (iii) the terms and provisions of any proposed sublease or assignment (including without limitation the rent and other economic provisions, term, improvement obligations and commencement date); (iv) evidence that the proposed assignee or subtenant will comply with the requirements of Exhibit D to this Lease; and (v) any other information requested by Landlord and reasonably related to the Transfer. Landlord shall not unreasonably withhold its consent, provided: (1) the use of the Premises will be consistent with the provisions of this Lease and with Landlord's commitment to other tenants of the Building and Project; (2) any proposed subtenant or assignee demonstrates that it is financially responsible by submission to Landlord of all reasonable information as Landlord may request concerning the proposed subtenant or assignee, including, but not limited to, a balance sheet of the proposed subtenant or assignee as of a date within 90 days of the request for Landlord's consent and statements of income or profit and loss of the proposed subtenant or assignee for the two-year period preceding the request for Landlord's consent; (3) the proposed assignee or subtenant is neither an existing tenant or occupant of the Building or Project with whom Landlord has been "actively negotiating" (as hereinafter defined) to expand or relocate within the Building or Project nor a prospective tenant with whom Landlord or Landlord's affiliate has been actively negotiating to become a tenant at the Building or Project; and (4) the proposed transferee is not an SDN (as defined below) and will not impose additional burdens or security risks on Landlord. If Landlord consents to the proposed Transfer, then the Transfer may be effected within 90 days after the date of the consent upon the terms described in the information furnished to Landlord; provided that any material change in the terms shall be subject to Landlord's consent as set forth in this Section 9.1(b). Landlord shall approve or disapprove any requested Transfer within 30 days following receipt of Tenant's written notice and the information set forth above. Except in connection with a Permitted Transfer (as defined below), if Landlord approves the Transfer Tenant shall pay a transfer fee of \$750.00 to Landlord concurrently with Tenant's execution of a Transfer consent prepared by Landlord. As used herein, "actively negotiating" shall mean that Landlord and the existing or prospective tenant have exchanged written communications concerning the leased space in the Building or the Project within 120 days prior to the date of Tenant's request for Landlord's consent.

(c) Notwithstanding the provisions of Subsection (b) above, and except in connection with a "*Permitted Transfer*" (as defined below), in lieu of consenting to a proposed assignment of this Lease or to a proposed subletting of all or substantially all of the Premises for all or substantially all of the remainder of the Term, Landlord may elect to terminate this Lease in its entirety, such termination to be effective on the date that such proposed sublease or assignment would have commenced. Landlord may thereafter, at its option, assign or re-let any space so recaptured to any third party, including without limitation the proposed transferee identified by Tenant.

(d) Should any Transfer occur, Tenant shall, except in connection with a Permitted Transfer, promptly pay or cause to be paid to landlord, as additional rent, 50% of any amounts paid by the assignee or subtenant, however described and whether funded during or after the Lease Term, to the extent such amounts are in excess of the sum of (i) the scheduled Basic Rent payable by Tenant hereunder (or, in the event of a subletting of only a portion of the Premises, the Basic Rent allocable to such portion and (ii) the direct out-of-pocket costs, as evidenced by third party invoices provided to Landlord, incurred by Tenant to effect the Transfer (including, but not limited to, costs of tenant improvements, brokerage fees, moving allowances), which costs shall be amortized over the remaining Term of this Lease or, if shorter, over the term of the sublease.

(e) The sale of all or substantially all of the assets of Tenant (other than bulk sales in the ordinary course of business), the merger or consolidation of Tenant, the sale of Tenant's capital stock, or any other direct or indirect change of control of Tenant, including, without limitation, change of control of Tenant's parent company or a merger by Tenant or its parent company, shall be deemed a Transfer within the meaning and provisions of this Article. Notwithstanding the foregoing, Tenant may assign this Lease to a successor to Tenant by merger, consolidation or the purchase of substantially all of Tenant's assets, or assign this Lease or sublet all or a portion of the Premises to an Affiliate (defined below), without the consent of Landlord but subject to the provisions of Section 9.2, provided that all of the following conditions are satisfied (a "Permitted Transfer"): (i) Tenant is not then in Default hereunder; (ii) Tenant gives Landlord written notice at least 10 business days before such Permitted Transfer; and (iii) the successor entity resulting from any merger or consolidation of Tenant or the sale of all or substantially all of the assets of Tenant, has a net worth (computed in accordance with generally accepted accounting principles, except that intangible assets such as goodwill, patents, copyrights, and trademarks shall be excluded in the calculation ("*Net Worth*")) at the time of the Permitted Transfer that is at least equal to the Net Worth of Tenant immediately before the Permitted Transfer. Tenant's notice to Landlord shall include reasonable information and documentation evidencing the Permitted Transfer and showing that each of the above conditions has been satisfied. If requested by Landlord, Tenant's successor shall sign and deliver to Landlord a commercially reasonable form of assumption agreement. "Affiliate" shall mean an entity controlled by, controlling or under common control with Tenant.

**9.2. EFFECT OF TRANSFER.** No subletting or assignment, even with the consent of Landlord, shall relieve Tenant, or any successor-in-interest to Tenant hereunder, of its obligation to pay rent and to perform all its other obligations under this Lease. Each assignee, other than Landlord, shall be deemed to assume all obligations of Tenant under this Lease and shall be liable jointly and severally with Tenant for the payment of all rent, and for the due performance of all of Tenant's obligations, under this Lease. Such joint and several liability shall not be discharged or impaired by any subsequent modification or extension of this Lease. Consent by Landlord to one or more transfers shall not operate as a waiver or estoppel to the future enforcement by Landlord of its rights under this Lease.

**9.3. SUBLEASE REQUIREMENTS.** Any sublease, license, concession or other occupancy agreement entered into by Tenant shall be subordinate and subject to the provisions of this Lease, and if this Lease is terminated during the term of any such agreement, Landlord shall have the right to: (i) treat such agreement as cancelled and repossess the subject space by any lawful means, or (ii) require that such transferee attorn to and recognize Landlord as its landlord (or licensor, as applicable) under such agreement. Landlord shall not, by reason of such attornment or the collection of sublease rentals, be deemed liable to the subtenant for the performance of any of Tenant's obligations under the sublease. If Tenant is in Default (hereinafter defined), Landlord is irrevocably authorized to direct any transferee under any such agreement to make all payments under such agreement directly to Landlord (which Landlord shall apply towards Tenant's obligations under this Lease) until such Default is cured. No collection or acceptance of rent by Landlord from any transferee shall be deemed a waiver of any provision of Article 9 of this Lease, an approval of any transferee, or a release of Tenant from any obligation under this Lease, whenever accruing. In no event shall Landlord's enforcement of any provision of this Lease against any transferee be deemed a waiver of Landlord's right to enforce any term of this Lease against Tenant or any other person.

#### **ARTICLE 10. INSURANCE AND INDEMNITY**

**10.1. TENANT'S INSURANCE.** Tenant, at its sole cost and expense, shall provide and maintain in effect the insurance described in Exhibit D. Evidence of that insurance must be delivered to Landlord prior to the Commencement Date.

**10.2.** LANDLORD'S INSURANCE. Landlord shall provide the following types of insurance, with or without deductible and in amounts and coverages as may be determined by Landlord in its discretion: property insurance, subject to standard exclusions (such as, but not limited to, earthquake and flood exclusions), covering the Building or Project. In addition, Landlord may, at its election, obtain insurance coverages for such other risks as Landlord or its Mortgagees may from time to time deem appropriate, including earthquake and commercial general liability coverage. Landlord shall not be required to carry insurance of any kind on any tenant improvements or Alterations in the Premises installed by Tenant or its contractors or otherwise removable by Tenant (collectively, "*Tenant Installations*"), or on any trade fixtures, furnishings, equipment, interior plate glass, signs or items of personal property in the Premises, and Landlord shall not be obligated to repair or replace any of the foregoing items should damage occur. All proceeds of insurance maintained by Landlord upon the Building and Project shall be the property of Landlord, whether or not Landlord is obligated to or elects to make any repairs.

**10.3. TENANT'S INDEMNITY.** To the fullest extent permitted by law, but subject to Section 10.5 below, Tenant shall defend, indemnify and hold harmless Landlord and Landlord's agents, employees, lenders, and affiliates, from and against any and all negligence, claims, liabilities, damages, costs or expenses arising either before or after the Commencement Date which arise from or are caused by Tenant's use or occupancy of the Premises, the Building or the Common Areas of the Project, or from the conduct of Tenant's business, or from any activity, work, or thing done, permitted or suffered by Tenant or Tenant's agents, employees, subtenants, vendors, contractors, invitees or licensees in or about the Premises, the Building or the Common Areas of the Project, or from any Default in the performance of any obligation on Tenant's part to be performed under this Lease, or from any act, omission or negligence on the part of Tenant or Tenant's agents, employees, subtenants, vendors, contractors, invitees or licensees. Landlord's defense in any action covered by this Section 10.3 through counsel reasonably satisfactory to Landlord. Notwithstanding the foregoing, Tenant shall not be obligated to indemnify Landlord against any liability or expense to the extent it is ultimately determined that the same was caused by the sole negligence or willful misconduct of Landlord, its agents, contractors or employees.

**10.4.** LANDLORD'S NONLIABILITY. Landlord shall not be liable to Tenant, its employees, agents and invitees, and Tenant hereby waives all claims against Landlord, its employees and agents for loss of or damage to any property, or any injury to any person, resulting from any condition including, but not limited to, acts or omissions (criminal or otherwise) of third parties and/or other tenants of the Project, or their agents, employees or invitees, fire, explosion, falling plaster, steam, gas, electricity, water or rain which may leak or flow from or into any part of the Premises or from the breakage, leakage, obstruction or other defects of the pipes, sprinklers, wires, appliances, plumbing, air conditioning, electrical works or other fixtures in the Building, whether the damage or injury results from conditions arising in the Premises or in other portions of the Building, regardless of the negligence of Landlord, its agents or any and all affiliates of Landlord in connection with the foregoing. It is understood that any such condition may require the temporary evacuation or closure of all or a portion of the Building. Should Tenant elect to receive any service from a concessionaire, licensee or third party tenant of Landlord, Tenant shall not seek recourse against Landlord for any breach or liability of that service provider. Notwithstanding anything to the contrary contained in this Lease, in no event shall Landlord be liable for Tenant's loss or interruption of business or income (including without limitation, Tenant's consequential damages, lost profits or opportunity costs), or for interference with light or other similar intangible interests.

**10.5. W**AIVER OF **SUBROGATION**. Landlord and Tenant each hereby waives all rights of recovery against the other on account of loss and damage occasioned to the property of such waiving party to the extent that the waiving party is entitled to proceeds for such loss and damage under any property insurance policies carried or otherwise required to be carried by this Lease; provided however, that the foregoing waiver shall not apply to the extent of Tenant's obligation to pay deductibles under any such policies and this Lease. By this waiver it is the intent of the parties that neither Landlord nor Tenant shall be liable to any insurance company (by way of subrogation or otherwise) insuring the other party for any loss or damage insured against under any property insurance policies, even though such loss or damage might be occasioned by the negligence of such party, its agents, employees, contractors or invitees. The foregoing waiver by Tenant shall also inure to the benefit of Landlord's management agent for the Building.

# **ARTICLE 11. DAMAGE OR DESTRUCTION**

# 11.1. RESTORATION.

(a) If the Building of which the Premises are a part is damaged as the result of an event of casualty, then subject to the provisions below, Landlord shall repair that damage as soon as reasonably possible unless Landlord reasonably determines that: (i) the Premises have been materially damaged and there is less than 1 year of the Term remaining on the date of the casualty; (ii) any Mortgagee (defined in Section 13.1) requires that the insurance proceeds be applied to the payment of the mortgage debt; or (iii) proceeds necessary to pay the full cost of the repair are not available from Landlord's insurance, including without limitation earthquake insurance. Should Landlord elect not to repair the damage for one of the preceding reasons, Landlord shall so notify Tenant in the "Casualty Notice" (as defined below), and this Lease shall terminate as of the date of delivery of that notice.

(b) As soon as reasonably practicable following the casualty event but not later than 60 days thereafter, Landlord shall notify Tenant in writing ("*Casualty Notice*") of Landlord's election, if applicable, to terminate this Lease. If this Lease is not so terminated, the Casualty Notice shall set forth the anticipated period for repairing the casualty damage. If the anticipated repair period exceeds 270 days and if the damage is so extensive as to reasonably prevent Tenant's substantial use and enjoyment of the Premises, then either party may elect to terminate this Lease by written notice to the other within 10 days following delivery of the Casualty Notice.

(c) In the event that neither Landlord nor Tenant terminates this Lease pursuant to Section 11.1(b), Landlord shall repair all material damage to the Premises or the Building as soon as reasonably possible and this Lease shall continue in effect for the remainder of the Term. Upon notice from Landlord, Tenant shall assign or endorse over to Landlord (or to any party designated by Landlord) all property insurance proceeds payable to Tenant under Tenant's insurance with respect to any Tenant Installations; provided if the estimated cost to repair such Tenant Installations exceeds the amount of insurance proceeds received by Landlord from Tenant's insurance carrier, the excess cost of such repairs shall be paid by Tenant to Landlord prior to Landlord's commencement of repairs. Within 15 days of demand, Tenant shall also pay Landlord for any additional excess costs that are determined during the performance of the repairs to such Tenant Installations.

(d) From and after the 6<sup>th</sup> business day following the casualty event, the rental to be paid under this Lease shall be abated in the same proportion that the Floor Area of the Premises that is rendered unusable by the damage from time to time bears to the total Floor Area of the Premises.

(e) Notwithstanding the provisions of subsections (a), (b) and (c) of this Section 11.1, but subject to Section 10.5, the cost of any repairs shall be borne by Tenant, and Tenant shall not be entitled to rental abatement or termination rights, if the damage is due to the fault or neglect of Tenant or its employees, subtenants, contractors, invitees or representatives. In addition, the provisions of this Section 11.1 shall not be deemed to require Landlord to repair any Tenant Installations, fixtures and other items that Tenant is obligated to insure pursuant to Exhibit D or under any other provision of this Lease.

**11.2.** LEASE GOVERNS. Tenant agrees that the provisions of this Lease, including without limitation Section 11.1, shall govern any damage or destruction and shall accordingly supersede any contrary statute or rule of law.

# **ARTICLE 12. EMINENT DOMAIN**

Either party may terminate this Lease if any material part of the Premises is taken or condemned for any public or quasipublic use under Law, by eminent domain or private purchase in lieu thereof (a "*Taking*") Landlord shall also have the right to terminate this Lease if there is a Taking of any portion of the Building or Project which would have a material adverse effect on Landlord's ability to profitably operate the remainder of the Building. The termination shall be effective as of the effective date of any order granting possession to, or vesting legal title in, the condemning authority. If this Lease is not terminated, Basic Rent and Tenant's Share of Operating Expenses shall be appropriately adjusted to account for any reduction in the square footage of the Building or Premises. All compensation awarded for a Taking shall be the property of Landlord and the right to receive compensation or proceeds in connection with a Taking are expressly waived by Tenant; provided, however, Tenant may file a separate claim for Tenant's personal property and Tenant's reasonable relocation expenses, provided the filing of the claim does not diminish the amount of Landlord's award. If only a part of the Premises is subject to a Taking and this Lease is not terminated, Landlord, with reasonable diligence, will restore the remaining portion of the Premises as nearly as practicable to the condition immediately prior to the Taking. Tenant agrees that the provisions of this Lease shall govern any Taking and shall accordingly supersede any contrary statute or rule of law.

# **ARTICLE 13. SUBORDINATION; ESTOPPEL CERTIFICATE**

**13.1. SUBORDINATION**. Tenant accepts this Lease subject and subordinate to any mortgage(s), deed(s) of trust, ground lease(s) or other lien(s) now or subsequently arising upon the Premises, the Building or the Project, and to renewals, modifications, refinancings and extensions thereof (collectively referred to as a "*Mortgage*"). The party having the benefit of a Mortgage shall be referred to as a "*Mortgagee*". This clause shall be self-operative, but upon request from a Mortgagee, Tenant shall execute a commercially reasonable subordination and attornment agreement in favor of the Mortgagee, provided such agreement provides a non-disturbance covenant benefiting Tenant. Alternatively, a Mortgagee shall have the right at any time to subordinate its Mortgage to this Lease. Upon request, Tenant, without charge, shall attorn to any successor to Landlord's interest in this Lease in the event of a foreclosure of any mortgage. Tenant agrees that any purchaser at a foreclosure sale or lender taking title under a deed in lieu of foreclosure shall not be responsible for any act or omission of a prior landlord, shall not be subject to any offsets or defenses Tenant may have against a prior landlord, and shall not be liable for the return of the Security Deposit not actually recovered by such purchaser nor bound by any rent paid in advance of the calendar month in which the transfer of title occurred; provided that the foregoing shall not release the applicable prior landlord from any liability for those obligations. Tenant acknowledges that Landlord's Mortgagees and their successors-in-interest are intended third party beneficiaries of this Section 13.1.

**13.2. ESTOPPEL CERTIFICATE.** Tenant shall, within 10 days after receipt of a written request from Landlord, execute and deliver a commercially reasonable estoppel certificate in favor of those parties as are reasonably requested, Landlord (including a Mortgagee or a prospective purchaser of the Building or the Project).

#### **ARTICLE 14. DEFAULTS AND REMEDIES**

**14.1. TENANT'S DEFAULTS.** In addition to any other event of default set forth in this Lease, the occurrence of any one or more of the following events shall constitute a "*Default*" by Tenant:

(a) The failure by Tenant to make any payment of Rent required to be made by Tenant, as and when due, where the failure continues for a period of 5 days after written notice from Landlord to Tenant. The term "*Rent*" as used in this Lease shall be deemed to mean the Basic Rent and all other sums required to be paid by Tenant to Landlord pursuant to the terms of this Lease.

(b) The assignment, sublease, encumbrance or other Transfer of the Lease by Tenant, either voluntarily or by operation of law, whether by judgment, execution, transfer by intestacy or testacy, or other means, without the prior written consent of Landlord unless otherwise authorized in Article 9 of this Lease.

(c) The discovery by Landlord that any financial statement provided by Tenant, or by any affiliate, successor or guarantor of Tenant, was materially false.

(d) Except where a specific time period is otherwise set forth for Tenant's performance in this Lease (in which event the failure to perform by Tenant within such time period shall be a Default), the failure or inability by Tenant to observe or perform any of the covenants or provisions of this Lease to be observed or performed by Tenant, other than as specified in any other subsection of this Section 14.1, where the failure continues for a period of 30 days after written notice from Landlord to Tenant. However, if the nature of the failure is such that more than 30 days are reasonably required for its cure, then Tenant shall not be deemed to be in Default if Tenant commences the cure within 30 days, and thereafter diligently pursues the cure to completion.

The notice periods provided herein are in lieu of, and not in addition to, any notice periods provided by law, and Landlord shall not be required to give any additional notice under California Code of Civil Procedure Section 1161, or any successor statute, in order to be entitled to commence an unlawful detainer proceeding.

# 14.2. LANDLORD'S REMEDIES.

(a) Upon the occurrence of any Default by Tenant, then in addition to any other remedies available to Landlord, Landlord may exercise the following remedies:

(i) Landlord may terminate Tenant's right to possession of the Premises by any lawful means, in which case this Lease shall terminate and Tenant shall immediately surrender possession of the Premises to Landlord. Such termination shall not affect any accrued obligations of Tenant under this Lease. Upon termination, Landlord shall have the right to reenter the Premises and remove all persons and property. Landlord shall also be entitled to recover from Tenant:

termination:

(1) The worth at the time of award of the unpaid Rent which had been earned at the time of

(2) The worth at the time of award of the amount by which the unpaid Rent which would have been earned after termination until the time of award exceeds the amount of such loss that Tenant proves could have been reasonably avoided;

(3) The worth at the time of award of the amount by which the unpaid Rent for the balance of the Term after the time of award exceeds the amount of such loss that Tenant proves could be reasonably avoided;

(4) Any other amount necessary to compensate Landlord for all the detriment proximately caused by Tenant's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result from Tenant's default, including, but not limited to, the cost of recovering possession of the Premises, commissions and other expenses of reletting, including necessary repair, renovation, improvement and alteration of the Premises for a new tenant, reasonable attorneys' fees, and any other reasonable costs; and

(5) At Landlord's election, all other amounts in addition to or in lieu of the foregoing as may be permitted by law. Any sum, other than Basic Rent, shall be computed on the basis of the average monthly amount accruing during the 24 month period immediately prior to Default, except that if it becomes necessary to compute such rental before the 24 month period has occurred, then the computation shall be on the basis of the average monthly amount during the shorter period. As used in subparagraphs (1) and (2) above, the "*worth at the time of award*" shall be computed by allowing interest at the rate of 10% per annum. As used in subparagraph (3) above, the "*worth at the time of award*" shall be computed by discounting the amount at the discount rate of the Federal Reserve Bank of San Francisco at the time of award plus 1%.

(ii) Landlord may elect not to terminate Tenant's right to possession of the Premises, in which event Landlord may continue to enforce all of its rights and remedies under this Lease, including the right to collect all rent as it becomes due. Efforts by the Landlord to maintain, preserve or relet the Premises, or the appointment of a receiver to protect the Landlord's interests under this Lease, shall not constitute a termination of the Tenant's right to possession of the Premises. In the event that Landlord elects to avail itself of the remedy provided by this subsection (ii), Landlord shall not unreasonably withhold its consent to an assignment or subletting of the Premises subject to the reasonable standards for Landlord's consent as are contained in this Lease.

The various rights and remedies reserved to Landlord in this Lease or otherwise shall be cumulative and, (b) except as otherwise provided by California law, Landlord may pursue any or all of its rights and remedies at the same time. No delay or omission of Landlord to exercise any right or remedy shall be construed as a waiver of the right or remedy or of any breach or Default by Tenant. The acceptance by Landlord of rent shall not be a (i) waiver of any preceding breach or Default by Tenant of any provision of this Lease, other than the failure of Tenant to pay the particular rent accepted, regardless of Landlord's knowledge of the preceding breach or Default at the time of acceptance of rent, or (ii) a waiver of Landlord's right to exercise any remedy available to Landlord by virtue of the breach or Default. The acceptance of any payment from a debtor in possession, a trustee, a receiver or any other person acting on behalf of Tenant or Tenant's estate shall not waive or cure a Default under Section 14.1. No payment by Tenant or receipt by Landlord of a lesser amount than the rent required by this Lease shall be deemed to be other than a partial payment on account of the earliest due stipulated rent, nor shall any endorsement or statement on any check or letter be deemed an accord and satisfaction and Landlord shall accept the check or payment without prejudice to Landlord's right to recover the balance of the rent or pursue any other remedy available to it. Tenant hereby waives any right of redemption or relief from forfeiture under California Code of Civil Procedure Section 1174 or 1179, or under any successor statute, in the event this Lease is terminated by reason of any Default by Tenant. No act or thing done by Landlord or Landlord's agents during the Term shall be deemed an acceptance of a surrender of the Premises, and no agreement to accept a surrender shall be valid unless in writing and signed by Landlord. No employee of Landlord or of Landlord's agents shall have any power to accept the keys to the Premises prior to the termination of this Lease, and the delivery of the keys to any employee shall not operate as a termination of the Lease or a surrender of the Premises.

**14.3.** LATE PAYMENTS. Any Rent due under this Lease that is not paid to Landlord within 5 business days of the date when due shall bear interest at the maximum rate permitted by law from the date due until fully paid. The payment of interest shall not cure any Default by Tenant under this Lease. In addition, Tenant acknowledges that the late payment by Tenant to Landlord of rent will cause Landlord to incur costs not contemplated by this Lease, the exact amount of which will be extremely difficult and impracticable to ascertain. Those costs may include, but are not limited to, administrative, processing and accounting charges, and late charges which may be imposed on Landlord by the terms of any ground lease, mortgage or trust deed covering the Premises. Accordingly, if any rent due from Tenant shall not be received by Landlord or Landlord's designee within 5 business days after the date due, then Tenant shall pay to Landlord, in addition to the interest provided above, a late charge for each delinquent payment equal to the greater of (i) 5% of that delinquent payment or (ii) \$100.00. Acceptance of a late charge by Landlord shall not constitute a waiver of Tenant's Default with respect to the overdue amount, nor shall it prevent Landlord from exercising any of its other rights and remedies.

**14.4. RIGHT OF LANDLORD TO PERFORM.** If Tenant is in Default of any of its obligations under the Lease, Landlord shall have the right to perform such obligations. Tenant shall reimburse Landlord for the cost of such performance upon demand together with an administrative charge equal to 10% of the cost of the work performed by Landlord.

**14.5. D**EFAULT BY LANDLORD. Landlord shall not be deemed to be in default in the performance of any obligation under this Lease unless and until it has failed to perform the obligation within 30 days after written notice by Tenant to Landlord specifying in reasonable detail the nature and extent of the failure, provided, however, that if the nature of Landlord's obligation is such that more than 30 days are required for its performance, then Landlord shall not be deemed to be in default if it commences performance within the 30 day period and thereafter diligently pursues the cure to completion. Tenant hereby waives any right to terminate or rescind this Lease as a result of any default by Landlord hereunder or any breach by Landlord of any promise or inducement relating hereto, and Tenant agrees that its remedies shall be limited to a suit for actual damages and/or injunction and shall in no event include any consequential damages, lost profits or opportunity costs.

**14.6.** EXPENSES AND LEGAL FEES. Should either Landlord or Tenant bring any action int connection with this Lease, the prevailing party shall be entitled to recover as a part of the action its reasonable attorneys' fees, and all other reasonable costs The prevailing party for the purpose of this paragraph shall be determined by the trier of the facts.

14.7. WAIVER OF JURY TRIAL/JUDICIAL REFERENCE.

(a) LANDLORD AND TENANT EACH ACKNOWLEDGES THAT IT IS AWARE OF AND HAS HAD THE ADVICE OF COUNSEL OF ITS CHOICE WITH RESPECT TO ITS RIGHT TO TRIAL BY JURY, AND EACH PARTY DOES HEREBY EXPRESSLY AND KNOWINGLY WAIVE AND RELEASE ALL SUCH RIGHTS TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY EITHER PARTY HERETO AGAINST THE OTHER (AND/OR AGAINST ITS OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, OR SUBSIDIARY OR AFFILIATED ENTITIES) ON ANY MATTERS WHATSOEVER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS LEASE, TENANT'S USE OR OCCUPANCY OF THE PREMISES, AND/OR ANY CLAIM OF INJURY OR DAMAGE.

(b) In the event that the jury waiver provisions of Section 14.7 (a) are not enforceable under California law, then, unless otherwise agreed to by the parties, the provisions of this Section 14.7 (b) shall apply. Landlord and Tenant agree that any disputes arising in connection with this Lease (including but not limited to a determination of any and all of the issues in such dispute, whether of fact or of law) shall be resolved (and a decision shall be rendered) by way of a general reference as provided for in Part 2, Title 8, Chapter 6 (§§ 638 et. seq.) of the California Code of Civil Procedure, or any successor California statute governing resolution of disputes by a court appointed referee. Nothing within this Section 14.7 shall apply to an unlawful detainer action.

**14.8. SATISFACTION OF JUDGMENT.** The obligations of Landlord do not constitute the personal obligations of the individual partners, trustees, directors, officers, members or shareholders of Landlord or its constituent partners or members. Should Tenant recover a money judgment against Landlord, such judgment shall be satisfied only from the interest of Landlord in the Project and out of the rent or other income from such property receivable by Landlord, and no action for any deficiency may be sought or obtained by Tenant.

#### ARTICLE 15. END OF TERM

**15.1. HOLDING OVER.** If Tenant holds over for any period after the Expiration Date (or earlier termination of the Term) without the prior written consent of Landlord, such tenancy shall constitute a tenancy at sufferance only and a Default by Tenant; such holding over with the prior written consent of Landlord shall constitute a month-to-month tenancy commencing on the 1<sup>st</sup> day following the termination of this Lease and terminating 30 days following delivery of written notice of termination by either Landlord or Tenant to the other. In either of such events, possession shall be subject to all of the terms of this Lease, except that the monthly rental shall be 150% of the total monthly rental for the month immediately preceding the date of termination, subject to Landlord's right to modify same upon 30 days notice to Tenant. The acceptance by Landlord of monthly hold-over rental in a lesser amount shall not constitute a waiver of Landlord's right to recover the full amount due unless otherwise agreed In writing by Landlord. If Tenant fails to surrender the Premises upon the expiration of this Lease despite demand to do so by Landlord, Tenant shall indemnify and hold Landlord harmless from all loss or liability, including without limitation, any claims made by any succeeding tenant relating to such failure to surrender. The foregoing provisions of this Section 15.1 are in addition to and do not affect Landlord's right of re-entry or any other rights of Landlord under this Lease or at law.

**15.2. SURRENDER OF PREMISES; REMOVAL OF PROPERTY.** Upon the Expiration Date or upon any earlier termination of this Lease, Tenant shall quit and surrender possession of the Premises to Landlord in as good order, condition and repair as when received or as hereafter may be improved by Landlord or Tenant, reasonable wear and tear and repairs which are Landlord's obligation excepted, and shall remove or fund to Landlord the cost of removing all wallpapering, voice and/or data transmission cabling installed by or for Tenant and Required Removables, together with all personal property and debris, and shall perform all work required under Section 7.3 of this Lease. If Tenant shall fail to comply with the provisions of this Section 15.2, Landlord may effect the removal and/or make any repairs, and the cost to Landlord shall be additional rent payable by Tenant upon demand.

#### **ARTICLE 16. PAYMENTS AND NOTICES**

All sums payable by Tenant to Landlord shall be paid, without deduction or offset, in lawful money of the United States to Landlord at its address set forth in Item 12 of the Basic Lease Provisions, or at any other place as Landlord may designate in writing. Unless this Lease expressly provides otherwise, as for example in the payment of rent pursuant to Section 4.1, all payments shall be due and payable within 5 days after demand. All payments requiring proration shall be prorated on the basis of the number of days in the pertinent calendar month or year, as applicable. Any notice, election, demand, consent, approval or other communication to be given or other document to be delivered by either party to the other may be delivered to the other party, at the address set forth in Item 12 of the Basic Lease Provisions, by personal service, or by any courier or "overnight" express mailing service. Either party may, by written notice to the other, served in the manner provided in this Article, designate a different address. The refusal to accept delivery of a notice, or the inability to deliver the notice (whether due to a change of address for which notice was not duly given or other good reason), shall be deemed delivery and receipt of the notice as of the date of attempted delivery. If more than one person or entity is named as Tenant under this Lease, service of any notice upon any one of them shall be deemed as service upon all of them.

# **ARTICLE 17. RULES AND REGULATIONS**

Tenant agrees to comply with the Rules and Regulations attached as Exhibit E, and any reasonable and nondiscriminatory amendments, modifications and/or additions as may be adopted and published by written notice to tenants by Landlord for the safety, care, security, good order, or cleanliness of the Premises, Building, Project and/or Common Areas Landlord shall not be liable to Tenant for any violation of the Rules and Regulations or the breach of any covenant or condition in any lease or any other act or conduct by any other tenant, and the same shall not constitute a constructive eviction hereunder. One or more waivers by Landlord of any breach of the Rules and Regulations by Tenant or by any other tenant(s) shall not be a waiver of any subsequent breach of that rule or any other. Tenant's failure to keep and observe the Rules and Regulations shall constitute a default under this Lease. In the case of any conflict between the Rules and Regulations and this Lease, this Lease shall be controlling.

#### **ARTICLE 18. BROKER'S COMMISSION**

The parties recognize as the broker(s) who negotiated this Lease the firm(s) whose name(s) is (are) stated in Item 10 of the Basic Lease Provisions, and agree that Landlord shall be responsible for the payment of brokerage commissions to those broker(s) unless otherwise provided in this Lease. It is understood that Landlord's Broker represents only Landlord in this transaction and Tenant's Broker (if any) represents only Tenant. Each party warrants that it has had no dealings with any other real estate broker or agent in connection with the negotiation of this Lease, and agrees to indemnify and hold the other party harmless from any cost, expense or liability (including reasonable attorneys' fees) for any compensation, commissions or charges claimed by any other real estate broker or agent employed or claiming to represent or to have been employed by the indemnifying party in connection with the negotiation of this Lease. The foregoing agreement shall survive the termination of this Lease.

#### **ARTICLE 19. TRANSFER OF LANDLORD'S INTEREST**

In the event of any transfer of Landlord's interest in the Premises, the transferor shall be automatically relieved of all obligations on the part of Landlord accruing under this Lease from and after the date of the transfer, provided that Tenant is duly notified of the transfer. Any funds held by the transferor in which Tenant has an interest, including without limitation, the Security Deposit, shall be turned over, subject to that interest, to the transferee. No Mortgagee to which this Lease is or may be subordinate shall be responsible in connection with the Security Deposit unless the Mortgagee actually receives the Security Deposit. It is intended that the covenants and obligations contained in this Lease on the part of Landlord shall, subject to the foregoing, be binding on Landlord, its successors and assigns, only during and in respect to their respective successive periods of ownership.

#### **ARTICLE 20. INTERPRETATION**

**20.1. NUMBER.** Whenever the context of this Lease requires, the words "*Landlord*" and "*Tenant*" shall include the plural as well as the singular.

**20.2. H**EADINGS. The captions and headings of the articles and sections of this Lease are for convenience only, are not a part of this Lease and shall have no effect upon its construction or interpretation.

**20.3. JOINT AND SEVERAL LIABILITY.** If more than one person or entity is named as Tenant, the obligations imposed upon each shall be joint and several and the act of or notice from, or notice or refund to, or the signature of, any one or more of them shall be binding on all of them with respect to the tenancy of this Lease, including, but not limited to, any renewal, extension, termination or modification of this Lease.

**20.4. S**UCCESSORS. Subject to Sections 13.1 and 22.3 and to Articles 9 and 19 of this Lease, all rights and liabilities given to or imposed upon Landlord and Tenant shall extend to and bind their respective heirs, executors, administrators, successors and assigns. Nothing contained in this Section 20.4 is intended, or shall be construed, to grant to any person other than Landlord and Tenant and their successors and assigns any rights or remedies under this Lease.

**20.5. TIME OF ESSENCE.** Time is of the essence with respect to the performance of every provision of this Lease in which time of performance is a factor.

**20.6. CONTROLLING LAW/VENUE**. This Lease shall be governed by and interpreted in accordance with the laws of the State of California. Should any litigation be commenced between the parties in connection with this Lease, such action shall be prosecuted in the applicable State Court of California in the county in which the Building is located.

**20.7. SEVERABILITY.** If any term or provision of this Lease, the deletion of which would not adversely affect the receipt of any material benefit by either party or the deletion of which is consented to by the party adversely affected, shall be held invalid or unenforceable to any extent, the remainder of this Lease shall not be affected and each term and provision of this Lease shall be valid and enforceable to the fullest extent permitted by law.

**20.8. WAIVER.** One or more waivers by Landlord or Tenant of any breach of any term, covenant or condition contained in this Lease shall not be a waiver of any subsequent breach of the same or any other term, covenant or condition. Consent to any act by one of the parties shall not be deemed to render unnecessary the obtaining of that party's consent to any subsequent act. No breach of this Lease shall be deemed to have been waived unless the waiver is in a writing signed by the waiving party.

**20.9. INABILITY TO PERFORM.** In the event that either party shall be delayed or hindered in or prevented from the performance of any work or in performing any act required under this Lease by reason of any cause beyond the reasonable control of that party, then the performance of the work or the doing of the act shall be excused for the period of the delay and the time for performance shall be extended for a period equivalent to the period of the delay. The provisions of this Section 20.9 shall not operate to excuse Tenant from the prompt payment of Rent.

**20.10.** ENTIRE AGREEMENT. This Lease and its exhibits and other attachments cover in full each and every agreement of every kind between the parties concerning the Premises, the Building, and the Project, and all preliminary negotiations, oral agreements, understandings and/or practices, except those contained in this Lease, are superseded and of no further effect. Tenant waives its rights to rely on any representations or promises made by Landlord or others which are not contained in this Lease. No verbal agreement or implied covenant shall be held to modify the provisions of this Lease, any statute, law, or custom to the contrary notwithstanding.

**20.11. QUIET ENJOYMENT.** Upon the observance and performance of all the covenants, terms and conditions on Tenant's part to be observed and performed, and subject to the other provisions of this Lease, Tenant shall have the right of quiet enjoyment and use of the Premises for the Term without hindrance or interruption by Landlord or any other person claiming by or through Landlord.

**20.12. SURVIVAL.** All covenants of Landlord or Tenant which reasonably would be intended to survive the expiration or sooner termination of this Lease, including without limitation any warranty or indemnity hereunder, shall so survive and continue to be binding upon and inure to the benefit of the respective parties and their successors and assigns.

# **ARTICLE 21. EXECUTION AND RECORDING**

**21.1. COUNTERPARTS; DIGITAL SIGNATURES.** This Lease may be executed in one or more counterparts, each of which shall constitute an original and all of which shall be one and the same agreement. The parties agree to accept a digital image (including but not limited to an image in the form of a PDF, JPEG, GIF file, or other e-signature) of this Lease, if applicable, reflecting the execution of one or both of the parties, as a true and correct original.

**21.2. CORPORATE AND PARTNERSHIP AUTHORITY.** If Tenant is a corporation, limited liability company or partnership, each individual executing this Lease on behalf of the entity represents and warrants that such individual is duly authorized to execute and deliver this Lease and that this Lease is binding upon the corporation, limited liability company or partnership in accordance with its terms. Tenant shall, at Landlord's request, deliver a certified copy of its organizational documents or an appropriate certificate authorizing or evidencing the execution of this Lease.

**21.3. EXECUTION OF LEASE; NO OPTION OR OFFER.** The submission of this Lease to Tenant shall be for examination purposes only, and shall not constitute an offer to or option for Tenant to lease the Premises. Execution of this Lease by Tenant and its return to Landlord shall not be binding upon Landlord, notwithstanding any time interval, until Landlord has in fact executed and delivered this Lease to Tenant, it being intended that this Lease shall only become effective upon execution by Landlord and delivery of a fully executed counterpart to Tenant.

**21.4. R**ECORDING. Tenant shall not record this Lease without the prior written consent of Landlord. Tenant, upon the request of Landlord, shall execute and acknowledge a "*short form*" memorandum of this Lease for recording purposes.

**21.5. AMENDMENTS.** No amendment or mutual termination of this Lease shall be effective unless in writing signed by authorized signatories of Tenant and Landlord, or by their respective successors in interest. No actions, policies, oral or informal arrangements, business dealings or other course of conduct by or between the parties shall be deemed to modify this Lease in any respect.

#### **ARTICLE 22. MISCELLANEOUS**

**22.1. NONDISCLOSURE OF LEASE TERMS.** Tenant acknowledges that the content of this Lease and any related documents are confidential information. Except to the extent disclosure is required by law, Tenant shall keep such confidential information strictly confidential and shall not disclose such confidential information to any person or entity other than Tenant's financial, legal and space-planning consultants, provided, however, that Tenant may disclose the terms to prospective subtenants or assignees under this Lease or pursuant to legal requirement.

**22.2. TENANT'S FINANCIAL STATEMENTS.** The application, financial statements and tax returns, if any, submitted and certified to by Tenant as an accurate representation of its financial condition have been prepared, certified and submitted to Landlord as an inducement and consideration to Landlord to enter into this Lease. Tenant shall during the Term furnish Landlord with current annual financial statements accurately reflecting Tenant's financial condition upon written request from Landlord within 10 days following Landlord's request; provided, however, that so long as Tenant is a publicly traded corporation on a nationally recognized stock exchange, the foregoing obligation to deliver the statements shall be waived.

**22.3. MORTGAGEE PROTECTION.** No act or failure to act on the part of Landlord which would otherwise entitle Tenant to be relieved of its obligations hereunder or to terminate this Lease shall result in such a release or termination unless (a) Tenant has given notice by registered or certified mail to any Mortgagee of a Mortgage covering the Building whose address has been furnished to Tenant and (b) such Mortgagee is afforded a reasonable opportunity to cure the default by Landlord (which shall in no event be less than 60 days), including, if necessary to effect the cure, time to obtain possession of the Building by power of sale or judicial foreclosure provided that such foreclosure remedy is diligently pursued. Tenant shall comply with any written directions by any Mortgagee to pay Rent due hereunder directly to such Mortgagee without determining whether a default exists under such Mortgagee's Mortgage.

**22.4. SDN UST**. Tenant hereby represents and warrants that neither Tenant nor any officer, director, employee, partner, member or other principal of Tenant (collectively, "*Tenant Parties*") is listed as a Specially Designated National and Blocked Person ("*SDN*") on the list of such persons and entities issued by the U.S. Treasury Office of Foreign Assets Control (OFAC). In the event Tenant or any Tenant Party is or becomes listed as an SDN, Tenant shall be deemed in breach of this Lease and Landlord shall have the right to terminate this Lease immediately upon written notice to Tenant.

#### LANDLORD:

# **GHOST MEDIA GROUP, LLC**, a Nevada limited liability company

# By/s/ Steven M. Case

Steven M. Case Executive Vice President, Office Properties

# By/s/ Michael T. Bennett

Michael T. Bennett Senior Vice President, Operations, Office Properties

#### TENANT:

THE IRVINE COMPANY LLC, a Delaware limited liability company

By/s/ Justin Hartfield

Printed Name Justin Hartfield

\_\_\_\_\_

Title Manager

By\_\_\_\_\_

Printed Name

Title

Ехнівіт А

# **DESCRIPTION OF PREMISES**

# 41 DISCOVERY

#### Ехнівіт В

## Operating Expenses (Net)

From and after the Commencement Date. Tenant shall pay to Landlord, as additional rent, Tenant's Share of (a) Operating Expenses, incurred by Landlord in the operation of the Building and the Project. The term "Tenant's Share" means 100% of the Operating Expenses determined by Landlord to benefit or relate substantially to the Building, plus that portion of any Operating Expenses determined by multiplying the cost of such item by a fraction, the numerator of which is the Floor Area and the denominator of which is the total rentable square footage, as determined from time to time by Landlord, of (i) the Building, for expenses determined by Landlord to benefit or relate substantially to the Building rather than the entire Project, and (ii) all or some of the buildings in the Project, for expenses determined by Landlord to benefit or relate substantially to all or some of the buildings in the Project rather than any specific building. Notwithstanding the foregoing, for the initial 6 months of the Term of the Lease only. Tenant's Share of Operating Expenses shall be reduced to 50% of the amount determined in the foregoing. Landlord reserves the right to allocate to the entire Project any Operating Expenses which may benefit or substantially relate to a particular building within the Project in order to maintain greater consistency of Operating Expenses among buildings within the Project. In the event that Landlord determines that the Premises or the Building incur a non-proportional benefit from any expense, or is the non-proportional cause of any such expense, Landlord may allocate a greater percentage of such Operating Expense to the Premises or the Building. In the event that any management and/or overhead fee payable or imposed by Landlord for the management of Tenant's Premises is calculated as a percentage of the rent payable by Tenant and other tenants of Landlord, then the full amount of such management and/or overhead fee which is attributable to the rent paid by Tenant shall be additional rent payable by Tenant, in full, provided, however, that Landlord may elect to include such full amount as part of Tenant's Share of Operating Expenses.

(b) Landlord estimates that the amount of Tenant's Share of Operating Expenses for the Expense Recovery Period ended June 30, 2014 shall be \$.42 per rentable square foot of the Premises. Commencing prior to the start of the first full "*Expense Recovery Period*" of the Lease (as defined in Item 7 of the Basic Lease Provisions), and prior to the start of each full or partial Expense Recovery Period thereafter, Landlord shall give Tenant a written estimate of the amount of Tenant's Share of Operating Expenses for the applicable Expense Recovery Period. Tenant shall pay the estimated amounts to Landlord in equal monthly installments, in advance, concurrently with payments of Basic Rent. If Landlord has not furnished its written estimate for any Expenses in effect during the prior Expense Recovery Period; provided that when the new estimate is delivered to Tenant, Tenant shall, at the next monthly payment date, pay any accrued estimated Tenant's Share of Operating Expenses based upon the new estimate. Not more than once in any calendar year, Landlord may from time to time change the Expense Recovery Period to reflect a calendar year or a new fiscal year of Landlord, as applicable, in which event Tenant's Share of Operating Expenses shall be equitably prorated for any partial year.

(c) Within 180 days after the end of each Expense Recovery Period, Landlord shall furnish to Tenant a statement (a "*Reconciliation Statement*") showing in reasonable detail the actual or prorated Tenant's Share of Operating Expenses incurred by Landlord during such Expense Recovery Period, and the parties shall within 30 days thereafter make any payment or allowance necessary to adjust Tenant's estimated payments of Tenant's Share of Operating Expenses, if any, to the actual Tenant's Share of Operating Expenses as shown by the Reconciliation Statement. Any delay or failure by Landlord in delivering any Reconciliation Statement shall not constitute a waiver of Landlord's right to require Tenant to pay Tenant's Share of Operating Expenses pursuant hereto. Any amount due Tenant shall be credited against installments next coming due under this Exhibit B, and any deficiency shall be paid by Tenant together with the next installment. Should Tenant fail to object in writing to Landlord's determination of Tenant's Share of Operating Expenses, or fail to give written notice of its intent to audit Landlord's Operating Expenses pursuant to the provisions of the next succeeding paragraph, within 120 days following delivery of Landlord's Reconciliation Statement, Landlord's determination of Tenant's Share of Operating Expenses for the applicable Expense Recovery Period shall be conclusive and binding on Tenant for all purposes and any future claims by Tenant to the contrary shall be barred.

Provided no Default has occurred, Tenant shall have the right to cause a certified public accountant, engaged on a noncontingency fee basis, to audit Operating Expenses by inspecting Landlord's general ledger of expenses not more than once during any Expense Recovery Period. However, to the extent that insurance premiums or any other component of Operating Expenses is determined by Landlord on the basis of an internal allocation of costs utilizing information Landlord in good faith deems proprietary, such expense component shall not be subject to audit so long as it does not exceed the amount per square foot typically imposed by landlords of other first class business parks in the vicinity of the Project. Tenant shall give notice to Landlord of Tenant's intent to audit within 120 days after delivery of Landlord's Reconciliation Statement which sets forth Tenant's Share of Landlord's actual Operating Expenses to Tenant Such audit shall be conducted at a mutually agreeable time during normal business hours at the office of Landlord or its management agent where such accounts are maintained. If Tenant's audit determines that actual Operating Expenses have been overstated by more than 5%, then subject to Landlord's right to review and/or contest the audit results, Landlord shall reimburse Tenant for the reasonable out-of-pocket costs of such audit. Tenant's rent shall be appropriately adjusted to reflect any overstatement in Operating Expenses. All of the information obtained by Tenant and/or its auditor in connection with such audit, as well as any compromise, settlement, or adjustment reached between Landlord and Tenant as a result thereof, shall be held in strict confidence and, except as may be required pursuant to litigation, shall not be disclosed to any third party, directly or indirectly, by Tenant or its auditor or any of their officers, agents or employees. Landlord may require Tenant's auditor to execute a separate confidentiality agreement affirming the foregoing as a condition precedent to any audit. In the event of a violation of this confidentiality covenant in connection with any audit, then in addition to any other legal or equitable remedy available to Landlord, Tenant shall forfeit its right to any reconciliation or cost reimbursement payment from Landlord due to said audit (and any such payment theretofore made by Landlord shall be promptly returned by Tenant), and Tenant shall have no further audit rights under this Lease. Notwithstanding the foregoing, Tenant shall have no right of audit with respect to any Expense Recovery Period unless the total Operating Expenses per square foot for such Expense Recovery Period, as set forth in Landlord's annual Reconciliation Statement, exceed the total Operating Expenses per square foot during the initial Expense Recovery Period during the Term, as increased by the percentage change in the United States Department of Labor, Bureau of Labor Statistics, Consumer Price Index for all Urban Consumers, Los Angeles -Riverside — Orange County Area Average, all items (1982-84 = 100) (the "Index"), which change in the Index shall be measured by comparing the Index published for January of the initial Expense Recovery Period during the Term with the Index published for January of the applicable Expense Recovery Period.

(d) Even though this Lease has terminated and the Tenant has vacated the Premises, when the final determination is made of Tenant's Share of Operating Expenses for the Expense Recovery Period in which this Lease terminates, Tenant shall within 30 days of written notice pay the entire increase over the estimated Tenant's Share of Operating Expenses already paid. Conversely, any overpayment by Tenant shall be rebated by Landlord to Tenant not later than 30 days after such final determination. However, in lieu thereof, Landlord may deliver a reasonable estimate of the anticipated reconciliation amount to Tenant prior to the Expiration Date of the Term, in which event the appropriate party shall fund the amount by the Expiration Date.

(e) If, at any time during any Expense Recovery Period (but not more than once during each Expense Recovery Period), any one or more of the Operating Expenses are increased to a rate(s) or amount(s) in excess of the rate(s) or amount(s) used in calculating the estimated Tenant's Share of Operating Expenses for the year, then the estimate of Tenants Share of Operating Expenses may be increased by written notice from Landlord for the month in which such rate(s) or amount(s) becomes effective and for all succeeding months by an amount equal to the estimated amount of Tenant's Share of the increase. Landlord shall give Tenant written notice of the amount or estimated amount of the increase, the month in which the increase will become effective, Tenant's Share thereof and the months for which the payments are due. Tenant shall pay the increase to Landlord as part of the Tenant's monthly payments of estimated expenses as provided in paragraph (b) above, commencing with the month in which effective.

(f) The term "*Operating Expenses*" shall mean and include all Project Costs, as defined in Section (g) below, and Property Taxes, as defined in Section (h) below.

The term "Project Costs" shall mean all expenses of operation, management, repair, replacement and (g) maintenance of the Building and the Project, including without limitation all appurtenant Common Areas (as defined in Section 6.2 of the Lease), and shall include the following charges by way of illustration but not limitation: water and sewer charges; insurance premiums, deductibles, or reasonable premium equivalents or deductible equivalents should Landlord elect to self insure any risk that Landlord is authorized to insure hereunder; license, permit, and inspection fees; light; power, window washing; trash pickup; janitorial services to any interior Common Areas; heating, ventilating and air conditioning; supplies; materials; equipment; tools; reasonable fees for consulting services; access control/security costs, inclusive of the reasonable cost of improvements made to enhance access control systems and procedures; establishment of reasonable reserves for replacement of the roof of the Building; costs incurred in connection with compliance with any laws or changes in laws applicable to the Building or the Project; the cost of any capital improvements or replacements (other than tenant improvements for specific tenants) to the extent of the amortized amount thereof over the useful life of such capital improvements or replacements (or, if such capital improvements or replacements are anticipated to achieve a cost savings as to the Operating Expenses, any shorter estimated period of time over which the cost of the capital improvements or replacements would be recovered from the estimated cost savings) calculated at a market cost of funds, all as determined by Landlord, for each year of useful life or shorter recovery period of such capital expenditure whether such capital expenditure occurs during or prior to the Term; costs associated with the maintenance of an air conditioning, heating and ventilation service agreement, and maintenance of any communications or networked data transmission equipment, conduit, cabling, wiring and related telecommunications facilitating automation and control systems, remote telecommunication or data transmission infrastructure within the Building and/or the Project, and any other maintenance, repair and replacement costs associated with such infrastructure; capital costs associated with a requirement related to demands on utilities by Project tenants, including without limitation the cost to obtain additional voice, data and modem connections; labor; reasonably allocated wages and salaries, fringe benefits, and payroll taxes for administrative and other personnel directly applicable to the Building and/or Project, including both Landlord's personnel and outside personnel; any expense incurred pursuant to Sections 6.1, 6.2, 7.2, 10.2, and Exhibits C and F of the Lease; and reasonable and marketcompetitive overhead and/or management fees for the professional operation of the Project. It is understood and agreed that Project Costs may include competitive charges for direct services (including, without limitation, management and/or operations services) provided by any subsidiary, division or affiliate of Landlord.

(h) The term "*Property Taxes*" as used herein shall include any form of federal, state, county or local government or municipal taxes, fees, charges or other impositions of every kind (whether general, special, ordinary or extraordinary) related to the ownership, leasing or operation of the Premises, Building or Project, including without limitation, the following: (i) all real estate taxes or personal property taxes levied against the Premises, the Building or Project, as such property taxes may be reassessed from time to time; and (ii) other taxes, charges and assessments which are levied with respect to this Lease or to the Building and/or the Project, and any improvements, fixtures and equipment and other property of Landlord located in the Building and/or the Project, (iii) all assessments and fees for public improvements, services, and facilities and impacts thereon, including without limitation arising out of any Community Facilities Districts, "Mello Roos" districts, similar assessment districts, and any traffic impact mitigation assessments or fees; (iv) any tax, surcharge or assessment which shall be levied in addition to or in lieu of real estate or personal property taxes, and (v) taxes based on the receipt of rent (including gross receipts or sales taxes applicable to the receipt of rent), and (vi) costs and expenses incurred in contesting the amount or validity of any Property Tax by appropriate proceedings. Notwithstanding the foregoing, general net income or franchise taxes imposed against Landlord shall be excluded.

- (i) Notwithstanding the foregoing provisions of this Exhibit B, Operating Expenses shall exclude the following:
- (1) All costs and expenses of operation of any child care, health club, restaurants and retail space in the Building;

(2) The wages and benefits of any employee who does not devote substantially all of his or her employed time to the Project unless such wages and benefits are prorated to reflect time spent on operating and managing the Project vis-a-vis time spent on matters unrelated to operating and managing the Project; provided that in no event shall Operating Costs include wages and/or benefits attributable to personnel above the level of portfolio property manager or chief engineer;

(3) Interest on debt or amortization on any Mortgage or Mortgages encumbering the Building;

(4) All costs relating to activities for the marketing, solicitation and execution or renewal of leases of space in the Project, including, without limitation, accounting and legal fees, advertising, printing costs and brochures, space planning, tenant allowances, leasehold improvements and other tenant concessions;

(5) Costs associated with the sale or refinancing of the Project, including, without limitation, attorneys' fees, accounting costs, closing costs, consulting or brokerage commissions, origination fees or points, and interest cost or charges;

(6) Costs associated with the acquisition of the fee, ground lease (including payments due under a ground lease), air rights or development rights with respect to the Project;

(7) Cost of decorating, redecorating, or tenant installations incurred in connection with preparing rentable space for a new tenant (or retaining a tenant);

(8) Expenses in connection with services or other benefits which are not provided to Tenant or for which Tenant is charged for directly but which are provided to another tenant or occupant of the Project;

(9) Costs incurred by Landlord for repairs, replacements and/or restoration to or of the Building to the extent that Landlord is actually reimbursed by insurance or condemnation proceeds or by tenants (other than through Operating Expense pass-through), warrantors or other third persons;

(10) Costs incurred by Landlord for improvements or replacements (including structural additions), repairs, equipment and tools which are of a "*capital*" nature and/or which are considered "*capital*" improvements or replacements under GAAP, except to the extent included in Project Costs by the express terms of Section (f) of this Exhibit B;

(11) Overhead and profit increments paid to subsidiaries or affiliates of Landlord for services provided to the Building or any other portion of the Project to the extent the same exceeds the costs that would generally be charged for such services if rendered on a competitive basis (based upon a standard of similar office buildings in the general market area of the Premises) by unaffiliated third parties capable of providing such service;

(12) The cost of alterations of rentable space in the Building leased to Tenant and other tenants;

(13) Costs arising from the gross negligence or intentional misconduct of Landlord or its employees, contractors or agents,

(14) Costs incurred to remove, remedy, contain, or treat any hazardous material, which hazardous material is brought onto the Project (A) before the date of this Lease and (B) after the date hereof by Landlord or any other tenant of the Project or any other person other than Tenant, its employees, agents, licensees, subtenants or invitees, and is of such a nature, at that time, that a federal, state or municipal governmental authority, if it had then had knowledge of the presence of such hazardous material, in the state, and under the conditions, that it then exists on the Project, would have then required the removal of such hazardous material or other remedial or containment action with respect thereto;

<sup>4</sup> 

period;

(15) Penalties and interest charges as a result of Landlord not paying bills when due or within any grace

Electric power costs or other utility costs for which any tenant directly contracts with the utility

- (16) Ground rent or similar payments to a ground lessor;
- (17) Costs related to Landlord's charitable or political contributions;

(18) Attorneys' fees and other costs and expenses incurred in connection with negotiations or disputes with present or prospective tenants or other occupants of the Project, except those attorneys' fees and other costs and expenses incurred in connection with negotiations, disputes or claims relating to items of Operating Expenses, enforcement of rules and regulations of the Project and such other matters relating to the maintenance of standards required of Landlord;

(19) Damage and repairs due to Landlord's, its employees', servants' or agents', gross negligence or willful misconduct;

company; and

(20)

(21) All costs associated with the operation of the business of the ownership or entity which constitutes "Landlord", as distinguished from the costs of operating, maintaining, repairing, replacing and managing the Building or Project, including, but not limited to, partnership accounting and legal matters, costs of defending any lawsuits with any mortgagee (except as the actions of Tenant may be in issue), costs of selling, syndicating, financing, mortgaging or hypothecating any of Landlord's interest in the Building, costs of any disputes between Landlord and its employees (if any) not engaged in Building operation, disputes of Landlord with Building management, or outside fees paid in connection with disputes with other tenants.

It is understood that Operating Costs shall be reduced by all cash discounts, trade discounts, or quantity discounts received by Landlord or Landlord's managing agent in the purchase of any goods, utilities, or services in connection with the operation of the Building. Landlord shall make payments for goods, utilities and services in a timely manner to obtain the maximum possible discount. In the calculation of any expenses hereunder, it is understood that no expense shall be charged more than once. Landlord shall use its best efforts to effect an equitable proration of bills for services rendered to the Building and to any other property owned by Landlord. Landlord agrees to keep books and records showing the Operating Costs in accordance with a system of accounts and accounting practices consistently maintained on a year-to-year basis.

#### Ехнівіт С

## UTILITIES AND SERVICE

Tenant shall be responsible for and shall pay promptly, directly to the appropriate supplier, all charges for electricity metered to the Premises, telephone, telecommunications service, janitorial service, interior landscape maintenance and all other utilities, materials and services furnished directly to Tenant or the Premises or used by Tenant in, on or about the Premises during the Term, together with any taxes thereon. Landlord shall make a reasonable determination of Tenant's proportionate share of the cost of water, gas, sewer, refuse pickup and any other utilities and services that are not separately metered to the Premises and services, and Tenant shall pay such amount to Landlord, as an item of additional rent, within 10 days after delivery of Landlord's statement or invoice therefor. Alternatively, Landlord may elect to include such cost in the definition of Project Costs in which event Tenant shall pay Tenant's proportionate share of such costs in the manner set forth in Section 4.2. Tenant shall also pay to Landlord as an item of additional rent, within 10 days after delivery of Landlord's statement or invoice therefor, Landlord's "standard charges" (as hereinafter defined, which shall be in addition to the electricity charge paid to the utility provider) for "after hours" usage by Tenant of each HVAC unit servicing the Premises. "After hours" shall mean more than 3,396 hours of usage during each calendar year of the Term, as equitably prorated for any partial calendar year. "After hours" usage shall be determined based upon the operation of the applicable HVAC unit during each calendar year on a "non-cumulative" basis (that is, without regard to Tenant's usage or nonusage of other unit(s) serving the Premises, or of the applicable unit during other years of the Term). As used herein, "standard charges" shall mean \$10.00 per hour for each hour of "after hours" use ( in addition to the applicable electricity charges paid to the utility provider).

#### Exhibit D

## **TENANT'S INSURANCE**

The following requirements for Tenant's insurance shall be in effect during the Term, and Tenant shall also cause any subtenant to comply with the requirements. Landlord reserves the right to adopt reasonable nondiscriminatory modifications and additions to these requirements.

1. Tenant shall maintain, at its sole cost and expense, during the entire Term: (i) commercial general liability insurance with respect to the Premises and the operations of Tenant in, on or about the Premises, on a policy form that is at least as broad as Insurance Service Office (ISO) CGL 00 01 (if alcoholic beverages are sold on the Premises, liquor liability shall be explicitly covered), which policy(ies) shall be written on an "occurrence" basis and for not less than \$2,000,000 combined single limit per occurrence for bodily injury, death, and property damage liability; (ii) workers' compensation insurance coverage as required by law, together with employers' liability insurance coverage of at least \$1,000,000 each accident and each disease; (iii) with respect to Alterations constructed by Tenant under this Lease, builder's risk insurance, in an amount equal to the replacement cost of the work; and (iv) insurance against fire, vandalism, malicious mischief and such other additional perils as may be included in a standard "special form" policy, insuring all Alterations, trade fixtures, furnishings, equipment and items of personal property in the Premises, in an amount equal to not less than 90% of their replacement cost (with replacement cost endorsement), which policy shall also include business interruption coverage in an amount sufficient to cover 1 year of loss. In no event shall the limits of any policy be considered as limiting the liability of Tenant under this Lease.

2. All policies of insurance required to be carried by Tenant pursuant to this Exhibit D shall be written by insurance companies authorized to do business in the State of California and with a genera( policyholder rating of not less than "**VIII**" in the most current Best's Insurance Report. The deductible or other retained limit under any policy carried by Tenant shall be commercially reasonable, and Tenant shall be responsible for payment of such deductible or retained limit with waiver of subrogation in favor of Landlord. Any insurance required of Tenant may be furnished by Tenant under any blanket policy carried by it or under a separate policy. A certificate of insurance, certifying that the policy has been issued, provides the coverage required by this Exhibit and contains the required provisions, together with endorsements acceptable to Landlord evidencing the waiver of subrogation and additional insured provisions required below, shall be delivered to Landlord prior to the date Tenant is given the right of possession of the Premises. Proper evidence of the renewal of any insurance coverage shall also be delivered to Landlord not less than thirty (30) days prior to the expiration of the coverage. In the event of a loss covered by any policy under which Landlord is an additional insured, Landlord shall be entitled to review a copy of such policy.

3. Tenant's commercial general liability insurance shall contain a provision that the policy shall be primary to and noncontributory with any policies carried by Landlord, together with a provision including Landlord and any other parties in interest designated by Landlord as additional insureds. Tenant's policies described in Subsections 1 (ii), (iii) and (iv) above shall each contain a waiver by the insurer of any right to subrogation against Landlord, its agents, employees, contractors and representatives. Tenant also waives its right of recovery for any deductible or retained limit under same policies enumerated above. All of Tenant's policies shall contain a provision that the insurer will not cancel or change the coverage provided by the policy without first giving Landlord 30 days prior written notice. Tenant shall also name Landlord as an additional insured on any excess or umbrella liability insurance policy carried by Tenant.

## NOTICE TO TENANT: IN ACCORDANCE WITH THE TERMS OF THIS LEASE, TENANT MUST PROVIDE EVIDENCE OF THE REQUIRED INSURANCE TO LANDLORD'S MANAGEMENT AGENT PRIOR TO BEING AFFORDED ACCESS TO THE PREMISES.

#### Ехнівіт Е

## **RULES AND REGULATIONS**

The following Rules and Regulations shall be in effect at the Building. Landlord reserves the right to adopt reasonable nondiscriminatory modifications and additions at any time. In the case of any conflict between these regulations and the Lease, the Lease shall be controlling.

1. The sidewalks, halls, passages, elevators, stairways, and other common areas shall not be obstructed by Tenant or used by it for storage, for depositing items, or for any purpose other than for ingress to and egress from the Premises. Should Tenant have access to any balcony or patio area, Tenant shall not place any furniture other personal property in such area without the prior written approval of Landlord.

2. Neither Tenant nor any employee or contractor of Tenant shall go upon the roof of the Building without the prior written consent of Landlord.

3. [Intentionally Deleted]

4. No antenna or satellite dish shall be installed by Tenant without the prior written agreement of Landlord.

5. The sashes, sash doors, windows, glass lights, solar film and/or screen, and any lights or skylights that reflect or admit light into the halls or other places of the Building shall not be covered or obstructed. If Landlord, by a notice in writing to Tenant, shall object to any curtain, blind, tinting, shade or screen attached to, or hung in, or used in connection with, any window or door of the Premises, the use of that curtain, blind, tinting, shade or screen shall be Immediately discontinued and removed by Tenant. No awnings shall be permitted on any part of the Premises.

6. The installation and location of any unusually heavy equipment in the Premises, including without limitation file storage units, safes and electronic data processing equipment, shall require the prior written approval of Landlord. The moving of large or heavy objects shall occur only between those hours as may be designated by, and only upon previous notice to, Landlord. No freight, furniture or bulky matter of any description shall be received into or moved out of the lobby of the Building or carried in any elevator other than the freight elevator (if available) designated by Landlord unless approved in writing by Landlord.

7. Any pipes or tubing used by Tenant to transmit water to an appliance or device in the Premises must be made of copper or stainless steel, and in no event shall plastic tubing be used for that purpose.

8. Tenant shall not place any lock(s) on any door in the Premises or Building without Landlord's prior written consent, which consent shall not be unreasonably withheld. Upon the termination of its tenancy, Tenant shall deliver to Landlord all the keys to offices, rooms and toilet rooms and all access cards which shall have been furnished to Tenant or which Tenant shall have had made.

9. Tenant shall not install equipment requiring electrical or air conditioning service in excess of that to be provided by Landlord under the Lease without prior written approval from Landlord.

10. Tenant shall not use space heaters within the Premises.

11. Tenant shall not do or permit anything to be done in the Premises, or bring or keep anything in the Premises, which shall in any way increase the insurance on the Building, or on the property kept in the Building, or interfere with the rights of other tenants, or conflict with any government rule or regulation.

12. Tenant shall not use or keep any foul or noxious gas or substance In the Premises.

13. Tenant shall not permit the Premises to be occupied or used in a manner offensive or objectionable to Landlord or other occupants of the Building by reason of noise, odors and/or vibrations, or interfere in any way with other tenants or those having business with other tenants.

14. Tenant shall not permit any animals or birds be kept by Tenant in or about the Building.

15. Neither Tenant nor its employees, agents, contractors, invitees or licensees shall bring any firearm, whether loaded or unloaded, into the Project at any time.

16. Smoking anywhere within the Premises or Building is strictly prohibited, and Landlord may enforce such prohibition pursuant to Landlord's leasehold remedies. Smoking is permitted outside the Building and within the project only in areas designated by Landlord.

17. Tenant shall not install an aquarium of any size in the Premises unless otherwise approved by Landlord.

18. Tenant shall not utilize any name selected by Landlord from time to time for the Building and/or the Project as any part of Tenant's corporate or trade name. Landlord shall have the right to change the name, number or designation of the Building or Project without liability to Tenant. Tenant shall not use any picture of the Building in its advertising, stationery or in any other manner.

19. Tenant shall, upon request by Landlord, supply Landlord with the names and telephone numbers of personnel designated by Tenant to be contacted on an after-hours basis should circumstances warrant.

20. Landlord may from time to time grant tenants individual and temporary variances from these Rules, provided that any variance does not have a material adverse effect on the use and enjoyment of the Premises by Tenant.

## Ехнівіт **F**

## PARKING

Tenant shall be entitled to the number of vehicle parking spaces set forth in Item 11 of the Basic Lease Provisions, which spaces shall be unreserved and unassigned, on those portions of the Common Areas designated by Landlord for parking, except that at Tenant's request, up to 5 of said spaces shall be marked as "reserved" for Tenant in locations designated by Landlord at the rear of the Premises. Tenant shall not use more parking spaces than such number. All parking spaces shall be used only for parking of vehicles no larger than full size passenger automobiles, sport utility vehicles or pickup trucks. Tenant shall not permit or allow any vehicles that belong to or are controlled by Tenant or Tenant's employees, suppliers, shippers, customers or invitees to be loaded, unloaded or parked in areas other than those designated by Landlord for such activities. If Tenant permits or allows any of the prohibited activities described above then Landlord shall have the right, without notice, in addition to such other rights and remedies that Landlord may have, to remove or tow away the vehicle involved and charge the costs to Tenant. Parking within the Common Areas shall be limited to striped parking stalls, and no parking shall be permitted in any driveways, access ways or in any area which would prohibit or impede the free flow of traffic within the Common Areas. There shall be no parking of any vehicles for longer than a forty-eight (48) hour period unless otherwise authorized by Landlord, and vehicles which have been abandoned or parked in violation of the terms hereof may be towed away at the owner's expense. Nothing contained in this Lease shall be deemed to create liability upon Landlord for any damage to motor vehicles of visitors or employees, for any loss of property from within those motor vehicles, or for any injury to Tenant, its visitors or employees, unless ultimately determined to be caused by the sole negligence or willful misconduct of Landlord. Landlord shall have the right to establish, and from time to time amend, and to enforce against all users all reasonable rules and regulations (including the designation of areas for employee parking) that Landlord may deem necessary and advisable for the proper and efficient operation and maintenance of parking within the Common Areas. Landlord shall have the right to construct, maintain and operate lighting facilities within the parking areas; to change the area, level, location and arrangement of the parking areas and improvements therein; to restrict parking by tenants, their officers, agents and employees to employee parking areas; to enforce parking charges (by operation of meters or otherwise); and to do and perform such other acts in and to the parking areas and improvements therein as, in the use of good business judgment, Landlord shall determine to be advisable, provided, however, that none of the foregoing shall deprive Tenant of reasonable access to the Premises Any person using the parking area shall observe all directional signs and arrows and any posted speed limits. In no event shall Tenant interfere with the use and enjoyment of the parking area by other tenants of the Project or their employees or invitees. Parking areas shall be used only for parking vehicles. Washing, waxing, cleaning or servicing of vehicles, or the storage of vehicles for longer than 48-hours, is prohibited unless otherwise authorized by Landlord. Tenant shall be liable for any damage to the parking areas caused by Tenant or Tenant's employees, suppliers, shippers, customers or invitees, including without limitation damage from excess oil leakage. Tenant shall have no right to install any fixtures, equipment or personal property in the parking areas.

In addition to the vehicle parking spaces allocated to Tenant in Item 11 of the Basic Lease Provisions, Landlord agrees that, subject to availability, Tenant may license the use of up to 22 unreserved spaces in addition to the number of vehicle parking spaces set forth in Item 11 of the Basic Lease Provisions during the initial Term of this Lease ("*Additional Parking*") at no additional rent payable by Tenant. Landlord shall have the right to designate location(s) within the Common Areas for Additional Parking from time to time by notice to Tenant. The Additional Parking shall be subject to all of the provisions of this Exhibit C. Tenant's Additional Parking rights shall belong solely to Ghost Media Group, LLC, a Nevada corporation, and any attempted assignment or transfer of such rights shall be void and of no force and effect.

#### EXHIBIT G

## ADDITIONAL PROVISIONS

The following additional provisions shall be binding on Landlord and Tenant:

1. **RIGHT TO EXTEND THIS LEASE**. Provided that no Default has occurred under any provision of this Lease, either at the time of exercise of the extension right granted herein or at the time of the commencement of such extension, and provided further that Tenant is occupying the entire Premises and has not assigned or sublet any of its interest in this Lease, then Tenant may extend the Term of this Lease for one (1) extension period of 60 months. Tenant shall exercise its right to extend the Term by and only by delivering to Landlord, not less than 9 months or more than 12 months prior to the Expiration Date of the Term, Tenant's irrevocable written notice of its commitment to extend (the "*Commitment Notice*"). The Basic Rent payable under the Lease during any extension of the Term shall be determined as provided in the following provisions.

If Landlord and Tenant have not by then been able to agree upon the Basic Rent for the extension of the Term, then not less than 90 days or more than 120 days prior to the Expiration Date of the Term, Landlord shall notify Tenant in writing of the Basic Rent that would reflect the prevailing market rental rate for a 60-month renewal of comparable space in the Project (together with any increases thereof during the extension period) as of the commencement of the extension period ("Landlord's Determination"). Should Tenant disagree with the Landlord's Determination, then Tenant shall, not later than 20 days thereafter, notify Landlord in writing of Tenant's determination of those rental terms ("Tenant's Determination"). Within 10 days following delivery of the Tenant's Determination, the parties shall attempt to agree on an appraiser to determine the fair market rental. If the parties are unable to agree in that time, then each party shall designate an appraiser within 10 days thereafter. Should either party fail to so designate an appraiser within that time, then the appraiser designated by the other party shall determine the fair market rental. Should each of the parties timely designate an appraiser, then the two appraisers so designated shall appoint a third appraiser who shall, acting alone, determine the fair market rental for the Premises. Any appraiser designated hereunder shall have an MAI certification with not less than 5 years experience in the valuation of commercial industrial buildings in the vicinity of the Project.

Within 30 days following the selection of the appraiser and such appraiser's receipt of the Landlord's Determination and the Tenant's Determination, the appraiser shall determine whether the rental rate determined by Landlord or by Tenant more accurately reflects the fair market rental rate for the 60-month renewal of the Lease for the Premises, as reasonably extrapolated to the commencement of the extension period. Accordingly, either the Landlord's Determination or the Tenant's Determination shall be selected by the appraiser as the fair market rental rate for the extension period. In making such determination, the appraiser shall consider rental comparables for the Project (provided that if there are an insufficient number of comparables within the Project, the appraiser shall consider rental comparables for similarly improved space owned by Landlord in the Irvine Spectrum area with appropriate adjustment for location and quality of project), but the appraiser shall not attribute any factor for brokerage commissions in making its determination of the fair market rental rate. At any time before the decision of the appraiser is rendered, either party may, by written notice to the other party, accept the rental terms submitted by the other party, in which event such terms shall be deemed adopted as the agreed fair market rental. The fees of the appraiser(s) shall be borne entirely by the party whose determination of the fair market rental rate was not accepted by the appraiser.

Within 20 days after the determination of the fair market rental, Landlord shall prepare an appropriate amendment to this Lease for the extension period, and Tenant shall execute and return same to Landlord within 10 days after Tenant's receipt of same. Should the fair market rental not be established by the commencement of the extension period, then Tenant shall continue paying rent at the rate In effect during the last month of the initial Term, and a lump sum adjustment shall be made promptly upon the determination of such new rental.

If Tenant fails to timely exercise the extension right granted herein within the time period expressly set forth for exercise by Tenant in the initial paragraph of this Section, Tenant's right to extend the Term shall be extinguished and the Lease shall automatically terminate as of the expiration date of the Term, without any extension and without any liability to Landlord. Tenant's rights under this Section shall belong solely to Ghost Media Group, LLC, a Nevada corporation, and any attempted assignment or transfer of such rights (except in connection with a Permitted Transfer) shall be void and of no force and effect. Tenant shall have no other right to extend the Term beyond the single 60 month extension period created by this Section. Unless agreed to in a writing signed by Landlord and Tenant, any extension of the Term, whether created by an amendment to this Lease or by a holdover of the Premises by Tenant, or otherwise, shall be deemed a part of, and not in addition to, any duly exercised extension period permitted by this Section.

2. TENANT'S SECURITY POLICIES, PROCEDURES AND EQUIPMENT. Notwithstanding anything to the contrary in this Lease, Tenant shall have the right, at its sole discretion and expense, to install its own electronic security systems within the Premises, including card key and CCTV system ("*Tenant's Security Systems*"), provided that any portions of Tenant's Security Systems visible from the exterior of the Premises shall be subject to the approval of Landlord as to appearance and conformity with the architectural integrity of the Building, and provided further that Landlord shall approve the operating attributes of the Tenant's Security Systems. Tenant's Security System shall be "Required Removables" as defined in Section 7.3 of this Lease.

## 3. LANDLORD'S WARRANTIES.

(a) Landlord shall correct, repair and/or replace any non-compliance of (i) the lobby restrooms of the Building, and (ii) the Common Areas of the Project exterior to the Building, with all building permits and codes in effect and applicable as of the execution of this Lease, including without limitation, the provisions of Title III of the Americans With Disabilities Act ("*ADA*"), which are "triggered" by the permitting and construction of the "Tenant Improvements" by Landlord pursuant to the attached Work Letter. Said costs of compliance shall be Landlord's sole cost and expense in addition to the "Landlord's Contribution" set forth in the attached Work Letter and shall not be part of Project Costs. Landlord shall correct, repair or replace any non-compliance of the areas described in Subsection (i) and (ii) above with any revisions or amendments to applicable building codes, including the ADA, becoming effective after the execution of this Lease, provided that the amortized cost of such repairs or replacements (amortized over the useful life thereof) shall be included as Project Costs payable by Tenant. All other ADA compliance issues which pertain to the Premises, including without limitation, in connection with Tenant's construction of the Tenant Improvements, any Alterations or other improvements to the Premises and the operation of Tenant's business and employment practices in the Premises, shall be the responsibility of Tenant at its sole cost and expense. The repairs, corrections or replacements required of Landlord or of Tenant under the foregoing provisions of this Section shall be made promptly following notice of non-compliance from any applicable governmental agency.

(b) Landlord warrants to Tenant that the plumbing, fire sprinkler system, lighting, heating, ventilation and air conditioning systems and electrical systems serving the Premises shall be in good operating condition on the Commencement Date of this Lease. Provided that Tenant shall notify Landlord of a non-compliance with the foregoing warranty not later than 30 days following the Commencement Date of the Lease, then Landlord shall, except as otherwise provided in this Lease, promptly after receipt of written notice from Tenant setting forth the nature and extent of such non-compliance, rectify same at Landlord's sole cost and expense (and not as a Project Cost).

(c) Landlord shall correct, repair and/or replace, at its sole cost and expense and not as a Project Cost, the structural components of the roof, the load-bearing walls and the foundations and footings of the Building. Notwithstanding the foregoing, Landlord's obligation contained in this Section to bear such costs and expenses shall not apply: (i) to the costs and expenses of periodic maintenance of the roof, walls, foundations and footings of the Building, (ii) to the cost of replacing the roof membrane and accompanying roof materials as and when such replacement is required, nor (iii) to the extent of the negligence or willful misconduct by Tenant, its employees, agents, contractors, licensees or invitees (in which case Tenant shall be responsible for the reasonable costs of such repairs and/or replacements). The repairs or replacements required of Landlord pursuant to this Section shall be made promptly following notice from Tenant.

4. **EXISTING CABLING.** Landlord shall request (but cannot guarantee) that the tenant in possession of the Premises at the date of execution of this Lease leaves its cabling installed in the Premises ("*Existing Cabling*") for Tenant's use during the Term. If the Existing Cabling is left in place for Tenant's use: (i) Landlord makes no representation or warranty whatsoever as to its condition or suitability or fitness for any purpose whatsoever, and Tenant shall operate and maintain the Existing Cabling during the Term at its sole cost and expense, and (ii) the Existing Cabling shall be "*Required Removables*" at Lease expiration or earlier termination.

#### Ехнівіт Н

### LANDLORD'S DISCLOSURES

### **SPECTRUM**

The capitalized terms used and not otherwise defined in this Exhibit shall have the same definitions as set forth in the Lease. The provisions of this Exhibit shall supersede any inconsistent or conflicting provisions of the Lease.

1. Landlord has been informed that the El Toro Marine Corps Air Station (MCAS) has been listed as a Federal Superfund site as a result of chemical releases occurring over many years of occupancy. Various chemicals including jet fuel, motor oil and solvents have been discharged in several areas throughout the MCAS site. A regional study conducted by the Orange County Water District has estimated that groundwaters beneath more than 2,900 acres have been impacted by Trichloroethlene (TCE), an industrial solvent, There is a potential that this substance may have migrated into the ground water underlying the Premises. The U.S. Environmental Protection Agency, the Santa Ana Region Quality Control Board, and the Orange County Health Care Agency are overseeing the investigation/cleanup of this contamination. To the Landlord's current actual knowledge, the ground water in this area is used for irrigation purposes only, and there is no practical impediment to the use or c) occupancy of the Premises due to the El Toro discharges.

### Ехнівіт Х

## WORK LETTER

#### **BUILD TO SUIT**

The tenant improvement work (the "*Tenant Improvements*" and the "*Tenant Improvement Work*") shall consist of the work, including work in place as of the date hereof, required to complete the improvements to the Premises as shown in the space plan (the "*Plan*") prepared by SAA, dated September 26, 2013, including any "Alternates" shown in the Plan and elected by Tenant, and the cost estimate (the "*Cost Estimate*") prepared by Casco Contractors, dated October 14, 2013. The Tenant Improvement Work shall be performed by a contractor selected by Landlord and In accordance with the requirements and procedures set forth below.

## I. ARCHITECTURAL AND CONSTRUCTION PROCEDURES

A. Landlord shall pay up to the amount of the "Landlord's Contribution" (as defined below) towards the cost of the Tenant Improvement Work. Any additional costs of the Tenant Improvement Work, including, without limitation, additional costs resulting from Changes (as hereinafter defined) requested by Tenant shall be borne solely by Tenant and paid to Landlord as hereinafter provided. To the extent applicable, the build-out of the Tenant Improvements shall include Landlord's building standard tenant improvements, materials and specifications for the Project as set forth in Schedule I attached hereto ("*Building Standard Improvements*"), except for those additions or variations to Building Standard Improvements shall be referred to herein as a "*Non-Standard Improvement*"). Should Landlord submit any additional plans, equipment specification sheets, or other matters to Tenant for approval or completion in connection with the Tenant Improvement Work, Tenant shall respond in writing, as appropriate, within 5 days unless a shorter period is provided herein. Tenant shall not unreasonably withhold its approval of any matter, and any disapproval shall be limited to items not previously approved by Tenant in the Plan or otherwise.

In the event that Tenant requests in writing a revision to the Plan ("*Change*"), and Landlord so approves such B. Change as provided in the Section next below, Landlord shall advise Tenant by written change order as soon as is practical of any increase in the cost to complete the Tenant Improvement Work that causes the total cost of the Tenant Improvement Work to exceed Landlord's Contribution. Tenant shall approve or disapprove such change order, if any, in writing within 2 business days following Tenant's receipt of such change order. If Tenant approves any such change order, Landlord, at its election, may either (i) require as a condition to the effectiveness of such change order that Tenant pay any such increase in the Completion Cost attributable to such change order concurrently with delivery of Tenant's approval of the change order, or (ii) defer Tenant's payment of such increase until the date 10 days after delivery of invoices for same, provided however, that the Tenant Contribution must in any event be paid in full prior to Tenant's commencing occupancy of the Premises. If Tenant disapproves any such change order, Tenant shall nonetheless be responsible for the reasonable architectural and/or planning fees incurred in preparing such change order that causes the total cost of the Tenant Improvement Work to exceed the Landlord's Contribution. Landlord shall have no obligation to interrupt or modify the Tenant Improvement Work pending Tenant's approval of a change order, but if Tenant fails to timely approve a change order, Landlord may (but shall not be required to) suspend the applicable Tenant Improvement Work, in which event any related critical path delays because of such suspension shall constitute Tenant Delays hereunder.

C. Landlord agrees that it shall not unreasonably withhold its consent to Tenant's requested Changes, provided that such consent may be withheld in all events if the requested Change (i) is of a lesser quality than the Tenant Improvements previously approved by Landlord, (ii) fails to conform to applicable governmental requirements, (iii) would result in the Premises requiring building services beyond the level normally provided to other tenants, (iv) would delay construction of the Tenant Improvements and Tenant declines to accept such delay in writing as a Tenant Delay, or (v) interferes in any manner with the proper functioning of, or Landlord's access to, any mechanical, electrical, plumbing or HVAC systems, facilities or equipment in or serving the Building, or (vi) would have an adverse aesthetic impact visible from the exterior of the Premises or would cause additional material expenses to Landlord in releting the Premises.

D. Notwithstanding any provision in the Lease to the contrary, and not by way of limitation of any other rights or remedies of Landlord, if Tenant fails to comply with any of the time periods specified in this Work Letter, fails otherwise to approve or reasonably disapprove any submittal within the time period specified herein for such response (or if no time period is so specified, within 5 business days following Tenant's receipt thereof), requests any Changes, furnishes inaccurate or erroneous specifications or other information, or otherwise delays in any manner the completion of the Tenant Improvements (including without limitation by specifying materials that are not readily available) or the issuance of an occupancy certificate (any of the foregoing being referred to in this Lease as a "*Tenant Delay*"), then Tenant shall bear any resulting additional construction cost or other expenses, and the Commencement Date of this Lease shall be deemed to have occurred for all purposes, including without limitation Tenant's obligation to pay rent, as of the date Landlord reasonably determines that it would have been able to deliver the Premises to Tenant but for the collective Tenant Delays, but not sooner than the Estimated Commencement Date. Should Landlord reasonably determine that the Commencement Date should be advanced in accordance with the foregoing, it shall so notify Tenant in writing. Landlords determination shall be conclusive unless Tenant notifies Landlord in writing, within 5 days thereafter, of Tenant's election to contest same by arbitration pursuant to the provisions of Section III below. Pending the outcome of such arbitration proceedings. Tenant shall make timely payment of all rent due under this Lease based upon the Commencement Date set forth in the aforesaid notice from Landlord.

E. All Tenant Improvements shall become the property of Landlord and shall be surrendered with the Premises at the end of the Term; except that Landlord may, by notice to Tenant given at the time of Landlord's approval of any change requested by Tenant, require Tenant either (i) to remove all or any of the Tenant Improvements approved by way of such Change requested by Tenant, to repair any damage to the Premises or the Common Area arising from such removal, and to replace any Non-Standard Improvements so approved by way of such change with the applicable Building Standard Improvements, or (ii) to reimburse Landlord for the reasonable cost of such removal, repair and replacement upon demand. Any such removals, repairs and replacements by Tenant shall be completed by the Expiration Date, or sooner termination of this Lease. Landlord confirms and agrees, however, that no such removal and/or replacement shall be required for any of the Tenant Improvements shown in the Plan, including any "Alternates" shown in the Plan and elected by Tenant.

Landlord shall permit Tenant and its agents to enter the Premises within 30 days prior to the Commencement Date F. of the Lease in order that Tenant may install fixtures, furniture and cabling through Tenant's own contractors prior to the Commencement Date. Any such work shall be subject to Landlord's prior written approval, and shall be performed in a manner and upon terms and conditions and at times satisfactory to Landlord's representative. The foregoing license to enter the Premises prior to the Commencement Date is, however, conditioned upon Tenant's contractors and their subcontractors and employees working in harmony and not interfering with the work being performed by Landlord. If at any time that entry shall cause disharmony or interfere with the work being performed by Landlord, this license may be withdrawn by Landlord upon 24 hours written notice to Tenant. That license is further conditioned upon the compliance by Tenant's contractors with all requirements imposed by Landlord on third party contractors, including without limitation the maintenance by Tenant and its contractors and subcontractors of workers' compensation and public liability and property damage insurance in amounts and with companies and on forms satisfactory to Landlord, with certificates of such insurance being furnished to Landlord prior to proceeding with any such entry. The entry shall be deemed to be under all of the provisions of the Lease except as to the covenants to pay rent. Landlord shall not be liable in any way for any personal injury, and/or loss or damage of property which may occur in connection with such entry by Tenant or in connection with such work being performed by Tenant, the same being solely at Tenant's risk. In no event shall the failure of Tenant's contractors to complete any work in the Premises extend the Commencement Date of this Lease.

G. Tenant hereby designates Doug Francis (*Tenant's Construction Representative*"), Telephone No. (949) 444-6243, as its representative, agent and attorney-in-fact for all matters related to the Tenant Improvement Work, including but not by way of limitation, for purposes of receiving notices, approving submittals and issuing requests for Changes, and Landlord shall be entitled to rely upon authorizations and directives of such person(s) as if given directly by Tenant. The foregoing authorization is intended to provide assurance to Landlord that it may rely upon the directives and decision making of the Tenant's Construction Representative with respect to the Tenant Improvement Work and is <u>not</u> intended to limit or reduce Landlord's right to reasonably rely upon any decisions or directives given by other officers or representatives of Tenant. Tenant may amend the designation of its Tenant's Construction Representative(s) at any time upon delivery of written notice to Landlord.

H. Landlord shall submit the construction drawings for the Tenant Improvement Work to a competitive bidding process involving at least 3 licensed and reputable general contractors. If requested by Tenant, Landlord shall provide copies of the bid responses to Tenant. After adjustments for any inconsistent assumptions to reflect an "*apples to apples*" comparison, Landlord shall select the lowest qualified bidder for construction of the Tenant Improvements. Upon selection of the lowest qualified bidder, Landlord shall enter into a construction contract with the contractor so selected in the bid amount for construction of the Tenant Improvements.

## II. COST OF THE TENANT IMPROVEMENTS WORK

A. Landlord shalt provide an allowance towards the "Completion Cost" (as defined below) of constructing the Tenant Improvement Work in the amount of \$1,011,825.00 (the "*Landlord's Contribution*"), based on \$25.00 per usable square foot of the Premises, with any excess cost of the Tenant Improvements Work to be borne solely by Tenant. If the actual cost of completion of the Tenant Improvements is less than the maximum amount provided for the Landlord's Contribution, such savings shall inure to the benefit of Landlord and Tenant shall not be entitled to any credit or payment. Notwithstanding the foregoing, Tenant may utilize a portion of the Landlord's Contribution, not to exceed the amount of \$121,419.00, based on 3.00 per usable square foot of the Premises, towards Tenant's out-of-pocket costs of cabling, furniture, architectural fees, signage, security system, and related moving expenses for Tenant's move to the Premises. Landlord shall reimburse Tenant for such costs within 30 days following invoicing by Tenant, provided, however, that Landlord shall have no obligation to fund any portion of the Landlord's Contribution towards invoices received from Tenant after 90 days following the Commencement Date of this Lease.

B. Tenant shall pay all Completion Costs attributable to any costs attributable to Tenant Delays and the amount, if any, by which aggregate Completion Cost of the Tenant Improvement Work exceeds the Landlord's Contribution. The amounts to be paid by Tenant for the Tenant Improvements pursuant to this Section II.C. are sometimes cumulatively referred to herein as the "**Tenant's Contribution**".

C. The "*Completion Cost*" shall mean all costs of Landlord in completing the Tenant Improvements Work, including but not limited to the following: (i) payments made to architects, engineers, contractors, subcontractors and other third party consultants in the performance of the Work, (ii) permit fees and other sums paid to governmental agencies, and (iii) costs of all materials incorporated into the Work or used in connection with the Work. The Completion Cost shall also include an administrative/supervision fee to be paid to Landlord or to Landlord's management agent in the amount of 3% of the Completion Cost. Unless expressly authorized in writing by Landlord, the Completion Cost shall not include (and no portion of the Landlord Contribution shall be paid for) any costs incurred by Tenant, including without limitation, any costs for space planners, managers, advisors or consultants retained by Tenant in connection with the Tenant Improvements.

D. Prior to start of construction of the Tenant Improvements, Tenant shall pay to Landlord in full the amount of the Tenant's Contribution, if any, set forth in the approved Cost Estimate. Following completion of the Tenant Improvements Work, Tenant shall pay (or be refunded) any difference between the estimated and the actual amount of the Tenants Contribution towards the Completion Cost. The balance of all sums due and owing and not otherwise paid by Tenant shall be due and payable on or before the Commencement Date of this Lease. If Tenant defaults in the payment of any sums due under this Work Letter, Landlord shall (in addition to all other remedies) have the same rights as in the case of Tenant's failure to pay rent under the Lease, including, without limitation, the right to terminate this Lease and recover damages from Tenant and/or to charge a late payment fee and to collect interest on delinquent payments, and Landlord may (but shall not be required to) suspend the Tenant Improvement Work following such default, in which event any delays because of such suspension shall constitute Tenant Delays hereunder.

### III. DISPUTE RESOLUTION

A. All claims or disputes between Landlord and Tenant arising out of, or relating to, this Work Letter shall be decided by the JAMS/ENDISPUTE ("JAMS"), or its successor, with such arbitration to be held in Orange County, California, unless the parties mutually agree otherwise. Within 10 business days following submission to JAMS, JAMS shall designate three arbitrators and each party may, within 5 business days thereafter, veto one of the three persons so designated. If two different designated arbitrators have been vetoed, the third arbitrator shall hear and decide the matter. If less than 2 arbitrators are timely vetoed, JAMS shall select a single arbitrator from the non-vetoed arbitrators originally designated by JAMS, who shall hear and decide the matter. Any arbitration pursuant to this section shall be decided within 30 days of submission to JAMS. The decision of the arbitrator shall be final and binding on the parties. All costs associated with the arbitration shall be awarded to the prevailing party as determined by the arbitrator.

B. Notice of the demand for arbitration by either party to the Work Letter shall be filed in writing with the other party to the Work Letter and with JAMS and shall be made within a reasonable time after the dispute has arisen. The award rendered by the arbitrator shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof. Except by written consent of the person or entity sought to be joined, no arbitration arising out of or relating to this Work Letter shall include, by consolidation, joinder or in any other manner, any person or entity not a party to the Work Letter unless (1) such person or entity is substantially involved in a common question of fact or law, (2) the presence of such person or entity is required if complete relief is to be accorded in the arbitration, or (3) the interest or responsibility of such person or entity in the matter is not insubstantial.

C. The agreement herein among the parties to arbitrate shall be specifically enforceable under prevailing law. The agreement to arbitrate hereunder shall apply only to disputes arising out of, or relating to, this Work Letter, and shall not apply to other matters of dispute under the Lease except as may be expressly provided in the Lease,

#### Tenant Improvement / Interior Construction Outline Specifications

TENANT STANDARD **GENERAL OFFICE:** 

#### CARPET

Direct glue, from one of the following options:

Designweave - Z6354 Tempest Esq.:		Designweave – Z6356 Techno:	
a)	553 Steel Wool	a)	336 Lido
b)	773 Melba Toast	b)	252 Topaz
c)	575 Silver Smoke	c)	518 Night Sky
d)	535 Dolphin	d)	997 Silver Plum
e)	454 Denim	e)	496 Galactic

#### VINYL COMPOSMON TILE (VCT)

12x12 VCT Armstrong Standard Excelon, from the following options:

a)	51803 Pearl White	c)	51908 Pewter
b)	51899 Cool White	d)	51899 Cool W

#### PAINT / WALLS

5/8" gypsum drywall on 2-1/2" x 25 ga. metal studs, floor to ceiling construction, no walls shall penetrate the grid unless required by code. All walls shall be straight, and parallel to building perimeter walls. All offices and rooms shall be constructed of a standard size and tangent to a building shell or core wall. Paint finish, one standard color to be Benjamin Moore AC-40, Glacier White, flat finish.

White

.

#### BASE

2-1/2" Burke rubber base color Pearl 137P, straight at cut pile carpet, coved at resilient flooring and loop carpet.

#### RUBBER TRANSITION STRIP

Transition strip between carpet and resilient flooring to be Burke #150, color: to match adjacent V.C.T.

#### PLASTIC LAMINATE

Plastic laminate color at millwork to be Nevamar "Smoky White", Textured #S-7-27T.

#### CEILING

2x4 USG Radar Illusions #2842 grid and scored tile on 9/16" T-bar grid. Continuous grid throughout.

#### PERIMETER WALLS

Furring, 25 ga. metal studs with 5/8" gypsum drywall, with batt insulation.

#### LIGHTING

2X4 fluorescent, 3-lamp energy saving ballast, 18-cell parabolic lens fixture.

#### DOORS

1-3/4" solid core, 3"-0" x 8-10", plain sliced white oak, Western Integrated clear anodized aluminum frames, Schlage "D" series "Sparta" latchset hardware, dull chrome finish.

#### **OFFICE SIDELITES**

All interior offices to have sidelite glazing adjacent to office entry door. 2' wide x door height, Western Integrated clear anodized aluminum frame Integral to door frame with clear tempered glass.

#### Schedule I

Tenant Standard General Office (continued):	<b>WINDOW COVERINGS</b> Vertical blinds: Mariak Industries PVC blinds at building perimeter windows, Model M-3000, Color: Light Grey.
Tenant Standard Mechanical:	<b>HVAC</b> Interior and Exterior zone VAV boxes shall be connected to the main supply air loop. Exterior zone VAV boxes shall be provided with single-row hot water reheat coil.
	Air distribution downstream of VAV boxes shall be provided complete with ductwork, 2'x2' perforated face ceiling diffusers, 2'x2' perforated return air grilles and air balance.
	Pneumatic thermostats with blank white cover shall be provided for each zone Thermostats shall be located adjacent to light switch at 48" above finished floor.
	Exterior corner spaces with more than one exposure shall be provided with a separate zone.
	Conference Room (or Training Room) 20'x13' or larger shall be provided with a separate zone.
	Exterior zone shall be limited to a single exposure and a maximum of 750 to 1000 square feet.
	Interior zone shall be limited to a maximum of 2000 square feet.
	<b>FIRE PROTECTION</b> Pendant satin chrome plated, recessed heads, adjustable canopies, minimum K factor to be 5.62, located at center of scored ceiling tile. Ceiling drops from shell supply loop.
Tenant Standard	
Electrical:	<b>ELECTRICAL SYSTEM</b> 277/480 volt, three phase, four wire metered distribution section added to main service at Main Electrical Room.
	Electrical tenant distribution capacity suitable for 22 watts per s.f. to accommodate HVAC, lighting, data processing, computer loads and convenience outlets.
	Tenant Electrical Room, located within the lease space, to include 270/480 volt and 120/208 volt panels, transformer. lighting control panel, as required.
	<b>LIGHTING</b> Double switch per Title 24, paired in double gang box, Leviton "Decors" white plastic coverplate, 42" AFF to switch centerline. Provide occupancy sensors as required by code. 2x4 fluorescent light fixtures, 3-lamp energy saving ballast, 18-cell parabolic lens fixture based upon one (1) fixture per 80 square feet.
	Exit signs Internally illuminated, white sign face with green text.
	2

TENANT STANDARD Electrical (continued):

#### **OUTLETS**

Power: 15-amp 125-volt specification grade duplex receptacle mounted vertically, 18" AFF to centerline, white plastic coverplate. Feeds to systems furniture by Tenant to be via walls, furred columns or ceiling J-box. Power poles and furniture by Tenant Ratio of one (1) feed per eight (8) workstations. Assumes four (4) circuits, eight (8) wire configuration of systems furniture.

Telephone/Data: Single gang box with mud ring and pull string, mounted vertically, 18" AFF to centerline, Cover plate by telephone and/or cabling company. Teflon cable by tenant.

One (1) empty 2" conduit to be routed from Tenant's Server Room, 4x8 backboard to building main telephone backboard.

TENANT STANDARD WAREHOUSE/SHIPPING AND RECEIVING:

#### FLOORS

Sealed concrete.

#### WALLS

5/8" gypsum wallboard standard partition. Paint to match Benjamin Moore AC-40 Glacier White; rated partition at occupancy separation as required by code.

#### CEILING

Exposed structure, non-painted.

#### WINDOWS

None

## ACCESS

7"-6" H x 7"-6" W glazed service doors. Glazing is bronze reflective glass.

HVAC None

### PLUMBING

Single accommodation restroom, if required.

Sheet vinyl flooring to be Armstrong Classic Carlon "Seagate" #86526 Oyster, with Smooth White FRP panel wainscot to 48" high. Painted walls and ceiling to be Benjamin Moore AC-40 Glacier White, semi-gloss finish.

#### LIGHTING

Chain hung florescent strip fixtures.

#### OTHER ELECTRICAL

Convenience outlets; surface mounted at exposed concrete walls.

### SECURITY

Lockable doors.

## Ехнівіт Ү

# DISCOVERY BUSINESS CENTER

## FIRST AMENDMENT TO LEASE AND CONSENT TO ASSIGNMENT

## I. PARTIES AND DATE.

This First Amendment to Lease (the "Amendment") dated January 27, 2016, is by and between **DISCOVERY BUSINESS CENTER LLC**, a Delaware limited liability company, as successor-in-interest to The Irvine Company LLC, a Delaware limited liability company ("Landlord"), and GHOST MANAGEMENT GROUP, LLC, a Delaware limited liability company ("Tenant"), as successor-in-interest to Ghost Media Group, LLC, a Nevada limited liability company ("Original Tenant").

## II. RECITALS.

On November 11, 2013, Landlord and Original Tenant entered into a lease ("*Lease*") for space in a building located at 41 Discovery, Irvine, California ("*Discovery Premises*" and "*Discovery Building*").

On or prior to the date of this Amendment, Original Tenant has assigned all of its right, title and interest under the Lease to Tenant. In connection with such assignment, Tenant desires to assume all of the obligations of Original Tenant under this Lease, and Landlord desires to confirm that such assignment and assumption constitutes a "*Permitted Transfer*" under the Lease. as more particularly provided in "**IV. ASSIGNMENT AND ASSUMPTION**" below.

Landlord and Tenant each desire to modify the Lease to add approximately 4,182 rentable square feet of space in a building located at 15420 Laguna Canyon Road, Suite 260, Irvine, California, more particularly described on **EXHIBIT A** attached to this Amendment and herein referred to as the "*Laguna Canyon Premises*", to adjust the Basic Rent and to make such other modifications as are set forth in "**Ill. MODIFICATIONS**" next below.

## III. MODIFICATIONS.

A. Premises/Building. Effective as of the "Commencement Date for the Laguna Canyon Premises" (as hereinafter defined), the "Premises" under the Lease shall consist of the Discovery Premises and the Laguna Canyon Premises (and references to Exhibit A to the Lease shall also include Exhibit A attached to this Amendment), and all references to the "Building" in the Lease shall be amended to refer to the two (2) buildings located at 41 Discovery, Irvine. California, and 15420 Laguna Canyon Road, Irvine, California ("Laguna Canyon Building"), either collectively or individually as the context may reasonably require; provided, that, for the avoidance of doubt, Tenant's obligations under the Lease and this Amendment with respect to the Laguna Canyon Building shall only extend to Suite 260 located in the Laguna Canyon Building and Tenant shall not have any obligations with respect to any of the other office space or areas located in or around the Laguna Canyon Building (unless such areas constitute Common Areas as defined under the Lease, then Tenant's obligations with respect to such Common Areas shall be a set forth in the Lease).

**B. Basic Lease Provisions**. The Basic Lease Provisions are hereby amended as follows:

**1.** Effective as of the Commencement Date for the Laguna Canyon Premises, Item 2 shall be deleted in its entirety and substituted therefore shall be the following:

"2. Premises: The Premises are more particularly described in Section 2.1.

Address of Buildings:	41 Discovery, Irvine, CA 92618	
	15420 Laguna Canyon Road, Suite 260,	
	Irvine, CA 92618	

Project Description: Discovery Business Center

**2.** Item 4 is hereby amended by adding the following:

"*Commencement Date for the Laguna Canyon Premises*" shall mean the earlier of (a) the date Tenant commences regular business activities in the Premises, or (b) March 1, 2016."

**3.** Item 5 is hereby amended by adding the following:

"5. Lease Term as to the Laguna Canyon Premises: The Term of this Lease as to the Laguna Canyon Premises shall expire conterminously with the Term of the Lease, that is at midnight on February 28, 2019."

**4.** Item 6 is hereby amended by adding the following:

"Basic Rent for the Laguna Canyon Premises:

Months of Term or Period for the Laguna Canyon Premises	Monthly Rate Per Rentable Square Foot for the Laguna Canyon Premises	Monthly Basic Rent (rounded to the nearest dollar) for the Laguna Canyon Premises
Commencement Date for the Laguna Canyon Premises to January 31. 2017	\$2.06	\$8,615.00
February 1, 2017 to January 31, 2018	\$2.15	\$8,991.00
February 1, 2018 to February 28, 2019	\$2.25	\$9,410.00

5. Effective as of the Commencement Date for the Laguna Canyon Premises, Item 8 shall be deleted in its entirety and substituted therefor shall be the following:

"8. Floor Area of Premises: Approximately 49,002 rentable square feet, comprised of the following:

Floor Area of Laguna Canyon Premises: Approximately 4,182 rentable square feet

Floor Area of Discovery Premises: Approximately 44,820 rentable square feet

Floor Area of the Laguna Canyon Building: Approximately 50,562 rentable square feet

Floor Area of the Discovery Building: Approximately 44,820 rentable square feet"

**6.** Item 9 is hereby deleted in its entirety and substituted therefore shall be the following:

"9. Security Deposit \$185,350.00"

7. Effective as of the Commencement Date for the Laguna Canyon Premises, Item 11 shall be deleted in its entirety and substituted therefor shall be the following:

"11. Parking: 216 parking spaces, together with the Additional Parking" in accordance with the provisions set forth in **Exhibit F** attached to this Lease"

**C. HVAC After Hours**. **Exhibit C** attached to the Lease is hereby amended to provide that for the HVAC unit(s) servicing the Laguna Canyon Premises "*after hours*" shall mean usage of said unit(s) before 8:00 A.M. or after 6:00 P.M. on Mondays through Fridays, before 9:00 A.M. or after 1:00 P.M. on Saturdays, and all day on Sundays and nationally-recognized holidays, subject to reasonable adjustment of said hours by Landlord.

**D. Right to Extend the Lease**. The provisions of Section 1 of Exhibit G attached to the Lease entitled "Right to Extend this Lease" shall remain in full force and effect and exercisable by Tenant in accordance with the terms of the Lease.

**E. Security Deposit**. Concurrently with Tenant's delivery of this Amendment, Tenant shall deliver the sum of \$10,350.00 to Landlord, which sum shall be added to the Security Deposit presently being held by Landlord in accordance with Section 4.3 of the Lease.

**F. Brokers**. Article 18 of the Lease is amended to provide that the parties recognize the following parties as the brokers who negotiated this Amendment, and agree that Landlord shall be responsible for payment of brokerage commissions to such brokers pursuant to its separate agreements with such brokers: Irvine Realty Company ("*Landlord's Broker*") is the agent of Landlord exclusively and Newmark Grubb Knight Frank ("*Tenant's Broker*") is the agent of Tenant exclusively. By the execution of this Amendment, each of Landlord and Tenant hereby acknowledge and confirm (a) receipt of a copy of a Disclosure Regarding Real Estate Agency Relationship conforming to the requirements of California Civil Code 2079.16, and (b) the agency relationships specified herein, which acknowledgement and confirmation is expressly made for the benefit of Tenant's Broker. If there is no Tenant's Broker so identified herein, then such acknowledgement and confirmation is expressly made for the benefit of the benefit of Landlord's Broker. By the execution of this Amendment, Landlord and Tenant are executing the confirmation of the agency relationships set forth herein. The warranty and indemnity provisions of Article 18 of the Lease, as amended hereby, shall be binding and enforceable in connection with the negotiation of this Amendment.

**G.** Acceptance of the Laguna Canyon Premises. Tenant acknowledges that the lease of the Laguna Canyon Premises pursuant to this Amendment shall be subject to the acknowledgement set forth in Section 2.2 of the Lease and shall be on an "*as-is*" basis without further obligation on Landlord's part as to improvements whatsoever.

**Inter-Building Conduit**. Tenant shall have the right to install data and/or telecommunications cabling within the H. existing underground conduit at the Project (the "Conduit") facilitating Tenant's between the Discovery Building and the Laguna Canyon Building for the purpose of facilitating Tenant's data and telecommunications networking, subject to and upon the following conditions: (i) Tenant's right to use and/or, install cabling within the Conduit shall be subject to the availability, from time to time, of sufficient space therein, (ii) the installation by or for Tenant of any cabling within the Conduit shall be at Tenant's sole risk and cost, and Landlord shall not be liable for any injury, loss, damage, cost, expense or liability that may occur or result in connection therewith, (iii) Tenant shall utilize a contractor either designated by, or otherwise reasonably acceptable to, Landlord to install any cabling within the Conduit, which contractor shall comply with Landlord's construction rules for the Project, including without limitation Landlord's standard insurance requirements, (iv) should Landlord determine in good faith at any time that any cabling installed within the Conduit by or for Tenant interferes with or other poses a risk to the normal functioning of Landlord's cabling, systems, or other equipment, Landlord may require Tenant to make arrangements satisfactory to Landlord to remove or mitigate such risk or interference or, if Tenant either fails or is unable to make such satisfactory arrangements, to remove its cabling, (v) any cabling installed by Tenant within the Conduit shall be considered an "Alteration" pursuant to Section 7.3 of the Lease, and (vi) upon the expiration or sooner termination of the Lease, Tenant shall remove any cabling installed within the Conduit and all related equipment and facilities, and shall repair any damage caused thereby, all at Tenant's sole cost and expense.

### IV. ASSIGNMENT AND ASSUMPTION.

Tenant has assumed all of the obligations and liabilities of "Tenant" under the Lease, as amended by this Amendment, as if Tenant were the original signatory of the Lease, and Landlord hereby confirms that the assignment of the Lease, as amended by this Amendment, to Tenant, constitutes a "*Permitted Transfer*" pursuant to Section 9.1(e) of the Lease and that all of Tenant's and Original Tenant's obligations with respect thereto have been satisfied.

## V. GENERAL.

**A. Effect of Amendments**. The Lease shall remain in full force and effect and unmodified except to the extent that it is modified by this Amendment.

**B. Entire Agreement**. This Amendment embodies the entire understanding between Landlord and Tenant with respect to the modifications set forth in "**III. MODIFICATIONS**" above and can be changed only by a writing signed by Landlord and Tenant.

**C. Defined Terms**. All words commencing with initial capital letters in this Amendment and defined in the Lease shall have the same meaning in this Amendment as in the Lease, unless they are otherwise defined in this Amendment.

**D. Corporate and Partnership Authority**. If Tenant is a corporation or partnership, or is comprised of either or both of them, each individual executing this Amendment for the corporation or partnership represents that he or she is duly authorized to execute and deliver this Amendment on behalf of the corporation or partnership and that this Amendment is binding upon the corporation or partnership in accordance with its terms.

**E. Counterparts; Digital Signatures**. If this Amendment is executed in counterparts, each is hereby declared to be an original; all, however, shall constitute but one and the same amendment. In any action or proceeding, any photographic, photostatic, or other copy of this Amendment may be introduced into evidence without foundation. The parties agree to accept a digital image (including but not limited to an image in the form of a PDF, JPEG, GIF file, or other e-signature) of this Amendment, if applicable, reflecting the execution of one or both of the parties, as a true and correct original.

**F. Certified Access Specialist**. As of the dale of this Amendment, there has been no inspection of the Building and Project by a Certified Access Specialist as referenced in Section 1938 of the California Civil Code.

## VI. EXECUTION.

## LANDLORD:

DISCOVERY BUSINESS CENTER LLC, a Delaware limited liability company

By:	/s/ Steven M. Case
Name:	Steven M. Case
Title:	EVP Office Properties
By:	/s/ Holly McManus
Name:	Holly McManus
Title:	VP, Operations Office Properties

5

## TENANT:

GHOST MANAGEMENT GROUP, LLC, a Delaware limited liability company

By:	/s/ Doug Francis	
Name:	Doug Francis	
Title:	President	
By:	/s/ Chris Beals	
Name:	Christopher Beals	
Title:	General Counsel	

## EXHIBIT A

## 15420 Laguna Canyon Road Suite 260

### SECOND AMENDMENT TO LEASE

#### I. PARTIES AND DATE.

This Second Amendment to Lease (this "**Amendment**") dated April 7, 2017, is by and between **DISCOVERY BUSINESS CENTER LLC**, a Delaware limited liability company ("**Landlord**"), and **GHOST MANAGEMENT GROUP**, **LLC**, a Delaware limited liability company ("**Tenant**").

#### II. RECITALS.

On November 11, 2013, Landlord and Tenant entered into a lease for space in a building located at 41 Discovery, Irvine, California ("**41 Discovery Premises**" and "**41 Discovery Building**"), which lease was amended by a First Amendment to Lease dated January 27, 2016 ("First Amendment"), wherein Tenant added approximately 4,182 rentable square feet of space in a building located at 15420 Laguna Canyon Road, Suite 260, Irvine, California ("Laguna Canyon Premises" and "Laguna Canyon Building"). The foregoing lease, as so amended, is hereinafter referred to as the "Lease".

Landlord and Tenant each desire to modify the Lease to add approximately 7,000 rentable square feet of space in a building located at 49 Discovery, Suite 200, Irvine, California, more particularly described on **EXHIBIT A** attached to this Amendment and herein referred to as the "**49 Discovery Premises**", to adjust the Basic Rent and to make such other modifications as are set forth in "**III. MODIFICATIONS**" next below.

### III. MODIFICATIONS.

A. <u>Premises/Building</u>. Effective as of the "**Commencement Date for the 49 Discovery Premises**" (as hereinafter defined), the "**Premises**" under the Lease shall consist of the 41 Discovery Premises, the Laguna Canyon Premises and the 49 Discovery Premises (and references to **Exhibit A** to the Lease shall also include **Exhibit A** attached to the First Amendment and **Exhibit A** attached to this Amendment), and all references to the "**Building**" in the Lease shall be amended to refer to the three (3) buildings located at 41 Discovery, Irvine, California, 15420 Laguna Canyon Road, Irvine, California, and 49 Discovery, Irvine, California ("49 Discovery Building"), either collectively or individually as the context may reasonably require; provided, that, for the avoidance of doubt, Tenant's obligations under the Lease and this Amendment with respect to the 49 Discovery Building shall only extend to Suite 200 located in the 49 Discovery Building and Tenant shall not have any obligations with respect to any of the other office space or areas located in or around the 49 Discovery Building (unless such areas constitute Common Areas as defined under the Lease, then Tenant's obligations with respect to such Common Areas shall be as set forth in the Lease).

B. <u>Basic Lease Provisions</u>. The Basic Lease Provisions are hereby amended as follows:

1. Effective as of the Commencement Date for the 49 Discovery Premises, Item 2 shall be deleted in its entirety and substituted therefore shall be the following:

"2. **Premises**: The Premises are more particularly described in Section 2.1.

Address of Buildings:	41 Discovery, Irvine, CA 92618	
	15420 Laguna Canyon Road, Suite	
	260, Irvine, CA 92618	
	49 Discovery, Suite 200, Irvine, CA	
	92618	

**Project Description:** Discovery Business Center

2. Item 4 is hereby amended by adding the following:

"**Commencement Date for the 49 Discovery Premises**" shall mean the <u>earlier</u> of (a) the date Tenant commences its regular business activities in the 49 Discovery Premises, <u>or</u> (b) June 1, 2017. Prior to Tenant's taking possession of the 49 Discovery Premises, the parties shall memorialize on a form provided by Landlord the actual Commencement Date for the 49 Discovery Premises, provided that Tenant's failure to execute that form shall not affect the validity of Landlord's determination of said date."

3. Item 5 is hereby amended by adding the following:

"5. **Lease Term as to the 49 Discovery Premises**: The Term of this Lease as to the 49 Discovery Premises shall expire coterminously with the Term of the Lease as to the 41 Discovery Premises and the Laguna Canyon Premises, that is at midnight on February 28, 2019."

4. Item 6 is hereby amended by adding the following:

"Basic Rent for the 49 Discovery Premises:

Months of Term or Period for the 49 Discovery Premises	Monthly Rate Per Rentable Square Foot for the 49 Discovery Premises	Monthly Basic Rent (rounded to the nearest dollar) for the 49 Discovery Premises
Commencement Date for the 49 Discovery Premises to April 30, 2018	\$2.25	\$15,750.00
May 1, 2018 to Expiration Date	\$2.35	\$16,450.00

5. Effective as of the Commencement Date for the 49 Discovery Premises, Item 8 shall be deleted in its entirety and substituted therefor shall be the following:

"8. Floor Area of Premises: Approximately 56,002 rentable square feet, comprised of the following:

Floor Area of Laguna Canyon Premises: Approximately 4,182 rentable square feet

Floor Area of 41 Discovery Premises: Approximately 44,820 rentable square feet

Floor Area of 49 Discovery Premises: Approximately 7,000 rentable square feet

Floor Area of the Laguna Canyon Building: Approximately 50,562 rentable square feet

Floor Area of the 41 Discovery Building: Approximately 44,820 rentable square feet

Floor Area of the 49 Discovery Building: Approximately 54,720 rentable square feet"

6. Item 9 is hereby deleted in its entirety and substituted therefore shall be the following:

### "9. Security Deposit: \$98,454.00"

7. Effective as of the Commencement Date for the 49 Discovery Premises, Item 11 shall be deleted in its entirety and substituted therefor shall be the following:

"11. **Parking**: 239 parking spaces, together with the "Additional Parking" in accordance with the provisions set forth in **Exhibit F** attached to this Lease"

C. <u>Signs</u>. Effective as of the Commencement Date for the 49 Discovery Premises, Section 5.2 of the Lease entitled "Signs", shall be amended by adding the following:

"Landlord shall affix and maintain a sign (restricted solely to Tenant's name as set forth herein or such other name as Landlord may consent to in writing) adjacent to the entry door of the 49 Discovery Premises, together with a directory strip listing Tenant's name as set forth herein in the lobby directory of the 49 Discovery Building. Tenant shall not place or allow to be placed any other sign, decoration or advertising matter of any kind that is visible from the exterior of the 49 Discovery Premises." D. <u>Utilities and Services</u>. Effective as of the Commencement Date for the 49 Discovery Premises, **Exhibit C** attached to the Lease, entitled "Utilities and Services", as amended by Section III.C of the First Amendment, shall be further amended by adding the following:

"Tenant shall contract for and shall pay promptly, directly to the appropriate supplier, all charges for telephone, telecommunications service, and all other utilities, materials and services furnished at the request of Tenant for the 49 Discovery Premises during the Term of the Lease as to the 49 Discovery Premises, together with any taxes thereon. Landlord shall contract for and shall make a reasonable determination of Tenant's proportionate share of the cost of water, gas, sewer, refuse pickup, electricity, interior janitorial services, HVAC services (other than "after hours" HVAC services described below), and any other utilities and services that are not separately metered to the 49 Discovery Premises, and Tenant shall pay such amount to Landlord, as an item of additional rent, within 10 days after delivery of Landlord's statement or invoice therefor. Alternatively, Landlord may elect to include such cost in the definition of Project Costs in which event Tenant shall pay Tenant's proportionate share of such costs in the manner set forth in Section 4.2. Tenant shall also pay to Landlord as an item of additional rent, within 10 days after delivery of Landlord's statement or invoice therefor, Landlord's "standard charges" (as hereinafter defined, which shall be in addition to the electricity charge paid to the utility provider) for "after hours" usage by Tenant of each HVAC unit servicing the 49 Discovery Premises. "After hours" shall mean usage of said unit(s) servicing the 49 Discovery Premises before 8:00 A.M. or after 6:00 P.M. on Mondays through Fridays, before 9:00 A.M. or after 1:00 P.M. on Saturdays, and all day on Sundays and nationally-recognized holidays, subject to reasonable adjustment of said hours by Landlord. "After hours" usage shall be determined based upon the operation of the applicable HVAC unit during each of the foregoing periods on a "non-cumulative" basis (that is, without regard to Tenant's usage or nonusage of other unit(s) serving the 49 Discovery Premises, or of the applicable unit during other periods of the Term of the Lease as to the 49 Discovery Premises). As used herein, "standard charges" shall mean \$25.25 for each hour of "after hours" use (inclusive of the applicable electricity charges paid to the utility provider)."

E. <u>Right to Extend the Lease</u>. The provisions of Section 1 of **Exhibit G** attached to the Lease entitled "Right to Extend this Lease", as amended by Section III.D of the First Amendment, shall remain in full force and effect and exercisable by Tenant in accordance with the terms of the Lease, as to the entire Premises.

F. <u>Security Deposit; First Month's Rent</u>. Concurrently with Tenant's delivery of this Amendment, Tenant shall deliver (i) the sum of \$18,095.00 to Landlord, which sum shall be added to the Security Deposit presently being held by Landlord in accordance with Section 4.3 of the Lease, and (ii) an installment in the amount of 1 full month's Basic Rent, at the initial rate specified in Section III.B.4 of this Amendment, and one month's estimated Tenant's Share of Operating Expenses, which shall be applied against the Basic Rent and Operating Expenses first due hereunder as to the 49 Discovery Premises.

G. <u>Exterior Visibility</u>. Other than Tenant's permitted signs, Tenant shall not place or allow to be placed any other sign, decoration or advertising matter of any kind that is visible from the exterior of the 49 Discovery Premises. Any violating sign or decoration may be immediately removed by Landlord at Tenant's expense without notice and without the removal constituting a breach of the Lease by Landlord or entitling Tenant to claim damages. The interior of the 49 Discovery Premises visible from the exterior must be maintained in a visually professional manner and consistent with a first class office building. Tenant shall not place any unsightly items (as determined by Landlord in its reasonable discretion) along the exterior glass line of the 49 Discovery Premises including, but not limited to, boxes and electrical and data cords.

Rules and Regulations. Tenant acknowledges that Tenant's covenant to comply with the Rules and Regulations H. attached to the Lease as Exhibit E, as the same may be amended from time to time by Landlord in accordance with the terms thereof, constitutes material consideration for Landlord's agreement to enter into this Amendment for the lease by Tenant of the 49 Discovery Premises, and that repeated or persistent violations of the Rules and Regulations by Tenant or its employees, agents, or contractors, whether or not such violations are subsequently cured, will deprive Landlord of the benefits to which it is entitled under this Amendment and the Lease. Accordingly, should Tenant or any of its employees, agents, or contractors violate any of the Rules and Regulations promulgated by Landlord (including, without limitation, any Rules and Regulations pertaining to the smoking or vaping of marijuana or marijuana derivatives in or about the Premises, Building or Project) on 3 or more occasions in any 12-month period (with each such occasion described in a written notice delivered by Landlord to Tenant), whether or not such violations are subsequently cured by Tenant, Landlord shall have the right (i) to terminate Tenant's lease of the 49 Discovery Premises upon 60 days' prior written notice to Tenant, and/or (ii) to declare Tenant to be in Default under the Lease as it pertains to the provisions of Section 1 of Exhibit G to the Lease only, but not with respect to Section 14.2 of the Lease unless such violation is not timely and fully cured as provided in Section 14.1(d) of the Lease. Any such termination pursuant to the immediately preceding sentence shall not be in derogation of, or otherwise limit, any of Landlord's remedies under the Lease or at law or in equity.

I. Fitness Center. Landlord currently provides a fitness center in a building located at 47 Discovery, Irvine, California, and provided: (a) Tenant is not in Default under any provision of the Lease, and (b) Tenant's employees execute Landlord's standard waiver of liability form, then Tenant's employees (the "Fitness Center Users") shall be entitled to use said fitness center (the "Fitness Center"). The use of the Fitness Center shall be free of charge (with the exception of towel/laundry fees) during the initial Term of the Lease, and shall be subject to the reasonable rules and regulations (including rules regarding hours of use) established from time to time by Landlord. Landlord and Tenant acknowledge that the use of the Fitness Center by the Fitness Center Users shall be at their own risk and that the terms and provisions of Section 10.3 of the Lease and shall apply to Tenant and the Fitness Center User's use of the Fitness Center. The costs of operating, maintaining and repairing the Fitness Center may be included as part of Operating Expenses. Tenant acknowledges that the provisions of this Section shall not be deemed to be a representation by Landlord that Landlord shall continuously maintain the Fitness Center (or any other fitness facility) throughout the Term of the Lease, and Landlord shall have the right, at Landlord's sole discretion, to expand, contract, eliminate or otherwise modify the Fitness Center. No expansion, contraction, elimination or modification of the Fitness Center, and no termination of Tenant's or the Fitness Center Users' rights to the Fitness Center shall entitle Tenant to an abatement or reduction in Basic Rent constitute a constructive eviction, or result in an event of default by Landlord under the Lease. Tenant hereby voluntarily releases, discharges, waives and relinquishes any and all actions or causes of action for personal injury or property damage occurring to Tenant or its employees or agents arising as a result of the use of the Fitness Center, or any activities incidental thereto, wherever or however the same may occur, and further agrees that Tenant will not prosecute any claim for personal injury or property damage against Landlord or any of its officers, agents, servants or employees for any said causes of action, except, in each case, arising out of or relating to any gross negligence or willful misconduct of Landlord or any of its employees, contractors or agents. It is the intention of Tenant with respect to the Fitness Center to exempt and relieve Landlord from liability for personal injury or property damage caused by negligence (but not gross negligence or willful misconduct). Tenant's rights under this Section shall belong solely to Ghost Management Group, LLC, and any attempted assignment or transfer of such rights shall be void and of no force and effect.

Conference Room. Landlord currently provides a conference room (the "Conference Room") in the 49 Discovery J. Building, which is capable of accommodating groups of people for use by Building tenants (including Tenant) on a reserved basis. Tenant shall, subject to availability, have the use of the Conference Room subject to Landlord's procedures and charges. The use of the Common Area Conference Room shall be subject to the reasonable rules and regulations (including rules regarding hours of use and priorities for the tenants of the particular building in which a Common Area Conference Room is located, set up and clean up charges, etc.) established from time to time by Landlord for the Common Area Conference Room. Landlord and Tenant acknowledge that the terms and provisions of Section 10.3 of the Lease shall apply to Tenant's use of the Conference Room. Further, Landlord shall have no liability whatsoever with respect to the existence, condition (except, in each case, arising out of or relating to any gross negligence or willful misconduct of Landlord or any of its employees, contractors or agents) or availability of any Common Area Conference Room(s) nor shall Landlord have any obligation or availability of any Common Area Conference Room(s) nor shall Landlord have any obligation whatsoever to enforce or make reservations thereof, and Tenant hereby expressly waives all claims against Landlord with respect to the same. No expansion, contraction, elimination, unavailability or modification of the Common Area Conference Room(s), and no termination of or interference with Tenant's rights to the Common Area Conference Room(s), shall entitle Tenant to an abatement or reduction in rent or constitute a constructive eviction or an event of default by Landlord under the Lease. Tenant's rights under this Section shall belong solely to Ghost Management Group, LLC, and any attempted assignment or transfer of such rights shall be void and of no force and effect.

K. <u>Game Room</u>. Landlord currently provides a game room (the "**Game Room**") in a building located at 15440 Laguna Canyon Road, Irvine, California, for use by Building tenants (including Tenant). Tenant shall, subject to availability, have the use of the Game Room subject to Landlord's procedures. The use of the Game Room shall be subject to the reasonable rules and regulations (including rules regarding hours of use and priorities for tenants who may reserve the Game Room. Landlord and Tenant acknowledge that the terms and provisions of Section 10.3 of the Lease shall apply to Tenant's use of the Game Room. Further, Landlord shall have no liability whatsoever with respect to the existence, condition (except, in each case, arising out of or relating to any gross negligence or willful misconduct of Landlord or any of its employees, contractors or agents) or availability of the Game Room, and Tenant hereby expressly waives all claims against Landlord with respect to the same. No expansion, contraction, elimination, unavailability or modification of the Game Room, and no termination of or interference with Tenant's rights to the Game Room, shall entitle Tenant to an abatement or reduction in rent or constitute a constructive eviction or an event of default by Landlord under the Lease. Tenant's rights under this Section shall belong solely to Ghost Management Group, LLC, and any attempted assignment or transfer of such rights shall be void and of no force and effect.

L. Inter-Building Conduit. Tenant shall have the right to install data and/or telecommunications cabling within the existing underground conduit at the Project (the "Conduit") between and among the 41 Discovery Building, the Laguna Canyon Building and the 49 Discovery Building for the purpose of facilitating Tenant's data and telecommunications networking, subject to and upon the following conditions: (i) Tenant's right to use and/or install cabling within the Conduit shall be subject to the availability, from time to time, of sufficient space therein, (ii) the installation by or for Tenant of any cabling within the Conduit shall be at Tenant's sole risk and cost, and Landlord shall not be liable for any injury, loss, damage, cost, expense or liability that may occur or result in connection therewith, (iii) Tenant shall utilize a contractor either designated by, or otherwise reasonably acceptable to, Landlord to install any cabling within the Conduit, which contractor shall comply with Landlord's construction rules for the Project, including without limitation Landlord's standard insurance requirements, (iv) should Landlord determine in good faith at any time that any cabling installed within the Conduit by or for Tenant interferes with or other poses a risk to the normal functioning of Landlord's cabling, systems, or other equipment, Landlord may require Tenant to make arrangements satisfactory to Landlord to remove or mitigate such risk or interference or, if Tenant either fails or is unable to make such satisfactory arrangements, to remove its cabling, (v) any cabling installed by Tenant within the Conduit shall be considered an "Alteration" pursuant to Section 7.3 of the Lease, and (vi) upon the expiration or sooner termination of the Lease, Tenant shall remove any cabling installed within the Conduit and all related equipment and facilities, and shall repair any damage caused thereby, all at Tenant's sole cost and expense.

M. <u>Early Access</u>. Landlord agrees that Tenant shall be permitted to enter the 49 Discovery Premises from and after the mutual execution of this Amendment and continuing until the Commencement Date for the 49 Discovery Premises in order that Tenant may install fixtures, furniture and cabling through Tenant's own contractors subject to the following terms and conditions: (a) concurrently with the execution and delivery of this Amendment, and prior to any such early access by Tenant, Tenant shall deliver to Landlord the required certificate(s) of insurance and payments required hereunder; and (b) Tenant's early access of the 49 Discovery Premises prior to the Commencement Date for the 49 Discovery Premises pursuant to this Section shall be subject to all of the covenants and conditions on Tenant's part contained in the Lease (including, without limitation, the covenants contained in Sections 5.3, 6.1, 7.1, 7.3, 7.4, 10.1 and 10.3 of the Lease).

N. <u>Acceptance of the 49 Discovery Premises</u>. Tenant acknowledges that the lease of the 49 Discovery Premises pursuant to this Amendment shall be subject to the acknowledgements set forth in Section 2.2 of the Lease, and shall be on an "as-is" basis without further obligation on Landlord's part as to improvements whatsoever.

O. <u>Brokers</u>. Article 18 of the Lease is amended to provide that the parties recognize the following parties as the brokers who negotiated this Amendment, and agree that Landlord shall be responsible for payment of brokerage commissions to such brokers pursuant to its separate agreements with such brokers: Irvine Realty Company ("Landlord's Broker") is the agent of Landlord exclusively and Newmark Grubb Knight Frank ("Tenant's Broker") is the agent of Tenant exclusively. By the execution of this Amendment, each of Landlord and Tenant hereby acknowledge and confirm (a) receipt of a copy of a Disclosure Regarding Real Estate Agency Relationship conforming to the requirements of California Civil Code 2079.16, and (b) the agency relationships specified herein, which acknowledgement and confirmation is expressly made for the benefit of Tenant's Broker. If there is no Tenant's Broker so identified herein, then such acknowledgement and confirmation is expressly made for the benefit of the benefit of Landlord's Broker. By the execution of this Amendment, Landlord and Tenant are executing the confirmation of the agency relationships set forth herein. The warranty and indemnity provisions of Article 18 of the Lease, as amended hereby, shall be binding and enforceable in connection with the negotiation of this Amendment.

## IV. GENERAL.

A. <u>Effect of Amendments</u>. The Lease shall remain in full force and effect and unmodified except to the extent that it is modified by this Amendment.

B. <u>Entire Agreement</u>. This Amendment embodies the entire understanding between Landlord and Tenant with respect to the modifications set forth in "III. MODIFICATIONS" above and can be changed only by a writing signed by Landlord and Tenant.

C. <u>Defined Terms</u>. All words commencing with initial capital letters in this Amendment and defined in the Lease shall have the same meaning in this Amendment as in the Lease, unless they are otherwise defined in this Amendment.

D. <u>Corporate and Partnership Authority</u>. If Tenant is a corporation or partnership, or is comprised of either or both of them, each individual executing this Amendment for the corporation or partnership represents that he or she is duly authorized to execute and deliver this Amendment on behalf of the corporation or partnership and that this Amendment is binding upon the corporation or partnership in accordance with its terms.

E. <u>Counterparts; Digital Signatures</u>. If this Amendment is executed in counterparts, each is hereby declared to be an original; all, however, shall constitute but one and the same amendment. In any action or proceeding, any photographic, photostatic, or other copy of this Amendment may be introduced into evidence without foundation. The parties agree to accept a digital image (including but not limited to an image in the form of a PDF, JPEG, GIF file, or other e-signature) of this Amendment, if applicable, reflecting the execution of one or both of the parties, as a true and correct original.

Certified Access Specialist. Pursuant to California Civil Code § 1938, Landlord hereby states that the Premises F. have not undergone inspection by a Certified Access Specialist (CASp) (defined in California Civil Code § 55.52(a)(3)). Pursuant to Section 1938 of the California Civil Code, Landlord hereby provides the following notification to Tenant: "A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction related accessibility standards within the premises." If Tenant requests to perform a CASp inspection of the Premises, Tenant shall, at its cost, retain a CASp approved by Landlord (provided that Landlord may designate the CASp, at Landlord's option) to perform the inspection of the Premises at a time agreed upon by the parties. Tenant shall provide Landlord with a copy of any report or certificate issued by the CASp (the "CASp Report") and Tenant shall, at its cost, promptly complete any modifications necessary to correct violations of construction related accessibility standards identified in the CASp Report, notwithstanding anything to the contrary in this Lease. Tenant agrees to keep the information in the CASp Report confidential except as necessary for the Tenant to complete such modifications or as required by law, subpoena or other legal process.

## V. EXECUTION.

#### LANDLORD:

**DISCOVERY BUSINESS CENTER LLC,** a Delaware limited liability company

By: /s/ Steven M. Case Printed Name: Steven M. Case Title: EVP Office Properties

By: <u>/s/ Holly McManus</u> Printed Name: Holly McManus Title: Vice President, Operations Office Properties **TENANT: GHOST MANAGEMENT GROUP, LLC,** a Delaware limited liability company

By: /s/ Doug Francis

Printed Name: Doug Francis Title: CEO

# EXHIBIT A

# THIRD AMENDMENT TO LEASE

#### I. PARTIES AND DATE.

This Third Amendment to Lease ("*Amendment*") dated December 29, 2017, is by and between **DISCOVERY BUSINESS CENTER LLC**, a Delaware limited liability company ("*Landlord*"), and **GHOST MANAGEMENT GROUP**, **LLC**, a Delaware limited liability company ("*Tenant*").

#### II. RECITALS.

On November 11, 2013, Landlord and Tenant entered into a lease for space in a building located at 41 Discovery, Irvine, California ("**41 Discovery Premises**" and "**41 Discovery Building**"), which lease was amended by a First Amendment to Lease dated January 27, 2016 ("*First Amendment*"), wherein Tenant added approximately 4,182 rentable square feet of space in a building located at 15420 Laguna Canyon Road, Suite 260, Irvine, California ("*Laguna Canyon Premises*" and "*Laguna Canyon Building*") to the Premises, and by a Second Amendment to Lease dated April 7, 2017, wherein Tenant added approximately 7,000 rentable square feet of space in a building located at 49 Discovery, Suite 200, Irvine, California ("49 Discovery Premises") to the Premises. The foregoing lease, as so amended, is hereinafter referred to as the "*Lease*".

Landlord and Tenant each desire to modify the Lease to terminate Tenant's leasing of the Laguna Canyon Premises and the 49 Discovery Premises (collectively, the "*Excess Space*") in exchange for leasing approximately 31,407 rentable square feet of space in a building located at 43 Discovery, Suite 200, Irvine, California, more particularly described on **EXHIBIT A** attached to this Amendment and herein referred to as the "**43 Discovery Premises**", to extend the Lease Term as to the remainder of the Premises (i.e., the 41 Discovery Premises), to adjust the Basic Rent and to make such other modifications as are set forth in "*III. MODIFICATIONS*" next below.

# III. MODIFICATIONS.

A. **Termination as to the Excess Space**. Notwithstanding anything to the contrary set forth in the Lease, the parties agree that Tenant's lease as to the Excess Space shall terminate on the day immediately preceding the Commencement Date for the 43 Discovery Premises (the "*Termination Date for the Excess Space*"), provided that such termination shall not relieve Tenant of (i) any rent or other charges owed by Tenant, or other obligations required of Tenant, as are set forth in the Lease with respect to the Excess Space from and after the date of this Amendment through and including the Termination Date for the Excess Space, (ii) any obligations which are set forth in this Amendment, and (iii) any indemnity or hold harmless obligations set forth in the Lease as to the Excess Space. Tenant shall quit and surrender possession of the Excess Space to Landlord on or before the date that is 7 days after the Termination Date for the Excess Space as required by the provisions of Section 15.2 of the Lease.

B. **Premises/Building**. Effective as of the "*Commencement Date for the 43 Discovery Premises*" (as hereinafter defined), the "*Premises*" under the Lease shall consist of the 41 Discovery Premises and the 43 Discovery Premises (and references to **Exhibit A** to the Lease shall also include **Exhibit A** attached to this Amendment), and all references to the "*Building*" in the Lease shall be amended to refer to the two (2) buildings located at 41 Discovery, Irvine, California and 43 Discovery, Irvine, California ("**43 Discovery Building**"), either collectively or individually as the context may reasonably require; provided, that, for the avoidance of doubt, Tenant's obligations under the Lease and this Amendment with respect to the 43 Discovery Building shall only extend to Suite 200 located in the 43 Discovery Building and Tenant shall not have any obligations with respect to any of the other office space or areas located in or around the 43 Discovery Building (unless such areas constitute Common Areas as defined under the Lease, then Tenant's obligations with respect to such Common Areas shall be as set forth in the Lease).

C. **Basic Lease Provisions**. The Basic Lease Provisions are hereby amended as follows:

1. Effective as of the Commencement Date for the 43 Discovery Premises, Item 2 shall be deleted in its entirety and substituted therefore shall be the following:

"2. **Premises**: The Premises are more particularly described in Section 2.1.

Address of Buildings:	41 Discovery, Irvine, CA 92618
	43 Discovery, Suite 200, Irvine, CA
	92618
Project Description:	Discovery Business Center

# 2. Item 4 is hereby amended by adding the following:

# "Estimated Commencement Date for the 43 Discovery Premises: November 1, 2018"

"Commencement Date for the 43 Discovery Premises" shall mean the earlier of (a) Landlord's tender of possession of the 43 Discovery Premises to Tenant with the "Tenant Improvements" (as hereinafter defined and as defined) for the 43 Discovery Premises substantially completed but for minor punch list items, or (b) the date Tenant commences its business operations in the 43 Discovery Premises. Prior to Tenant's taking possession of the 43 Discovery Premises, the parties shall memorialize on a form provided by Landlord the actual Commencement Date for the 43 Discovery Premises, provided that Tenant's failure to execute that form shall not affect the validity of Landlord's determination of said date. Tenant acknowledges that neither Landlord nor any representative of Landlord has made any representation or warranty with respect to the 43 Discovery Premises, the 43 Discovery Building or the Project or the suitability or fitness of either for any purpose, except as expressly set forth in the Lease. As of the Commencement Date for the 43 Discovery Premises, Tenant shall be conclusively deemed to have accepted that the 43 Discovery Premises is in satisfactory condition and in conformity with the provisions of the Lease, subject only to those defective or incomplete portions of the Tenant Improvements for the 43 Discovery Premises which Tenant shall have itemized on a written punch list and delivered to Landlord within thirty (30) days after the Commencement Date for the 43 Discovery Premises. Landlord shall promptly complete all punch list matters of which it is so notified."

3. Item 5 is hereby amended by adding the following:

"5. **Lease Term as to the Entire Premises**: The Term of this Lease as to the entire Premises shall expire on February 28, 2025."

4. Item 6 is hereby amended by adding the following:

# "Basic Rent for the 43 Discovery Premises:

Months of Term or Period	Monthly Rate Per Rentable Square Foot	Monthly Basic Rent (rounded to the nearest dollar)
Commencement Date for		
the 43 Discovery		
Premises to February 28,		
2019	\$2.13	\$66,897.00

Notwithstanding the above schedule of Basic Rent to the contrary (or the schedule of Basic Rent set forth below, if the Commencement Date for the 43 Discovery Premises occurs after November 1, 2018), as long as Tenant is not in Default (as defined in Section 14.1) under this Lease, Tenant shall be entitled to an abatement of 4 full calendar months of Basic Rent for the 43 Discovery Premises in the aggregate amount of \$267,588.00 (i.e. \$66,897.00 per month) (the "*Abated Basic Rent*") for the initial four full calendar months of the Term as to the 43 Discovery Premises (the "*Abatement Period*"). In the event Tenant Defaults at any time during the Term beyond any applicable "**cure**" period with the result that Tenant's right to possession of the Premises is terminated, then unamortized Abated Basic Rent to the date of such termination (amortized over the initial 76 months of the Term as to the 43 Discovery Premises) shall immediately become due and payable. The payment by Tenant of the unamortized Abated Basic Rent in the event of a Default shall not limit or affect any of Landlord's other rights, pursuant to this Lease or at law or in equity. Only Basic Rent shall be abated during the Abatement Period and all other additional rent and other costs and charges specified in this Lease shall remain as due and payable pursuant to the provisions of this Lease.

# "Basic Rent for the entire Premises:

Months of Term or Period	Monthly Rate Per Rentable Square Foot	Monthly Basic Rent (rounded to the nearest dollar)
March 1, 2019 toFebruary 29, 2020	\$2.13	\$162,364.00
March 1, 2020 toFebruary 29, 2021	\$2.23	\$169,987.00
March 1, 2021 toFebruary 29, 2022	\$2.33	\$177,609.00
March 1, 2022 toFebruary 29, 2023	\$2.43	\$185,232.00
March 1, 2023 toFebruary 29, 2024	\$2.54	\$193,617.00
March 1, 2024 toExpiration Date	\$2.65	\$202,002.00

For the avoidance of doubt, the Basic Rent for the 41 Discovery Premises for the period commencing on the Commencement Date for the 43 Discovery Premises and ending on February 28, 2019 shall be the amount set forth in the Lease with respect to the 41 Discovery Premises.

5. Effective as of the Commencement Date for the 43 Discovery Premises, Item 8 shall be deleted in its entirety and substituted therefor shall be the following:

"8. **Floor Area of Premises**: Approximately 76,227 rentable square feet, comprised of the following:

Floor Area of 41 Discovery Premises: Approximately 44,820 rentable square feet

Floor Area of 43 Discovery Premises: Approximately 31,407 rentable square feet

Floor Area of the 41 Discovery Building: Approximately 44,820 rentable square feet

Floor Area of the 43 Discovery Building: Approximately 62,814 rentable square feet"

6. Item 9 is hereby deleted in its entirety and substituted therefore shall be the following:

"9. Security Deposit: \$222,202.00

7. Effective as of the Commencement Date for the 43 Discovery Premises, Item 11 shall be deleted in its entirety and substituted therefor shall be the following:

"11. **Parking**: 319 parking spaces, together with the "Additional Parking" in accordance with the provisions set forth in **Exhibit** F attached to this Lease"

D. **Security Deposit**. Concurrently with Tenant's delivery of this Amendment, Tenant shall deliver the sum of \$123,748.00 to Landlord, which sum shall be added to the Security Deposit presently being held by Landlord in accordance with Section 4.3 of the Lease.

# E. Signs.

(i) Effective as of the Commencement Date for the 43 Discovery Premises, the first sentence of Section 5.2 of the Lease entitled "*Signs*", shall be deleted in its entirety and substituted therefor shall be the following:

"Provided Tenant continues to occupy the entire Premises, Tenant shall have (i) the exclusive right to 2 exterior "building top" signs on the 41 Discovery Building and a location on the building monument signage in front of the 41 Discovery Building, and (ii) the non-exclusive right to 1 exterior "building top" sign on the 43 Discovery Building, for Tenant's name and graphics in locations designated by Landlord, subject to Landlord's right of prior approval that such exterior signage is in compliance with the Signage Criteria (defined below). In the event Tenant leases the entire 43 Discovery Building, for Tenant's name and graphics in locations designated by Landlord, subject to Landlord's right of prior approval that such exterior signage is in compliance with the Signage Criteria (below). In the event Tenant leases the entire 43 Discovery Building, for Tenant's name and graphics in locations designated by Landlord, subject to Landlord's right of prior approval that such exterior signage is in compliance with the Signage Criteria (defined below). "

(ii) Effective as of the Commencement Date for the 43 Discovery Premises, Section 5.2 of the Lease entitled "*Signs*", shall be amended by adding the following:

"Landlord shall affix and maintain a sign (restricted solely to Tenant's name as set forth herein or such other name as Landlord may consent to in writing) adjacent to the entry door of the 43 Discovery Premises, together with a directory strip listing Tenant's name as set forth herein in the lobby directory of the 43 Discovery Building. Tenant shall not place or allow to be placed any other sign, decoration or advertising matter of any kind that is visible from the exterior of the 43 Discovery Premises."

F. **Utilities and Services**. Effective as of the Commencement Date for the 43 Discovery Premises, both Section III.C of the First Amendment and Section III.D of the Second Amendment, each entitled "*Utilities and Services*", shall be deleted in their entirety and substituted therefor shall be the following:

"Tenant shall contract for and shall pay promptly, directly to the appropriate supplier, all charges for telephone, telecommunications service, and all other utilities, materials and services furnished at the request of Tenant for the 43 Discovery Premises during the Term of the Lease as to the 43 Discovery Premises, together with any taxes thereon. Landlord shall contract for and shall make a reasonable determination of Tenant's proportionate share of the cost of water, gas, sewer, refuse pickup, electricity, interior janitorial services, HVAC services (other than "after hours" HVAC services described below), and any other utilities and services that are not separately metered to the 43 Discovery Premises, and Tenant shall pay such amount to Landlord, as an item of additional rent, within 10 days after delivery of Landlord's statement or invoice therefor. Alternatively, Landlord may elect to include such cost in the definition of Project Costs in which event Tenant shall pay Tenant's proportionate share of such costs in the manner set forth in Section 4.2. Tenant shall also pay to Landlord as an item of additional rent, within 10 days after delivery of Landlord's statement or invoice therefor, Landlord's "standard charges" (as hereinafter defined, which shall be in addition to the electricity charge paid to the utility provider) for "after hours" usage by Tenant of each HVAC unit servicing the 43 Discovery Premises. After hours" shall mean more than 66 hours of usage during each week. "After hours" usage shall be determined based upon the operation of the applicable HVAC unit during each calendar year on a "non-cumulative" basis (that is, without regard to Tenant's usage or nonusage of other unit(s) serving the Premises, or of the applicable unit during other years of the Term). As used herein, "standard charges" shall mean \$10.00 per hour for each hour of "after hours" use (in addition to the applicable electricity charges paid to the utility provider)."

G. **Right to Extend the Lease**. The provisions of Section 1 of Exhibit G attached to the Lease entitled "*Right to Extend this Lease*", as amended by Section III.D of the First Amendment and by Second III.F of the Second Amendment, shall remain in full force and effect and exercisable by Tenant during the Term of the Lease as extended by this Amendment and shall apply to Tenant's leasing of the 43 Discovery Premises pursuant to this Amendment, except that the reference in the second sentence of the first paragraph thereof to "9 months" is hereby amended to "12 months".

H. Right of First Refusal. Provided Tenant is not then in Default hereunder beyond any applicable cure period, and provided further that Tenant is occupying the entire Premises and has not assigned or sublet any of its interest in the Lease (except in connection with a Permitted Transfer of this Lease to an Affiliate as described in Section 9.1(e) of the Lease, Tenant shall have a continuing right ("*First Refusal Right*") to lease, during the initial approximately 77 month Term of the Lease as to the 43 Discovery Premises, approximately 31,407 rentable square feet of office space on the first floor of the Building as shown on the attached Exhibit A-1 ("First Right Space") in accordance with and subject to the provisions of this Section. Following the receipt by Landlord of a bona fide letter of intent, request for proposal or other written expression of interest to lease the First Right Space, then provided Landlord intends to pursue such leasing opportunity, Landlord shall give Tenant written notice ("First *Right Notice*") of the basic economic terms, including but not limited to the Basic Rent, term, operating expense base, security deposit, and tenant improvement allowance (collectively, the "Economic Terms"), upon which Landlord intends to lease such First Right Space to the applicable third party; provided that the Economic Terms shall exclude brokerage commissions and other Landlord payments that do not directly inure to the tenant's benefit. It is understood that should Landlord intend to lease other office space in addition to the First Right Space as part of a single transaction, then the First Right Notice shall so provide and all such space shall collectively be subject to the following provisions. Within 3 business days after receipt of the First Right Notice, Tenant may, by written notice to Landlord, elect to lease all, but not less than all, of the space specified in the First Right Notice (the "Designated First Right Space") upon such Economic Terms set forth in the First Right Notice and the same non-Economic Terms as set forth in this Lease. In the event that Tenant does not timely commit in writing to lease the Designated First Right Space on the foregoing terms, then Landlord shall be free to lease same thereafter without any constraint, and Tenant shall have no further rights to any such Designated First Right Space. Should Tenant timely elect to lease the Designated First Right Space, then Landlord shall promptly prepare and deliver to Tenant an amendment to this Lease consistent with the foregoing, and Tenant shall execute and return same to Landlord within 10 days. Tenant's failure to timely return the amendment shall entitle Landlord to specifically enforce Tenant's commitment to lease the Designated First Right Space, to lease such space to a third party without any obligation pursuant to this Section, and/or to pursue any other available legal remedy. In the event that Landlord leases the First Right Space, or any portion thereof, to a third party in accordance with the provisions of this Section, and during the effective period of this First Right the First Right Space, or any portion thereof, shall again become available for releasing, then prior to Landlord entering into any such new lease with a third party other than the then-current occupant thereof for the First Right Space, Landlord shall repeat the procedures specified above in this Section. Notwithstanding the foregoing, it is understood that Tenant's First Refusal Right shall be subject to any extension rights previously granted by Landlord to any third party tenant in the Building, as well as to any such rights which may hereafter be granted by Landlord to any third party tenant occupying the First Right Space or any portion thereof, and Landlord shall in no event be obligated to initiate this First Refusal Right prior to leasing any portion of the First Right Space to the then-current occupant thereof. Tenant's rights under this Section shall be personal to the original Tenant named in this Lease and may not be assigned or transferred (except in connection with a Permitted Transfer as described in Section 9.1(e) of the Lease). Any other attempted assignment or transfer shall be void and of no force or effect. Time is specifically made of the essence of this Section.

I. **Exterior Visibility**. The provisions of Section III.G of the Second Amendment shall remain in full force and effect during the Term of the Lease as extended by this Amendment and shall apply to Tenant's leasing of the 43 Discovery Premises pursuant to this Amendment, except that the references in said Section III.G to the 49 Discovery Premises shall be amended to instead mean and refer to the 43 Discovery Premises.

J. **Rules and Regulations**. The provisions of Section III.H of the Second Amendment shall remain in full force and effect during the Term of the Lease as extended by this Amendment and shall apply to Tenant's leasing of the 43 Discovery Premises pursuant to this Amendment, except that the references in said Section III.G to the 49 Discovery Premises shall be amended to instead mean and refer to the 43 Discovery Premises.

K. **Inter-Building Conduit**. The provisions of Section III.L of the Second Amendment shall remain in full force and effect during the Term of the Lease as extended by this Amendment and shall apply to Tenant's leasing of the 43 Discovery Premises pursuant to this Amendment, except that the references in said Section III.L to the Laguna Canyon Building and the 49 Discovery Building shall be amended to instead mean and refer to the 43 Discovery Building. Not later than the Termination Date for the Excess Space, Tenant shall remove any existing cabling within the Conduit servicing the Laguna Canyon Building and the 49 Discovery Building and all related equipment and facilities at Tenant's sole cost and expense.

L. **Acceptance of Premises**. Tenant acknowledges that the lease of the Premises pursuant to this Amendment shall be on an "as-is" basis without further obligation on Landlord's part as to improvements whatsoever, except that Landlord hereby agrees to complete the Tenant Improvements for the Premises in accordance with the provisions of Exhibit X, Work Letter, attached hereto.

M. **Brokers**. Article 18 of the Lease is amended to provide that the parties recognize the following parties as the brokers who negotiated this Amendment, and agree that Landlord shall be responsible for payment of brokerage commissions to such brokers pursuant to its separate agreements with such brokers: Irvine Realty Company ("*Landlord's Broker*") is the agent of Landlord exclusively and Newmark Grubb Knight Frank / Newport Beach ("*Tenant's Broker*") is the agent of Tenant exclusively. By the execution of this Amendment, each of Landlord and Tenant hereby acknowledge and confirm (a) receipt of a copy of a Disclosure Regarding Real Estate Agency Relationship conforming to the requirements of California Civil Code 2079.16, and (b) the agency relationships specified herein, which acknowledgement and confirmation is expressly made for the benefit of Tenant's Broker. If there is no Tenant's Broker so identified herein, then such acknowledgement and confirmation is expressly made for the benefit of Landlord's Broker. By the execution of this Amendment, Landlord and Tenant are executing the confirmation of the agency relationships set forth herein. The warranty and indemnity provisions of Article 18 of the Lease, as amended hereby, shall be binding and enforceable in connection with the negotiation of this Amendment.



# IV. GENERAL.

A. **Effect of Amendment**. The Lease shall remain in full force and effect and unmodified except to the extent that it is modified by this Amendment.

B. **Entire Agreement**. This Amendment embodies the entire understanding between Landlord and Tenant with respect to the modifications set forth in "*III. MODIFICATIONS*" above and can be changed only by a writing signed by Landlord and Tenant.

C. **Defined Terms**. All words commencing with initial capital letters in this Amendment and defined in the Lease shall have the same meaning in this Amendment as in the Lease, unless they are otherwise defined in this Amendment.

D. **Corporate and Partnership Authority**. If Tenant is a corporation or partnership, or is comprised of either or both of them, each individual executing this Amendment for the corporation or partnership represents that he or she is duly authorized to execute and deliver this Amendment on behalf of the corporation or partnership and that this Amendment is binding upon the corporation or partnership in accordance with its terms.

E. **Counterparts; Digital Signatures**. If this Amendment is executed in counterparts, each is hereby declared to be an original; all, however, shall constitute but one and the same amendment. In any action or proceeding, any photographic, photostatic, or other copy of this Amendment may be introduced into evidence without foundation. The parties agree to accept a digital image (including but not limited to an image in the form of a PDF, JPEG, GIF file, or other e-signature) of this Amendment, if applicable, reflecting the execution of one or both of the parties, as a true and correct original.

California Certified Access Specialist Inspection. Pursuant to California Civil Code § 1938, Landlord hereby F. states that the Premises have not undergone inspection by a Certified Access Specialist (CASp) (defined in California Civil Code § 55.52(a)(3)). Pursuant to Section 1938 of the California Civil Code, Landlord hereby provides the following notification to Tenant: "A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction related accessibility standards within the premises." If Tenant requests to perform a CASp inspection of the Premises, Tenant shall, at its cost, retain a CASp approved by Landlord (provided that Landlord may designate the CASp, at Landlord's option) to perform the inspection of the Premises at a time agreed upon by the parties. Tenant shall provide Landlord with a copy of any report or certificate issued by the CASp (the "CASp Report") and Tenant shall, at its cost, promptly complete any modifications necessary to correct violations of construction related accessibility standards identified in the CASp Report, notwithstanding anything to the contrary in this Lease. Tenant agrees to keep the information in the CASp Report confidential except as necessary for the Tenant to complete such modifications.

# V. EXECUTION.

# LANDLORD:

**DISCOVERY BUSINESS CENTER LLC,** a Delaware limited liability company

By: /s/ Steven M. Case Printed Name: Steven M. Case Title: EVP Office Properties

By: /s/ Holly McManus Printed Name: Holly McManus Title: Vice President, OperationsOffice Properties TENANT: GHOST MANAGEMENT GROUP, LLC, a Delaware limited liability company

By: /s/ **Doug Francis** Printed Name: Doug Francis Title: CEO

By: /s/ Chris Beals Printed Name: Chris Beals Title: President & General Counsel

Ехнівіт А

# 43 DISCOVERY PREMISES

43 Discovery

2nd Floor

Ехнівіт А-1

FIRST RIGHT SPACE

43 Discovery

**First Floor** 

# Ехнівіт Х

# WORK LETTER

#### **DOLLAR ALLOWANCE**

The tenant improvement work to be contracted for by Landlord hereunder ("*Tenant Improvement Work*") shall consist of the design and construction of all tenant improvements ("*Tenant Improvements*"), including work in place as of the date hereof, required for the Premises pursuant to the approved final Working Drawings and Specifications (as hereinafter defined). All of the Tenant Improvement Work shall be performed by a contractor selected by Landlord and in accordance with the procedures and requirements set forth below. It is understood that the Tenant Improvement Work may be performed in one or two phases (i.e., one for each of the 41 Discovery Premises and the 43 Discovery Premises), and in such event the provisions of this Work Letter shall apply to each such phase, but there shall be only one Landlord's Maximum Contribution and Moving Allowance as set forth in Section II.B below.

# I. ARCHITECTURAL AND CONSTRUCTION PROCEDURES.

Tenant and Landlord have approved, or shall approve within the time period set forth below, both (i) a detailed A. space plan for the Premises, prepared by Landlord's architect (to be selected by Tenant from among LPA, Inc., Gensler, SAA, or H. Hendy Associates), which includes interior partitions, ceilings, interior finishes, interior doors, suite entrance, floor coverings, window coverings, lighting, electrical and telephone outlets, plumbing connections, heavy floor loads and other special requirements of Tenant ("*Preliminary Plan*"), and (ii) an estimate of the cost to complete the Tenant Improvements in accordance with the Preliminary Plan ("Preliminary Cost Estimate"), which Preliminary Cost Estimate will be based upon estimated costs provided by Landlord's contractor. To the extent applicable, the Preliminary Plan shall include Landlord's building standard tenant improvements, materials and specifications for the Project as set forth in Schedule I attached hereto ("Building Standard *Improvements*"), except for changes and additions specifically requested by Tenant and approved by Landlord in writing (any such addition or variation from the Standard Improvements shall be referred to herein as a ("Non-Standard Improvement"). All Tenant requests for Non-Standard Improvements must be in writing and shall provide sufficient specifications and details for Landlord to reasonably evaluate impacts upon the Preliminary Plan and Preliminary Cost Estimate. Tenant shall approve or provide suggested revisions to the Preliminary Plan and the Preliminary Cost Estimate by signing copies of each or specifying in reasonable detail any suggested revisions thereto as hereinabove provided, and delivering each to Landlord within 5 business days of receipt thereof by Tenant. Landlord shall attempt in good faith to modify the Preliminary Plan and the Preliminary Cost Estimate to incorporate Tenant's suggested revisions in a mutually satisfactory manner. In all events, Tenant shall approve in all respects a Preliminary Plan and a Preliminary Cost Estimate (as either may have been modified as herein provided to incorporate Tenant's suggested revisions) for the 43 Discovery Premises not later than May 30, 2018 ("Plan Approval Date"), it being understood that Tenant's failure to do so shall constitute a "Tenant Delay" (as defined below).

**B.** On or before the Plan Approval Date, Tenant shall provide in writing to Landlord or Landlord's architect all specifications and information requested by Landlord for the 43 Discovery Premises for the preparation of final construction documents and costing, including without limitation Tenant's final selection of paint and floor finishes, complete specifications and locations (including electrical, load and HVAC requirements) of Tenant's equipment, and details of all Non-Standard Improvements (as defined above) which have been approved by Landlord as part of the Preliminary Plan (collectively, "*Programming Information*"). Tenant's failure to provide the Programming Information for the 43 Discovery Premises by the Plan Approval Date shall constitute a Tenant Delay for purposes hereof. Tenant understands that final construction documents for the Tenant Improvements shall be predicated on the Programming Information, and accordingly that such information must be accurate and complete and that any defects or problems due to incomplete or inaccurate Programming Information shall be the responsibility of the Tenant and that the Landlord shall have no obligation or liability for such defects or problems arising from any incomplete or inaccurate Programming Information. Any delays in the completion of the Tenant Improvements due to incomplete or inaccurate Programming Information shall constitute a Tenant Delay.

C. Upon Tenant's approval of the Preliminary Plan and Preliminary Cost Estimate and delivery of the complete Programming Information, Landlord's architect and engineers shall prepare and deliver to Tenant working drawings and specifications ("Working Drawings and Specifications"). Landlord shall submit the Working Drawings and Specifications for the Tenant Improvement Work to a competitive bidding process involving at least 3 licensed and reputable general contractors. If requested by Tenant, Landlord shall provide copies of the bid responses to Tenant. After adjustments for any inconsistent assumptions to reflect an "apples to apples" comparison, Landlord shall select the lowest qualified bidder for construction of the Tenant Improvements. In the event Landlord selects other than the lowest bidder, it shall do so based on commercially reasonable factors which it shall demonstrate to Tenant. Upon selection of the bidder, Landlord shall enter into a construction contract with the contractor so selected in the bid amount for construction of the Tenant Improvements (which bid amount shall serve as the "Final Cost Estimate"). Tenant shall have 5 business days from the receipt thereof to approve or disapprove the Working Drawings and Specifications. Tenant shall not unreasonably withhold or delay its approval, and any disapproval or requested modification shall be limited to items not consistent with the approved Preliminary Plan or Preliminary Cost Estimate. Should Tenant disapprove the Working Drawings and Specifications, such disapproval shall be accompanied by specific reasons for disapproval and a detailed list of requested revisions. Any revision requested by Tenant and accepted by Landlord, shall be incorporated into a revised set of Working Drawings and Specifications and the Final Cost Estimate, and Tenant shall approve same in writing within 5 business days of receipt without further revision.

**D.** In the event that Tenant requests in writing a revision to the Working Drawings and Specifications ("*Change*"), and Landlord so approves such Change as provided in the Section next below, Landlord shall advise Tenant by written change order as soon as is practical of any increase in the Completion Cost such Change would cause. Tenant shall approve or disapprove such change order, if any, in writing within 2 business days following Tenant's receipt of such change order. If Tenant approves any such change order, and such change order causes Final Cost Estimate to exceed (or further exceed) the Landlord's Maximum Contribution, Landlord, at its election, may either (i) require as a condition to the effectiveness of such change order that Tenant pay the increase in the Completion Cost attributable to such change order concurrently with delivery of Tenant's approval of the change order, or (ii) defer Tenant's payment of such increase until the date 10 business days after delivery of invoices for same. If Tenant disapproves any such change order, Tenant shall nonetheless be responsible for the reasonable architectural and/or planning fees incurred in preparing such change order. Landlord shall have no obligation to interrupt or modify the Tenant Improvement Work pending Tenant's approval of a change order, but if Tenant fails to timely approve a change order, Landlord may (but shall not be required to) suspend the applicable Tenant Improvement Work, in which event any related critical path delays because of such suspension shall constitute Tenant Delays hereunder.

**E.** Landlord agrees that it shall not unreasonably withhold its consent to Tenant's requested Changes, provided that such consent may be withheld in all events if the requested Change (i) is of a lesser quality than the Tenant Improvements previously approved by Landlord, (ii) fails to conform to applicable governmental requirements, (iii) would result in the Premises requiring building services beyond the level Landlord has agreed to provide Tenant under the Lease, (iv) would delay construction of the Tenant Improvements and Tenant declines to accept such delay in writing as a Tenant Delay, (v) interferes in any manner with the proper functioning of, or Landlord's access to, any mechanical, electrical, plumbing or HVAC systems, facilities or equipment in or serving the Building, or (vi) would have an adverse aesthetic impact to the Premises or would cause additional expenses to Landlord in reletting the Premises.

Notwithstanding any provision in the Lease to the contrary, and not by way of limitation of any other rights or F. remedies of Landlord, if Tenant fails to comply with any of the time periods specified in this Work Letter, fails otherwise to approve or reasonably disapprove any submittal within the time period specified herein for such response (or if no time period is so specified, within 5 business days following Tenant's receipt thereof), fails to approve in writing both the Preliminary Plan and Preliminary Cost Estimate for the Tenant Improvements for the 43 Discovery Premises by the Plan Approval Date, fails to provide all of the Programming Information for the 43 Discovery Premises requested by Landlord by the Plan Approval Date, fails to approve in writing the Working Drawings and Specifications or the Final Cost Estimate within the time provided herein, fails to timely deliver the Tenant's Contribution as required hereunder, requests any Changes, furnishes inaccurate or erroneous Programming Information, specifications or other information, or otherwise delays in any manner the completion of the Tenant Improvements (including without limitation by specifying materials that are not readily available) or the issuance of an occupancy certificate (any of the foregoing being referred to in this Lease as a "Tenant Delay"), then Tenant shall bear any resulting additional construction cost or other expenses, and, to the extent the delay relates to the Tenant Improvements for the 43 Discovery Premises, the Commencement Date for the 43 Discovery Premises shall be deemed to have occurred for all purposes, including without limitation Tenant's obligation to pay rent for the 43 Discovery Premises, as of the date Landlord reasonably determines that it would have been able to deliver the 43 Discovery Premises to Tenant but for the collective Tenant Delays. Should Landlord determine that the Commencement Date for the 43 Discovery Premises should be advanced in accordance with the foregoing, it shall so notify Tenant in writing. Landlord's determination shall be conclusive unless Tenant notifies Landlord in writing, within 5 business days thereafter of Tenant's election to contest same pursuant to Section 14.7 of the Lease. Pending the outcome of such proceedings, Tenant shall make timely payment of all rent due under this Lease based upon the Commencement Date for the 43 Discovery Premises set forth in the aforesaid notice from Landlord.

**G.** All of the Tenant Improvements shall become the property of Landlord and shall be surrendered with the Premises at the expiration or sooner termination of this Lease, except that Landlord shall have the right, by notice to Tenant given at the time of Landlord's approval of the Preliminary Plan, the Working Drawings and Specifications and any Change, to require Tenant either to remove all or any of the Tenant Improvements approved in the Preliminary Plan or in the Working Drawings and Specifications or by way of such Change, to repair any damage to the Premises or the Common Areas arising from such removal, and to replace any Non-Standard Improvements so approved with the applicable Standard Improvement, or to reimburse Landlord for the reasonable cost of such removal, repair and replacement upon demand. Any such removals, repairs and replacements by Tenant shall be completed by the Expiration Date or sooner termination of the Lease.

X-2

H. Landlord shall permit Tenant and its agents to enter the 43 Discovery Premises at least 30 days prior to the Commencement Date for the 43 Discovery Premises in order that Tenant may install its cabling and related communication equipment through Tenant's own contractors prior to the Commencement Date for the 43 Discovery Premises. Any such work shall be subject to Landlord's prior written approval, and shall be performed in a manner and upon terms and conditions and at times satisfactory to Landlord's representative. The foregoing license to enter the 43 Discovery Premises prior to the Commencement Date for the 43 Discovery Premises is, however, conditioned upon Tenant's contractors and their subcontractors and employees working in harmony and not interfering with the work being performed by Landlord. If at any time Landlord determines that such entry shall cause disharmony or interfere with the work being performed by Landlord, this license may be withdrawn by Landlord upon 24-hours written notice to Tenant. That license is further conditioned upon the compliance by Tenant's contractors with all requirements imposed by Landlord on third party contractors, including without limitation the maintenance by Tenant and its contractors and subcontractors of workers' compensation and public liability and property damage insurance in amounts and with companies and on forms satisfactory to Landlord, with certificates of such insurance being furnished to Landlord prior to proceeding with any such entry. The entry shall be deemed to be under all of the provisions of the Lease except as to the covenants to pay rent. Landlord shall not be liable in any way for any injury, loss or damage which may occur to any such work being performed by Tenant, the same being solely at Tenant's risk. In no event shall the failure of Tenant's contractors to complete any work in the 43 Discovery Premises extend the Commencement Date for the 43 Discovery Premises.

I. Tenant hereby designates Dale Foster ("*Tenant's Construction Representative*"), Telephone No. (949) 486-7456, Email: dfoster@weedmaps.com, as its representative, agent and attorney-in-fact for all matters related to the Tenant Improvement Work, including but not by way of limitation, for purposes of receiving notices, approving submittals and issuing requests for Changes, and Landlord shall be entitled to rely upon authorizations and directives of such person(s) as if given directly by Tenant. The foregoing authorization is intended to provide assurance to Landlord that it may rely upon the directives and decision making of the Tenant's Construction Representative with respect to the Tenant Improvement Work and is not intended to limit or reduce Landlord's right to reasonably rely upon any decisions or directives given by other officers or representatives of Tenant. Any notices or submittals to, or requests of, Tenant related to this Work Letter and/or the Tenant Improvement Work may be sent to Tenant's Construction Representative at the email address above provided. Tenant may amend the designation of its Tenant's Construction Representative(s) at any time upon delivery of written notice to Landlord.

#### II. COST OF TENANT IMPROVEMENTS

**A.** Landlord shall complete, or cause to be completed, the Tenant Improvements, at the construction cost shown in the Final Cost Estimate (subject to increases for Landlord approved Changes and as otherwise provided in this Work Letter), in accordance with final Working Drawings and Specifications approved by both Landlord and Tenant.

**B.** Landlord shall pay up to \$2,493,576.00, based on \$48.00 per rentable square foot of the 43 Discovery Premises and \$22.00 per rentable square foot of the 41 Discovery Premises ("*Landlord's Maximum Contribution*"), of the final "*Completion Cost*" (as defined below). Tenant acknowledges that the Landlord's Maximum Contribution is intended only as the maximum amount Landlord will pay toward approved Tenant Improvements, and not by way of limitation, any partitions, modular office stations, fixtures, cabling, furniture and equipment requested by Tenant are in no event subject to payment as part of Landlord's Maximum Contribution except as provided below. In the event the Completion Cost of the Tenant Improvement Work is less than the Landlord's Maximum Contribution, Landlord's actual contribution toward the Completion Cost ("*Landlord's Contribution*") shall equal such lesser amount, and Tenant shall have no right to receive any credit, refund or allowance of any kind for any unused portion of the Landlord's Maximum Contribution. Notwithstanding the foregoing, Tenant may utilize a portion of the Landlord's Contribution, not to exceed the amount of \$228,681.00, based on \$3.00 per rentable square foot of the Premises (the "*Moving Allowance*"), towards Tenant's cost of cabling, furniture, fixtures and equipment, or related moving expenses for Tenant's move to the 43 Discovery Premises. It is further understood and agreed that the Tenant Improvements shall be substantially completed not later than June 30, 2019 to be eligible for funding by Landlord, and that Landlord shall not be obligated to fund any portion of the Landlord's Contribution towards the Tenant Improvements commenced after such date.

X-3

**C.** Tenant shall pay any costs due to inaccurate or incomplete Programming Information and the amount, if any, by which aggregate Completion Cost for the Tenant Improvement Work exceeds the Landlord's Maximum Contribution. The amounts to be paid by Tenant for the Tenant Improvements pursuant to this Section II.C are sometimes cumulatively referred to herein as the "*Tenant's Contribution*".

**D.** The "*Completion Cost*" shall mean all costs of Landlord in completing the Tenant Improvements in accordance with the approved Working Drawings and Specifications and with any approved Changes thereto, including but not limited to the following costs: (i) payments made to architects, engineers, contractors, subcontractors and other third party consultants in the performance of the work, (ii) permit fees and other sums paid to governmental agencies, and (iii) costs of all materials incorporated into the work or used in connection with the work. The Completion Cost shall also include a construction management fee to be paid to Landlord or to Landlord's management agent in the amount of three percent (3%) of the Completion Cost. Unless expressly authorized in writing by Landlord, and except for a fee to be paid to Tenant's construction manager not to exceed 2% of the Completion Cost, the Completion Cost shall not include (and no portion of the Landlord's Contribution shall be paid for) any costs incurred by Tenant, including without limitation, any costs for space planners, managers, advisors or consultants retained by Tenant in connection with the Tenant Improvements.

**E.** Prior to start of construction of the Tenant Improvements, Tenant shall pay to Landlord in full the amount of the Tenant's Contribution set forth in the approved Preliminary Cost Estimate or in the Final Cost Estimate (once approved by Tenant). If the actual Completion Cost of the Tenant Improvements is greater than the Final Cost Estimate because of Changes, modifications or extras not reflected on the approved Working Drawings and Specifications, or because of Tenant Delays, then Tenant shall pay all such additional costs within 10 business days after written demand for same. The balance of any sums not otherwise paid by Tenant shall be due and payable on or before the Commencement Date for the 43 Discovery Premises. If Tenant defaults in the payment of any sums due under this Work Letter, Landlord shall (in addition to all other remedies) have the same rights as in the case of Tenant's failure to pay rent under the Lease, including, without limitation, the right to terminate this Lease and recover damages from Tenant and/or to charge a late payment fee and to collect interest on delinquent payments, and Landlord may (but shall not be required to) suspend the Tenant Improvement Work following such default, in which event any delays because of such suspension shall constitute Tenant Delays hereunder.

X-4

# **TENANT IMPROVEMENT / INTERIOR CONSTRUCTION OUTLINE SPECIFICATIONS**

# SCHEDULE I

#### (By Tenant/Tenant Allowance)

**Note** During preliminary walk throughs, construction management is to confirm re-use of existing building components and provide direction to: 1) match existing, or 2) provide new building standard at all remodel conditions; or 3) provide upgrade to building standard based on project team input. Each suite to be reviewed on a case-by-case basis.

TENANT STANDARD GENERAL OFFICE: **CARPET** Direct glue broadloom carpet.

#### VINYL COMPOSITION TILE (VCT)

12" x 12" VCT Armstrong Standard Excelon.

#### WALLS

Standard Walls: 5/8" gypsum drywall on 2-1/2" x 25 ga. metal studs 16" o.c., floor to ceiling construction. No walls shall penetrate the grid unless required by code. All walls shall be straight, and parallel to building exterior walls. All offices and rooms shall be constructed of a standard size and tangent to a building shell or core wall.

**Exterior Walls (First Generation Only):** 5/8" gypsum drywall furring on 25 ga. metal studs, with R-13 insulation.

#### PAINT

Paint finish, one standard color to be Benjamin Moore AC-40, Glacier White, flat finish. Dark colors subject to Landlord approval.

#### BASE

2-1/2" Burke rubber base; straight at cut pile carpet, coved at resilient flooring and loop carpet.

#### **RUBBER TRANSITION STRIP**

Transition strip between carpet and resilient flooring to be Burke #150, color: to match adjacent V.C.T.

#### PLASTIC LAMINATE

Plastic laminate color at millwork: Nevamar "Smoky White", Textured #S-7-27T.

#### CEILING

2x4 USG Radar Illusions #2842 scored ceiling tile, installed in building standard 9/16" or 15/16" T-bar grid. Continuous grid throughout.

#### LIGHTING

All spaces are to be illuminated with building standard 2 x 4 direct/indirect fixtures, approved by the Landlord.

#### DOORS

1-3/4" solid core, 3"-0" x 8'-10" plain sliced white oak, Western Integrated clear anodized aluminum frames, Schlage "D" series "Sparta" latchset hardware, dull chrome finish.

#### **OFFICE SIDELITES**

All interior offices to have sidelite glazing adjacent to office entry door, 4' wide x door height, Western Integrated clear anodized aluminum frame integral to door frame with clear tempered glass.

#### WINDOW COVERINGS

Vertical blinds: Mariak Industries PVC blinds at building perimeter windows, Model M-3000, Color: Light Grey.

# TENANT IMPROVEMENT / INTERIOR CONSTRUCTION OUTLINE SPECIFICATIONS (Continued)

TENANT STANDARD MECHANICAL:	<b>HVAC</b> <i>General</i> : Exterior corner spaces with more than one exposure shall be provided with a separate zone. Conference Room (or Training Room) 20' x 13' or larger shall be provided with a separate zone. Exterior zone shall be limited to a single exposure and a maximum of 750 to 1000 square feet.
	<i>Campus Office Building</i> : Interior and Exterior zone VAV boxes shall be connected to the main supply air loop. Exterior zone VAV boxes shall be provided with two-row hot water reheat coil. Interior zone shall be limited to a maximum of 2000 square feet.
	Air distribution downstream of VAV boxes shall be provided complete with ductwork, 2'x2' perforated face ceiling diffusers, 2'x2' perforated return air grilles and air balance. All ductwork shall be sheet metal constructed per SMACNA standards and insulated per the latest Title 24 requirements.
	Pneumatic thermostats with blank white cover shall be provided for each zone. Thermostats shall be located adjacent to light switch at 48" above finished floor. When the building utilizes DDC zone control, DDC system shall be Andover and installed by AAS. DDC system shall be interfaced to the existing Irvine Company network.
	<i>Mid-Tech / Manufacturing Building</i> : Air distribution downstream of packaged rooftop units and/or split system fan coil units shall be provided complete with ductwork, 2'x2' perforated face ceiling diffusers, 2'x2' perforated return air grilles and air balance. All ductwork shall be sheet metal constructed per SMACNA standards and insulated per the latest Title 24 requirements. Interior zone shall be limited to a maximum of 2500 square feet.
	Packaged rooftop units and/or split system units shall be connected to existing Irvine Company Energy Management System. Thermostats shall be located adjacent to light switch at 48" above finished floor. EMS shall be Andover and installed by AAS.
	New packaged rooftop units larger than 5-ton shall be provided with seismic isolation curb with minimum 1- inch spring deflection. New packaged rooftop units larger than 6.25 ton shall be provided with economizer with barometric relief damper.
TENANT STANDARD FIRE PROTECTION:	<b>FIRE PROTECTION</b> Pendant satin chrome plated, recessed heads, adjustable canopies, minimum K factor to be 5.62, located at center of 2' x 2' section of scored ceiling tile. Ceiling drops from shell supply loop.
TENANT STANDARD FIRE SPRINKLER:	FIRE SPRINKLER - Hard pipe to be used. Any substitutions to be submitted for Landlord review and approval prior to install. - Center sprinkler head in 2x2 ceiling tile.

Schedule I-2

#### TENANT IMPROVEMENT / INTERIOR CONSTRUCTION OUTLINE SPECIFICATIONS (Continued)

TENANT STANDARD
ELECTRICAL:

#### ELECTRICAL SYSTEM

A 277/480 volt, three phase, four wire tenant metered distribution section will be added to main service at Main Electrical Room.

Tenant Electrical Room, located within the lease space, as directed by the Landlord, to include 277/480 volt and 120/208 volt panels, transformer, lighting control panel, as required. All newly installed panels and distribution boards shall have all branch circuit loads appropriately disaggregated per 2013 Title 24 requirements.

Standard tenant electrical capacity will be provided in the following capacity:

- Lighting 277V: Minimum of 1.2 watt watts per s.f.
- General 277V Power: As required to accommodate tenant loads.
- HVAC Power 277/480V: As required to accommodate the HVAC equipment.
- General 120/208V Power: Minimum of 8.0 watts per s.f.

#### LIGHTING

All spaces are to be illuminated with building standard 2' x 4', direct/indirect fixtures based on one (1) fixture per 96 square feet. All lighting in newly renovated areas (and associated existing areas with renovations mandated by 2013 Title 24 requirements) are to be illuminated with building standard 2'x4' direct/indirect LED 0-10V dimmable fixtures based on (1) fixture per 96 square feet.

Fixture to be Focal Point TICLED-24-4000L-35 (FLUL-24-PS-4000L-35K-1C-VOLT-LD1-GRID TYPE-EQ-WH) - All Fixtures should be ordered via Southern California Illumination, contact rep at 949-622-3000.

Any substitutions to these fixtures must be reviewed/approved by the Landlord.

All lighting in newly renovated areas (and associated existing areas with renovations mandated by 2013 Title 24 requirements) are to be controlled by 2013 Title 24 compliant digital lighting system, complete with room controller capable of full range 0-10V LED dimming, occupancy sensors, daylight sensors (as required), and low voltage digital switches as required for each respective enclosed space. Locate switches at 48" to switch centerline. Digital control system shall be by Greengate or equal by Wattstopper. Projects in excess of 10,000 square feet shall also have demand responsive controls via input / output interface at each room controller location with applicable low voltage conductors routed to tenant electrical room for future connection to demand response system per 2013 Title 24 requirements.

Exit signs: Internally illuminated, white sign face with green text.

#### OUTLETS

Power: Leviton "*Decora*" style 15 / 20 amp 125-volt specification grade white duplex receptacle mounted vertically, 18" AFF to centerline, with a white plastic coverplate.

2013 Title 24 controlled receptacles are to be plug load controllable decorator receptacle, 15A, half control, white in color Legrand #26252CHW. Receptacle relay shall be wired to room controller in respective vicinity or enclosed space for controlled receptacle to shut off during periods of vacancy.

Schedule I-3

#### TENANT IMPROVEMENT / INTERIOR CONSTRUCTION OUTLINE SPECIFICATIONS (Continued)

# **TENANT STANDARD**ELECTRICAL (CONTINUED):All furniture systems will be assumed to be a four (4) circuit / eight<br/>(8) wire configuration. All furniture system workstations are assumed to have personal computers only and<br/>will be connected at a ratio of eight (8) workstations per four (4) circuit / eight (8) wire homerun. For each<br/>four circuit homerun, the two "general" circuits shall be controlled circuits per 2013 Title 24 requirements<br/>and shall be controlled by relays connected to the room controller in respective vicinity or enclosed space for<br/>controlled receptacles in partitions to shut off during periods of vacancy.All wall mounted furniture system communication feeds will be provided with (2) 1 1/2" conduit (non-fire<br/>rated / non-insulated walls) OR (2) 1-1/4" conduit (fire rated / insulated walls); a 4S/DP box and a double-<br/>gang mud ring in the wall. One (1) furniture system communication feeds will be assumed to be capable of

providing enough cabling capacity for eight (8) workstations.

Power and Telecom Feeds to systems furniture by Tenant to be via walls, furred columns or ceiling J-box.

All wall mounted general communication outlets in non-fire rated / non-insulated walls will be provided a 2gang mud ring and a pull string in the wall. All wall mounted communication outlets in fire-rated and insulated walls will be provided with 3/4" conduit (voice and / or data only) OR a 1" conduit (combination voice / data), stubbed into the accessible ceiling space, 4S/DP box and a single gang mud ring in the wall. Cover plate, jacks and cables by tenant.

A single tenant telecom room will be provided with a single 4' x 8' backboard. An empty 2" conduit will be routed from this backboard to the building's main telephone backboard. An empty 4" conduit sleeve will be stubbed into the accessible ceiling space.

TENANT STANDARD WAREHOUSE/SHIPPING AND RECEIVING (IF APPLICABLE):

#### FLOORS

Sealed concrete.

#### WALLS

5/8" gypsum wallboard standard partition, height and construction subject to Landlord approval. At furred walls, paint to match Benjamin Moore AC-40 Glacier White. Provide rated partition at occupancy separation, as required by code.

# CEILING

Exposed structure, non-painted.

#### WINDOWS

None.

#### ACCESS

7'-6" H x 7'-6" W glazed service doors. Glazing is bronze reflective glass.

#### HVAC

None.

#### PLUMBING

Single accommodation restroom, if required.

Sheet vinyl flooring to be Armstrong Classic Corlon "Seagate" #86526 Oyster, with Smooth White FRP panel wainscot to 48" high. Painted walls and ceiling to be Benjamin Moore AC-40 Glacier White, semigloss finish.

#### **LIGHTING** T5 High Bay, 2 x 4 fixtures.

#### **OTHER ELECTRICAL**

Convenience outlets; surface mounted at exposed concrete walls.

# SECURITY

Lockable doors.

Schedule I-4

# FOURTH AMENDMENT TO LEASE

#### I. PARTIES AND DATE.

This Fourth Amendment to Lease ("*Amendment*") dated May 3, 2018, is by and between DISCOVERY BUSINESS CENTER LLC, a Delaware limited liability company ("*Landlord*"), and GHOST MANAGEMENT GROUP, LLC, a Delaware limited liability company ("*Tenant*").

#### II. RECITALS.

On November 11, 2013, Landlord and Tenant entered into a lease for space in a building located at 41 Discovery, Irvine, California ("41 Discovery Premises" and "41 Discovery Building"), which lease was amended by a First Amendment to Lease dated January 27, 2016 ("First Amendment"), wherein Tenant added approximately 4,182 rentable square feet of space in a building located at 15420 Laguna Canyon Road, Suite 260, Irvine, California ("Laguna Canyon Premises" and "Laguna Canyon Building") to the Premises, by a Second Amendment to Lease dated April 7, 2017, wherein Tenant added approximately 7,000 rentable square feet of space in a building located at 49 Discovery, Suite 200, Irvine, California {"49 Discovery Premises") to the Premises, and by a Third Amendment to Lease dated December 29, 2017 ("Third Amendment"), wherein Tenant terminated its leasing of the Laguna Canyon Premises and the 49 Discovery Premises in exchange for leasing approximately 31,407 rentable square feet of space in a building located at 43 Discovery, Suite 200, Irvine, California, and herein referred to as the "43 Discovery Premises". The foregoing lease, as so amended, is hereinafter referred to as the "Lease".

Landlord and Tenant each desire to make such modifications as are set forth in "Ill. MODIFICATIONS" next below.

# III. MODIFICATIONS.

A. Work Letter Modifications. Landlord and Tenant desire that the architectural design services for the Tenant Improvements for the 43 Discovery Premises be performed by an architect selected and retained by Tenant (not by Landlord). Accordingly, Sections I.A., I.B., and I.C. of the Work Letter attached to the Third Amendment are hereby deleted in their entirety and the following substituted therefor.

# "I. ARCHITECTURAL AND CONSTRUCTION PROCEDURES.

A. Tenant shall engage Design Blitz for the preparation of a detailed space plan for the Premises which includes interior partitions, ceilings, interior finishes, interior office doors, suite entrance, floor coverings, window coverings, lighting, electrical and telephone outlets, plumbing connections, heavy floor loads and other special requirements ("*Preliminary Plan*"). Upon completion of the Preliminary Plan, Landlord and Tenant shall approve an estimate of the cost to complete the Tenant Improvements in accordance with the Preliminary Plan ("*Preliminary Cost Estimate*"), which Preliminary Cost Estimate will be based upon estimated costs provided by Landlord's contractor. To the extent applicable, the Preliminary Plan shall include Landlord's building standard tenant improvements, materials and specifications for the Project as set forth in Schedule I attached hereto ("*Standard Improvements*"), except for changes and additions specifically requested by Tenant and approved by Landlord in writing (any such addition or variation from the Standard Improvements shall be referred to herein as a "*Non-Standard Improvement*"). In all events, Tenant shall approve in all respects a Preliminary Plan and a Preliminary Cost Estimate for the 43 Discovery Premises not later than May 31, 2018 ("*Plan Approval Date*"), it being understood that Tenant's failure to do so shall constitute a "*Tenant Delay*" (as defined below).

- B. Not later than July 31, 2018 (the "Working Drawings Delivery Date"), Tenant shall (i) cause its architect to prepare final construction drawings and specifications for the Tenant Improvements ("Working Drawings and Specifications") and (ii) approve in writing the Working Drawings and Specifications and deliver same to Landlord for review and approval. It is understood that Tenant's architect shall utilize Landlord's mechanical, electrical, and plumbing engineer (i.e., TK1SC), and fire/life safety and fire sprinkler engineers for the Project. Tenant shall cause Landlord to be provided with CAD files of the final as-built construction drawings that are compatible with Landlord's CAD/Revit standards, which standards shall be provided to Tenant or its architect upon request; should Tenant fail to deliver same by the substantial completion of the Tenant Improvement work, then Landlord may cause such CAD files to be prepared by its architect, the cost of which shall be funded by Tenant to Landlord within 10 days after invoice therefor. The cost incurred by Tenant in preparing the Preliminary Plan and the Working Drawings and Specifications (but only to the extent the amount thereof is furnished to Landlord prior to the receipt of the contractors' bids as described in Section I.C. below), together with the reasonable cost not to exceed \$7,500.00 of Landlord's architect to review same and the accompanying CAD files, shall be funded by Landlord from the "Landlord's Contribution" set forth below upon delivery of invoices and payment authorization by Tenant. Tenant's failure to deliver to Landlord the Tenant-approved Working Drawings and Specifications for the 43 Discovery Premises by the Working Drawings Delivery Date shall constitute a Tenant Delay for purposes hereof.
- C. Upon approval of the Working Drawings and Specifications, Landlord shall submit the Working Drawings and Specifications for the Tenant Improvement Work to a competitive bidding process involving at least 3 licensed and reputable general contractors. If requested by Tenant, Landlord shall provide copies of the bid responses to Tenant. After adjustments for any inconsistent assumptions to reflect an "apples to apples" comparison, Landlord shall select the lowest qualified bidder for construction of the Tenant Improvements. In the event Landlord selects other than the lowest bidder, it shall do so based on commercially reasonable factors which it shall demonstrate to Tenant. Upon selection of the bidder, Landlord shall enter into a construction contract with the contractor so selected in the bid amount for construction of the Tenant Improvements (which bid amount shall serve as the "*Final Cost Estimate*")."

In addition, Section I.F of the Work Letter attached to the Third Amendment is hereby deleted in its entirety and the following substituted therefor:

"F. Notwithstanding any provision in the Lease to the contrary, and not by way of limitation of any other rights or remedies of Landlord, if Tenant fails to comply with any of the time periods specified in this Work Letter, fails otherwise to approve or reasonably disapprove any submittal within the time period specified herein for such response (or if no time period is so specified, within 5 business days following Tenant's receipt thereof), fails to approve in writing both the Preliminary Plan and Preliminary Cost Estimate for the Tenant Improvements for the 43 Discovery Premises by the Plan Approval Date, fails to deliver to Landlord for Landlord's approval the Working Drawings and Specifications by the Working Drawings Delivery Date, fails to timely deliver the Tenant's Contribution as required hereunder, requests any Changes, furnishes inaccurate or erroneous specifications or other information, or otherwise delays in any manner the completion of the Tenant Improvements (including without limitation by specifying materials that are not readily available) or the issuance of an occupancy certificate (any of the foregoing being referred to in this Lease as a "Tenant Delay"), then Tenant shall bear any resulting additional construction cost or other expenses, and, to the extent the delay relates to the Tenant Improvements for the 43 Discovery Premises, the Commencement Date for the 43 Discovery Premises ,shall be deemed to have occurred for all purposes, including without limitation Tenant's obligation to pay rent for the 43 Discovery Premises, as of the date Landlord reasonably determines that it would have been able to deliver the 43 Discovery Premises to Tenant but for the collective Tenant Delays. Should Landlord determine that the Commencement Date for the 43 Discovery Premises should be advanced in accordance with the foregoing, it shall so notify Tenant in writing. Landlord's determination shall be conclusive unless Tenant notifies Landlord in writing, within 5 business days thereafter of Tenant's election to contest same pursuant to Section 14.7 of the Lease. Pending the outcome of such proceedings, Tenant shall make timely payment of all rent due under this Lease based upon the Commencement Date for the 43 Discovery Premises set forth in the aforesaid notice from Landlord."

Further in addition, the last sentence of Section II.D of the Work Letter attached to the Third Amendment is hereby deleted in its entirety and the following substituted therefor:

"Unless expressly authorized in writing by Landlord, and except for (i) a fee to be paid to Tenant's construction manager not to exceed 2% of the Completion Cost, and (ii) the fees to be paid to Tenant's architect, the Completion Cost shall not include (and no portion of the Landlord's Contribution shall be paid for) any costs incurred by Tenant, including without limitation, any costs for managers, advisors or consultants retained by Tenant in connection with the Tenant Improvements."

IV. GENERAL.

**A. Effect of Amendment**. The Lease shall remain in full force and effect and unmodified except to the extent that it is modified by this Amendment.

**B.** Entire Agreement. This Amendment embodies the entire understanding between Landlord and Tenant with respect to the modifications set forth in "Ill. MODIFICATIONS" above and can be changed only by a writing signed by Landlord and Tenant.

**C. Defined Terms.** All words commencing with initial capital letters in this Amendment and defined in the Lease shall have the same meaning in this Amendment as in the Lease, unless they are otherwise defined in this Amendment.

**D. Corporate and Partnership Authority**. If Tenant is a corporation or partnership, or is comprised of either or both of them, each individual executing this Amendment for the corporation or partnership represents that he or she is duly authorized to execute and deliver this Amendment on behalf of the corporation or partnership and that this Amendment is binding upon the corporation or partnership in accordance with its terms.

**E. Counterparts- Digital Signatures.** If this Amendment is executed in counterparts, each is hereby declared to be an original; all, however, shall constitute but one and the same amendment. In any action or proceeding, any photographic, photostatic, or other copy of this Amendment may be introduced into evidence without foundation. The parties agree to accept a digital image (including but not limited to an image in the form of a PDF, JPEG, GIF file, or other e-signature) of this Amendment, if applicable, reflecting the execution of one or both of the parties, as a true and correct original.

California Certified Access Specialist Inspection. Pursuant to California Civil Code § 1938, Landlord hereby F. states that the Premises have not undergone inspection by a Certified Access Specialist (CASp) (defined in California Civil Code § 55.52(a)(3)) Pursuant to Section 1938 of the California Civil Code, Landlord hereby provides the following notification to Tenant: "A Certified Access Specialist (CASp) can inspect the subject premises and determine whether the subject premises comply with all of the applicable construction-related accessibility standards under state law. Although state law does not require a CASp inspection of the subject premises, the commercial property owner or lessor may not prohibit the lessee or tenant from obtaining a CASp inspection of the subject premises for the occupancy or potential occupancy of the lessee or tenant, if requested by the lessee or tenant. The parties shall mutually agree on the arrangements for the time and manner of the CASp inspection, the payment of the fee for the CASp inspection, and the cost of making any repairs necessary to correct violations of construction related accessibility standards within the premises." If Tenant requests to perform a CASp inspection of the Premises, Tenant shall, at its cost, retain a CASp approved by Landlord (provided that Landlord may designate the CASp, at Landlord's option) to perform the inspection of the Premises at a time agreed upon by the parties. Tenant shall provide Landlord with a copy of any report or certificate issued by the CASp (the "CASp Report") and Tenant shall, at its cost, promptly complete any modifications necessary to correct violations of construction related accessibility standards identified in the CASp Report, notwithstanding anything to the contrary in this Lease. Tenant agrees to keep the information in the CASp Report confidential except as necessary for the Tenant to complete such modifications.

<sup>3</sup> 

# V. EXECUTION.

Landlord and Tenant executed this Amendment on the date as set forth in "I. PARTIES AND DATE." above.

#### LANDLORD:

**DISCOVERY BUSINESS CENTER LLC**, a Delaware limited liability company

By: /s/ Steven M. Case Name: Steven M. Case Title: EVP Office Properties

By: <u>/s/ Holly McManus</u> Name: Holly McManus Title: VP, Operations Office Properties

#### TENANT:

**GHOST MANAGEMENT GROUP, LLC**, a Delaware limited liability company

By: /s/ Chris Beals Name: Chris Beals Title: President

By: /s/ Doug Francis Name: Doug Francis Title: CEO

# CONFIDENTIAL

# STRATEGIC ADVISOR AGREEMENT

THIS STRATEGIC ADVISOR AGREEMENT ("*Agreement*") is made and entered into between Ghost Management Group, LLC, a Delaware limited liability company (the "*Company*"), on the one hand, and Steven Jung, an individual ("*Employee*"), on the other hand. Each of Company and Employee may be referred to individually as a "*Party*" and collectively as the "*Parties*."

# RECITALS

- A. Employee is an at-will employee of the Company;
- **B.** As part of this Agreement, Employee and Company have agreed that June 30, 2021 will be Employee's last day of active employment in his current role as President and Chief Operating Officer of the Company, WM Holding Company, LLC, and WM Technology, Inc. and he will remain in a paid employee advisory role from July 1, 2021 through July 6, 2022 (the "*Advisory Services Period*").
- **C.** The Company and Employee wish to effectuate a release of any and all claims that Employee holds against the Company and to resolve all controversies and disputes as hereinafter set forth.

NOW, THEREFORE, in consideration of the Recitals, covenants and representations contained in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which is acknowledged, the Parties agree as follows:

# 1. Advisory Services Period & Termination of Employment.

Advisory Services Period: Beginning on the first day of the Advisory Services Period and extending (a) through last day of the Advisory Services Period (the "Services Completion Date"), Employee will be relieved of all of his responsibilities and duties except that he will continue to provide advisory services as an employee to the Company as requested in writing by the Company's CEO. The Parties acknowledge and agree that during the Advisory Services Period, Employee is free to take on additional outside work or employment, including in a project, consulting or full-time role; *provided*, that any outside work is accepted and provided (x) in Employee's individual capacity and not as a representative of the Company, and (y) subject to the Company's Confidential Information, Non-Solicitation and Inventions Assignment Agreement signed by Employee as a condition of his employment with the Company, as well as Employee's obligations pursuant to this Agreement, including the confidentiality obligations in Section 9. As consideration for the advisory services provided during the Advisory Services Period and the Release of Claims in Section 3, and subject to the terms of this Agreement, (i) Employee will continue to be paid his regular salary on the Company's normal payroll dates during the Advisory Services Period ")(less standard deductions and withholdings)(the "Advisory Services Compensation"), (ii) the Class P Units of WM Holding Company, LLC, a Delaware limited liability company and parent company of the Company ("WM Holding") held by Employee pursuant to the Equity Award Agreements (defined below) will continue to vest on their normal schedule during the Advisory Services Period (the "Advisory Services Units"), and (iii) his health insurance and 401k benefits will be maintained by the Company (the "Insurance Benefit"). Except as stated in this Section 1(a), Employee will not be eligible for any other wages, payments, bonuses, incentives, equity vesting, or benefits (including sick leave or accrual of vacation time) during the Advisory Services Period. In the event Employee breaches any provision of this Agreement, the Company will immediately cease providing the payments and benefits stated in this Section 1(a), and process Employee's separation of employment and Employee will not be entitled to any further payments or benefits under this Agreement. Employee acknowledges and agrees that Employee will promptly notify the Company if he acquires alternative health insurance coverage at any point during the Advisory Services Period, in which case, the Company will cease providing health insurance benefits to Employee. Notwithstanding anything to the contrary contained herein, (i) the Advisory Services Compensation will not exceed an aggregate amount of \$609,863 and (ii) a maximum of 278,897 Class P Units will vest as part of the Advisory Services Units.

(b) <u>Termination of Employment</u>: Employee's employment with the Company in any and all capacities will be terminated effective as of the Services Completion Date. Employee acknowledges that Employee has received all wages, benefits and other amounts due up to and including the first day of the Advisory Services Period, including without limitation any and all wages, compensation, vacation, sick leave, overtime, commissions, options, bonuses, profit sharing, benefits, insurance, or any other form of payment. Employee acknowledges that payment any portion of Employee's wages, including vacation pay, was not conditioned on Employee signing this Agreement.

(c) <u>Alternative Insurance Benefit</u>: If the Company determines at any point during the Advisory Services Period that Employee may not remain covered by the Company's employer-provided health insurance policies then Employee will no longer be entitled to participate in any Company healthcare benefit plan, and the Company shall provide Employee notice regarding Employee's right to continue Employee's healthcare coverage pursuant to COBRA. If and for so long as Employee enrolls in COBRA, the Company will pay directly Employee's COBRA premium for health insurance for the remainder of the Advisory Services Period (the "*Alternative Insurance Benefit*"). Company will not enroll Employee in COBRA and Employee will be fully responsible for completing any and all paperwork necessary for COBRA enrollment. Employee acknowledges and agrees that Employee will promptly notify the Company if he acquires alternative health insurance coverage at any point during the Advisory Services Period, in which case, the Company will cease providing the Alternative Insurance Benefit to Employee.

2. **Return of All Company Materials**. Employee hereby certifies that Employee has returned to Company <u>all</u> Company records, documents, electronically stored information, and tangible embodiments of such, prepared by Employee or coming into Employee's possession by virtue of Employee's employment with Company. To the extent Employee is unable to return such materials that are in a digital form, Employee shall remove all copies of such digital material and certify the destruction or deletion of such digital copies. Employee further certifies that Employee has returned to Company all property of Company, including but not limited to pagers, keys, key cards, cellular phones, credit cards, personal and laptop computers, and other electronic equipment. Employee agrees that should Employee at any future time discover additional items of property belonging to Company, Employee will promptly return such property to Company.

General Release. Except as to such rights or claims as may be created by this Agreement, Employee and 3. Employee's respective heirs, administrators, successors in interest, assigns and agents, hereby release and forever discharge the Company and its current and former officers, directors, shareholders, employees, representatives, attorneys, agents, members, trustees, administrators, owners, partners, insurers, fiduciaries, subsidiaries, parent companies, affiliates, related entities, assigns, predecessors and successors in interest, jointly and severally (referred to collectively hereafter as the "Releasees"), from any and all claims, demands, liabilities, suits, causes of action, charges, complaints, obligations, costs, losses, damages, injuries, penalties, interest, attorneys' fees, and other legal responsibilities, of any form whatsoever, whether known or unknown, unforeseen, unanticipated, unsuspected or latent (referred to collectively hereafter as "Claim" or "Claims"), which Employee has at any time owned or held up to and including the date Employee signs this Agreement (including during the Revocation Period (as defined below)), including, and without limiting the generality of the foregoing, any and all Claims arising out of, connected with, or relating to: (1) Employee's employment with the Company or the termination of that employment; (2) any act or omission by or on the part of any of the Releasees; (3) any alleged violation of California Labor Code, applicable California Wage Order, Fair Labor Standards Act or the Consolidated Omnibus Budget Reconciliation Act (COBRA), as amended; (4) any Claim arising under the Labor Code Private Attorneys General Act (PAGA), Labor Code §§ 2699, et seq.; (5) any federal, state or local law regulating compensation, salaries, wages, meal periods, rest periods, itemized wage statements, pay stubs or payroll records, hours, bonuses, commissions, overtime, benefits, monies, pay, allowances, benefits, sick pay, severance pay, retention pay or benefits, paid leave benefits, vacation pay, penalties, interest or damages; (6) any Claim for violation of any federal, state or local law or regulation prohibiting discrimination, harassment or retaliation of any kind; (7) breach of any express or implied employment contract or agreement, wrongful discharge, breach of the implied covenant of good faith and fair dealing, intentional or negligent infliction of emotional distress, fraud, misrepresentation, defamation, trespass, conversion, interference with prospective economic advantage, and invasion of privacy; and (8) any Claim for attorneys' fees, costs or expenses. The foregoing general release does not apply to any Claim that cannot be released as a matter of law. Nothing in this Agreement prohibits or prevents Employee from filing a charge with or participating, testifying, or assisting in any investigation, hearing, or other proceeding before any federal, state, or local government agency. However, to the maximum extent permitted by law, Employee agrees that if such an administrative claim is made, Employee shall not be entitled to recover any individual monetary relief or other individual remedies.

Employee agrees that the release provided by this Agreement applies to any Claims brought by any person or agency on behalf of Employee or any class action, representative action or collective action pursuant to which Employee may have any right or benefit. Employee agrees not to participate in any class action, representative action or collective action that may include or encompass any of the Claims released by this Agreement. With respect to any Claims released by this Agreement, Employee further agrees not to accept any recovery or benefit that may be obtained on Employee's behalf by any other person or agency or in any class action, representative action, and does hereby assign any such recovery or benefit to the Company. In addition, Employee agrees that in the event Employee receives any notice from any claims administrator, attorney, law firm or other person purporting to represent Employee or any of the Releasees that encompasses any of the Claims released by this Agreement, Employee will promptly notify such claims administrator, attorney, law firm, or other person that Employee will promptly notify such claims administrator, attorney, law firm, or other person that Employee does not wish to participate in and specifically "opts out" of any class action, representative action or collective action against the Company or any of the Releasees. Finally, by signing this Agreement, Employee acknowledges and agrees that Employee is not an "aggrieved employee," as that term is defined in the Labor Code Private Attorneys General Act (PAGA), with respect to any Claims released by this Agreement.

Notwithstanding anything to the contrary herein, the foregoing release shall not cover, and Employee does not release, any rights of Employee under this Agreement or the Equity Award Agreements, dated as of June 26, 2017, May 10, 2018 and May 7, 2020, respectively (collectively, as the same may be amended and/or restated from time to time, the "Equity Award Agreements"), between Employee and WM Holding, or any rights of Employee as a holder of Class P Units of WM Holding under the Certificate of Formation or the Fourth Amended and Restated Operating Agreement of WM Holding, dated as of June 16, 2021 (as the same may be amended and/or restated from time to time, the "Amended and Restated LLC Agreement"). Employee acknowledges and agrees that Employee will own an aggregate total of 1,035,514 vested Class P Units of WM Holding as of the first day of the Advisory Services Period, and, if Employee accepts and executes this Agreement, WM Holding will permit continued vesting of the Advisory Services Units during the Advisory Services Period, so that Employee will have vested in an aggregate total of 1,314,411 vested Class P Units as of the Services Completion Date, all 1,314,411 of which shall be retained by Employee following the Services Completion Date pursuant and subject to the Equity Award Agreements, the Amended and Restated LLC Agreement and the WM Holding Third Amended and Restated Equity Incentive Plan, dated as of August 15, 2018 as may be amended, restated, supplemented, or otherwise modified from time to time (the "Plan"). Except for the Advisory Services Units, Employee acknowledges that Employee forfeits all Class P Units granted to Employee under the Equity Award Agreement and which are unvested as of the Services Completion Date pursuant to the terms of the Equity Award Agreement, the Amended and Restated LLC Agreement and the Plan. Except as set forth in this paragraph, Employee acknowledges that Employee has no right, title or interest to any other equity, options or equivalents of Company or its affiliates.

4. **Release of All Unknown Claims.** The general release above is intended to be a full and final release covering all unsuspected, unknown, undisclosed and unanticipated Claims which may have arisen, or may arise, from any act or omission up to and including the date Employee signs this Agreement, including during the Revocation Period, and which arise out of or are related, directly or indirectly, to the dealings between the Parties or any matters described in this Agreement. Employee, and on behalf of anyone or any entity claiming through Employee, each waive any and all rights or benefits which they may now have, or in the future may have, under the terms of Section 1542 of the California Civil Code, which provides as follows:

# A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY.

Notwithstanding the provision of Section 1542, and for the purpose of implementing a full and complete release and discharge, Employee expressly acknowledges that the general release above is intended to include and does include in its effect, without limitation, all Claims which Employee does not know or suspect to exist in Employee's favor against any of the Releasees up to and including the date Employee signs this Agreement and that the release in this Agreement expressly contemplates the extinguishment of all such Claims.

5. **No Admissions**. Neither this Agreement nor the furnishing of the consideration for this Agreement shall be deemed or construed as an admission of liability or wrongdoing on the part of either Party, nor shall they be admissible as evidence in any proceeding other than for the enforcement of this Agreement.

6. **Covenant Not to Sue**. Employee has not and will not directly or indirectly institute any legal action against the Releasees based upon, arising out of, or relating to any Claims released herein. Employee has not and will not directly or indirectly encourage and/or solicit any third party to institute any legal action against the Releasees.

7. **Intellectual Property of the Company.** As additional consideration for the execution and delivery of this Agreement and only to the extent not already obligated to do so with respect to the Company and without limitation to any other agreement Employee has with the Company with respect to the following matters:

(a) Employee agrees to disclose in writing to the Company all source code, passwords, inventions, products, designs, drawings, notes, information, documentation, improvements, works of authorship, processes, techniques, know-how, technical specifications, hardware, computer programs, databases, user interfaces, encoding techniques, and other materials or innovations of any kind that Employee has made, conceived, developed or reduced to practice, alone or jointly with others, in connection with Employee's term of service to the Company prior to the Services Completion Date or that resulted from or that related thereto, whether or not they are eligible for patent, copyright, mask work, trade secret, trademark or other legal protection ("Innovations").

(b) Employee and the Company agree that, to the fullest extent legally possible, all Innovations (as defined in Section 7(a)) will be "works for hire" as that term is defined in Section 101 of the 1976 Copyright Act, and shall be owned exclusively by the Company. Employee agrees that, regardless of whether the Innovations are legally works for hire, all Innovations will be the sole and exclusive property of the Company. Employee hereby irrevocably transfers and assigns to the Company, and agrees to irrevocably transfer and assign to the Company, all of Employee's right, title and interest in and to the Innovations, including all worldwide patent rights (including patent applications and disclosures), copyright rights, mask work rights, trade secret rights, know-how, and any and all other intellectual property or proprietary rights therein (collectively, "*Intellectual Property Rights*"). At the Company in all respects and will execute documents (including, without limitation, assignments as to any Intellectual Property Rights which become the property of the Company pursuant to this Section 7) and, subject to the reasonable availability of Employee, will give testimony and take such further acts reasonably requested by the Company to enable the Company to acquire, transfer, maintain, perfect and enforce its Intellectual Property Rights and other legal protections for the Innovations. Employee hereby appoints the Company as attorney-in-fact to execute documents on behalf of Employee for this limited purpose.

(c) Employee hereby irrevocably transfers and assigns to the Company, and agrees to irrevocably transfer and assign to the Company, and waives and agrees never to assert, any and all Moral Rights (as defined below) that Employee may have in or with respect to any Innovation, during and after the term of this Agreement. "*Moral Rights*" mean any rights to claim authorship of an Innovation (as defined in Section 7(a)), to object to or prevent the modification or destruction of any Innovation, to withdraw from circulation or control the publication or distribution of any Innovation, and any similar right, existing under judicial or statutory law of any country in the world, or under any treaty, regardless of whether or not such right is called or generally referred to as a "moral right."

(d) To the extent that Employee owns or controls any patent rights, copyright rights, mask work rights, trade secret rights, or any other intellectual property or proprietary rights that block or interfere with the rights assigned to the Company under this Agreement (collectively, "*Related Rights*"), Employee hereby grants or will cause to be granted to the Company a non-exclusive, royalty-free, irrevocable, worldwide license to make, have made, use, offer to sell, sell, import, copy, modify, create derivative works based upon, distribute, sublicense, display, perform and transmit any products, software, hardware, methods or materials of any kind that are covered by such Related Rights, to the extent necessary to enable the Company to exercise all of the rights assigned to the Company under this Agreement.

8. **Cooperation**. Employee agrees that, during the period, if any, between the Effective Date and the Services Completion Date, he will act in support of the Company and its business interests and assist with the smooth transition of his responsibilities. Employee agrees to cooperate with the Company and make themselves generally available in connection with disputes between the Company and third parties. This cooperation may include, but is not limited to, conferring with and assisting the Company in preparatory work in litigation matters, providing factual information to the Company, and giving depositions and testimony in judicial and administrative proceedings. Employee agrees that Employee will not be paid by the Company for Employee's cooperation. Employee further agrees to make themselves available on a reasonable basis with respect to any outstanding legal, business or strategic matters as necessary following the Effective Date.

9. **Confidentiality**. Except as otherwise prohibited by law, Employee agrees that neither Employee nor any of Employee's agents or representatives will disclose, disseminate and/or publicize, or cause or permit to be disclosed, disseminated or publicized, any attorney client privileged information, attorney work product, confidential information or trade secrets of the Company, or any claims or allegations which Employee could have made or asserted against the Company, to any person, corporation, association or governmental agency or other entity except: (a) to the extent necessary to report income to appropriate taxing authorities: (b) to members of Employee's immediate family: (c) in response to an order of a court of competent jurisdiction or subpoena issued under the authority thereof; or (d) in response to any inquiry or subpoena issued by a state or federal governmental agency; provided, however, that notice of receipt of such judicial order or subpoena shall be immediately communicated by Employee to the Company telephonically, and confirmed immediately thereafter in writing, so that the Company will have the opportunity to assert what rights it has to non-disclosure prior to Employee's response to the order, inquiry or subpoena. Should Employee disclose to Employee's immediate family members any of the terms this Agreement, Employee will advise them that they are under the same obligation of confidentiality as described in this Section. Employee also agrees not to disclose any confidential or proprietary information pertaining to the business of the Company and/or its customers. For the avoidance of doubt, the confidentiality obligations of this Section shall extend to any information Employee has with respect to the businesses or activities of the Company Entities and any prospective opportunities which such equity holders have discussed or considered pursuing as well as any information received from any actual or potential customers of the Company.

Any violation of the confidentiality provision contained in this Agreement by Employee shall be considered a material breach of this Agreement.

Non-Disparagement. Employee agrees that on a permanent basis Employee will not make any negative, critical, 10. disparaging, detrimental or derogatory comments (whether verbally or in writing) to anyone, including any employee or third party concerning the Company, or any of its current and former officers, directors, shareholders, employees, representatives, attorneys and agents, as well as its predecessors, parents, subsidiaries, affiliates, divisions, and successors-in-interest. With respect to the Company, the foregoing includes, but is not limited to, the Company's services, products, processes, policies, practices, standards of business conduct, and areas or techniques of research and/or development and the actual or planned activities, the plans, processes or business practices of the equity holders of the Company or any Company Entities. The first sentence of this Section 10 shall not apply to any communication or disclosure to the extent required to institute any proceedings to enforce the terms of this Agreement or as otherwise expressly authorized and required by law or lawful process. Company will respond to any inquiries about Employee's employment by providing only Employee's dates of employment, job title, and, if authorized by Employee in writing, Employee's last rate of pay. Employee will direct all such inquiries only to Company's People Operations Department. Employee agrees that Employee's failure to comply with this non-disparagement provision shall be deemed a material breach of this Agreement and shall entitle Company to immediately (a) stop providing the Advisory Services Period benefits set forth in Sections 1(a) and 1(c) including the Advisory Services Compensation, Insurance Benefit, and, if applicable, the Alternative Insurance Benefit, (b) seek recoupment of any amounts already paid to Employee, and (c) cancel the vesting of the Advisory Services Units, in addition to recovering damages for breach.

11. **Right to an Attorney; Time to Consider**. This Agreement constitutes a knowing and voluntary waiver of any and all rights or claims that Employee has or may have under the federal Age Discrimination in Employment Act (ADEA), as amended by the Older Workers' Benefit Protection Act of 1990, 29 U.S.C. §§ 621 *et seq*. This Section and this Agreement are written in a manner calculated to be understood by Employee. Employee acknowledges and agrees that Employee has had twenty-one (21) days to consider this Agreement and to consult with counsel, and the Company has advised Employee of Employee's right to do so. To the extent that Employee has taken less than twenty-one (21) days to consider this Agreement, Employee acknowledges that Employee has had sufficient time to consider the Agreement and to consult with counsel and that Employee did not desire additional time. The terms of this Agreement will not become enforceable for seven (7) calendar days following the date of execution of this Agreement (the "*Revocation Period*"), during which time Employee may revoke this Agreement by notifying the undersigned representative of the Company in writing by registered letter. In the event of any such revocation, the Employee acknowledges and agrees that the Equity Award Agreements shall be deemed revoked as well by the Employee.

12. **Cancellation and Repayment of Advisory Services Benefits.** In the event of breach of this Agreement by Employee, including, but not limited to, the confidentiality and non-disparagement provisions during the period that begins on the Effective Date, then in addition to any remedies available to the Company at law or equity, the Company shall be immediately entitled to terminate Employee's employment (if during the Advisory Services Period) and stop all payments and benefits under this Agreement. This means that Company may (i) withhold any and all payments of Insurance Benefits or Alternative Insurance Benefits to Employee or owed to cover COBRA premiums pursuant to Section 1 of this Agreement, (ii) cancel the vesting of the Advisory Services Units, and (iii) cancel payments of the Advisory Services Compensation. Employee agrees to promptly return to Company the full monetary value of the Advisory Services Compensation, Insurance Benefits and Alternative Insurance Benefits that was received by Employee and to indemnify and hold harmless the Releasees for and against any and all costs, losses, or liability whatsoever, including reasonable attorneys' fees, caused by Employee's breach of this Agreement. Furthermore, Employee agrees that the return or obligation to return the amount of the Advisory Services Compensation, Insurance Benefit, and Alternative Insurance Benefits to the Company and the cancellation of the vesting of the Advisory Services Units will not abrogate or affect in any way Employee's full release of any and all Claims against the Releasees. Employee acknowledges and agrees that the provisions contained in this Section 12 are reasonable and necessary for the protection of the Company's legitimate business interests, and further agrees not to challenge the reasonableness of such restrictions.

13. **Section 409A of the Code**. While the tax treatment of the payments provided under this Agreement is not warranted or guaranteed, it is intended that such payments shall either be exempt from, or comply with, the requirements of Section 409A of the Internal Revenue Code of 1986, as amended. This Agreement shall be construed, administered and governed in a manner that effects such intent.

14. **Waiver and Modification**. The failure to enforce any provision of this Agreement shall not be construed to be a waiver of such provision or to affect either the validity of this Agreement or the right of any Party to enforce the Agreement. This Agreement may be modified or amended only by a written agreement executed by Employee and the Company.

15. **Severability and Savings Provision**. In the event that any provision of this Agreement should be held to be void, voidable, or unenforceable, the remaining portions hereof shall remain in full force and effect. Additionally, in the event that any portion of this Agreement is deemed void or unenforceable, the Parties will be excused from performing that portion of the Agreement.

16. **Entire Agreement**. This Agreement constitutes the entire agreement between Employee and the Company and supersedes and cancels all prior agreements, oral or written, if any, between Employee and the Company, except that this Agreement does not alter, modify, or impact any pre-existing confidentiality provisions and restrictive covenants between the Parties, nor does it affect Employee's obligation to comply with those provisions and covenants.

17. **Arbitration**. The Parties hereby agree to submit any claim or dispute arising out of the terms of this Agreement to private and confidential arbitration by a single neutral arbitrator through Judicial Arbitration and Mediation Services, Inc. ("*JAMS*"). This arbitration provision covers all claims that Employee may have against Company, or that Company may have against Employee. The JAMS Streamlined Arbitration Rules & Procedures in effect at the time the claim or dispute is arbitrated will govern the procedure for the arbitration proceedings between the Parties. The arbitration shall take place in Los Angeles, California. The arbitrator in this matter shall not have the power to modify any of the provisions of this Agreement. The decision of the arbitrator shall be final and binding on all Parties to this Agreement, and judgment thereon may be entered in any court having jurisdiction. Company shall initially advance the arbitrator's fee and all costs of services provided by the arbitrator and arbitration organization. However, the arbitrator's fee and all costs of services provided by the arbitrator, as well as reasonable attorneys' fees and recoverable costs, shall be paid as the arbitrator or court awards in accordance with applicable law. The Parties hereby waive any right to a jury trial on any dispute or claim covered by this Agreement.

18. **Governing Law**. This Agreement shall be construed in accordance with, and be governed by the laws of the State of California, both procedural and substantive.

19. **Review Period and Effective Date**. Employee has until June 28, 2021 to review, sign and return this Agreement to the Company. This Agreement is effective as of the first day of the Advisory Services Period (the "*Effective Date*").

# PLEASE READ CAREFULLY. THIS AGREEMENT CONTAINS A RELEASE OF ALL KNOWN AND UNKNOWN CLAIMS. THE UNDERSIGNED AGREE TO THE TERMS OF THIS AGREEMENT AND VOLUNTARILY ENTER INTO IT WITH THE INTENT TO BE BOUND THEREBY.

The undersigned have carefully read this Agreement and understand that it contains a release of known and unknown claims. The undersigned acknowledge and agree to all of the terms and conditions of this Agreement. The undersigned further acknowledge that they are entering into this Agreement voluntarily as of the Effective Date with a full understanding of its terms.

/s/ Steven Jung
Steven Jung
Dated: June 21, 2021
Ghost Management Group, LLC
By: /s/ Christopher Beals
Christopher Beals Its Authorized Signatory
Dated: June 21, 2021

June 21, 2021

Securities and Exchange Commission 100 F Street, N.E. Washington, DC 20549

Commissioners:

We have read the statements made by WM Technology, Inc. (f/k/a Silver Spike Acquisition Corp.), under Item 4.01 of its Form 8-K filed June 21, 2021. We agree with the statements concerning our Firm under Item 4.01, in which we were informed of our dismissal on June 16, 2021, effective immediately. We are not in a position to agree or disagree with other statements of WM Technology, Inc. (f/k/a Silver Spike Acquisition Corp.) contained therein.

Very truly yours,

/s/ Marcum LLP Marcum LLP

# Legal Name

WM Holding Company, LLC Weedmaps Spain, S.L.U. Weedmaps Germany GmbH Ghost Management Group, LLC GMG Holdco, Inc. Weedmaps Media, LLC Discovery Opco, LLC WM Enterprise, LLC WM Marketplace, LLC WM Canada Holdings, Inc. WM Museum, LLC WM Teal, LLC WM Retail, LLC Grow One Software (Canada), Inc.

# Jurisdiction of Organization

Delaware Spain Germany Delaware Delaware Delaware Delaware Delaware British Columbia Delaware Delaware Delaware Delaware British Columbia

# UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

Capitalized terms used but not defined in this Exhibit 99.1 shall have the meanings ascribed to them in the Current Report on Form 8-K to which this Exhibit 99.1 is attached. Unless the context otherwise requires, the "Company" refers to WM Technology, Inc. (f/k/a Silver Spike Acquisition Corp.) and its subsidiaries after the Closing, and Silver Spike Acquisition Corp. ("Silver Spike") prior to the Closing.

The following unaudited pro forma condensed combined financial information presents the combination of the financial information of Silver Spike and Legacy WMH, adjusted to give effect to the Business Combination and related transactions. The following unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X as amended by the final rule, Release 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses."

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 combines the historical balance sheet of Silver Spike and the historical balance sheet of Legacy WMH on a pro forma basis as if the Business Combination and related transactions, summarized below, had been consummated on March 31, 2021. The unaudited pro forma condensed combined statements of operations for the three months ended March 31, 2021 and year ended December 31, 2019, combine the historical statements of operations of Silver Spike and historical statements of operations of Legacy WMH for such periods on a pro forma basis as if the Business Combination and related transactions, summarized below, had been consummated on January 1, 2020, the beginning of the earliest period presented:

- the Business Combination;
- the Domestication; and
- the issuance and sale of 32,500,000 shares of Class A Common Stock for a purchase price of \$10.00 per share and an aggregate purchase price of \$325.0 million in the PIPE subscription financing pursuant to the Subscription Agreements.

The unaudited pro forma condensed combined financial statements have been developed from and should be read in conjunction with:

- the accompanying notes to the unaudited pro forma condensed combined financial information;
- the historical audited financial statements of Silver Spike as of and for the year ended December 31, 2020 and the related notes, included in the Proxy Statement/Prospectus;
- the historical audited financial statements of Legacy WMH as of and for the year ended December 31, 2020 and the related notes, included in the Proxy Statement/Prospectus;
- the historical unaudited financial statements of Silver Spike as of and for the three months ended March 31, 2021 and the related notes, included in the Proxy Statement/Prospectus;
- the historical unaudited financial statements of Legacy WMH as of and for the three months ended March 31, 2021 and the related notes, included in the Proxy Statement/Prospectus;
- other information relating to Silver Spike and Legacy WMH contained in the Proxy Statement/Prospectus, including the Merger Agreement and the description of certain terms thereof set forth in the section entitled "*The Business Combination*."

Pursuant to Silver Spike's then-existing organizational documents, Silver Spike's public shareholders were offered the opportunity to redeem, upon the Closing, shares of Class A ordinary shares then held by them for cash equal to their pro rata share of the aggregate amount on deposit (as of two business days prior to the Closing) in the Trust Account (as defined in the Proxy Statement/Prospectus). The unaudited condensed combined pro forma financial statements reflect actual redemptions of 10,012 shares of Class A ordinary shares at \$10.17 per share.

Notwithstanding the legal form of the Business Combination pursuant to the Merger Agreement, the Business Combination is accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, Silver Spike is treated as the acquired company and Legacy WMH is treated as the acquirer for financial statement reporting purposes. Legacy WMH has been determined to be the accounting acquirer based on evaluation of the following facts and circumstances:

- Legacy WMH Class A Unit holders, through their ownership of the Class V Common Stock, have the greatest voting interest in the Company with over 50% of the voting interest;
- Legacy WMH's directors represent the majority of the new board of directors of the Company;
- Legacy WMH's senior management is the senior management of the Company; and
- Legacy WMH is the larger entity based on historical operating activity and has the larger employee base.

Assumptions and estimates underlying the unaudited pro forma adjustments set forth in the unaudited pro forma condensed combined financial statements are described in the accompanying notes. The unaudited pro forma condensed combined financial statements have been presented for illustrative purposes only and are not necessarily indicative of the operating results and financial position that would have been achieved had the Business Combination occurred on the dates indicated. Further, the unaudited pro forma condensed combined financial statements do not purport to project the future operating results or financial position of the Company following the completion of the Business Combination. The unaudited pro forma adjustments represent management's estimates based on information available as of the date of these unaudited pro forma condensed combined financial information becomes available and analyses are performed.

# UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF MARCH 31, 2021 (in thousands)

			Transaction Accounting Adjustments (Note 2)		Pro Forma Combined			
Assets								
Cash	\$	19,604	\$	101	254,203	(a)	\$	104,529
					(8,750)	(b)		
					(30,345)	(C)		
					325,000	(d)		
					(102)	(f)		
					(455,182)	(1)		
Accounts receivable, net		7,553			()	(-)		7,553
Prepaid expenses and other current assets		7,344		102	(3,192)	(c)		4,254
Total current assets		34,501		203	81,632	(C)		116,336
		54,501						110,550
Marketable securities held in Trust Account		-		254,203	(254,203)	(a)		-
Property and equipment, net		6,892						6,892
Goodwill		3,961						3,961
Intangible assets, net		4,280						4,280
Right of use assets		42,113						42,113
Deferred tax asset		-			157,104	(j)		157,104
Other assets		3,874						3,874
Total assets	\$	95,621	\$	254,406	\$ (15,467)		\$	334,560
	÷		÷		¢ (10,107)			55 1,500
T :- L :!! .:-								
Liabilities	*	10	<i>c</i>	0 F 15	<b>d</b>		*	10.5
Accounts payable and accrued expenses	\$	13,496	\$	3,549	\$ (3,659)	(c)	\$	13,386
Deferred revenue		6,189						6,189
Operating lease liabilities, current portion		4,884						4,884
Notes payable to members, current portion		205						205
Promissory note - related party		-		200				200
Total current liabilities		24,774		3,749	(3,659)			24,864
Operating lease liabilities, non-current portion		43,558		· ·				43,558
Other long-term liabilities		906						906
Long-term payable under Tax Receivable Agreement		-			133,538	(j)		133,538
Warrant liability		_		141,900	155,550	0)		141,900
Deferred underwriting fee payable		-		8,750	(9.750)	(b)		141,900
		-			(8,750)	(b)		-
Total liabilities		69,238	_	154,399	121,129			344,766
Commitments and contingencies								
Class A ordinary shares subject to possible redemptions				95,006	(95,006)	(e)		-
Stockholders' equity (deficit)/ Members' equity								
Preferred stock								-
Ordinary shares								
Class A				2	1	(e)		-
					-	(f)		
					(3)	(g)		
Class B				1	(1)	(g)		-
Common stock				-	(1)	(5)		
Class A					3	(d)		7
Cidss A								/
Close W					4	(g)		7
Class V					7	(i)		7
Members' units		18,809		400.0	(18,809)	(l)		-
Additional paid in capital				130,948	(26,908)	(c)		(3,337)
					324,997	(d)		
					95,005	(e)		
					(102)	(f)		
					(125,950)	(h)		
					(7)	(i)		
					23,566	(j)		
					13,200	(k)		
					(455,182)	(l)		
					17,096	(1)		
Retained earnings (accumulated deficit)		7,574		(125,950)	125,950	(1) (h)		(1,034)
ואבומוווכע כמווווואָס (מכרטוווטומוכע עכווכוו)		7,374		(120,900)				(1,034)
					(2,970)	(c)		
					(13,200)	(k)		
					7,562	(l)		
Total stockholders' equity (deficit) attributable to common								
shareholders / members' equity	_	26,383		5,001	(35,741)			(4,357)
Noncontrolling interests					(5,849)	(l)		(5,849)
Total stockholders' equity (deficit)/ members' equity		26,383		5,001	(41,590)	~ /		(10,206)
round of the second sec		-0,000	_	5,001	(41,000)		_	(10,200)

Total liabilities and stockholders' equity (deficit)/ members' equity $\$$	95,621	\$ 254,406	\$ (15,467)	\$	334,560
	3				

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE THREE MONTHS ENDED MARCH 31, 2021 (in thousands, except share and per share data)

	WMH (Historical)	Silver Spike (Historical)	Transaction Accounting Adjustments (Note 2)			o Forma ombined
Revenues	\$ 41,154	\$ -			\$	41,154
Operating expenses						
Cost of revenues	1,857	-				1,857
Sales and marketing	9,117	_				9,117
Product development	7,868	-				7,868
General and administrative	13,366	738	(60)	(bb)		13,589
	,		(455)	(cc)		
Depreciation and amortization	1,002	-				1,002
Total operating expenses	33,210	738	(515)			33,433
Income (Loss) from operations	7,944	(738)	515			7,721
Other income (expense):						
Interest earned on marketable securities held in Trust Account	-	30	(30)	(aa)		-
Unrealized gain on marketable securities held in Trust Account	-	-				-
Change in fair value of warrant liability		(78,220)				(78,220)
Interest expense	-	-				-
Other expense, net	28					28
Total other income (expense)	28	(78,190)	(30)			(78,192)
Income (Loss) before provision for income taxes	7,972	(78,928)	485			(70,471)
Provision for income taxes	241	-	(8,422)	(ff)		(8,181)
Net income (loss)	7,731	(78,928)	8,907			(62,290)
Net income (loss) attributable to noncontrolling interests	-	-	(40,372)	(gg)		(40,372)
Net income (loss) attributable to common shareholders	\$ 7,731	\$ (78,928)	\$ 49,279		\$	(21,918)
		<u> </u>			<u> </u>	
Basic and diluted weighted average shares outstanding - Class A					(	63,738,563
Basic and diluted net loss per share - Class A					\$	(0.34)
Sance and analed her roos per share Chass 11					Ψ	(0.01)

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE YEAR ENDED DECEMBER 31, 2020 (in thousands, except share and per share data)

	WMH (Historical)	Silver Spike (Historical)	Transaction Accounting Adjustments (Note 2)			o Forma ombined
Revenues	\$ 161,791	\$ -			\$	161,791
Operating expenses						
Cost of revenues	7,630	-				7,630
Sales and marketing	30,716	-				30,716
Product development	27,142	-				27,142
General and administrative	51,127	3,864	(240)	(bb)		65,380
			(2,571)	(cc)		
			13,200	(dd)		
Depreciation and amortization	3,978					3,978
Total operating expenses	120,593	3,864	10,389			134,846
Income (Loss) from operations	41,198	(3,864)	(10,389)			26,945
Other income (expense):						
Interest earned on marketable securities held in Trust Account	-	2,258	(2,258)	(aa)		-
Unrealized gain on marketable securities held in Trust Account	-	5	(5)	(aa)		-
Change in fair value of warrant liability		(50,420)				(50,420)
Interest expense	(2)	-				(2)
Other expense, net	(2,366)	-	(2,970)	(ee)		(5,336)
Total other income (expense)	(2,368)	(48,157)	(5,233)			(55,758)
Income (Loss) before provision for income taxes	38,830	(52,021)	(15,622)			(28,813)
Provision for income taxes	-	-	(3,443)	(ff)		(3,443)
Net income (loss)	38,830	(52,021)	(12,179)			(25,370)
Net income (loss) attributable to noncontrolling interests	-	-	(16,507)	(gg)		(16,507)
Net income (loss) attributable to common shareholders	\$ 38,830	\$ (52,021)	\$ 4,328		\$	(8,863)
Designed diluted evidence change states diag. Close A						
Basic and diluted weighted average shares outstanding - Class A Basic and diluted net loss per share - Class A					\$	63,738,563
Basic and unuted net loss per share - Class A					Ф	(0.14)

# Notes to Unaudited Pro Forma Condensed Combined Financial Statements

# 1. Basis of Presentation

The Business Combination is accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, Silver Spike is treated as the "acquired" company for financial reporting purposes. Accordingly, for accounting purposes, the Business Combination will be treated as the equivalent of Legacy WMH issuing stock for the net assets of Silver Spike, accompanied by a recapitalization. The net assets of Silver Spike are stated at historical cost, with no goodwill or other intangible assets recorded.

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 gives pro forma effect to the Business Combination as if it had been consummated on March 31, 2021. The unaudited pro forma condensed combined statements of operations for the three months ended March 31, 2021 and year ended December 31, 2020, give pro forma effect to the Business Combination as if it had been consummated on January 1, 2020.

The unaudited pro forma condensed combined balance sheet as of March 31, 2021 has been prepared using, and should be read in conjunction with, the following:

- Silver Spike's unaudited balance sheet as of March 31, 2021 and the related notes, included in the Proxy Statement/Prospectus; and
- Legacy WMH's unaudited balance sheet as of March 31, 2021 and the related notes, included in the Proxy Statement/Prospectus.

The unaudited pro forma condensed combined statement of operations for the three months ended March 31, 2021 has been prepared using, and should be read in conjunction with, the following:

- Silver Spike's unaudited statement of operations for the three months ended March 31, 2021 and the related notes, included in the Proxy Statement/Prospectus; and
- Legacy WMH's unaudited statement of operations for the three months ended March 31, 2021 and the related notes, included in the Proxy Statement/Prospectus.

The unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 has been prepared using, and should be read in conjunction with, the following:

- Silver Spike's audited statement of operations for the year ended December 31, 2020 and the related notes, included in the Proxy Statement/Prospectus; and
- Legacy WMH's audited consolidated statement of operations for the year ended December 31, 2020 and the related notes, included in the Proxy Statement/Prospectus.

Management has made significant estimates and assumptions in its determination of the pro forma adjustments. As the unaudited pro forma condensed combined financial information has been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The pro forma adjustments reflecting the consummation of the Business Combination are based on certain currently available information and certain assumptions and methodologies that management believes are reasonable under the circumstances. The unaudited condensed pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the actual adjustments will differ from the pro forma adjustments and it is possible the difference may be material. Management believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Business Combination based on information available to management at the time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed combined financial information.

The unaudited pro forma condensed combined financial information is not necessarily indicative of what the actual results of operations and financial position would have been had the Business Combination taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of the post-combination company. They should be read in conjunction with the historical financial statements and notes thereto of Silver Spike and Legacy WMH.

# 2. Adjustments to Unaudited Pro Forma Condensed Combined Financial Information

The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Business Combination and has been prepared for informational purposes only.

The following unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses." Release No. 33-10786 replaces the existing pro forma adjustment criteria with simplified requirements to depict the accounting for the transaction (*"Transaction Accounting Adjustments"*) and present the reasonably estimable synergies and other transaction effects that have occurred or are reasonably expected to occur (*"Management's Adjustments"*). The Company has elected not to present Management's Adjustments and will only be presenting Transaction Accounting Adjustments in the following unaudited pro forma condensed combined financial information.

Silver Spike and Legacy WMH have not had any historical relationship prior to the Business Combination. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

# Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet

The Transaction Accounting Adjustments included in the unaudited pro forma condensed combined balance sheet as of March 31, 2021 are as follows:

- a Reflects the reclassification of marketable securities held in the Trust Account that became available following the Business Combination.
- b Reflects the settlement of \$8.75 million in deferred underwriting fees.
- c Represents estimated transaction costs incurred by Silver Spike and Legacy WMH of approximately \$32.9 million for legal, financial advisory and other professional fees incurred in consummating the Business Combination. Of this amount, approximately:
  - \$2.6 million was capitalized within Prepaid expenses and other current assets and paid by Legacy WMH as of March 31, 2021.
  - \$0.6 was capitalized within Prepaid expenses and other current assets and accrued within Accounts payable and accrued expenses by Legacy WMH as of March 31, 2021.
  - \$3.0 million was accrued by Silver Spike in Accounts payable and accrued expenses as of March 31, 2021 and previously recognized in expense.
  - \$30.3 million was reflected as a reduction of cash, which represents the estimated transaction costs of \$32.9 million less the \$2.6 million previously paid by Legacy WMH.
  - \$3.0 million were not capitalized as part of the Business Combination and reflected as a decrease in accumulated deficit. The costs expensed through accumulated deficit are included in the unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 as discussed in Note (ee) below.

- \$26.9 million were capitalized and offset against the proceeds from the Business Combination and PIPE subscription financing and reflected as a decrease in additional paid-in capital. This amount represents total estimated transaction costs less: 1) \$3.0 million previously recognized in expense by Silver Spike and reclassified to additional paid-in capital in Note 2(h) of this unaudited pro forma condensed combined financial information; and 2) \$3.0 million that were not capitalized as part of the Business Combination and reflected as a decrease in accumulated deficit.
- Reflects the proceeds of \$325.0 million from the issuance and sale of 32,500,000 shares of Class A Common Stock at \$10.00 per share in d the PIPE subscription financing pursuant to the terms of the Subscription Agreements.
- Reflects the reclassification of \$95.0 million of Class A ordinary shares subject to possible redemption to permanent equity. e
- Represents share redemptions of 10,012 shares of Class A ordinary shares for \$101,822 allocated to Class A ordinary shares and f additional paid-in capital using par value of \$0.0001 per share and at a redemption price of \$10.17 per share.
- Reflects the conversion of Class A ordinary shares and Class B ordinary shares, on a one-for-one basis, into shares of Class A Common g Stock upon the Domestication.
- h Reflects the elimination of Silver Spike's historical accumulated deficit.

- Reflects the issuance of 65,502,349 shares of Class V Common Stock upon the Closing. The Class V Common Stock, par value \$0.0001, i entitle their holder to one vote per share but not any right to dividends or distributions.
- Reflects pursuant to the terms of the tax receivable agreement (1) the estimated deferred tax asset related to the tax basis step-up on the j exchange of common units for cash in the Business Combination, and (2) the tax receivable agreement liability for amounts payable to post-merger WMH equity holders for tax benefits received by the Company on the step-up. The adjustment to deferred tax asset was calculated based on the estimated tax basis step-up multiplied by an estimated effective tax rate of 27.98%. The adjustment to long-term payable under Tax Receivable Agreement is 85% of the estimated tax benefit, in accordance with the terms of the tax receivable agreement. The remaining difference between the deferred tax asset and tax receivable agreement liability is reflected as additional paidin capital.
- k Represents approximately \$13.2 million of share-based expense associated with vested Legacy WMH Class B units, which expense will be recorded upon the Closing. The cost expensed through accumulated deficit is included in the unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 as discussed in Note (dd) below.
- Represents the pro forma adjustment to: 1) reflect the payment of the cash consideration of \$455.2 million to the Legacy WMH equity 1 holders, and 2) to present the post-merger WMH units as noncontrolling interest upon the reorganization of the post-combination company into an Up-C structure. The noncontrolling interest adjustment reflects the allocation of the post-combination company's total equity (deficit) to the holders of post-merger WMH units approximate 57.3% economic interest in the post-combination company, as follows:

	Equity Ir		tockholders' Noncontr Equity Intere		Noncontrolling S Interest (57.3%)		Common ockholders' Equity (42.7%)
Historical Legacy WMH members' equity	\$	26,383	\$	15,114	\$ 11,269		
Historical Silver Spike total stockholders' equity		5,001		2,865	2,136		
Class A common stock issued in the PIPE subscription financing		325,000		186,186	138,814		
Reclass of redeemable public shares to permanent equity		95,006		54,427	40,579		
Redemption of public shares		(102)		(58)	(44)		
Payment of transaction costs		(29,878)		(17,117)	(12,761)		
Equity adjustment related to deferred tax asset and Tax Receivable Agreement liabilitiy		23,566		13,501	10,065		
Payment of aggregate cash consideration		(455,182)		(260,767)	 (194,415)		
	\$	(10,206)	\$	(5,849)	\$ (4,357)		

The adjustment to noncontrolling interest was recorded with an offset: (1) to eliminate WMH equity holders' units balance of \$18.8 million; (2) to retained earnings for \$7.6 million, representing the portion of the share-based expense in Note (k) allocated to noncontrolling interests; and (3) to additional paid-in capital for the residual amount of \$15.3 million.

# Transaction Accounting Adjustments to Unaudited Pro Forma Condensed Combined Statements of Operations

The Transaction Accounting Adjustments included in the unaudited pro forma condensed combined statements of operations for the three months ended March 31, 2021 and year ended December 31, 2020 as follows:

- aa Represents pro forma adjustment to eliminate interest income and unrealized gains on marketable securities held in the Trust Account.
- bb Represents pro forma adjustment to eliminate historical expenses related to Silver Spike's office space, administrative and support services paid to the sponsor, which terminated upon consummation of the Business Combination.
- cc Reflects pro forma adjustment to eliminate transaction costs expensed by Silver Spike during the three months ended March 31, 2021 and year ended December 31, 2020, which will be capitalized as part of the Business Combination or recognized in expense in Note (ee) below.
- dd Represents approximately \$13.2 million of share-based expense associated with vested Legacy WMH Class B units, which expense is recorded upon the Closing. These costs are reflected as incurred on January 1, 2020, the date the Business Combination occurred for purposes of the unaudited pro forma condensed combined statements of operations. This is a non-recurring item.
- ee Reflects estimated transaction costs allocated to the public and private placement warrant liabilities that are assumed as part of the Business Combination. These costs are reflected as if incurred on January 1, 2020, the date the Business Combination occurred for the purposes of the unaudited pro forma condensed combined statements of operations. This is a non-recurring item.
- ff Represents the pro forma adjustment for income taxes, applying an estimated statutory tax rate of 27.98%.
- gg Represents the pro forma adjustment to allocate net loss to the noncontrolling interests. As the provision for income taxes was all attributable to common shareholders, the allocation of net loss to the noncontrolling interests was determined using Loss before provision for income taxes, as follows:

	]	ee Months Ended ch 31, 2021	Year Ended cember 31, 2020
Loss before provision for income taxes	\$	(70,471)	\$ (28,813)
Economic interest held by noncontrolling interest holders		57.3%	57.3%
Net loss attributable to noncontrolling interests	\$	(40,372)	\$ (16,507)

#### 3. Loss per Share

Represents the net loss per share calculated using the historical weighted average shares outstanding, and the issuance of additional shares in connection with the PIPE subscription financing, assuming the shares were outstanding since January 1, 2020. As the Business Combination is being reflected as if it had occurred at the beginning of the periods presented, the calculation of weighted average shares outstanding for basic and diluted net loss per share assumes that the shares issuable relating to the PIPE subscription financing have been outstanding for the entire periods presented.

	Three Months Ended March 31, 2021	Year Ended December 31, 2020
Pro forma net loss attributable to common shareholders (in thousands)	\$ (21,918)	\$ (8,863)
Pro forma weighted average shares outstanding, basic and diluted - Class A	63,738,563	63,738,563
Pro forma net loss per share, basic and diluted - Class A (1)(2)	\$ (0.34)	\$ (0.14)
Pro forma weighted average shares calculation, basis and diluted - Class A		
Silver Spike public shareholders - Class A	24,988,563	24,988,563
Holders of founder shares - Class A	6,250,000	6,250,000
Subscription investors - Class A	32,500,000	32,500,000
	63,738,563	63,738,563

- (1) Shares of Class V Common Stock will not share in the earnings or losses of the Company and are therefore not participating securities. As such, separate calculations of basic and diluted net loss per share for Class V Common Stock under the two-class method has not been presented.
- (2) For the purpose of calculating diluted net loss per share, it was assumed that all outstanding public and private placement warrants are exchanged for shares of Class A Common Stock. However, since this results in anti-dilution, the effect of such exchange was not included in the calculation of diluted net loss per share.